SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM S-4 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

INTEGRATED ELECTRICAL SERVICES, INC. * (exact name of registrant as specified in its charter)

DEL AWARE (State or other jurisdiction of incorporation or

1731 (Primary Standard Industrial Classification Code Number)

76-0542208 (I.R.S. Employer Identification Number)

JOHN F. WOMBWELL EXECUTIVE VICE PRESIDENT AND GENERAL COUNSEL 1800 WEST LOOP SOUTH, SUITE 500 HOUSTON, TEXAS 77027 (713) 860-1500 (Name, address, including zip code, and telephone number, including area code, of agent for service)

organization)

1800 WEST LOOP SOUTH SUITE 500 HOUSTON, TEXAS 77027 (713) 860-1500 (Address, including zip code, and telephone number, including area code, of Registrant's principal executive offices)

> Copy to: DAVID P. OELMAN, ESQ. VINSON & ELKINS L.L.P. 2300 FIRST CITY TOWER 1001 FANNIN STREET HOUSTON, TEXAS 77002-6760 713-758-3708 713-615-5861 (FAX)

APPROXIMATE DATE OF COMMENCEMENT OF PROPOSED SALE TO THE PUBLIC: As soon as practicable after the effective date of this Registration Statement

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box. [X]

If this Form is filed to registered additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration number of the earlier effective registration statement for the same offering. [X]

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [X]

	CALCULATION OF REGISTRATION FEE			
TITLE OF EACH CLASS OF SECURITIES TO BE REGISTERED	AMOUNT TO BE REGISTERED	PROPOSED MAXIMUM OFFERING PRICE PER NOTE(1)	PROPOSED MAXIMUM AGGREGATE OFFERING PRICE(1)	AMOUNT OF REGISTRATION FEE
9 3/8% Senior Notes Due 2009	\$125,000,000	100%	\$125,000,000	\$31,250

(1) Estimated solely for the purpose of calculating the registration fee in accordance with Rule 457(f) under the Securities Act of 1933.

* Includes certain subsidiaries of Integrated Electrical Services, Inc. identified on the following pages.

UNDER REGISTRATION STATEMENT ON FORM S-4

The following subsidiaries of Integrated Electrical Services, Inc. are co-registrants under this registration statement for the purpose of providing guarantees, if any, of payments on debt securities registered hereunder:

<pre>1st Group Telecommunications, Inc. (f/k/a Bryant Acquisition Corporation)</pre>	Delaware	742930927
Ace Electric, Inc.	Georgia	581233590
Aladdin Ward Electric & Air, Inc.	Florida	592137098
Amber Electric, Inc.	Florida	591888807
Anderson & Wood Construction Co., Inc.	Delaware	742918934
ARC Electric, Incorporated	Delaware	760581695
B. Rice Electric LP	Texas	760619043
Bachofner Electric, Inc.	Delaware	760593514
Bartley & Devary Electric, Inc.	Delaware	742916903
Bear Acquisition Corporation	Delaware	742959621
Bexar Electric Company, Ltd.	Texas	742767532
Brink Electric Construction Co.	South Dakota	460322078
Britt Rice Electric, Inc.	Delaware	760616944
Britt Rice Holdings LLC	Arizona	522216042
Britt Rice Management LLC	Arizona	760618685
Bryant Electric Company, Inc.	North Carolina	561054780
BW Consolidated, Inc.	Nevada	741769791
BW/BEC, Inc.	Texas	742835288
BW/BEC, L.L.C.	Nevada	860873929
Canova Electrical Contracting, Inc.	Delaware	742913069
Carroll Holdings LLC	Arizona	742916337
Carroll Management LLC	Arizona	742916336
Carroll Systems LP	Texas	760601730

PAGE 1 OF 5

Carroll Systems, Inc. (f/k/a Pan American Acquisition Corporation)	Delaware	760597830
Charles P. Bagby Co., Inc.	Alabama	630751092
Collier Electric Company, Inc.	Florida	742923443
Commercial Electrical Contractors, Inc.	Delaware	760587343
Cross State Electric, Inc.	California	953657116
Cypress Electrical Contractors, Inc.	Delaware	721028256
Daniel Electrical Contractors, Inc.	Florida	592622624
Daniel Electrical of Treasure Coast, Inc.	Florida	650548129
Davis Electrical Constructors, Inc.	South Carolina	570474303
Delco Electric, Inc.	Delaware	731563953
DKD Electric Company, Inc.	New Mexico	850245113
Electro-Tech, Inc.	Nevada	880200302
EMC Acquisition Corporation	Delaware	742908723
Ernest P. Breaux Electrical, Inc.	Delaware	742916899
Federal Communications Group, Inc.	Delaware	850461441
Florida Industrial Electric, Inc.	Florida	593508913
General Partner, Inc.	Alabama	631080687
Goss Electric Company, Inc.	Delaware	760581878
H.R. Allen, Inc.	South Carolina	570695117
Hatfield Reynolds Electric Company (f/k/a Hatfield Electric, Inc.)	Arizona	860565738
Haymaker Electric, Ltd.	Alabama	631044169
Holland Electrical Systems, Inc.	Delaware	760576826
Houston Stafford Holdings, LLC	Arizona	522097492
Houston-Stafford Electric, Inc.	Texas	741774028
Houston-Stafford Electrical Contractors, LP	Texas	522095983
Houston-Stafford Management LLC	Arizona	522095981
Howard Brothers Electric Co., Inc.	Delaware	760570227
I.C.G. Electric, Inc.	Delaware	742918936

PAGE 2 OF 5

4

ICS Holdings LLC	Arizona	
ICS Integrated Communication Services LP	Texas	522114914
IES Communications Group, Inc. (f/k/a IES Communications Inc.)	Delaware	760656305
IES Contractors Holdings LLC	Arizona	522131430
IES Contractors LP	Texas	522129299
IES Contractors Management LLC	Arizona	522129827
IES Electrical Group, Inc. (f/k/a Integrated Communication Services, Inc.)	Delaware	522110684
IES Holdings, LLC	Arizona	522097490
IES Management, LP	Texas	760569183
IES Residential Group, Inc.	Delaware	760656307
IES Specialty Lighting, Inc. (f/k/a Modern Acquisition Corporation)	Delaware	731592395
IES Ventures Inc.	Delaware	760656308
Innovative Electric Company, Inc. (f/k/a Thurman & O'Connell Corp.)	Kentucky	611145474
Integrated Electrical Finance, Inc.	Texas	760559059
Integrated Electrical Services, Inc.	Delaware	760542208
Intelligent Building Solutions, Inc.	Delaware	742910189
J.W. Gray Electric Company, Inc.	Delaware	760573295
J.W. Gray Electrical Contractors, LP	Texas	522097983
J.W. Gray Holdings, LLC	Arizona	522097988
J.W. Gray Management, LLC	Arizona	522097977
Kayton Electric, Inc.	Nebraska	470623159
Key Electrical Supply, Inc.	Texas	760285442
Linemen, Inc. (d/b/a California Communications)	Delaware	742912738
Mark Henderson, Incorporated	Delaware	760576830
Menninga Electric, Inc.	Delaware	760575872
Midlands Electrical Contractors, Inc.	Delaware	742918935
Mid-States Electric Company, Inc.	Delaware	621746956
Mills Electric LP	Texas	522095984

PAGE 3 OF 5

Mills Electrical Contractors, Inc.	Texas	751394916
Mills Electrical Holdings, LLC	Arizona	522097491
Mills Management LLC	Arizona	522095982
Mitchell Electric Company, Inc.	Arizona	860141057
M-S Systems, Inc.	Tennessee	621404226
Murray Electrical Contractors, Inc.	Delaware	742913067
Muth Electric, Inc.	South Dakota	460324448
NBH Holding Co., Inc. (f/k/a DKD Acquisition Corporation)	Delaware	850461866
Neal Electric LP	Texas	760657784
Neal Electric Management LLC (f/k/a ICS Management LLC)	Arizona	522114906
New Technology Electrical Contractors, Inc.	Delaware	742918933
Newcomb Electric Company, Inc.	Delaware	760611653
Pan American Electric Company, Inc., a New Mexico	New Mexico	742618624
Pan American Electric, Inc.	Tennessee	620985675
Paulin Electric Company, Inc.	Delaware	610608088
Pollock Electric, Inc.	Texas	760078839
Pollock Summit Electric, LP	Texas	760569180
Pollock Summit Holdings, Inc.	Arizona	522097493
PrimeNet, Inc. (f/k/a Stutts Acquisition Corporation)	Delaware	742902100
Primo Electric Company (f/k/a Hamer Electric Acquisition, Inc.)	Delaware	742902099
Putzel Electrical Contractors, Inc.	Delaware	760604195
Raines Electric Co., Inc.	Delaware	760581935
Raines Electric LP	Texas	522132532
Raines Holdings LLC	Arizona	522132528
Raines Management LLC	Arizona	522132530
RKT Electric, Inc.	Delaware	760585981
Rockwell Electric, Inc.	Delaware	760593890
Rodgers Electric Company, Inc.	Washington	911004905

PAGE 4 OF 5

6

Ron's Electric, Inc.	Delaware	742925506
Spectrol, Inc.	Delaware	760576823
<pre>Spoor Electric, Inc. (d/b/a SEI Electrical Contractor)</pre>	Florida	742899568
Summit Electric of Texas, Incorporated	Texas	760214796
T&H Electrical Corporation	Delaware	760583746
Tech Electric Co., Inc.	Delaware	742912739
Tesla Power (Nevada), Inc.	Nevada	760604875
Tesla Power and Automation, LP	Texas	760592351
Tesla Power G.P., Inc.	Texas	760604876
Tesla Power Properties, LP	Texas	760592352
Thomas Popp & Company	Ohio	311112666
Valentine Electrical, Inc.	Delaware	742916344
Wolfe Electric Co., Inc.	Delaware	742925512
Wright Electrical Contracting, Inc.	Delaware	631203022

PAGE 5 OF 5

THE REGISTRANT HEREBY AMENDS THIS REGISTRATION STATEMENT ON SUCH DATES AS MAY BE NECESSARY TO DELAY ITS EFFECTIVE DATE UNTIL THE REGISTRANT SHALL FILE A FURTHER AMENDMENT WHICH SPECIFICALLY STATES THAT THIS REGISTRATION STATEMENT SHALL THEREAFTER BECOME EFFECTIVE IN ACCORDANCE WITH SECTION 8(a) OF THE SECURITIES ACT OF 1933, AS AMENDED, OR UNTIL THE REGISTRATION STATEMENT SHALL BECOME EFFECTIVE ON SUCH DATE AS THE COMMISSION, ACTING PURSUANT TO SAID SECTION 8(a), MAY DETERMINE.

The information in this prospectus is not complete and may be changed. We may not exchange for these securities until the registration statement filed with the Securities and Exchange Commission is effective. This prospectus is not an offer to sell these securities and is not soliciting an offer to buy these securities in any state where the offer or sale is not permitted.

PROSPECTUS

INTEGRATED ELECTRICAL SERVICES, INC.

OFFER TO EXCHANGE UP TO \$125,000,000 OF 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009

FOR

125,000,000 OF 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009 THAT HAVE BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933

TERMS OF THE EXCHANGE OFFER

- We are offering to exchange up to \$125,000,000 of our outstanding
 9 3/8% Senior Subordinated Notes due 2009 for new notes with substantially identical terms that have been registered under the Securities Act and are freely tradable.
- We will exchange all outstanding notes that you validly tender and do not validly withdraw before the exchange offer expires for an equal principal amount of new notes.
- o The exchange offer expires at 5:00 p.m., New York City time, on _____, 2001, unless extended. We do not currently intend to extend the exchange offer.
- Tenders of outstanding notes may be withdrawn at any time prior to the expiration of the exchange offer.
- The exchange of outstanding notes for new notes will not be a taxable event for U.S. federal income tax purposes.

TERMS OF THE NEW 9 3/8% SENIOR SUBORDINATED NOTES OFFERED IN THE EXCHANGE OFFER

MATURITY

o The new notes will mature on February 1, 2009.

INTEREST

- o Interest on the new notes is payable on February 1 and August 1 of each year, beginning August 1, 2001.
- o Interest will accrue from May 29, 2001.

REDEMPTION

- o We may redeem some or all of the new notes at any time on or after February 1, 2004 at redemption prices listed in "Description of the New Notes--Optional Redemption."
- o We may also redeem up to 35% of the new notes using the proceeds of certain equity offerings completed before February 1, 2004.

CHANGE OF CONTROL

o If we sell assets or experience a change of control, subject to certain conditions, we must offer to purchase the new notes.

RANKING

o The new notes will be subordinated to all existing and future senior indebtedness. The new notes will rank equally with all of our other existing and future senior subordinated indebtedness and will rank senior to all our subordinated indebtedness.

SEE "RISK FACTORS" ON PAGE 6 FOR A DISCUSSION OF FACTORS YOU SHOULD CONSIDER BEFORE PARTICIPATING IN THE EXCHANGE OFFER.

THESE SECURITIES HAVE NOT BEEN APPROVED OR DISAPPROVED BY THE SECURITIES AND EXCHANGE COMMISSION OR ANY STATE SECURITIES COMMISSION NOR HAS THE SECURITIES AND EXCHANGE COMMISSION PASSED UPON THE ACCURACY OR ADEQUACY OF THIS PROSPECTUS. ANY REPRESENTATION TO THE CONTRARY IS A CRIMINAL OFFENSE. This prospectus is part of a registration statement we filed with the Securities and Exchange Commission. In making your investment decision, you should rely only on the information contained in this prospectus and in the accompanying letter of transmittal. We have not authorized anyone to provide you with any other information. If you receive any unauthorized information, you must not rely on it. We are not making an offer to sell these securities in any state where the offer is not permitted. You should not assume that the information contained in this prospectus is accurate as of any date other than the date on the front cover of this prospectus.

TABLE OF CONTENTS

PROSPECTUS SUMMARY	.1
RISK FACTORS	.6
EXCHANGE OFFER	13
USE OF PROCEEDS	24
BUSINESS	25
DESCRIPTION OF THE NEW NOTES	26
FEDERAL INCOME TAX CONSIDERATIONS	58
PLAN OF DISTRIBUTION	61
LEGAL MATTERS	63
EXPERTS	63
WHERE YOU CAN FIND MORE INFORMATION	63
FORWARD-LOOKING STATEMENTS	64
Item 20. Indemnification Of Officers And DirectorsII	
Item 21. Exhibits And Financial Statement SchedulesII	-1
Item 22. UndertakingsII	- 3

i

PROSPECTUS SUMMARY

This summary may not contain all the information that may be important to you. You should read this entire prospectus and the documents to which we have referred you before making an investment decision. You should carefully consider the information set forth under "Risk Factors." In addition, certain statements include forward-looking information which involves risks and uncertainties. See "Forward-Looking Statements." Unless this prospectus otherwise indicates or the context otherwise requires, the terms "we," "our," "us" "IES" or the "Company" as used in this prospectus refer to Integrated Electrical Services, Inc. and its subsidiaries.

THE COMPANY

We are the second largest provider of electrical contracting services in the United States. We are also a growing provider of solutions in the data communications and utilities markets. We provide a broad range of services including designing, building and maintaining electrical, data communications and utilities systems for commercial, industrial and residential customers.

Our executive offices are located at 1800 West Loop South, Suite 500, Houston, Texas 77027, and our telephone number is (713) 860-1500.

THE EXCHANGE OFFER

On May 29, 2001, we completed a private offering of the outstanding notes. We entered into a registration rights agreement with the initial purchasers in the private offering in which we agreed to deliver to you this prospectus and to use our reasonable best efforts to complete the exchange offer within 180 days after the date we issued the outstanding notes.

Exchange Offer	We are offering to exchange new notes for outstanding notes.
Expiration Date	The exchange offer will expire at 5:00 p.m. New York City time, on, 2001, unless we decide to extend it.
Condition to the Exchange Offer	The registration rights agreement does not require us to accept outstanding notes for exchange if the exchange offer or the making of any exchange by a holder of the outstanding notes would violate any applicable law or interpretation of the staff of the Securities and Exchange Commission. A minimum aggregate principal amount of outstanding notes being tendered is not a condition to the exchange offer.
Procedures for Tendering Outstanding Notes	To participate in the exchange offer, you must complete, sign and date the letter of transmittal, or a facsimile of the letter of transmittal, and transmit it together with all other documents required by the letter of transmittal, including the outstanding notes that you wish to exchange, to State Street Bank and Trust Company, as exchange agent, at the address indicated on the cover page of the letter of transmittal. In the alternative, you can tender your outstanding notes by following the procedures for book-entry transfer described in this prospectus. If your outstanding notes are held through The Depository Trust Company and you wish to participate in the exchange offer, you may do so through the automated tender offer program of The Depository Trust Company. If you tender under this program, you will agree to be bound by the letter of transmittal that we are

	providing with this prospectus as though you had signed the letter of transmittal. If a broker, dealer, commercial bank, trust company or other nominee is the registered holder of your outstanding notes, we urge you to contact that person promptly to tender your outstanding notes in the exchange offer. For more information on tendering your outstanding notes, please refer to the sections in this prospectus entitled "Exchange OfferTerms of the Exchange Offer," "Procedures for Tendering" and "Book-Entry Transfer."
Guaranteed Delivery Procedures	If you wish to tender your outstanding notes and you cannot get your required documents to the exchange agent on time, you may tender your outstanding notes according to the guaranteed delivery procedures described in "Exchange OfferGuaranteed Delivery Procedures."
Withdrawal of Tenders	You may withdraw your tender of outstanding notes at any time prior to the expiration date. To withdraw, you must have delivered a written or facsimile transmission notice of withdrawal to the exchange agent at its address indicated on the cover page of the letter of transmittal before 5:00 p.m. New York City time on the expiration date of the exchange offer.
Acceptance of Outstanding Notes and Delivery of New Notes	If you fulfill all conditions required for proper acceptance of outstanding notes, we will accept any and all outstanding notes that you properly tender in the exchange offer on or before 5:00 p.m. New York City time on the expiration date. We will return any outstanding note that we do not accept for exchange to you without expense as promptly as practicable after the expiration date. We will deliver the new notes as promptly as practicable after the expiration date and acceptance of the outstanding notes for exchange. Please refer to the section in this prospectus entitled "Exchange OfferTerms of the Exchange Offer."
Fees and Expenses	We will bear all expenses related to the exchange offer. Please refer to the section in this prospectus entitled "Exchange OfferFees and Expenses."
Use of Proceeds	The issuance of the new notes will not provide us with any new proceeds. We are making this exchange offer solely to satisfy our obligations under our registration rights agreement.
Consequences of Failure to Exchange Outstanding Notes	If you do not exchange your outstanding notes in this exchange offer, you will no longer be able to require us to register the outstanding notes under the Securities Act of 1933 except in the limited circumstances provided under our registration rights agreement. In addition, you will not be able to resell, offer to resell or otherwise transfer the outstanding notes unless we have registered the outstanding notes under the Securities Act of 1933, or unless you resell, offer to resell or otherwise transfer them under an exemption from the registration requirements of, or in a transaction not subject to, the Securities Act of 1933.

U.S. Federal Income Tax Considerations	The exchange of new notes for outstanding notes in the exchange offer should not be a taxable event for U.S. federal income tax purposes. Please read "Federal Income Tax Considerations."
Exchange Agent	We have appointed State Street Bank and Trust Company as exchange agent for the exchange offer. You should direct questions and requests for assistance, requests for additional copies of this prospectus or the letter of transmittal and requests for the notice of guaranteed delivery to the exchange agent addressed as follows: Goodwin Square, 225 Asylum Street, 23rd Floor, Hartford, Connecticut 06103. Eligible institutions may make requests by facsimile at (617) 662-1452.

TERMS OF THE NEW NOTES

The new notes will be identical to the outstanding notes except that the new notes are registered under the Securities Act of 1933 and will not have restrictions on transfer, registration rights or provisions for additional interest and will contain different administrative terms. The new notes will evidence the same debt as the outstanding notes, and the same indenture will govern the new notes and the outstanding notes.

The following summary contains basic information about the new notes and is not intended to be complete. It does not contain all the information that is important to you. For a more complete understanding of the new notes, please refer to the section of this document entitled "Description of the New Notes."

Issuer	Integrated Electrical Services, Inc.
Notes Offered	\$125 million in aggregate principal amount of 93/8% Senior Subordinated Notes due 2009.
Maturity	February 1, 2009.
Interest on the New Notes	Annual Rate 9 3/8%.
Interest Payment Dates	February 1 and August 1 of each year, commencing on August 1, 2001.
Sinking Fund	None.
Optional Redemption	On or after February 1, 2004, we may redeem some or all of the new notes at the redemption prices listed in the "Description of the New NotesOptional Redemption" section of this prospectus, plus accrued but unpaid interest to the date of redemption. Before February 1, 2002, we may redeem up to 35% of the aggregate principal amount of the new notes with the net proceeds of certain sales of equity at the price listed in the section "Description of New Notes" under the heading "Optional Redemption," provided at least 65% of the aggregate principal amount of the new notes originally issued remains outstanding after such redemption.
Change of Control	If we sell assets or if a change of control occurs, subject to certain conditions, we may be required to offer to repurchase the notes at prices listed in the section "Description of New NotesChange of Control." The term "Change of Control" is defined in the "Description of the New NotesCertain Definitions" section of this prospectus.
Guarantees	Each of our subsidiaries will fully and unconditionally guarantee the new notes on a senior subordinated basis. Future subsidiaries also may be required to guarantee the new notes.

The new notes will be subordinated to all existing and future senior indebtedness. The new notes will rank equally with all our other existing and future senior subordinated indebtedness and Ranking..... will rank senior to all our subordinated indebtedness. The guarantees will be subordinated to all existing and future senior indebtedness of such subsidiaries. Because the new notes are subordinated, in the event of bankruptcy, liquidation or dissolution, holders of the new notes will not receive any payment until holders of senior indebtedness and guarantor senior indebtedness have been paid in full. See "Description of the New Notes--Ranking." Specified Covenants..... The indenture governing the new notes will contain covenants that, among other things, restrict our ability and the ability of our restricted subsidiaries to: o borrow money; o pay dividends or repurchase stock; o make investments; o use assets as security in other transactions; o sell certain assets or merge with or into other companies; o sell stock in our subsidiaries; and o limit our subsidiaries from making dividends and other payments. All of these limitations are subject to a number of important exceptions and qualifications, which are described in the "Description of the New Notes--Certain Covenants" section of this prospectus. Transfer Restrictions; Absence of a Public The new notes generally will be freely transferable, but will also be new securities for which there will not initially be a market. Market for the Notes..... There can be no assurance as to the development or liquidity of any market for the new notes.

RISK FACTORS

See "Risk Factors," beginning on page 6 hereof, for a discussion of certain factors that you should consider before participating in the exchange offer.

In addition to the other information set forth elsewhere or incorporated by reference in this prospectus, the following factors relating to our company and the exchange offer and the new notes should be considered carefully in deciding whether to participate in the exchange offer.

RISKS RELATED TO OUR BUSINESS

DOWNTURNS IN CONSTRUCTION COULD ADVERSELY AFFECT OUR BUSINESS BECAUSE MORE THAN HALF OF OUR BUSINESS IS DEPENDENT ON LEVELS OF NEW CONSTRUCTION ACTIVITY.

More than half of our business is the installation of electrical systems in newly constructed and renovated buildings, plants and residences. Our ability to maintain or increase revenues from new installation services will depend on the number of new construction starts and renovations, which will likely be correlated with the cyclical nature of the construction industry. The number of new building starts will be affected by general and local economic conditions, changes in interest rates and other factors, including the following:

- o employment and income levels;
- interest rates and other factors affecting the availability and cost of financing;
- o tax implications for homebuyers;
- o consumer confidence; and
- o housing demand.

Additionally, a majority of our business is focused in the southeastern and southwestern portion of the United States, concentrating our exposure to local economic conditions in those regions. Downturns in levels of construction or housing starts could result in a material reduction in our activity levels.

THE ESTIMATES WE USE IN PLACING BIDS COULD BE MATERIALLY INCORRECT. THE USE OF INCORRECT ESTIMATES COULD RESULT IN LOSSES ON A FIXED PRICE CONTRACT. THESE LOSSES COULD BE MATERIAL TO OUR BUSINESS.

More than half of our revenues are generated under fixed price contracts. We must estimate the costs of completing a particular project to bid for these fixed price contracts. The cost of labor and materials, however, may vary from the costs we originally estimated. Variations from estimated contract costs along with other risks inherent in performing fixed price contracts may result in actual revenue and gross profits for a project differing from those we originally estimated and could result in losses on projects. Depending upon the size of a particular project, actual costs that are higher than estimated contract costs can have a significant impact on our operating results for any fiscal quarter or year. Our results in 2000 were adversely affected by losses recorded on fixed-priced contracts at one of our subsidiaries.

A SIGNIFICANT AMOUNT OF OUR HISTORIC GROWTH HAS OCCURRED THROUGH THE ACQUISITION OF EXISTING BUSINESSES; HOWEVER, FUTURE ACQUISITIONS WILL BE MADE ON A SELECTIVE BASIS AND MAY BE DIFFICULT TO IDENTIFY AND INTEGRATE AND MAY DISRUPT OUR BUSINESS AND ADVERSELY AFFECT OUR OPERATING RESULTS.

Historically, a significant amount of our growth has come through acquisitions. From April 1998 to our last significant acquisition in December 1999, we made 67 acquisitions. Although we have significantly diminished our acquisition activity, we intend to continue to pursue selected acquisition opportunities in the future. We continually evaluate acquisition prospects to complement and expand our existing business platforms. The timing, size or success of any acquisition effort and the associated potential capital commitments cannot be predicted. If we are unable to find appropriate acquisitions, our future ability to grow our revenues and profitability may be diminished. Each acquisition, however, involves a number of risks. These risks include:

- the diversion of our management's attention from our existing businesses to integrating the operations and personnel of the acquired business;
- possible adverse effects on our operating results during the integration process; and
- o our possible inability to achieve the intended objectives of the combination.

We may seek to finance an acquisition through borrowings under our credit facility or through the issuance of new debt or equity securities. There can be no assurance that we will be able to secure this financing if and when it is needed or on terms we consider acceptable. If we should proceed with a relatively large cash acquisition, we could deplete a substantial portion of our financial resources to the possible detriment of our other operations. Any future acquisitions could also dilute the equity interests of our stockholders, require us to write off assets for accounting purposes or create other undesirable accounting issues, such as significant expenses for amortization of goodwill or other intangible assets.

WE MAY EXPERIENCE DIFFICULTIES IN MANAGING INTERNAL GROWTH.

In order to continue to grow internally, we expect to expend significant time and effort managing and expanding existing operations. We cannot guarantee that our systems, procedures and controls will be adequate to support our expanding operations, including the timely receipt of financial information. Our growth imposes significant added responsibilities on our senior management, such as the need to identify, recruit and integrate new senior managers and executives. If we are unable to manage our growth, or if we are unable to attract and retain additional qualified management, our operations could be materially adversely effected.

THERE IS CURRENTLY A SHORTAGE OF QUALIFIED ELECTRICIANS. SINCE THE MAJORITY OF OUR WORK IS PERFORMED BY ELECTRICIANS, THIS SHORTAGE MAY NEGATIVELY IMPACT OUR BUSINESS, INCLUDING OUR ABILITY TO GROW.

There is currently a shortage of qualified electricians in the United States. In order to conduct our business, it is necessary to employ electricians. Over the last few years, the growth of the U.S. economy has increased the demand for electricians, making it difficult for us to attract, hire and retain competent electricians. While overall economic growth has diminished, our ability to increase productivity and profitability may be limited by the difficulty of employing, training and retaining the number of skilled electricians required to meet our needs. Accordingly, we cannot assure you that, among other things:

- o we will be able to maintain the skilled labor force necessary to operate efficiently; and
- o our labor expenses will not increase as a result of a shortage in the skilled labor supply.

DUE TO SEASONALITY AND DIFFERING REGIONAL ECONOMIC CONDITIONS, OUR RESULTS MAY FLUCTUATE FROM PERIOD TO PERIOD.

Our business can be subject to seasonal variations in operations and demand that affect the construction business, particularly in residential construction. Our revenues in the winter are typically significantly lower than in the summer. Accordingly, our performance in any particular quarter may not be indicative of the results that can be expected for any other quarter or for the entire year.

TO SERVICE OUR INDEBTEDNESS AND TO FUND WORKING CAPITAL, WE WILL REQUIRE A SIGNIFICANT AMOUNT OF CASH. OUR ABILITY TO GENERATE CASH DEPENDS ON MANY FACTORS.

Our ability to make payments on and to refinance our indebtedness and to fund planned capital expenditures will depend on our ability to generate cash in the future. This is subject to our operational performance, as well as general economic, financial, competitive, legislative, regulatory and other factors that are beyond our control. Our new credit facility will expire in May 2004.

We cannot assure you that our business will generate sufficient cash flow from operations or that future borrowings will be available to us under our credit facility in an amount sufficient to enable us to pay our indebtedness, or to fund our other liquidity needs. We may need to refinance all or a portion of our indebtedness, on or before maturity. We cannot assure you that we will be able to refinance any of our indebtedness on commercially reasonable terms or at all. Our inability to refinance our debt on commercially reasonable terms could materially adversely affect our business.

THE HIGHLY COMPETITIVE NATURE OF OUR INDUSTRY COULD AFFECT OUR PROFITABILITY BY REDUCING OUR PROFIT MARGINS.

The electrical contracting industry is served by small, owner-operated private companies, public companies and several large regional companies. We could also face competition in the future from other competitors entering these markets. Some of our competition in the rulate from other competitors checking theorem mechanical construction, plumbing and heating, ventilation and air conditioning services. Competition in our markets depends on a number of factors, including price. Some of our competitors may have lower overhead cost structures and may, therefore, be able to provide services comparable to ours at lower rates than we do. If we are unable to offer our services at competitive prices or if we have to reduce our prices to remain competitive, our profitability would be impaired.

OUR OPERATIONS ARE SUBJECT TO NUMEROUS PHYSICAL HAZARDS ASSOCIATED WITH THE CONSTRUCTION OF ELECTRICAL SYSTEMS. IF AN ACCIDENT OCCURS, IT COULD RESULT IN AN ADVERSE EFFECT ON OUR BUSINESS.

Hazards related to our industry include, but are not limited to, electrocutions, fires, mechanical failures or transportation accidents. These hazards can cause personal injury and loss of life, severe damage to or destruction of property and equipment and may result in suspension of operations. Our insurance does not cover all types or amounts of liabilities. Our third-party insurance is subject to large deductibles for which we establish reserves and, accordingly, we effectively self insure for much of our insurance coverage. No assurance can be given either that our insurance or our provisions for incurred claims and incurred but not reported claims will be adequate to cover all losses or liabilities we may incur in our operations or that we will be able to maintain adequate insurance at reasonable rates or at all.

WE HAVE A SUBSTANTIAL AMOUNT OF DEBT. OUR CURRENT DEBT LEVEL COULD LIMIT OUR ABILITY TO FUND FUTURE WORKING CAPITAL NEEDS AND INCREASE OUR EXPOSURE DURING ADVERSE ECONOMIC CONDITIONS.

We have a significant amount of debt. The following chart shows certain important credit statistics and is presented assuming we had completed this offering as of March 31, 2001 and applied the proceeds as described in "Use of Proceeds." The ratio of earnings to fixed charges assumes we had completed this offering at the beginning of the 12-month period.

Total indebtedness	\$276.9	million
Stockholders' equity	\$523.5	million
Debt to total capitalization	0.35x	
Ratio of earnings to fixed charges	3.0x	

Our substantial indebtedness could have important consequences to you. For example, it could:

- make it more difficult for us to satisfy our obligations with respect 0 to the notes;
- increase our vulnerability to general adverse economic and industry 0 conditions:
- limit our ability to fund future working capital, capital expenditures, 0 acquisitions and other general corporate requirements;
- limit our flexibility in planning for, or reacting to, changes in our business and the industry in which we operate; 0
- place us at a disadvantage compared to our competitors that have less 0 debt; and
- limit our ability to borrow additional funds. 0

Additionally, the indenture and other documents governing our indebtedness contain financial and other restrictive covenants. Failing to comply with those covenants could result in an event of default which, if not cured or waived, could have a material adverse effect on us. See "Description of the Notes."

We may incur additional indebtedness in the future which may intensify the risks described above. The terms of the indenture do not prohibit us or our subsidiaries from incurring additional indebtedness, including indebtedness that is senior in right of payment to the notes, although the indenture does contain certain limitations on additional indebtedness. As of July 2, 2001, we had outstanding letters of credit of \$5.3 million and additional available borrowings of up to \$144.7 million under our credit facility, subject to customary borrowing conditions.

THE NEW NOTES AND THE SUBSIDIARY GUARANTEES RANK BEHIND ALL OF OUR AND OUR SUBSIDIARY GUARANTORS' EXISTING AND FUTURE SENIOR INDEBTEDNESS.

As a result of the subordinated nature of the new notes and guarantees, upon any distribution to our creditors or the creditors of the guarantors in a bankruptcy, liquidation or reorganization or similar proceeding relating to us or the guarantors or our or their property, the holders of senior indebtedness of our company and the guarantors will be entitled to be paid in full in cash before any payment may be made with respect to the new notes or the subsidiary guarantees.

In addition, all payments on the new notes and the guarantees will be blocked in the event of a payment default on certain senior indebtedness and may be blocked for up to 179 of 360 consecutive days in the event of certain non-payment defaults on senior indebtedness.

In the event of a bankruptcy, liquidation or reorganization or similar proceeding relating to our company or the guarantors, holders of the new notes will participate with trade creditors and all other holders of subordinated indebtedness of our company and the guarantors in the assets remaining after we and the subsidiary guarantors have paid all of the senior indebtedness. In any of these cases, we and the subsidiary guarantors may not have sufficient funds to pay all of our creditors and holders of new notes may receive less, ratably, than the holders of senior indebtedness. However, because the indenture requires that amounts otherwise payable to holders of senior indebtedness instead, holders of the new notes may receive less, ratably, than holders of trade payables in any such proceeding.

As of July 2, 2001, the new notes and subsidiary guarantees would have been subordinated to approximately \$1.9 senior indebtedness and up to \$144.7 would have been available for borrowing as additional senior indebtedness under our credit facility, subject to customary borrowing conditions. We will be permitted to borrow substantial additional indebtedness, including senior indebtedness, in the future under the terms of the indenture.

THE LOSS OF A GROUP OF KEY PERSONNEL, EITHER AT THE CORPORATE OR OPERATING LEVEL, COULD ADVERSELY AFFECT OUR BUSINESS.

The loss of key personnel or the inability to hire and retain qualified employees could have an adverse effect on our business, financial condition and results of operations. Our operations depend on the continued efforts of our current and future executive officers, senior management and management personnel at the companies we have acquired. A criteria we use in evaluating acquisition candidates is the quality of their management. We cannot guarantee that any member of management at the corporate or subsidiary level will continue in their capacity for any particular period of time. If we lose a group of key personnel, our operations could be adversely affected. We do not maintain key man life insurance.

SOME SIGNIFICANT RESTRUCTURING TRANSACTIONS MAY NOT CONSTITUTE A CHANGE OF CONTROL, WHICH WOULD OBLIGATE US TO OFFER TO REPURCHASE YOUR NEW NOTES, AND EVEN IF A TRANSACTION DOES TRIGGER THAT OBLIGATION, WE MAY NOT BE ABLE TO REPURCHASE THE NEW NOTES. Upon the occurrence of certain specific kinds of change of control events, we will be required to offer to repurchase all outstanding new notes. However, certain important corporate events, such as leveraged recapitalizations that would increase the level of our indebtedness, would not constitute a "Change of Control" under the indenture. Moreover, even if a transaction triggers a repurchase obligation, it is possible that we will not have sufficient funds at the time of the change of control to make the required repurchase of new notes. In addition, restrictions in our credit facility will not allow such repurchases. See "Description of Notes -- Change of Control."

A SUBSIDIARY GUARANTEE COULD BE VOIDED OR SUBORDINATED BECAUSE OF FEDERAL BANKRUPTCY LAW OR COMPARABLE STATE LAW PROVISIONS.

Under the federal bankruptcy law and comparable provisions of state fraudulent transfer laws, one or more of the subsidiary guarantees could be voided, or claims in respect of a subsidiary guarantee could be subordinated to all other debts of that guarantor if, among other things, the guarantor, at the time it incurred the indebtedness evidenced by its guarantee received less than reasonably equivalent value or fair consideration for the incurrence of such guarantee; and

- o was insolvent or rendered insolvent by reason of such incurrence; or
- was engaged in a business or transaction for which the guarantor's remaining assets constituted unreasonably small capital; or
- o intended to incur, or believed that it would incur, debts beyond its ability to pay such debts as they mature.

In addition, any payment by that guarantor pursuant to its guarantee could be voided and required to be returned to the guarantor or to a fund for the benefit of the creditors of the guarantor.

The measures of insolvency for purposes of these fraudulent transfer laws will vary depending upon the law applied in any proceeding to determine whether a fraudulent transfer has occurred. Generally, however, a guarantor would be considered insolvent if:

- o the sum of its debts, including contingent liabilities, were greater than the fair saleable value of all of its assets; or
- o the present fair saleable value of its assets were less than the amount that would be required to pay its probable liability on its existing debts, including contingent liabilities, as they become absolute and mature; or
- o it could not pay its debts as they become due.

We cannot be sure as to the standards that a court would use to determine whether or not the guarantors were solvent at the relevant time, or, regardless of the standard that the court uses, that the issuance of the guarantee of the notes would not be voided or the guarantee of the notes would not be subordinated to that guarantor's other debt.

If the guarantees were legally challenged, any guarantee could also be subject to the claim that, since the guarantee was incurred for our benefit, and only indirectly for the benefit of the guarantor, the obligations of the applicable guarantor were incurred for less than fair consideration.

A court could thus void the obligations under the guarantee or subordinate the guarantee to the applicable guarantor's other debt or take other action detrimental to holders of the new notes.

OUR RESULTS OF OPERATIONS COULD BE ADVERSELY AFFECTED AS A RESULT OF GOODWILL AMORTIZATION OR WRITE-OFFS.

When we acquire a business, we record an asset called "goodwill" equal to the excess amount we pay for the business, including liabilities assumed, over the fair value of the tangible assets of the business we acquire. Pursuant to accounting principles generally accepted in the United States, we amortize this goodwill over its estimated useful

life of 40 years following the acquisition, which directly impacts our earnings in those years. Furthermore, we continually evaluate whether events or circumstances have occurred that indicate that the remaining useful life of goodwill may warrant revision or that the remaining balance may not be recoverable.

On February 14, 2001 the Financial Accounting Standards Board issued a revised limited Exposure Draft of its 1999 proposed Statement, Business Combinations and Intangible Assets. The Exposure Draft contains the FASB's tentative decision reached in December 2000 to eliminate amortization of purchased goodwill. Under that approach, goodwill would not be amortized to earnings. Instead, goodwill would be reviewed for impairment, meaning that it would be written down and expensed against earnings in the periods in which the recorded value of goodwill exceeds its fair value. FASB expects to issue its final statement regarding the treatment of purchased goodwill by the end of July 2001. Because the proposal has not been finalized and certain details regarding its use have yet to be defined, we are unable to predict the way in which its adoption will affect our financial reports. Because we have made numerous acquisitions since our founding, we carry significant amounts of goodwill on our books for which we would no longer be required to recognize a periodic amortization charge as a result of this proposal.

Should we be required to accelerate the amortization of goodwill or write it off completely because of impairments, our results from operations may be materially adversely affected.

RISKS RELATED TO THE EXCHANGE OFFER AND THE NEW NOTES

IF YOU DO NOT PROPERLY TENDER YOUR OUTSTANDING NOTES, YOU WILL CONTINUE TO HOLD UNREGISTERED OUTSTANDING NOTES AND YOUR ABILITY TO TRANSFER OUTSTANDING NOTES WILL BE ADVERSELY AFFECTED.

We will only issue new notes in exchange for outstanding notes that you timely and properly tender. Therefore, you should allow sufficient time to endure timely delivery of the outstanding notes and you should carefully follow the instructions on how to tender your outstanding notes. Neither we nor the exchange agent is required to tell you of any defects or irregularities with respect to your tender of outstanding notes.

If you do not exchange your outstanding notes for new notes pursuant to the exchange offer, the outstanding notes you hold will continue to be subject to the existing transfer restrictions. In general, you may not offer or sell the outstanding notes except under an exemption from, or in a transaction not subject to, the Securities Act of 1933 and applicable state securities laws. We do not plan to register outstanding notes under the Securities Act of 1933 unless our registration rights agreement with the initial purchasers of the outstanding notes requires us to do so. Further, if you continue to hold any outstanding notes after the exchange offer is consummated, you may have trouble selling them because there will be fewer outstanding notes outstanding.

IF AN ACTIVE TRADING MARKET DOES NOT DEVELOP FOR THE NEW NOTES, YOU MAY BE UNABLE TO SELL THE NEW NOTES OR TO SELL THE NEW NOTES AT A PRICE THAT YOU DEEM SUFFICIENT.

The new notes will be new securities for which there currently is no established trading market. Although we will register the new notes under the Securities Act of 1933, we do not intend to apply for listing of the new notes on any securities exchange or for quotation of the new notes in any automated dealer quotation system. In addition, although the initial purchasers of the outstanding notes have informed us that they intend to make a market in the new notes after the exchange offer, the initial purchasers may stop making a market at any time. Finally, if a large number of holders of outstanding notes do not tender outstanding notes or tender outstanding notes improperly, the limited amount of new notes that would be issued and outstanding after we consummate the exchange offer could adversely affect the development of a market for these new notes.

A GUARANTEE COULD BE VOIDED IF THE GUARANTOR FRAUDULENTLY TRANSFERRED THE GUARANTEE AT THE TIME IT INCURRED THE INDEBTEDNESS, WHICH COULD RESULT IN THE NOTEHOLDERS BEING ABLE TO RELY ON ONLY INTEGRATED ELECTRICAL SERVICES, INC. TO SATISFY CLAIMS.

Under U.S. bankruptcy law and comparable provisions of state fraudulent transfer laws, a guarantee can be voided, or claims under a guarantee may be subordinated to all other debts of that guarantor if, among other things, the guarantor, at the time it incurred the indebtedness evidenced by its guarantee:

- intended to hinder, delay or defraud any present or future creditor or received less than reasonably equivalent value or fair consideration for the incurrence of the guarantee;
- o was insolvent or rendered insolvent by reason of such incurrence;
- was engaged in a business or transaction for which the guarantor's remaining assets constituted unreasonably small capital; or
- o intended to incur, or believed that it would incur, debts beyond its ability to pay those debts as they mature.

In addition, any payment by that guarantor under a guarantee could be voided and required to be returned to the guarantor or to a fund for the benefit of the creditors of the guarantor.

The measures of insolvency for purposes of fraudulent transfer laws vary depending upon the governing law. Generally, a guarantor would be considered insolvent if:

- o the sum of its debts, including contingent liabilities, was greater than the fair saleable value of all of its assets;
- o the present fair saleable value of its assets was less than the amount that would be required to pay its probable liability on its existing debts, including contingent liabilities, as they became absolute and mature: or
- o it could not pay its debts as they became due.

On the basis of historical financial information, recent operating history and other factors, we believe that the subsidiary guarantees are being incurred for proper purposes and in good faith and that each subsidiary guarantor, after giving effect to its guarantee of the notes, will not be insolvent, have unreasonably small capital for the business in which it is engaged or have incurred debts beyond their ability to pay those debts as they mature. We cannot be certain, however, that a court would agree with our conclusions in this regard.

HOLDERS OF THE NEW NOTES WILL BE EFFECTIVELY SUBORDINATED TO ALL OF OUR AND THE SUBSIDIARY GUARANTORS' SECURED INDEBTEDNESS AND TO ALL LIABILITIES OF OUR NON-GUARANTOR SUBSIDIARIES.

Holders of our secured indebtedness, including the indebtedness under our credit facilities, have claims with respect to our assets constituting collateral for their indebtedness that are prior to your claims under the new notes. In the event of a default on the new notes or our bankruptcy, liquidation or reorganization, those assets would be available to satisfy obligations with respect to the indebtedness secured thereby before any payment could be made on the new notes. Accordingly, the secured indebtedness would effectively be senior to the new notes to the extent of the value of the collateral securing the indebtedness. While the indenture governing the new notes places some limitations on our ability to create liens, there are significant exceptions to these limitations, including with respect to sale and leaseback transactions, that will allow us to secure some kinds of indebtedness without equally and ratably securing the new notes. To the extent the value of the collateral is not sufficient to satisfy the secured indebtedness, the holders of that indebtedness would be entitled to share with the holders of the new notes and the holders of other claims against us with respect to our other assets.

In addition, the new notes are not guaranteed by all of our subsidiaries, and our non-guarantor subsidiaries are permitted to incur additional indebtedness under the indenture. As a result, holders of the new notes will be effectively subordinated to claims of third party creditors, including holders of indebtedness, and preferred shareholders of these non-guarantor subsidiaries. Claims of those other creditors, including trade creditors, secured creditors, authorities, holders of indebtedness or guarantees issued by the non-guarantor subsidiaries and preferred shareholders of the non-guarantor subsidiaries, will generally have priority as to the assets of the non-guarantor subsidiaries over our claims and equity interests. As a result, holders of our indebtedness, including the holders of the new notes, will be effectively subordinated to all those claims.

EXCHANGE OFFER

PURPOSE AND EFFECT OF THE EXCHANGE OFFER

In connection with the issuance of the outstanding notes, we entered into a registration rights agreement. Under the registration rights agreement, we agreed to:

- o within 60 days after the original issuance of the outstanding notes, file a registration statement with the SEC with respect to a registered offer to exchange each outstanding note for a new note having terms substantially identical in all material respects to such note except that the new note will not contain terms with respect to transfer restrictions);
- use our reasonable best efforts to cause the registration statement to be declared effective under the Securities Act of 1933 within 150 days after the original issuance of the outstanding notes;
- o promptly following the effectiveness of the registration statement, offer the new notes in exchange for surrender of the outstanding notes; and
- o keep the exchange offer open for not less than 20 business days (or longer if required by applicable law) after the date notice of the exchange offer is mailed to the holders of the outstanding notes.

We have fulfilled the agreements described in the first two of the preceding bullet points, and as soon as practicable after this registration statement becomes effective, we will offer eligible holders of the outstanding notes the opportunity to exchange their outstanding notes for new notes registered under the Securities Act. Holders are eligible if they are not prohibited by any law or policy of the SEC from participating in this exchange offer. The new notes will be substantially identical to the outstanding notes except that the new notes will not contain terms with respect to transfer restrictions, registration rights or additional interest.

Under limited circumstances, we agreed to use our reasonable best efforts to cause the SEC to declare effective a shelf registration statement for the resale of the outstanding notes. We also agreed to use our reasonable best efforts to keep the shelf registration statement effective for up to two years after its effective date. The circumstances include if:

- we are not permitted to effect the exchange offer because of any change in law or applicable interpretations of the law by the staff of the SEC;
- o for any other reason the exchange offer is not consummated within 180 days after the date of issuance of the notes;
- any notes validly tendered pursuant to the exchange offer are not exchanged by us for exchange notes within ten days of being accepted in the exchange offer;
- the initial purchasers so request with respect to notes held by the initial purchasers that are not eligible to be exchanged for exchange notes in the exchange offer;
- o any applicable law or interpretations do not permit any holder of notes to participate in the exchange offer; or
- any holder of notes that participates in the exchange offer does not receive freely transferable exchange notes in exchange for tendered notes.

We will pay additional cash interest on the applicable outstanding notes, subject to certain exceptions if:

22

- o the exchange offer registration statement is not filed with the Securities and Exchange Commission on or before 60 days after the date of the original issuance of the outstanding notes or the shelf registration statement is not filed with the SEC on or before the shelf filing date;
- o the exchange offer registration statement is not declared effective within 150 days after the date of the original issuance of the outstanding notes or the shelf registration statement is not declared effective within 120 days after the shelf filing date;
- o the exchange offer is not consummated on or before 180 days after the date of the original issuance of the outstanding notes; or
- the shelf registration statement is filed and declared effective within 120 days after the shelf filing date but thereafter ceases to be effective, at any time that we and our guarantor subsidiaries are obligated to maintain its effectiveness, without being succeeded within 30 days by an additional registration statement filed and declared effective (each such event referred to in the preceding clauses being a "registration default")

from and including the date on which any such registration default occurs to but excluding the date on which all registration defaults have been cured.

The rate of additional interest will be increased by an additional 0.25% per annum upon the occurrence of the registration default, and shall be further increased by 0.25% on the 91st day after a registration default has occurred, until the applicable registration statement is filed, the exchange offer registration statement is declared effective and the exchange offer is consummated, or the shelf registration statement is declared effective or again becomes effective, as the case may be. All accrued additional interest will be paid to holders in the same manner as interest payments on the notes on semi-annual payment dates that correspond to interest payment dates for the notes. Additional interest only accrues during a registration default.

Upon the effectiveness of this registration statement, the consummation of the exchange offer, the effectiveness of a shelf registration statement, or the effectiveness of a succeeding registration statement, as the case may be, the interest rate borne by the notes from the date of such effectiveness or consummation, as the case may be, will be reduced to the original interest rate. However, if after any such reduction in interest rate, a different event specified in the clauses above occurs, the interest rate may again be increased pursuant to the preceding provisions.

To exchange your outstanding notes for transferable new notes in the exchange offer, you will be required to make the following representations:

- o any new notes will be acquired in the ordinary course of your business;
- you have no arrangement or understanding with any person or entity to participate in the distribution of the new notes;
- you are not engaged in and do not intend to engage in the distribution of the new notes;
- o if you are a broker-dealer that will receive new notes for your own account in exchange for outstanding notes, you acquired those notes as a result of market-making activities or other trading activities and you will deliver a prospectus, as required by law, in connection with any resale of such new notes; and
- o you are not our "affiliate," as defined in Rule 405 of the Securities Act.

In addition, we may require you to provide information to be used in connection with the shelf registration statement to have your outstanding notes included in the shelf registration statement and benefit from the provisions regarding additional interest described in the preceding paragraphs. A holder who sells outstanding notes under the shelf registration statement generally will be required to be named as a selling securityholder in the related prospectus and to deliver a prospectus to purchasers. Such a holder will also be subject to the civil liability

The description of the registration rights agreement contained in this section is a summary only. For more information, you should review the provisions of the registration rights agreement that we filed with the SEC as an exhibit to the registration statement of which this prospectus is a part.

RESALE OF NEW NOTES

Based on no action letters of the SEC staff issued to third parties, we believe that new notes may be offered for resale, resold and otherwise transferred by you without further compliance with the registration and prospectus delivery provisions of the Securities Act if:

- o you are not our "affiliate" within the meaning of Rule 405 under the Securities Act;
- o such new notes are acquired in the ordinary course of your business; and
- o you do not intend to participate in a distribution of the new notes.

The SEC, however, has not considered the exchange offer for the new notes in the context of a no action letter, and the SEC may not make a similar determination as in the no action letters issued to these third parties.

If you tender in the exchange offer with the intention of participating in any manner in a distribution of the new notes, you

- o cannot rely on such interpretations by the SEC staff; and
- must comply with the registration and prospectus delivery requirements of the Securities Act in connection with a secondary resale transaction.

Unless an exemption from registration is otherwise available, any security holder intending to distribute new notes should be covered by an effective registration statement under the Securities Act. This registration statement should contain the selling security holder's information required by Item 507 of Regulation S-K under the Securities Act. This prospectus may be used for an offer to resell, resale or other retransfer of new notes only as specifically described in this prospectus. Only broker-dealers that acquired the outstanding notes as a result of market-making activities or other trading activities may participate in the exchange offer. Each broker-dealer that receives new notes for its own account in exchange for outstanding notes, where such outstanding notes were acquired by such broker-dealer as a result of market-making activities or other trading activities, must acknowledge in the letter of transmittal that it will deliver a prospectus in connection with any resale of the new notes. Please read the section captioned "Plan of Distribution" for more details regarding the transfer of new notes.

TERMS OF THE EXCHANGE OFFER

Subject to the terms and conditions described in this prospectus and in the letter of transmittal, we will accept for exchange any outstanding notes properly tendered and not withdrawn prior to 5:00 p.m. New York City time on the expiration date. We will issue new notes in principal amount equal to the principal amount of outstanding notes surrendered under the exchange offer. Outstanding notes may be tendered only for new notes and only in integral multiples of \$1,000.

The exchange offer is not conditioned upon any minimum aggregate principal amount of outstanding notes being tendered for exchange.

As of the date of this prospectus, \$125,000,000 in aggregate principal amount of the outstanding notes are outstanding. This prospectus and the letter of transmittal are being sent to all registered holders of outstanding

notes. There will be no fixed record date for determining registered holders of outstanding notes entitled to participate in the exchange offer.

We intend to conduct the exchange offer in accordance with the provisions of the registration rights agreement, the applicable requirements of the Securities Act and the Securities Exchange Act of 1934 and the rules and regulations of the SEC. Outstanding notes that the holders thereof do not tender for exchange in the exchange offer will remain outstanding and continue to accrue interest. These outstanding notes will be entitled to the rights and benefits such holders have under the indenture relating to the notes and the registration rights agreement.

We will be deemed to have accepted for exchange properly tendered outstanding notes when we have given oral or written notice of the acceptance to the exchange agent and complied with the applicable provisions of the registration rights agreement. The exchange agent will act as agent for the tendering holders for the purposes of receiving the new notes from us.

If you tender outstanding notes in the exchange offer, you will not be required to pay brokerage commissions or fees or, subject to the letter of transmittal, transfer taxes with respect to the exchange of outstanding notes. We will pay all charges and expenses, other than certain applicable taxes described below, in connecting with the exchange offer. It is important that you read the section labeled "--Fees and Expenses" for more details regarding fees and expenses incurred in the exchange offer.

We will return any outstanding notes that we do not accept for exchange for any reason without expense to their tendering holder as promptly as practicable after the expiration or termination of the exchange offer.

EXPIRATION DATE

The exchange offer will expire at 5:00 p.m. New York City time on _ 2001, unless, in our sole discretion, we extend it.

EXTENSIONS, DELAYS IN ACCEPTANCE, TERMINATION OR AMENDMENT

We expressly reserve the right, at any time or various times, to extend the period of time during which the exchange offer is open. We may delay acceptance of any outstanding notes by giving oral or written notice of such extension to their holders. During any such extensions, all outstanding notes previously tendered will remain subject to the exchange offer, and we may accept them for exchange.

In order to extend the exchange offer, we will notify the exchange agent orally or in writing of any extension. We will notify the registered holders of outstanding notes of the extension no later than 9:00 a.m., New York City time, on the business day after the previously scheduled expiration date.

If any of the conditions described below under "--Conditions to the Exchange Offer" have not been satisfied, we reserve the right, in our sole discretion

to delay accepting for exchange any outstanding notes, 0

to extend the exchange offer, or 0

to terminate the exchange offer. 0

by giving oral or written notice of such delay, extension or termination to the exchange agent. Subject to the terms of the registration rights agreement, we also reserve the right to amend the terms of the exchange offer in any manner.

Any such delay in acceptance, extension, termination or amendment will be followed as promptly as practicable by oral or written notice thereof to the registered holders of outstanding notes. If we amend the exchange offer in a manner that we determine to constitute a material change, we will promptly disclose such amendment by means of a

25

prospectus supplement. The supplement will be distributed to the registered holders of the outstanding notes. Depending upon the significance of the amendment and the manner of disclosure to the registered holders, we will extend the exchange offer if the exchange offer would otherwise expire during such period.

CONDITIONS TO THE EXCHANGE OFFER

We will not be required to accept for exchange, or exchange any new notes for, any outstanding notes if the exchange offer, or the making of any exchange by a holder of outstanding notes, would violate applicable law or any applicable interpretation of the staff of the SEC. Similarly, we may terminate the exchange offer as provided in this prospectus before accepting outstanding notes for exchange in the event of such a potential violation.

In addition, we will not be obligated to accept for exchange the outstanding notes of any holder that has not made to us the representations described under "--Purpose and Effect of the Exchange Offer," "--Procedures for Tendering" and "Plan of Distribution" and such other representations as may be reasonably necessary under applicable SEC rules, regulations or interpretations to allow us to use an appropriate form to register the new notes under the Securities Act.

We expressly reserve the right to amend or terminate the exchange offer, and to reject for exchange any outstanding notes not previously accepted for exchange, upon the occurrence of any of the conditions to the exchange offer specified above. We will give oral or written notice of any extension, amendment, non-acceptance or termination to the holders of the outstanding notes as promptly as practicable.

These conditions are for our sole benefit, and we may assert them or waive them in whole or in part at any time or at various times in our sole discretion. If we fail at any time to exercise any of these rights, this failure will not mean that we have waived our rights. Each such right will be deemed an ongoing right that we may assert at any time or at various times.

In addition, we will not accept for exchange any outstanding notes tendered, and will not issue new notes in exchange for any such outstanding notes, if at such time any stop order has been threatened or is in effect with respect to the registration statement of which this prospectus constitutes a part or the qualification of the indenture relating to the notes under the Trust Indenture Act of 1939.

PROCEDURES FOR TENDERING

How to Tender Generally

Only a holder of outstanding notes may tender such outstanding notes in the exchange offer. To tender in the exchange offer, a holder must:

- complete, sign and date the letter of transmittal, or a facsimile of the letter of transmittal;
- o have the signature on the letter of transmittal guaranteed if the letter of transmittal so requires; and
 - o mail or deliver such letter of transmittal or facsimile to the exchange agent prior to 5:00 p.m. New York City time on the expiration date; or
 - o comply with the automated tender offer program procedures of The Depository Trust Company, or DTC, described below.

In addition, either:

- the exchange agent must receive outstanding notes along with the letter of transmittal;
- o the exchange agent must receive, prior to 5:00 p.m. New York City time on the expiration date, a timely confirmation of book-entry transfer of such outstanding notes into the exchange agent's account at DTC

according to the procedure for book-entry transfer described below or a properly transmitted agent's message; or

 the holder must comply with the guaranteed delivery procedures described below.

To be tendered effectively, the exchange agent must receive any physical delivery of the letter of transmittal and other required documents at its address indicated on the cover page of the letter of transmittal. The exchange agent must receive such documents prior to 5:00 p.m. New York City time on the expiration date.

The tender by a holder that is not withdrawn prior to 5:00 p.m. New York City time on the expiration date will constitute an agreement between the holder and us in accordance with the terms and subject to the conditions described in this prospectus and in the letter of transmittal.

THE METHOD OF DELIVERY OF OUTSTANDING NOTES, THE LETTER OF TRANSMITTAL AND ALL OTHER REQUIRED DOCUMENTS TO THE EXCHANGE AGENT IS AT YOUR ELECTION AND RISK. RATHER THAN MAIL THESE ITEMS, WE RECOMMEND THAT YOU USE AN OVERNIGHT OR HAND DELIVERY SERVICE. IN ALL CASES, YOU SHOULD ALLOW SUFFICIENT TIME TO ASSURE DELIVERY TO THE EXCHANGE AGENT BEFORE 5:00 P.M. NEW YORK CITY TIME ON THE EXPIRATION DATE. YOU SHOULD NOT SEND THE LETTER OF TRANSMITTAL OR OUTSTANDING NOTES TO US. YOU MAY REQUEST YOUR BROKERS, DEALERS, COMMERCIAL BANKS, TRUST COMPANIES OR OTHER NOMINEES TO EFFECT THE ABOVE TRANSACTIONS FOR YOU.

How to Tender if You Are a Beneficial Owner

If you beneficially own outstanding notes that are registered in the name of a broker, dealer, commercial bank, trust company or other nominee and you wish to tender those notes, you should contact the registered holder promptly and instruct it to tender on your behalf. If you are a beneficial owner and wish to tender on your own behalf, you must, prior to completing and executing the letter of transmittal and delivering your outstanding notes, either:

- o make appropriate arrangements to register ownership of the outstanding notes in your name; or
- o obtain a properly completed bond power from the registered holder of outstanding notes.

The transfer of registered ownership, if permitted under the indenture for the notes, may take considerable time and may not be completed prior to the expiration date.

Signatures and Signature Guarantees

You must have signatures on a letter of transmittal or a notice of withdrawal (as described below) guaranteed by a member firm of a registered national securities exchange or of the National Association of Securities Dealers, Inc., a commercial bank or trust company having an office or correspondent in the United States, or an "eligible guarantor institution" within the meaning of Rule 17Ad-15 under the Securities Exchange Act. In addition, such entity must be a member of one of the recognized signature guarantee programs identified in the letter of transmittal. Signature guarantees are not required, however, if the notes are tendered:

- by a registered holder who has not completed the box entitled "Special Issuance Instructions" or "Special Delivery Instructions" on the letter of transmittal;
- o for the account of a member firm of a registered national securities exchange or of the National Association of Securities Dealers, Inc., a commercial bank or trust company having an office or correspondence in the United States, or an eligible guarantor institution.

When You Need Endorsements or Bond Powers

If the letter of transmittal is signed by a person other than the registered holder of any outstanding notes, the outstanding notes must be endorsed or accompanied by a properly completed bond power. The bond power must be

signed by the registered holder as the registered holder's name appears on the outstanding notes. A member firm of a registered national securities exchange or of the National Association of Securities Dealers, Inc., a commercial bank or trust company having an office or correspondent in the United States, or an eligible guarantor institution must guarantee the signature on the bond power.

28

If the letter of transmittal or any outstanding notes or bond powers are signed by trustees, executors, administrators, guardians, attorneys-in-fact, officers of corporations or others acting in a fiduciary or representative capacity, those persons should so indicate when signing. Unless waived by us, they should also submit evidence satisfactory to us of their authority to deliver the letter of transmittal.

Tendering Through DTC's Automated Tender Offer Program

The exchange agent and DTC have confirmed that any financial institution that is a participant in DTC's system may use DTC's automated tender offer program to tender. Participants in the program may, instead of physically completing and signing the letter of transmittal and delivering it to the exchange agent, transmit their acceptance of the exchange offer electronically. They may do so by causing DTC to transfer the outstanding notes to the exchange agent in accordance with its procedures for transfer. DTC will then send an agent's message to the exchange agent.

The term "agent's message" means a message transmitted by DTC, received by the exchange agent and forming part of the book-entry confirmation, to the effect that:

- DTC has received an express acknowledgment from a participant in its automated tender offer program that is tendering outstanding notes that are the subject of such book-entry confirmation;
- o such participant has received and agrees to be bound by the terms of the letter of transmittal or, in the case of an agent's message relating to guaranteed delivery, that such participant has received and agrees to be bound by the applicable notice of guaranteed delivery; and
- o the agreement may be enforced against such participant.

Determinations Under the Exchange Offer

We will determine in our sole discretion all questions as to the validity, form, eligibility, time of receipt, acceptance of tendered outstanding notes and withdrawal of tendered outstanding notes. Our determination will be final and binding. We reserve the absolute right to reject any outstanding notes not properly tendered or any outstanding notes our acceptance of which would, in the opinion of our counsel, be unlawful. We also reserve the right to waive any defect, irregularities or conditions of tender as to particular outstanding notes. Our interpretation of the terms and conditions of the exchange offer, including the instructions in the letter of transmittal, will be final and binding on all parties. Unless waived, all defects or irregularities in connection with tenders of outstanding notes must be cured within such time as we shall determine. Although we intend to notify holders of defects or irregularities with respect to tenders of outstanding notes, neither we, the exchange agent nor any other person will incur any liability for failure to give such notification. Tenders of outstanding notes will not be deemed made until such defects or irregularities have been cured or waived. Any outstanding notes received by the exchange agent that are not properly tendered and as to which the defects or irregularities have not been cured or waived will be returned to the tendering holder, unless otherwise provided in the letter of transmittal, as soon as practicable following the expiration date.

When We Will Issue New Notes

In all cases, we will issue new notes for outstanding notes that we have accepted for exchange under the exchange offer only after the exchange agent timely receives:

 o outstanding notes or a timely book-entry confirmation of such outstanding notes into the exchange agent's account at DTC; and

 a properly completed and duly executed letter of transmittal and all other required documents or a properly transmitted agent's message.

Return of Outstanding Notes Not Accepted or Exchanged

If we do not accept any tendered outstanding notes for exchange or if outstanding notes are submitted for a greater principal amount than the holder desires to exchange, the unaccepted or non-exchanged outstanding notes will be returned without expense to their tendering holder. In the case of outstanding notes tendered by book-entry transfer in the exchange agent's account at DTC according to the procedures described below, such non-exchanged outstanding notes will be credited to an account maintained with DTC. These actions will occur as promptly as practicable after the expiration or termination of the exchange offer.

Your Representations to Us

By signing or agreeing to be bound by the letter of transmittal, you will represent to us that, among other things:

- any new notes that you receive will be acquired in the ordinary course of your business;
- you have no arrangement or understanding with any person or entity to participate in the distribution of the new notes;
- you are not engaged in and do not intend to engage in the distribution of the new notes;
- o if you are a broker-dealer that will receive new notes for your own account in exchange for outstanding notes, you acquired those notes as a result of market-making activities or other trading activities and you will deliver a prospectus, as required by law, in connection with any resale of such new notes; and
- o you are not our "affiliate," as defined in Rule 405 of the Securities Act.

BOOK-ENTRY TRANSFER

The exchange agent will establish an account with respect to the outstanding notes at DTC for purposes of the exchange offer promptly after the date of this prospectus. Any financial institution participating in DTC's system may make book-entry delivery of outstanding notes by causing DTC to transfer such outstanding notes into the exchange agent's account at DTC in accordance with DTC's procedures for transfer. Holders of outstanding notes who are unable to deliver confirmation of the book-entry tender of their outstanding notes into the exchange agent's account at DTC or all other documents required by the letter of transmittal to the exchange agent on or prior to 5:00 p.m. New York City time on the expiration date must tender their outstanding notes according to the guaranteed delivery procedures described below.

GUARANTEED DELIVERY PROCEDURES

If you wish to tender your outstanding notes but your outstanding notes are not immediately available or you cannot deliver your outstanding notes, the letter of transmittal or any other required documents to the exchange agent or comply with the applicable procedures under DTC's automated tender offer program prior to the expiration date, you may tender if:

- the tender is made through a member firm of a registered national securities exchange or of the National Association of Securities Dealers, Inc., a commercial bank or trust company having an office or correspondent in the United States, or an eligible guarantor institution,
- o prior to the expiration date, the exchange agent receives from such member firm of a registered national securities exchange or of the National Association of Securities Dealers, Inc., commercial bank or trust company having a office or correspondent in the United States, or eligible guarantor institution either a

properly completed and duly executed notice of guaranteed delivery by facsimile transmission, mail or hand delivery or a properly transmitted agent's message and notice of guaranteed delivery:

- setting forth your name and address, the registered number(s) of your outstanding notes and the principal amount of outstanding notes tendered,
- o stating that the tender is being made thereby, and
- o guaranteeing that, within three (3) New York Stock Exchange ("NYSE") trading days after the expiration date, the letter of transmittal or facsimile thereof, together with the outstanding notes or a book-entry confirmation, and any other documents required by the letter of transmittal will be deposited by the eligible guarantor institution with the exchange agent, and
- the exchange agent receives such properly completed and executed letter of transmittal or facsimile thereof, as well as all tendered outstanding notes in proper form for transfer or a book-entry confirmation, and all other documents required by the letter of transmittal, within three (3) NYSE trading days after the expiration date.

Upon request to the exchange agent, a notice of guaranteed delivery will be sent you if you wish to tender your outstanding notes according to the guaranteed delivery procedures described above.

WITHDRAWAL OF TENDERS

Except as otherwise provided in this prospectus, you may withdraw your tender at any time prior to 5:00 p.m. New York City time on the expiration date.

For a withdrawal to be effective:

- o the exchange agent must receive a written notice of withdrawal at the address indicated on the cover page of the letter of transmittal or
- o you must comply with the appropriate procedures of DTC's automated tender offer program system.

Any notice of withdrawal must:

- o $% \left({{{\rm{specify}}} } \right)$ specify the name of the person who tendered the outstanding notes to be withdrawn, and
- o identify the outstanding notes to be withdrawn, including the principal amount of such outstanding notes.

If outstanding notes have been tendered under the procedure for book-entry transfer described above, any notice of withdrawal must specify the name and number of the account at DTC to be credited with withdrawn outstanding notes and otherwise comply with the procedures of DTC.

We will determine all questions as to the validity, form, eligibility and time of receipt of notice of withdrawal. Our determination shall be final and binding on all parties. We will deem any outstanding notes so withdrawn not to have been validly tendered for exchange for purposes of the exchange offer.

Any outstanding notes that have been tendered for exchange but that are not exchanged for any reason will be returned to their holder without cost to the holder. In the case of outstanding notes tendered by book-entry transfer into the exchange agent's account at DTC according to the procedures described above, such outstanding notes will be credited to an account maintained with DTC for the outstanding notes. This return or crediting will take place as soon as practicable after withdrawal, rejection of tender or termination of the exchange offer. You may retender properly withdrawn outstanding notes by following one of the procedures described under "--Procedures for Tendering" above at any time on or prior to the expiration date.

FEES AND EXPENSES

We will bear the expenses of soliciting tenders. The principal solicitation is being made by mail; however, we may make additional solicitation by telegraph, telephone or in person by our officers and regular employees and those of our affiliates.

We have not retained any dealer-manager in connection with the exchange offer and will not make any payments to broker-dealers or others soliciting acceptances of the exchange offer. We will, however, pay the exchange agent reasonable and customary fees for its services and reimburse it for its related reasonable out-of-pocket expenses.

We will pay the cash expenses to be incurred in connection with the exchange offer. They include:

- o SEC registration fees;
- o fees and expenses of the exchange agent and trustee;
- o accounting and legal fees and printing costs; and
- o related fees and expenses.

TRANSFER TAXES

We will pay all transfer taxes, if any, applicable to the exchange of outstanding notes under the exchange offer. The tendering holder, however, will be required to pay any transfer taxes, whether imposed on the registered holder or any other person, if:

- certificates representing outstanding notes for principal amounts not tendered or accepted for exchange are to be delivered to, or are to be issued in the name of, any person other than the registered holder of outstanding notes tendered;
- o tendered outstanding notes are registered in the name of any person other than the person signing the letter of transmittal; or
- o a transfer tax is imposed for any reason other than the exchange of outstanding notes under the exchange offer.

If satisfactory evidence of payment of any transfer taxes payable by a note holder is not submitted with the letter of transmittal, the amount of such transfer taxes will be billed directly to that tendering holder.

CONSEQUENCES OF FAILURE TO EXCHANGE

If you do not exchange new notes for your outstanding notes under the exchange offer, you will remain subject to the existing restrictions on transfer of the outstanding notes. In general, you may not offer or sell the outstanding notes unless they are registered under the Securities Act, or if the offer or sale is exempt from the registration under the Securities Act and applicable state securities laws. Except as required by the registration rights agreement, we do not intend to register resales of the outstanding notes under the Securities Act.

ACCOUNTING TREATMENT

We will record the new notes in our accounting records at the same carrying value as the outstanding notes. This carrying value is the aggregate principal amount of the outstanding notes less the original issue discount, as reflected in our accounting records on the date of exchange. Accordingly, we will not recognize any gain or loss for accounting purposes in connection with the exchange offer.

OTHER

Participation in the exchange offer is voluntary, and you should carefully consider whether to accept. You are urged to consult your financial and tax advisors in making your own decision on what action to take.

We may in the future seek to acquire untendered outstanding notes in open market or privately negotiated transactions, through subsequent exchange offers or otherwise. We have no present plans to acquire any outstanding notes that are not tendered in the exchange offer or to file a registration statement to permit resales of any untendered outstanding notes.

USE OF PROCEEDS

The exchange offer is intended to satisfy our obligations under the registration rights agreement. We will not receive any cash proceeds from the issuance of the new notes in the exchange offer. In consideration for issuing the new notes as contemplated by this prospectus, we will receive outstanding notes in a like principal amount. The form and terms of the new notes are identical in all respects to the form and terms of the outstanding notes, except the new notes do not include certain transfer restrictions. Outstanding notes surrendered in exchange for the new notes will be retired and cancelled and will not be reissued. Accordingly, the issuance of the new notes will not result in any change in our outstanding indebtedness.

BUSINESS

We are the second largest provider of electrical contracting services in the United States. We are also a growing provider of solutions in the data communications and utilities markets. We provide a broad range of services including designing, building and maintaining electrical, data communications and utilities systems for commercial, industrial and residential customers.

Additional information concerning our company is included in our reports and other documents incorporated by reference in this prospectus. See "Where You Can Find More Information."

DESCRIPTION OF THE NEW NOTES

The new notes will be issued, and the outstanding notes were issued, pursuant to an indenture dated as of May 29, 2001 (the "Indenture") among the Company, as issuer, the Subsidiaries, as guarantors, and State Street Bank and Trust Company, as trustee (the "Trustee"). The terms of the new notes include those stated in the Indenture and those made part of the Indenture by the Trust Indenture Act of 1939, as amended (the "Trust Indenture Act"). The definitions of certain capitalized terms used in the following summary are set forth below under "--Certain Definitions." References to the "notes" in this section of the prospectus include both the outstanding notes and the new notes.

The following description is a summary of the material provisions of the Indenture. It does not restate that agreement in its entirety. The Company urges Holders to read the Indenture because it, and not this description, defines the rights of Holders of these notes. The Company has filed the Indenture as an exhibit to the registration statement which includes this prospectus.

If the exchange offer contemplated by this prospectus (the "Exchange Offer") is consummated, Holders of outstanding notes who do not exchange those notes for new notes in the Exchange Offer will vote together with Holders of new notes for all relevant purposes under the Indenture. In that regard, the Indenture requires that certain actions by the Holders thereunder (including acceleration following an Event of Default) must be taken, and certain rights must be exercised, by specified minimum percentages of the aggregate principal amount of the outstanding securities issued under the Indenture. In determining whether Holders of the requisite percentage in principal amount have given any notice, consent or waiver or taken any other action permitted under the Indenture, any outstanding notes that remain outstanding after the Exchange Offer will be aggregated with the new notes, and the Holders of such outstanding notes and the new notes will vote together as a single series for all such purposes. Accordingly, all references herein to specified percentages in aggregate principal amount of the notes outstanding shall be deemed to mean, at any time after the Exchange Offer is consummated, such percentages in aggregate principal amount of the outstanding notes and the new notes then outstanding.

GENERAL

The new notes will be unsecured senior subordinated obligations of the company initially limited to \$125.0 million aggregate principal amount and will be guaranteed by each of the Guarantors (which will initially be all of the subsidiaries) on a senior subordinated basis as described below. The Indenture will permit the company to issue an unlimited amount of notes, subject to compliance with the terms of the covenants described under "-- Material Covenants -- Limitation on Indebtedness."

The new notes will be issued only in registered form without coupons, in denominations of \$1,000 and integral multiples thereof. Principal of, premium, if any, and interest on the notes will be payable, and the notes will be transferable, at the corporate trust office or agency of the Trustee in the City of New York maintained for such purposes. In addition, interest may be paid at our option on any notes not issued in global form by check mailed to the person entitled thereto as shown on the security register subject to the right of any holder of notes in the principal amount of \$500,000 or more to request payment by wire transfer. No service charge will be made for any transfer, exchange or redemption of notes, except in certain circumstances for any tax or other governmental charge that may be imposed in connection therewith. The notes may be issued with original issue discount for federal income tax purposes. The notes will be substantially identical to the 9 3/8% senior subordinated notes due 2009 that we issued in 1999, except that this series of notes will be issued under a different indenture.

MATURITY, INTEREST AND PRINCIPAL

The notes will mature on February 1, 2009. Interest on the notes will accrue at the rate of 9 3/8% per annum and will be payable semi-annually in arrears on each February 1 and August 1 commencing August 1, 2001, to the holders of record of notes at the close of business on the January 15 and July 15, respectively, immediately preceding such interest payment date. Interest on the notes will accrue from the most recent date to which interest has been paid or, if no interest has been paid, from the original date of issuance of the outstanding notes (the "Issue Date"). Interest will be computed on the basis of a 360-day year comprised of twelve 30-day months.

OPTIONAL REDEMPTION

Optional Redemption. The notes will be redeemable at our option, in whole or in part, at any time on or after February 1, 2004, at the redemption prices (expressed as percentages of principal amount) set forth below, plus accrued and unpaid interest, if any, to the redemption date, if redeemed during the 12-month period beginning February 1 of the years indicated below:

YEAR			REDEMPTION PRICE
2004			104.688%
2005			103.125%
2006			101.563%
2007	and	thereafter	100.000%

In addition, at any time, or from time to time, on or prior to February 1, 2002, we may, at our option, use the net cash proceeds of one or more Qualified Equity Offerings to redeem up to an aggregate of 35% of the principal amount of the notes originally issued, at a redemption price equal to 109.375% of the principal amount thereof plus accrued and unpaid interest, if any, thereon to the redemption date; provided that at least 65% of the originally issued principal amount of notes remains outstanding immediately after the occurrence of such redemption. In order to effect the foregoing redemption with the proceeds of any Qualified Equity Offering, we must consummate such redemption not later than 60 days after the closing of any such Qualified Equity Offering.

Selection and Notice. In the event that less than all of the notes are to be redeemed at any time, selection of notes for redemption will be made by the Trustee in compliance with the requirements of the principal national securities exchange, if any, on which the notes are listed or, if the notes are not then listed on a national securities exchange, on a pro rata basis, by lot or by such method as the Trustee shall deem fair and appropriate (subject to the rules of DTC); provided, however, that notes shall only be redeemable in principal amounts of \$1,000 or an integral multiple of \$1,000. Notice of redemption shall be mailed by first-class mail at least 30 but not more than 60 days before the redemption date to each holder of notes to be redeemed at its registered address. If any note is to be redeemed in part only, the notice of redemption that relates to such note shall state the portion of the principal amount thereof to be redeemed. A new note in a principal amount equal to the unredeemed portion of the original note will be issued in the name of the holder thereof upon surrender for cancellation of the original note. On and after the redemption date, interest will cease to accrue on notes or portions thereof called for redemption, unless we default in the payment of the redemption price.

SINKING FUND

The notes will not be entitled to the benefit of any mandatory sinking fund.

VOTING

The Indenture will provide that the holders of the outstanding notes and the new notes will vote and consent together on all matters (to which such holders are entitled to vote or consent) as one class and that none of the holders of the outstanding notes and the new notes shall have the right to vote or consent as a separate class on any matter (to which such holders are entitled to vote or consent).

CHANGE OF CONTROL

Upon the occurrence of a Change of Control we shall be obligated to make an offer to purchase (a "Change of Control Offer"), and shall purchase, on a business day (the "Change of Control Purchase Date") not more than 60 nor less than 30 days following the mailing of notice of such Change of Control Offer as described below, all of the then outstanding notes tendered at a purchase price in cash (the "Change of Control Purchase Price") equal to 101% of the principal amount thereof plus accrued and unpaid interest, if any, thereon to the Change of Control Offer and not withdrawn. The Change of Control Offer will be required to remain open for at least 20 business days.

In order to effect such Change of Control Offer, we shall, not later than the 30th day after the Change of Control, mail to each holder of notes notice of the Change of Control Offer, which notice shall govern the terms of the Change of Control Offer and shall state, among other things, the procedures that holders of notes must follow to accept the Change of Control Offer.

If a Change of Control Offer is made, there can be no assurance that we will have available funds sufficient to pay the Change of Control Purchase Price for all of the notes that might be delivered by holders of notes seeking to accept the Change of Control Offer. In addition, there can be no assurance that our debt instruments will permit such offer to be made. The Credit Facility does not permit us to make a Change of Control Offer and, in order to make such offer, we would be required to pay off the Credit Facility in full or seek a waiver from the lenders under the Credit Facility to allow us to make the Change of Control Offer. The occurrence of a Change of Control is also an event of default under the Credit Facility and would entitle the lenders thereunder to accelerate all amounts owing under the Credit Facility. Any future credit agreements or other agreements relating to Senior Indebtedness to which we become a party may contain similar restrictions and provisions. Moreover, the exercise by the holders of their rights to require us to repurchase the notes could cause a default under such indebtedness, even if the Change of Control itself does not, due to the financial effect of such repurchase on us. Failure to make a Change of Control Offer, even if prohibited by our debt instruments, would constitute a default under the Indenture. See "Risk Factors -- Possible Inability to Repurchase Notes upon Change of Control.'

We shall not be required to make a Change of Control Offer upon a Change of Control if a third party makes the Change of Control Offer at the same purchase price, at the same time and otherwise in compliance with the requirements applicable to a Change of Control Offer made by the company and purchases all notes validly tendered and not withdrawn under such Change of Control Offer.

The provisions of the Indenture may not afford note holders protection in the event of a highly leveraged transaction, reorganization, restructuring, merger or similar transaction involving the company if such transaction is not a transaction defined as a "Change of Control." The existence of a holder's right to require, subject to certain conditions, the company to repurchase the notes upon a Change of Control may deter a third party from acquiring us in a transaction that constitutes, or results in, a Change of Control.

The use of the term "all or substantially all" in provisions of the Indenture such as clause (b) of the definition of "Change of Control" and under "-- Consolidation, Merger, Sale of Assets, Etc." has no clearly established meaning under New York law (which will govern the Indenture) and has been the subject of limited judicial interpretation in only a few jurisdictions. Accordingly, there may be a degree of uncertainty in ascertaining whether any particular transaction would involve a disposition of "all or substantially all" of the assets of a person, which uncertainty should be considered by prospective purchasers of notes.

We will comply with Rule 14e-1 under the Exchange Act and any other securities laws and regulations thereunder, to the extent such laws or regulations are applicable, in the event that a Change of Control occurs and we are required to purchase notes as described above, and any violation of the provisions of the Indenture relating to such Offer to Purchase occurring as a result of such compliance shall not be deemed a Default or an Event of Default.

SUBORDINATION

The indebtedness evidenced by the notes will be subordinated in right of payment, to the extent set forth in the Indenture, to the prior payment in full in cash of all Senior Indebtedness (including Indebtedness under the Credit Facility). The notes will be unsecured senior subordinated indebtedness of the company ranking senior in right of payment to all existing and future Subordinated Indebtedness of the company. The Credit Facility provides that the subordination provisions of the notes may not be modified or amended without the prior written consent of the lenders under the Credit Facility. The notes will rank equal to right of payment to the \$150 million in aggregate principal amount of 9 3/8% senior subordinated notes issued by the company in 1999.

The Indenture will provide that in the event of (a) any insolvency or bankruptcy case or proceeding, or any receivership, liquidation, reorganization or other similar case or proceeding in connection therewith, relating to the company or its assets, (b) any liquidation, dissolution or other winding-up of the company, whether voluntary or involuntary, or (c) any assignment for the benefit of creditors or other marshaling of assets or liabilities of the company, all Senior Indebtedness (including, in the case of Designated Senior Indebtedness, any interest accruing subsequent to the filing of a petition for bankruptcy regardless of whether such interest is an allowed claim in the bankruptcy proceeding) must be paid in full in cash before any direct or indirect payment (other than payments from trusts previously created pursuant to the provisions described under "-- Legal Defeasance or Covenant Defeasance of Indenture") by or on behalf of the company of any kind or character (excluding Permitted Junior Securities) may be made on account of the principal of, premium, if any, or interest on, or the purchase, redemption or other acquisition of, the notes.

During the continuance of any default in the payment of principal, premium, if any, or interest on any Senior Indebtedness, when the same becomes due, and after receipt by the Trustee and the company from representatives of holders of such Senior Indebtedness of written notice of such default, no direct or indirect payment (other than payments from trusts previously created pursuant to the provisions described under "-- Legal Defeasance or Covenant Defeasance of Indenture") by or on behalf of the company of any kind or character (excluding Permitted Junior Securities) may be made on account of the principal of, premium, if any, or interest on, or the purchase, redemption or other acquisition of, the notes unless and until such default has been cured or waived or has ceased to exist or such Senior Indebtedness shall have been paid in full in cash or indefeasibly discharged, after which the company shall promptly resume making any and all required payments in respect of the notes, including any missed payments.

In addition, during the continuance of any other default with respect to any Designated Senior Indebtedness that permits, or would permit with the passage of time or the giving of notice or both, the maturity thereof to be accelerated (a "Non-payment Default") and upon the earlier to occur of (a) receipt by the Trustee and the company from the representatives of holders of such Designated Senior Indebtedness of a written notice of such Non-payment Default or (b) if such Non-payment Default results from the acceleration of the maturity of the notes, the date of such acceleration, no direct or indirect payment (other than payments from trusts previously created pursuant to the provisions described under "-- Legal Defeasance or Covenant Defeasance of Indenture") of any kind or character (excluding Permitted Junior Securities) may be made on account of the principal of, premium, if any, or interest on, or the purchase, redemption, or other acquisition of, the notes for the period specified below (the "Payment Blockage Period").

The Payment Blockage Period shall commence upon the receipt of notice of a Non-payment Default by the Trustee and the company from the representatives of holders of Designated Senior Indebtedness or the date of the acceleration referred to in clause (b) of the preceding paragraph, as the case may be, and shall end on the earliest to

occur of the following events: (i) 179 days have elapsed since the receipt of such notice or the date of the acceleration referred to in clause (b) of the preceding paragraph (provided the maturity of such Designated Senior Indebtedness shall not theretofore have been accelerated), (ii) such default is cured or waived or ceases to exist or such Designated Senior Indebtedness is paid in full in cash or indefeasibly discharged, or (iii) such Payment Blockage Period has been terminated by written notice to the company or the Trustee from the representatives of holders of Designated Senior Indebtedness initiating such Payment Blockage Period, after which the company shall promptly resume making any and all required payments in respect of the notes, including any missed payments. Only one Payment Blockage Period with respect to the notes may be commenced within any 360 consecutive day period. No Non-payment Default with respect to Designated Senior Indebtedness that existed or was continuing on the date of the commencement of any Payment Blockage Period with respect to the Designated Senior Indebtedness initiating such Payment Blockage Period will be, or can be, made the basis for the commencement of a second Payment Blockage Period, whether or not within a period of 360 consecutive days, unless such default has been cured or waived for a period of not less than 90 consecutive days. In no event will a Payment Blockage Period extend beyond 179 days from the date of the receipt by the Trustee of the notice or the date of the acceleration initiating such Payment Blockage Period, and there must be a 181 consecutive day period in any 360 day period during which no Payment Blockage Period is in effect.

If we fail to make any payment on the notes when due on account of the payment blockage provisions referred to above, such failure would constitute an Event of Default under the Indenture and would enable the holders of the notes to accelerate the maturity thereof. See "-- Events of Default."

By reason of such subordination, in the event of liquidation or insolvency, creditors of the company who are holders of Senior Indebtedness may recover more, ratably, than the holders of the notes, and funds which would be otherwise payable to the holders of the notes will be paid to the holders of Senior Indebtedness to the extent necessary to pay the Senior Indebtedness in full, and we may be unable to meet its obligations fully with respect to the notes.

On a pro forma basis after giving effect to the sale of the notes offered hereby and the application of the estimated net proceeds, the company and the Guarantors would have had, without duplication, \$1.9 million of Senior Indebtedness and Guarantor Senior Indebtedness (as defined) outstanding as of March 31, 2001. The Indenture will limit, but not prohibit, the incurrence by the company of additional Indebtedness which is senior to the notes and will prohibit the incurrence by the company of Indebtedness which is subordinated in right of payment to any other Indebtedness of the company and senior in right of payment to the notes. As of March 31, 2001, after giving effect to this offering and our new credit facility, approximately \$144.7 million would have been available for additional borrowing under our new credit facility.

"Designated Senior Indebtedness" means (i) all Senior Indebtedness under the Credit Facility and (ii) any other Senior Indebtedness which (a) at the time of the determination is equal to or greater than \$25 million in aggregate principal amount and (b) is specifically designated by the company in the instrument evidencing such Senior Indebtedness as "Designated Senior Indebtedness."

"Permitted Junior Securities" means Capital Stock of the company or debt securities that are subordinated to all Senior Indebtedness (and any debt securities issued in exchange for Senior Indebtedness) to at least the same extent as the notes are subordinated to Senior Indebtedness.

"Senior Indebtedness" means the principal of, premium, if any, and interest on any Indebtedness of the company, whether outstanding on the Issue Date or thereafter created, incurred or assumed, unless, in the case of any particular Indebtedness, the instrument creating or evidencing the same or pursuant to which the same is outstanding expressly provides that such Indebtedness shall not be senior in right of payment to the notes. Without limiting the generality of the foregoing, (x) "Senior Indebtedness" shall include the principal of, premium, if any, and interest on all obligations of every nature of the company from time to time owed to the lenders under the Credit Facility, including, without limitation, principal of and interest on, and all fees, indemnities and expenses payable under, the Credit Facility, and (y) in the case of Designated Senior Indebtedness, "Senior Indebtedness" shall include interest accruing thereon subsequent to the occurrence of any Event of Default specified in clause (vii) or (viii) under "-- Events of Default" relating to the company, whether or not the claim for such interest is allowed under any applicable Bankruptcy Code. Notwithstanding the foregoing, "Senior Indebtedness" shall not include (a) Indebtedness evidenced by the notes, (b) Indebtedness that is expressly subordinate or junior in right of payment to any other Indebtedness of the company, (c) Indebtedness which, when incurred and without respect to any election under Section 1111(b) of Title 11, United States Code, is by its terms without recourse to the company, (d) Indebtedness which is represented by Redeemable Capital Stock, (e) to the extent it constitutes Indebtedness, any liability for federal, state, local or other taxes owed or owing by the company, (f) Indebtedness of the company to a Subsidiary of the company or any other Affiliate of the company or any of such Affiliate's Subsidiaries, and (g) that portion of any Indebtedness which is incurred by the company in violation of the Indenture.

GUARANTEES

Each of our Subsidiaries (each, a "Guarantor") will fully and unconditionally guarantee, on a senior subordinated basis, jointly and severally, to each holder of notes and the Trustee, the full and prompt performance of the company's obligations under the Indenture and the notes, including the payment of principal of and interest on the notes. The Guarantees will be subordinated to Guarantor Senior Indebtedness on the same basis as the notes are subordinated to Senior Indebtedness.

The obligations of each Guarantor are limited to the maximum amount which, after giving effect to all other contingent and fixed liabilities of such Guarantor and after giving effect to any collections from or payments made by or on behalf of any other Guarantor in respect of the obligations of such other Guarantor under its Guarantee or pursuant to its contribution obligations under the Indenture, will result in the obligations of such Guarantor under the Guarantee not constituting a fraudulent conveyance or fraudulent transfer under federal or state law. See "Risk Factors -- Fraudulent Transfer Considerations."

Each Guarantor that makes a payment or distribution under a Guarantee shall be entitled to a contribution from each other Guarantor in an amount pro rata, based on the net assets of each Guarantor, determined in accordance with GAAP.

Each Guarantor may consolidate with or merge into or sell its assets to the company or another Guarantor without limitation, or with other persons upon the terms and conditions set forth in the Indenture. See "-- Consolidation, Merger, Sale of Assets, Etc." In the event all or substantially all of the assets or the Capital Stock of a Guarantor is sold and the sale complies with the provisions set forth in "-- Material Covenants -- Limitation on Disposition of Proceeds of Asset Sales" or the Guarantor is designated an Unrestricted Subsidiary in accordance with the terms of the Indenture, then the Guarantor's Guarantee will be automatically and unconditionally discharged and released.

"Guarantor Senior Indebtedness" of a Guarantor means the principal of, premium, if any, and interest on any Indebtedness of such Guarantor, whether outstanding on the Issue Date or thereafter created, incurred or assumed, unless, in the case of any particular Indebtedness, the instrument creating or evidencing the same or pursuant to which the same is outstanding expressly provides that such Indebtedness shall not be senior in right of payment to such Guarantor's Guarantee. Without limiting the generality of the foregoing, (x) "Guarantor Senior Indebtedness" shall include the principal of, premium, if if anv. and interest on all obligations of every nature of such Guarantor from time to time owed to the lenders under the Credit Facility, including, without limitation, principal of and interest on, and all fees, indemnities and expenses payable under, the Credit Facility, and (y) in the case of amounts owing under the Credit Facility and guarantees of Designated Senior Indebtedness, "Guarantor Senior Indebtedness" shall include interest accruing thereon subsequent to the occurrence of any Event of Default specified in clause (vii) or (viii) under Events of Default" relating to such Guarantor, whether or not the claim for such interest is allowed under any applicable Bankruptcy Code. Notwithstanding the foregoing, "Guarantor Senior Indebtedness" shall not include (a) Indebtedness evidenced by the notes or the Guarantees, (b) Indebtedness that is expressly subordinate or junior in right of payment to any other Indebtedness of such Guarantor, (c) Indebtedness which, when incurred and without respect to any election under Section 1111(b) of Title 11, United States Code, is by its terms without recourse to such Guarantor, (d) Indebtedness which is represented by Redeemable Capital Stock, (e) to the extent it constitutes Indebtedness, any liability for federal, state, local or other taxes owed or owing by such Guarantor, (f) Indebtedness of such Guarantor to the company or a Subsidiary of the company or any other Affiliate of the company or any of such Affiliate's Subsidiaries, and (g) that portion of any Indebtedness which is incurred by such Guarantor in violation of the Indenture.

We are structured as a holding company and substantially all of our assets are held and substantially all of our operations are conducted by our subsidiaries. There are currently no significant restrictions on our ability to obtain funds from our subsidiaries by dividend or loan. Separate financial statements of the Guarantors are not included herein because the Guarantors will be jointly and severally liable with respect to the company's obligations pursuant to the notes, and the aggregate net assets, earnings and equity of the Guarantors and the company are substantially equivalent to the net assets, earnings and equity of the company on a consolidated basis.

MATERIAL COVENANTS

Limitation on Indebtedness. Our company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create, incur, issue, assume, guarantee or in any manner become directly or indirectly liable, contingently or otherwise (in each case, to "incur"), for the payment of any Indebtedness (including any Acquired Indebtedness) other than Permitted Indebtedness; provided, however, that the company and any Guarantor will be permitted to incur Indebtedness (including Acquired Indebtedness), if (A) the Consolidated Fixed Charge Coverage Ratio of the company is at least 2.0 to 1 and (B) no Default or Event of Default would occur or be continuing. As of March 31, 2001, after giving proforma effect to this offering, the Consolidated Fixed Charge Coverage Ratio of our company would be 4.4.

Limitation on Restricted Payments. Our company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly:

(a) declare or pay any dividend or make any other distribution or payment on or in respect of Capital Stock of the company or any of its Restricted Subsidiaries or make any payment to the direct or indirect holders (in their capacities as such) of Capital Stock of the company or any of its Restricted Subsidiaries (other than dividends or distributions payable solely in Capital Stock of the company (other than Redeemable Capital Stock) or in options, warrants or other rights to purchase Capital Stock of the company (other than Redeemable Capital Stock)) (other than the declaration or payment of dividends or other distributions to the extent declared or paid to the company or any Restricted Subsidiary),

(b) purchase, redeem or otherwise acquire or retire for value any Capital Stock of the company or any of its Restricted Subsidiaries or any options, warrants or other rights to purchase any such Capital Stock (other than any such securities owned by the company or a Restricted Subsidiary),

(c) make any principal payment on, or purchase, defease, redeem or otherwise acquire or retire for value, prior to any scheduled maturity, scheduled repayment, scheduled sinking fund payment or other Stated Maturity, any Subordinated Indebtedness (other than any such Subordinated Indebtedness owed to the company or a Restricted Subsidiary), or

(d) make any Investment (other than any Permitted Investment) in any person

(such payments or Investments described in the preceding clauses (a), (b), (c) and (d) are collectively referred to as "Restricted Payments"), unless, after giving effect to the proposed Restricted Payment (the amount of any such Restricted Payment, if other than cash, shall be the Fair Market Value of the asset(s) proposed to be transferred by the company or such Restricted Subsidiary, as the case may be, pursuant to such Restricted Payment), (A) no Default or Event of Default shall have occurred and be continuing, (B) after giving pro forma effect to such Restricted Payment, the company would be able to incur \$1.00 of additional Indebtedness (other than Permitted Indebtedness) in accordance with the "Limitation on Indebtedness" covenant described above and (C) the aggregate amount of all Restricted Payments declared or made from January 28, 1999 would not exceed the sum of:

(1) 50% of the aggregate Consolidated Net Income of the company accrued on a cumulative basis during the period beginning on January 1, 1999 and ending on the last day of the fiscal quarter of the company ending immediately prior to the date of such proposed Restricted Payment (or, if such aggregate cumulative Consolidated Net Income of the company for such period shall be a loss, minus 100% of such loss); (2) the aggregate net cash proceeds received by the company as capital contributions to the company after January 28, 1999 and which constitute shareholders' equity of the company in accordance with GAAP;

(3) the aggregate net cash proceeds received by the company from the issuance or sale of Capital Stock (excluding Redeemable Capital Stock) of the company to any person (other than to a Subsidiary of the company) after January 28, 1999;

(4) the aggregate net cash proceeds received by the company from any person (other than a Subsidiary of the company) upon the exercise of any options, warrants or rights to purchase shares of Capital Stock (other than Redeemable Capital Stock) of the company after January 28, 1999;

(5) the aggregate net cash proceeds received after January 28, 1999 by the company from any person (other than a Subsidiary of the company) for debt securities that have been converted into or exchanged for Capital Stock of the company (other than Redeemable Capital Stock) to the extent such debt securities were originally sold for cash, plus the aggregate amount of cash received by the company (other than from a Subsidiary of the company) at the time of such conversion or exchange;

(6) to the extent not otherwise included in the company's Consolidated Net Income, in the case of the disposition or repayment of any Investment constituting a Restricted Payment after January 28, 1999, an amount equal to the lesser of the return of capital with respect to such Investment and the initial amount of such Investment, in either case, less the cost of the disposition of such Investment;

(7) so long as the Designation thereof was treated as a Restricted Payment made after January 28, 1999, with respect to any Unrestricted Subsidiary that has been redesignated as a Restricted Subsidiary after January 28, 1999 in accordance with "-- Limitation on Designations of Unrestricted Subsidiaries" below, the Fair Market Value of the company's interest in such Subsidiary at the time of such redesignation; provided that such amount shall not in any case exceed the Designation Amount with respect to such Restricted Subsidiary upon its Designation; and

(8) \$10 million.

For purposes of the preceding clause (C)(4), the value of the aggregate net proceeds received by the company upon the issuance of Capital Stock upon the exercise of options, warrants or rights will be the net cash proceeds received upon the issuance of such options, warrants or rights plus the incremental amount received by the company upon the exercise thereof. As of March 31, 2001, under the preceding clause (C), we had approximately 35.1 million available for future Restricted Payments, including amounts available under clause (8) above.

None of the foregoing provisions will prohibit, so long as, in the case of clause (v) below, there is no Default or Event of Default continuing, (i) the payment of any dividend or distribution within 60 days after the date of its declaration, if at the date of declaration such payment would be permitted by the first paragraph of this covenant; (ii) the redemption, repurchase or other acquisition or retirement of any shares of any class of Capital Stock of the company in exchange for, or out of the net cash proceeds of a substantially concurrent issue and sale of, other shares of Capital Stock of the company (other than Redeemable Capital Stock) to any person (other than to a Subsidiary from clause (C) of the first paragraph of this covenant; (iii) any redemption, repurchase or other acquisition or retirement of Subordinated Indebtedness of the company in exchange for, or out of the net cash proceeds of a substantially concurrent issue and sale of, (1) Capital Stock (other than Redeemable Capital Stock) of the company to any person (other than to a Subsidiary of the company); provided, however, that any such net cash proceeds are excluded from clause (C) of the first paragraph of this covenant; or (2) other Subordinated Indebtedness of the company which (w) has no scheduled principal payment prior to the 91st day after the Maturity Date, (x) has an Average Life to Stated Maturity greater than the remaining Average Life to Stated Maturity of the notes and (y) is subordinated to the notes to at least the same extent as the Subordinated Indebtedness so purchased, exchanged, redeemed, acquired or retired; (iv) payments to purchase Capital Stock of the company from management or employees of the company or any of its Subsidiaries, or their authorized representatives, upon the death, disability or termination of employment of such employees, in aggregate amounts under this clause (iv) not to exceed \$3 million in any fiscal year of the company; (v) payments relating to Permitted Founder Stock Repurchases so long as the Consolidated Fixed Charge Coverage Ratio of the

company is at least 3.0 to 1; (vi) cash payments in lieu of fractional shares issuable as dividends on preferred securities of the company or any of its Restricted Subsidiaries, in aggregate amounts under this clause (vi) not to exceed \$20,000 in any fiscal year of the company; (vii) repurchases of Capital Stock deemed to occur upon exercise of stock options if such Capital Stock represents a portion of the exercise price of such options; and (viii) the payment of the redemption price of rights issued pursuant to any shareholders' rights plan not in excess of \$0.05 per right and not in excess of \$1,000,000 in the aggregate. Any payments made pursuant to clause (i) of this paragraph shall be taken into account in calculating the amount of Restricted Payments made from and after January 28, 1999.

In computing Consolidated Net Income of the company under clause (C)(1) of the first paragraph of this covenant, (1) the company shall use audited financial statements for the portions of the relevant period for which audited financial statements are available on the date of determination and unaudited financial statements and other current financial data based on the books and records of the company for the remaining portion of such period and (2) the company shall be permitted to rely in good faith on the financial statements and other financial data derived from the books and records of the company that are available on the date of determination. If the company makes a Restricted Payment which, at the time of the making of such Restricted Payment would in the good faith determination of the company be permitted under the requirements of the Indenture, such Restricted Payment shall be deemed to have been made in compliance with the Indenture notwithstanding any subsequent adjustments made in good faith to the company's financial statements affecting Consolidated Net Income of the company for any period.

Limitation on Liens. Our company will not, and will not permit any of its Restricted Subsidiaries to, create, incur, assume or suffer to exist any Liens of any kind securing Indebtedness upon any of its property or assets, or any proceeds therefrom, unless the notes are equally and ratably secured (except that Liens securing Subordinated Indebtedness shall be expressly subordinate to Liens securing the notes to the same extent such Subordinated Indebtedness is subordinate to the notes), except for (a) Liens securing Senior Indebtedness and Guarantor Senior Indebtedness; (b) Liens securing the notes; (c) Liens securing Indebtedness which is incurred to refinance Indebtedness which has been secured by a Lien (other than a Lien in favor of the company or a Restricted Subsidiary) permitted under the Indenture and which has been incurred in accordance with the provisions of the Indenture; provided, however, that such Liens do not extend to or cover any property or assets of the company or any of its Restricted Subsidiaries not securing the Indebtedness so refinanced; and (d) Permitted Liens.

Disposition of Proceeds of Asset Sales. Our company will not, and will not permit any of its Restricted Subsidiaries to, make any Asset Sale unless (a) the company or such Restricted Subsidiary, as the case may be, receives consideration at the time of such Asset Sale at least equal to the Fair Market Value of the shares or assets sold or otherwise disposed of and (b) at least 75% January 28, 1999 on a cumulative basis, consists of cash or Cash Equivalents; provided, however, that the amount of any Indebtedness (as shown on the most recent balance sheet of the company or such Restricted Subsidiary) of the company or such Restricted Subsidiary that is assumed by the transferee of such assets as a result of which the company and its Restricted Subsidiaries are no longer liable thereon, and any securities, notes or other obligations received by the company or such Restricted Subsidiary from such transferee that are converted within 60 days into cash or Cash Equivalents (to the extent of the cash or Cash Equivalents received) shall be deemed to be cash for the purposes of this provision. To the extent that the Net Cash Proceeds, or portion thereof, of any Asset Sale are not applied to repay, and permanently reduce the commitments under Senior Indebtedness or Guarantor Senior Indebtedness in accordance with the terms thereof, the company or such Restricted Subsidiary, as the case may be, may apply the Net Cash Proceeds, or portion thereof, from such Asset Sale, within 360 days of such Asset Sale, to an investment in properties and assets that replace the properties and assets that were the subject of such Asset Sale or in properties and assets that (as determined in good faith by the Board of Directors of the company or the Restricted Subsidiary, as the case may be) are used or useful in the business of the company and its Restricted Subsidiaries conducted at such time or in businesses reasonably related thereto or in Capital Stock of a person, the principal portion of whose assets consist of such property or assets ("Replacement Assets"). Any Net Cash Proceeds from any Asset Sale that has occurred on or after January 28, 1999, that are neither used to repay, and permanently reduce the commitments under, Senior Indebtedness or Guarantor Senior Indebtedness in accordance with the terms thereof nor invested in Replacement Assets within such 360-day period will constitute "Excess Proceeds" subject to disposition as provided below. As of March 31, 2001, we have no Excess Proceeds.

34

When the aggregate amount of Excess Proceeds equals or exceeds \$10 million, the company shall make an offer to purchase, from all holders of the notes and any then outstanding Pari Passu Indebtedness required to be repurchased or repaid on a permanent basis in connection with an Asset Sale, an aggregate principal amount of notes and any then outstanding Pari Passu Indebtedness equal to such Excess Proceeds as follows:

(i) (A) We shall make an offer to purchase (an "Asset Sale Offer") from all holders of the notes in accordance with the procedures set forth in the Indenture the maximum principal amount (expressed as a multiple of \$1,000) of notes that may be purchased out of an amount (the "Asset Sale Offer Amount") equal to the product of such Excess Proceeds, multiplied by a fraction, the numerator of which is the outstanding principal amount of the notes and the denominator of which is the sum of the outstanding principal amount of the notes and such Pari Passu Indebtedness, if any (subject to proration in the event such amount is less than the aggregate Offered Price (as defined herein) of all notes tendered), and (B) to the extent required by such Pari Passu Indebtedness and provided there is a permanent reduction in the principal amount of such Pari Passu Indebtedness, the company shall make an offer to purchase Pari Passu Indebtedness (a "Pari Passu Offer") in an amount (the "Pari Passu Indebtedness Amount") equal to the excess of the Excess Proceeds over the Asset Sale Offer Amount;

(ii) The offer price for the notes shall be payable in cash in an amount equal to 100% of the principal amount of the notes tendered pursuant to an Asset Sale Offer, plus accrued and unpaid interest, if any, to the date such Asset Sale Offer is consummated (the "Offered Price"), in accordance with the procedures set forth in the Indenture. To the extent that the aggregate Offered Price of the notes tendered pursuant to an Asset Sale Offer is less than the Asset Sale Offer Amount relating thereto or the aggregate amount of the Pari Passu Indebtedness that is purchased or repaid pursuant to the Pari Passu Offer is less than the Pari Passu Indebtedness that curchased or repaid pursuant (such shortfall constituting an "Asset Sale Deficiency"), the company may use such Asset Sale Deficiency for general corporate purposes, subject to the limitations described above under the caption "-- Limitation on Restricted Payments;" and

(iii) If the aggregate Offered Price of notes validly tendered and not withdrawn by holders thereof exceeds the Asset Sale Offer Amount, notes to be purchased will be selected on a pro rata basis. Upon completion of such Asset Sale Offer and Pari Passu Offer, the amount of Excess Proceeds shall be reset to zero.

Our company will comply with Rule 14e-1 under the Exchange Act and any other securities laws and regulations thereunder, to the extent such laws and regulations are applicable, in the event that an Asset Sale occurs and the company is required to purchase notes as described above, and any violation of the provisions of the Indenture relating to such Asset Sale Offer occurring as a result of such compliance shall not be deemed a Default or an Event of Default.

Limitation on Issuances and Sales of Restricted Subsidiary Stock. Our company (i) will not permit any Restricted Subsidiary to issue any Capital Stock (other than to the company or a Restricted Subsidiary) and (ii) will not permit any Person (other than the company and/or one or more Restricted Subsidiaries) to own any Capital Stock of any Restricted Subsidiary; provided, however, that this covenant shall not prohibit (1) the issuance and sale of all, but not less than all, of the issued and outstanding Capital Stock of any Restricted Subsidiary owned by the company or any of its Restricted Subsidiaries in compliance with the other provisions of the Indenture, or (2) the ownership by directors of directors' qualifying shares or the ownership by foreign nationals of Capital Stock of any Restricted Subsidiary, to the extent mandated by applicable law.

Limitation on Transactions with Affiliates. Our company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, enter into any transaction or series of related transactions (including, without limitation, the sale, transfer, disposition, purchase, exchange or lease of assets, property or services) with, or for the benefit of, any of its Affiliates, except (a) on terms that are no less favorable to the company or such Restricted Subsidiary, as the case may be, than those which could have been obtained at the time in a comparable transaction or series of related transactions from persons who are not Affiliates of the company, (b) with respect to a transaction or series of related transactions involving aggregate payments or value equal to or greater than \$5 million, the company shall have delivered an officers' certificate to the Trustee certifying that such transaction or transactions comply with the preceding clause (a) and have been approved by the Board of Directors of the company, and (c) with respect to a transaction or series of related transactions involving aggregate payments or value equal to or greater than \$10

million, the officers' certificate referred to in clause (b) above also includes a certification that such transaction or transactions have been approved by a majority of the Disinterested Members of the Board of Directors of the company or, in the event there are no such Disinterested Members of the Board of Directors, that the company has obtained a written opinion from an independent nationally recognized investment banking firm, accounting firm or appraisal firm, in each case specializing or having a specialty in the type and subject matter of the transaction or series of transactions at issue, which opinion shall be to the effect set forth in clause (a) above or shall state that such transaction or series of related transactions is fair from a financial point of view to the company or such Restricted Subsidiary.

Notwithstanding the foregoing, the restrictions set forth in this covenant shall not apply to (i) transactions between or among the company and its Restricted Subsidiaries, (ii) customary directors' fees, indemnification and similar arrangements, consulting fees, employee salaries, bonuses or employment agreements, compensation or employee benefit arrangements and incentive arrangements with any officer, director or employee of the company or any Restricted Subsidiary entered into in the ordinary course of business, (iii) any dividends made in compliance with "-- Limitation on Restricted Payments" above, (iv) loans and advances to officers, directors and employees of the company or any Restricted Subsidiary made in the ordinary course of business in an aggregate amount not to exceed \$1,000,000 outstanding at any one time, (v) transactions pursuant to agreements in effect on January 28, 1999, (vi) written agreements entered into or assumed in connection with acquisitions of other businesses with persons who were not Affiliates prior to such transactions, or (vii) leases of property or equipment entered into in the ordinary course of business on terms that are substantially similar to those which could have been obtained at the time in a comparable transaction with non-Affiliates.

Limitation on Dividends and Other Payment Restrictions Affecting Restricted Subsidiaries. Our company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create or otherwise cause or suffer to exist or become effective any consensual encumbrance or restriction on the ability of any Restricted Subsidiary to (a) pay dividends, in cash or otherwise, or make any other distributions on or in respect of its Capital Stock to the company or any other Restricted Subsidiary, (b) pay any Indebtedness owed to the company or any other Restricted Subsidiary, (c) make loans or advances to the company or any other Restricted Subsidiary, (d) transfer any of its properties or assets to the company or any other Restricted Subsidiary or (e) guarantee any Indebtedness of the company or any other Restricted Subsidiary, except for such encumbrances or restrictions existing under or by reason of (i) applicable law or any applicable rule, regulation or order, (ii) customary nonassignment provisions of any contract or any lease governing a leasehold interest of the company or any Restricted Subsidiary, (iii) customary restrictions on transfers of property subject to a Lien permitted under the Indenture (including purchase money Liens permitted under the Indenture), (iv) any agreement or other instrument of a person acquired by the company or any Restricted Subsidiary in existence at the time of such acquisition (but not created in contemplation thereof), which encumbrance or restriction is not applicable to any person, or the properties or assets of any person, other than the person, or the property or assets of the person, so acquired, (v) an agreement entered into for the sale or disposition of Capital Stock or assets of a Restricted Subsidiary or an agreement entered into for the sale of specified assets (in either case, so long as such encumbrance or restriction, by its terms, terminates on the earlier of the termination of such agreement or the consummation of such agreement and so long as such restriction applies only to the Capital Stock or assets to be sold), (vi) any agreement in effect on January 28, 1999, (vii) the Indenture and the Guarantees, and (viii) any agreement that amends, extends, refinances, renews or replaces any agreement described in the foregoing clauses; provided that the terms and conditions of any such agreement are not materially less favorable to the holders of the notes with respect to such encumbrances or restrictions than those under or pursuant to the agreement amended, extended, refinanced, renewed or replaced.

Limitation on Designations of Unrestricted Subsidiaries. Our company may designate after the Issue Date any Restricted Subsidiary as an "Unrestricted Subsidiary" under the Indenture (a "Designation") only if:

(i) no Default shall have occurred and be continuing at the time of or after giving effect to such Designation;

(ii) the company would be permitted to make an Investment (other than a Permitted Investment covered by clause (x) of the definition thereof) at the time of Designation (assuming the effectiveness of such Designation) pursuant to the first paragraph of "-- Limitation on Restricted Payments" above in an amount (the "Designation Amount") equal to the Fair Market Value of the company's interest in such Subsidiary on such date; and

(iii) the company would be permitted under the Indenture to incur \$1.00 of additional Indebtedness (other than Permitted Indebtedness) pursuant to the covenant described under "-- Limitation on Indebtedness" at the time of such Designation (assuming the effectiveness of such Designation).

In the event of any such Designation, the company shall be deemed to have made an Investment constituting a Restricted Payment pursuant to the covenant "-- Limitation on Restricted Payments" for all purposes of the Indenture in the Designation Amount.

Our ability to designate a Subsidiary as an Unrestricted Subsidiary allows us to more freely pursue additional financing for future projects. To the extent we participate in projects that require financing in addition to the notes, we may need the ability to create Subsidiaries that can pursue the project financing without being subject to the restrictions of the indenture and the obligations under the guarantees. A Subsidiary that is not subject to the restrictions of the indenture and the guarantees may be able to procure project financing more easily than a Subsidiary that has guaranteed the notes.

We shall not, and shall not cause or permit any Restricted Subsidiary to, at any time (x) provide credit support for or subject any of its property or assets (other than the Capital Stock of any Unrestricted Subsidiary) to the satisfaction of, any Indebtedness of any Unrestricted Subsidiary (including any undertaking, agreement or instrument evidencing such Indebtedness), (y) be directly or indirectly liable for any Indebtedness of any Unrestricted Subsidiary or (z) be directly or indirectly liable for any Indebtedness which provides that the holder thereof may (upon notice, lapse of time or both) declare a default thereon or cause the payment thereof to be accelerated or payable prior to its final Stated Maturity upon the occurrence of a default with respect to any Indebtedness of any Unrestricted Subsidiary (including any right to take enforcement action against such Unrestricted Subsidiary). All Subsidiaries of Unrestricted Subsidiaries shall automatically be deemed to be Unrestricted Subsidiaries.

Our company may revoke any Designation of a Subsidiary as an Unrestricted Subsidiary (a "Revocation") if:

(i) no Default shall have occurred and be continuing at the time of and after giving effect to such Revocation; and

(ii) all Liens and Indebtedness of such Unrestricted Subsidiary outstanding immediately following such Revocation would, if incurred at such time, have been permitted to be incurred under the Indenture.

All Designations and Revocations must be evidenced by Board Resolutions of the company delivered to the Trustee certifying compliance with the foregoing provisions.

Limitation on the Issuance of Subordinated Indebtedness. Our company will not, and will not permit any Guarantor to, directly or indirectly, incur any Indebtedness (including Acquired Indebtedness) that is expressly subordinate or junior in right of payment to any other Indebtedness of the company or such Guarantor and senior in right of payment to the notes or the Guarantee of such Guarantor, as the case may be.

Additional Subsidiary Guarantees. If the company or any of its Restricted Subsidiaries acquires, creates or designates another Restricted Subsidiary organized under the laws of the United States or any possession or territory thereof, any State of the United States or the District of Columbia, then such newly acquired, created or designated Restricted Subsidiary shall, within 30 days after the date of its acquisition, creation or designation, whichever is later, execute and deliver to the Trustee a supplemental indenture in form reasonably satisfactory to the Trustee pursuant to which such Subsidiary shall unconditionally guarantee (on a senior subordinated basis) all of the company's obligations under the notes and the Indenture on the terms set forth in the Indenture; provided, that such Restricted Subsidiary shall not be obligated to become a Guarantor in the manner set forth above if such Restricted Subsidiary is not, either individually or when considered in the aggregate with all othem rRestricted Subsidiaries that are not Guarantors, a Significant Subsidiary. Thereafter, such Restricted Subsidiary shall be a Guarantor for all purposes of the Indenture. The Indenture will also provide that any Restricted Subsidiary that is not a Guarantor shall become a Guarantor in the manner provided above within 30 days of such time as it becomes, either individually or when considered in the aggregate with all other Restricted Subsidiaries that are not Guarantors, a

Significant Subsidiary. Our company at its option may also cause any other Restricted Subsidiary to so become a Guarantor.

Reporting Requirements. For so long as the notes are outstanding, whether or not the company is subject to Section 13(a) or 15(d) of the Exchange Act, or any successor provision thereto, the company shall file with the SEC (if permitted by SEC practice and applicable law and regulations) the annual reports, quarterly reports and other documents which the company would have been required to file with the SEC pursuant to such Section 13(a) or 15(d) or any successor provision thereto if the company were so subject, such documents to be filed with the SEC on or prior to the respective dates (the "Required Filing Dates") by which the company would have been required so to file such documents if the company were so subject. Our company shall also in any event within 15 days after each Required Filing Date (whether or not permitted or required to be filed with the SEC) file with the Trustee, copies of the annual reports, quarterly reports and other documents which the company would be required to file with the SEC if the notes were then registered under the Exchange Act and to make such information available to holders of notes upon request. In addition, if the company is not subject to the reporting requirements of the Exchange Act, for so long as any notes remain outstanding, the company will furnish to the holders of notes and prospective investors, upon their request, the information required to be delivered pursuant to Rule 144A(d)(4) under the Securities Act.

CONSOLIDATION, MERGER, SALE OF ASSETS, ETC.

The Indenture will provide that the company will not, in any transaction or series of transactions, merge or consolidate with or into, or sell, assign, convey, transfer, lease or otherwise dispose of all or substantially all of its properties and assets to, any person or persons, and the company will not permit any of its Restricted Subsidiaries to enter into any such transaction or series of transactions if such transaction or series of transactions, in the aggregate, would result in a sale, assignment, conveyance, transfer, lease or other disposition of all or substantially all of the properties and assets of the company and its Restricted Subsidiaries, on a consolidated basis, to any other person or persons, unless at the time and after giving effect thereto (a) either (i) if the transaction or series of transactions is a merger or consolidation, the company or such Restricted Subsidiary, as the case may be, shall be the surviving person of such merger or consolidation, or (ii) the person formed by such consolidation or into which the company or such Restricted Subsidiary, as the case may be, is merged or to which the properties and assets of the company or such Restricted Subsidiary, as the case may be, are disposed of (any such surviving person or transferee person being the "Surviving Entity") shall be a corporation organized and existing under the laws of the United States of America, any State thereof or the District of Columbia and shall expressly assume by a supplemental indenture executed and delivered to the Trustee, in form satisfactory to the Trustee, all the obligations of the company under the notes, the Indenture and the registration rights agreement, and in each case, the Indenture, the notes and the registration rights agreement shall remain in full force and effect; (b) immediately after giving effect to such transaction or series of transactions on a pro forma basis (including, without limitation, any Indebtedness incurred or anticipated to be incurred in connection with or in respect of such transaction or series of transactions), no Default or Event of Default shall have occurred and be continuing; and (c) except in the case of any merger of the company with any Restricted Subsidiary or any merger of Guarantors (and, in each case, no other persons), the company or the Surviving Entity, as the case may be, after giving effect to such transaction or series of transactions on a pro forma basis on the assumption that the transaction or transactions had occurred on the first day of the period of four fiscal quarters ending immediately prior to the consummation of such transaction or transactions, with the appropriate adjustments with respect to such transaction or transactions being included in such pro forma calculation, could incur \$1.00 of additional Indebtedness (other than Permitted Indebtedness) in accordance with the "Limitation of Indebtedness" covenant described above (assuming a market rate of interest with respect to such additional Indebtedness).

In connection with any consolidation, merger, sale, assignment, conveyance, transfer, lease, assignment or other disposition contemplated hereby, the company shall deliver, or cause to be delivered, to the Trustee, in form and substance reasonably satisfactory to the Trustee, an officers' certificate and an opinion of counsel, each stating that such consolidation, merger, transfer, lease, assignment or other disposition and the supplemental indenture in respect thereof comply with the requirements under the Indenture.

Upon any such consolidation, merger, or any sale, assignment, conveyance, transfer, lease or other disposition in accordance with the immediately preceding paragraphs, the successor person formed by such consolidation or into which the company or a Restricted Subsidiary, as the case may be, is merged or the successor person to which

such sale, assignment, conveyance, transfer, lease or other disposition is made shall succeed to, and be substituted for, and may exercise every right and power of the company under the notes, the Indenture and/or the registration rights agreement, as the case may be, with the same effect as if such successor had been named as the company in the notes, the Indenture and/or the registration rights agreement, as the case may be, and, except in the case of a lease, the company or such Restricted Subsidiary shall be released and discharged from its obligations thereunder.

The Indenture will provide that for all purposes of the Indenture and the notes (including the provision of this covenant and the covenants described in "-- Material Covenants -- Limitation on Indebtedness," "-- Limitation on Restricted Payments," and "-- Limitation on Liens"), Subsidiaries of any surviving person shall, upon such transaction or series of related transactions, become Restricted Subsidiaries unless and until designated Unrestricted Subsidiaries pursuant to and in accordance with "-- Limitation on Designations of Unrestricted Subsidiaries" and all Indebtedness, and all Liens on property or assets, of the company and the Restricted Subsidiaries in existence immediately after such transaction or series of related transactions will be deemed to have been incurred upon such transaction or series of related transactions.

EVENTS OF DEFAULT

The following will be "Events of Default" under the Indenture:

(i) default in the payment of the principal of or premium, if any, when due and payable, on any of the notes (at Stated Maturity, upon optional redemption, required purchase or otherwise); or

(ii) default in the payment of an installment of interest on any of the notes, when due and payable, for 30 days; or

(iii) default in the performance, or breach, of any covenant or agreement of the company under the Indenture (other than a default in the performance or breach of a covenant or agreement which is specifically dealt with in clause (i), (ii) or (iv)) and such default or breach shall continue for a period of 30 days after written notice has been given, by certified mail, (x) to the company by the Trustee or (y) to the company and the Trustee by the holders of at least 25% in aggregate principal amount of the outstanding notes; or

(iv) (a) there shall be a default in the performance or breach of the provisions of "-- Consolidation, Merger and Sale of Assets, Etc."; (b) the company shall have failed to make or consummate an Asset Sale Offer in accordance with the provisions of the Indenture described under "-- Material Covenants -- Dispositions of Proceeds of Asset Sales"; or (c) the company shall have failed to make or consummate a Change of Control Offer in accordance with the provisions of the Indenture described under "-- Change of Control"; or

(v) default or defaults under one or more agreements, instruments, mortgages, bonds, debentures or other evidences of Indebtedness under which the company or any Restricted Subsidiary then has outstanding Indebtedness in excess of \$10 million, individually or in the aggregate, and (a) such default or defaults include a failure to make a payment of principal, (b) such Indebtedness is already due and payable in full or (c) such default or defaults have resulted in the acceleration of the maturity of such Indebtedness; provided that if any such default is cured or waived or any such acceleration rescinded, or such Indebtedness is repaid, within a period of 10 days from the continuation of such default beyond the applicable grace period or the occurrence of such acceleration, as the case may be, such Event of Default under the Indenture and any consequential acceleration of the notes shall be automatically rescinded, so long as such rescission does not conflict with any judgment or decree; or

(vi) one or more judgments, orders or decrees of any court or regulatory or administrative agency of competent jurisdiction for the payment of money in excess of \$10 million, either individually or in the aggregate (net of applicable insurance coverage which is acknowledged in writing by the insurer or which has been determined to be applicable by a final nonappealable determination by a court of competent jurisdiction), shall be entered against the company or any Restricted Subsidiary or any of their respective properties and shall not be discharged and there shall have been a period of 60 days after the date on which any period for appeal has expired and during which a stay of enforcement of such judgment, order or decree shall not be in effect; or (vii) the entry of a decree or order by a court having jurisdiction in the premises (A) for relief in respect of the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary, in an involuntary case or proceeding under the Federal Bankruptcy Code or any other federal, state or foreign bankruptcy, insolvency, reorganization or similar law or (B) adjudging the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary, bankrupt or insolvent, or approving a petition seeking reorganization, arrangement, adjustment or composition of or in respect of the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary, under the Federal Bankruptcy Code or any other similar federal, state or foreign law, or appointing a custodian, receiver, liquidator, assignee, trustee or sequestrator (or other similar official) of the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiaries that, taken together, would constitute a Significant subsidiary or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiaries that, taken together, would constitute a Significant Subsidiaries that, taken together, would constitute a Significant Subsidiary or of any substantial part of any of their properties, or ordering the winding up or liquidation of any of their affairs, and the continuance of any such decree or order unstayed and in effect for a period of 60 consecutive days; or

(viii) the institution by the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary of a voluntary case or proceeding under the Federal Bankruptcy Code or any other similar federal, state or foreign law or any other case or proceedings to be adjudicated a bankrupt or insolvent, or the consent by the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary to the entry of a decree or order for relief in respect of the company or such Significant Subsidiary or group of Restricted Subsidiaries in any involuntary case or proceeding under the Federal Bankruptcy Code or any other similar federal, state or foreign law or to the institution of bankruptcy or insolvency proceedings against the company or such Significant Subsidiary or group of Restricted Subsidiaries, or the filing by the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary of a petition or answer or consent seeking reorganization or relief under the Federal Bankruptcy Code or any other similar federal, state or foreign law, or the consent by it to the filing of any such petition or to the appointment of or taking possession by a custodian, receiver, liquidator, assignee, trustee or sequestrator (or other similar official) of any of the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary or of any substantial part of its property, or the making by it of an assignment for the benefit of creditors, or the admission by it in writing of its inability to pay its debts generally as they become due or the taking of corporate action by the company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary in furtherance of any such action; or

(ix) any of the Guarantees of any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary ceases to be in full force and effect or any of such Guarantees is declared to be null and void and unenforceable or any of such Guarantees is found to be invalid or any of such Guarantors denies its liability under its Guarantee (other than by reason of release of such Guarantor in accordance with the terms of the Indenture).

If an Event of Default (other than those covered by clause (vii) or (viii) above) shall occur and be continuing, the Trustee, by notice to the company, or the holders of at least 25% in aggregate principal amount of the notes then outstanding, by notice to the Trustee and the company, may declare the principal of, premium, if any, and accrued and unpaid interest, if any, on all of the outstanding notes due and payable immediately, upon which declaration, all amounts payable in respect of the notes shall be due and payable. If an Event of Default specified in clause (vii) or (viii) above occurs and is continuing, then the principal of, premium, if any, and accrued and unpaid interest, if any, on all the outstanding notes shall ipso facto become and be immediately due and payable without any declaration or other act on the part of the Trustee or any holder of notes.

After a declaration of acceleration under the Indenture, but before a judgment or decree for payment of the money due has been obtained by the Trustee, the holders of a majority in aggregate principal amount of the outstanding notes, by written notice to the company and the Trustee, may rescind such declaration if: (a) the company or any Guarantor has paid or deposited with the Trustee a sum sufficient to pay (i) all sums paid or advanced by the Trustee under the Indenture and the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel, (ii) all overdue interest on all notes, (iii) the principal of and

premium, if any, on any notes which have become due otherwise than by such declaration of acceleration and interest thereon at the rate borne by the notes, and (iv) to the extent that payment of such interest is lawful, interest upon overdue interest and overdue principal at the rate borne by the notes which has become due otherwise than by such declaration of acceleration; (b) the rescission would not conflict with any judgment or decree of a court of competent jurisdiction; and (c) all Events of Default, other than the non-payment of principal of, premium, if any, and interest on the notes that has become due solely by such declaration of acceleration, have been cured or waived.

No holder of any of the notes will have any right to institute any proceeding with respect to the Indenture or any remedy thereunder, unless the holders of at least 25% in aggregate principal amount of the outstanding notes have made written request, and offered reasonable indemnity, to the Trustee to institute such proceeding as Trustee under the notes and the Indenture, the Trustee has failed to institute such proceeding within 45 days after receipt of such notice and the Trustee, within such 45-day period, has not received directions inconsistent with such written request by holders of a majority in aggregate principal amount of the outstanding notes. Such limitations will not apply, however, to a suit instituted by a holder of a note for the enforcement of the payment of the principal of, premium, if any, or interest on such note on or after the respective due dates expressed in such note. These limitations would also not affect a noteholder's ability to enforce the provisions of the guarantee on or after the respective due dates expressed in a note.

During the existence of an Event of Default, the Trustee will be required to exercise such rights and powers vested in it under the Indenture and use the same degree of care and skill in its exercise thereof as a prudent person would exercise under the circumstances in the conduct of such person's own affairs. Subject to the provisions of the Indenture relating to the duties of the Trustee, in case an Event of Default shall occur and be continuing, the Trustee under the Indenture will not be under any obligation to exercise any of its rights or powers under the Indenture at the request or direction of any of the holders unless such holders shall have offered to the Trustee reasonable security or indemnity. Subject to certain provisions concerning the rights of the Trustee, the holders of a majority in aggregate principal amount of the outstanding notes will have the right to direct the time, method and place of conducting any proceeding for any remedy available to the Trustee, or exercising any trust or power conferred on the Trustee under the Indenture.

If a Default or an Event of Default occurs and is continuing and is known to the Trustee, the Trustee shall mail to each holder of the notes notice of the Default or Event of Default within 30 days after obtaining knowledge thereof. Except in the case of a Default or an Event of Default in payment of principal of, premium, if any, or interest on any notes, the Trustee may withhold the notice to the holders of such notes if a committee of its trust officers in good faith determines that withholding the notice is in the interest of the note holders.

The Indenture will require the company to furnish to the Trustee annual and quarterly statements as to the performance by the company of its obligations under the Indenture and as to any default in such performance. Our company also will be required to notify the Trustee within five business days of any event which is, or after notice or lapse of time or both would become, an Event of Default.

NO LIABILITY FOR CERTAIN PERSONS

No director, officer, employee or stockholder of the company, nor any director, officer or employee of any Guarantor, as such, will have any liability for any obligations of the company or any Guarantor under the notes, the Guarantees or the Indenture based on, in respect of or by reason of such obligations or their creation. Each holder by accepting a note waives and releases all such liability. The foregoing waiver and release are an integral part of the consideration for the issuance of the notes. Such waiver may not be effective to waive liabilities under the federal securities laws.

LEGAL DEFEASANCE OR COVENANT DEFEASANCE OF INDENTURE

Our company may, at its option and at any time, terminate the obligations of the company and the Guarantors with respect to the outstanding notes ("legal defeasance") to the extent set forth below. Such legal defeasance means that the company shall be deemed to have paid and discharged the entire Indebtedness represented by the outstanding notes, except for (i) the rights of holders of outstanding notes to receive solely from the trust fund described below payment in respect of the principal of, premium, if any, and interest on such notes when such payments are due, (ii) the company's obligations to issue temporary notes, register the transfer or exchange of any notes, replace mutilated, destroyed, lost or stolen notes and maintain an office or agency for payments in respect of the notes, (iii) the rights, powers, trusts, duties and immunities of the Trustee, and (iv) the legal defeasance provisions of the Indenture. In addition, the company may, at its option and at any time, elect to terminate the obligations of the company and the Guarantors with respect to certain covenants that are set forth in the Indenture, some of which are described under "--Material Covenants" above, and any subsequent failure to comply with such obligations shall not constitute a Default or an Event of Default with respect to the notes ("covenant defeasance").

In order to exercise either legal defeasance or covenant defeasance, (i) the company or any Guarantor must irrevocably deposit with the Trustee, in trust, for the benefit of the holders of the notes, cash in United States dollars, U.S. Government Obligations (as defined in the Indenture), or a combination thereof, in such amounts as will be sufficient, in the opinion of a nationally recognized firm of independent public accountants, to pay the principal of, premium, if any, and interest on the outstanding notes to redemption or maturity (except lost, stolen or destroyed notes which have been replaced or paid); (ii) the company shall have delivered to the Trustee an opinion of counsel to the effect that the holders of the outstanding notes will not recognize income, gain or loss for federal income tax purposes as a result of such legal defeasance or covenant defeasance and will be subject to federal income tax on the same amounts, in the same manner and at the same times as would have been the case if such legal defeasance or covenant defeasance had not occurred (in the case of legal defeasance, such opinion must refer to and be based upon a ruling of the Internal Revenue Service or a change in applicable federal income tax laws); (iii) no Default or Event of Default shall have occurred and be continuing on the date of such deposit (other than a Default or Event of Default relating to the borrowing of funds to be applied to such deposit); (iv) such legal defeasance or covenant defeasance shall not cause the Trustee to have a conflicting interest with respect to any securities of the company; (v) such legal defeasance or covenant defeasance shall not result in a breach or violation of, or constitute a default under, any agreement or instrument to which the company is a party or by which it is bound; (vi) the company shall have delivered to the Trustee an opinion of counsel to the effect that after the 91st day following the deposit, the trust funds will not be subject to the effect of any applicable bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (vii) the company shall have delivered to the Trustee an officers' certificate stating that the deposit was not made by the company with the intent of preferring the holders of the notes over the other creditors of the company with the intent of hindering, delaying or defrauding creditors of the company or others; (viii) no event or condition shall exist that would prevent the company from making payments of the principal of, premium, if any, and interest on the notes on the date of such deposit or at any time ending on the 91st day after the date of such deposit; and (ix) the company shall have delivered to the Trustee an officers certificate and an opinion of counsel, each stating that all conditions precedent under the Indenture to either legal defeasance or covenant defeasance, as the case may be, have been complied with.

Our company may exercise its legal defeasance option notwithstanding its prior exercise of its covenant defeasance option.

SATISFACTION AND DISCHARGE

The Indenture will be discharged and will cease to be of further effect (except as to surviving rights or registration of transfer or exchange of the notes, as expressly provided for in the Indenture) as to all outstanding notes when: (i) either (a) all the notes theretofore authenticated and delivered (except lost, stolen or destroyed notes which have been replaced or repaid and notes for whose payment money has theretofore been deposited in trust or segregated and held in trust by the company and thereafter repaid to the company or discharged from such trust) have been delivered to the Trustee for cancellation or (b) all notes not theretofore delivered to the Trustee for cancellation (except lost, stolen or destroyed notes which have been replaced or paid) have become due and payable or will become due and payable at their Stated Maturity within one year, or are to be called for redemption within one year under arrangements satisfactory to the Trustee for the serving of notice of redemption by the Trustee in the name, and at the expense, of the company, and the company or any Guarantor has irrevocably deposited or caused to be deposited with the Trustee funds in an amount sufficient to pay and discharge the entire Indebtedness on the notes not theretofore delivered to the Trustee for cancellation, for principal of, premium, if any, and interest on the notes to the date of deposit (in the case of notes which have become due and payable) or to the Stated Maturity or date for redemption, as the case may be, together with irrevocable instructions from the company directing the Trustee to apply such funds to the payment thereof at Stated Maturity or redemption, as the case may be; (ii) the company or the Guarantors have paid all other sums payable under the Indenture by the company or the Guarantors; and (iii) the

company has delivered to the Trustee an officers' certificate and an opinion of counsel which, taken together, state that all conditions precedent under the Indenture relating to the satisfaction and discharge of the Indenture have been complied with.

AMENDMENTS AND WAIVERS

From time to time, the company and the Guarantors, when authorized by a resolution of its Board of Directors, and the Trustee may, without the consent of the holders of any outstanding notes, amend or modify the Indenture or the notes for certain specified purposes, including, among other things, curing ambiguities, defects or inconsistencies, qualifying, or maintaining the qualification of, the Indenture under the Trust Indenture Act, as long as any such change does not adversely affect the rights of any holder of notes. Other amendments and modifications of the Indenture or the notes may be made by the company, the Guarantors and the Trustee with the consent of the holders of not less than a majority of the aggregate principal amount of the outstanding notes; provided, however, that no such modification or amendment may, without the consent of the holder of each outstanding note affected thereby, (i) change the Stated Maturity of the principal of, or any installment of interest on, any note or alter the redemption provisions of the notes, (ii) reduce the principal amount of (or the premium, if any, on), or interest on, any notes, (iii) change the currency in which any notes or any premium or the interest thereon is payable, (iv) reduce the above-stated percentage in principal amount of outstanding notes that must consent to an amendment or modification of the Indenture or the notes, (v) impair the right to institute suit for the enforcement of any payment on or with respect to the notes or the Guarantees, (vi) reduce the percentage in aggregate principal amount of outstanding notes necessary to waive compliance with certain provisions of the Indenture or to waive certain defaults under the Indenture, (vii) amend or modify the obligation of the company to make and consummate a Change of Control Offer after the occurrence of a Change of Control or make and consummate the Asset Sale Offer with respect to any Asset Sale that has been consummated or modify any of the provisions or definitions with respect thereto, (viii) to modify or amend any provision of the Indenture relating to the Guarantees in a manner adverse to the holders of the notes or (ix) modify or change any provision of the Indenture or the related definitions affecting the subordination or ranking of the notes or any Guarantee in a manner which adversely affects the note holders.

The holders of not less than a majority in aggregate principal amount of the outstanding notes may on behalf of the holders of all the notes waive (i) compliance by the company with certain restrictive provisions of the Indenture and (ii) any past defaults under the Indenture, except a default in the payment of the principal of, premium, if any, or interest on any note, or in respect of a covenant or provision which under the Indenture cannot be modified or amended without the consent of the holder of each note outstanding.

FURTHER ISSUES

We may from time to time, without notice to or the consent of the holders of the notes, create and issue further notes ranking equally and ratably with the notes we are offering hereby in all respects, so that such further notes shall be consolidated and form a single series with the notes and shall have the same terms as to status, redemption or otherwise as the notes.

THE TRUSTEE

The Indenture will provide that, except during the continuance of an Event of Default, the Trustee thereunder will perform only such duties as are specifically set forth in the Indenture. If an Event of Default has occurred and is continuing, the Trustee will exercise such rights and powers vested in it under the Indenture and use the same degree of care and skill in its exercise as a prudent person would exercise under the circumstances in the conduct of such person's own affairs.

The Indenture and provisions of the Trust Indenture Act incorporated by reference therein will contain limitations on the rights of the Trustee thereunder, should it become a creditor of the company, to obtain payment of claims in certain cases or to realize on certain property received by it in respect of any such claims, as security or otherwise. The Indenture will permit the Trustee to engage in other transactions; provided, however, that if it acquires any conflicting interest (as defined in the Trust Indenture Act) it must eliminate such conflict or resign.

State Street Bank and Trust Company, Goodwin Square, 225 Asylum Street, 23rd Floor, Hartford, Connecticut 06103, will be the Trustee under the Indenture. State Street is also the trustee under the Indenture governing our \$150 million principal amount of 9 3/8% senior subordinated notes due 2009 that we issued in 1999.

GOVERNING LAW

The Indenture and the notes will be governed by the laws of the State of New York.

CERTAIN DEFINITIONS

"Acquired Indebtedness" means Indebtedness of a person (a) assumed in connection with an Asset Acquisition from such person or (b) existing at the time such person becomes or is merged into a Subsidiary of any other person.

"Affiliate" means, with respect to any specified person, any other person directly or indirectly controlling or controlled by or under direct or indirect common control with such specified person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by" and "under common control with"), as used with respect to any person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such person, whether through the ownership of Voting Stock, by agreement or otherwise; provided that beneficial ownership of 10% of more of the Voting Stock of a person shall be deemed to be control.

"Asset Acquisition" means (a) an Investment by or company or any Restricted Subsidiary in any other person pursuant to which such person shall become a Restricted Subsidiary, or shall be merged with or into the company or any Restricted Subsidiary, or (b) the acquisition by the company or any Restricted Subsidiary of the assets of any person which constitute all or substantially all of the assets of such person, any division or line of business of such person or, other than in the ordinary course of business, any other properties or assets of such person.

"Asset Sale" means any sale, issuance, conveyance, transfer, lease or other disposition by the company or any Restricted Subsidiary to any person other than the company or a Restricted Subsidiary, of (a) any Capital Stock of any Restricted Subsidiary; (b) all or substantially all of the properties and assets of any division or line of business of the company or any Restricted Subsidiary; or (c) any other properties or assets of the company or any Restricted Subsidiary outside of the ordinary course of business, other than (i) sales of obsolete, damaged or used equipment or other equipment or inventory sales in the ordinary course of business, (ii) sales of assets in one or a series of related transactions for an aggregate consideration of less than \$2 million and (iii) sales of accounts receivable for financing purposes. For the purposes of this definition, the term "Asset Sale" shall not include (i) any sale, issuance, conveyance, transfer, lease or other disposition of properties or assets that is governed by the provisions described under "-- Consolidation, Merger, Sale of Assets, Etc.," (ii) a Restricted Payment that is permitted by the covenant described under "-- Material Covenants -- Limitation on Restricted Payments," or (iii) the trade or exchange by the company or any Restricted Subsidiary of any property or assets owned or held by the company or such Restricted Subsidiary for any property or assets owned or held by another person, provided that the Fair Market Value of the properties traded or exchanged by the company or such Restricted Subsidiary (including any cash or Cash Equivalents to be delivered by the company or such Restricted Subsidiary) is reasonably equivalent to the Fair Market Value of the properties (together with any cash or Cash Equivalents) to be received by the company or such Restricted Subsidiary, and provided further that any such cash or Cash Equivalents shall be deemed to constitute Net Cash Proceeds of an Asset Sale for purposes of the covenant described under "Material Covenants -- Disposition of Proceeds of Asset Sales.

"Average Life to Stated Maturity" means, with respect to any Indebtedness, as at any date of determination, the quotient obtained by dividing (i) the sum of the products of (a) the number of years (and any portion thereof) from such date of such determination to the date or dates of each successive scheduled principal payment (including, without limitation, any sinking fund or mandatory redemption payment requirements) of such Indebtedness, and (b) the amount of each such principal payment by (ii) the sum of all such principal payments.

"Board of Directors" means the board of directors of a company or its equivalent, including managers of a limited liability company, general partners of a partnership or trustees of a business trust, or any duly authorized committee thereof.

"Capital Stock" means, with respect to any person, any and all shares, interests, participations, rights in or other equivalents (however designated) of such person's capital stock or equity participations, and any rights (other than debt securities convertible into capital stock), warrants or options exchangeable for or convertible into such capital stock and including, without limitation, with respect to partnerships, limited liability companies or business trusts, ownership interests (whether general or limited) and any other interest or participation that confers on a person the right to receive a share of the profits and losses of, or distributions of assets of, such partnerships, limited liability companies or business trusts.

"Capitalized Lease Obligation" means any obligation to pay rent or other amounts under a lease of (or other agreement conveying the right to use) any property (whether real, personal or mixed) that is required to be classified and accounted for as a capital lease obligation under GAAP, and, for the purpose of the Indenture, the amount of such obligation at any date shall be the capitalized amount thereof at such date, determined in accordance with GAAP.

"Cash Equivalents" means, at any time, (a) any evidence of Indebtedness, maturing not more than one year after such time, issued or guaranteed by the United States Government or any agency thereof (provided that the full faith and credit of the United States of America is pledged in support thereof), (b) commercial paper, maturing not more than one year from the date of issue, rated at least A-2 by Standard & Poor's Ratings Group or P-2 by Moody's Investors Service, Inc., (c) any certificate of deposit (or time deposits represented by such certificates of deposit) or bankers acceptance, maturing not more than one year after such time, or overnight Federal Funds transactions that are issued or sold by a banking institution that is a member of the Federal Reserve System and has a combined capital and surplus and undivided profits of not less than \$500 million, (d) repurchase obligations with a term of not more than seven days for underlying securities of the types described in clause (a) above entered into with any bank meeting the specifications of clause (c) above, and (e) investments in funds investing primarily in investments of the types described in clauses (a) through (d) above.

"Change of Control" means the occurrence of any of the following events: (a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) is or becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a person shall be deemed to have "beneficial ownership" of all securities that such person has the right to acquire, whether such right is exercisable immediately or only after the passage of time), directly or indirectly, of more than 50% of the total Voting Stock of the company; (b) the company consolidates with, or merges with or into, another person or sells, assigns, conveys, transfers, leases or otherwise disposes of all or substantially all of its assets to any person, or any person consolidates with, or merges with or into, the company, in any such event pursuant to a transaction in which the outstanding Voting Stock of the company is converted into or exchanged for cash, securities or other property, other than any such transaction where (i) the outstanding Voting Stock of the company is converted into or exchanged for Voting Stock (other than Redeemable Capital Stock) of the surviving or transferee corporation and (ii) immediately after such transaction no "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) is the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a person shall be deemed to have "beneficial ownership" of all securities that such person has the right to acquire, whether such right is exercisable immediately or only after the passage of time), directly or indirectly, of more than 50% of the total Voting Stock of the surviving or transferee corporation; (c) during any consecutive two-year period, individuals who at the beginning of such period constituted the Board of Directors of the company (together with any new directors whose election by such Board of Directors or whose nomination for election by the stockholders of the company was approved by a vote of 66 2/3% of the directors then still in office who were either directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason to constitute a majority of the Board of Directors of the company then in office; or (d) the company is liquidated or dissolved or adopts a plan of liquidation.

"Common Stock" means the common stock of the company, par value \$0.01 per share, and the company's Restricted Voting Common Stock, par value \$0.01 per share.

"Consolidated Cash Flow Available for Fixed Charges" as of any date of determination means, with respect to any person for any period, the sum of, without duplication, the amounts for such period, taken as a single accounting period, of (a) Consolidated Net Income, (b) Consolidated Non-cash Charges, (c) Consolidated Interest Expense, and (d) Consolidated Income Tax Expense (other than income tax expense (either positive or negative) attributable to extraordinary gains or losses).

'Consolidated Fixed Charge Coverage Ratio" as of any date of determination means, with respect to any person, the ratio of the aggregate amount of Consolidated Cash Flow Available for Fixed Charges of such person for the four full fiscal quarters, treated as one period, for which financial information in respect thereof is available immediately preceding the date of the transaction (the "Transaction Date") giving rise to the need to calculate the Consolidated Fixed Charge Coverage Ratio (such four full fiscal quarter period being referred to herein as the "Four Quarter Period") to the aggregate amount of Consolidated Fixed Charges of such person for the Four Quarter Period. For purposes of making the computation referred to above, Consolidated Cash Flow Available for Fixed Charges and Consolidated Fixed Charges shall be calculated giving pro forma effect (in a manner consistent with Rule 11-02 of Regulation S-X) to the following events (without duplication): (i) any Asset Sale or Asset Acquisition occurring since the first day of the Four Quarter Period (including to the date of calculation) as if such acquisition or disposition occurred at the beginning of the Four Quarter Period (including giving effect to (A) the amount of any reduction in expenses related to any compensation, remuneration or other benefit paid or provided to any employee, consultant, Affiliate or equity owner of the entity involved in any such Asset Sale or Asset Acquisition to the extent such costs are eliminated or reduced (or public announcement has been made of the intent to eliminate or reduce such costs) prior to the date of such calculation and not replaced and (B) the amount of any reduction in general, administrative or overhead costs of the entity involved in any such Asset Sale or Asset Acquisition), (ii) the incurrence of Indebtedness giving rise to the need to calculate the Consolidated Fixed Charge Coverage Ratio and (if applicable) the application of the net proceeds therefrom, including to refinance other Indebtedness, as if such Indebtedness were incurred at the beginning of the Four Quarter Period, (iii) the incurrence, repayment or retirement of any other Indebtedness by the company and its Restricted Subsidiaries since the first day of the Four Quarter Period and prior to the date of making this calculation as if such Indebtedness or obligations were incurred, repaid or retired at the beginning of the Four Quarter Period (except that, in making such computation, the amount of Indebtedness under any revolving credit facility shall be computed based upon the average daily balance of such Indebtedness during the Four Quarter Period), (iv) elimination of Consolidated Cash Flow Available for Fixed Charges and Consolidated Fixed Charges attributable to discontinued operations, as determined in accordance with GAAP, but, with respect to Consolidated Fixed Charges, only to the extent that the obligations giving rise to such Consolidated Fixed Charges will not be obligations of the referent person or any of its Restricted Subsidiaries following the Transaction Date. In calculating Consolidated Fixed Charges for purposes of determining the denominator (but not the numerator) of the Consolidated Fixed Charge Coverage Ratio, (i) interest on outstanding Indebtedness determined on a fluctuating basis as of the Transaction Date and which will continue to be so determined thereafter shall be deemed to have accrued at a fixed rate per annum equal to the rate of interest on such Indebtedness in effect on the Transaction Date; and (ii) if interest on any Indebtedness actually incurred on the Transaction Date may optionally be determined at an interest rate based upon a factor of a prime or similar rate, a Eurocurrency interbank offered rate, or other rates, then the interest rate in effect on the Transaction Date will be deemed to have been in effect during the Four Quarter Period. If such person or any of its Restricted Subsidiaries directly or indirectly guarantees Indebtedness of a third person, the above provisions shall give effect to the incurrence of such guaranteed Indebtedness as if such person or such Subsidiary had directly incurred or otherwise assumed such guaranteed Indebtedness.

"Consolidated Fixed Charges" means, with respect to any person for any period, the sum of, without duplication, the amounts for such period of (i) Consolidated Interest Expense, and (ii) the aggregate amount of dividends and other distributions paid or accrued during such period in respect of Redeemable Capital Stock or Preferred Stock of such person and its Restricted Subsidiaries on a consolidated basis.

"Consolidated Income Tax Expense" means, with respect to any person for any period, the provision for federal, state, local and foreign income taxes of such person and its Restricted Subsidiaries for such period as determined on a consolidated basis in accordance with GAAP.

"Consolidated Interest Expense" means, with respect to any person for any period, without duplication, the sum of (i) the interest expense of such person and its Restricted Subsidiaries for such period as determined on a

consolidated basis in accordance with GAAP, including, without limitation, (a) any amortization of debt discount and capitalized debt issuance costs, (b) the net cost under Interest Rate Protection Obligations (including any amortization of discounts), (c) the interest portion of any deferred payment obligation, (d) all commissions, discounts and other fees and charges owed with respect to letters of credit, bankers' acceptance financing or similar facilities and (e) all accrued interest and (ii) the interest component of Capitalized Lease Obligations paid, accrued and/or scheduled to be paid or accrued by such person and its Restricted Subsidiaries during such period as determined on a consolidated basis in accordance with GAAP.

"Consolidated Net Income" means, with respect to any person, for any period, the consolidated net income (or loss) of such person and its Restricted Subsidiaries for such period as determined in accordance with GAAP, adjusted, to the extent included in calculating such net income, by excluding, without duplication, (i) all extraordinary gains or losses (net of fees and expenses relating to the transaction giving rise thereto), (ii) the portion of net income of such person and its Restricted Subsidiaries allocable to minority interests in unconsolidated persons or to Investments in Unrestricted Subsidiaries to the extent that cash dividends or distributions have not actually been received by such person or one of its Restricted Subsidiaries, (iii) net income (or loss) of any person combined with such person or one of its Restricted Subsidiaries on a "pooling of interests" basis attributable to any period prior to the date of combination, (iv) gains or losses in respect of any Asset Sales by such person or one of its Restricted Subsidiaries (net of fees and expenses relating to the transaction giving rise thereto), on an after-tax basis, (v) the net income of any Restricted Subsidiary of such person to the extent that the declaration of dividends or similar distributions by that Restricted Subsidiary of that income is not at the time permitted, directly or indirectly, by operation of the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or governmental regulation applicable to that Restricted Subsidiary or its stockholders and (vi) any gain or loss realized as a result of the cumulative effect of a change in accounting principles.

"Consolidated Non-cash Charges" means, with respect to any person for any period, the aggregate depreciation, amortization (including amortization of goodwill and other intangibles) and other non-cash charges of such person and its Restricted Subsidiaries reducing Consolidated Net Income of such person and its Restricted Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP (excluding any such non-cash charges constituting an extraordinary item or loss).

"Credit Facility" means the Credit Agreement, dated as of May 22, 2001, among our company, The Chase Manhattan Bank, as Administrative Agent and the banks named therein (which replaces the Credit Agreement dated as of July 30, 1998 among the company, NationsBank, N.A., as the Agent, and the banks named therein), including any notes, guarantees, collateral documents, instruments and agreements executed in connection therewith, and in each case as amended (including any amendment and restatement thereof), modified, extended, renewed, refunded, substituted or replaced or refinanced from time to time, including any agreement extending the maturity of, refinancing, replacing or otherwise restructuring (including increasing the amount of available borrowings thereunder or adding Subsidiaries of the company as additional borrowers or guarantors thereunder) all or any portion of the Indebtedness under such agreement or any successor or replacement agreement and whether by the same or any other agents, creditor, lender or group of creditors or lenders.

"Default" means any event that is, or after notice or passage of time or both would be, an Event of Default.

"Disinterested Member of the Board of Directors of the company" means, with respect to any transaction or series of related transactions, a member of the Board of Directors of the company other than a member who has any material direct or indirect financial interest in or with respect to such transaction or series of related transactions or is an Affiliate, or an officer, director or an employee of any person (other than the company) who has any direct or indirect financial interest in or with respect to such transaction or series of related transactions (in each case other than an interest arising solely from the beneficial ownership of Capital Stock of the company).

"Event of Default" has the meaning set forth under "-- $\ensuremath{\mathsf{Events}}$ of Default" herein.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Fair Market Value" means, with respect to any asset, the price (after taking into account any liabilities relating to such assets) which could be negotiated in an arm's length free market transaction between a willing seller and a

willing buyer, neither of which is under pressure or compulsion to complete the transaction. Fair Market Value shall be determined by the Board of Directors of the company in good faith.

"GAAP" means generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other entity as may be approved by a significant segment of the accounting profession of the United States of America, which are in effect from time to time.

"Guarantee" means, as applied to any obligation, (i) a guarantee (other than by endorsement of negotiable instruments for collection in the ordinary course of business), direct or indirect, in any manner, of any part or all of such obligation and (ii) an agreement, direct or indirect, contingent or otherwise, the practical effect of which is to assure in any way the payment or performance (or payment of damages in the event of nonperformance) of all or any part of such obligation, including, without limiting the foregoing, the payment of amounts available to be drawn down under letters of credit of another person. When used as a verb, "guarantee" shall have a corresponding meaning.

"Indebtedness" means, with respect to any person, without duplication, (a) all liabilities of such person for borrowed money or for the deferred purchase price of property or services, excluding any trade payables and other accrued current liabilities incurred in the ordinary course of business, but including, without limitation, all obligations, contingent or otherwise, of such person in connection with any letters of credit, banker's acceptance or other similar credit transaction, if, and to the extent, any of the foregoing would appear as a liability on a balance sheet of such person prepared in accordance with GAAP, (b) all obligations of such person evidenced by bonds, notes, debentures or other similar instruments, if, and to the extent, any of the foregoing would appear as a liability on a balance sheet of such person prepared in accordance with GAAP, (c) all indebtedness of such person created or arising under any conditional sale or other title retention agreement with respect to property acquired by such person (even if the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), but excluding consignments and trade accounts payable arising in the ordinary course of business, (d) all Capitalized Lease Obligations of such person, (e) all Indebtedness referred to in the preceding clauses of other persons and all dividends of other persons, the payment of which is secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien upon property (including, without limitation, accounts and contract rights) owned by such person, even though such person has not assumed or become liable for the payment of such Indebtedness (the amount of such obligation being deemed to be the lesser of the Fair Market Value of such property or asset or the amount of the obligation so secured), (f) all guarantees of Indebtedness referred to in this definition by such person, (q) all Redeemable Capital Stock of such person valued at the greater of its voluntary or involuntary maximum fixed repurchase price plus accrued dividends, and (h) all Interest Rate Protection Obligations of such person; provided, however, that Indebtedness shall not include (i) Indebtedness arising from agreements of the company or any Restricted Subsidiary providing for indemnification, adjustment or holdback of purchase price or similar obligations, in each case, incurred or assumed in connection with the acquisition or disposition of any business, assets or a Subsidiary, other than guarantees of Indebtedness incurred by any person acquiring all or any portion of such business, assets or Subsidiary for the purpose of financing such acquisition, or (ii) obligations under performance bonds, performance guarantees, surety bonds, appeal bonds, security deposits or similar obligations. For purposes hereof, the "maximum fixed repurchase price" of any Redeemable Capital Stock which does not have a fixed repurchase price shall be calculated in accordance with the terms of such Redeemable Capital Stock as if such Redeemable Capital Stock were purchased on any date on which Indebtedness shall be required to be determined pursuant to the Indenture, and if such price is based upon, or measured by, the fair market value of such Redeemable Capital Stock, such fair market value shall be approved in good faith by the board of directors of the issuer of such Redeemable Capital Stock; provided, however, that if such Redeemable Capital Stock is not at the date of determination permitted or required to be repurchased, the "maximum fixed repurchase price" shall be the book value of such Redeemable Capital Stock.

"Interest Rate Protection Agreement" means, with respect to any person, any arrangement with any other person whereby, directly or indirectly, such person is entitled to receive from time to time periodic payments calculated by applying either a floating or a fixed rate of interest on a stated notional amount in exchange for periodic payments made by such person calculated by applying a fixed or a floating rate of interest on the same notional amount and

shall include, without limitation, interest rate swaps, caps, floors, collars and similar agreements or arrangements designed to protect against or manage such person's exposure to fluctuations in interest rates.

"Interest Rate Protection Obligations" means the net obligations of any person pursuant to any Interest Rate Protection Agreements.

"Investment" means, with respect to any person, any direct or indirect loan or other extension of credit (including, without limitation, a guarantee) or capital contribution to (by means of any transfer of cash or other property to others or any payment for property or services for the account or use of others), or any purchase or acquisition by such person of any Capital Stock, bonds, notes, debentures or other securities or evidences of Indebtedness issued by, any other person, provided that the term "Investment" shall not include (a) extensions of trade credit on commercially reasonable terms in accordance with normal trade practices and (b) Interest Rate Protection Obligations entered into in the ordinary course of business.

"Lien" means any mortgage, charge, pledge, lien (statutory or other), security interest, hypothecation, assignment for security, claim or other encumbrance upon or with respect to any property of any kind. A person shall be deemed to own subject to a Lien any property which such person has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement.

"Maturity Date" means February 1, 2009.

"Net Cash Proceeds" means, with respect to any Asset Sale, the proceeds thereof received by the company or any Restricted Subsidiary in the form of cash or Cash Equivalents including payments in respect of deferred payment obligations when received in the form of cash or Cash Equivalents (except to the extent that such obligations are financed or sold with recourse to the company or any Restricted Subsidiary) net of (i) brokerage commissions and other fees and expenses (including, without limitation, fees and expenses of legal counsel and investment bankers, recording fees, transfer fees and appraisers' fees) related to such Asset Sale, (ii) provisions for all taxes payable as a result of such Asset Sale, (iii) amounts required to be paid to any person (other than the company or any Restricted Subsidiary) owning a beneficial interest in the assets subject to the Asset Sale, (iv) payments made to permanently retire Indebtedness where payment of such Indebtedness is secured by the assets or properties the subject of such Asset Sale, and (v) appropriate amounts to be provided by the company or any Restricted Subsidiary, as the case may be, as a reserve required in accordance with GAAP against any liabilities associated with such Asset Sale and retained by the company or any Restricted Subsidiary, as the case may be, after such Asset Sale, including, without limitation, pension and other post-employment benefit liabilities, liabilities related to environmental matters and liabilities under any indemnification obligations associated with such Asset Sale; provided, however, that any amounts remaining after adjustments, revaluations or liquidations of such reserves shall constitute Net Cash Proceeds.

"Pari Passu Indebtedness" means any Indebtedness of the company that is pari passu in right of payment to the notes.

"Permitted Founder Stock Repurchases" means one or more repurchases by the company, for an aggregate purchase price not to exceed \$10 million since January 29, 1999, of shares of Common Stock owned by former owners of Subsidiaries of the company, that were, as of the date of acquisition of such stock by such persons, subject to contractual agreements with the company restricting their resale.

"Permitted Indebtedness" means, without duplication:

(a) Indebtedness of the company and the Guarantors evidenced by the outstanding notes issued on the Issue Date, the new notes issued in exchange for those notes and the Guarantees thereof;

(b) Indebtedness of the company and any Guarantor under the Credit Facility in an aggregate principal amount at any one time outstanding not to exceed \$250 million, less any amounts permanently repaid in accordance with the covenant described under "-- Material Covenants --Disposition of Proceeds of Asset Sales";

(c) Indebtedness of the company or any Guarantor outstanding as of the Issue Date after giving effect to the application of the proceeds of this offering as contemplated by this offering memorandum;

(d) Indebtedness of the company or any Restricted Subsidiary incurred in respect of bankers' acceptances and letters of credit in the ordinary course of business, including Indebtedness evidenced by letters of credit issued in the ordinary course of business to support the insurance or self-insurance obligations of the company or any of its Restricted Subsidiaries (including to secure workers' compensation and other similar insurance coverages), in an aggregate amount not to exceed \$15 million at any time, but excluding letters of credit issued in respect of or to secure money borrowed;

(e) (i) Interest Rate Protection Obligations of the company or a Guarantor covering Indebtedness of the company or a Guarantor and (ii) Interest Rate Protection Obligations of any Restricted Subsidiary covering Permitted Indebtedness or Acquired Indebtedness of such Restricted Subsidiary; provided that, in the case of either clause (i) or (ii), (x) any Indebtedness to which any such Interest Rate Protection Obligations correspond bears interest at fluctuating interest rates and is otherwise permitted to be incurred under the "Limitation on Indebtedness" covenant and (y) the notional principal amount of the principal amount of the Indebtedness to which such Interest Rate Protection Obligations relate shall not constitute Permitted Indebtedness;

(f) Indebtedness of a Restricted Subsidiary owed to and held by the company or another Restricted Subsidiary, except that (i) any transfer of such Indebtedness by the company or a Restricted Subsidiary (other than to the company or another Restricted Subsidiary), (ii) the sale, transfer or other disposition by the company or any Restricted Subsidiary of Capital Stock of a Restricted Subsidiary which is owed Indebtedness of another Restricted Subsidiary such that it shall no longer be a Restricted Subsidiary and (iii) the designation of a Restricted Subsidiary which is owed Indebtedness of another Restricted Subsidiary as an Unrestricted Subsidiary shall, in each case, be an incurrence of Indebtedness by such Restricted Subsidiary subject to the other provisions of the Indenture;

(g) Indebtedness of the company owed to and held by a Restricted Subsidiary which is unsecured and expressly subordinated in right of payment to the payment and performance of the obligations of the company under the Indenture and the notes, except that (i) any transfer of such Indebtedness by a Restricted Subsidiary (other than to another Restricted Subsidiary) and (ii) the sale, transfer or other disposition by the company or any Restricted Subsidiary of Capital Stock of a Restricted Subsidiary which is owed Indebtedness of the company such that it shall no longer be a Restricted Subsidiary and (iii) the designation of a Restricted Subsidiary which is owed Indebtedness of the company shall, in each case, be an incurrence of Indebtedness by the company, subject to the other provisions of the Indenture;

(h) Indebtedness of the company or any Guarantor represented by Capitalized Lease Obligations, mortgage financings or purchase money obligations, in each case incurred for the purpose of financing all or any part of the purchase price or cost of construction or improvement of property, plant or equipment used in the business of the company or such Guarantor, in an aggregate principal amount not to exceed \$25 million at any time outstanding;

(i) Subordinated Indebtedness of the company, in an aggregate principal amount not to exceed \$10 million at any time outstanding, that is convertible into Common Stock and issued in connection with an Asset Acquisition of a business engaged in the provision of electrical contracting and maintenance services to the commercial, industrial, power line and data cabling markets and any other businesses reasonably related thereto;

(j) Indebtedness of the company, in addition to that described in clauses (a) through (i) of this definition, in an aggregate principal amount not to exceed \$30 million at any time outstanding;

(k) (i) Indebtedness of the company the proceeds of which are used solely to refinance (whether by amendment, renewal, extension or refunding) Indebtedness of the company or any of the Guarantors incurred pursuant to the Consolidated Fixed Charge Coverage Ratio test of the proviso of the "Limitation on Indebtedness" covenant or clause (a), (c) or (k) of this definition and (ii) Indebtedness of any Guarantor the proceeds of which are used solely to refinance (whether by amendment, renewal, extension or refunding) Indebtedness of such Guarantor incurred pursuant to the Consolidated Fixed Charge Coverage Ratio test of the proviso of the "Limitation on Indebtedness" covenant or clause (c) or (k) of this definition; provided, however, that (x) the principal amount of Indebtedness incurred pursuant to this clause (k) (or if such Indebtedness provides for an amount less than the principal amount thereof to be due and payable upon a declaration of acceleration of maturity thereof, the original issue price of such Indebtedness) shall not exceed the sum of the principal amount of Indebtedness so refinanced, plus the amount of any premiums and fees required to be paid in connection with such refinancing pursuant to the terms of such Indebtedness, and (y) any Indebtedness incurred pursuant to this clause (k) (A) has no scheduled principal payment prior to the 91st day after the Maturity Date, (B) has an Average Life to Stated Maturity greater than the remaining Average Life to Stated Maturity of the notes and (C) is subordinated to the notes or the Guarantees, as the case may be, at least to the same extent that the Indebtedness being refinanced is subordinated to the notes or the Guarantees, as the case may be;

(1) Indebtedness of any Restricted Subsidiary that constitutes Acquired Indebtedness not incurred in contemplation of the acquisition of such Restricted Subsidiary; provided that such Indebtedness is repaid within 90 days following the consummation of the Asset Acquisition in which the company acquired such Restricted Subsidiary; and

(m) Guarantees by the company or guarantees by a Guarantor of Indebtedness that was permitted to be incurred under the Indenture.

For purposes of determining compliance with the "Limitation on Indebtedness" covenant, (A) in the event that an item of Indebtedness meets the criteria of more than one of the types of Indebtedness described in the clauses of the preceding paragraph, the company, in its sole discretion, shall classify such item of Indebtedness and only be required to include the amount and type of such Indebtedness in one such clause, and (B) the amount of Indebtedness issued at a price that is either less or greater than the principal amount thereof shall be equal to the amount of the liability in respect thereof determined in conformity with GAAP.

"Permitted Investments" means any of the following: (i) Investments in the company or in a Restricted Subsidiary; (ii) Investments in another person, if as a result of such Investment (A) such other person becomes a Restricted Subsidiary or (B) such other person is merged or consolidated with or into, or transfers or conveys all or substantially all of its assets to the company or a Restricted Subsidiary; (iii) Investments representing Capital Stock or obligations issued to the company or any of its Restricted Subsidiaries in settlement of debts created in the ordinary course of business or claims against any other person by reason of a composition or readjustment of debt or a reorganization of any debtor of the company or such Restricted Subsidiary or in satisfaction of judgments; (iv) Investments in Interest Rate Protection Agreements on commercially reasonable terms entered into by the company or any of its Restricted Subsidiaries in the ordinary course of business in connection with the operations of the business of the company or any Restricted Subsidiaries (v) Investments in interest rates on its outstanding Indebtedness; (vi) Investments acquired by the company or any Restricted Subsidiary in connection with an Asset Sale permitted under "-- Material Covenants -- Disposition of Proceeds of Asset Sales" to the extent such Investment to the extent that the consideration therefor is Capital Stock (other than Redeemable Capital Stock) of the company; (ix) any loans or other advances approved by the Board of Directors of the company; in its good faith judgment, not to exceed \$5 million at any one time outstanding; and (x) other Investments not to exceed \$5 million at any time outstanding.

"Permitted Liens" means the following types of Liens:

(a) any Lien existing as of the date of the Indenture;

(b) any Lien securing Acquired Indebtedness created prior to (and not created in connection with, or in contemplation of) the incurrence of such Indebtedness by the company or any Restricted Subsidiary, if such Lien does not attach to any property or assets of the company or any Restricted Subsidiary other than the property or assets subject to the Lien prior to such incurrence;

(c) Liens in favor of the company or a Restricted Subsidiary;

(d) Liens on and pledges of the Capital Stock of any Unrestricted Subsidiary securing any Indebtedness of such Unrestricted Subsidiary;

(e) Liens for taxes, assessments or governmental charges or claims, to the extent any such changes or claims constitute Indebtedness, either (i) not delinquent or (ii) contested in good faith by appropriate proceedings and as to which the company or its Restricted Subsidiaries shall have set aside on its books such reserves as may be required pursuant to GAAP;

(f) Liens incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance or other kinds of social security, old age pension or public liability obligations;

(g) Liens to secure Indebtedness (including Capitalized Lease Obligations) permitted by clause (h) under the definition of Permitted Indebtedness covering only the assets acquired with such indebtedness;

(h) Liens securing Interest Rate Protection Obligations permitted to be entered into pursuant to the debt incurrence covenant;

 (i) judgment and attachment Liens not giving rise to an Event of Default or Liens created by or existing from any litigation or legal proceeding that are currently being contested in good faith by appropriate proceedings and for which adequate reserves have been made;

(j) Liens in favor of collecting or payor banks having a right of setoff, revocation, refund or chargeback with respect to money or instruments of the company or any Subsidiary on deposit with or in possession of such bank; and

(k) Liens not otherwise permitted by clauses (a) through (j) that are incurred in the ordinary course of business of the company or any Restricted Subsidiary with respect to Indebtedness that does not exceed \$5 million at any one time outstanding.

"person" means any individual, corporation, partnership (general or limited), limited liability company, joint venture, association, joint-stock company, trust, unincorporated organization or government or any agency or political subdivision thereof.

"Preferred Stock," as applied to any person, means Capital Stock of any class or series (however designated) which is preferred as to the payment of dividends or distributions, or as to the distribution of assets upon any voluntary or involuntary liquidation or dissolution of such person, over shares of Capital Stock of any other class or series of such person.

"Qualified Equity Offering" means (i) any public sale of Common Stock of the company pursuant to a registration statement filed with the SEC in accordance with the Securities Act (other than any public offerings with respect to the company's Common Stock registered on Form S-8 or Form S-4) or (ii) any private placement for aggregate proceeds of at least \$25 million to a third party of Common Stock or Capital Stock (other than Redeemable Capital Stock) that is convertible into Common Stock.

"Redeemable Capital Stock" means any class or series of Capital Stock that, either by its terms, by the terms of any security into which it is convertible or exchangeable or by contract or otherwise, is or upon the happening of an event or passage of time would be required to be redeemed prior to the 91st day after the Maturity Date or is redeemable at the option of the holder thereof at any time prior to the 91st day after the Maturity Date, or is convertible into or exchangeable for debt securities at any time prior to the 91st day after the Maturity Date;

61

provided that Capital Stock will not constitute Redeemable Capital Stock solely because the holders thereof have the right to require the company to repurchase or redeem such Capital Stock upon the occurrence of a Change of Control or an Asset Sale.

"Restricted Subsidiary" means any Subsidiary of the company that is not an Unrestricted Subsidiary.

"Significant Subsidiary" means any Restricted Subsidiary that would be a "significant subsidiary" as defined in Article 1, Rule 1-02 of Regulation S-X, promulgated pursuant to the Securities Act, as such Regulation is in effect on the date of the Indenture.

"Stated Maturity" means, when used with respect to any note or any installment of interest thereon, the date specified in such note as the fixed date on which the principal of such note or such installment of interest is due and payable, and when used with respect to any other Indebtedness, means the date specified in the instrument governing such Indebtedness as the fixed date on which the principal of such Indebtedness, or any installment of interest thereon, is due and payable.

"Subordinated Indebtedness" means, with respect to the company, Indebtedness of the company which is expressly subordinated in right of payment to the notes.

"Subsidiary" means, with respect to any person, (i) a corporation a majority of whose Voting Stock is at the time, directly or indirectly, owned by such person, by one or more Subsidiaries of such person or by such person and one or more Subsidiaries thereof and (ii) any other person (other than a corporation), including, without limitation, a partnership, limited liability company, business trust or joint venture, in which such person, one or more Subsidiaries thereof or such person and one or more Subsidiaries thereof, directly or indirectly, at the date of determination thereof, have at least majority ownership interest entitled to vote in the election of directors, managers or trustees thereof (or other person performing similar functions). For purposes of this definition, any directors' qualifying shares or investments by foreign nationals mandated by applicable law shall be disregarded in determining the ownership of a Subsidiary.

"Unrestricted Subsidiary" means (i) each Subsidiary of the company designated as such pursuant to and in compliance with the covenant described under "-- Material Covenants -- Limitation on Designations of Unrestricted Subsidiaries" and (ii) each Subsidiary of any Subsidiary described in clause (i) of this definition.

"Voting Stock" means any class or classes of Capital Stock pursuant to which the holders thereof have the general voting power under ordinary circumstances to elect at least a majority of the board of directors, managers or trustees of any person (irrespective of whether or not, at the time, stock of any other class or classes shall have, or might have, voting power by reason of the happening of any contingency), and, with respect to the company, shall be deemed to include the Common Stock.

BOOK-ENTRY; DELIVERY AND FORM

The new notes will be issued in the form of one or more global securities. The global securities will be deposited with, or on behalf of, DTC and registered in the name of DTC or its nominee. Except as set forth below, the global securities may be transferred, in whole and not in part, only to DTC or another nominee of DTC. Investors may hold their beneficial interests in the global securities directly through DTC if they have an account with DTC or indirectly through organizations which have accounts with DTC.

Notes that are issued as described below under "-- Certificated Notes" will be issued in definitive form. Upon the transfer of notes in definitive form, such notes will, unless the global securities have previously been exchanged for notes in definitive form, be exchanged for an interest in the global securities representing the aggregate principal amount of notes being transferred.

63 BOOK-ENTRY PROCEDURES FOR THE GLOBAL NOTES

The descriptions of the operations and procedures of DTC, Euroclear and Clearstream set forth below are provided solely as a matter of convenience. These operations and procedures are solely within the control of these settlement systems and are subject to change by them from time to time. Neither we nor the initial purchasers take any responsibility for these operations or procedures, and investors are urged to contact the relevant system or its participants directly to discuss these matters.

DTC has advised us that it is:

- a limited purpose trust company organized under the laws of the State of New York;
- a "banking organization" within the meaning of the New York Banking Law;
- o a member of the Federal Reserve System;
- o a "clearing corporation" within the meaning of the Uniform Commercial Code, as amended; and
- o a "clearing agency" registered under Section 17A of the Securities Exchange Act of 1934.

DTC was created to hold securities for its participants and facilitates the clearance and settlement of securities transactions between its participants through electronic book-entry changes to the accounts of its participants, which eliminates the need for physical transfer and delivery of certificates. Participants in DTC include securities brokers

and dealers, including the initial purchasers; banks and trust companies; clearing corporations and some other organizations. Indirect access to DTC's system is also available to other entities such as banks, brokers, dealers and trust companies; these indirect participants clear through or maintain a custodial relationship with a participant in DTC, either directly or indirectly. Investors who are not DTC participants may beneficially own securities held by or on behalf of DTC only through participants or indirect participants in DTC.

We expect that pursuant to procedures established by DTC:

- o upon deposit of each global note, DTC will credit the accounts of participants in DTC designated by the initial purchasers with an interest in the global note; and
- o ownership of the notes will be shown on, and the transfer of ownership of the notes will be effected only through, records maintained by DTC, with respect to the interests of participants in DTC, and the records of participants and indirect participants, with respect to the interests of persons other than participants in DTC.
- o The laws of some jurisdictions may require that some purchasers of securities take physical delivery of the securities in definitive form. Accordingly, the ability to transfer interests in the notes represented by a global note to these persons may be limited. In addition, because DTC can act only on behalf of its participants, who in turn act on behalf of persons who hold interests through participants, the ability of a person having an interest in notes represented by a global note to pledge or transfer that interest to persons or entities that do not participate in DTC's system, or to otherwise take actions in respect of that interest.

So long as DTC or its nominee is the registered owner of a global note, DTC or the nominee, as the case may be, will be considered the sole owner or holder of the notes represented by the global note for all purposes under the indenture. Except as provided below, owners of beneficial interests in a global note:

- will not be entitled to have notes represented by the global note registered in their names;
- will not receive or be entitled to receive physical delivery of certificated notes; and
- o will not be considered the owners or holders of the notes under the indenture for any purpose, including with respect to the giving of any direction, instruction or approval to the trustee under the indenture.

Accordingly, each holder owning a beneficial interest in a global note must rely on the procedures of DTC and, if the holder is not a participant or an indirect participant in DTC, on the procedures of the DTC participant through which the holder owns its interest, to exercise any rights of a holder of notes under the indenture or the global note. We understand that under existing industry practice, if we request any action of holders of notes, or a holder that is an owner of a beneficial interest in a global note desires to take any action that DTC, as the holder of the global note, is entitled to take, then DTC would authorize its participants to take the action and the participants would authorize holders owning through participants to take the action or would otherwise act upon the instruction of such holders. Neither we nor the trustee will have any responsibility or liability for any aspect of the records relating to or payments made on account of DTC relating to the notes.

Payments with respect to the principal of, and premium, if any, liquidated damages, if any, and interest on, any notes represented by a global note registered in the name of DTC or its nominee on the applicable record date will be payable by the trustee to or at the direction of DTC or its nominee in its capacity as the registered holder of the global note representing those notes under the indenture. Under the terms of the indenture, we and the trustee may treat the persons in whose names the notes, including the global notes, are registered as the owners of the notes for the purpose of receiving payment on the notes and for any and all other purposes whatsoever. Accordingly, neither we nor the trustee has or will have any responsibility or liability for the payment of amounts to owners of beneficial interests in a global note, including principal, premium, if any, liquidated damages, if any, and interest. Payments by the participants and the indirect participants in DTC to the owners of beneficial interests in a global note.

governed by standing instructions and customary industry practice and will be the responsibility of the participants or the indirect participants and DTC.

Transfers between participants in DTC will be effected in accordance with DTC's procedures, and will be settled in same-day funds. Transfers between participants in Euroclear or Clearstream will be effected in the ordinary way in accordance with their respective rules and operating procedures.

Subject to compliance with the transfer restrictions applicable to the notes, cross-market transfers between the participants in DTC, on the one hand, and Euroclear or Clearstream participants, on the other hand, will be effected through DTC in accordance with DTC's rules on behalf of Euroclear or Clearstream, as the case may be, by its respective depositary. These cross-market transactions, however, will require delivery of instructions to Euroclear or Clearstream, as the case may be, by the counterparty in that system in accordance with the rules and procedures and within the established deadlines, Brussels time, of that system. If the transaction meets its settlement requirements, Euroclear or Clearstream, as the case may be, will deliver instructions to its respective depositary to take action to effect final settlement on its behalf by delivering or receiving interests in the relevant global notes in DTC, and making or receiving payment in accordance with normal procedures for same-day funds settlement applicable to DTC. Euroclear participants and Clearstream participants may not deliver instructions directly to the depositaries for Euroclear or Clearstream.

Because of time zone differences, the securities account of a Euroclear or Clearstream participant purchasing an interest in a global note from a participant in DTC will be credited, and any crediting will be reported to the relevant Euroclear or Clearstream participant, during the securities settlement processing day, which must be a business day for Euroclear and Clearstream, immediately following the settlement date of DTC. Cash received in Euroclear or Clearstream as a result of sales of interest in a global note by or through a Euroclear or Clearstream participant to a participant in DTC will be received with value on the settlement date of DTC but will be available in the relevant Euroclear or Clearstream cash account only as of the business day for Euroclear or Clearstream following DTC's settlement date.

Although DTC, Euroclear and Clearstream have agreed to the above procedures to facilitate transfers of interests in the global notes among participants in DTC, Euroclear and Clearstream, they are under no obligation to perform or to continue to perform the procedures, and the procedures may be discontinued at any time. Neither we nor the trustee will have any responsibility for the performance by DTC, Euroclear or Clearstream or their respective participants or indirect participants of their respective obligations under the rules and procedures governing their operations.

CERTIFICATED NOTES

If:

- o DTC notifies us that it is at any time unwilling or unable to continue as a depositary or DTC ceases to be registered as a clearing agency under the Securities Exchange Act of 1934 and a successor depositary is not appointed within 90 days of such notice or cessation;
- o we, at our option, notify the trustee in writing that we elect to cause the issuance of notes in definitive form under the indenture; or
- o upon the occurrence of some other events as provided in the indenture;

then, upon surrender by DTC of the global notes, certificated notes will be issued to each person that DTC identifies as the beneficial owner of the notes represented by the global notes. Upon the issuance of certificated notes, the trustee is required to register the certificated notes in the name of that person or persons, or their nominee, and cause the certificated notes to be delivered thereto.

Neither we nor the trustee will be liable for any delay by DTC or any participant or indirect participant in DTC in identifying the beneficial owners of the related notes and each of those persons may conclusively rely on, and will be protected in relying on, instructions from DTC for all purposes, including with respect to the registration and delivery, and the respective principal amounts, of the notes to be issued.

FEDERAL INCOME TAX CONSIDERATIONS

FEDERAL INCOME TAX CONSIDERATIONS OF THE EXCHANGE OF OUTSTANDING NOTES FOR NEW NOTES

The following discussion is a summary of certain federal income tax considerations relevant to the exchange of outstanding notes for new notes, but does not purport to be a complete analysis of all potential tax effects. The discussion is based upon the Internal Revenue Code of 1986, as amended (the "Code"), Treasury Regulations, Internal Revenue Service rulings and pronouncements and judicial decisions now in effect, all of which may be subject to change at any time by legislative, judicial or administrative action. These changes may be applied retroactively in a manner that could adversely affect a holder of new notes. The description does not consider the effect of any applicable foreign, state, local or other tax laws or estate or gift tax considerations.

We believe that the exchange of outstanding notes for new notes should not be an exchange or otherwise a taxable event to a holder for United States federal income tax purposes. Accordingly, a holder should have the same adjusted issue price, adjusted basis and holding period in the new notes as it had in the outstanding notes immediately before the exchange.

FEDERAL INCOME TAX CONSIDERATIONS TO NON-U.S. HOLDERS OF OWNERSHIP AND DISPOSITION OF NEW NOTES

The following discussion summarizes certain U.S. federal income tax consequences of the ownership and disposition of the new notes by an initial holder of outstanding notes who is a non-U.S. holder. This discussion is based upon the Code, existing and proposed Treasury Regulations, and judicial decisions and administrative interpretations thereunder, as of the date hereof, all of which are subject to change, possibly with retroactive effect, or are subject to different interpretations. We cannot assure you that the Internal Revenue Service (the "IRS") will not challenge one or more of the tax consequences described herein, and we have not obtained, nor do we intend to obtain, a ruling from the IRS or an opinion of counsel with respect to the U.S. federal tax consequences of acquiring, holding or disposing of the notes.

In this discussion, we do not purport to address all tax considerations that may be important to a particular non-U.S. holder in light of the non-U.S. holder's circumstances, or to certain categories of investors (such as certain financial institutions, insurance companies, tax-exempt organizations, dealers in securities, persons who hold the notes through partnerships or other pass-through entities, U.S. expatriates, or persons who hold the new notes as part of a hedge, conversion transaction, straddle or other risk reduction transaction) that may be subject to special rules. This discussion is limited to initial non-U.S. holders who purchased the outstanding notes for cash at the original offering price and who held those notes and will hold the new notes received in exchange therefor as capital assets. This discussion also does not address the tax considerations arising under the laws of any foreign, state or local jurisdiction.

YOU SHOULD CONSULT YOUR OWN TAX ADVISORS AS TO THE PARTICULAR TAX CONSEQUENCES TO YOU OF THE ACQUISITION, OWNERSHIP AND DISPOSITION OF THE NEW NOTES, INCLUDING THE EFFECT AND APPLICABILITY OF STATE, LOCAL OR FOREIGN TAX LAWS.

You are a non-U.S. holder for purposes of this discussion if you are not:

- o an individual U.S. citizen or resident alien;
- a corporation, or other entity taxable as a corporation for U.S.
 federal income tax purposes, that was created or organized in or under
 U.S. law (federal or state);
- an estate whose world-wide income is subject to U.S. federal income taxation; or
- o a trust that either is subject to the supervision of a court within the United States and which has one or more U.S. persons with authority to control all substantial decisions, or has a valid election in effect under applicable U.S. Treasury regulation to be treated as a U.S. person.

If a partnership holds notes, the tax treatment of a partner generally will depend upon the status of the partner and upon the activities of the partnership. If you are a partner of a partnership holding outstanding notes, we suggest that you consult your tax advisor.

U.S. Federal Withholding Tax

The 30% U.S. federal withholding tax will not apply to any payment of principal or interest (which, for purposes of this discussion, includes the accrual of original issue discount) to you on the new notes provided that:

- you do not actually (or constructively) own 10% or more of the total combined voting power of all classes of our voting stock within the meaning of the Code and the U.S. Treasury regulations;
- you are not a controlled foreign corporation that is related to us through stock ownership; and
- o you are not a bank whose receipt of interest on the new notes is pursuant to a loan agreement entered into in the ordinary course of business.

In each case, (a) you must provide your name and address on an IRS Form W-8BEN (or successor form), and certify under penalty of perjury, that you are not a U.S. person, (b) a financial institution holding the new notes on your behalf must certify, under penalty of perjury, that it has received an IRS Form W-8BEN (or successor form) from you and must provide us with a copy, or (c) you must hold your new notes directly through a "qualified intermediary," and the qualified intermediary must have sufficient information in its files indicating that you are a non-U.S. holder. A qualified intermediary is a bank, broker or other intermediary that is acting out of a non-U.S. branch or office and has signed an agreement with the IRS providing that it will administer all or part of the U.S. tax withholding rules under specified procedures.

If you cannot satisfy the requirements described above, payments of principal and interest made to you will be subject to the 30% U.S. federal withholding tax, unless you provide us with a properly executed (1) IRS Form W-8BEN (or successor form) claiming an exemption from (or a reduction of) withholding under the benefit of a tax treaty or (2) IRS Form W-8ECI (or successor form) stating that interest paid on the new notes is not subject to withholding tax because it is effectively connected with your conduct of a trade or business in the United States.

The 30% U.S. federal withholding tax generally will not apply to any gain or income that you realize on the sale, exchange, or other disposition of the new notes.

U.S. Federal Estate Tax

If you are an individual, your estate will not be subject to U.S. federal estate tax on new notes beneficially owned by you at the time of your death, provided that (1) you do not own 10% or more of the total combined voting power of all classes of our voting stock (within the meaning of the Code and the U.S. Treasury Regulations) and (2) interest on such notes would not have been, if received at the time of your death, effectively connected with the conduct by you of a trade or business in the United States.

U.S. Federal Income Tax

If you are engaged in a trade or business in the United States and interest on the new notes is effectively connected with the conduct of that trade or business, you will be subject to U.S. federal income tax on the interest on a net income basis (although exempt from the 30% withholding tax) in the same manner as if you were a U.S. person as defined under the Code. In addition, if you are a foreign corporation, you may be subject to a branch profits tax equal to 30% (or lower applicable treaty rate) of your earnings and profits for the taxable year, including earnings and profits from an investment in the new notes, that are effectively connected with the conduct by you of a trade or business in the United States.

Any gain or income realized on the sale, exchange, or redemption of the new notes generally will not be subject to U.S. federal income tax unless:

67

- o that gain or income is effectively connected with the conduct of a trade or business in the United States by you,
- o you are an individual who is present in the United States for 183 days or more in the taxable year of that disposition, and certain other conditions are present, or
- o the gain represents accrued interest, in which case the rules for interest would apply.

Backup Withholding and Information Reporting

Backup withholding and information reporting will not apply to payments of principal and interest on the new notes by us or our agent to you if you certify as to your non-U.S. holder status under penalties of perjury or you otherwise qualify for an exemption (provided that neither we nor our agent know or have reason to know that you are a U.S. person or that the conditions of any other exemptions are not in fact satisfied).

The payment of the proceeds of the disposition of new notes to or through the U.S. office of a U.S. or foreign broker will be subject to information reporting and backup withholding unless you provide the certification described above or you otherwise qualify for an exemption. The proceeds of a disposition effected outside the United States by a non-U.S. holder to or through a foreign office of a broker generally will not be subject to backup withholding or information reporting. However, if such broker is a U.S. person, a controlled foreign corporation for U.S. tax purposes, a foreign person 50% or more of whose gross income from all sources for certain periods is effectively connected with a trade or business in the United States, or a foreign partnership that is engaged in the conduct of a trade or business in the United States or that has one or more partners that are U.S. persons who in the aggregate hold more than 50 percent of the income or capital interests in the partnership, information reporting requirements will apply unless such broker has documentary evidence in its files of your non-U.S. status and has no actual knowledge or reason to know to the contrary or unless you otherwise qualify for an exemption. Any amount withheld under the backup withholding rules will be refunded or is allowable as a credit against your federal income tax liability, if any, provided the required information or appropriate claim for refund is provided to the IRS.

PLAN OF DISTRIBUTION

Based on interpretations by the staff of the Securities and Exchange Commission in no action letters issued to third parties, we believe that you may transfer new notes issued under the exchange offer in exchange for the outstanding notes if:

- o you acquire the new notes in the ordinary course of your business; and
- o you are not engaged in, and do not intend to engage in, and have no arrangement or understanding with any person to participate in, a distribution of such new notes.

You may not participate in the exchange offer if you are:

- o our "affiliate" within the meaning of Rule 405 under the Securities Act of 1933; or
- o a broker-dealer that acquired outstanding notes directly from us.

Each broker-dealer that receives new notes for its own account pursuant to the exchange offer must acknowledge that it will deliver a prospectus in connection with any resale of such new notes. To date, the staff of the Securities and Exchange Commission has taken the position that broker-dealers may fulfill their prospectus delivery requirements with respect to transactions involving an exchange of securities such as this exchange offer, other than a resale of an unsold allotment from the original sale of the outstanding notes, with the prospectus contained in this registration statement. This prospectus, as it may be amended or supplemented from time to time, may be used by a broker-dealer in connection with resales of new notes received in exchange for outstanding notes where such outstanding notes were acquired as a result of market-making activities or other trading activities. We have agreed that, for a period of up to 180 days after the effective date of this registration statement, we will make this prospectus, as amended or supplemented, available to any broker-dealer for use in connection with any such resale. In addition, until such date, all dealers effecting transactions in new notes may be required to deliver a prospectus.

If you wish to exchange new notes for your outstanding notes in the exchange offer, you will be required to make representations to us as described in "Exchange Offer--Purpose and Effect of the Exchange Offer" and "--Procedures for Tendering--Your Representations to Us" in this prospectus and in the letter of transmittal. In addition, if you are a broker-dealer who receives new notes for your own account in exchange for outstanding notes that were acquired by you as a result of market-making activities or other trading activities, you will be required to acknowledge that you will deliver a prospectus in connection with any resale by you of such new notes.

We will not receive any proceeds from any sale of new notes by broker-dealers. New notes received by broker-dealers for their own account pursuant to the exchange offer may be sold from time to time in one or more transactions in the over-the-counter market:

- o in negotiated transactions;
- through the writing of options on the new notes or a combination of such methods of resale;
- o at market prices prevailing at the time of resale; and
- at prices related to such prevailing market prices or negotiated prices.

Any such resale may be made directly to purchasers or to or through brokers or dealers who may receive compensation in the form of commissions or concessions from any such broker-dealer or the purchasers of any such new notes. Any broker-dealer that resells new notes that were received by it for its own account pursuant to the exchange offer and any broker or dealer that participates in a distribution of such new notes may be deemed to be an "underwriter" within the meaning of the Securities Act of 1933. The letter of transmittal states that by

acknowledging that it will deliver and by delivering a prospectus, a broker-dealer will not be deemed to admit that it is an "underwriter" within the meaning of the Securities Act of 1933.

For a period of 180 days after the effective date of this registration statement, we will promptly send additional copies of this prospectus and any amendment or supplement to this prospectus to any broker-dealer that requests such documents in the letter of transmittal. We have agreed to pay all expenses incident to the exchange offer (including the expenses of one counsel for the holders of the outstanding notes) other than commissions or concessions of any broker-dealers and will indemnify the holders of the outstanding notes (including any broker-dealers) against certain liabilities, including liabilities under the Securities Act of 1933.

LEGAL MATTERS

The validity of the new notes offered in this exchange offer will be passed upon for us by Vinson & Elkins L.L.P.

EXPERTS

The annual consolidated financial statements incorporated by reference in this prospectus have been audited by Arthur Andersen LLP, independent public accountants, as indicated in their report with respect thereto, and included herein in reliance upon the authority of said firm as experts in giving said report.

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the SEC. You may inspect and copy such material at the public reference facilities maintained by the SEC at Room 1024, Judiciary Plaza, 450 Fifth Street, N.W, Washington, D.C. 20549, as well as at the SEC's regional offices at 500 West Madison Street, Suite 1400, Chicago, Illinois 60661 and 7 World Trade Center, Suite 1300, New York, New York 10048. Please call the SEC at 1-800-SEC-0330 for more information on the public reference rooms. You can also find our SEC filings at the SEC's website at www.sec.gov and on our website at www.ielectric.com. Information contained on our website is not part of this prospectus.

In addition, reports, proxy statements and other information concerning us can be inspected at the NYSE, 20 Broad Street, New York, New York 10005, where our common stock is listed.

The following documents we filed with the SEC pursuant to the Exchange \mbox{Act} are incorporated herein by reference:

- Annual Report on Form 10-K for the fiscal year ended September 30, 2000;
- Proxy Statement dated December 28, 2000 for our Annual Meeting of Stockholders;
- 3. Current Report on Form 8-K dated March 24, 2001; and
- Quarterly Reports on Form 10-Q for the quarters ended December 31, 2000 and March 31, 2001.

All documents filed by us pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act subsequent to the date of this prospectus and prior to the termination of the notes offering shall be deemed to be incorporated in this prospectus and to be a part hereof from the date of the filing of such document. Any statement contained in a document incorporated by reference herein shall be deemed to be modified or superseded for all purposes to the extent that a statement contained in this prospectus, or in any other subsequently filed document which is also incorporated or deemed to be incorporated by reference, modifies or supersedes such statement. Any statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this prospectus.

We will provide without charge to each person to whom this prospectus is delivered, upon written or oral request of such person, a copy of any or all documents incorporated by reference in this prospectus. Requests for such copies should be directed to John F. Wombwell, Executive Vice President and General Counsel, Integrated, Electrical Services, Inc., 1800 West Loop South, Suite 500, Houston, Texas 77027, by mail, and if by telephone at (713) 860-1500.

FORWARD-LOOKING STATEMENTS

This prospectus includes forward-looking statements. We have based these forward-looking statements on our current expectations and projections about future events. These forward-looking statements are subject to risks, uncertainties and assumptions about us, including, among other things:

- o inherent uncertainties relating to estimating future results;
- o fluctuation in operating results because of downturns in levels of construction;
- o incorrect estimates used in entering into fixed price contracts;
- difficulty in managing the operation and growth of existing and newly acquired businesses;
- o our ability to incur additional debt in order to fund working capital or acquisitions;
- $\ensuremath{\mathsf{o}}$ $\ensuremath{\mathsf{the}}$ high level of competition in the construction industry; and
- o seasonality.

We undertake no obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise. In light of these risks, uncertainties and assumptions, the forward-looking events discussed in this prospectus might not occur.

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

ITEM 20. INDEMNIFICATION OF OFFICERS AND DIRECTORS

Subsection (a) of section 145 of the General Corporation Law of the State of Delaware empowers a corporation to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation) by reason of the fact that he is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him in connection with such action, suit or proceeding if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful

Subsection (b) of Section 145 empowers a corporation to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, or suit by or in the right of the corporation to procure a judgment in its favor by reason of the fact that such person acted in any of the capacities set forth above, against expenses (including attorneys' fees) actually and reasonably incurred by him in connection with the defense or settlement of such action or suit if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the corporation, except that no indemnification may be made in respect of any claim, issue or matter as to which such person shall have been made to be liable to the corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

Section 145 further provides that to the extent a director or officer of a corporation has been successful on the merits or otherwise in the defense of any action, suit or proceeding referred to in subsections (a) and (b) of Section 145 in the defense of any claim, issue or matter therein, he shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by him in connection therewith; that indemnification provided for by Section 145 shall not be deemed exclusive of any other rights to which the indemnified party may be entitled; that indemnification provided for by Section 145 shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of such person's heirs, executors and administrators; and empowers the corporation to purchase and maintain insurance on behalf of a director or officer of the corporation against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such whether or not the corporation would have the power to indemnify him against such liabilities under Section 145.

Section 102(b)(7) of the General Corporation Law of the State of Delaware provides that a certificate of incorporation may contain a provision eliminating or limiting the personal liability of a director to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director provided that such provision shall not eliminate or limit the liability of a director (1) for any breach of the director's duty of loyalty to the corporation or its stockholders, (2) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (3) under Section 174 of the Delaware General Corporation Law, or (4) for any transaction from which the director derived an improper personal benefit.

ITEM 21. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

(a)Exhibits. The following exhibits are filed herewith pursuant to the requirements of Item 601 of Regulation S-K:

Exhibit	
No.	Description
3.1	Amended and Restated Certificate of Incorporation as amended. (Incorporated herein by reference to Exhibit 3.1 to the Registration Statement on Form S-1 (File No. 333-38715) of the Company).
3.2* 4.2	Bylaws, as amended. Indenture, dated January 28, 1999, by and among Integrated Electrical Services, Inc. and the subsidiaries named therein and State Street Bank and Trust Company covering up to \$150,000,000 9 3/8% Senior Subordinated Notes due 2009. (Incorporated herein by reference to Exhibit 4.2 to Post-Effective Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-50031) of the Company).
4.3*	Indenture, dated as of May 29, 2001 by and among Integrated Electrical Services, Inc., the subsidiaries name therein and State Street Bank and Trust Company.
4.4	Form of Integrated Electrical Services, Inc. 9 3/8% Senior Subordinated Note due 2009 (Series A) (Included in Exhibit A to Exhibit 4.2 to Post-Effective Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-50031) of the Company).
4.5	Form of Integrated Electrical Services, Inc. 9 3/8% Senior Subordinated Note due 2009 (Series B) (Included in Exhibit A to Exhibit 4.2 to Post-Effective Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-50031) of the Company).
4.6*	Form of Integrated Electrical Services, Inc. 9[]% Senior Subordinated Note due 2009 (Series C) (Included in Exhibit A to the Indenture, dated as of May 29, 2001, filed herewith as Exhibit 4.3)
4.7* 5.1*	Exchange and Registration Rights Agreement dated as of May 29, 2001 by and between Integrated Electrical Services, Inc. and the initial purchasers named therein. Opinion of Vinson & Elkins L.L.P. regarding the validity of
10.1+	the securities being registered. Form of Employment Agreement (Incorporated herein by
10.2	reference to Exhibit 10.1 to the Registration Statement on Form S-1 (File No. 333-38715) of the Company). Form of Officer and Director Indemnification Agreement (Incorporated herein by reference to Exhibit 10.2 to the
10.3+	Registration Statement on Form S-1 (File No. 333-38715) of the Company). Integrated Electrical Services, Inc. 1997 Stock Plan, as amended. (Incorporated herein by reference to Exhibit 10.1 of the Company's Quarterly Report on Form 10-Q for the quarter
10.4+	ended June 30, 1999). Integrated Electrical Services, Inc. 1997 Directors Stock Plan (Incorporated herein by reference to Exhibit 16.4 of the Company's Annual Report on Form 10-K for the year ended
10.5	September 30, 2000). Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A., including Guaranty, Pledge Agreement, Security Agreement, form of promissory note, and form of swing line note. (Incorporated herein by reference to Exhibit 10.5 to Post-Effective Amendment No. 1 to the Registration Statement
10.6	on Form S-1 (File No. 333-50031) of the Company). Amendment No. 1 dated September 30, 1998, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.6 to the Company's Annual Report on Form 10-K/A for the year
10.7	ended September 30, 1998). Amendment No. 2 dated January 18, 1999, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.7 to Post-Effective Amendment No. 2 to the Registration
10.8	Statement on Form S-1 (Reg. No. 333-50031) of the Company). Amendment No. 3 dated August 19, 1999, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.9 of the Company's Annual Report on Form 10-K for the year ended September 30, 2000).
10.9	Amendment No. 4 dated March 31, 2000, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.1 of the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2000).
10.10+	March 31, 2000). Employment Agreement between the Company and H. David Ramm dated March 20, 2000 (Incorporated herein by reference to Exhibit 10.2 of the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2000).

75

- 10.11+ Integrated Electrical Services, Inc. 1999 Incentive Compensation Plan (Incorporation herein by reference to Exhibit 10.11 of the Company's Annual Report on Form 10-K for the year ended September 30, 2000).
- 10.12* Credit Agreement dated as of May 22, 2001 among Integrated Electrical Services, Inc., as borrower, the financial institutions named therein, as banks, Credit Lyonnais and the Bank of Nova Scotia, as syndication agents, Toronto Dominion (Texas), Inc., as documentation agent and The Chase Manhattan Bank, as administrative agent.
- 12* Ratio of Earnings to Fixed Charges.
- 21.1 List of Subsidiaries (Incorporated herein by reference to Exhibit 21.1 of the Company's Annual Report on Form 10-K for the year ended September 30, 2000).
- 23.1* Consent of Arthur Andersen LLP
- 23.7* Consent of Vinson & Elkins L.L.P. (included in Exhibit 5.1).
- 24.1* Power of Attorney (included in the signature pages of this Registration Statement).
- 25.1* Statement of Eligibility on Form T-1 of State Street Bank and Trust Company.
- 99.1* Form of Letter of Transmittal.
- 99.2* Form of Letter to Clients.
- 99.3* Form of Letter to Registered Holders and DTC Participants.
- 99.4* Form of Notice of Guaranteed Delivery.

* Filed herewith.

Management contract or compensatory plan or arrangement.

(b) Financial Statement Schedules. Incorporated herein by reference to Item 8 of our annual report on Form 10-K for the year ended September 30, 2000, as amended.

ITEM 22. UNDERTAKINGS

Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of any Registrant, we have been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by any Registrant of expenses incurred or paid by a director, officer or controlling person of such Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, such Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

Each Registrant hereby undertakes:

(1) To respond to requests for information that is Incorporated herein by reference into the prospectus pursuant to Items 4, 10(b), 11 or 13 of Form S-4, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the Registration Statement through the date of responding to the request;

(2) That, for purposes of determining any liability under the Securities Act of 1933, each filing of the Registrant's annual report pursuant to section 13(a) or section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to section 15(d) of the Securities Exchange Act of 1934) that is Incorporated herein by reference in the Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof; and

(3) To supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired therein, that was not the subject of and included in the Registration Statement when it became effective.

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

INTEGRATED ELECTRICAL SERVICES, INC.

By: /s/ H. DAVID RAM

H. David Ramm President, Chief Executive Officer and Director

Signature	Capacity	Date
/s/ H. DAVID RAMM	President, Chief Executive Officer and Director (Principal Executive Officer)	July 5, 2001
H. David Ramm		
/s/ WILLIAM W. REYNOLDS	Executive Vice President and Chief Financial	July 5, 2001
William W. Reynolds	Officer (Principal Financial Officer)	
/s/ NEIL J. DePASCAL, JR.	Vice President, Treasurer and Chief	July 5, 2001
Neil J. DePascal, Jr.	Accounting Officer (Principal Accounting Officer)	
/s/ C. BYRON SNYDER	Chairman of the Board of Directors	July 5, 2001
C. Byron Snyder		
/s/ HERBERT R. ALLEN	Director	July 5, 2001
Herbert R. Allen		
/s/ RICHARD L. CHINA	Director	July 5, 2001
Richard L. China		
/s/ JOHN A. COSENTINO, JR.	Director	July 5, 2001
John A. Cosentino, Jr.		
/s/ DONALD PAUL HODEL	Director	July 5, 2001
Donald Paul Hodel	DITECTO	JULY 3, 2001
Donatu Paul houel		
/s/ ROBERT C. KELLY	Director	July 5, 2001
Robert C. Kelly		
/s/ BEN L. MUELLER	Director	July 5, 2001
Ben L. Mueller		

Signature	Capacity	Date
/s/ RICHARD MUTH	Director	July 5, 2001
Richard Muth		
Alan R. Sielbeck	Director	
/s/ RICHARD L. TUCKER	Director	July 5, 2001
Richard L. Tucker		
/s/ BOB WEIK	Director	July 5, 2001
Bob Weik		
/s/ JIM P. WISE	Director	July 5, 2001
Jim P. Wise		
/s/ JAMES D. WOODS	Director	July 5, 2001
James D. Woods		

II-5

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

> DKD ELECTRIC COMPANY, INC. NBH HOLDING CO., INC. (F/K/A DKD ACQUISITION CORPORATION) POLLOCK SUMMIT HOLDINGS, INC. TESLA POWER (NEVADA), INC.

BY: /s/ ADRIANNE HORNE

ADRIANNE HORNE CHIEF EXECUTIVE OFFICER AND DIRECTOR

Each person whose signature appears below authorizes Adrianne Horne, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE

CAPACITY

DATE

/S/ ADRIANNE HORNE

Adrianne Horne

Chief Executive Officer and Director (Principal Executive, Financial and Accounting Officer) July 5, 2001

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

> BRITT RICE ELECTRIC, INC. BRYANT ELECTRIC COMPANY, INC. BW CONSOLIDATED, INC. BW/BEC, INC. CROSS STATE ELECTRIC, INC. CYPRESS STATE ELECTRIC, INC. CYPRESS ELECTRICAL CONTRACTORS, INC. DAVIS ELECTRICAL CONSTRUCTORS, INC. ELECTRO-TECH, INC. ERNEST P. BREAUX ELECTRICAL, INC. H.R. ALLEN, INC. HOUSTON-STAFFORD ELECTRIC, INC. HOWARD BROTHERS ELECTRIC CO., INC. I.C.G. ELECTRIC, INC. INNOVATIVE ELECTRIC COMPANY, INC. (F/K/A THURMAN & O'CONNELL CORP.) J.W. GRAY ELECTRIC COMPANY, INC. KEY ELECTRICAL SUPPLY, INC. MITCHELL ELECTRIC COMPANY, INC. PAN AMERICAN ELECTRIC COMPANY, INC., NEW MEXICO PAN AMERICAN ELECTRIC, INC. PAULIN ELECTRIC COMPANY, INC. RKT ELECTRIC, INC. RNT ELECTRIC, INC. ROCKWELL ELECTRIC, INC. SPECTROL, INC. SPOOR ELECTRIC, INC. (D/B/A SEI ELECTRICAL CONTRACTOR) T&H ELECTRICAL CORPORATION TECH ELECTRIC CO., INC. WRIGHT ELECTRICAL CONTRACTING, INC.

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(Principal Executive Orricer)	
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds	(Frincipal Financial Officer)	
/s/ NEIL J. DePASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.		
/s/ BEN L. MUELLER	Director	July 5, 2001
Ben L. Mueller		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001. IST GROUP TELECOMMUNICATIONS, INC. (F/K/A BRYANT ACQUISITION CORPORATION) ACE ELECTRIC, INC. ALADDIN WARD ELECTRIC & AIR, INC. AMBER ELECTRIC, INC. ANDERSON & WOOD CONSTRUCTION CO., INC. BACHOFNER ELECTRIC, INC. BRINK ELECTRIC, INC. (F/K/A PAN AMERICAN ACQUISITION CORPORATION) CHARLES P. BAGBY CO., INC. COLLIER ELECTRIC COMPANY, INC. DANIEL ELECTRIC COMPANY, INC. DANIEL ELECTRICAL OF TREASURE COAST, INC. FEDERAL COMMUNICATIONS GROUP, INC. FLORIDA INDUSTRIAL ELECTRIC, INC. GOSS ELECTRIC COMPANY, INC. HOLLAND ELECTRIC COMPANY, INC.

KAYTON ELECTRIC, INC.

MENNINGA ELECTRIC, INC. MIDLANDS ELECTRICAL CONTRACTORS, INC. MURRAY ELECTRICAL CONTRACTORS, INC. MUTH ELECTRIC, INC. NEW TECHNOLOGY ELECTRICAL CONTRACTORS, INC. PRIMENET, INC. (F/K/A STUTTS ACQUISITION CORPORATION) PUTZEL ELECTRICAL CONTRACTORS, INC. RODGERS ELECTRIC COMPANY, INC. RODGERS ELECTRIC COMPANY, INC. RON'S ELECTRIC, INC. WOLFE ELECTRIC CO., INC. BY: /s/ H. DAVID RAMM H. DAVID RAMM CHIEF EXECUTIVE OFFICER AND DIRECTOR

LINEMEN, INC. (D/B/A CALIFORNIA COMMUNICATIONS) MARK HENDERSON, INCORPORATED

INTELLIGENT BUILDING SOLUTIONS, INC.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer and Director (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds		
/s/ NEIL J. DePASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.	(FILICIPAL ACCOUNTING UTILET)	

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

> ARC ELECTRIC, INCORPORATED BARTLEY & DEVARY ELECTRIC, INC. BEAR ACQUISITION CORPORATION CANOVA ELECTRICAL CONTRACTING, INC. COMMERCIAL ELECTRICAL CONTRACTORS, INC. DELCO ELECTRIC, INC. EMC ACQUISITION CORPORATION HATFIELD REYNOLDS ELECTRIC COMPANY (F/K/A HATFIELD ELECTRIC, INC.) (F/K/A HATFIELD ELECTRIC, INC.)
> IES COMMUNICATIONS GROUP, INC.
> (F/K/A IES COMMUNICATIONS INC.)
> IES ELECTRICAL GROUP, INC.
> (F/K/A INTEGRATED COMMUNICATION SERVICES, INC.) IES RESIDENTIAL GROUP, INC. IES SPECIALTY LIGHTING, INC. (F/K/A MODERN ACQUISITION CORPORATION) IES VENTURES INC. INTEGRATED ELECTRICAL FINANCE, INC. MID-STATES ELECTRIC COMPANY, INC. MILLS ELECTRICAL CONTRACTORS, INC. M-S SYSTEMS, INC. NEWCOMB ELECTRIC COMPANY, INC. POLLOCK ELECTRIC, INC. PRIMO ELECTRIC COMPANY (F/K/A HAMER ELECTRIC ACQUISITION, INC.) RAINES ELECTRIC CO., INC. SUMMIT ELECTRIC OF TEXAS, INCORPORATED TESLA POWER G.P., INC. THOMAS POPP & COMPANY VALENTINE ELECTRICAL, INC.

BY: /s/ H. DAVID RAMM H. DAVID RAMM CHIEF EXECUTIVE OFFICER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds /s/ NEIL J. DEPASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr. /s/ JOHN F. WOMBWELL	Director	July 5, 2001
John F. Wombwell		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

> BRITT RICE MANAGEMENT LLC HOUSTON-STAFFORD MANAGEMENT LLC J.W. GRAY MANAGEMENT, LLC

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
/s/ NEIL J. DEPASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
/s/ BEN L. MUELLER	Manager	July 5, 2001

Ben L. Mueller

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

CARROLL MANAGEMENT LLC

BY: /s/ H. DAVID RAMM H. DAVID RAMM CHIEF EXECUTIVE OFFICER AND MANAGER

SIGNATURE	CAPACITY		DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer and Manager (Principal Executive Officer)	July	5, 2001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July	5, 2001
/s/ NEIL J. DEPASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July	5, 2001

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

IES CONTRACTORS MANAGEMENT LLC MILLS MANAGEMENT LLC NEAL ELECTRIC MANAGEMENT LLC (F/K/A ICS MANAGEMENT LLC) RAINES MANAGEMENT LLC

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY		DATE	
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July	5, 20	901
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July	5, 20	901
/s/ NEIL J. DEPASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July	5, 20	901
/s/ JOHN F. WOMBWELL John F. Wombwell	Manager	July	5, 20	901

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

BRITT RICE HOLDINGS LLC BW/BEC, L.L.C. CARROLL HOLDINGS LLC HOUSTON STAFFORD HOLDINGS, LLC ICS HOLDINGS LLC IES CONTRACTORS HOLDINGS LLC IES HOLDINGS, LLC J.W. GRAY HOLDINGS, LLC MILLS ELECTRICAL HOLDINGS, LLC RAINES HOLDINGS LLC

BY: /s/ ADRIANNE HORNE ADRIANNE HORNE CHIEF EXECUTIVE OFFICER AND MANAGER

SIGNATURE	CAPACITY	DATE
/s/ ADRIANNE HORNE	Chief Executive Officer and Manager (Principal Executive, Financial and	July 5, 2001
Adrianne Horne	Accounting Officer)	

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

B. RICE ELECTRIC LP

BY: BRITT RICE MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm /s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds /s/ NEIL J. DEPASCAL, JR.	Chief Accounting Officer	July 5, 2001
Neil J. DePascal, Jr.	(Principal Accounting Officer)	_
/s/ BEN L. MUELLER Ben L. Mueller	Manager	July 5, 2001

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

BEXAR ELECTRIC COMPANY LTD.

BY: BW/BEC, INC., ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes Adrianne Horne, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
/s/ NEIL J. DEPASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
/s/ BEN L. MUELLER Ben L. Mueller	Director	July 5, 2001

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

CARROLL SYSTEMS LP

BY: CARROLL MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER AND MANAGER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer and Manager (Principal Executive Officer)	July 5, 2001
H. David Ramm		
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds		
/s/ NEIL J. DePASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

HAYMAKER ELECTRIC, LTD

BY: GENERAL PARTNER, INC., ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM

CHIEF EXECUTIVE OFFICER AND DIRECTOR

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer and Director (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
/s/ NEIL J. DePASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

HOUSTON-STAFFORD ELECTRICAL CONTRACTORS LP

BY: HOUSTON-STAFFORD MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds	(Frincipal Financial Officer)	
/s/ NEIL J. DePASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.		
/s/ BEN L. MUELLER	Manager	July 5, 2001
Ben L. Mueller		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

ICS INTEGRATED COMMUNICATIONS SERVICES LP

BY: NEAL ELECTRIC MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
/s/ NEIL J. DePASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
/s/ JOHN F. WOMBWELL John F. Wombwell	Manager	July 5, 2001

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

IES CONTRACTORS LP

BY: IES CONTRACTORS MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds /s/ NEIL J. DEPASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr. /s/ JOHN F. WOMBWELL	Manager	July 5, 2001
John F. Wombwell		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

IES MANAGEMENT LP

BY: INTEGRATED ELECTRICAL FINANCE, INC., ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(Trinoipar Excouring officially	
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds	(Frincipal Financial Officer)	
/s/ NEIL J. DePASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.	(Frincipal Accounting Officer)	
/s/ JOHN F. WOMBWELL	Manager	July 5, 2001
John F. Wombwell		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

J.W. GRAY ELECTRICAL CONTRACTORS LP

BY: J.W. GRAY MANAGEMENT, LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
/s/ NEIL J. DePASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
/s/ JOHN F. WOMBWELL	Manager	July 5, 2001

John F. Wombwell

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

MILLS ELECTRIC LP

BY: MILLS MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(Frincipal Executive officer)	
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer	July 5, 2001
William W. Reynolds	(Principal Financial Officer)	
/s/ NEIL J. DEPASCAL, JR.	Chief Accounting Officer	July 5, 2001
Neil J. DePascal, Jr.	(Principal Accounting Officer)	
/s/ JOHN F. WOMBWELL	Manager	July 5, 2001
John F. Wombwell		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

NEAL ELECTRIC LP

BY: BW/BEC, INC., ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(Trinoipal Excountre officier)	
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds	(Frincipal Financial Officer)	
/s/ NEIL J. DEPASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.	(Frincipal Accounting Officer)	
/s/ BEN L. MUELLER	Director	July 5, 2001
Ben L. Mueller		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

POLLOCK SUMMIT ELECTRIC LP

BY: POLLOCK ELECTRIC, INC., ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

- BY: SUMMIT ELECTRIC OF TEXAS, INCORPORATED, ITS GENERAL PARTNER
- BY: /s/ H. DAVID RAMM
 - H. DAVID RAMM CHIEF EXECUTIVE OFFICER

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(Trincipal Executive Officer)	
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer (Principal Financial Officer)	July 5, 2001
William W. Reynolds	(FILICIPAL FINANCIAL OFFICE)	
/s/ NEIL J. DEPASCAL, JR.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2001
Neil J. DePascal, Jr.		
/s/ JOHN F. WOMBWELL	Manager	July 5, 2001
John F. Wombwell		

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

RAINES ELECTRIC LP

BY: RAINES MANAGEMENT LLC, ITS GENERAL PARTNER

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE	
/s/ H. DAVID RAMM H. David Ramm	Chief Executive Officer (Principal Executive Officer)	July 5, 2	001
/s/ WILLIAM W. REYNOLDS William W. Reynolds	Chief Financial Officer (Principal Financial Officer)	July 5, 2	001
/s/ NEIL J. DEPASCAL, JR. Neil J. DePascal, Jr.	Chief Accounting Officer (Principal Accounting Officer)	July 5, 2	001
/s/ JOHN F. WOMBWELL	Manager	July 5, 2	001

John F. Wombwell

Pursuant to the requirements of the Securities Act of 1933, as amended, the company has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Houston, State of Texas, on the 5th day of July, 2001.

TESLA POWER AND AUTOMATION, LP TESLA POWER PROPERTIES, LP

BY: TESLA POWER, GP INC., THE GENERAL PARTNER OF EACH RESPECTIVE ENTITY

BY: /s/ H. DAVID RAMM

H. DAVID RAMM CHIEF EXECUTIVE OFFICER

Each person whose signature appears below authorizes H. David Ramm and William W. Reynolds, and each of them, each of whom may act without joinder of the other, to execute in the name of each such person who is then an officer or director of the company and to file any amendments to this registration statement necessary or advisable to enable the company to comply with the Securities Act of 1933, as amended, and any rules, regulations and requirements of the Securities and Exchange Commission in respect thereof, in connection with the registration of the securities which are the subject of this registration statement, which amendments may make such changes in the registration statement as such attorney may deem appropriate. Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following person in the capacities and on the date indicated.

SIGNATURE	CAPACITY	DATE
/s/ H. DAVID RAMM	Chief Executive Officer (Principal Executive Officer)	July 5, 2001
H. David Ramm	(Principal Executive Officer)	
/s/ WILLIAM W. REYNOLDS	Chief Financial Officer	July 5, 2001
William W. Reynolds	(Principal Financial Officer)	
/s/ NEIL J. DePASCAL, JR.	Chief Accounting Officer	July 5, 2001
Neil J. DePascal, Jr.	(Principal Accounting Officer)	
/s/ JOHN F. WOMBWELL	Director	July 5, 2001
John F. Wombwell	-	

Exhibit	
No.	Description
3.1	Amended and Restated Certificate of Incorporation as amended. (Incorporated herein by reference to Exhibit 3.1 to the Registration Statement on Form S-1 (File No. 333-38715) of the Company).
3.2* 4.2	Bylaws, as amended. Indenture, dated January 28, 1999, by and among Integrated Electrical Services, Inc. and the subsidiaries named therein and State Street Bank and Trust Company covering up to \$150,000,000 9 3/8% Senior Subordinated Notes due 2009. (Incorporated herein by reference to Exhibit 4.2 to Post-Effective Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-50031) of the Company).
4.3*	Indenture, dated as of May 29, 2001 by and among Integrated Electrical Services, Inc., the subsidiaries name therein and State Street Bank and Trust Company.
4.4	Form of Integrated Electrical Services, Inc. 9 3/8% Senior Subordinated Note due 2009 (Series A) (Included in Exhibit A to Exhibit 4.2 to Post-Effective Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-50031) of the Company).
4.5	Form of Integrated Electrical Services, Inc. 9 3/8% Senior Subordinated Note due 2009 (Series B) (Included in Exhibit A to Exhibit 4.2 to Post-Effective Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-50031) of the Company).
4.6*	Form of Integrated Electrical Services, Inc. 9[]% Senior Subordinated Note due 2009 (Series C) (Included in Exhibit A to the Indenture, dated as of May 29, 2001, filed herewith as Exhibit 4.3)
4.7*	Exchange and Registration Rights Agreement dated as of May 29, 2001 by and between Integrated Electrical Services, Inc. and the initial purchasers named therein.
5.1*	Opinion of Vinson & Elkins L.L.P. regarding the validity of the securities being registered.
10.1+	Form of Employment Agreement (Incorporated herein by reference to Exhibit 10.1 to the Registration Statement on Form S-1 (File No. 333-38715) of the Company).
10.2	Form of Officer and Director Indemnification Agreement (Incorporated herein by reference to Exhibit 10.2 to the Registration Statement on Form S-1 (File No. 333-38715) of the Company).
10.3+	Integrated Electrical Services, Inc. 1997 Stock Plan, as amended. (Incorporated herein by reference to Exhibit 10.1 of the Company's Quarterly Report on Form 10-Q for the quarter
10.4+	ended June 30, 1999). Integrated Electrical Services, Inc. 1997 Directors Stock Plan (Incorporated herein by reference to Exhibit 16.4 of the Company's Annual Report on Form 10-K for the year ended
10.5	September 30, 2000). Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A., including Guaranty, Pledge Agreement, Security Agreement, form of promissory note, and form of swing line note. (Incorporated herein by reference to Exhibit 10.5 to
10.6	Post-Effective Amendment No. 1 to the Registration Statement on Form S-1 (File No. 333-50031) of the Company). Amendment No. 1 dated September 30, 1998, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.6 to the Company's Annual Report on Form 10-K/A for the year
10.7	ended September 30, 1998). Amendment No. 2 dated January 18, 1999, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.7 to Post-Effective Amendment No. 2 to the Registration
10.8	Statement on Form S-1 (Reg. No. 333-50031) of the Company). Amendment No. 3 dated August 19, 1999, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.9 of the Company's Annual Report on Form 10-K for the year
10.9	ended September 30, 2000). Amendment No. 4 dated March 31, 2000, to the Credit Agreement dated July 30, 1998, among the Company, the Financial Institutions named therein and NationsBank of Texas, N.A. (Incorporated herein by reference to Exhibit 10.1 of the Company's Quarterly Report on Form 10-Q for the quarter ended
10.10+	March 31, 2000). Employment Agreement between the Company and H. David Ramm dated March 20, 2000 (Incorporated herein by reference to Exhibit 10.2 of the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2000).

104

- 10.11+ Integrated Electrical Services, Inc. 1999 Incentive Compensation Plan (Incorporation herein by reference to Exhibit 10.11 of the Company's Annual Report on Form 10-K for the year ended September 30, 2000).
- 10.12* Credit Agreement dated as of May 22, 2001 among Integrated Electrical Services, Inc., as borrower, the financial institutions named therein, as banks, Credit Lyonnais and the Bank of Nova Scotia, as syndication agents, Toronto Dominion (Texas), Inc., as documentation agent and The Chase Manhattan Bank, as administrative agent.
- 12* Ratio of Earnings to Fixed Charges.
- 21.1 List of Subsidiaries (Incorporated herein by reference to Exhibit 21.1 of the Company's Annual Report on Form 10-K for the year ended September 30, 2000).
- 23.1* Consent of Arthur Andersen LLP
- 23.7* Consent of Vinson & Elkins L.L.P. (included in Exhibit 5.1).
- 24.1* Power of Attorney (included in the signature pages of this Registration Statement).
- 25.1* Statement of Eligibility on Form T-1 of State Street Bank and Trust Company.
- 99.1* Form of Letter of Transmittal.
- 99.2* Form of Letter to Clients.
- 99.3* Form of Letter to Registered Holders and DTC Participants.
- 99.4* Form of Notice of Guaranteed Delivery.

- -----

* Filed herewith.

+ Management contract or compensatory plan or arrangement.

(b) Financial Statement Schedules. Incorporated herein by reference to Item 8 of our annual report on Form 10-K for the year ended September 30, 2000, as amended.

BYLAWS

0F

INTEGRATED ELECTRICAL SERVICES, INC.

(AS AMENDED)

May 8, 2001

ARTICLE I

OFFICES

Section 1. The registered office of Integrated Electrical Services, Inc.(the "Corporation") shall be in the City of Wilmington, County of New Castle, State of Delaware.

Section 2. The Corporation may also have offices at such other places both within and without the state of Delaware as the Board of Directors may from time to time determine or the business of the Corporation may require.

ARTICLE II

MEETINGS OF STOCKHOLDERS

Section 1. All meetings of the stockholders for the election of Directors shall be held at such place as may be fixed from time to time by the Board of Directors and stated in the notice of the meeting. Meetings of stockholders for any other purpose may be held at such time and place, within or without the State of Delaware, as shall be stated in the notice of the meeting or in a duly executed waiver of notice thereof

Section 2. Annual meetings of stockholders shall be held on such date and at such time as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting. At the annual meeting, the stockholders shall elect by a plurality vote the Directors pursuant to Article III of these Bylaws, and transact such other business as may properly be brought before the meeting.

Section 3. Written notice of the annual meeting stating the place, date and hour of the meeting shall be given to each stockholder entitled to a vote at such meeting not less than ten (10) nor more than sixty (60) days before the date of the meeting.

At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly brought before an annual meeting, business must be (i) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors, (ii) otherwise properly brought before the meeting by or at the direction of the Board of Directors, or (iii) otherwise properly brought before an annual meeting by a stockholder. For business to be properly brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation. To be timely, a stockholder's notice must be delivered to or mailed to and received at the principal executive offices of the Corporation not less than 80 days prior to the meeting; provided, however, that in the event that less than 90 days' notice or prior public disclosure of the date of the meeting is given or made to stockholders, notice by the stockholder to be timely must be so received not later than the close of business on the tenth day following the date on which such notice of the date of the annual meeting was mailed or such public disclosure made. A stockholder's notice to the Secretary shall set forth as to each matter the stockholder proposes to bring before the annual meeting (a) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting, (b) the name and address, as they appear on the Corporation's books, of the stockholder proposing such business, (c) the class and number of shares of the Corporation which are beneficially owned by the stockholder, and (d) any material interest of the stockholder in such business. Notwithstanding anything in the Bylaws to the contrary, no business shall be conducted at an annual meeting except in accordance with the procedures set forth in this Section 3.

The presiding officer of an annual meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting in accordance with this Section 3, and if the presiding officer should so determine, the presiding officer shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

Section 4. The officer who has charge of the stock ledger of the Corporation shall prepare and make, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten (10) days prior to the meeting, either at a place within the city where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not so specified, at the place where the meeting is to be held. The list shall also be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present.

Section 5. Special meetings of the stockholders for any purpose may be called only by the Chairman of the Board of Directors and shall be called within 10 days after receipt of the written request of the Board of Directors, pursuant to a resolution approved by a majority of the entire Board of Directors. The business permitted to be conducted at any special meeting of the stockholders is limited to the business brought before the meeting by the Chairman or by the Secretary at the request of a majority of the entire Board of Directors.

Section 6. Written notice of a special meeting stating the place, date and hour of the meeting, and the purpose or purposes for which the meeting is called, shall be given not less than ten (10) nor more than sixty (60) days before the date of the meeting, to each stockholder entitled to vote at such meeting.

Section 7. The holders of a majority of the stock issued, outstanding and entitled to vote, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business except as otherwise provided by statute or by the certificate of incorporation. If, however, such quorum shall not be present or represented at any meeting of the stockholders, the stockholders entitled to vote thereat, present in person or

-2-

represented by proxy, shall have power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present or represented.

Section 8. When a meeting is adjourned to another time or place, notice need not be given of the adjourned meeting, except as otherwise required by this Section 8, if the time and place thereof are announced at the meeting at which the adjournment is taken. At such adjourned meeting the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting.

Section 9. If a quorum exists, action on a matter (other than the election of directors) shall be approved if the votes cast in favor of the matter exceed the votes cast opposing the matter. In determining the number of votes cast, shares abstaining from voting or not voted on a matter will not be treated as votes cast. The provisions of this paragraph will govern with respect to all votes of stockholders except as otherwise provided for in these Bylaws or in the certificate of incorporation or by a specific statutory provision superseding the provisions contained in these Bylaws or the certificate of incorporation.

Section 10. Each stockholder shall at every meeting of the stockholders, subject to any restriction or qualification set forth in the Certificate of Incorporation, be entitled to one vote in person or by proxy for each share of the capital stock having voting power held by such stockholder, but no proxy shall be voted after three years from its date, unless the proxy provides for a longer period.

Section 11. After March 1, 1998, any action required or permitted to be taken by the stockholders of the Corporation must be affected at a duly called annual or special meeting of stockholders of the Corporation and may not be effected by any consent in writing of such stockholders.

Section 12. At each meeting of stockholders, the Chairman or Vice-Chairman of the Board of Directors shall preside, and the secretary shall keep records, and in the absence of either such officer, his duty shall be performed by a person appointed at the meeting.

ARTICLE III

DIRECTORS

Number, Nomination, Removal

Section 1. The number of Directors shall be fixed from time to time by the Board of Directors, but shall not be less than 1 nor more than 15 persons. The Directors shall be elected at the annual meeting of the stockholders in accordance with the provisions of Section 2 of this Article, and each Director elected shall hold office until his successor is elected and qualified. Directors need not be stockholders.

5

Section 2. Subject to the rights of holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation, nominations for the election of Directors may be made by the Board of Directors or a committee appointed by the Board of Directors or by any stockholder entitled to vote in the election of Directors generally. Any stockholder entitled to vote in the election of Directors generally may nominate one or more persons for election as Directors at a meeting only if written notice of such stockholder's intent to make such nomination or nominations has been given, either by personal delivery or by United States mail, postage prepaid, to the Secretary of the Corporation not later than 80 days prior to the date of any annual or special meeting. In the event that the date of such annual or special meeting was not publicly announced by the Corporation by mail, press release or otherwise more than 90 days prior to the meeting, notice by the stockholder to be timely must be delivered to the Secretary of the Corporation not later than the close of business on the tenth day following the day on which such announcement of the date of the meeting was communicated to the stockholders.

Each such notice shall set forth: (a) the name and address of the stockholder who intends to make the nomination and of the person or persons to be nominated; (b) a representation that the stockholder is a holder of record of stock of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to nominate the person or persons specified in the notice; (c) a description of all arrangements or understandings between the stockholder and each nominee and any other person or persons (naming such person or persons) pursuant to which the nomination regarding each nominee proposed by such stockholder as would be required to be included in a proxy statement filed pursuant to the proxy rules of the Securities and Exchange Commission had the nominee been nominated, or intended to be nominated, by the Board of Directors, and (e) the consent of each nominee to serve as a Director of the Corporation if so elected.

If the presiding officer of the meeting for the election of Directors determines that a nomination of any candidate for election as a Director at such meeting was not made in accordance with the applicable provisions of these Bylaws, such nomination shall be void.

Section 3. Subject to the rights of the holders of any class or series of stock having a preference over the Common Stock as to dividends or upon liquidation to elect additional Directors under specified circumstances, newly created directorships resulting from any increase in the number of Directors and any vacancy on the Board of Directors resulting from death, resignation, disqualification, removal or other cause shall be filled solely by the affirmative vote of a majority of the remaining Directors then in office, even though less than a quorum of the Board of Directors. or by a sole remaining Director. Any Director elected or chosen as provided herein shall hold office until the sooner of the following events: (i) the expiration of the term of the directorship to which he is appointed, (ii) such time as his successor is elected and qualified or (iii) his resignation or

-4-

removal. No decrease in the number of Directors constituting the Board of Directors shall shorten the term of an incumbent Director.

Section 4. Subject to the rights of the holders of any class or series of stock having preference over the Common Stock as to dividends or upon liquidation to elect additional Directors under specified circumstances, any Director may be removed from office only for cause by the stockholders in the manner provided in this Section 4. At any annual meeting of the stockholders of the Corporation or at any special meeting of the stockholders of the Corporation, the notice of which shall state that the removal of a Director or Directors is among the purposes of the meeting, the affirmative vote of the holders of at least 66 2/3 percent of the combined voting power of the outstanding shares of Voting Stock (as defined below), voting together as a single class, may remove such Director or Directors for cause.

For the purpose of this Section 4, "Voting Stock" shall mean the outstanding shares of capital stock of the Corporation entitled to vote generally in the election of Directors. In any vote required by or provided for in this Section 4, each share of Voting Stock shall have the number of votes granted to it generally in the election of Directors.

Section 5. The business of the Corporation shall be managed by its Board of Directors, which may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the certificate of incorporation or by these Bylaws directed or required to be exercised or done by the stockholders.

Meetings of the Board of Directors

6

Section 6. The Board of Directors of the Corporation may hold meetings, both regular and special, either within or without the State of Delaware.

Section 7. Meetings of the Board of Directors may be held at such time and place as shall be specified in a notice given in the manner hereinafter provided, or as shall be specified in a written waiver signed by all of the Directors.

Section 8. Regular meetings of the Board of Directors may be held without notice at such time and at such place as shall from time to time be determined by the Board of Directors.

Section 9. Special meetings of the Board of Directors may be called by the Chairman of the Board on 24 hours' notice to each Director, either personally or by telecopy or telegram; special meetings shall be called by the president, chief executive officer or secretary in like manner and on like notice on the written request of three Directors.

Section 10. Except as provided in these Bylaws to the contrary, at all meetings of the board a majority of the total number of Directors shall constitute a quorum for the transaction of business and the vote of a majority of the Directors entitled to vote and present at a meeting at which a

-5-

quorum is present shall be the act of the Board of Directors, unless the certificate of incorporation shall require a vote of a greater number. If a quorum shall not be present at any meeting of the Board of Directors, the Directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present.

Section 11. Unless otherwise restricted by the certificate of incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board of Directors, or of any committee thereof, may be taken without a meeting, if all members of the board or committee, as the case may be, consent thereto in writing, and the writing or writings are filed with the minutes of proceedings of the Board of Directors or committee.

Section 12. At all meetings of the Board of Directors, business shall be transacted in such order as from time to time the Board of Directors may determine.

At all meetings of the Board of Directors, the Chairman or Vice-Chairman of the Board of Directors shall preside, and in the absence of either such Director a person shall be chosen by the board from among the Directors present to act as chairman of the meeting.

The secretary of the Corporation shall act as secretary of the meeting of the Board of Directors, but in the absence of the secretary, the presiding officer may appoint any person to act as secretary of the meeting.

Committees of Directors

7

Section 13. The Board of Directors may, by resolution adopted by a majority of the whole board, designate one (1) or more committees, each committee to consist of one (1) or more Directors. The board may designate one (1) or more directors as alternate members of any committee, who may replace any absent or disqualified member of any meeting of the committee. In the absence or disqualification of a member, and the alternate or alternates, if any, designated for such member, of any committee, the member or members thereof present at the meetings and not disqualified from voting, whether or not he or they constitute a quorum, may unanimously appoint another director to act at the meeting in the place of any such absent or disqualified member.

Any such committee, to the extent provided in the resolution establishing such committee, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to the following matters: (i) approving or adopting, or recommending to the stockholders, any action or matter expressly required by the Delaware General Corporation Law to be submitted to stockholders for approval or (ii) adopting, amending or repealing any bylaw of the Corporation. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors.

-6-

Compensation of Directors

8

Section 15. The Directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary or retainer as Director. No such payment shall preclude any Director from serving the Corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for attending committee meetings.

The Chairman of the Board of Directors

Section 16. The Chairman of the Board of Directors of the Corporation shall preside at all meetings of stockholders and the Board of Directors. He shall perform such duties and have such powers as usually appertain to the office or as the Board of Directors may from time to time prescribe.

The Vice Chairman of the Board of Directors

Section 17. The Vice Chairman of the Board of Directors of the Corporation shall perform such duties and have such powers as the Board of Directors or Chief Executive Officer may from time to time prescribe.

ARTICLE IV

NOTICES

Section 1. Whenever notice is required to be given to any Director or stockholder pursuant to a statutory provision or the certificate of incorporation or these Bylaws, it shall not be construed to mean personal notice, but such notice may be given in writing, by mail, addressed to such Director or stockholder, at his address as it appears in the records of the Corporation, with postage thereon prepaid, and such notice shall be deemed to be given at the time when the same shall be deposited in the United States mail. Notice to Directors may also be given personally or by telegram or telecopy.

Section 2. Whenever notice is required to be given pursuant to a statutory provision or the certificate of incorporation or Bylaws, a waiver thereof in writing, signed by the person or persons entitled to said notice, whether before or after the time stated therein, shall be deemed equivalent thereto.

-7-

ARTICLE V

OFFICERS

Section 1. The officers of the Corporation shall be chosen by the Board of Directors and shall be a chief executive officer, a president, a vice president, a secretary and a treasurer. The Board of Directors may also appoint chief operating officers, additional vice presidents and one or more assistant secretaries and assistant treasurers. Any number of offices may be held by the same person, unless the certificate of incorporation or these Bylaws otherwise provide.

Section 2. The Board of Directors at its first meeting after each annual meeting of stockholders shall choose a chief executive officer, a president, one or more chief operating officers, one or more vice presidents, a secretary and a treasurer.

Section 3. The Board of Directors may appoint such other officers and agents as it shall deem necessary who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the board.

Section 4. The salaries of all officers and agents of the Corporation shall be fixed by the Board of Directors, a committee thereof or any such party to which either of them may delegate such authority.

Section 5. The officers of the Corporation shall hold office until their successors are chosen and qualify. Any officer elected or appointed by the Board of Directors may be removed at any time by the affirmative vote of a majority of the Board of Directors. Any vacancy occurring in any office of the Corporation shall be filled by the Board of Directors.

The Chief Executive Officer

Section 6. The Chief Executive Officer of the Corporation shall have general and active management of the business of the Corporation and shall see that all orders and resolutions of the Board of Directors are carried into effect. He shall have the authority to execute all documents and instruments necessary to carry out the management of the business of the Corporation. He shall execute bonds, mortgages and other contracts requiring a seal, under the seal of the Corporation, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of this Corporation. He shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe. He shall report to the Board of Directors.

The President

Section 7. The President of the Corporation shall perform such duties and have such powers as usually appertain to the office or as the Chief Executive Officer or the Board of Directors may from time to time prescribe. He shall have the authority to execute all documents and instruments necessary to carry out the management of the business of the Corporation. He shall report to the Chief Executive Officer.

The Chief Operating Officers

Section 8. The chief operating officers of the Corporation shall be responsible for the day-to-day operations of the Corporation and shall have the authority to execute all documents and instruments necessary to carry out such operations. They shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe. They shall report to the Board of Directors.

The Vice Presidents

Section 9. In the absence of the president or in the event of his inability or refusal to act, the vice president (or in the event there is more than one, the vice presidents in the order determined by the Board of Directors, or, if there be no such determination, then in the order of their election), shall perform the duties of the president, and when so acting, shall have all the powers of and be subject to all the restrictions imposed upon the president. The vice presidents shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

The Secretary and the Assistant Secretary

Section 10. The secretary shall attend all meetings of the Board of Directors and all meetings of the stockholders and record all the proceedings of the meetings to be kept for that purpose and shall perform like duties for the standing committees when required. He shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors, and shall perform such other duties as may be prescribed by the Board of Directors or president, under whose supervision he shall be. He shall have custody of the corporate seal of the Corporation, if any such seal be adopted by resolution of the Board of Directors, and he, or an assistant secretary, shall have authority to affix the same to any instrument requiring it and when so affixed, it may be attested by his signature or by the signature of such assistant secretary. The Board of Directors may give general authority to any other officer to affix the seal of the Corporation and to attest the affirming thereof by his signature.

Section 11. The assistant secretary (or if there be more than one, the assistant secretaries in the order determined by the Board of Directors, or, if there be no such determination, then in the order of their election) shall, in the absence of the secretary or in the event of his inability or refusal to act, perform the duties and exercise the powers of the secretary and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

-9-

The Treasurer and Assistant Treasurer

Section 12. The treasurer shall have the custody of the corporate funds and securities and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. He shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the president and the Board of Directors, at its regular meetings, or when the Board of Directors so requires, an account of all his transactions as treasurer and of the financial condition of the Corporation.

Section 13. The assistant treasurer (or, if there shall be more than one, the assistant treasurers in the order determined by the Board of Directors, or, if there be no such determination, then in the order of their election) shall, in the absence of the treasurer or in the event of his inability or refusal to act, perform the duties and exercise the powers of the treasurer and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

ARTICLE VI

CERTIFICATES OF STOCK

Section 1. Every holder of stock in the Corporation shall be entitled to a certificate, signed by, or in the name of the Corporation by, the Chairman of the Board, the chief executive officer, the president or a vice president and the secretary or an assistant secretary of the Corporation, certifying the number of shares owned by him in the Corporation. Any signature on the certificate may be a facsimile. If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class of stock, the designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or back of the certificate which the Corporation shall issue to represent such class or series of stock, provided that, except as otherwise provided in Section 202 of the General Corporation Law of Delaware, in lieu of the foregoing requirements, there may be set forth on the face or back of the certificate which the Corporation shall issue to represent such class or series of stock, a statement that the Corporation will furnish without charge to each stockholder who so requests the designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

Section 2. Where a certificate is countersigned (1) by a transfer agent other than the Corporation or its employee or, (2) by a registrar other than the Corporation or its employee, any signature on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the

-10-

Lost Certificates

Section 3. The Board of Directors may direct a new certificate or certificates to be issued in place of any certificate or certificates theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. When authorizing such issue of a new certificate or certificates, the Board of Directors may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificate or certificates, or his legal representative, to advertise the same in such manner as it shall require and/or to give the Corporation a bond in such sum as it may direct as indemnity against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed.

Transfers of Stock

Section 4. Upon surrender to the Corporation or the transfer agent of the Corporation of a certificate for shares duly endorsed or accompanied by a proper evidence of succession, assignment or authority to transfer, it shall be the duty of the Corporation to issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books.

Fixing Record Date

Section 5. In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or to express consent to corporate action in writing without a meeting prior to March 1, 1998, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting, nor more than sixty (60) days prior to any other action. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

Registered Stock Holders

Section 6. The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote as such owner, and to hold liable for calls and assessments a person registered on its books as the owner of shares, and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of Delaware.

-11-

GENERAL PROVISIONS

Dividends

Section 1. Dividends upon the capital stock of the Corporation, subject to the provisions of the certificate of incorporation, if any, may be declared by the Board of Directors at any regular or special meetings, pursuant to law. Dividends may be paid in cash, in property, or in shares of the capital stock, subject to the provisions of the certificate of incorporation.

Section 2. Before payment of any dividend, there may be set aside out of any funds of the Corporation available for dividends such sum or sums as the Directors from time to time, in their absolute discretion, think proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the Corporation, or for such other purpose as the Directors shall think conducive to the interest of the Corporation, and the Directors may modify or abolish any such reserve in the manner in which it was created.

Checks

Section 3. All checks or demands for money and notes of the Corporation shall be signed by such officer or officers or such other person or persons as the Board of Directors may from time to time designate.

Fiscal Year

Section 4. The fiscal year of the Corporation shall begin on the first day of October of each year and end on the last day of September of each year, unless otherwise determined by the Board of Directors.

Seal

Section 5. The corporate seal, if any such seal be adopted by resolution of the Board of Directors, will be in such form as the Board of Directors may prescribe. The seal may be used by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise placed thereon.

-12-

Section 6.

(a) No contract or transaction between the Corporation and one or more of its Directors or officers, or between the Corporation and any other corporation, partnership, association, or other organization in which one or more of its Directors or officers are Directors or officers, or have a financial interest, shall be void or voidable solely for this reason, or solely because the Director or officer is present at or participates in the meeting of the board or committee thereof which authorizes the contract or transaction, or solely because his or their votes are counted for such purposes, if;

> (1) the material facts as to his relationship or interest and as to the contract or transaction are disclosed or are known to the Board of Directors or the committee, and the board or committee in good faith authorizes the contract or transaction by the affirmative votes of a majority of the disinterested Directors, even though the disinterested Directors be less than a quorum; or

> (2) the material facts as to his relationship or interest and as to the contract or transaction are disclosed or are known to the stockholders entitled to vote thereon, and the contract for transaction is specifically approved in good faith by vote of the stockholders; or

(3) the contract or transaction is fair as to the Corporation as of the time it is authorized, approved or ratified, by the Board of Directors, a committee thereof, or the stockholder.

(b) Common or interested Directors may be counted in determining the presence of a quorum at a meeting of the Board of Directors or of a committee which authorizes the contract or transaction.

ARTICLE VIII

AMENDMENTS

These Bylaws may be altered, amended or repealed, or new Bylaws may be adopted by the affirmative vote of a majority of the entire Board of Directors at any meeting and without the consent or vote of the stockholders. These Bylaws may be altered, amended or repealed, or new Bylaws may be adopted by the stockholders at any regular meeting of the stockholders or at any special meeting of the stockholders, if notice of such alteration, amendment, repeal or adoption of new Bylaws is contained in the notice of such meeting, by the holders of at least 66% of the total voting power of all shares of stock of the Corporation entitled to vote in the election of directors, considered for purposes of this Article VIII as one class.

-13-

ARTICLE IX

INDEMNIFICATION AND INSURANCE

Section 1. The Corporation shall, to the full extent permitted by Section 145 of Title 8 of the General Corporation Law of the State of Delaware, as amended from time to time, indemnify all officers and directors of the Corporation whom it may indemnify pursuant thereto. The provisions of this Article IX shall apply to acts or omissions occurring before or after the adoption hereof. The right of indemnification herein provided for shall not be exclusive of any other right to which any Director or officer may now or hereafter be entitled under any statute, bylaw, agreement, vote of stockholders or disinterested Directors or otherwise, shall continue as to a person who has ceased to be such Director or officer entitled to indemnification pursuant to this Article IX and shall inure to the benefit of the heirs, executors and administrators of such Director or officer.

Section 2. The Corporation may purchase and maintain insurance on behalf of any person who is or was a Director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a Director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not the Corporation would have the power to indemnify him against such liability under the provisions of this Article IX or of Section 145 of the General Corporation Law of the State of Delaware.

Section 3. The indemnification provided by this Article IX shall be subject to all valid and applicable laws, and, in the event this Article IX or any of the provisions hereof or the indemnification contemplated hereby are found to be inconsistent with or contrary to any such valid laws, the latter shall be deemed to control, and this Article IX shall be regarded as modified accordingly and, as so modified, shall continue in full force and effect.

-14-

EXHIBIT 4.3

EXECUTION COPY

INTEGRATED ELECTRICAL SERVICES, INC.

AS THE COMPANY

AND

THE SUBSIDIARIES NAMED HEREIN

AS GUARANTORS

т0

STATE STREET BANK AND TRUST COMPANY

AS TRUSTEE

INDENTURE

DATED AS OF MAY 29, 2001

9 3/8% SENIOR SUBORDINATED NOTES, SERIES C, DUE 2009

9 3/8% SENIOR SUBORDINATED NOTES, SERIES D, DUE 2009

CROSS-REFERENCE TABLE

(NOTE - this table is not part of the Indenture)

TIA Section	Indenture Section
Section 310(a)(1) 310(a)(2) 310(a)(3) 310(a)(4) 310(a)(5) 310(b) 310(c) 311(b) 311(c) 312(a) 312(b) 312(c) 313(a) 313(d) 313(d) 314(b) 314(c)(1) 314(c)(2) 314(c)(2) 314(c)(3) 314(e) 314(f) 314(f) 314(f) 314(f) 315(b) 315(d)	Section 6.9
315(d) 315(e) 316(a)(1)(A) 316(a)(1)(B) 316(a)(2)	5.14 5.12 5.13
316(a)(last sentence) 316(b) 316(c) 317(a)(1) 317(a)(2)	5.7; 5.8 1.4 5.3 5.4
317(b) 317(a)	

ARTICLE I Definitions and Other Provisions of General Application1
SECTION 1.1. Definitions1SECTION 1.2. Compliance Certificates and Opinions21SECTION 1.3. Form of Documents Delivered to Trustee21SECTION 1.4. Acts of Holders; Record Dates22SECTION 1.5. Notices to Trustee, the Company or a Guarantor23SECTION 1.6. Notice to Holders; Waiver23SECTION 1.7. Conflict with Trust Indenture Act24SECTION 1.8. Effect of Headings and Table of Contents24SECTION 1.10. Separability Clause24SECTION 1.11. Benefits of Indenture24SECTION 1.12. Governing Law24SECTION 1.13. Legal Holidays24
ARTICLE II Security Forms
SECTION 2.1. Forms Generally25
ARTICLE III The Securities
SECTION 3.1. Title and Terms.25SECTION 3.2. Denominations.26SECTION 3.3. Execution, Authentication, Delivery and Dating.26SECTION 3.4. Temporary Securities.27SECTION 3.5. Registration, Registration of Transfer and Exchange.28SECTION 3.6. Mutilated, Destroyed, Lost and Stolen Securities.29SECTION 3.7. Payment of Interest; Rights Preserved.29SECTION 3.8. Persons Deemed Owners.30SECTION 3.9. Cancellation.30SECTION 3.10. Computation of Interest.30SECTION 3.11. CUSIP and CINS Numbers.31SECTION 3.12. Deposits of Monies.31SECTION 3.13. Book-Entry Provisions for Global Securities.31SECTION 3.14. Transfer and Exchange.32
ARTICLE IV Satisfaction and Discharge
SECTION 4.1. Satisfaction and Discharge of Indenture
ARTICLE V Remedies
SECTION 5.1. Events of Default

SECTION 5.2. Acceleration of Maturity; Rescission and Annulment
SECTION 5.3. Collection of Indebtedness and Suits for Enforcement by Trustee
SECTION 5.4. Trustee May File Proofs of Claim
SECTION 5.5. Trustee May Enforce Claims Without Possession of Securities
SECTION 5.6. Application of Money Collected
SECTION 5.7. Limitation on Suits
SECTION 5.8. Unconditional Right of Holders to Receive Principal, Premium and Interest
SECTION 5.9. Restoration of Rights and Remedies
SECTION 5.10. Rights and Remedies Cumulative
SECTION 5.11. Delay or Omission Not Waiver40
SECTION 5.12. Control by Holders40
SECTION 5.13. Waiver of Past Defaults41
SECTION 5.14. Undertaking for Costs41
SECTION 5.15. Waiver of Stay or Extension Laws
ARTICLE VI The Trustee
SECTION 6.1. Certain Duties and Responsibilities42
SECTION 6.2. Notice of Defaults
SECTION 6.3. Certain Rights of Trustee
SECTION 6.4. Not Responsible for Recitals or Issuance of Securities
SECTION 6.5. May Hold Securities
SECTION 6.6. Money Held in Trust
SECTION 6.7. Compensation and Reimbursement44
SECTION 6.8. Conflicting Interests
SECTION 6.9. Corporate Trustee Required; Eligibility45
SECTION 6.10. Resignation and Removal; Appointment of Successor
SECTION 6.11. Acceptance of Appointment by Successor
SECTION 6.12. Merger, Conversion, Consolidation or Succession to Business
SECTION 6.13. Preferential Collection of Claims Against the Company or a Guarantor
SECTION 6.14. Appointment of Authenticating Agent
Section 0.14. Appointment of Authenticating Agent
ARTICLE VII Holders' Lists and Reports by Trustee and Company
ARTICLE VII HOTUETS LISTS and Reports by Trustee and Company
CENTER 7.4. Company to Evenich Tructes Names and Addresses of Valdaus
SECTION 7.1. Company to Furnish Trustee Names and Addresses of Holders
SECTION 7.2. Preservation of Information; Communications to Holders
SECTION 7.3. Reports by Trustee
SECTION 7.4. Reports by Company
ARTICLE VIII Consolidation, Merger, Conveyance, Transfer or Lease
SECTION 8.1. Company May Merge, Consolidate, Etc. Only on Certain Terms
SECTION 8.2. Successor Substituted
ARTICLE IX Amendments; Waivers; Supplemental Indentures
, , , , , , , , , , , , , , , , , , ,
SECTION 9.1. Amendments, Waivers and Supplemental Indentures Without Consent of Holders
SECTION 9.2. Modifications, Amendments and Supplemental Indentures with Consent of Holders
elected of the house of the north and ouppremental indentation with consent of houders in the north the no

-ii-

4

SECTION 9.3. Execution of Supplemental Indentures.52SECTION 9.4. Effect of Supplemental Indentures.52SECTION 9.5. Conformity with Trust Indenture Act.52SECTION 9.6. Reference in Securities to Supplemental Indentures.53SECTION 9.7. Waiver of Certain Covenants.53SECTION 9.8. No Liability for Certain Persons.53 SECTION 10.17. Additional Subsidiary Guarantees......62

SECTION 12.5. Deposited Money and U.S. Government Obligations To Be Held in Trust; Miscellaneous

-iii-

SECTION 12.6. Reinstatement	
ARTICLE XIII Guaranty	
SECTION 13.1. Guaranty	
ARTICLE XIV Subordination	
SECTION 14.1. Securities Subordinate to Senior Indebtedness and Senior to Subordinated Indebtedness	
SECTION 14.12. Article Applicable to Paying Agents	

Schedule A	List of Guarantors
Exhibit A-1	Form of Series C Security
Exhibit A-2	Form of Series D Security
Exhibit B	Global Securities Legend
Exhibit C	Transfer Letter
Exhibit D	Form of Supplemental Indenture
Exhibit E	Form of Certificate from Acquiring
	Institutional Accredited Investor

-iv-

INDENTURE, dated as of May 29, 2001, among INTEGRATED ELECTRICAL SERVICES, INC., a corporation duly organized and existing under the laws of the State of Delaware (herein called the "Company"), the Subsidiaries of the Company named in Schedule A as of the date of issuance (herein called the "Initial Guarantors") and STATE STREET BANK AND TRUST COMPANY, as trustee (herein called the "Trustee").

RECITALS OF THE COMPANY

The Company has duly authorized the execution and delivery of this Indenture to provide for the issuance of (i) \$125,000,000 aggregate principal amount of the Company's 9[]% Senior Subordinated Notes, Series C, due 2009, issued on the date hereof (the "Initial Securities"), (ii) if and when issued, an unlimited principal amount of additional 9[]% Senior Subordinated Notes, Series C, due 2009 or 9[]% Senior Subordinated Notes, Series D, due 2009 ("Series D Securities") of the Company that may be offered from time to time subsequent to the Issue Date (the "Additional Securities") and (iii) if and when issued, the Company's 9[]% Senior Subordinated Notes, Series D, due 2009 that may be issued from time to time in exchange for Initial Securities or any Additional Securities in an offer registered under the Securities Act as provided in a Registration Rights Agreement or as Private Exchange Securities," and together with the Initial Securities and Additional Securities, the "Securities").

Each Guarantor desires to make the Guaranty provided herein and has duly authorized the execution and delivery of this Indenture.

All things necessary to make the Securities, when executed by the Company, authenticated and delivered hereunder and duly issued by the Company, and the Guarantees, when executed and delivered hereunder by each Guarantor, the valid obligations of the Company and each Guarantor, and to make this Indenture a valid agreement of the Company and each Guarantor, in accordance with their and its terms, have been done.

NOW, THEREFORE, THIS INDENTURE WITNESSETH:

For and in consideration of the premises and the purchase of the Securities by the Holders (as defined herein) thereof, it is mutually covenanted and agreed, for the equal and proportionate benefit of all Holders of the Securities, as follows:

ARTICLE I

Definitions and Other Provisions of General Application

SECTION 1.1. Definitions.

For all purposes of this Indenture, except as otherwise expressly provided or unless the context otherwise requires:

(1) the terms defined in this Article have the meanings assigned to them in this Article and include the plural as well as the singular;

(2) all other terms used herein which are defined in the Trust Indenture Act, either directly or by reference therein, have the meanings assigned to them therein;

(3) all accounting terms not otherwise defined herein have the meanings assigned to them in accordance with GAAP (whether or not such is indicated herein); (4) unless the context otherwise requires, any reference to an "Article" or a "Section" refers to an Article or Section, as the case may be, of this Indenture;

(5) the words "herein," "hereof" and "hereunder" and other words of similar import refer to this Indenture as a whole and not to any particular Article, Section or other subdivision; and

(6) each reference herein to a rule or form of the Commission shall mean such rule or form and any rule or form successor thereto, in each case as amended from time to time.

Whenever this Indenture requires that a particular ratio or amount be calculated with respect to a specified period after giving effect to certain transactions or events on a pro forma basis, such calculation shall be made as if the transactions or events occurred on the first day of such period, unless otherwise specified.

"Acquired Indebtedness" means Indebtedness of a Person (a) assumed in connection with an Asset Acquisition from such Person or (b) existing at the time such Person becomes or is merged into a Subsidiary of any other Person.

"Act," when used with respect to any Holder, has the meaning specified in Section 1.4.

"Additional Securities" has the meaning set forth in the first paragraph of the RECITALS OF THE COMPANY of this Indenture.

"Affiliate" means, with respect to any specified Person, any other Person directly or indirectly controlling or controlled by or under direct or indirect common control with such specified Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling," "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through the ownership of Voting Stock, by agreement or otherwise; provided that beneficial ownership of 10% of more of the Voting Stock of a Person shall be deemed to be control.

"Asset Acquisition" means (a) an Investment by the Company or any Restricted Subsidiary of the Company in any other Person pursuant to which such Person shall become a Restricted Subsidiary of the Company, or shall be merged with or into the Company or any Restricted Subsidiary of the Company, or (b) the acquisition by the Company or any Restricted Subsidiary of the Company of the assets of any Person which constitute all or substantially all of the assets of such Person, any division or line of business of such Person or, other than in the ordinary course of business, any other properties or assets of such Person.

"Asset Sale" means any sale, issuance, conveyance, transfer, lease or other disposition by the Company or any Restricted Subsidiary to any Person other than the Company or a Restricted Subsidiary, of (a) any Capital Stock of any Restricted Subsidiary; (b) all or substantially all of the properties and assets of any division or line of business of the Company or any Restricted Subsidiary; or (c) any other properties or assets of the Company or any Restricted Subsidiary outside of the ordinary course of business, other than (i) sales of obsolete, damaged or used equipment or other equipment or inventory sales in the ordinary course of business, (ii) sales of assets in one or a series of related transactions for an aggregate consideration of less than \$2,000,000 and (iii) sales of accounts receivable for financing purposes. For the purposes of this definition, the term "Asset Sale" shall not include (i) any sale, issuance, conveyance, transfer, lease or other disposition of properties or assets that is governed by the provisions of Article VIII, (ii) a Restricted Payment that is permitted under Section 10.9, or (iii) the trade or exchange by the Company or any Restricted Subsidiary of any property or assets owned or held by the Company or such Restricted Subsidiary for any property or assets owned or held by another Person, provided that the Fair Market Value of the properties traded or exchanged by the Company or such Restricted Subsidiary (including any cash or Cash Equivalents to be delivered by the Company or such Restricted Subsidiary) is reasonably equivalent to the Fair Market Value of the properties (together with any cash or Cash Equivalents) to be received by the Company or such Restricted Subsidiary, and provided further that any such cash or Cash Equivalents shall be deemed to constitute Net Cash Proceeds of an Asset Sale for purposes of Section 10.14.

"Asset Sale Deficiency" has the meaning specified in Section 10.14.

"Asset Sale Offer" has the meaning specified in Section 10.14.

"Asset Sale Offer Amount" has the meaning specified in Section 10.14.

"Asset Sale Purchase Date" means the Purchase Date relating to an Asset Sale Offer.

"Authenticating Agent" means any Person authorized by the Trustee pursuant to Section 6.14 hereof to act on behalf of the Trustee to authenticate Securities.

"Average Life to Stated Maturity" means, with respect to any Indebtedness, as at any date of determination, the quotient obtained by dividing (i) the sum of the products of (a) the number of years (and any portion thereof) from such date of such determination to the date or dates of each successive scheduled principal payment (including, without limitation, any sinking fund or mandatory redemption payment requirements) of such Indebtedness, and (b) the amount of each such principal payment by (ii) the sum of all such principal payments.

"Board of Directors" means the board of directors of a company or its equivalent, including managers of a limited liability company, general partners of a partnership or trustees of a business trust, or any duly authorized committee thereof.

"Board Resolution" means a copy of a resolution certified by the Secretary or an Assistant Secretary of a company to have been duly adopted by the Board of Directors of such company and to be in full force and effect on the date of such certification, and delivered to the Trustee.

"Business Day" means each Monday, Tuesday, Wednesday, Thursday and Friday which is not a day on which banking institutions in the Borough of Manhattan, The City of New York, Hartford, Connecticut or Boston, Massachusetts are authorized or obligated by law or executive order to close.

"Capital Stock" means, with respect to any Person, any and all shares, interests, participations, rights in or other equivalents (however designated) of such Person's capital stock or equity participations, and any rights (other than debt securities convertible into capital stock), warrants or options exchangeable for or convertible into such capital stock and, including, without limitation, with respect to partnerships, limited liability companies or business trusts, ownership interests (whether general or limited) and any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, such partnerships, limited liability companies or business trusts.

"Capitalized Lease Obligation" means any obligation to pay rent or other amounts under a lease of (or other agreement conveying the right to use) any property (whether real, personal or mixed) that is required to be classified and accounted for as a capital lease obligation under GAAP, and, for the purpose of this Indenture, the amount of such obligation at any date shall be the capitalized amount thereof at such date, determined in accordance with GAAP. "Cash Equivalents" means, at any time, (a) any evidence of Indebtedness, maturing not more than one year after such time, issued or guaranteed by the United States Government or any agency thereof (provided that the full faith and credit of the United States of America is pledged in support thereof), (b) commercial paper, maturing not more than one year from the date of issue, rated at least A-2 by Standard & Poor's Ratings Group or P-2 by Moody's Investors Service, Inc., (c) any certificate of deposit (or time deposits represented by such certificates of deposit) or bankers acceptance, maturing not more than one year after such time, or overnight Federal Funds transactions that are issued or sold by a banking institution that is a member of the Federal Reserve System and has a combined capital and surplus and undivided profits of not less than \$500,000,000, (d) repurchase obligations with a term of not more than seven days for underlying securities of the types described in clause (a) above entered into with any bank meeting the specifications of clause (c) above, and (e) investments in funds investing primarily in investments of the types described in clauses (a) through (d) above.

"Change of Control" means the occurrence of any of the following events: (a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) is or becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a Person shall be (as defined deemed to have "beneficial ownership" of all securities that such Person has the right to acquire, whether such right is exercisable immediately or only after the passage of time), directly or indirectly, of more than 50% of the total Voting Stock of the Company; (b) the Company consolidates with, or merges with or into, another Person or sells, assigns, conveys, transfers, leases or otherwise disposes of all or substantially all of its assets to any Person, or any Person consolidates with, or merges with or into, the Company, in any such event pursuant to a transaction in which the outstanding Voting Stock of the Company is converted into or exchanged for cash, securities or other property, other than any such transaction where (i) the outstanding Voting Stock of the Company is converted into or exchanged for Voting Stock (other than Redeemable Capital Stock) of the surviving or transfere corporation and (ii) immediately after such transaction no "Person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) is the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a Person shall be deemed to have "beneficial ownership" of all securities that such Person has the right to acquire, whether such right is exercisable immediately or only after the passage of time), directly or indirectly, of more than 50% of the total Voting Stock of the surviving or transferee corporation; (c) during any consecutive two-year period, individuals who at the beginning of such period constituted the Board of Directors of the Company (together with any new directors whose election by such Board of Directors or whose nomination for election by the stockholders of the Company was approved by a vote of 66 2/3% of the directors then still in office who were either directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason to constitute a majority of the Board of Directors of the Company then in office; or (d) the Company is liquidated or dissolved or adopts a plan of liquidation.

"Change of Control Date" has the meaning specified in Section 10.13.

"Change of Control Offer" has the meaning specified in Section 10.13.

"Change of Control Purchase Date" has the meaning specified in Section 10.13.

"Change of Control Purchase Price" has the meaning specified in Section 10.13.

"Clearstream" means Clearstream Banking, societe anonyme.

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations thereunder.

"Commission" means the Securities and Exchange Commission, as from time to time constituted, created under the Exchange Act, or, if at any time after the execution of this instrument such Commission is not existing and performing the duties now assigned to it under the Trust Indenture Act, then the body performing such duties at such time.

"Common Stock" means the common stock of the Company, par value 0.01 per share, and the Company's restricted voting common stock, par value 0.01 per share.

"Company" means the Person named as the "Company" in the first paragraph of this instrument until a successor Person shall have become such pursuant to the applicable provisions of this Indenture and thereafter "Company" shall mean such successor Person.

"Company Request" or "Company Order" means a written request or order signed in the name of the Company by its Chairman of the Board, its Chief Executive Officer, its Chief Financial Officer, its President or a Vice President, and by its Treasurer, an Assistant Treasurer, its Secretary or an Assistant Secretary, and delivered to the Trustee or Paying Agent, as applicable.

"Consolidated Cash Flow Available for Fixed Charges" as of any date of determination means, with respect to any Person for any period, the sum of, without duplication, the amounts for such period, taken as a single accounting period, of (a) Consolidated Net Income, (b) Consolidated Non-cash Charges, (c) Consolidated Interest Expense, and (d) Consolidated Income Tax Expense (other than income tax expense (either positive or negative) attributable to extraordinary gains or losses).

"Consolidated Fixed Charge Coverage Ratio" as of any date of determination means, with respect to any Person, the ratio of the aggregate amount of Consolidated Cash Flow Available for Fixed Charges of such Person for the four full fiscal quarters, treated as one period, for which financial information in respect thereof is available immediately preceding the date of the transaction (the "Transaction Date") giving rise to the need to calculate the Consolidated Fixed Charge Coverage Ratio (such four full fiscal quarter period being referred to herein as the "Four Quarter Period") to the aggregate amount of Consolidated Fixed Charges of such Person for the Four Quarter Period. For purposes of making the computation referred to above, Consolidated Cash Flow Available for Fixed Charges and Consolidated Fixed Charges shall be calculated giving pro forma effect (in a manner consistent with Rule 11-02 of Regulation S-X) to the following events (without duplication): (i) any Asset Sale or Asset Acquisition occurring since the first day of the Four Quarter Period (including to the date of calculation) as if such acquisition or disposition occurred at the beginning of the Four Quarter Period (including giving effect to (A) the amount of any reduction in expenses related to any compensation, remuneration or other benefit paid or provided to any employee, consultant, Affiliate or equity owner of the entity involved in any such Asset Sale or Asset Acquisition to the extent such costs are eliminated or reduced (or public announcement has been made of the intent to eliminate or reduce such costs) prior to the date of such calculation and not replaced and (B) the amount of any reduction in general, administrative or overhead costs of the entity involved in any such Asset Sale or Asset Acquisition), (ii) the incurrence of Indebtedness giving rise to the need to calculate the Consolidated Fixed Charge Coverage Ratio and (if applicable) the application of the net proceeds therefrom, including to refinance other Indebtedness, as if such Indebtedness were incurred at the beginning of the Four Quarter Period, (iii) the incurrence, repayment or subsidiaries since the first day of the Four Quarter Period and prior to the making of this calculation as if such Indebtedness or obligations were incurred, repaid or retired at the beginning of the Four Quarter Period (except that, in making such computation, the amount of Indebtedness under any revolving credit facility shall be computed based upon the average daily balance of such Indebtedness during the Four Quarter Period), (iv) elimination of Consolidated Cash Flow Available for Fixed Charges and Consolidated Fixed Charges attributable to discontinued operations, as determined in accordance with GAAP, but, with respect to Consolidated Fixed Charges, only to the extent that the obligations giving rise to such Consolidated Fixed Charges will not be obligations of the referent Person or any of its Restricted Subsidiaries following the Transaction Date. In calculating Consolidated Fixed Charges for purposes of determining the denominator (but not the numerator) of the Consolidated Fixed Charge Coverage Ratio, (i) interest on

"Consolidated Fixed Charges" means, with respect to any Person for any period, the sum of, without duplication, the amounts for such period of (i) Consolidated Interest Expense and (ii) the aggregate amount of dividends and other distributions paid or accrued during such period in respect of Redeemable Capital Stock or Preferred Stock of such Person and its Restricted Subsidiaries on a consolidated basis.

"Consolidated Income Tax Expense" means, with respect to any Person for any period, the provision for federal, state, local and foreign income taxes of such Person and its Restricted Subsidiaries for such period as determined on a consolidated basis in accordance with GAAP.

"Consolidated Interest Expense" means, with respect to any Person for any period, without duplication, the sum of (i) the interest expense of such Person and its Restricted Subsidiaries for such period as determined on a consolidated basis in accordance with GAAP, including, without limitation, (a) any amortization of debt discount and capitalized debt issuance costs, (b) the net cost under Interest Rate Protection Obligations (including any amortization of discounts), (c) the interest portion of any deferred payment obligation, (d) all commissions, discounts and other fees and charges owed with respect to letters of credit, bankers' acceptance financing or similar facilities and (e) all accrued interest and (ii) the interest component of Capitalized Lease Obligations paid, accrued and/or scheduled to be paid or accrued by such Person and its Restricted Subsidiaries during such period as determined on a consolidated basis in accordance with GAAP.

"Consolidated Net Income" means, with respect to any Person, for any period, the consolidated net income (or loss) of such Person and its Restricted Subsidiaries for such period as determined in accordance with GAAP, adjusted, to the extent included in calculating such net income, by excluding, without duplication, (i) all extraordinary gains or losses (net of fees and expenses relating to the transaction giving rise thereto), (ii) the portion of net income of such Person and its Restricted Subsidiaries allocable to minority interests in unconsolidated Persons or to Investments in Unrestricted Subsidiaries to the extent that cash dividends or distributions have not actually been received by such Person or one of its Restricted Subsidiaries, (iii) net income (or loss) or any Person combined with such Person or one of its Restricted Subsidiaries on a of "pooling of interests" basis attributable to any period prior to the date of combination, (iv) gains or losses in respect of any Asset Sales by such Person or one of its Restricted Subsidiaries (net of fees and expenses relating to the transaction giving rise thereto), on an after-tax basis, (v) the net income of any Restricted Subsidiary of such Person to the extent that the declaration of dividends or similar distributions by that Restricted Subsidiary of that income is not at the time permitted, directly or indirectly, by operation of the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or governmental regulations applicable to that Restricted Subsidiary or its stockholders and (vi) any gain or loss realized as a result of the cumulative effect of a change in accounting principles.

"Consolidated Non-cash Charges" means, with respect to any Person for any period, the aggregate depreciation, amortization (including amortization of goodwill and other intangibles) and other non-cash charges of such Person and its Restricted Subsidiaries reducing Consolidated Net Income of "Corporate Trust Office" means the office of the Trustee at which at any particular time its corporate trust business shall be administered, which address as of the date of this Indenture is located at Goodwin Square, 23rd Floor, 225 Asylum Street, Hartford, CT 06103, Attention: Corporate Trust, Administration.

"Covenant Defeasance" has the meaning specified in Section 12.3.

"Credit Facility" means the Credit Agreement dated as of May 22, 2001 among the Company, The Chase Manhattan Bank, as Administrative Agent, and the Banks named therein (which replaced the Credit Agreement dated as of July 30, 1998 among the Company, NationsBank, N.A., as the Agent, and the banks named therein), including any notes, guarantees, collateral documents, instruments and agreements executed in connection therewith, and in each case as amended (including any amendment and restatement thereof), modified, extended, renewed, refunded, substituted or replaced or refinanced from time to time, including any agreement extending the maturity of, refinancing, replacing or otherwise restructuring (including increasing the amount of available borrowings thereunder or adding Subsidiaries of the Company as additional borrowers or guarantors thereunder) all or any portion of the Indebtedness under such agreement or any successor or replacement agreement and whether by the same or any other agents, creditor, lender or group of creditors or lenders.

"Default" means any event that is, or after notice or passage of time, or both, would be, an Event of Default.

"Defeasance" has the meaning specified in Section 12.2.

"Depository" means The Depository Trust Company, or its successor.

"Designated Guarantor Senior Indebtedness" means, with respect to a Guarantor, amounts owing by such Guarantor under the Credit Facility and guarantees by such Guarantor of Designated Senior Indebtedness.

"Designated Senior Indebtedness" means (i) all Indebtedness under the Credit Facility and (ii) any other issue of Senior Indebtedness which (a) at the time of the determination is equal to or greater than \$25,000,000 in aggregate principal amount and (b) is specifically designated by the Company in the instrument evidencing such Senior Indebtedness as "Designated Senior Indebtedness."

"Disinterested Member of the Board of Directors of the Company" means, with respect to any transaction or series of related transactions, a member of the Board of Directors of the Company other than a member who has any material direct or indirect financial interest in or with respect to such transaction or series of transactions or who is an Affiliate, or an officer, director or an employee of any Person (other than the Company) who has any direct or indirect financial interest in or with respect to such transaction or series of transactions (in each case other than an interest arising solely from the beneficial ownership of Capital Stock of the Company).

"Euroclear" means Euroclear Bank S.A./N.V., as operator of the Euroclear System.

"Event of Default" has the meaning specified in Section 5.1.

"Excess Proceeds" has the meaning specified in Section 10.14.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Expiration Date" shall have the meaning set forth in the definition of "Offer to Purchase."

"Fair Market Value" means, with respect to any asset, the price (after taking into account any liabilities relating to such assets) which could be negotiated in an arm's-length free market transaction between a willing seller and a willing buyer, neither of which is under pressure or compulsion to complete the transaction. Fair Market Value shall be determined by the Board of Directors of the Company in good faith.

"Federal Bankruptcy Code" means Title 11, U.S. Code.

"GAAP" means generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other entity as may be approved by a significant segment of the accounting profession of the United States of America, which are in effect from time to time.

"Global Securities" means one or more Regulation S Global Securities, 144A Global Securities, IAI Global Securities and Series D Global Securities.

"guarantee" means, as applied to any obligation, (i) a guarantee (other than by endorsement of negotiable instruments for collection in the ordinary course of business), direct or indirect, in any manner, of any part or all of such obligation and (ii) an agreement, direct or indirect, contingent or otherwise, the practical effect of which is to assure in any way the payment or performance (or payment of damages in the event of nonperformance) of all or any part of such obligation, including, without limiting the foregoing, the payment of amounts available to be drawn down under letters of credit of another Person. The term "guarantee" used as a verb has a corresponding meaning. The term "guarantor" shall mean any Person providing a guarantee of any obligation.

"Guarantor Senior Indebtedness" of a Guarantor means the principal of, premium, if any, and interest on any Indebtedness of such Guarantor, whether outstanding on the Issue Date or thereafter created, incurred or assumed, unless, in the case of any particular Indebtedness, the instrument creating or evidencing the same or pursuant to which the same is outstanding expressly provides that such Indebtedness shall not be senior in right of payment to such Guarantor's Guarantee. Without limiting the generality of the foregoing, (x) "Guarantor Senior Indebtedness" shall include the principal of, premium, if any, and interest on all obligations of every nature of such Guarantor from time to time owed to the lenders under the Credit Facility, including, without limitation, principal of and interest on, and all fees, indemnities and expenses payable under, the Credit Facility, and (y) in the case of amounts owing under the Credit Facility and guarantees of Designated Senior Indebtedness, "Guarantor Senior Indebtedness" shall include interest accruing thereon subsequent to the occurrence of any Event of Default specified in clause (7) or (8) of Section 5.1 relating to such Guarantor, whether or not the claim for such interest is allowed under any applicable Bankruptcy Code. Notwithstanding the foregoing, "Guarantor Senior Indebtedness" shall not include (a) Indebtedness evidenced by the Securities or the Guarantees, (b) Indebtedness that is expressly subordinate or junior in right of payment to any other Indebtedness of such Guarantor, (c) Indebtedness which, when incurred and without respect to any election under Section 1111(b) of Title 11, United States Code, is by its terms without recourse to such Guarantor, (d) Indebtedness which is represented by Redeemable Capital Stock, (e) to the extent it constitutes Indebtedness, any liability for foderel, state, lead or state for the state to be a state of the state. federal, state, local or other taxes owed or owing by such Guarantor, (f) Indebtedness of such Guarantor to the Company or a Subsidiary of the Company or any other Affiliate of the Company or any of such Affiliate's Subsidiaries, and (g) that portion of any Indebtedness which is incurred by such Guarantor in violation of the Indenture.

"Guarantor Subordinated Indebtedness" means, with respect to a Guarantor, indebtedness and other obligations of such Guarantor which are expressly subordinated in right of payment to such Guarantor's Guaranty.

"Guarantors" shall mean each Initial Guarantor and each future Subsidiary designated a Guarantor in accordance with Section 10.17 herein.

"Guaranty" means each guaranty of the Securities contained in Article XIII given by each Guarantor.

"Guaranty Obligations" means, with respect to each Guarantor, the obligations of such Guarantor under Article XIII.

"Holder" means a Person in whose name a Security is registered in the Security Register.

"IAI Global Security" means a permanent global Security in registered form representing the aggregate principal amount of Securities sold to Institutional Accredited Investors.

"IAI Security" has the meaning set forth in Section 3.3.

"Indebtedness" means, with respect to any Person, without duplication, (a) all liabilities of such Person for borrowed money or for the deferred purchase price of property or services, excluding any trade payables and other accrued current liabilities incurred in the ordinary course of business, but including, without limitation, all obligations, contingent or otherwise, of such Person in connection with any letters of credit, banker's acceptance or other similar credit transaction, if, and to the extent, any of the foregoing would appear as a liability on a balance sheet of such Person prepared in accordance with GAAP, (b) all obligations of such Person evidenced by bonds, notes, debentures or other similar instruments, if, and to the extent, any of the foregoing would appear as a liability on a balance sheet of such Person prepared in accordance with GAAP, (c) all indebtedness of such Person created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person (even if the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), but excluding consignments and trade accounts payable arising in the ordinary course of business, (d) all Capitalized Lease Obligations of such Person, (e) all Indebtedness referred to in the preceding clauses of other Persons and all dividends of other Persons, the payment of which is secured by (or for which the Holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien upon property (including, without limitation, accounts and contract rights) owned by such Person, even though such Person has not assumed or become liable for the payment of such Indebtedness (the amount of such obligation being deemed to be the lesser of the Fair Market Value of such property or asset or the amount of the obligation so secured), (f) all guarantees of Indebtedness referred to in this definition by such Person, (g) all Redeemable Capital Stock of such Person valued at the greater of its voluntary or involuntary maximum fixed repurchase price plus accrued dividends, and (h) all Interest Rate Protection Obligations of such Person; provided, however, that Indebtedness shall not include (i) Indebtedness arising from agreements of the Company or any Restricted Subsidiary providing for indemnification, adjustment or holdback of purchase price or similar obligations, in each case, incurred or assumed in connection with the acquisition or disposition of any business, assets or a Subsidiary, other than guarantees of Indebtedness incurred by any Person acquiring all or any portion of such business, assets or Subsidiary for the purpose of financing such acquisition, or (ii) obligations under performance bonds, performance guarantees, surety bonds, appeal bonds, security deposits or similar obligations. For purposes hereof, the "maximum fixed repurchase price" of any Redeemable Capital Stock which does not have a fixed repurchase price shall be calculated in accordance with the terms of such Redeemable Capital Stock as if such Redeemable Capital Stock were purchased on any date on which Indebtedness shall be required to be determined pursuant to the Indenture, and if such price is based upon, or measured by, the fair market value of such Redeemable Capital Stock, such fair market

"Indenture" means this instrument as originally executed or as it may from time to time be supplemented or amended by one or more indentures supplemental hereto entered into pursuant to the applicable provisions hereof, including, for all purposes of this instrument and any such supplemental indenture, the provisions of the Trust Indenture Act that are deemed to be a part of and govern this instrument and any such supplemental indenture, respectively.

"Initial Guarantors" has the meaning set forth in the introduction to this Indenture.

"Initial Purchasers" means J.P.Morgan Securities Inc., Merrill Lynch, Pierce, Fenner & Smith Incorporated, Credit Lyonnais Securities (USA) Inc., Scotia Capital (USA) Inc. and TD Securities (USA) Inc.

"Initial Securities" has the meaning set forth in the first paragraph of the RECITALS OF THE COMPANY of this Indenture.

"Institutional Accredited Investor" or "IAI" means an institution that is an "accredited investor" as defined in Rule 501(a)(1), (2), (3) or (7) under the Securities Act, that is not also a QIB.

"Interest Payment Date" means the Stated Maturity of an installment of interest on the Securities.

"Interest Rate Protection Agreement" means, with respect to any Person, any arrangement with any other Person whereby, directly or indirectly, such Person is entitled to receive from time to time periodic payments calculated by applying either a floating or a fixed rate of interest on a stated notional amount in exchange for periodic payments made by such Person calculated by applying a fixed or a floating rate of interest on the same notional amount and shall include, without limitation, interest rate swaps, caps, floors, collars and similar agreements or arrangements designed to protect against or manage such Person's exposure to fluctuations in interest rates.

"Interest Rate Protection Obligations" means the obligations of any Person pursuant to any Interest Rate Protection Agreements.

"Investment" means, with respect to any Person, any direct or indirect loan or other extension of credit (including, without limitation, a guarantee) or capital contribution to (by means of any transfer of cash or other property to others or any payment for property or services for the account or use of others), or any purchase or acquisition by such Person of any Capital Stock, bonds, notes, debentures or other securities or evidences of Indebtedness issued by, any other Person, provided that the term "Investment" shall not include (a) extensions of trade credit on commercially reasonable terms in accordance with normal trade practices and (b) Interest Rate Protection Obligations entered into in the ordinary course of business.

"Issue Date" means May 29, 2001.

"Lien" means any mortgage, charge, pledge, lien (statutory or other), security interest, hypothecation, assignment for security, claim or other encumbrance upon or with respect to any property of any kind. A Person shall be deemed to own subject to a Lien any property which such Person has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement. "Moody's" means Moody's Investors Service, Inc. and its successors.

"Net Cash Proceeds" means, with respect to any Asset Sale, the proceeds thereof received by the Company or any Restricted Subsidiary in the form of cash or Cash Equivalents including payments in respect of deferred payment obligations when received in the form of cash or Cash Equivalents (except to the extent that such obligations are financed or sold with recourse to the Company or any Restricted Subsidiary) net of (i) brokerage commissions and other fees and expenses (including, without limitation, fees and expenses of legal counsel and investment bankers, recording fees, transfer fees and appraisers' fees) related to such Asset Sale, (ii) provisions for all taxes payable as a result of such Asset Sale, (iii) amounts required to be paid to any Person (other than the Company or any Restricted Subsidiary) owning a beneficial interest in the assets subject to the Asset Sale, and (v) appropriate amounts to be provided by the Company or any Restricted Subsidiary, as the case may be, as a reserve required in accordance with GAAP against any liabilities associated with such Asset Sale, including, without limitation, pension and other post-employment benefit liabilities, liabilities related to environmental matters and liabilities under any indemnification obligations associated with such Asset Sale, norwided, however, that any amounts remaining after adjustments, revaluations or liquidations of such reserves shall constitute Net Cash Proceeds.

"1999 Indenture" has the meaning specified in Section 10.15.

"Non-U.S. Person" means a Person that is not a U.S. Person as such term is defined in Regulation S.

"Notice of Default" means a written notice of the kind specified in Section 5.2.

"Offer" means a Change of Control Offer or an Asset Sale Offer.

"Offer to Purchase" means an Offer, sent by or on behalf of the Company by first-class mail, postage prepaid, to each Holder of Securities at its address appearing in the register for the Securities on the date of the Offer, offering to purchase up to the principal amount of Securities specified in such Offer at the purchase price specified in such Offer (as determined pursuant to this Indenture). Unless otherwise provided in Section 10.13 or 10.14 or otherwise required by applicable law, the Offer shall specify an expiration date (the "Expiration Date") of the Offer to Purchase, which shall be not less than 20 Business Days nor more than 60 days after the date of such Offer (or such later date as may be necessary for the Company to comply with the Exchange Act), and a settlement date (the "Purchase Date") for purchase of Securities to occur no later than five Business Days after the Expiration Date. The Company shall notify the Trustee at least 15 Business Days (or such shorter period as is acceptable to the Trustee) prior to the mailing of the Offer of the Company's obligation to make an Offer to Purchase, and the Offer shall be mailed by the Company or, at the Company's request, by the Trustee in the name and at the expense of the Company. The Offer shall contain all instructions and materials necessary to enable such Holders to tender Securities pursuant to the Offer to Purchase. The Offer shall also state:

> (1) the Section of this Indenture pursuant to which the Offer to Purchase is being made;

> > (2) the Expiration Date and the Purchase Date;

(3) the purchase price to be paid by the Company for each \$1,000 aggregate principal amount of Securities accepted for payment (as specified pursuant to this Indenture) (the "Purchase Price"); and the amount of accrued and unpaid interest to be paid;

(4) that the Holder may tender all or any portion of the Securities registered in the name of such Holder and that any portion of a Security tendered must be tendered in an integral multiple of \$1,000 principal amount;

(5) the place or places where Securities are to be surrendered for tender pursuant to the Offer to Purchase;

(6) that interest on any Security not tendered or tendered but not purchased by the Company pursuant to the Offer to Purchase will continue to accrue;

(7) that on the Purchase Date the Purchase Price will become due and payable upon each Security being accepted for payment pursuant to the Offer to Purchase and that interest thereon shall cease to accrue on and after the Purchase Date;

(8) that each Holder electing to tender all or any portion of a Security pursuant to the Offer to Purchase will be required to surrender such Security at the place or places specified in the Offer prior to the close of business on the Expiration Date (such Security being, if the Company or the Trustee so requires, duly endorsed by, or accompanied by a written instrument of transfer in form satisfactory to the Company and the Trustee duly executed by the Holder thereof or his attorney duly authorized in writing);

(9) that Holders will be entitled to withdraw all or any portion of Securities tendered if the Company (or its Paying Agent) receives, not later than the close of business on the fifth Business Day next preceding the Expiration Date, a facsimile transmission or letter setting forth the name of the Holder, the principal amount of the Security the Holder tendered, the certificate number of the Security the Holder tendered and a statement that such Holder is withdrawing all or a portion of his tender;

(10) that (a) if Securities purchasable at an aggregate Purchase Price less than or equal to the Purchase Amount are duly tendered and not withdrawn pursuant to the Offer to Purchase, the Company shall purchase all such Securities and (b) if Securities purchasable at an aggregate Purchase Price in excess of the Purchase Amount are tendered and not withdrawn pursuant to the Offer to Purchase, the Company shall purchase Securities on a pro rata basis based on the Purchase Price therefor (subject in each case to applicable rules of the Depositary and any securities exchange upon which the Securities may then be listed), with such adjustments as may be deemed appropriate so that only Securities in denominations of \$1,000 principal face amount or integral multiples thereof shall be purchased; and

(11) that in the case of a Holder whose Security is purchased only in part, the Company shall execute and the Trustee shall authenticate and deliver to the Holder of such Security without service charge, a new Security or Securities, of any authorized denomination as requested by such Holder, in an aggregate principal amount equal to and in exchange for the unpurchased portion of the Security so tendered.

An Offer to Purchase shall be governed by and effected in accordance with the provisions of this Indenture pertaining to the type of Offer to which it relates.

"Offered Price" has the meaning specified in Section 10.14.

"Offering Memorandum" means the Offering Memorandum dated May 23, 2001 pursuant to which the Initial Securities were offered, and any supplement thereto.

"Officer's Certificate" means a certificate signed by the Chairman of the Board, the Chief Executive Officer, the President or a Vice President, the Chief Financial Officer, the Treasurer, an Assistant Treasurer, the Secretary or an Assistant Secretary, of the Company, and delivered to the Trustee. One of the officers signing an Officer's Certificate given pursuant to Section 10.20 shall be the principal executive, financial or accounting officer of the Company.

"144A Global Security" means a permanent global security in registered form representing the aggregate principal amount of Securities sold in reliance on Rule 144A under the Securities Act.

"Opinion of Counsel" means a written opinion of counsel, who may be counsel for the Company, and who shall be acceptable to the Trustee.

"Outstanding," when used with respect to Securities, means, as of the date of determination, all Securities theretofore authenticated and delivered under this Indenture, except:

(i) Securities theretofore canceled by the Trustee or delivered to the Trustee for cancellation;

(ii) Securities for whose payment or redemption money in the necessary amount has been theretofore deposited with the Trustee or any Paying Agent (other than the Company) in trust or set aside and segregated in trust by the Company (if the Company shall act as its own Paying Agent) for the Holders of such Securities; provided that, if such Securities are to be redeemed, notice of such redemption has been duly given pursuant to this Indenture or provision therefor satisfactory to the Trustee has been made;

> (iii) Securities which have been paid pursuant to Section 3.6 or in exchange for or in lieu of which other Securities have been authenticated and delivered pursuant to this Indenture, other than any such Securities in respect of which there shall have been presented to the Trustee proof satisfactory to it that such Securities are held by a bona fide purchaser in whose hands such Securities are valid obligations of the Company; and

(iv) Securities as to which Defeasance has been effected pursuant to Section 12.2;

provided, however, that in determining whether the Holders of the requisite principal amount of the Outstanding Securities have given, made or taken any request, demand, authorization, direction, notice, consent, waiver or other action hereunder as of any date, Securities owned by the Company or any other obligor upon the Securities or any Affiliate of the Company or of such other obligor shall be disregarded and deemed not to be Outstanding (it being understood that Securities to be acquired by the Company pursuant to an Offer or other offer to purchase shall not be deemed to be owned by the Company until legal title to such Securities passes to the Company), except that, in determining whether the Trustee shall be protected in relying upon any such request, demand, authorization, direction, notice, consent, waiver or other action, only Securities which a Responsible Officer of the Trustee actually knows to be so owned shall be so disregarded. Securities so owned which have been pledged in good faith may be regarded as Outstanding if the pledgee establishes to the satisfaction of the Trustee the pledgee's right so to act with respect to such Securities and that the pledgee is not the Company or any other obligor upon the Securities or any Affiliate of the Company or of such other obligor.

"Pari Passu Indebtedness" means any Indebtedness of the Company that is pari passu in right of payment to the Securities.

"Pari Passu Indebtedness Amount" has the meaning specified in Section 10.14.

"Pari Passu Offer" has the meaning specified in Section 10.14.

"Paying Agent" means any Person authorized by the Company to pay the principal of (and premium, if any) or interest on any Securities on behalf of the Company.

"Permitted Founder Stock Repurchases" means one or more repurchases by the Company, for an aggregate purchase price not to exceed \$10,000,000 since January 29, 1999, of shares of Common Stock owned by former owners of Subsidiaries of the Company, that were, as of the date of acquisition of such stock by such Persons, subject to contractual agreements with the Company restricting their resale.

"Permitted Indebtedness" means, without duplication:

 (a) Indebtedness of the Company and the Guarantors evidenced by the Securities issued on the Issue Date, the Exchange Securities issued in exchange thereof and the Guarantees in respect thereof;

(b) Indebtedness of the Company and any Guarantor under the Credit Facility in an aggregate principal amount at any one time outstanding not to exceed \$250,000,000, less any amounts permanently repaid in accordance with Section 10.14;

(c) Indebtedness of the Company or any Guarantor outstanding as of the Issue Date, after giving effect to the application of proceeds contemplated by the Offering Memorandum;

(d) Indebtedness of the Company or any Restricted Subsidiary incurred in respect of bankers' acceptances and letters of credit in the ordinary course of business, including Indebtedness evidenced by letters of credit issued in the ordinary course of business to support the insurance or self-insurance obligations of the Company or any of its Restricted Subsidiaries (including to secure workers' compensation and other similar insurance coverages), in an aggregate amount not to exceed \$15,000,000 at any time, but excluding letters of credit issued in respect of or to secure money borrowed;

(e) (i) Interest Rate Protection Obligations of the Company or a Guarantor covering Indebtedness of the Company or a Guarantor and (ii) Interest Rate Protection Obligations of any Restricted Subsidiary covering Permitted Indebtedness or Acquired Indebtedness of such Restricted Subsidiary; provided that, in the case of either clause (i) or (ii), (x) any Indebtedness to which any such Interest Rate Protection Obligations correspond bears interest at fluctuating interest rates and is otherwise permitted to be incurred under Section 10.8 and (y) the notional principal amount of any such Interest Rate Protection Obligations that exceeds 105% of the principal amount of the Indebtedness to which such Interest Rate Protection Obligations relate shall not constitute Permitted Indebtedness;

(f) Indebtedness of a Restricted Subsidiary owed to and held by the Company or another Restricted Subsidiary, except that (i) any transfer of such Indebtedness by the Company or a Restricted Subsidiary (other than to the Company or another Restricted Subsidiary), (ii) the sale, transfer or other disposition by the Company or any Restricted Subsidiary of Capital Stock of a Restricted Subsidiary which is owed Indebtedness of another Restricted Subsidiary such that it shall no longer be a Restricted Subsidiary and (iii) the designation of a Restricted Subsidiary which is owed Indebtedness of another Restricted Subsidiary as an Unrestricted Subsidiary shall, in each case, be an incurrence of Indebtedness by such Restricted Subsidiary subject to the other provisions of the Indenture; (h) Indebtedness of the Company or any Guarantor represented by Capitalized Lease Obligations, mortgage financings or purchase money obligations, in each case incurred for the purpose of financing all or any part of the purchase price or cost of construction or improvement of property, plant or equipment used in the business of the Company or such Guarantor, in an aggregate principal amount not to exceed \$25,000,000 at any time outstanding;

(i) Subordinated Indebtedness of the Company, in an aggregate principal amount not to exceed \$10,000,000 at any time outstanding, that is convertible into Common Stock and issued in connection with an Asset Acquisition of a business engaged in the provision of electrical contracting and maintenance services to the commercial, industrial, power line and data cabling markets and any other businesses reasonably related thereto;

(j) Indebtedness of the Company, in addition to that described in clauses (a) through (i) of this definition, in an aggregate principal amount not to exceed \$30,000,000 at any time outstanding;

(k) (i) Indebtedness of the Company the proceeds of which are used solely to refinance (whether by amendment, renewal, extension or refunding) Indebtedness of the Company or any of the Guarantors incurred pursuant to the Consolidated Fixed Charge Coverage Ratio test of the proviso of Section 10.8 or clause (a), (c) or (k) of this definition and (ii) Indebtedness of any Guarantor the proceeds of which are used solely to refinance (whether by amendment, renewal, extension or refunding) Indebtedness of such Guarantor incurred pursuant to the Consolidated Fixed Charge Coverage Ratio test of the proviso of Section 10.8 or clause (c) or (k) of this definition; provided, however, that (x) the principal amount of Indebtedness incurred pursuant to this clause (k) (or if such Indebtedness provides for an amount less than the principal amount thereof to be due and payable upon a declaration of acceleration of maturity thereof, the original issue price of such Indebtedness) shall not exceed the sum of the principal amount of Indebtedness so refinanced, plus the amount of any premiums and fees required to be paid in connection with such refinancing pursuant to the terms of such Indebtedness, and (y) any Indebtedness incurred pursuant to this clause (k) (A) has no scheduled principal payment prior to the 91st day after the Maturity Date, (B) has an Average Life to Stated Maturity greater than the remaining Average Life to Stated Maturity of the Securities and (C) is subordinated to the Securities or the Guarantees, as the case may be, at least to the same extent that the Indebtedness being refinanced is subordinated to the Securities or the Guarantees, as the case may be;

(1) Indebtedness of any Restricted Subsidiary that constitutes Acquired Indebtedness not incurred in contemplation of the acquisition of such Restricted Subsidiary; provided that such Indebtedness is repaid within 90 days following the consummation of the Asset Acquisition in which the Company acquired such Restricted Subsidiary; and

(m) Guarantees by the Company or guarantees by a Guarantor of Indebtedness that was permitted to be incurred under the Indenture.

"Permitted Investments" means any of the following: (i) Investments in the Company or in a Restricted Subsidiary; (ii) Investments in another Person, if as a result of such Investment (A) such other Person becomes a Restricted Subsidiary or (B) such other Person is merged or consolidated with or into, or transfers or conveys all or substantially all of its assets to the Company or a Restricted Subsidiary; (iii) Investments representing Capital Stock or obligations issued to the Company or any of its Restricted Subsidiaries in settlement of debts created in the ordinary course of business or claims against any other Person by reason of a composition or readjustment of debt or a reorganization of any debtor of the Company or such Restricted Subsidiary or in Agreements on commercially reasonable terms entered into by the Company or any of its Restricted Subsidiaries in the ordinary course of business in connection with the operations of the business of the Company or its Restricted Subsidiaries to hedge against fluctuations in interest rates on its outstanding Indebtedness; (v) Investments in the Securities; (vi) Investments in Cash Equivalents; (vii) Investments acquired by the Company or any Restricted Subsidiary in connection with an Asset Sale permitted under Section 10.14 to the extent such Investments are non-cash proceeds as permitted under Section 10.14; (viii) any Investment to the extent that the consideration therefor is Capital Stock (other than Redeemable Capital Stock) of the Company; (ix) any loans or other advances made pursuant to any employee benefit plans (including plans for the benefit of directors) or employment agreements or other compensation arrangements (including for the purchase of Capital Stock by such employees), in each case as approved by the Board of Directors of the Company in its good faith judgment, not to exceed \$1,000,000 at any one time outstanding; and (x) other Investments not to exceed \$5,000,000 at any time outstanding.

"Permitted Junior Securities" means Capital Stock of the Company or debt securities that are subordinated to all Senior Indebtedness (and any debt securities issued in exchange for Senior Indebtedness) to at least the same extent as the Securities are subordinated to Senior Indebtedness.

"Permitted Liens" means the following types of Liens:

(a) any Lien existing as of the date of the Indenture;

(b) any Lien securing Acquired Indebtedness created prior to (and not created in connection with, or in contemplation of) the incurrence of such Indebtedness by the Company or any Restricted Subsidiary, if such Lien does not attach to any property or assets of the Company or any Restricted Subsidiary other than the property or assets subject to the Lien prior to such incurrence;

(c) Liens in favor of the Company or a Restricted Subsidiary;

 (d) Liens on and pledges of the Capital Stock of any Unrestricted Subsidiary securing any Indebtedness of such Unrestricted Subsidiary;

(e) Liens for taxes, assessments or governmental charges or claims, to the extent any such changes or claims constitute Indebtedness, either (i) not delinquent or (ii) contested in good (f) Liens incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance or other kinds of social security, old age pension or public liability obligations;

(g) Liens to secure Indebtedness (including Capitalized Lease Obligations) permitted by clause (h) under the definition of Permitted Indebtedness covering only the assets acquired with such indebtedness;

(h) Liens securing Interest Rate Protection Obligations permitted to be entered into pursuant to Section 10.8;

(i) judgment and attachment Liens not giving rise to an Event of Default or Liens created by or existing from any litigation or legal proceeding that are currently being contested in good faith by appropriate proceedings and for which adequate reserves have been made;

(j) Liens in favor of collecting or payor banks having a right of setoff, revocation, refund or chargeback with respect to money or instruments of the Company or any Subsidiary on deposit with or in possession of such bank; and

(k) Liens not otherwise permitted by clauses (a) through (j) that are incurred in the ordinary course of business of the Company or any Restricted Subsidiary with respect to Indebtedness that does not exceed \$5,000,000 at any one time outstanding.

"Person" means any individual, corporation, partnership (general or limited), limited liability company, joint venture, association, joint-stock company, trust, unincorporated organization or government or any agency or political subdivision thereof.

"Preferred Stock," as applied to any Person, means Capital Stock of any class or series (however designated) which is preferred as to the payment of dividends or distributions, or as to the distribution of assets upon any voluntary or involuntary liquidation or dissolution of such Person, over shares of Capital Stock of any other class or series of such Person.

"Private Exchange Securities" has the meaning set forth in the Registration Rights Agreement.

"Private Placement Legend" shall mean the legend initially set forth on the Securities in the form set forth on Exhibit A-1.

"Purchase Amount" means, with respect to an Offer to Purchase, the maximum aggregate amount payable by the Company for Securities under the terms of such Offer to Purchase, if such Offer to Purchase were accepted in respect of all Securities.

"Purchase Date" shall have the meaning set forth in the definition of "Offer to Purchase."

"Qualified Equity Offering" means (i) any public sale of Common Stock of the Company pursuant to a registration statement filed with the Commission in accordance with the Securities Act (other than any public offerings with respect to the Company's Common Stock registered on Form S-8 or Form S-4) or (ii) any private placement for aggregate proceeds of at least \$25,000,000 to a third party of Common Stock or Capital Stock (other than Redeemable Capital Stock) that is convertible into Common Stock. "Qualified Institutional Buyer" or "QIB" has the meaning specified in Rule 144A under the Securities Act.

"Record Expiration Date" has the meaning specified in Section 1.4.

"Redeemable Capital Stock" means any class or series of Capital Stock that, either by its terms, by the terms of any security into which it is convertible or exchangeable or by contract or otherwise, is or upon the happening of an event or passage of time would be required to be redeemed prior to the 91st day after the Maturity Date or is redeemable at the option of the holder thereof at any time prior to the 91st day after the Maturity Date, or is convertible into or exchangeable for debt securities at any time prior to the 91st day after the Maturity Date; provided that Capital Stock will not constitute Redeemable Capital Stock solely because the holders thereof have the right to require the Company to repurchase or redeem such Capital Stock upon the occurrence of a Change of Control or an Asset Sale.

"Redemption Date," when used with respect to any Security to be redeemed, means the date fixed for such redemption by or pursuant to this Indenture.

"Redemption Price," when used with respect to any Security to be redeemed, means the price at which it is to be redeemed pursuant to this Indenture.

"Registration Rights Agreement" means the Exchange and Registration Rights Agreement relating to the Securities dated May 29, 2001 by and among the Company, the Guarantors and the Initial Purchasers, as the same may be amended, supplemented or otherwise modified from time to time in accordance with the terms thereof, and, with respect to any Additional Securities, one or more substantially similar registration rights agreements between the Company and the other parties thereto, as such agreement(s) may be amended, from time to time.

"Regular Record Date" for the interest payable on any Interest Payment Date means the January 15 or July 15 (whether or not a Business Day), as the case may be, next preceding such Interest Payment Date.

"Regulation S" means Regulation S under the Securities Act.

"Regulation S Global Security" means a permanent global Security in registered form representing the aggregate principal amount of Securities sold in reliance on Regulation S under the Securities Act.

"Regulation S Security" has the meaning set forth in Section 3.3.

"Replacement Assets" has the meaning specified in Section 10.14.

"Required Filing Dates" has the meaning specified in Section 10.19.

"Resale Restriction Termination Date" has the meaning specified in Section 3.14(a).

"Responsible Officer," when used with respect to the Trustee, means any officer within the Corporate Trust Office, including, any vice president, any assistant vice president, any assistant secretary, any assistant treasurer, or any other officer of the Trustee customarily performing functions similar to those performed by any of the above designated officers and also means, with respect to a particular corporate trust matter, any other officer to whom such matter is referred because of his knowledge of and familiarity with the particular subject.

"Restricted Payments" has the meaning specified in Section 10.9.

"Restricted Security" means a Security that constitutes a "restricted security" within the meaning of Rule 144(a)(3) under the Securities Act; provided, however, that the Trustee shall be entitled to request and conclusively rely on an opinion of counsel with respect to whether any Security constitutes a Restricted Security.

"Restricted Subsidiary" means any Subsidiary of the Company that is not an Unrestricted Subsidiary.

"Revocation" has the meaning set forth in Section 10.18.

"Rule 144A" means Rule 144A under the Securities Act.

"Rule 144A Security" has the meaning set forth in Section 3.3.

"S&P" means Standard & Poor's Ratings Group, and its successors.

"Securities" has the meaning set forth in the first paragraph of the RECITALS OF THE COMPANY of this Indenture.

"Securities Act" means the Securities Act of 1933 and any statute successor thereto, in each case as amended from time to time.

"Security Register" and "Security Registrar" have the respective meanings specified in Section 3.5.

"Senior Indebtedness" means the principal of, premium, if any, and interest on any Indebtedness of the Company, whether outstanding on the Issue Date or thereafter created, incurred or assumed, unless, in the case of any particular Indebtedness, the instrument creating or evidencing the same or pursuant to which the same is outstanding expressly provides that such Indebtedness shall not be senior in right of payment to the Securities. Without Indebtedness shall not be senior in right of payment to the Securities. Without limiting the generality of the foregoing, (x) "Senior Indebtedness" shall include the principal of, premium, if any, and interest on all obligations of every nature of the Company from time to time owed to the lenders under the Credit Facility, including, without limitation, principal of and interest on, and all fees, indemnities and expenses payable under, the Credit Facility, and (y) in the case of Designated Senior Indebtedness, "Senior Indebtedness" shall include interest accruing thereon subsequent to the occurrence of any Event of befault specified in clause (7) or (8) of Section 5.1 relating to the Company, whether or not the claim for such interest is allowed under any applicable Bankruptcy Code. Notwithstanding the foregoing, "Senior Indebtedness" shall not include (a) Indebtedness evidenced by the Securities, (b) Indebtedness that is expressly subordinate or junior in right of payment to any other Indebtedness of the Company, (c) Indebtedness which, when incurred and without respect to any election under Section 1111(b) of Title 11, United States Code, is by its terms without recourse to the Company, (d) Indebtedness which is represented by Redeemable Capital Stock, (e) to the extent it constitutes Indebtedness, any liability for federal, state, local or other taxes owed or owing by the Company, (f) Indebtedness of the Company to a Subsidiary of the Company or any other Affiliate of the Company or any of such Affiliate's Subsidiaries, and (g) that portion of any Indebtedness which is incurred by the Company in violation of this Indenture.

"Significant Subsidiary" of any Person means, as of any date of determination, a Restricted Subsidiary of such Person which would be a significant subsidiary of such Person as of such date as determined in accordance with the definition in Rule 1-02(w) of Regulation S-X promulgated by the Commission and as in effect on the date of this Indenture.

"Special Record Date" for the payment of any Defaulted Interest means a date fixed by the Trustee pursuant to Section 3.7.

"Stated Maturity" means, when used with respect to any Security or any installment of interest thereon, the date specified in such Security as the fixed date on which the principal of such Security or such installment of interest is due and payable, and when used with respect to any other Indebtedness, means the date specified in the instrument governing such Indebtedness as the fixed date on which the principal of such Indebtedness, or any installment of interest thereon, is due and payable.

"Subordinated Indebtedness" means, with respect to the Company, Indebtedness of the Company which is expressly subordinated in right of payment to the Securities.

"Subsidiary" means, with respect to any Person, (i) a corporation a majority of whose Voting Stock is at the time, directly or indirectly, owned by such Person, by one or more Subsidiaries of such Person or by such Person and one or more Subsidiaries thereof and (ii) any other Person (other than a corporation), including, without limitation, a partnership, limited liability company, business trust or joint venture, in which such Person, one or more Subsidiaries thereof or such Person and one or more Subsidiaries thereof, directly or indirectly, at the date of determination thereof, have at least majority ownership interest entitled to vote in the election of directors, managers or trustees thereof (or other Person performing similar functions). For purposes of this definition, any directors' qualifying shares or investments by foreign nationals mandated by applicable law shall be disregarded in determining the ownership of a Subsidiary.

"Trustee" means the Person named as the "Trustee" in the first paragraph of this instrument until a successor Trustee shall have become such pursuant to the applicable provisions of this Indenture, and thereafter "Trustee" shall mean such successor Trustee.

"Trust Indenture Act" means the Trust Indenture Act of 1939 as in force at the date as of which this instrument was executed; provided, however, that in the event the Trust Indenture Act of 1939 is amended after such date, "Trust Indenture Act" means, to the extent required by any such amendment, the Trust Indenture Act of 1939 as so amended.

"Unrestricted Subsidiary" means each Subsidiary of the Company designated as such pursuant to and in compliance with Section 10.18 and each Subsidiary of any such Subsidiary.

"U.S. Government Obligation" has the meaning specified in Section 12.4.

"Vice President," when used with respect to the Company or the Trustee, means any vice president, whether or not designated by a number or a word or words added before or after the title "vice president."

"Voting Stock" means any class or classes of Capital Stock pursuant to which the holders thereof have the general voting power under ordinary circumstances to elect at least a majority of the board of directors, managers or trustees of any Person (irrespective of whether or not, at the time, stock of any other class or classes shall have, or might have, voting power by reason of the happening of any contingency) and, with respect to the Company, shall be deemed to include the Common Stock.

SECTION 1.2. Compliance Certificates and Opinions.

Upon any application or request by the Company or a Guarantor to the Trustee to take any action under any provision of this Indenture, the Company or the Guarantor shall furnish to the Trustee such certificates and opinions as may be required under the Trust Indenture Act. Each such certificate or opinion shall be given in the form of an Officer's Certificate, if to be given by an officer of the Company or a Guarantor, or an Opinion of Counsel, if to be given by counsel, and shall comply with the requirements of the Trust Indenture Act and any other requirement set forth in this Indenture.

Every certificate or opinion with respect to compliance with a condition or covenant provided for in this Indenture shall include:

(i) a statement that each individual signing such certificate or opinion has read such covenant or condition and the definitions herein relating thereto;

(ii) a brief statement as to the nature and scope of the examination or investigation upon which the statements or opinions contained in such certificate or opinion are based;

(iii) a statement that, in the opinion of each such individual, he has made such examination or investigation as is necessary to enable him to express an informed opinion as to whether or not such covenant or condition has been complied with; and

(iv) a statement as to whether, in the opinion of each such individual, such condition or covenant has been complied with.

SECTION 1.3. Form of Documents Delivered to Trustee.

In any case where several matters are required to be certified by, or covered by an opinion of, any specified Person, it is not necessary that all such matters be certified by, or covered by the opinion of, only one such Person, or that they be so certified or covered by only one document, but one such Person may certify or give an opinion with respect to some matters and one or more other such Persons as to other matters, and any such Person may certify or give an opinion as to such matters in one or several documents.

Any certificate or opinion of an officer of the Company or a Guarantor may be based, insofar as it relates to legal matters, upon a certificate or opinion of, or representations by, counsel, unless such officer knows, or in the exercise of reasonable care should know, that the certificate or opinion or representations with respect to the matters upon which his certificate or opinion is based are erroneous. Any such certificate or opinion of counsel may be based, insofar as it relates to factual matters, upon a certificate or opinion of, or representations by, an officer or officers of the Company or a Guarantor stating that the information with respect to such factual matters is in the possession of the Company or such Guarantor, unless such counsel knows, or in the exercise of reasonable care should know, that the certificate or opinion or representations with respect to such matters are erroneous.

Where any Person is required to make, give or execute two or more applications, requests, consents, certificates, statements, opinions or other instruments under this Indenture, they may, but need not, be consolidated and form one instrument.

SECTION 1.4. Acts of Holders; Record Dates.

Any request, demand, authorization, direction, notice, consent, waiver or other action provided or permitted by this Indenture to be given or taken by Holders may be embodied in and evidenced by one or more instruments of substantially similar tenor signed by such Holders in person or by agent duly appointed in writing; and, except as herein otherwise expressly provided, such action shall become effective when such instrument or instruments are delivered to the Trustee and, where it is hereby expressly required, to the Company or a Guarantor, as applicable. Such instrument or instruments (and the action embodied therein and evidenced thereby) are herein sometimes referred to as the "Act" of the Holders signing such instrument or instruments. Proof of execution of any such instrument or of a writing appointing any such agent shall be sufficient for any purpose of this Indenture and (subject to Section 6.1) conclusive in favor of the Trustee and the Company, if made in the manner provided in this Section.

The fact and date of the execution by any Person of any such instrument or writing may be proved by the affidavit of a witness of such execution or by a certificate of a notary public or other officer authorized by law to take acknowledgments of deeds, certifying that the individual signing such instrument or writing acknowledged to him the execution thereof. Where such execution is by a signer acting in a capacity other than his individual capacity, such certificate or affidavit shall also constitute sufficient proof of his authority. The fact and date of the execution of any such instrument or writing, or the authority of the Person executing the same, may also be proved in any other manner which the Trustee deems sufficient.

The ownership of Securities shall be proved by the Security Register.

Any request, demand, authorization, direction, notice, consent, waiver or other Act of the Holder of any Security shall bind every future Holder of the same Security and the Holder of every Security issued upon the registration of transfer thereof or in exchange therefor or in lieu thereof in respect of anything done, omitted or suffered to be done by the Trustee, the Company or a Guarantor in reliance thereon, whether or not notation of such action is made upon such Security.

The Company may set any day as a record date for the purpose of determining the Holders of Outstanding Securities entitled to give or take any request, demand, authorization, direction, notice, consent, waiver or other action provided or permitted by this Indenture to be given or taken by Holders of Securities, provided that the Company may not set a record date for, and the provisions of this paragraph shall not apply with respect to, the giving or making of any notice, declaration, request or direction referred to in the next paragraph. If any record date is set pursuant to this paragraph, the Holders of Outstanding Securities on such record date, and no other Holders, shall be entitled to take the relevant action, whether or not such Holders remain Holders after such record date; provided that no such action shall be effective hereunder unless taken on or prior to the applicable Record Expiration Date by Holders of the requisite principal amount of Outstanding Securities on such record date. Nothing in this paragraph shall prevent the Company from setting a new record date for any action for which a record date has previously been set pursuant to this paragraph (whereupon the record date previously set shall automatically and with no action by any Person be cancelled and of no effect), nor shall anything in this paragraph be construed to render ineffective any action taken pursuant to or in accordance with any other provision of this Indenture by Holders of the requisite principal amount of Outstanding Securities on the date such action is taken. Promptly after any record date is set pursuant to this paragraph, the Company, at its own expense, shall cause notice of such record date, the proposed action by Holders and the applicable Record Expiration Date to be given to the Trustee in writing and to each Holder of Securities in the manner set forth in Section 1.6.

The Trustee may but need not set any day as a record date for the purpose of determining the Holders of Outstanding Securities entitled to join in the giving or making of (i) any Notice of Default,

(ii) any declaration of acceleration referred to in Section 5.2, (iii) any request to institute proceedings referred to in Section 5.7(ii) or (v) any direction referred to in Section 5.12. If any record date is set pursuant to this paragraph, the Holders of Outstanding Securities on such record date, and no other Holders, shall be entitled to join in such notice, declaration, request or direction, whether or not such Holders remain Holders after such record date; provided that no such action shall be effective hereunder unless taken on or prior to the applicable Record Expiration Date by Holders of the requisite principal amount of Outstanding Securities on such record date. Nothing in this paragraph shall be construed to prevent the Trustee from setting a new record date for any action (whereupon the record date previously set shall automatically and without any action by any Person be cancelled and of no effect), nor shall anything in this paragraph be construed to render ineffective any action taken pursuant to or in accordance with any other provision of this Indenture by Holders of the requisite principal amount of Outstanding Securities on the date such action is taken. Promptly after any record date is set pursuant to this paragraph, the Trustee, at the Company's expense, shall cause notice of such record date, the matter(s) to be submitted for potential action by Holders and the applicable Record Expiration Date to be given to the Company in writing and to each Holder of Securities in the manner set forth in Section 1.6.

With respect to any record date set pursuant to this Section, the party hereto that sets such record date may designate any day as the "Record Expiration Date" and from time to time may change the Record Expiration Date to any earlier or later day, provided that no such change shall be effective unless notice of the proposed new Record Expiration Date is given to the other party hereto in writing, and to each Holder of Securities in the manner set forth in Section 1.6, on or before the existing Record Expiration Date. If a Record Expiration Date is not designated with respect to any record date set pursuant to this Section, the party hereto that set such record date shall be deemed to have initially designated the 180th day after such record date as the Record Expiration Date with respect thereto, subject to its right to change the Record Expiration Date as provided in this paragraph. Notwithstanding the foregoing, no Record Expiration Date shall be later than the 180th day after the applicable record date.

Without limiting the foregoing, a Holder entitled hereunder to take any action hereunder with regard to any particular Security may do so with regard to all or any part of the principal amount of such Security or by one or more duly appointed agents each of which may do so pursuant to such appointment with regard to all or any part of such principal amount.

SECTION 1.5. Notices to Trustee, the Company or a Guarantor.

Any request, demand, authorization, direction, notice, consent, waiver or Act of Holders or other document provided or permitted by this Indenture to be made upon, given or furnished to, or filed with,

> (i) the Trustee by any Holder or by the Company or a Guarantor shall be sufficient for every purpose hereunder if made, given, furnished or filed in writing or mailed, first class postage prepaid, or sent by overnight courier, or sent by telecopier, to or with the Trustee at its Corporate Trust Office, Attention: Corporate Trust Administration. or

> (ii) the Company or a Guarantor by the Trustee or by any Holder shall be sufficient for every purpose hereunder (unless otherwise herein expressly provided) if in writing and mailed, first-class postage prepaid, to the Company or such Guarantor addressed to it at the address of the Company's principal office specified in the first paragraph of this instrument, or at any other address previously furnished in writing to the Trustee by the Company.

SECTION 1.6. Notice to Holders; Waiver.

Where this Indenture provides for notice to Holders of any event, such notice shall be sufficiently given (unless otherwise herein expressly provided) if in writing and mailed, first-class postage prepaid, to each Holder affected by such event, at his address as it appears in the Security Register, not later than the latest date (if any), and not earlier than the earliest date (if any), prescribed for the giving of such notice. In any case where notice to Holders is given by mail, neither the failure to mail or receive such notice, nor any defect in any such notice, to any particular Holder shall affect the sufficiency or validity of such notice. Where this Indenture provides for notice in any manner, such notice may be waived in writing by the Person entitled to receive such notice, either before or after the event, and such waiver shall be the equivalent of such notice. Waivers of notice by Holders shall be filed with the Trustee, but such filing shall not be a condition precedent to the validity of any action taken in reliance upon such waiver.

In case by reason of the suspension of regular mail service or by reason of any other cause it shall be impracticable to give such notice by mail, then such notification as shall be made with the approval of the Trustee shall constitute a sufficient notification for every purpose hereunder.

SECTION 1.7. Conflict with Trust Indenture Act.

If any provision hereof limits, qualifies or conflicts with a provision of the Trust Indenture Act that is required under the Trust Indenture Act to be part of and govern this Indenture, such provision of the Trust Indenture Act shall control. If any provision of this Indenture modifies or excludes any provision of the Trust Indenture Act that may be so modified or excluded, such provision shall be deemed to be so modified or excluded, as the case may be.

SECTION 1.8. Effect of Headings and Table of Contents.

The Article and Section headings herein and the Table of Contents are for convenience only and shall not affect the construction hereof.

SECTION 1.9. Successors and Assigns.

Without limiting Articles VIII and XIII hereof, all covenants and agreements in this Indenture by each of the Company or the Guarantors shall bind their respective successors and assigns, whether so expressed or not.

SECTION 1.10. Separability Clause.

In case any provision in this Indenture or in the Securities shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby.

SECTION 1.11. Benefits of Indenture.

Nothing in this Indenture or in the Securities, express or implied, shall give to any Person, other than the parties hereto and their successors hereunder and the Holders of Securities, any benefit or any legal or equitable right, remedy or claim under this Indenture.

SECTION 1.12. Governing Law.

This Indenture and the Securities shall be governed by and construed in accordance with the laws of the State of New York.

SECTION 1.13. Legal Holidays.

In any case where any Interest Payment Date, Redemption Date, Purchase Date or Stated Maturity of any Security shall not be a Business Day, then (notwithstanding any other provision of this Indenture or of the Securities) payment of interest or principal (and premium, if any) need not be made on such date, but may be made on the next succeeding Business Day with the same force and effect

(including with respect to the accrual of interest) as if made on the Interest Payment Date, Redemption Date or Purchase Date, or at the Stated Maturity.

ARTICLE II

SECURITY FORMS

SECTION 2.1. Forms Generally.

The Securities and the Trustee's certificates of authentication shall be in substantially the forms set forth or referenced in Exhibit A-1 and Exhibit A-2 annexed hereto, with such appropriate insertions, omissions, substitutions and other variations as are required or permitted by this Indenture, and may have such letters, numbers or other marks of identification and such legends or endorsements placed thereon as may be required to comply with the rules of any securities exchange or the Depository or as may, consistently herewith, be determined by the officers executing such Securities, as evidenced by their execution thereof.

ARTICLE III

THE SECURITIES

SECTION 3.1. Title and Terms.

The aggregate principal amount of Securities which may be authenticated and delivered under this Indenture is unlimited. The Initial Securities issued on the date hereof will be in an aggregate principal amount of \$125,000,000. In addition, the Company may issue, from time to time in accordance with the provisions of this Indenture, including, without limitation, Section 10.8 hereof, Additional Securities and Exchange Securities. Furthermore, Securities may be authenticated and delivered upon registration or transfer or in lieu of, other Securities pursuant to Section 3.4, 3.5, 3.6, 9.6 or 11.8 or in connection with an Offer pursuant to Sections 10.13 or 10.14.

The Initial Securities shall be known and designated as "9[]% Senior Subordinated Notes, Series C, due 2009" of the Company, Additional Securities issued as Restricted Securities shall be known and designated as "9[]% Senior Subordinated Notes, Series C, due 2009" of the Company, Additional Securities issued other than as Restricted Securities shall be known and designated as "9[]% Senior Subordinated Notes, Series D, due 2009" of the Company, and Exchange Securities shall be known and designated as "9[]% Senior Subordinated Notes, D due 2009" of the Company, and Notes due 2009, Series D" of the Company.

With respect to any Additional Securities, the Company shall set forth in a Board Resolution and an Officer's Certificate, the following information:

 (i) the aggregate principal amount of such Additional Securities to be authenticated and delivered pursuant to this Indenture;

(ii) the issue price and the issue date of such Additional Securities; and

(iii) whether such Additional Securities shall be Restricted Securities issued in the form of Exhibit A-1 hereto and/or shall be issued in the form of Exhibit A-2 hereto. The Stated Maturity for payment of principal of the Securities shall be February 1, 2009. Interest on the Securities shall accrue at the rate of 9[]% per annum (subject, in the case of the Initial Securities or any Additional Securities issued as Restricted Securities, to increase in the circumstances contemplated in the Registration Rights Agreement relating thereto) and shall be payable semi-annually on each February 1 and August 1, commencing August 1, 2001, to the Holders of record of Securities at the close of business on the January 15 and July 15, respectively, immediately preceding such Interest Payment Date. Interest on the Securities will accrue from the most recent date to which interest has been paid or, if no interest has been paid, from the Issue Date of such Securities. Interest will be computed on the basis of a 360-day year comprised of twelve 30-day months.

The Initial Securities, the Additional Securities and the Exchange Securities shall be considered collectively as a single class for all purposes of this Indenture. Holders of the Initial Securities, the Additional Securities and the Exchange Securities will vote and consent together on all matters to which such Holders are entitled to vote or consent as one class, and none of the Holders of the Initial Securities, the Additional Securities or the Exchange Securities shall have the right to vote or consent as a separate class on any matter to which such Holders are entitled to vote or consent.

The principal of (and premium, if any) and interest on the Securities shall be payable at the office or agency of the Trustee in the Borough of Manhattan, The City of New York or such other office maintained by the Trustee for such purpose and at any other office or agency maintained by the Company for such purpose; provided, however, that, at the option of the Company, payment of interest on Securities that are not held in global form may be made by check mailed to the address of the Person entitled thereto as such address shall appear in the Security Register subject to the right of any Holder of Securities in the principal amount of \$500,000 or more to request payment by wire transfer.

The Company may be required to make a Change of Control Offer as provided in Section 10.13, or an Asset Sale Offer as provided in Section 10.14.

The Securities shall be redeemable as provided in Article XI and the Securities.

The Securities shall be subject to Defeasance and/or Covenant Defeasance as provided in Article XII.

The other terms and provisions contained in the Securities shall constitute, and are hereby expressly made, a part of this Indenture.

SECTION 3.2. Denominations.

The Securities shall be issuable only in registered form without coupons and only in denominations of 1,000 principal amount and any integral multiple thereof.

SECTION 3.3. Execution, Authentication, Delivery and Dating.

The Initial Securities and the Trustee's certificate of authentication shall be substantially in the form of Exhibit A-1 hereto. The Exchange Securities and the Trustee's certificate of authentication shall be substantially in the form of Exhibit A-2 hereto; provided that any Exchange Securities issued as Private Exchange Securities shall contain the Private Placement Legend and shall be subject to the provisions of Section 3.14. Any Additional Securities (if issued as Restricted Securities) and the Trustee's certificate of authentication shall each be substantially in the form of Exhibit A-1 hereto. Any Additional Securities issued other than as Restricted Securities and the Trustee's certificate of authentication shall each be substantially in the form of Exhibit A-2 hereto. The terms and provisions contained in the Securities annexed hereto as Exhibits A-1 and A-2 shall constitute, and are hereby expressly, made, a part of this Indenture and, to the extent applicable, the Company and the Trustee, by their execution and delivery of this Indenture, expressly agree to such terms and provisions and to be bound thereby.

Securities offered and sold in reliance on Rule 144A (each, a "Rule 144A Security"), Securities offered and sold in reliance on Regulation S (each, a "Regulation S Security") and Securities resold to Institutional Accredited Investors (each, an "IAI Security") shall be issued initially in the form of one or more Global Securities, substantially in the form set forth in Exhibit A-1, deposited with the Trustee, as custodian for the Depository, duly executed by the Company and authenticated by the Trustee as hereinafter provided and shall bear the legend set forth in Exhibit B. The Series D Securities shall be issued initially in the form of one or more Global Securities, substantially in the form set forth in Exhibit A-2, deposited with the Trustee, as custodian for the Depository, duly executed by the Company and authenticated by the Trustee as hereinafter provided and shall bear the legend set forth in Exhibit B. The aggregate principal amount of the Global Securities may from time to time be increased or decreased by adjustments made on the records of the Trustee, as custodian for the Depository, as hereinafter provided.

All Securities shall remain in the form of a Global Security, except as provided herein.

The Securities shall be executed on behalf of the Company by its Chairman of the Board, its Chief Executive Officer, its President or one of its Vice Presidents, or its Chief Financial Officer, attested by its Secretary or one of its Assistant Secretaries. The signature of any of these officers on the Securities may be manual or facsimile.

Securities bearing the manual or facsimile signatures of individuals who were at any time the proper officers of the Company shall bind the Company, notwithstanding that such individuals or any of them have ceased to hold such offices prior to the authentication and delivery of such Securities or did not hold such offices at the date of such Securities.

At any time and from time to time after the execution and delivery of this Indenture, the Company may deliver Securities executed by the Company to the Trustee for authentication, together with a Company Order for the authentication and delivery of such Securities; and the Trustee in accordance with such Company order shall authenticate and deliver such Securities as in this Indenture provided and not otherwise.

Each Security shall be dated the date of its authentication.

No Security shall be entitled to any benefit under this Indenture or be valid or obligatory for any purpose unless there appears on such Security a certificate of authentication substantially in the form provided for herein executed by the Trustee by manual signature, and such certificate upon any Security shall be conclusive evidence, and the only evidence, that such Security has been duly authenticated and delivered hereunder.

SECTION 3.4. Temporary Securities.

Pending the preparation of definitive Securities, the Company may execute, and upon Company Order the Trustee shall authenticate and deliver, temporary Securities which are printed, lithographed, typewritten, mimeographed or otherwise produced, in any authorized denomination, substantially of the tenor of the definitive Securities in lieu of which they are issued and with such appropriate insertions, omissions, substitutions and other variations as the officers executing such Securities may determine, as evidenced by their execution of such Securities. If temporary Securities are issued, the Company will cause definitive Securities to be prepared without unreasonable delay. After the preparation of definitive Securities, the temporary Securities shall be exchangeable for definitive Securities upon surrender of the temporary Securities at any office or agency of the Company designated pursuant to Section 10.2, without charge to the Holder. Upon surrender for cancellation of any one or more temporary Securities the Company shall execute and the Trustee shall authenticate and deliver in exchange therefor a like principal amount of definitive Securities of authorized denominations and of a like tenor. Until so exchanged the temporary Securities shall in all respects be entitled to the same benefits under this Indenture as definitive Securities.

SECTION 3.5. Registration, Registration of Transfer and Exchange.

The Company shall cause to be kept at the Corporate Trust Office of the Trustee a register (the register maintained in such office and in any other office or agency designated pursuant to Section 10.2 being herein sometimes collectively referred to as the "Security Register") in which, subject to such reasonable regulations as the Company may prescribe, the Company shall provide for the registration of Securities and of transfers of Securities. The Trustee is hereby appointed the initial "Security Registrar" for the purpose of registering Securities and transfers of Securities as herein provided.

Subject to Sections 3.13 and 3.14 of this Indenture, upon surrender for registration of transfer of any Security at an office or agency of the Company designated pursuant to Section 10.2 for such purpose, the Company shall execute, and the Trustee shall authenticate and deliver, in the name of the designated transferee or transferees, one or more new Securities of any authorized denomination and of a like aggregate principal amount and tenor.

At the option of the Holder, Securities may be exchanged for other Securities of any authorized denominations and of a like aggregate principal amount and tenor, upon surrender of the Securities to be exchanged at such office or agency. Whenever any Securities are so surrendered for exchange, the Company shall execute, and the Trustee shall authenticate and deliver, the Securities which the Holder making the exchange is entitled to receive.

All Securities issued upon any registration of transfer or exchange of Securities shall be the valid obligations of the Company, evidencing the same debt, and entitled to the same benefits under this Indenture, as the Securities surrendered upon such registration of transfer or exchange.

Every Security presented or surrendered for registration of transfer or for exchange shall (if so required by the Company or the Trustee) be duly endorsed, or be accompanied by a written instrument of transfer in form satisfactory to the Company and the Security Registrar duly executed, by the Holder thereof or his attorney duly authorized in writing.

No service charge shall be made for any registration of transfer or exchange of Securities, but the Company may require payment of a sum sufficient to cover any tax or other governmental charge that may be imposed in connection with any registration of transfer or exchange of Securities, other than exchanges pursuant to Section 3.4, 9.6 or 11.8 or in accordance with any Change of Control Offer pursuant to Section 10.13 or any Asset Sale Offer pursuant to Section 10.14, and in any such case not involving any transfer.

The Company shall not be required (i) to issue, register the transfer of or exchange any Security during a period beginning at the opening of business 15 days before the day of the mailing of a notice of redemption of Securities selected for redemption under Section 11.4 and ending at the close of business on the day of such mailing, (ii) to register the transfer of or exchange any Security so selected for redemption in whole or in part, except the unredeemed portion of any Security being redeemed in part or (iii) to register the transfer of any Securities other than Securities having a principal amount of \$1,000 or integral multiples thereof. If any mutilated Security is surrendered to the Trustee, the Company shall execute and the Trustee shall authenticate and deliver in exchange therefor a new Security of like tenor and principal amount and bearing a number not contemporaneously outstanding.

If there shall be delivered to the Company and the Trustee (i) evidence to their satisfaction of the destruction, loss or theft of any Security and (ii) such security or indemnity as may be required by them to save each of them and any agent of each of them harmless, then, in the absence of notice to the Company or the Trustee that such Security has been acquired by a bona fide purchaser, the Company shall execute, and upon its request the Trustee shall authenticate and deliver, in lieu of any such destroyed, lost or stolen Security, a new Security of like tenor and principal amount and bearing a number not contemporaneously outstanding.

In case any such mutilated, destroyed, lost or stolen Security has become or is about to become due and payable, the Company in its discretion may, instead of, issuing a new Security, pay such Security.

Upon the issuance of any new Security under this Section, the Company may require the payment of a sum sufficient to cover any tax or other governmental charge that may be imposed in relation thereto and any other expenses (including the fees and expenses of the Trustee) connected therewith.

Every new Security issued pursuant to this Section in lieu of any destroyed, lost or stolen Security shall constitute an original additional contractual obligation of the Company, whether or not the destroyed, lost or stolen Security shall be at any time enforceable by anyone, and shall be entitled to all the benefits of this Indenture equally and proportionately with any and all other Securities duly issued hereunder.

The provisions of this Section are exclusive and shall preclude (to the extent lawful) all other rights and remedies with respect to the replacement or payment of mutilated, destroyed, lost or stolen Securities.

SECTION 3.7. Payment of Interest; Rights Preserved.

Interest on any Security which is payable, and is punctually paid or duly provided for, on any Interest Payment Date shall be paid to the Person in whose name that Security (or one or more predecessor securities) is registered at the close of business on the Regular Record Date for such interest payment.

Any interest on any Security which is payable, but is not punctually paid or duly provided for, on any Interest Payment Date (herein called "Defaulted Interest") shall forthwith cease to be payable to the Holder on the relevant Regular Record Date by virtue of having been such Holder, and such Defaulted Interest may be paid by the Company, at its election in each case, as provided in clause (1) or (2) below:

> (1) The Company may elect to make payment of any Defaulted Interest to the Persons in whose names the Securities (or their respective predecessor Securities) are registered at the close of business on a Special Record Date for the payment of such Defaulted Interest, which shall be fixed in the following manner. The Company shall notify the Trustee in writing of the amount of Defaulted Interest proposed to be paid on each Security and the date of the proposed payment, and at the same time the Company shall deposit with the Trustee an amount of money equal to the aggregate amount proposed to be paid in respect of such Defaulted Interest or shall make arrangements satisfactory to the Trustee for such deposit prior to the date of the proposed payment, such money when deposited to be held in trust for the benefit of the Persons

entitled to such Defaulted Interest as in this clause provided. Thereupon the Trustee shall fix a Special Record Date for the payment of such Defaulted Interest which shall be not more than 15 days and not less than 10 days prior to the date of the proposed payment and not less than 15 days after the receipt by the Trustee of the notice of the proposed payment. The Trustee shall promptly notify the Company of such Special Record Date and, in the name and at the expense of the Company, shall cause notice of the proposed payment of such Defaulted Interest and the Special Record Date therefor to be given to each Holder in the manner specified in Section 1.6, not less than 10 days prior to such Special Record Date. Notice of the proposed payment of such Defaulted Interest and the Special Record Date therefor having been so mailed, such Defaulted Interest shall be paid to the Persons in whose names the Securities (or their respective predecessor Securities) are registered at the close of business on such Special Record Date and shall no longer be payable pursuant to the following clause (2).

(2) The Company may make payment of any Defaulted Interest in any other lawful manner not inconsistent with the requirements of any securities exchange on which the Securities may be listed, and upon such notice as may be required by such exchange, if, after notice given by the Company to the Trustee of the proposed payment pursuant to this Clause, such manner of payment shall be deemed practicable by the Trustee.

Subject to the foregoing provisions of this Section, each Security delivered under this Indenture upon registration of transfer of or in exchange for or in lieu of any other Security shall carry the rights to interest accrued and unpaid, and to accrue, which were carried by such other Security.

SECTION 3.8. Persons Deemed Owners.

Prior to due presentment of a Security for registration of transfer, the Company, the Trustee and any agent of the Company or the Trustee may treat the Person in whose name such Security is registered as the owner of such Security for the purpose of receiving payment of principal of (and premium, if any) and (subject to Section 3.7) interest on such Security and for all other purposes whatsoever, whether or not such Security be overdue, and neither the Company, the Trustee nor any agent of the Company or the Trustee shall be affected by notice to the contrary.

SECTION 3.9. Cancellation.

All Securities surrendered for payment, redemption, registration of transfer or exchange or tendered and accepted pursuant to any Change of Control Offer pursuant to Section 10.13 or any Asset Sale Offer pursuant to Section 10.14 shall, if surrendered to any Person other than the Trustee, be delivered to the Trustee and shall be promptly cancelled by it. The Company may at any time deliver to the Trustee for cancellation any Securities previously authenticated and delivered hereunder which the Company may have acquired in any manner whatsoever, and all Securities so delivered shall be promptly cancelled by the Trustee. No Securities shall be authenticated in lieu of or in exchange for any Securities cancelled as provided in this Section, except as expressly permitted by this Indenture. All cancelled Securities held by the Trustee shall be disposed of in the customary manner by the Trustee unless otherwise directed by a Company Order.

SECTION 3.10. Computation of Interest.

Interest on the Securities shall be computed on the basis of a 360-day year comprised of twelve 30-day months.

SECTION 3.11. CUSIP and CINS Numbers.

The Company in issuing the Securities may use "CUSIP" and "CINS" numbers (if then generally in use), and, if so, the Trustee shall use the CUSIP or CINS numbers in notices of redemption or repurchase as a convenience to Holders; provided that any such notice may state that no representation is made as to the correctness of such numbers either as printed on the Securities or as contained in any notice of a redemption or repurchase and that reliance may be placed only on the other identification numbers printed on the Securities, and any such redemption or repurchase shall not be affected by any defect in or omission of such numbers.

SECTION 3.12. Deposits of Monies.

Except to the extent payment of interest is made by the Company's check or wire transfer pursuant to Section 3.1, prior to 11:00 a.m. New York City time on each Interest Payment Date, Redemption Date, Stated Maturity, and Purchase Date, the Company shall deposit with the Paying Agent in immediately available funds money sufficient to make cash payments, if any, due on such Interest Payment Date, Redemption Date, Stated Maturity and Purchase Date, as the case may be, in a timely manner which permits the Paying Agent to remit payment to the Holders on such Interest Payment Date, Redemption Date, Stated Maturity, and Purchase Date, as the case may be.

SECTION 3.13. Book-Entry Provisions for Global Securities.

(a) The Global Securities initially shall (i) be registered in the name of the Depository or the nominee of such Depository, (ii) be delivered to the Trustee as custodian for such Depository and (iii) bear legends as set forth in Exhibit B hereto.

Members of, or participants in, the Depository ("Agent Members") shall have no rights under this Indenture with respect to any Global Security held on their behalf by the Depository, or the Trustee as its custodian, or under any Global Security, and the Depository may be treated by the Company, the Trustee and any agent of the Company or the Trustee as the absolute owner of the Global Securities for all purposes whatsoever. Notwithstanding the foregoing, nothing herein shall prevent the Company, the Trustee or any agent of the Company or the Trustee from giving effect to any written certification, proxy or other authorization furnished by the Depository or impair, as between the Depository and its Agent Members, the operation of customary practices governing the exercise of the rights of a Holder of any Security.

(b) Transfer of Global Securities shall be limited to transfers in whole, but not in part, to the Depository, its successors or their respective nominees. Interests of beneficial owners in the Global Securities may not be transferred or exchanged for physical securities, except that physical securities shall be transferred to all beneficial owners in exchange for their beneficial interests in Global Securities if (i) the Depository notifies the Company that it is unwilling or unable to continue as Depository for any Global Security, or that it will cease to be a "Clearing Agency" under the Exchange Act, and in either case a successor Depository is not appointed by the Company within 90 days of such notice, (ii) an Event of Default has occurred and is continuing and the Security Registrar has received a written request from the Depository to issue physical securities or (iii) the Company has notified the Trustee in writing that it elects to cause the issuance of Securities as physical securities.

(c) The Holder of any Global Security may grant proxies and otherwise authorize any Person, including Agent Members and Persons that may hold interests through Agent Members, to take any action which a Holder is entitled to take under this Indenture or the Securities. (d) If any beneficial interest in a Global Security is exchanged for or transferred to a Person who will take delivery thereof in the form of a beneficial interest in another Global Security or for a physical Security, the principal amount of Securities represented by such Global Security shall be reduced accordingly on the records of the Trustee, as custodian for the Depository; and if the beneficial interest is being exchanged for or transferred to a Person who will take delivery thereof in the form of a beneficial interest in another Global Security, such other Global Security shall be increased accordingly on the records of the Trustee, as custodian for the Depository.

SECTION 3.14. Transfer and Exchange.

(a) The following provisions shall apply with respect to any proposed transfer of a 144A Security or an IAI Security or a beneficial interest therein prior to the date which is two years after the later of the date of its original issue and the last date on which the Company or any affiliate of the Company was the owner of such Securities (or any predecessor thereto) (the "Resale Restriction Termination Date"):

(i) a transfer of a 144A Security or an IAI Security or a beneficial interest therein to a QIB shall be made upon the representation of the transferee in the form as set forth on the reverse of the Security that it is purchasing for its own account or an account with respect to which it exercises sole investment discretion and that it and any such account is a "qualified institutional buyer" within the meaning of Rule 144A, and is aware that the sale to it is being made in reliance on Rule 144A and acknowledges that it has received such information regarding the Company as it has requested pursuant to Rule 144A or has determined not to request such information and that it is aware that the transferor is relying upon its foregoing representations in order to claim the exemption from registration provided by Rule 144A;

(ii) a transfer of a 144A Security or an IAI Security or a beneficial interest therein to an IAI shall be made upon receipt by the Trustee or its agent of a certificate substantially in the form set forth in Exhibit E from the proposed transferee and, if requested by the Company or the Trustee, the delivery of an opinion of counsel, certification and/or other information satisfactory to each of them; and

(iii) a transfer of a 144A Security or an IAI Security or a beneficial interest therein to a Non-U.S. Person shall be made upon receipt by the Trustee or its agent of a certificate substantially in the form set forth in Exhibit C from the proposed transferor and, if requested by the Company or the Trustee, the delivery of an opinion of counsel, certification and/or other information satisfactory to each of them.

(b) The following provisions shall apply with respect to any proposed transfer of a Regulation S Security or a beneficial interest therein prior to the expiration of the Restricted Period:

(i) a transfer of a Regulation S Security or a beneficial interest therein to a QIB shall be made upon the representation of the transferee, in the form of assignment on the reverse of the certificate, that it is purchasing the Security for its own account or an account with respect to which it exercises sole investment discretion and that it and any such account is a "qualified institutional buyer" within the meaning of Rule 144A, and is aware that the sale to it is being made in reliance on Rule 144A and acknowledges that it has received such information regarding the Company as the undersigned has requested pursuant to Rule 144A or has determined not to request such information and that it is aware that the transferor is relying upon its foregoing representations in order to claim the exemption from registration provided by Rule 144A;

(ii) a transfer of a Regulation S Security or a beneficial interest therein to an Institutional Accredited Investor shall be made upon receipt by the Trustee or its agent of a certificate substantially in the form set forth in Exhibit E from the proposed transferee and, if requested by the Company or the Trustee, the delivery of an opinion of counsel, certification and/or other information satisfactory to each of them; and

(iii) a transfer of a Regulation S Security or a beneficial interest therein to a Non-U.S. Person shall be made upon receipt by the Trustee or its agent of a certificate substantially in the form set forth in Exhibit C hereof from the proposed transferor and, if requested by the Company or the Trustee, receipt by the Trustee or its agent of an opinion of counsel, certification and/or other information satisfactory to each of them.

After the expiration of the Restricted Period, beneficial interests in the Regulation S Security may be transferred without requiring certification set forth in Exhibit C or Exhibit E or any additional certification.

(c) Restricted Securities Legend. Upon the transfer, exchange or replacement of Securities not bearing the Private Placement Legend, the Security Registrar shall deliver Securities that do not bear the Private Placement Legend. Upon the transfer, exchange or replacement of Securities bearing the Private Placement Legend, the Security Registrar shall deliver only Securities that bear the Private Placement Legend unless (i) there is delivered to the Security Registrar an Opinion of Counsel to the effect that neither such legend nor the related restrictions on transfer are required in order to maintain compliance with the provisions of the Securities Act, (ii) such transfer, exchange or replacement occurs after the Resale Restriction Termination Date or (iii) such Security has been sold pursuant to an effective registration statement under the Securities Act.

(d) Other Transfers. If a Holder proposes to transfer a Security constituting a Restricted Security pursuant to any exemption from the registration requirements of the Securities Act other than as provided for by Section 3.14(a) and (b), the Security Registrar shall only register such transfer or exchange if such transferor delivers an Opinion of Counsel satisfactory to the Company and the Security Registrar that such transfer is in compliance with the Securities Act and the terms of this Indenture.

(e) General. By its acceptance of any Security bearing the Private Placement Legend and by its ownership of a beneficial interest therein, each Holder of such a Security and each owner of a beneficial interest therein acknowledges the restrictions on transfer of such Security and of beneficial interests therein set forth in this Indenture and in the Private Placement Legend and agrees that it will transfer such Security and beneficial interests therein only as provided in this Indenture.

The Security Registrar shall retain copies of all letters, notices and other written communications received pursuant to Section 3.13 or this Section 3.14. The Company shall have the right to inspect and make copies of all such letters, notices or other written communications at any reasonable time upon the giving of reasonable prior written notice to the Security Registrar.

ARTICLE IV

SATISFACTION AND DISCHARGE

SECTION 4.1. Satisfaction and Discharge of Indenture.

This Indenture shall cease to be of further effect (except as to any surviving rights of registration of transfer or exchange of Securities herein expressly provided for), and the Trustee, on demand of and at the expense of the Company, shall execute proper instruments acknowledging satisfaction and discharge of this Indenture, when

(1) either

(A) all Securities theretofore authenticated and delivered (other than (i) Securities which have been destroyed, lost or stolen and which have been replaced or repaid as provided in Section 3.6 and (ii) Securities for whose payment money has theretofore been deposited in trust or segregated and held in trust by the Company and thereafter repaid to the Company or discharged from such trust, as provided in Section 10.3) have been delivered to the Trustee for cancellation; or

(B) all such Securities not theretofore delivered to the Trustee for cancellation (other than Securities which have been destroyed, lost or stolen and which have been replaced or repaid as provided in Section 3.6),

(i) have become due and payable, or

(ii) will become due and payable at their Stated Maturity within one year, or

(iii) are to be called for redemption within one year under arrangements satisfactory to the Trustee for the giving of notice of redemption by the Trustee in the name, and at the expense, of the Company,

and the Company or any Guarantor, in the case of (i), (ii) or (iii) above, has irrevocably deposited or caused to be deposited with the Trustee as trust funds in trust for the purpose an amount sufficient to pay and discharge the entire Indebtedness on such Securities not theretofore delivered to the Trustee for cancellation, for principal of, premium, if any, and interest on the Securities to the date of deposit (in the case of Securities which have become due and payable) or to the Stated Maturity or Redemption Date, as the case may be, together with irrevocable instructions from the Company directing the Trustee to apply such funds to the payment thereof at maturity or redemption, as the case may be;

(2) the Company or the Guarantors have paid or caused to be paid all other sums payable hereunder by the Company or the Guarantors; and

(3) the Company has delivered to the Trustee an Officer's Certificate and an Opinion of Counsel, which, taken together, state that all conditions precedent herein provided for relating to the satisfaction and discharge of this Indenture have been complied with.

Notwithstanding the satisfaction and discharge of this Indenture pursuant to this Article IV, the obligations of the Company to the Trustee under Section 6.7, the obligations of the Company to any Authenticating Agent under Section 6.14 and, if money shall have been deposited with the Trustee

SECTION 4.2. Application of Trust Money.

Subject to the provisions of the last paragraph of Section 10.3, all money deposited with the Trustee pursuant to Section 4.1 shall be held in trust and applied by it, in accordance with the provisions of the Securities and this Indenture, to the payment, either directly or through any Paying Agent (including the Company acting as its own Paying Agent) as the Trustee may determine, to the Persons entitled thereto, of the principal (and premium, if any) and interest for whose payment such money has been deposited with the Trustee.

ARTICLE V

REMEDIES

SECTION 5.1. Events of Default.

"Event of Default," wherever used herein, means any one of the following events (whatever the reason for such Event of Default and whether it shall be voluntary or involuntary or be effected by operation of law or pursuant to any judgment, decree or order of any court or any order, rule or regulation of any administrative or governmental body):

> (1) default in the payment of the principal of or premium, if any, when due and payable, on any of the Securities (at Stated Maturity, upon optional redemption, required purchase or otherwise); or

(2) default in the payment of an installment of interest on any of the Securities, when due and payable, for 30 days; or

(3) default in the performance, or breach, of any covenant or agreement of the Company under this Indenture (other than a default in the performance or breach of a covenant or agreement which is specifically dealt with in clauses (1), (2) or (4)) and such default or breach shall continue for a period of 30 days after written notice has been given, by certified mail, (x) to the Company by the Trustee or (y) to the Company and the Trustee by the holders of at least 25% in aggregate principal amount of the Outstanding Securities; or

(4) (a) there shall be a default in the performance or breach of the provisions of Section 8.1 with respect to the Company; (b) the Company shall have failed to make or consummate an Asset Sale Offer in accordance with the provisions of Section 10.14; or (c) the Company shall have failed to make or consummate a Change of Control Offer in accordance with the provisions of Section 10.13; or

(5) default or defaults under one or more agreements, instruments, mortgages, bonds, debentures or other evidences of Indebtedness under which the Company or any Restricted Subsidiary then has outstanding Indebtedness in excess of \$10,000,000, individually or in the aggregate, and (a) such default or defaults include a failure to make a payment of principal, (b) such Indebtedness is already due and payable in full or (c) such default or defaults have resulted in the acceleration of the maturity of such Indebtedness; provided that if any such default is cured or waived or any such acceleration rescinded, or such Indebtedness is repaid, within a period of 10 days from the continuation of such default beyond the applicable grace period or the

36

(6) one or more judgments, orders or decrees of any court or regulatory or administrative agency of competent jurisdiction for the payment of money in excess of \$10,000,000, either individually or in the aggregate (net of applicable insurance coverage which is acknowledged in writing by the insurer or which has been determined to be applicable by a final nonappealable determination by a court of competent jurisdiction), shall be entered against the Company or any Restricted Subsidiary or any of their respective properties and shall not be discharged and there shall have been a period of 60 days after the date on which any period for appeal has expired and during which a stay of enforcement of such judgment, order or decree, shall not be in effect; or

(7) the entry of a decree or order by a court having jurisdiction in the premises (A) for relief in respect of the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary, in an involuntary case or proceeding under the Federal Bankruptcy Code or any other federal, state or foreign bankruptcy, insolvency, reorganization or similar law or (B) adjudging the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary, bankrupt or insolvent, or approving a petition seeking reorganization, arrangement, adjustment or composition of or in respect of the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary, under the Federal Bankruptcy Code or any other similar federal, state or foreign law, or appointing a custodian, receiver, liquidator, assignee, trustee or sequestrator (or other similar official) of the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary or any Significant of their properties, or ordering the winding up or liquidation of any of their properties, or ordering the winding up or liquidation of any of their affairs, and the continuance of any such decree or order unstayed and in effect for a period of 60 consecutive days; or

(8) the institution by the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary of a voluntary case or proceeding under the Federal Bankruptcy Code or any other similar federal, state or foreign law or any other case or proceedings to be adjudicated a bankrupt or insolvent, or the consent by the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary to the entry of a decree or order for relief in respect of the Company or such Significant Subsidiary or group of Restricted Subsidiaries in any involuntary case or proceeding under the Federal Bankruptcy Code or any other similar federal, state or foreign law or to the institution of bankruptcy or insolvency proceedings against the Company or such Significant Subsidiary or group of Restricted Subsidiaries, or the filing by the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary of a petition or answer or consent seeking reorganization or relief under the Federal Bankruptcy Code or any other similar federal, state or foreign law, or the consent by it to the filing of any such petition or to the appointment of or taking possession by a custodian, receiver, liquidator, assignee, trustee or sequestrator (or other similar official) of any of the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary or of any substantial part of its property, or the making by it of an assignment for the benefit of creditors, or the admission by it in writing of its inability to pay its debts generally as they become due or the taking of corporate action by the Company or any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary in furtherance of any such action; or

(9) any of the Guarantees of any Significant Subsidiary or one or more Restricted Subsidiaries that, taken together, would constitute a Significant Subsidiary ceases to be in full force and effect or any of such Guarantees is declared to be null and void and unenforceable or any of such Guarantees is found to be invalid or any of such Guarantors denies its liability under its Guaranty (other than by reason of release of such Guarantor in accordance with the terms of this Indenture).

SECTION 5.2. Acceleration of Maturity; Rescission and Annulment.

If an Event of Default (other than those covered by clause (7) or (8) of Section 5.1 with respect to the Company or a Significant Subsidiary or one or more Restricted Subsidiaries that taken together would constitute a Significant Subsidiary) shall occur and be continuing, the Trustee, by notice to the Company, or the Holders of at least 25% in aggregate principal amount of the Securities then Outstanding, by notice to the Trustee and the Company, may declare the principal of, premium, if any, and accrued and unpaid interest, if any, on all of the outstanding Securities due and payable immediately, upon which declaration, all amounts payable in respect of the Securities shall be due and payable. If an Event of Default specified in clause (7) or (8) of Section 5.1 with respect to the Company or a Significant Subsidiary or one or more Restricted Subsidiaries that taken together would constitute a Significant Subsidiary occurs and is continuing, then the principal of, premium, if any, and accrued and unpaid interest, if any, on all the outstanding Securities shall be due accrued and unpaid interest, if any, on all the outstanding Securities shall ipso facto become and be immediately due and payable without any declaration or other act on the part of the Trustee or any Holder of Securities.

After a declaration of acceleration under the Indenture, but before a judgment or decree for payment of the money due has been obtained by the Trustee, the Holders of a majority in aggregate principal amount of the Outstanding Securities, by written notice to the Company and the Trustee, may rescind such declaration if

(1) the Company or any Guarantor has paid or deposited with the Trustee a sum sufficient to pay

(A) all sums paid or advanced by the Trustee under this Indenture and the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel,

(B) all overdue interest on all Securities,

(C) the principal of and premium, if any, on any Securities which have become due otherwise than by such declaration of acceleration and interest thereon at the rate borne by the Securities, and

(D) to the extent that payment of such interest is lawful, interest upon overdue interest and overdue principal at the rate set forth in the Securities which has become due otherwise than by such declaration of acceleration;

(2) the rescission would not conflict with any judgment or decree of a court of competent jurisdiction; and

(3) all Events of Default, other than the nonpayment of principal of, premium, if any, and interest on the Securities that have become due solely by such declaration of acceleration, have been cured or waived.

No such rescission shall affect any subsequent default or impair any right consequent thereto.

The Company and each Guarantor covenants that if

(i) default is made in the payment of any interest on any Security when such interest becomes due and payable and such default continues for a period of 30 days, or

(ii) default is made in the payment of the principal of (or premium, if any, on) any Security on the due date for payment thereof, including, with respect to any Security required to have been purchased pursuant to a Change of Control Offer or an Asset Sale Offer made by the Company, at the Purchase Date thereof,

the Company or such Guarantor will, upon demand of the Trustee, pay to it, for the benefit of the Holders of such Securities, the whole amount then due and payable on such Securities for principal (and premium, if any) and interest, and, to the extent that payment of such interest shall be legally enforceable, interest on any overdue principal (and premium, if any) and on any overdue interest, at the rate provided by the Securities, and, in addition thereto, such further amount as shall be sufficient to cover the costs and expenses of collection, including the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel.

In addition to the rights and powers set forth in Section 317(a) of the Trust Indenture Act, the Trustee shall be entitled to file such other papers or documents as may be necessary or advisable in order to have the claims of the Trustee and of the Holders of the Securities allowed in any judicial proceeding relative to the Company, any Guarantor or any other obligor upon the Securities, its creditors, or its property, and to collect and receive any moneys or other property payable or deliverable on any such claims, and to distribute the same after the deduction of its charges and expenses; and any receiver, assignee or trustee in bankruptcy or reorganization is hereby authorized by each of the Holders to make such payments to the Trustee, and, in the event that the Trustee shall consent to the making of such payments directly to the Holders, to pay to the Trustee any amount due it for compensation and expenses, including counsel fees incurred by it up to the date of such distribution.

If an Event of Default occurs and is continuing, the Trustee may in its discretion proceed to protect and enforce its rights and the rights of the Holders by such appropriate judicial proceedings as the Trustee shall deem most effectual to protect and enforce any such rights, whether for the specific enforcement of any covenant or agreement in this Indenture or in aid of the exercise of any power granted herein, or to enforce any other proper remedy.

SECTION 5.4. Trustee May File Proofs of Claim.

In case of any judicial proceeding relative to the Company, a Guarantor (or any other obligor upon the Securities), its property or its creditors, the Trustee shall be entitled and empowered, by intervention in such proceeding or otherwise, to take any and all actions authorized under the Trust Indenture Act in order to have claims of the Holders and the Trustee allowed in any such proceeding. In particular, the Trustee shall be authorized to collect and receive any moneys or other property payable or deliverable on any such claims and to distribute the same; and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Holder to make such payments to the Trustee and, in the event that the Trustee shall consent to the making of such payments directly to the Holders, to pay to the Trustee any amount due it for the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel, and any other amounts due the Trustee under Section 6.7.

No provision of this Indenture shall be deemed to authorize the Trustee to authorize or consent to or accept or adopt on behalf of any Holder any plan of reorganization, arrangement, adjustment or

SECTION 5.5. Trustee May Enforce Claims Without Possession of Securities.

All rights of action and claims under this Indenture or the Securities may be prosecuted and enforced by the Trustee without the possession of any of the Securities or the production thereof in any proceeding relating thereto, and any such proceeding instituted by the Trustee shall be brought in its own name as trustee of an express trust, and any recovery of judgment shall, after provision for the payment of the reasonable compensation, expenses, distributions and advances of the Trustee, its agents and counsel, be for the ratable benefit of the Holders of the Securities in respect of which such judgment has been recovered.

SECTION 5.6. Application of Money Collected.

Any money collected by the Trustee pursuant to this Article shall be applied in the following order, at the date or dates fixed by the Trustee and, in case of the distribution of such money on account of principal (or premium, if any) or interest, upon presentation of the Securities and the notation thereon of the payment if only partially paid and upon surrender thereof if fully paid:

FIRST: To the payment of all amounts due the Trustee under Section 6.7;

SECOND: To the payment of the amounts then due and unpaid for principal of (and premium, if any) and interest on the Securities in respect of which or for the benefit of which such money has been collected, ratably, without preference or priority of any kind, according to the amounts due and payable on such Securities for principal (and premium, if any) and interest, respectively;

THIRD: To the payment of any and all other amounts due under the Indenture, the Securities or the Guarantees; and

 $\ensuremath{\mathsf{FOURTH}}$: To the Company (or such other Person as a court of competent jurisdiction may direct).

SECTION 5.7. Limitation on Suits.

Subject to Section 5.8, no Holder of any Security shall have and right to institute any proceeding, judicial or otherwise, with respect to this Indenture, or for the appointment of a receiver or trustee, or for any other remedy hereunder, unless

(i) such Holder has previously given written notice to the Trustee of a continuing Event of Default;

(ii) the Holders of not less than 25% in principal amount of the Outstanding Securities shall have made written request to the Trustee to institute proceedings in respect of such Event of Default in its own name as Trustee hereunder;

(iii) such Holder or Holders have offered to the Trustee reasonable indemnity against the costs, expenses and liabilities to be incurred in compliance with such request;

(iv) the Trustee for 45 days after its receipt of such notice, request and offer of indemnity has failed to institute any such proceeding; and it being understood and intended that no one or more Holders shall have any right in any manner whatever by virtue of, or by availing of, any provision of this Indenture to affect, disturb or prejudice the rights of any other Holders, or to obtain or to seek to obtain priority or preference over any other Holders or to enforce any right under this Indenture, except in the manner herein provided and for the equal and ratable benefit of all the Holders.

SECTION 5.8. Unconditional Right of Holders to Receive Principal, Premium and Interest.

Notwithstanding any other provision in this Indenture, the Holder of any Security shall have the right, which is absolute and unconditional, to receive payment of the principal of (and premium, if any) and (subject to Section 3.7) interest on such Security on the respective Stated Maturities expressed in such Security (or, in the case of redemption, on the Redemption Date or in the case of a Change of Control Offer or an Asset Sale Offer made by the Company and required to be accepted as to such Security, on the relevant Purchase Date) and to institute suit for the enforcement of any such payment, and such rights shall not be impaired without the consent of such Holder.

SECTION 5.9. Restoration of Rights and Remedies.

If the Trustee or any Holder has instituted any proceeding to enforce any right or remedy under this Indenture and such proceeding has been discontinued or abandoned for any reason, or has been determined adversely to the Trustee or to such Holder, then and in every such case, subject to any determination in such proceeding, the Company, each Guarantor, the Trustee and the Holders shall be restored severally and respectively to their former positions hereunder and thereafter all rights and remedies of the Trustee and the Holders shall continue as though no such proceeding had been instituted, subject to the determination in such proceeding.

SECTION 5.10. Rights and Remedies Cumulative.

Except as otherwise provided with respect to the replacement or payment of mutilated, destroyed, lost or stolen Securities in Section 3.6, no right or remedy herein conferred upon or reserved to the Trustee or to the Holders is intended to be exclusive of any other right or remedy, and every right and remedy shall, to the extent permitted by law, be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other appropriate right or remedy.

SECTION 5.11. Delay or Omission Not Waiver.

No delay or omission of the Trustee or of any Holder of any Security to exercise any right or remedy accruing upon any Event of Default shall impair any such right or remedy or constitute a waiver of any such Event of Default or an acquiescence therein. Every right and remedy given by this Article or by law to the Trustee or to the Holders may be exercised from time to time, and as often as may be deemed expedient, by the Trustee or by the Holders, as the case may be.

SECTION 5.12. Control by Holders.

The Holders of a majority in principal amount of the Outstanding Securities shall have the right to direct the time, method and place of conducting any proceeding for any remedy available to the Trustee or exercising any trust or power conferred on the Trustee, provided that (i) such direction shall not be in conflict with any rule of law or with this Indenture, and

(ii) the Trustee may take any other action deemed proper by the Trustee which is not inconsistent with such direction.

SECTION 5.13. Waiver of Past Defaults.

The Holders of not less than a majority in principal amount of the Outstanding Securities may on behalf of the Holders of all the Securities waive any past default hereunder and its consequences, except a default

> (i) in the payment of the principal of (or premium, if any) or interest on any Security (including any Security which is required to have been purchased pursuant to a Change of Control Offer or an Asset Sale Offer which has been made by the Company), or

(ii) in respect of a covenant or provision hereof which under Article IX cannot be modified or amended without the consent of the Holder of each Outstanding Security affected.

Upon any such waiver, such default shall cease to exist, and any Event of Default arising therefrom shall be deemed to have been cured, for every purpose of this Indenture; but no such waiver shall extend to any subsequent or other default or impair any right consequent thereon.

SECTION 5.14. Undertaking for Costs.

In any suit for the enforcement of any right or remedy under this Indenture, or in any suit against the Trustee for any action taken, suffered or omitted by it as Trustee, a court may require any party litigant in such suit to file an undertaking to pay the costs of such suit, and may assess costs against any such party litigant, in the manner and to the extent provided in the Trust Indenture Act; provided, that neither this Section nor the Trust Indenture Act shall be deemed to authorize any court to require such an undertaking or to make such an assessment in any suit instituted by the Company or a Guarantor, in any suit instituted by the Trustee, in any suit instituted by any Holder or group of Holders, holding in the aggregate more than 10% in principal amount of the Outstanding Securities, or in any suit instituted by any Holder for the enforcement of the payment of the principal of (or premium, if any) or interest on any Security on or after the Stated Maturity expressed in such Security (or, in the case of redemption, on or after the Redemption Date or, in the case of a Change of Control Offer or an Asset Sale Offer, made by the Company and required to be accepted as to such Security, on the applicable Purchase Date, as the case may be).

SECTION 5.15. Waiver of Stay or Extension Laws.

The Company and each Guarantor covenants (to the extent that it may lawfully do so) that it will not at any time insist upon, or plead, or in any manner whatsoever claim or take the benefit or advantage of, any usury, stay or extension law wherever enacted, now or at any time hereafter in force, which may affect the covenants or the performance of this Indenture; and the Company and each Guarantor (to the extent that it may lawfully do so) hereby expressly waives all benefit or advantage of any such law and covenants that it will not hinder, delay or impede the execution of any power herein granted to the Trustee, but will suffer and permit the execution of every such power as though no such law had been enacted.

ARTICLE VI

THE TRUSTEE

SECTION 6.1. Certain Duties and Responsibilities.

(a) Except during the continuance of an Event of Default,

(i) the Trustee undertakes to perform such duties and only such duties as are specifically set forth in this Indenture, and no implied covenants or obligations shall be read into this Indenture against the Trustee; and

(ii) in the absence of bad faith on its part, the Trustee may conclusively rely, as to the truth of the statements and the correctness of the opinions expressed therein, upon certificates or opinions furnished to the Trustee and conforming to the requirements of this Indenture; but in the case of any such certificates or opinions which by the provisions hereof are specifically required to be furnished to the Trustee, the Trustee shall be under a duty to examine the same to determine whether or not they conform to the requirements of this Indenture but need not verify the contents thereof.

(b) In case an Event of Default has occurred and is continuing, the Trustee shall exercise such of the rights and powers vested in it by this Indenture, and use the same degree of care and skill in their exercise, as a prudent Person would exercise or use under the circumstances in the conduct of such Person's own affairs.

(c) No provision of this Indenture shall be construed to relieve the Trustee from liability for its own negligent misconduct, except that no provision of this Indenture shall require the Trustee to expend or risk its own funds or otherwise incur any financial liability in the performance of any of its duties hereunder, or in the exercise of any of its rights or powers under this Indenture, unless the Trustee has received security and indemnity satisfactory to it against any loss, liability or expense. The Trustee shall not be liable for any error of judgment unless it is proved that the Trustee was negligent in the performance of its duties hereunder.

(d) Whether or not therein expressly so provided, every provision of this Indenture relating to the conduct or affecting the liability of or affording protection to the Trustee shall be subject to the provisions of this Section 6.1.

SECTION 6.2. Notice of Defaults.

Within 30 days after the occurrence of any Default, the Trustee shall transmit by mail to all Holders, as their names and addresses appear in the Security Register, notice of such Default hereunder known to the Trustee, unless such Default shall have been cured or waived; provided, however, that, except in the case of a Default in the payment of the principal of, premium, if any, or interest on any Security, the Trustee shall be protected in withholding such notice if and so long as a trust committee of Responsible Officers of the Trustee in good faith determines that the withholding of such notice is in the interest of the Holders.

SECTION 6.3. Certain Rights of Trustee.

Subject to the provisions of Section 6.1:

(a) the Trustee may conclusively rely as to the truth of the statements and correctness of the opinions expressed therein and shall be fully protected in acting or refraining $% \left({\left[{{{\mathbf{x}}_{i}} \right]_{i}} \right)$

(b) any request or direction of the Company mentioned herein shall be sufficiently evidenced by a Company Request or Company Order and any resolution of the Board of Directors of the Company may be sufficiently evidenced by a Board Resolution of the Company;

(c) whenever in the administration of this Indenture the Trustee shall deem it desirable that a matter be proved or established prior to taking, suffering or omitting any action hereunder, the Trustee (unless other evidence be herein specifically prescribed) may, in the absence of bad faith on its part, rely upon an Officer's Certificate;

(d) the Trustee may consult with counsel of its selection and the advice of such counsel or any Opinion of Counsel shall be full and complete authorization and protection in respect of any action taken, suffered or omitted by it hereunder in good faith and in reliance thereon;

(e) the Trustee shall be under no obligation to exercise any of the rights or powers vested in it by this Indenture at the request or direction of any of the Holders pursuant to this Indenture, unless such Holders shall have offered to the Trustee reasonable security or indemnity against the costs, expenses and liabilities which might be incurred by it in compliance with such request or direction;

(f) the Trustee shall not be bound to make any investigation into the facts or matters stated in any resolution, certificate, statement, instrument, opinion, report, notice, request, direction, consent, order, bond, debenture, note, other evidence of indebtedness or other paper or document, but the Trustee, in its discretion, may make such further inquiry or investigation into such facts or matters as it may see fit, and, if the Trustee shall determine to make such further inquiry or investigation, it shall be entitled (subject to reasonable confidentiality arrangements as may be proposed by the Company or any Guarantor) to make reasonable examination (upon prior notice and during regular business hours) of the books, records and premises of the Company or a Guarantor, personally or by agent or attorney;

(g) the Trustee may execute any of the trusts or powers hereunder or perform any duties hereunder either directly or by or through agents or attorneys or custodians or nominees and the Trustee shall not be responsible for the supervision of, or any misconduct or negligence on the part of any agent or attorney appointed with due care by it hereunder;

(h) the Trustee shall not be liable for any action taken, suffered, or omitted to be taken by it in good faith and reasonably believed by it to be authorized or within the discretion or rights or powers conferred upon it by this Indenture; and

(i) in the event that the Trustee is also acting as Authenticating Agent, Paying Agent or Security Registrar hereunder, the rights and protections afforded to the Trustee pursuant to this Article VI shall also be afforded to such Authenticating Agent, Paying Agent and Security Registrar.

SECTION 6.4. Not Responsible for Recitals or Issuance of Securities.

The recitals contained herein and in the Securities, except the Trustee's certificates of authentication, shall be taken as the statements of the Company, and the Trustee or any Authenticating

43

Agent assumes no responsibility for their correctness. The Trustee makes no representations as to the validity or sufficiency of this Indenture or of the Securities. The Trustee shall not be accountable for the use or application by the Company of Securities or the proceeds thereof.

SECTION 6.5. May Hold Securities.

The Trustee, any Authenticating Agent, any Paying Agent, any Security Registrar or any other agent of the Company or any Guarantor, in its individual or any other capacity, may become the owner or pledgee of Securities and, subject to Sections 6.8 and 6.13, may otherwise deal with the Company or a Guarantor with the same rights it would have if it were not Trustee, Authenticating Agent, Paying Agent, Security Registrar or such other agent.

SECTION 6.6. Money Held in Trust.

Money held by the Trustee in trust hereunder need not be segregated from other funds except to the extent required by law. The Trustee shall be under no liability for interest on any money received by it hereunder except as otherwise agreed in writing with the Company.

SECTION 6.7. Compensation and Reimbursement.

The Company agrees

(1) to pay to the Trustee from time to time such reasonable compensation as the Company and the Trustee shall from time to time agree in writing for all services rendered by it hereunder (which compensation shall not be limited by any provision of law in regard to the compensation of a trustee of an express trust);

(2) except as otherwise expressly provided herein, to promptly reimburse the Trustee upon its request for all reasonable expenses, disbursements and advances incurred or made by the Trustee in accordance with any provision of this Indenture (including the reasonable compensation and the expenses and disbursements of its agents and counsel), except any such expense, disbursement or advance as may be attributable to its negligence or bad faith; and

(3) to indemnify the Trustee, its directors, officers, agents and employees for, and to hold them harmless against, any and all loss, damage, claim, liability or expense incurred without negligence or bad faith on its part, including taxes (other than taxes based upon, measured by or determined by the revenue or income of the Trustee), arising out of or in connection with the acceptance or administration of this trust, including the costs and expenses of defending itself against any claim or liability in connection with the exercise or performance of any of its powers or duties hereunder.

The Trustee shall have a lien prior to the Securities as to all property and funds held by it hereunder for any amount owing to it pursuant to this Section 6.7, except with respect to funds held in trust for the benefit of the Holders of particular Securities.

When the Trustee incurs expenses or renders services in connection with an Event of Default specified in Section 5.1(7) or Section 5.1(8), the expenses (including the reasonable charges and expenses of its counsel) and the compensation for the services are intended to constitute expenses of administration under any applicable Federal or State bankruptcy, insolvency or other similar law.

The provisions of this Section shall survive any termination of this Indenture.

If the Trustee has or shall acquire a conflicting interest within the meaning of the Trust Indenture Act, the Trustee shall either eliminate such interest or resign, to the extent and in the manner provided by, and subject to the provisions of, the Trust Indenture Act and this Indenture.

SECTION 6.9. Corporate Trustee Required; Eligibility.

There shall at all times be a Trustee hereunder which shall be a Person that is eligible pursuant to the Trust Indenture Act to act as such and has, or is a wholly-owned subsidiary of a bank holding company that has, a combined capital and surplus of at least \$100,000.000. If such Person publishes reports of condition at least annually, pursuant to law or to the requirements of a Federal or State supervising or examining authority, then for the purposes of this Section and to the extent permitted by the Trust Indenture Act, the combined capital and surplus of such Person shall be deemed to be its combined capital and surplus as set forth in its most recent report of condition so published. If at any time the Trustee shall cease to be eligible in accordance with the provisions of this Section, it shall resign immediately in the manner and with the effect hereinafter specified in this Article. In determining whether the Truste has a conflicting interest as defined in Section 310(b) of the Trust Indenture Act with respect to the Securities, there shall be excluded all securities issued under the 1999 Indenture.

SECTION 6.10. Resignation and Removal; Appointment of Successor.

(a) No resignation or removal of the Trustee and no appointment of a successor Trustee pursuant to this Article shall become effective until the acceptance of appointment by the successor Trustee in accordance with the applicable requirements of Section 6.9.

(b) The Trustee may resign at any time by giving written notice thereof to the Company. If an instrument of acceptance by a successor Trustee in accordance with the applicable requirements of Section 6.9 shall not have been delivered to the Company and the resigning Trustee within 30 days after the giving of such notice of resignation, the resigning Trustee may petition any court of competent jurisdiction for the appointment of a successor Trustee.

(c) The Trustee may be removed at any time by Act of the Holders of a majority in principal amount of the Outstanding Securities, delivered to the Trustee and to the Company.

(d) If at any time:

(i) the Trustee shall fail to comply with Section 6.8 after written request therefor by the Company or by any Holder who has been a bona fide Holder of a Security for at least six months, or

(ii) the Trustee shall cease to be eligible under Section 6.9 and shall fail to resign after written request therefor by the Company, any Guarantor or by any such Holder, or

(iii) the Trustee shall become incapable of acting or shall be adjudged a bankrupt or insolvent or a receiver of the Trustee or of its property shall be appointed or any public officer shall take charge or control of the Trustee or of its property or affairs for the purpose of rehabilitation, conservation or liquidation,

then, in any such case, (i) the Company or any Guarantor, in each case by a Board Resolution, may remove the Trustee, or (ii) subject to Section 5.14, any Holder who has been a bona fide Holder of a

Security for at least six months may, on behalf of himself and all others similarly situated, petition any court of competent jurisdiction for the removal of the Trustee and the appointment of a successor Trustee.

(e) If the Trustee shall resign, be removed or become incapable of acting, or if a vacancy shall occur in the office of Trustee for any cause, the Company, by a Board Resolution, shall promptly appoint a successor Trustee. If, within one year after such resignation, removal or incapability, or the occurrence of such vacancy, a successor Trustee shall be appointed by Act of the Holders of a majority in principal amount of the Outstanding Securities delivered to the Company and the retiring Trustee, the successor Trustee so appointed shall, forthwith upon its acceptance of such appointment in accordance with the applicable requirements of Section 6.11, become the successor Trustee and supersede the successor Trustee appointed by the Company or the Holders and accepted appointment in accordance with the applicable requirements of Section 6.11, any Holder who has been a bona fide Holder of a Security for at least six months may, on behalf of himself and all others similarly situated, petition any court of competent jurisdiction for the appointment of a successor Trustee.

(f) The Company shall give notice of each resignation and each removal of the Trustee and each appointment of a successor Trustee to all Holders in the manner provided in Section 1.6. Each notice shall include the name of the successor Trustee and the address of its Corporate Trust Office.

(g) The resignation or removal of the Trustee pursuant to this Section 6.10 shall not affect the obligation of the Company to indemnify the Trustee pursuant to Section 6.7(3) in connection with the exercise or performance by the Trustee prior to its resignation or removal of any of its powers or duties hereunder.

(h) No Trustee under this Indenture shall be liable for any action or omission of any successor Trustee.

SECTION 6.11. Acceptance of Appointment by Successor.

Every successor Trustee appointed hereunder shall execute, acknowledge and deliver to the Company and to the retiring Trustee an instrument accepting such appointment, and thereupon the resignation or removal of the retiring Trustee shall become effective and successor Trustee, without any further act, deed or conveyance, shall become vested with all the rights, powers, trusts and duties of the retiring Trustee; but, on request of the Company or the successor Trustee, such retiring Trustee shall, upon payment of its charges, execute and deliver an instrument transferring to such successor Trustee all the rights, powers and trusts of the retiring Trustee and shall duly assign, transfer and deliver to such successor Trustee all property and money held by such retiring Trustee hereunder. Upon request of any such successor Trustee, the Company shall execute any and all instruments for more fully and certainly vesting in and confirming to such successor Trustee all such rights, powers and trusts.

No successor Trustee shall accept its appointment unless at the time of such acceptance such successor Trustee shall be qualified and eligible under this Article.

SECTION 6.12. Merger, Conversion, Consolidation or Succession to Business.

Any corporation into which the Trustee may be merged or converted or with which it may be consolidated, or any corporation resulting from any merger, conversion or consolidation to which the Trustee shall be a party, or any corporation succeeding to all or substantially all the corporate trust business of the Trustee, shall be the successor of the Trustee hereunder, provided such corporation shall be otherwise qualified and eligible under this Article, without the execution or filing of any paper or any further act on the part of any of the parties hereto. In case any Securities shall have been authenticated, but not delivered, by the Trustee then in office, any successor by merger, conversion or consolidation to such authenticating Trustee may adopt such authentication and deliver the Securities so authenticated with the same effect as if such successor Trustee had itself authenticated such Securities.

SECTION 6.13. Preferential Collection of Claims Against the Company or a Guarantor.

If and when the Trustee shall be or become a creditor of the Company or a Guarantor (or any other obligor upon the Securities), the Trustee shall be subject to the provisions of the Trust Indenture Act regarding the collection of claims against the Company or such Guarantor (or any such other obligor).

SECTION 6.14. Appointment of Authenticating Agent.

The Trustee may appoint an Authenticating Agent or Agents which shall be authorized to act on behalf of the Trustee to authenticate Securities issued upon original issue and upon exchange, registration of transfer or partial redemption or partial purchase or pursuant to Section 3.6, and Securities so authenticated shall be entitled to the benefits of this Indenture and shall be valid and obligatory for all purposes as if authenticated by the Trustee hereunder. Wherever reference is made in this Indenture to the authentication and delivery of Securities by the Trustee or the Trustee's certificate of authentication, such reference shall be deemed to include authentication and delivery on behalf of the Trustee by an Authenticating Agent and a certificate of authentication executed on behalf of the Trustee by an Authenticating Agent. Each Authenticating Agent shall be acceptable to the Company and shall at all times be a corporation organized and doing business under the laws of the United States of America, any State thereof or the District of Columbia, authorized under such laws to act as Authenticating Agent, having a combined capital and surplus of not less than \$100,000,000 and subject to supervision or examination by Federal or State authority. If such Authenticating Agent publishes reports of condition at least annually, pursuant to law or to the requirements of said supervising or examining authority, then for the purposes of this Section, the combined capital and surplus of such Authenticating Agent shall be deemed to be its combined capital and surplus as set forth in its most recent report of condition so published. If at any time an Authenticating Agent shall cease to be Agent shall resign immediately in the manner and with the effect specified in this Section.

Any corporation into which an Authenticating Agent may be merged or converted or with which it may be consolidated, or any corporation resulting from any merger, conversion or consolidation to which such Authenticating Agent shall be a party, or any corporation succeeding to all or substantially all of the corporate agency or corporate trust business of an Authenticating Agent, shall continue to be an Authenticating Agent, provided such corporation shall be otherwise eligible under this Section, without the execution or filing of any paper or any further act on the part of the Trustee or the Authenticating Agent.

An Authenticating Agent may resign at any time by giving written notice thereof to the Trustee and to the Company. The Trustee may at any time terminate the agency of an Authenticating Agent by giving written notice thereof to such Authenticating Agent and to the Company. Upon receiving such a notice of resignation or upon such a termination, or in case at any time such Authenticating Agent shall cease to be eligible in accordance with the provisions of this Section, the Trustee may appoint a successor Authenticating Agent which shall be acceptable to the Company and shall give notice of such appointment in the manner provided in Section 1.6, to all Holders as their names and addresses appear in the Security Register. Any successor Authenticating Agent upon acceptance of its appointment hereunder shall become vested with all the rights, powers and duties of its predecessor hereunder, with like effect as if originally named as an Authenticating Agent. No successor Authenticating Agent shall be appointed unless eligible under the provisions of this Section. If an appointment is made pursuant to this Section, the Securities may have endorsed thereon, in addition to the Trustee's certificate of authentication, an alternative certificate of authentication in the following form:

 $% \left({{{\rm{This}}}} \right)$ This is one of the Securities described in the within-mentioned Indenture.

Dated:

State Street Bank and Trust Company, As Trustee

By:

As Authenticating Agent

Ву:

Authorized Signatory

ARTICLE VII

HOLDERS' LISTS AND REPORTS BY TRUSTEE AND COMPANY

SECTION 7.1. Company to Furnish Trustee Names and Addresses of Holders.

The Company will furnish or cause to be furnished to the Trustee at such times as the Trustee may reasonably request in writing, within 30 days after the receipt by the Company of any such request, a list of similar form and content as of a date not more than 15 days prior to the time such list is furnished; excluding from any such list names and addresses received by the Trustee in its capacity as Security Registrar.

SECTION 7.2. Preservation of Information; Communications to Holders.

(a) The Trustee shall preserve, in as current a form as is reasonably practicable, the names and addresses of Holders contained in the most recent list furnished to the Trustee as provided in Section 7.1 and the names and addresses of Holders received by the Trustee in its capacity as Security Registrar, if so acting.

(b) The rights of Holders to communicate with other Holders with respect to their rights under this Indenture or under the Securities, and the corresponding rights and duties of the Trustee, shall be as provided by the Trust Indenture Act.

(c) Every Holder of Securities, by receiving and holding the same, agrees with the Company and the Trustee that neither the Company, any Guarantor nor the Trustee nor any agent of either of them shall be held accountable by reason of any disclosure of information as to the names and addresses of Holders made pursuant to the Trust Indenture Act.

SECTION 7.3. Reports by Trustee.

(a) Within 60 days after May 15 of each year commencing May 15, 2002, the Trustee shall transmit to Holders such reports concerning the Trustee and its actions under this Indenture to the extent required pursuant to the Trust Indenture Act at the times and in the manner provided pursuant thereto.

(b) A copy of each such report shall, at the time of such transmission to Holders, be filed by the Trustee with each stock exchange upon which the Securities are listed, with the Commission and with the Company. The Company will promptly notify the Trustee when the Securities are listed on any stock exchange.

SECTION 7.4. Reports by Company.

The Company shall file with the Trustee and the Commission, and transmit to Holders, such information, documents and other reports, and such summaries thereof, as may be required pursuant to the Trust Indenture Act at the times and in the manner provided pursuant to the Trust Indenture Act; provided that any such information, documents or reports required to be filed with the Commission pursuant to Section 13 or 15(d) of the Exchange Act shall be filed with the Trustee within 15 days after the same is so required to be filed with the Commission.

ARTICLE VIII

CONSOLIDATION, MERGER, CONVEYANCE, TRANSFER OR LEASE

SECTION 8.1. Company May Merge, Consolidate, Etc. Only on Certain Terms.

(A) The Company will not, in any transaction or series of transactions, merge or consolidate with or into, or sell, assign, convey, transfer, lease or otherwise dispose of all or substantially all of its properties and assets to, any Person or Persons, and (B) the Company will not permit any of its Restricted Subsidiaries to enter into any such transaction or series of transactions if such transaction or series of transactions of all or substantially ally ally all of its properties and assets to, any Person or Persons, and (B) the Company will not permit any of its Restricted Subsidiaries to enter into any such transaction or series of transactions if such transaction or series of transactions, in the aggregate, would result in a sale, assignment, conveyance, transfer, lease or other disposition of all or substantially all of the properties and assets of the Company or the Company and its Restricted Subsidiaries, on a consolidated basis, to any other Person or Persons, unless, in each of cases (A) and (B), at the time and after giving effect thereto:

(1) either:

(x) if the transaction or transactions is a merger or consolidation, the Company or such Restricted Subsidiary, as the case may be, shall be the surviving Person of such merger or consolidation, or

(y) the Person formed by such consolidation or into which the Company or such Restricted Subsidiary, as the case may be, is merged or to which the properties and assets of the Company or such Restricted Subsidiary, as the case may be, are disposed of (any such surviving Person or transferee Person being the "Surviving Entity") shall be a corporation organized and existing under the laws of the United States of America, any state thereof or the District of Columbia and shall expressly assume by a supplemental indenture executed and delivered to the Trustee, in form satisfactory to the Trustee, all the obligations of (2) immediately after giving effect to such transaction or series of transactions on a pro forma basis (including, without limitation, any Indebtedness incurred or anticipated to be incurred in connection with or in respect of such transaction or series of transactions), no Default or Event of Default shall have occurred and be continuing; and

(3) except in the case of any merger of the Company with any Restricted Subsidiary of the Company or any merger of Guarantors, in each case with no other Person, the Company or the Surviving Entity, as the case may be, after giving effect to such transaction or series of transactions on a pro forma basis on the assumption that the transaction or transactions had occurred on the first day of the period of four fiscal quarters ending immediately prior to the consummation of such transaction or transactions, with the appropriate adjustments with respect to such transaction or transactions being included in such pro forma calculation, could incur \$1.00 of additional Indebtedness (other than Permitted Indebtedness) under Section 10.8 (assuming a market rate of interest with respect to such additional Indebtedness).

In connection with any consolidation, merger, transfer, lease, assignment or other disposition contemplated by the foregoing provisions of this Section 8.1, the Company shall deliver, or cause to be delivered, to the Trustee, in form and substance reasonably satisfactory to the Trustee, an Officer's Certificate and an Opinion of Counsel, each stating that such consolidation, merger, transfer, lease, assignment, or other disposition and the supplemental indenture in respect thereof (required under clause (1)(y) of this Section 8.1) comply with the requirements of this Indenture. Each such Officer's Certificate shall set forth the manner of determination of the ability to incur Indebtedness in accordance with clause (3) of this Section 8.1.

SECTION 8.2. Successor Substituted.

Upon any consolidation or merger, or any sale, assignment, conveyance, transfer, lease or disposition subject to and in accordance with Section 8.1, the successor Person formed by such consolidation or into which the Company or Restricted Subsidiary, as the case may be, is merged or the successor Person to which such sale, assignment, conveyance, transfer, lease or disposition is made shall succeed to, and be substituted for, and may exercise every right and power of the Company under the Securities, this Indenture and/or the Registration Rights Agreement, as applicable, with the same effect as if such successor had been named as the Company in the Securities, this Indenture and/or in the Registration Rights Agreement, as the case may be, and, except in the case of a lease, the Company or such Restricted Subsidiary, as the case may be, shall be automatically and unconditionally released and discharged from its obligations thereunder.

For all purposes of this Indenture and the Securities (including the provisions of this Article VIII and Sections 10.8, 10.9 and 10.12), Subsidiaries of any Surviving Entity shall, upon consummation of such transaction or series of related transactions, become Restricted Subsidiaries unless and until designated Unrestricted Subsidiaries pursuant to and in accordance with Section 10.18 and all Indebtedness, and all Liens on property or assets, of the Company and the Restricted Subsidiaries in existence immediately prior to such transaction or series of related transactions will be deemed to have been incurred upon consummation of such transaction or series of related transactions.

AMENDMENTS; WAIVERS; SUPPLEMENTAL INDENTURES

SECTION 9.1. Amendments, Waivers and Supplemental Indentures Without Consent of Holders.

Without the consent of any Holders, the Company and each Guarantor, when authorized by Board Resolutions, and the Trustee, at any time and from time to time, may together amend, waive or supplement this Indenture, for any of the following purposes:

> (i) to evidence the succession of another Person to the Company or a Guarantor and the assumption by any such successor of the covenants of the Company or such Guarantor herein and in the Securities and to evidence the assumption of obligations under this Indenture and a Guaranty pursuant to Section 10.17; or

> (ii) to add to the covenants of the Company or a Guarantor for the benefit of the Holders, or to surrender any right or power herein conferred upon the Company or a Guarantor; or

> (iii) to secure the Securities pursuant to the requirements of Section 10.12 or otherwise; or

(iv) to comply with any requirements of the Commission in order to effect or maintain the qualification of this Indenture under the Trust Indenture Act; or

(v) to cure any ambiguity, to correct or supplement any provision herein which may be defective or inconsistent with any other provision herein, or to make any other provisions with respect to matters or questions arising under this Indenture which shall not be inconsistent with the provisions of this Indenture,

provided that (a) such amendment, waiver or supplement does not adversely affect the rights of any Holder of Securities and (b) the Company shall have delivered to the Trustee an Opinion of Counsel stating that such action pursuant to clauses (i), (ii), (iii), (iv) or (v) above is permitted by this Indenture. The Trustee shall not be obligated to enter into any such amendment or supplemental indenture that adversely affects its own rights, duties or immunities under this Indenture to therwise.

SECTION 9.2. Modifications, Amendments and Supplemental Indentures with Consent of Holders.

With the consent of the Holders of not less than a majority in principal amount of the Outstanding Securities, by Act of said Holders delivered to the Company and the Trustee, the Company and the Guarantors, when authorized by Board Resolutions, and the Trustee may together modify, amend or supplement this Indenture for the purpose of adding any provisions to or changing in any manner or eliminating any of the provisions of this Indenture or of modifying in any manner the rights of the Holders under this Indenture; provided, however, that no such modification, amendment or supplemental indenture shall, without the consent of the Holder of each Outstanding Security affected thereby,

(i) change the Stated Maturity of the principal of, or any installment of interest on, any Security, or reduce the principal amount of (or the premium, if any, on), or interest on, any Securities or alter the redemption provisions of the Securities,

(ii) change the currency in which any Securities or any premium or the interest thereon is payable,

(iii) reduce the above-stated percentage in principal amount of outstanding Securities that must consent to an amendment or modification of this Indenture or the Securities.

(iv) reduce the specified percentage in aggregate principal amount of outstanding Securities necessary to waive compliance with provisions of the Indenture or to waive defaults under the Indenture,

 (ν) impair the right to institute suit for the enforcement of any payment on or with respect to the Securities or any Guaranty,

(vi) amend, change or modify the obligation of the Company to make and consummate a Change of Control Offer after the occurrence of a Change of Control or make and consummate an Asset Sale Offer with respect to any Asset Sale that has been consummated or modify any of the provisions or definitions with respect thereto,

 (\mbox{vii}) modify or change any provision of this Indenture relating to the Guarantees in a manner adverse to the Holders of the Securities, or

(viii) modify or change any provision of this Indenture affecting the Subordination or ranking of the Securities or any Guaranty in a manner which adversely affects the holders of Securities.

It shall not be necessary for any Act of Holders under this Section to approve the particular form of any proposed amendment or supplemental indenture, but it shall be sufficient if such Act shall approve the substance thereof.

The Trustee shall join with the Company and each Guarantor in the execution of such amended or supplemental indenture unless such amended or supplemental indenture affects the Trustee's own rights, duties or immunities under this Indenture or otherwise, in which case the Trustee may in its discretion but shall not be obligated to enter into such amendment or supplemental indenture.

SECTION 9.3. Execution of Supplemental Indentures.

In executing, or accepting the additional trusts created by, any supplemental indenture permitted by this Article or the modifications thereby of the trusts created by this Indenture, the Trustee shall be entitled to receive, and (subject to Section 6.1) shall be fully protected in relying upon, an Opinion of Counsel stating that the execution of such supplemental indenture is authorized or permitted by this Indenture. The Trustee may, but shall not be obligated to, enter into any such supplemental indenture which affects the Trustee's own rights, duties or immunities under this Indenture or otherwise; provided that the Trustee shall enter into and execute all other supplemental indentures which satisfy all applicable conditions under this Article IX.

SECTION 9.4. Effect of Supplemental Indentures.

Upon the execution of any supplemental indenture under this Article, this Indenture shall be modified in accordance therewith, and such supplemental indenture shall form a part of this Indenture for all purposes; and every Holder of Securities theretofore or thereafter authenticated and delivered hereunder shall be bound thereby.

SECTION 9.5. Conformity with Trust Indenture Act.

Every supplemental indenture executed pursuant to this Article shall conform to the requirements of the Trust Indenture Act as then in effect.

Securities authenticated and delivered after the execution of any supplemental indenture pursuant to this Article may, and shall if required by the Trustee, bear a notation in form approved by the Trustee as to any matter provided for in such supplemental indenture, provided that any failure by the Trustee to make such notation shall not affect the validity of the matter provided for in such supplemental indenture or any Security or Guaranty hereunder. If the Company shall so determine, new Securities so modified as to conform, in the opinion of the Trustee, the Guarantors and the Company, to any such supplemental indenture may be prepared and executed by the Company or Guarantor and authenticated and delivered by the Trustee in exchange for Outstanding Securities.

SECTION 9.7. Waiver of Certain Covenants.

The Company may omit in any particular instance to comply with any covenant or condition set forth in Section 8.1, any covenant or condition provided pursuant to Section 9.1(ii) or any covenant or condition set forth in Sections 10.4 to 10.12 and 10.15 to 10.18, inclusive, if before the time for such compliance the Holders of at least a majority in principal amount of the Outstanding Securities shall, by Act of such Holders, either waive such compliance in such instance or generally waive compliance with such covenant or condition, but no such waiver shall extend to or affect such covenant or condition except to the extent so expressly waived, and, until such waiver shall become effective, the obligations of the Company and the duties of the Trustee in respect of any such covenant or condition shall remain in full force and effect.

SECTION 9.8. No Liability for Certain Persons.

No director, officer, employee, or stockholder of the Company, nor any director, officer or employees of any Guarantor, as such, shall have any liability for any obligations of the Company or any Guarantor under the Securities, the Guarantees or this Indenture based on or by reason of such obligations or their creation. Each Holder by accepting a Security waives and releases all such liability. The foregoing waiver and release is an integral part of the consideration for the issuance of the Securities and the Guarantees.

ARTICLE X

COVENANTS

SECTION 10.1. Payment of Principal, Premium and Interest.

The Company shall duly and punctually pay the principal of (and premium, if any) and interest on the Securities in accordance with the terms of the Securities, this Indenture and the Registration Rights Agreement.

SECTION 10.2. Maintenance of Office or Agency.

The Company shall maintain an office or agency where Securities may be presented or surrendered for payment, where Securities may be surrendered for registration of transfer or exchange and where notices and demands to or upon the Company or any Guarantor in respect of the Securities, the Guarantees and this Indenture may be served. The Company shall give prompt written notice to the Trustee of the location, and any change in the location, of such office or agency. If at any time the Company shall fail to maintain any such required office or agency or shall fail to furnish the Trustee with the address thereof, such presentations, surrenders, notices and demands may be made or served at the 60

Corporate Trust Office of the Trustee, and the Company hereby appoints the Trustee as its agent to receive all such presentations, surrenders, notices and demands. In the event any such notice or demands are so made or served on the Trustee, the Trustee shall promptly forward copies thereof to the Company.

The Company may also from time to time designate one or more other offices or agencies where the Securities may be presented or surrendered for any or all such purposes and may from time to time rescind such designations. Further, if at any time there shall be no such office or agency in the City of New York where the Securities may be presented or surrendered for payment, the Company shall forthwith designate and maintain such an office or agency in the City of New York in order that the Securities shall at all times be payable in the City of New York. The Company shall give prompt written notice to the Trustee of any such designation or rescission and of any change in the location of any such other office or agency.

SECTION 10.3. Money for Security Payments to be Held in Trust.

If the Company shall at any time act as its own Paying Agent, it will, on or before each due date of the principal of (and premium, if any) or interest on any of the Securities, segregate and hold in trust for the benefit of the Persons entitled thereto a sum sufficient to pay the principal (and premium, if any) or interest so becoming due until such sums shall be paid to such Persons or otherwise disposed of as herein provided and will promptly notify the Trustee of its action or failure so to act.

Whenever the Company shall have one or more Paying Agents, the Company will, prior to 11:00 a.m. New York City time on each due date of the principal of (and premium, if any) or interest on any Securities, deposit with a Paying Agent a sum sufficient to pay the principal (and premium, if any) or interest so becoming due, such sum to be held as provided by the Trust Indenture Act, and (unless such Paying Agent is the Trustee) the Company will promptly notify the Trustee of its action or failure so to act.

The Company shall cause each Paying Agent other than the Trustee to execute and deliver to the Trustee an instrument in which such Paying Agent shall agree with the Trustee, subject to the provisions of this Section, that such Paying Agent will: (i) comply with the provisions of the Trust Indenture Act applicable to it as Paying Agent and (ii) during the continuance of any default by the Company (or any other obligor upon the Securities) in the making of any payment in respect of the Securities, upon the written request of the Trustee, forthwith pay to the Trustee all sums held in trust by such Paying Agent as such.

The Company may at any time, for the purpose of obtaining the satisfaction and discharge of this Indenture or for any other purpose, pay, or by Company Order direct any Paying Agent to pay, to the Trustee all sums held in trust by such Paying Agent, such sums to be held by the Trustee upon the same trusts as those upon which such sums were held by such Paying Agent; and, upon such payment by any Paying Agent (other than the Company) to the Trustee, such Paying Agent shall be released from all further liability with respect to such money.

Any money deposited with the Trustee or any Paying Agent, or then held by the Company, in trust for the payment of the principal of (and premium, if any) or interest on any Security and remaining unclaimed for two years after such principal (and premium, if any) or interest has become due and payable shall be paid to the Company on Company Request, or (if then held by the Company) shall be discharged from such trust; and the Holder of such Security shall thereafter, as an unsecured general creditor, look only to the Company for payment thereof, and all liability of the Trustee or such Paying Agent with respect to such trust money, and all liability of the Company as trustee thereof, shall thereupon cease; provided, however, that the Trustee or such Paying Agent, before being required to make any such repayment, may at the expense of the Company cause to be published once, in a newspaper published in the English language, customarily published on each Business Day and of general circulation in The City of New York, notice that such money remains unclaimed and that, after a date specified therein, which shall not be less than 30 days from the date of such publication, any unclaimed balance of such money then remaining will be repaid to the Company.

SECTION 10.4. Existence; Activities.

Subject to Article VIII, the Company shall do or cause to be done all things necessary to preserve and keep in full force and effect its existence, rights (charter and statutory) and material franchises; provided, however, that the Company shall not be required to preserve any such right or franchise if the Board of Directors of the Company in good faith shall determine that the preservation thereof is no longer desirable in the conduct of the business of the Company and that the loss thereof is not disadvantageous in any material respect to the Holders.

SECTION 10.5. Maintenance of Properties.

The Company shall cause all material properties used in the conduct of its business or the business of any Restricted Subsidiary to be maintained and kept in good condition, repair and working order (regular wear and tear excepted), all as in the judgment of the Company may be necessary so that the business carried on in connection therewith may be properly and advantageously conducted at all times; provided, however, that nothing in this Section shall prevent the Company from disposing of any asset (subject to compliance with Section 10.14) or from discontinuing the operation or maintenance of any of such material properties if such discontinuance is, as determined by the Company in good faith, desirable in the conduct of its business or the business of any Restricted Subsidiary and not disadvantageous in any material respect to the Holders.

SECTION 10.6. Payment of Taxes and Other Claims.

The Company shall pay or discharge or cause to be paid or discharged, before the same shall become delinquent, (1) all taxes, assessments and governmental charges levied or imposed upon the Company or any of its Restricted Subsidiaries or upon the income, profits or property of the Company or any of its Restricted Subsidiaries, and (2) all lawful material claims for labor, materials and supplies which, if unpaid, might by law become a lien upon property of the Company or any of its Restricted Subsidiaries; provided, however, that the Company shall not be required to pay or discharge or cause to be paid or discharged any such tax, assessment, charge or claim whose amount, applicability or validity is being contested in good faith by appropriate proceedings.

SECTION 10.7. Maintenance of Insurance.

The Company shall, and shall cause its Restricted Subsidiaries to, keep at all times all of their material properties which are of an insurable nature insured against loss or damage with insurers believed by the Company to be responsible to the extent that property of similar character is usually so insured by corporations similarly situated and owning like properties in accordance with good business practice. The Company shall, and shall cause its Restricted Subsidiaries to, use the proceeds from any such insurance policy to repair, replace or otherwise restore all material properties to which such proceeds relate, provided, however, that the Company shall not be required to repair, replace or otherwise restore any such material property if the Company in good faith determines that such inaction is desirable in the conduct of the business of the Company or any Restricted Subsidiary and not disadvantageous in any material respect to the Holders.

SECTION 10.8. Limitation on Indebtedness.

The Company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create, incur, issue, assume, guarantee or in any manner become directly or indirectly liable, contingently or otherwise (in each case, to "incur"), for the payment of any Indebtedness (including any Acquired Indebtedness) other than Permitted Indebtedness; provided, however, that the Company and any Guarantor will be permitted to incur Indebtedness (including Acquired Indebtedness) if (A) the Consolidated Fixed Charge Coverage Ratio of the Company is at least 2.0 to 1 and (B) no Default or Event of Default would occur or be continuing.

SECTION 10.9. Limitation on Restricted Payments.

The Company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly:

(a) declare or pay any dividend or make any other distribution or payment on or in respect of Capital Stock of the Company or any of its Restricted Subsidiaries or make any payment to the direct or indirect holders (in their capacities as such) of Capital Stock of the Company or any of its Restricted Subsidiaries (other than dividends or distributions payable solely in Capital Stock of the Company (other than Redeemable Capital Stock) or in options, warrants or other rights to purchase Capital Stock of the Company (other than Redeemable Capital Stock)) (other than the declaration or payment of dividends or other distributions to the extent declared or paid to the Company or any Restricted Subsidiary);

(b) purchase, redeem, or otherwise acquire or retire for value any Capital Stock of the Company or any of its Restricted Subsidiaries or any options, warrants, or other rights to purchase any such Capital Stock (other than any securities owned by the Company or a Restricted Subsidiary);

(c) make any principal payment on, or purchase, defease, repurchase, redeem or otherwise acquire or retire for value, prior to any scheduled maturity, scheduled repayment, scheduled sinking fund payment or other Stated Maturity, any Subordinated Indebtedness (other than any such Subordinated Indebtedness owed to the Company or a Restricted Subsidiary); or

(d) make any Investment (other than any Permitted Investment) in any $\ensuremath{\mathsf{Person}}$,

(such payments or Investments described in the preceding clauses (a), (b), (c) and (d) are collectively referred to as "Restricted Payments"), unless, after giving effect to the proposed Restricted Payment (the amount of any such Restricted Payment, if other than cash, shall be the Fair Market Value of the asset(s) proposed to be transferred by the Company or such Restricted Subsidiary, as the case may be, pursuant to such Restricted Payment), (A) no Default or Event of Default shall have occurred and be continuing, (B) after giving pro forma effect to such Restricted Payment, the Company would be able to incur \$1.00 of additional Indebtedness (other than Permitted Indebtedness) pursuant to Section 10.8 and (C) the aggregate amount of all Restricted Payments declared or made from and after January 28, 1999 would not exceed the sum of:

(1) 50% of the aggregate Consolidated Net Income of the Company accrued on a cumulative basis during the period beginning on January 1, 1999 and ending on the last day of the fiscal quarter of the Company ending immediately prior to the date of such proposed Restricted Payment (or, if such aggregate cumulative Consolidated Net Income of the Company for such period shall be a loss, minus 100% of such loss);

(2) the aggregate net cash proceeds received by the Company as capital contributions to the Company after January 28, 1999 and which constitute shareholders' equity of the Company in accordance with GAAP;

(3) the aggregate net cash proceeds received by the Company from the issuance or sale of Capital Stock (excluding Redeemable Capital Stock) of the Company to any Person (other than to a Subsidiary of the Company) after January 28, 1999;

(4) the aggregate net cash proceeds received by the Company from any Person (other than a Subsidiary of the Company) upon the exercise of any options, warrants or rights to purchase shares of Capital Stock (other than Redeemable Capital Stock) of the Company after January 28, 1999;

(5) the aggregate net cash proceeds received after January 28, 1999 by the Company from any Person (other than a Subsidiary of the Company) for debt securities that have been converted into or exchanged for Capital Stock of the Company (other than Redeemable Capital Stock) to the extent such debt securities were originally sold for cash, plus the aggregate amount of cash received by the Company (other than from a Subsidiary of the Company) at the time of such conversion or exchange;

(6) to the extent not otherwise included in the Company's Consolidated Net Income, in the case of the disposition or repayment of any Investment constituting a Restricted Payment after January 28, 1999, an amount equal to the lesser of the return of capital with respect to such Investment and the initial amount of such Investment, in either case, less the cost of the disposition of such Investment; and

(7) so long as the Designation (as defined in Section 10.18) thereof was treated as a Restricted Payment made after January 28, 1999, with respect to any Unrestricted Subsidiary that has been redesignated as a Restricted Subsidiary after January 28, 1999 in accordance with Section 10.18 below, the Fair Market Value of the Company's interest in such Subsidiary at the time of such redesignation, provided that such amount shall not in any case exceed the Designation Amount (as defined in Section 10.18) with respect to such Restricted Subsidiary upon its Designation; and

(8) \$10,000,000.

For purposes of the preceding clause (C)(4), the value of the aggregate net proceeds received by the Company upon the issuance of Capital Stock upon the exercise of options, warrants or rights will be the net cash proceeds received upon the issuance of such options, warrants or rights plus the incremental amount received by the Company upon the exercise thereof.

None of the foregoing provisions will prohibit, so long as, in the case of clause (v) below, there is no Default or Event of Default continuing, (i) the payment of any dividend or distribution within 60 days after the date of its declaration, if at the date of declaration such payment would be permitted by the first paragraph of this covenant; (ii) the redemption, repurchase or other acquisition or retirement of any shares of any class of Capital Stock of the Company in exchange for, or out of the net cash proceeds of a substantially concurrent issue and sale of, other shares of Capital Stock of the Company (other than Redeemable Capital Stock) to any Person (other than to a Subsidiary of the Company); provided, however, that such net cash proceeds are excluded from clause (C) of the first paragraph of this covenant; (iii) any redemption, repurchase or other acquisition or retirement of Subordinated Indebtedness of the Company in exchange for, or out of the net cash proceeds of a substantially concurrent issue and sale of, (1) Capital Stock (other than Redeemable Capital Stock) of the Company to any Person (other than to a Subsidiary of the Company); provided, however, that any such net cash proceeds are excluded from clause (C) of the first paragraph of this covenant; or (2) other Subordinated Indebtedness of the Company which (w) has no scheduled principal payment prior to the 91st day after the Maturity Date, (x) has an Average Life to Stated Maturity of the Securities and (y) is subordinated to the Securities to at least the same extent as the Subordinated Indebtedness so purchased, exchanged, redeemed, acquired or retired; (iv) payments to purchase Capital Stock of the Company from management or employees of the Company or any of its Subsidiaries, or their authorized representatives, upon the death, disability or termination of employment of such employees, in aggregate amounts under this clause (iv) not to exceed \$3,000,000 in any fiscal year of the Company; (v) payments relating to Permitted Founder Stock Repurchases so long as the Consolidated Fixed Charge Coverage Ratio of the Company is at least 3.0 to 1.0; (vi) cash payments in lieu of fractional shares issuable as dividends on preferred securities of the Company or any of its Restricted Subsidiaries, in aggregate amounts under this clause (vi) not to exceed \$20,000 in any fiscal year of the Company; (vii) repurchases of Capital Stock deemed to occur upon exercise of stock options if such Capital Stock represents a portion of the exercise price of such options; and (viii) the payment of the redemption price of rights issued pursuant to any shareholders' rights plan not in excess of \$0.05 per right and not in excess of \$1,000,000 in the aggregate. Any payments made pursuant to clause (i) of this paragraph shall be taken into account in calculating the amount of Restricted Payments made from and after January 28, 1999.

In computing Consolidated Net Income of the Company under clause (C)(1) of the first paragraph of this covenant, (1) the Company shall use audited financial statements for the portions of the relevant period for which audited financial statements are available on the date of determination and unaudited financial statements and other current financial data based on the books and records of the Company for the remaining portion of such period and (2) the Company shall be permitted to rely in good faith on the financial statements and other financial data derived from the books and records of the Company that are available on the date of determination. If the Company makes a Restricted Payment which, at the time of the making of such Restricted Payment would in the good faith determination of the Company be permitted under the requirements of the Indenture, such Restricted Payment shall be deemed to have been made in compliance with the Indenture notwithstanding any subsequent adjustments made in good faith to the Company for any period.

SECTION 10.10. Limitation on Issuances and Sales of Restricted Subsidiary Stock.

The Company (i) will not permit any Restricted Subsidiary to issue any Capital Stock (other than to the Company or a Restricted Subsidiary) and (ii) will not permit any Person (other than the Company and/or one or more Restricted Subsidiaries) to own any Capital Stock of any Restricted Subsidiary; provided, however, that this covenant shall not prohibit (1) the issuance and sale of all, but not less than all, of the issued and outstanding Capital Stock of any Restricted Subsidiary owned by the Company or any of its Restricted Subsidiaries in compliance with the other provisions of this Indenture, or (2) the ownership by directors of directors' qualifying shares or the ownership by foreign nationals of Capital Stock of any Restricted Subsidiary, to the extent mandated by applicable law.

SECTION 10.11. Limitation on Transactions with Affiliates.

The Company shall not, and shall not permit any of its Restricted Subsidiaries to, directly or indirectly, enter into any transaction or series of related transactions (including, without limitation, the sale, transfer, disposition, purchase, exchange or lease of assets, property or services) with, or for the benefit of, any of its Affiliates, except (a) on terms that are no less favorable to the Company or such Restricted Subsidiary, as the case may be, than those which could have been obtained at the time in a comparable transaction or series of related transactions from Persons who are not Affiliates of the Company, (b) with respect to a transaction or series of related transactions involving aggregate payments or value equal to or greater than \$5,000,000, the Company shall have delivered an Officer's Certificate to the Trustee certifying that such transaction or transactions comply with the preceding clause (a) and have been approved by the Board of Directors of the Company, and (c) with respect to a transaction or series of related transactions or value equal to or greater than \$10,000,000, the Officers' Certificate referred to in clause (b) above also includes a certification that such transaction or transactions have been approved by a majority of the Disinterested Members of the Board of Directors of Notwithstanding the foregoing, the restrictions set forth in this covenant shall not apply to (i) transactions between or among the Company and its Restricted Subsidiaries, (ii) customary directors' fees, indemnification and similar arrangements, consulting fees, employee salaries, bonuses or employment agreements, compensation or employee benefit arrangements and incentive arrangements with any officer, director or employee of the Company or any Restricted Subsidiary entered into in the ordinary course of business, (iii) any dividends made in compliance with Section 10.9, (iv) loans and advances to officers, directors and employees of the Company or any Restricted Subsidiary made in the ordinary course of business in an aggregate amount not to exceed \$1,000,000 outstanding at any one time, (v) transactions pursuant to agreements in effect on January 28, 1999, (vi) written agreements entered into or assumed in connection with acquisitions of other businesses with Persons who were not Affiliates prior to such transactions, or (vii) leases of property or equipment entered into in the ordinary course of business on terms that are substantially similar to those which could have been obtained at the time in a comparable transaction with non-Affiliates.

SECTION 10.12. Limitation on Liens.

The Company will not, and will not permit any of its Restricted Subsidiaries to, create, incur, assume or suffer to exist any Liens of any kind securing Indebtedness upon any of its property or assets, or any proceeds therefrom, unless the Securities are equally and ratably secured (except that Liens securing Subordinated Indebtedness shall be expressly subordinate to Liens securing the Securities to the same extent such Subordinated Indebtedness is subordinate to the Securities), except for (a) Liens securing Senior Indebtedness and Guarantor Senior Indebtedness; (b) Liens securing the Securities; (c) Liens securing Indebtedness which is incurred to refinance Indebtedness which has been secured by a Lien (other than a Lien in favor of the Company or a Restricted Subsidiary) permitted under the Indenture and which has been incurred in accordance with the provisions of this Indenture; provided, however, that such Liens do not extend to or cover any property or assets of the Company or any its Restricted Subsidiaries not securing the Indebtedness so refinanced; and (d) Permitted Liens.

SECTION 10.13. Change of Control.

On or before the 30th day after the date of the occurrence of a Change of Control (the "Change of Control Date"), the Company shall make an Offer to Purchase (a "Change of Control Offer") on a Business Day not more than 60 nor less than 30 days following the mailing of such Offer to Purchase, (the "Change of Control Purchase Date") all of the then Outstanding Securities tendered at a purchase price in cash (the "Change of Control Purchase Price") equal to 101% of the principal amount thereof plus accrued and unpaid interest, if any, thereon to the Change of Control Purchase Date. The Company shall be required to purchase all Securities tendered into the Change of Control Offer and not withdrawn.

On the Change of Control Purchase Date, the Company shall (i) accept for payment Securities or portions thereof (not less than \$1,000 principal amount and integral multiples thereof) tendered pursuant to the Change of Control Offer, (ii) deposit with the Paying Agent money, in immediately available funds, sufficient to pay the purchase price of all Securities or portions thereof so tendered and accepted and (iii) deliver to the Trustee the Securities or portions thereof tendered to and accepted for payment by the Company. The Paying Agent shall promptly mail or deliver to the Holders of Securities so accepted payment in an amount equal to the purchase price, and the Trustee shall promptly authenticate and make available for delivery to such The Company shall not be required to make a Change of Control offer upon a Change of Control if a third party makes the Change of Control Offer in the manner, at the times and otherwise in compliance with the requirements applicable to a Change of Control Offer made by the Company and purchases all Securities validly tendered and not withdrawn under such Change of Control Offer.

The Company shall comply with Rule 14e-1 under the Exchange Act and any other securities laws and regulations thereunder, to the extent such laws or regulations are applicable, in the event that a Change of Control occurs and the Company is required to purchase Securities as described above, and any violation of the provisions of the Indenture relating to such Offer to Purchase occurring as a result of such compliance shall not be deemed a Default or an Event of Default.

SECTION 10.14. Disposition of Proceeds of Asset Sales.

The Company will not, and will not permit any of its Restricted Subsidiaries to, make any Asset Sale unless (a) the Company or such Restricted Subsidiary, as the case may be, receives consideration at the time of such Asset Sale at least equal to the Fair Market Value of the shares or assets sold or otherwise disposed of and (b) at least 75% of the consideration in such Asset Sale, plus all other Asset Sales since January 28, 1999 on a cumulative basis, consists of cash or Cash Equivalents; provided, however, that the amount of any Indebtedness (as shown on the most recent balance sheet of the Company or such Restricted Subsidiary) of the Company or such Restricted Subsidiary that is assumed by the transferee of such assets as a result of which the Company and its Restricted Subsidiaries are no longer liable thereon, and any securities, notes or other obligations received by the Company or such Restricted Subsidiary from such transferee that are converted within 60 days into cash or Cash Equivalents (to the extent of the cash or Cash Equivalents received) shall be deemed to be cash for the purposes of this provision. To the extent that the Net Cash Proceeds, or portion thereof, of any Asset Sale are not applied to repay, and permanently reduce the commitments under Senior Indebtedness or Guarantor Senior Indebtedness in accordance with the terms thereof, the Company or such Restricted Subsidiary, as the case may be, may apply the Net Cash Proceeds, or portion thereof, from such Asset Sale, within 360 days of such Asset Sale, to an investment in properties and assets that replace the properties and assets that were the subject of such Asset Sale or in properties and assets that (as determined in good faith by the Board of Directors of the Company or the Restricted Subsidiary, as the case may be) are used or useful in the business of the Company and its Restricted Subsidiaries conducted at such time or in businesses reasonably related thereto or in Capital Stock of a Person, the principal portion of whose assets consist of such property or assets ("Replacement Assets"). Any Net Cash Proceeds from any Asset Sale that has occurred on or after January 28, 1999 that are neither used to repay, and permanently reduce the commitments under, Senior Indebtedness or Guarantor Senior Indebtedness in accordance with the terms thereof nor invested in Replacement Assets within such 360-day period will constitute "Excess Proceeds" subject to disposition as provided below.

When the aggregate amount of Excess Proceeds equals or exceeds \$10,000,000, the Company shall make an Offer to Purchase, from all Holders of the Securities and any then outstanding Pari Passu Indebtedness required to be repurchased or repaid on a permanent basis in connection with an Asset Sale, an aggregate principal amount of Securities and any then outstanding Pari Passu Indebtedness equal to such Excess Proceeds as follows:

> (i) (A) The Company shall make an Offer to Purchase (an "Asset Sale Offer") from all Holders of the Securities the maximum principal amount (expressed as a multiple of \$1,000) of Securities that may be purchased out of an amount (the "Asset Sale Offer Amount") equal to

the product of such Excess Proceeds, multiplied by a fraction, the numerator of which is the outstanding principal amount of the Securities and the denominator of which is the sum of the outstanding principal amount of the Securities and such Pari Passu Indebtedness, if any (subject to proration in the event such amount is less than the aggregate Offered Price (as defined below) of all Securities tendered), and (B) to the extent required by such Pari Passu Indebtedness and provided there is a permanent reduction in the principal amount of such Pari Passu Indebtedness, the Company shall make an Offer to Purchase Pari Passu Indebtedness (a "Pari Passu Offer") in an amount (the "Pari Passu Indebtedness Amount") equal to the excess of the Excess Proceeds over the Asset Sale Offer Amount;

(ii) The offer price for the Securities shall be payable in cash in an amount equal to 100% of the principal amount of the Securities tendered pursuant to an Asset Sale Offer, plus accrued and unpaid interest, if any, to the date such Asset Sale Offer is consummated (the "Offered Price"), in accordance with the procedures set forth in the Indenture. To the extent that the aggregate Offered Price of the Securities tendered pursuant to an Asset Sale Offer is less than the Asset Sale Offer Amount relating thereto or the aggregate amount of the Pari Passu Indebtedness that is purchased or repaid pursuant to the Pari Passu Offer is less than the Pari Passu Indebtedness Amount (such shortfall constituting an "Asset Sale Deficiency"), the Company may use such Asset Sale Deficiency for general corporate purposes, subject to the limitations in Section 10.9, and

(iii) If the aggregate Offered Price of Securities validly tendered and not withdrawn by Holders thereof exceeds the Asset Sale Offer Amount, Securities to be purchased will be selected on a pro rata basis. Upon completion of such Asset Sale Offer and Pari Passu Offer, the amount of Excess Proceeds shall be reset to zero.

On the Asset Sale Offer Purchase Date, the Company shall (i) accept for payment (subject to proration as described in the Offer to Purchase) Securities or portions thereof validly tendered pursuant to the Asset Sale Offer, respectively, (ii) deposit with the Paying Agent money, in immediately available funds, sufficient to pay the purchase price of all such Securities or portions thereof so tendered and accepted and (iii) deliver to the Trustee the Securities so accepted together with an Officer's Certificate setting forth the Securities or portions thereof tendered to and accepted for payment by the Company. The Paying Agent shall promptly mail or deliver to the Holders of Securities so accepted payment in an amount equal to the purchase price, and the Trustee shall promptly authenticate and make available for delivery to such Holders a new Security of like tenor equal in principal amount to any unpurchased portion of the Security surrendered. Any Securities not so accepted shall be promptly mailed or delivered by the Company to the Holder thereof. The Company shall publicly announce the results of the Asset Sale Offer and Pari Passu Offer not later than the third Business Day following the Asset Sale Offer Purchase Date.

Whenever the aggregate amount of Excess Proceeds received by the Company and its Restricted Subsidiaries exceeds \$10,000,000, such Excess Proceeds shall, prior to the purchase of Securities, be set aside by the Company or such Restricted Subsidiary, as the case may be, in a separate account pending (i) deposit with the Paying Agent of the amount required to purchase the Securities tendered in an Asset Sale Offer or (ii) delivery by the Company of the Offered Price to the Holders of the Securities validly tendered and not withdrawn pursuant to an Asset Sale Offer. Such Excess Proceeds may be invested in Cash Equivalents, as directed by the Company, having a maturity date which is not later than the earliest possible date for purchase of Securities pursuant to the Asset Sale Offer. The Company will be entitled to any interest or dividends accrued, earned or paid on such Cash Equivalents.

The Company shall comply with Rule 14e-1 under the Exchange Act and any other securities laws and regulations thereunder, to the extent such laws and regulations are applicable, in the event that an Asset Sale occurs and the Company is required to purchase Securities as described above, and any violation of the provisions of the Indenture relating to such Offer to Purchase occurring as a result of such compliance shall not be deemed a Default or an Event of Default.

SECTION 10.15. Limitation on Dividends and Other Payment Restrictions Affecting Restricted Subsidiaries.

The Company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly, create or otherwise cause or suffer to exist or become effective any consensual encumbrance or restriction on the ability of any Restricted Subsidiary to (a) pay dividends, in cash or otherwise, ability of any Restricted Subsidiary to (a) pay dividends, in cash of otherwise, or make any other distributions on or in respect of its Capital Stock to the Company or any other Restricted Subsidiary, (b) pay any Indebtedness owed to the Company or any other Restricted Subsidiary of the Company, (c) make loans or advances to the Company or any other Restricted Subsidiary of the Company, (d) transfer any of its properties or assets to the Company or any other Restricted Subsidiary of the Company or (e) guarantee any Indebtedness of the Company or any other Restricted Subsidiary of the Company, except for such encumbrances or restrictions existing under or by reason of (i) applicable law or any applicable rule, regulation or order, (ii) customary non-assignment provisions of any contract or any lease governing a leasehold interest of the Company or any Restricted Subsidiary of the Company, (iii) customary restrictions on transfers of property subject to a Lien permitted under this Indenture (including purchase money Liens permitted under this Indenture), (iv) any agreement or other instrument of a Person acquired by the Company or any Restricted Subsidiary in existence at the time of such acquisition (but not created in contemplation thereof), which encumbrance or restriction is not applicable to any Person, or the properties or assets of any Person, other than the Person, or the property or assets of the Person, so acquired, (v) an agreement entered into for the sale or disposition of Capital Stock or assets of a Restricted Subsidiary or an agreement entered into for the sale of specified assets (in either case, so long as such encumbrance or restriction, by its terms, terminates on the earlier of the termination of such agreement or the consummation of such agreement and so long as such restriction applies only to the Capital Stock or assets to be sold), (vi) any agreement in effect on January 28, 1999, including, without limitation, the Indenture, dated as of January 28, 1999 among the Company, State Street Bank and Trust Company and the guarantors named therein (the "1999 Indenture") and the Credit Facility, as defined in the 1999 Indenture, (vii) this Indenture and the Guarantees, and (viii) any agreement that amends, extends, refinances, renews or replaces any agreement described in the foregoing clauses; provided that the terms and conditions of any such agreement are not materially less favorable to the Holders of the Securities with respect to such encumbrances or restrictions than those under or pursuant to the agreement amended, extended, refinanced, renewed or replaced.

SECTION 10.16. Limitation on Issuance of Subordinated Indebtedness.

The Company shall not, and shall not permit any Guarantor to, directly or indirectly, incur any Indebtedness (including Acquired Indebtedness) that is expressly subordinate or junior in right of payment to any Indebtedness of the Company or such Guarantor and senior in right of payment to the Securities or the Guaranty of such Guarantor, as the case may be.

SECTION 10.17. Additional Subsidiary Guarantees.

If the Company or any of its Restricted Subsidiaries acquires, creates or designates another Restricted Subsidiary organized under the laws of the United States or any possession or territory thereof, any State of the United States or the District of Columbia, then such newly acquired, created or designated Restricted Subsidiary shall, within 30 days after the date of its acquisition, creation or designation, whichever is later, execute and deliver to the Trustee a supplemental indenture in substantially the form set forth in Exhibit D, pursuant to which such Restricted Subsidiary shall become a Guarantor of the Securities in the manner set forth in Article XIII; provided, that such Restricted Subsidiary shall not be obligated to become a Guarantor in the manner set forth above if such Restricted Subsidiary is not, either individually or when considered in the aggregate with all other Restricted Subsidiaries that are not Guarantors, a Significant Subsidiary. Thereafter, such Restricted Subsidiary shall be a Guarantor for all purposes of this Indenture. In addition to the foregoing, the Company shall cause any Restricted Subsidiary that is not a Guarantor to become a Guarantor in the manner provided above within 30 days of such time as it becomes, either individually or when considered in the aggregate with all other Restricted Subsidiaries that are not Guarantors, a Significant Subsidiary. Thereafter, such Restricted Subsidiary shall be a Guarantor for all purposes of this Indenture. The Company at its option may also cause any other Restricted Subsidiary to become a Guarantor in the manner provided in this Section.

SECTION 10.18. Limitations on Designation of Unrestricted Subsidiaries.

(a) The Company may designate after the Issue Date any Restricted Subsidiary as an "Unrestricted Subsidiary" under this Indenture (a "Designation") only if:

(i) no Default shall have occurred and be continuing at the time of or after giving effect to such Designation;

(ii) the Company would be permitted to make an Investment (other than a Permitted Investment covered by clause (x) of the definition thereof) at the time of Designation (assuming the effectiveness of such Designation) pursuant to the first paragraph of Section 10.9 in an amount (the "Designation Amount") equal to the Fair Market Value of the Company's interest in such Subsidiary on such date; and

(iii) the Company would be permitted under this Indenture to incur \$1.00 of additional Indebtedness (other than Permitted Indebtedness) pursuant to Section 10.8 at the time of such Designation (assuming the effectiveness of such Designation).

In the event of any such Designation, the Company shall be deemed to have made an Investment constituting a Restricted Payment pursuant to Section 10.9 for all purposes of this Indenture in the Designation Amount.

The Company shall not, and shall not cause or permit any Restricted Subsidiary to, at any time (x) provide credit support for or subject any of its property or assets (other than the Capital Stock of any Unrestricted Subsidiary) to the satisfaction of, any Indebtedness of any Unrestricted Subsidiary (including any undertaking, agreement or instrument evidencing such Indebtedness), (y) be directly or indirectly liable for any Indebtedness of any Unrestricted Subsidiary or (z) be directly or indirectly liable for any Indebtedness which provides that the holder thereof may (upon notice, lapse of time or both) declare a default thereon or cause the payment thereof to be accelerated or payable prior to its final Stated Maturity upon the occurrence of a default with respect to any Indebtedness of any Unrestricted Subsidiary (including any right to take enforcement action against such Unrestricted Subsidiary). All Subsidiaries of Unrestricted Subsidiaries shall automatically be deemed to be Unrestricted Subsidiaries.

(b) The Company may revoke any Designation of a Subsidiary as an Unrestricted Subsidiary (a "Revocation") if:

(i) no Default shall have occurred and be continuing at the time of and after giving effect to such Revocation, and

(ii) all Liens and Indebtedness of such Unrestricted Subsidiary outstanding immediately following such Revocation would, if incurred at such time by a Restricted Subsidiary, have been permitted to be incurred for all purposes of this Indenture.

63

(c) All Designations and Revocations must be evidenced by Board Resolutions of the Company delivered to the Trustee certifying compliance with the foregoing provisions.

SECTION 10.19. Provision of Financial Information.

For so long as the Securities are outstanding, whether or not the Company is subject to Section 13(a) or 15(d) of the Exchange Act, or any successor provision thereto, the Company shall file with the Commission (if permitted by Commission practice and applicable law and regulations) the annual reports, quarterly reports and other documents which the Company would have been required to file with the Commission pursuant to such Section 13(a) or 15(d) or any successor provision thereto if the Company were so subject, such documents to be filed with the Commission on or prior to the respective dates (the "Required Filing Dates") by which the Company would have been required so to file such documents if the Company were so subject. The Company shall also in any event within 15 days after each Required Filing Date (whether or not permitted or required to be filed with the Commission) file with the Trustee, copies of the annual reports, quarterly reports and other documents which the Company would be required to file with the Commission if the Securities were then registered under the Exchange Act and to make such information available to Holders of Securities upon request. In addition, if the Company is not subject to the reporting requirements of the Exchange Act, for so long as any Securities remain outstanding, the Company will furnish to the Holders of Securities and prospective investors, upon their request, the information required to be delivered pursuant to Rule 144A(d)(4) under the Securities Act.

SECTION 10.20. Statement by Officers as to Default; Compliance Certificates.

(a) The Company shall deliver to the Trustee, prior to March 31 in each year commencing March 31, 2002, an Officer's Certificate, stating whether or not to the best knowledge of the signers thereof the Company is in default in the performance and observance of any of the terms, provisions and conditions of this Indenture (without regard to any period of grace or requirement of notice provided hereunder), and if the Company shall be in default, specifying all such defaults and the nature and status thereof of which he may have knowledge.

(b) The Company shall deliver to the Trustee, as soon as possible and in any event within five business days after the Company becomes aware of the occurrence of a Default or an Event of Default, an Officer's Certificate setting forth the details of such Default or Event of Default, and the action which the Company proposes to take with respect thereto.

ARTICLE XI

REDEMPTION OF SECURITIES

SECTION 11.1. Right of Redemption.

The Securities may be redeemed at the election of the Company, in the amounts, at the times, at the Redemption Prices (together with any applicable accrued and unpaid interest to the Redemption Date), and subject to the conditions specified in the form of Security and hereinafter set forth.

SECTION 11.2. Applicability of Article.

Redemption of Securities at the election of the Company, as permitted by this Indenture and the provisions of the Securities, shall be made in accordance with such provisions and this Article. The election of the Company to redeem any Securities pursuant to Section 11.1 shall be evidenced by a Board Resolution. In the event of any redemption at the election of the Company pursuant to Section 11.1, the Company shall notify the Trustee, in case of a redemption of less than all the Securities, at least 60 days, and in the case of a redemption of all the Securities, at least 40 days, prior to the Redemption Date fixed by the Company (in each case, unless a shorter notice shall be satisfactory to the Trustee) of such Redemption Date and of the principal amount of Securities to be redeemed.

SECTION 11.4. Selection by Trustee of Securities to Be Redeemed.

In the event that less than all of the Securities are to be redeemed at any time, selection of such Securities for redemption will be made by the Trustee in compliance with the requirements of the principal national securities exchange, if any, on which the Securities are listed or, if the Securities are not then listed on a national securities exchange, on a pro rata basis, by lot or by such method as the Trustee shall deem fair and appropriate (subject to the rules of the Depository); provided, however, that Securities shall only be redeemable in amounts of \$1,000 or an integral multiple of \$1,000.

The Trustee shall promptly notify the Company and each Security Registrar in writing of the Securities selected for redemption and, in the case of any Securities selected for partial redemption, the principal amount thereof to be redeemed.

For all purposes of this Indenture and of the Securities, unless the context otherwise requires, all provisions relating to the redemption of Securities shall relate, in the case of any Securities redeemed or to be redeemed only in part, to the portion of the principal amount of such Securities which has been or is to be redeemed.

SECTION 11.5. Notice of Redemption.

Notice of redemption shall be given by first class mail, postage prepaid, mailed not less than 30 nor more than 60 days prior to the Redemption Date, to each Holder of Securities to be redeemed, at his address appearing in the Security Register.

All notices of redemption shall identify the Securities to be redeemed (including, if used, CUSIP or CINS numbers) and shall state:

(i) the Redemption Date,

(ii) the Redemption Price,

(iii) if less than all the Outstanding Securities are to be redeemed, the identification (and, in the case of partial redemption, the principal amounts) of the particular Securities to be redeemed,

 $({\rm iv})$ that on the Redemption Date the Redemption Price will become due and payable upon each such Security to be redeemed and that interest thereon will cease to accrue on and after such Redemption Date,

(v) the place or places where such Securities are to be surrendered for payment of the Redemption Price, and

(vi) if the redemption is being made pursuant to the provisions of the Securities regarding a Qualified Equity Offering, a brief description of the transaction or transactions giving rise to such redemption, the nature and amount of Capital Stock sold by the Company thereto in such transaction or transactions, the aggregate purchase price thereof and the net cash proceeds therefrom available for such redemption, the date or dates on which such transaction or transactions were completed and the percentage of the aggregate principal amount of Outstanding Securities being redeemed.

Notice of redemption of Securities to be redeemed at the election of the Company shall be given by the Company or, at the Company's request, by the Trustee in the name and at the expense of the Company and shall be irrevocable.

SECTION 11.6. Deposit of Redemption Price.

72

Prior to any Redemption Date, the Company shall deposit with the Trustee or with a Paying Agent (or, if the Company is acting as its own Paying Agent, segregate and hold in trust as provided in Section 10.3) an amount of money sufficient to pay the Redemption Price of, and (except if the Redemption Date shall be an Interest Payment Date) any applicable accrued interest on, all the Securities which are to be redeemed on that date.

SECTION 11.7. Securities Payable on Redemption Date.

Notice of redemption having been given as aforesaid, the Securities so to be redeemed shall, on the Redemption Date, become due and payable at the Redemption Price therein specified, and from and after such date (unless the Company shall default in the payment of the Redemption Price and any applicable accrued interest) such Securities shall not bear interest. Upon surrender of any such Security for redemption in accordance with said notice, such Security shall be paid by the Company at the Redemption Price, together with any applicable accrued and unpaid interest to the Redemption Date; provided, however, that installments of interest whose Stated Maturity is on or prior to the Redemption Date shall be payable to the Holders of such Securities, or one or more predecessor securities, registered as such at the close of business on the relevant record dates according to their terms and the provisions of Section 3.7.

If any Security called for redemption in accordance with the election of the Company made pursuant to Section 11.1 shall not be so paid upon surrender thereof for redemption, the principal (and premium, if any) shall, until paid, bear interest from the Redemption Date at the rate provided by the Security.

SECTION 11.8. Securities Redeemed in Part.

Any Security which is to be redeemed only in part shall be surrendered at an office or agency of the Company designated for that purpose pursuant to Section 10.2 (with, if the Company or the Trustee so requires, due endorsement by, or a written instrument of transfer in form satisfactory to the Company and the Trustee duly executed by, the Holder thereof or his attorney duly authorized in writing), and the Company shall execute, and the Trustee shall authenticate and deliver to the Holder of such Security without service charge, a new Security or Securities, of any authorized denomination as requested by such Holder, in aggregate principal amount at Stated Maturity equal to and in exchange for the unredeemed portion of the principal amount at Stated Maturity of the Security so surrendered.

DEFEASANCE AND COVENANT DEFEASANCE

SECTION 12.1. Company's Option To Effect Defeasance or Covenant Defeasance.

The Company may elect, at its option at any time, to have Section 12.2 or Section 12.3 applied to the Outstanding Securities (as a whole and not in part) upon compliance with the conditions set forth below in this Article. Any such election shall be evidenced by a Board Resolution.

SECTION 12.2. Defeasance and Discharge.

Upon the Company's exercise of its option to have this Section applied to the Outstanding Securities (as a whole and not in part), the Company shall be deemed to have been discharged from its obligations with respect to such Securities as provided in this Section on and after the date the conditions set forth in Section 12.4 are satisfied (hereinafter called "Defeasance"). For this purpose, such Defeasance means that the Company shall be deemed to have paid and discharged the entire Indebtedness represented by such Securities and to have satisfied all its other obligations under such Securities and this Indenture insofar as such Securities are concerned (and the Trustee, at the expense of the Company, shall execute proper instruments acknowledging the same), subject to the following which shall survive until otherwise terminated or discharged hereunder: (1) the rights of Holders of Outstanding Securities to receive, solely from the trust fund described in Section 12.4 and as more fully set forth in such Section, payments in respect of the principal of, premium, if any, and interest on such Securities when payments are due, (2) the Company's obligations with respect to such Securities under Sections 3.4, 3.5, 3.6, 10.2 and 10.3, (3) the rights, powers, trusts, duties and immunities of the Trustee hereunder and (4) this Article. Subject to compliance with this Article, the Company may exercise its option to have this Section applied to the Outstanding Securities (as a whole and not in part) notwithstanding the prior exercise of its option to have Section 12.3 applied to such Securities.

SECTION 12.3. Covenant Defeasance.

Upon the Company's exercise of its option to have this Section applied to the Outstanding Securities (as a whole and not in part), (i) the Company and the Guarantors shall be released from their obligations under Section 8.1(A)(3), Sections 10.5 through 10.19, inclusive, and any covenant provided pursuant to Section 9.1(ii), (ii) the occurrence of any event specified in Sections 5.1(3) and 5.1(4) (with respect to Section 8.1(A)(3), Shall be deemed not to be or result in an Event of Default, in each case with respect to such Securities as provided in this Section on and after the date the conditions set forth in Section 12.4 are satisfied (hereinafter called "Covenant Defeasance"). For this purpose, such Covenant Defeasance means that, with respect to such Securities, the Company may omit to comply with and shall have no liability in respect of any term, condition or limitation set forth in any such specified Section (to the extent so specified in the case of Section 5.1(3) or 5.1(4), whether directly or indirectly, by reason of any reference elsewhere herein to any such Section or by reason of any reference in any such Section to any other provision herein or in any other document, but the remainder of this Indenture and such Securities shall be unaffected thereby.

SECTION 12.4. Conditions to Defeasance or Covenant Defeasance.

The following shall be the conditions to the application of Section 12.2 or Section 12.3 to the Outstanding Securities:

(1) The Company or any Guarantor shall irrevocably have deposited or caused to be deposited with the Trustee (or another trustee which satisfies the requirements contemplated by Section 6.9 and agrees to comply with the provisions of this Article applicable to it) as trust funds in trust for the purpose of making the following payments, specifically pledged as security for, and dedicated solely to, the benefits of the Holders of such Securities, (A) money in an amount, or (B) U.S. Government Obligations which through the scheduled payment of principal and interest in respect thereof in accordance with their terms will provide, not later than one day before the due date of any payment, money in an amount, or (C) a combination thereof, in each case sufficient, in the opinion of a nationally recognized firm of independent public accountants expressed in a written certification thereof delivered to the Trustee, to pay and discharge, and which shall be applied by the Trustee (or any such other qualifying trustee) to pay and discharge, the principal of, premium, if qualifying trustee, to pay and discharge, the principal of, principal of, principal of, principal of, principal of, any, and any installment of interest on such Securities on the respective Stated Maturities thereof, in accordance with the terms of this Indenture and such Securities. As used herein, "U.S. Government obligation" means (x) any security which is (i) a direct Obligation of the United States of America for the payment of which the full faith and credit of the United States of America is pledged or (ii) an obligation of a Person controlled or supervised by and acting as an agency or instrumentality of the United States of America the payment of which is unconditionally guaranteed as a full faith and credit obligation by the United States of America, which, in either case (i) or (ii), is not callable or redeemable at the option of the issuer thereof, and (y) any depositary receipt issued by a bank (as defined in Section 3(a)(2) of the Securities Act) as custodian with respect to any U.S. Government Obligation which is specified in clause (x) above and held by such bank for the account of the holder of such depositary receipt, or with respect to any specific payment of principal of or interest on any U.S. Government Obligation which is so specified and held, provided (except as required by law) such custodian is not authorized to make any deduction from the amount payable to the holder of such depositary receipt from any amount received by the custodian in respect of the U.S. Government Obligation or the specific payment of principal or interest evidenced by such depositary receipt.

(2) In the event of an election to have Section 12.2 apply to the Outstanding Securities, the Company shall have delivered to the Trustee an Opinion of Counsel stating that (A) the Company has received from, or there has been published by, the Internal Revenue Service a ruling or (B) since the date of this Indenture, there has been a change in the applicable Federal income tax law, in either case to the effect that, and based thereon such opinion shall confirm that, the Holders of such Securities will not recognize gain or loss for Federal income tax purposes as a result of the deposit, Defeasance and discharge to be effected with respect to such Securities and will be subject to Federal income tax on the same amount, in the same manner and at the same times as would be the case if such deposit, Defeasance and discharge were not to occur.

(3) In the event of an election to have Section 12.3 apply to the Outstanding Securities, the Company shall have delivered to the Trustee an Opinion of Counsel to the effect that the Holders of such Securities will not recognize gain or loss for Federal income tax purposes as a result of the deposit and Covenant Defeasance to be effected with respect to such Securities and will be subject to Federal income tax on the same amount, in the same manner and at the same times as would be the case if such deposit and Covenant Defeasance were not to occur.

(4) No Default or Event of Default with respect to the Outstanding Securities shall have occurred and be continuing at the time of such deposit (excluding a Default or Event of Default due to a breach of Section 10.8 which arises solely due to the borrowing of funds entirely and immediately applied to such deposit).

(5) Such Defeasance or Covenant Defeasance shall not cause the Trustee to have a conflicting interest with respect to any securities of the Company or any Guarantor.

(6) Such Defeasance or Covenant Defeasance shall not result in a breach or violation of, or constitute a default under, any other agreement or instrument to which the Company is a party or by which it is bound.

(7) The Company shall have delivered to the Trustee an Opinion of Counsel (which opinion may be subject to customary assumptions and exceptions) to the effect that after the 91st day following the deposit, the trust funds will not be subject to the effect of any applicable bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally.

(8) The Company shall have delivered to the Trustee an Officer's Certificate stating that the deposit was not made by the Company with the intent of preferring the Holders of the Securities over the other creditors of the Company or any Guarantor with the intent of defeating, hindering, delaying or defrauding creditors of the Company or any Guarantor or others.

(9) No event or condition shall exist that would prevent the Company from making payments of the principal of, premium, if any, and interest on the Securities on the date of such deposit or at any time ending on the 91st day after the date of such deposit.

(10) The Company shall have delivered to the Trustee an Officer's Certificate and an Opinion of Counsel, each stating that all conditions precedent under this Indenture to either Defeasance or Covenant Defeasance, as the case may be, have been complied with.

SECTION 12.5. Deposited Money and U.S. Government Obligations To Be Held in Trust; Miscellaneous Provisions.

Subject to the provisions of the last paragraph of Section 10.3, all money and U.S. Government Obligations (including the proceeds thereof) deposited with the Trustee or other qualifying trustee (solely for purposes of this Section and Section 12.6, the Trustee and any such other trustee are referred to collectively as the "Trustee") pursuant to Section 12.4 in respect of the Outstanding Securities shall be held in trust and applied by the Trustee, in accordance with the provisions of such Securities and this Indenture, to the payment, either directly or through any such Paying Agent (including the Company acting as its own Paying Agent) as the Trustee may determine, to the Holders of such Securities, of all sums due and to become due thereon in respect of principal and any premium and interest, but money so held in trust need not be segregated from other funds except to the extent required by law.

The Company shall pay and indemnify the Trustee against any tax, fee or other charge imposed on or assessed against the U.S. Government Obligations deposited pursuant to Section 12.4 or the principal and interest received in respect thereof other than any such tax, fee or other charge which by law is for the account of the Holders of Outstanding Securities.

Anything in this Article to the contrary notwithstanding, the Trustee shall deliver or pay to the Company from time to time upon Company Request any money or U.S. Government Obligations held by it as provided in Section 12.4 which, in the opinion of a nationally recognized firm of independent public accountants expressed in a written certification thereof delivered to the Trustee, are in excess of the amount thereof which would then be required to be deposited to effect the Defeasance or Covenant Defeasance, as the case may be, with respect to the Outstanding Securities.

SECTION 12.6. Reinstatement.

If the Trustee or the Paying Agent is unable to apply any money in accordance with this Article with respect to any Securities by reason of any order or judgment of any court or governmental authority enjoining, restraining, or otherwise prohibiting such application, then the obligations under this Indenture, such Securities and the Guarantees from which the Company and the Guarantors have been discharged or released pursuant to Section 12.2 or 12.3 shall be revived and reinstated as though no deposit had occurred pursuant to this Article with respect to such Securities, until such time as the Trustee or Paying Agent is permitted to apply all money held in trust pursuant to Section 12.5 with respect to such Securities in accordance with this Article; provided, however, that if the Company makes any payment of principal of or any premium or interest on any such Security following such reinstatement of its obligations, the Company shall be subrogated to the rights (if any) of the Holders of such Securities to receive such payment from the money so held in trust.

ARTICLE XIII

GUARANTY

SECTION 13.1. Guaranty.

Each Guarantor hereby unconditionally and irrevocably guarantees on a senior subordinated basis, jointly and severally, to each Holder and to the Trustee and its successors and assigns (a) the full and prompt payment (within applicable grace periods) of principal of and interest on the Securities when due, whether at maturity, by acceleration, by redemption or otherwise, and all other monetary obligations of the Company under this Indenture and the Securities and (b) the full and prompt performance within applicable grace periods of all other obligations of the Company under this Indenture and the Securities (all the foregoing being hereinafter collectively called the "Guaranty Obligations"). Each Guarantor further agrees that the Guaranty Obligations may be extended or renewed, in whole or in part, without notice or further assent from such Guarantor, and that such Guarantor will remain bound under this Article XIII notwithstanding any extension or renewal of any Guaranty Obligation.

To the extent that any Guarantor shall be required to pay any amounts on account of the Securities pursuant to a Guaranty in excess of an amount calculated as the product of (i) the aggregate amount payable by the Guarantors on account of the Securities pursuant to the Guarantees times (ii) the proportion (expressed as a fraction) that such Guarantor's net assets (determined in accordance with GAAP) at the date enforcement of the Guarantees is sought bears to the aggregate net assets (determined in accordance with GAAP) of all Guarantors at such date, then such Guarantor shall be reimbursed by the other Guarantors for the amount of such excess, pro rata, based upon the respective net assets (determined in accordance with GAAP) of such other Guarantors at the date enforcement of the Guarantees is sought. This paragraph is intended only to define the relative rights of Guarantors as among themselves, and nothing set forth in this paragraph is intended to or shall impair the joint and several obligations of the Guarantors under their respective Guarantees.

The Guarantors shall have the right to seek contribution from any non-paying Guarantor so long as the exercise of such right does not impair the rights of the Holders under any Guaranty.

Each Guarantor waives presentation to, demand of payment from and protest to the Company of any of the Guaranty Obligations and also waives notice of protest for nonpayment. Each Guarantor waives notice of any default under the Securities or the Guaranty Obligations. The obligations of each Guarantor hereunder shall not be affected by (a) the failure of any Holder or the Trustee to assert any claim or demand or to enforce any right or remedy against the Company or any other Person under this Indenture, the Securities or any other agreement or otherwise; (b) any extension or renewal of any thereof; (c) any rescission, waiver, amendment or modification of any of the terms or provisions of this Indenture, the Securities or any other agreement; (d) the release of any security held by any Holder or the Trustee for the Guaranty Obligations or any of them; (e) the failure of any Holder or Trustee to exercise any right or remedy against any other guarantor of the Guaranty Obligations; or (f) any change in the ownership of any Guarantor (subject to Section 13.5(b)). Each Guarantor further agrees that its Guaranty herein constitutes a guaranty of payment, performance and compliance when due (and not a guaranty of collection) and waives any right to require that any resort be had by any Holder or the Trustee to any security held for payment of the Guaranty Obligations.

To the fullest extent permitted by law, the obligations of each Guarantor hereunder shall not be subject to any reduction, limitation, impairment or termination for any reason, including any claim of waiver, release, surrender, alteration or compromise, and shall not be subject to any defense of setoff, counterclaim, recoupment or termination whatsoever or by reason of the invalidity, illegality or unenforceability of the Guaranty Obligations or otherwise. Without limiting the generality of the foregoing, to the fullest extent permitted by law, the obligations of each Guarantor herein shall not be discharged or impaired or otherwise affected by the failure of any Holder or the Trustee to assert any claim or demand or to enforce any remedy under this Indenture, the Securities or any other agreement, by any waiver or modification of any thereof, by any default, failure or delay, willful or otherwise, in the performance of the Guaranty Obligations, or by any other act or thing or omission or delay to do any other act or thing which may or might in any manner or to any extent vary the risk of such Guarantor or would otherwise operate as a discharge of each Guarantor as a matter of law or equity.

Each Guarantor further agrees that its Guaranty herein shall continue to be effective or be reinstated, as the case may be, if at any time payment, or any part thereof, of principal of or interest on any Guaranty Obligation is rescinded or must otherwise be restored by any Holder or the Trustee upon the bankruptcy or reorganization of the Company or otherwise.

In furtherance of the foregoing and not in limitation of any other right which any Holder or the Trustee has at law or in equity against each Guarantor by virtue hereof, upon the failure of the Company to pay the principal of or interest on any Guaranty Obligation when and as the same shall become due, whether at maturity, by acceleration, by redemption or otherwise (within applicable grace periods), or to perform or comply with any other Guaranty Obligation (within applicable grace periods), each Guarantor hereby promises to and shall, upon receipt of written demand by the Trustee, forthwith pay, or cause to be paid, in cash, to the Holders or the Trustee an amount equal to the sum of (i) the unpaid principal amount of such Guaranty Obligations (ii) accrued and unpaid interest on such Guaranty Obligations (but only to the extent not prohibited by law) and (iii) all other monetary Guaranty Obligations of the Company to the Holders and the Trustee.

Each Guarantor agrees that it shall not be entitled to any right of subrogation in relation to the Holders in respect of any Guaranty Obligations guarantied hereby until payment in full of all Guaranty Obligations. Each Guarantor further agrees that, as between the Guarantors, on the one hand, and the Holders and the Trustee, on the other hand, (x) the maturity of the Guaranty Obligations guarantied hereby may be accelerated as provided in Article V for the purposes of its Guaranty herein, notwithstanding any stay, injunction or other prohibition preventing such acceleration in respect of the Guaranty Obligations guaranteed hereby, and (y) in the event of any declaration of acceleration of such Guaranty Obligations as provided in Article V, such Guaranty Obligations (whether or not due and payable) shall forthwith become due and payable by each Guarantor for the purposes of this Section.

Each Guarantor also agrees to pay any and all costs and expenses (including reasonable attorneys' fees and expenses) incurred by the Trustee or any Holder in enforcing any rights under this Section.

SECTION 13.2. Limitation on Liability.

Any term or provision of this Indenture to the contrary notwithstanding, the obligations of each Guarantor hereunder are limited to the maximum amount which, after giving effect to all other contingent and fixed liabilities of such Guarantor and after giving effect to any collections from or payments made by or on behalf of any other Guarantor in respect of the obligations of such other Guarantor under its Guaranty or pursuant to its contribution obligations under this Indenture, will result in the obligations of such Guarantor under this Indenture or its Guaranty not being voidable as a fraudulent conveyance or fraudulent transfer under federal or state law.

SECTION 13.3. Execution and Delivery of Guarantees; Supplemental Indentures.

The Guaranty of each Guarantor shall be conclusively evidenced by the execution and delivery by such Guarantor of this Indenture. The execution and delivery of any Restricted Subsidiary of the Company that is required to become a Guarantor pursuant to Section 10.17 shall be conclusively evidenced by its execution and delivery of the supplemental indenture referred to therein.

Each Guarantor hereby agrees that its Guaranty shall be and shall remain in full force and effect notwithstanding any failure to endorse on each Security a notation of such Guaranty.

The delivery of any Security by the Trustee, after the authentication thereof hereunder, shall constitute due delivery of the Guarantees on behalf of the Guarantors.

SECTION 13.4. Guarantors May Consolidate, Etc., on Certain Terms.

(a) Except as may be provided in Articles VIII and X, nothing contained in this Indenture or in any of the Securities shall prevent any consolidation or merger of a Guarantor with or into the Company or another Guarantor or shall prevent any sale or conveyance of the assets of a Guarantor as an entirety or substantially as an entirety or the Capital Stock of a Guarantor to the Company or another Guarantor.

(b) No Guarantor shall consolidate with or merge with or into (whether or not such Guarantor is the surviving Person), another Person (other than the Company or another Guarantor), whether or not affiliated with such Guarantor, unless, (i) subject to Section 13.5, the Person formed by or surviving any such consolidation or merger (if other than such Guarantor) assumes all the obligations of such Guarantor pursuant to a supplemental indenture, substantially in the form of Exhibit D hereto, under the Securities and this Indenture; and (ii) immediately after giving effect to such transaction, no Default or Event of Default exists. In the case of any such consolidation or merger and upon the assumption by the successor Person by supplemental indenture of the applicable Guaranty, such successor Person shall succeed to and be substituted for the Guarantor with the same effect as if it had been named herein as a Guarantor.

SECTION 13.5. Release of Guarantors.

(a) Concurrently with any consolidation or merger of a Guarantor or any sale or conveyance of all or substantially all of the assets of a Guarantor, in each case as permitted by the other provisions of this Indenture (including, without limitation, Sections 10.9, 10.11 and 10.14) and as a result of which such Guarantor ceases to be a Restricted Subsidiary of the Company, upon delivery by the Company to the Trustee of an Officer's Certificate to the effect that such consolidation, merger, sale or conveyance was made in accordance with this Indenture and an Opinion of Counsel to the effect that such transaction is permitted by this Indenture (which opinion may be subject to customary assumptions and limitations), the Trustee shall execute any documents reasonably required in order to evidence the release of such Guarantor from its obligations under its Guaranty under this Article XIII. Any Guarantor not released from its obligations under its Guaranty under this Article XIII shall remain liable for the full amount of principal of and premium, if any, and interest on the Securities and for the other obligations of a Guarantor under its Guaranty under this Article XIII.

(b) In the event that the Company designates a Guarantor to be an Unrestricted Subsidiary in accordance with the terms of this Indenture (including Section 10.18), then such Guarantor shall be released and relieved of any obligations under its Guaranty.

SECTION 13.6. Successors and Assigns.

This Article XIII shall be binding upon each Guarantor and its successors and assigns and shall inure to the benefit of the successors and assigns of the Trustee and the Holders and, in the event of any transfer or assignment of rights by any Holder or the Trustee, the rights and privileges conferred upon that party in this Indenture and in the Securities shall automatically extend to and be vested in such transferee or assignee, all subject to the terms and conditions of this Indenture.

SECTION 13.7. No Waiver, etc.

Neither a failure nor a delay on the part of either the Trustee or the Holders in exercising any right, power or privilege under this Article XIII shall operate as a waiver thereof, nor shall a single or partial exercise thereof preclude any other or further exercise of any right, power or privilege. The rights, remedies and benefits of the Trustee and the Holders herein expressly specified are cumulative and not exclusive of any other rights, remedies or benefits which either may have under this Article XIII at law, in equity, by statute or otherwise.

SECTION 13.8. Modification, etc.

No modification, amendment or waiver of any provision of this Article, nor the consent to any departure by a Guarantor therefrom, shall in any event be effective unless the same shall be in writing and signed by the Trustee, and then such waiver or consent shall be effective only in the specific instance and for the purpose for which given. No notice to or demand on a Guarantor in any case shall entitle such Guarantor or any other guarantor to any other or further notice or demand in the same, similar or other circumstances.

SECTION 13.9. Subordination of Guarantees.

The obligations of each Guarantor pursuant to its Guaranty and this Article XIII shall be (a) junior and subordinated in right of payment to the prior payment in full in cash of all Guarantor Senior Indebtedness of such Guarantor and (b) senior in right of payment to all existing and future Guarantor Subordinated Indebtedness of such Guarantor, in each case on the same basis as the Securities and the obligations of the Company hereunder are junior and subordinated to all Senior Indebtedness and senior in right of payment to all Subordinated Indebtedness. For the purposes of this Section 13.9, Article XIV shall apply to the obligations of each Guarantor under its Guaranty, this Article XIII and the other provisions of this Indenture as if references therein to the Company, the Securities, Senior Indebtedness, Subordinated Indebtedness and Designated Senior Indebtedness were references to such Guarantor, such Guarantor's Guaranty, Guarantor Senior Indebtedness, respectively.

ARTICLE XIV

SUBORDINATION

SECTION 14.1. Securities Subordinate to Senior Indebtedness and Senior to Subordinated Indebtedness.

The Company covenants and agrees, and each Holder of a Security, by his acceptance thereof, likewise covenants and agrees that, to the extent and in the manner hereinafter set forth in this

SECTION 14.2. Payment Over of Proceeds Upon Dissolution, Etc.

In the event of any insolvency or bankruptcy case or proceeding, or any receivership, liquidation, reorganization or other similar case or proceeding in connection therewith, relating to the Company or its assets, or any liquidation, dissolution or other winding-up of the Company, whether voluntary or involuntary, or any assignment for the benefit of creditors or other marshaling of assets or liabilities of the Company, all Senior Indebtedness (including, in the case of Designated Senior Indebtedness, any interest accruing subsequent to the filing of a petition for bankruptcy (regardless of whether such interest is an allowed claim in the bankruptcy proceeding)) must be paid in full in cash before any direct or indirect payment (other than payments from a trust previously created pursuant to Article XII) by or on behalf of the Company of any kind or character (excluding Permitted Junior Securities) may be made on account of the principal of, premium, if any, or interest on, or the purchase, redemption or other acquisition of, the Securities.

SECTION 14.3. No Payment When Designated Senior Indebtedness in Default.

During the continuance of any default in the payment of principal, or premium, if any, or interest on any Senior Indebtedness, when the same becomes due, and after receipt by the Trustee and the Company from representatives of holders of such Senior Indebtedness of written notice of such default, no direct or indirect payment (other than payments from trusts previously created pursuant to Article XII) by or on behalf of the Company of any kind or character (other than Permitted Junior Securities) may be made on account of the principal of, premium, if any, or interest on, or the purchase, redemption or other acquisition of, the Securities unless and until such default has been cured or waived or has ceased to exist or such Senior Indebtedness shall have been paid in full in cash or indefeasibly discharged, after which the Company shall promptly resume making any and all required payments in respect of the Securities, including any missed payments.

In addition, during the continuance of any other default with respect to any Designated Senior Indebtedness that permits, or would permit with the passage of time or the giving of notice or both, the maturity thereof to be accelerated (a "Non-payment Default") and upon the earlier to occur of (a) receipt by the Trustee and the Company from the representatives of holders of such Designated Senior Indebtedness of a written notice of such Non-payment Default or (b) if such Non-payment Default results from the acceleration of the maturity of the Securities, the date of such acceleration, no payment (other than payments from trusts previously created pursuant to Article XII) of any kind or character (excluding Permitted Junior Securities) may be made by the Company on account of the principal of, premium, if any, or interest on, or the purchase, redemption, or other acquisition of, the Securities for the period specified below (the "Payment Blockage Period").

The Payment Blockage Period shall commence upon the receipt of notice of a Non-payment Default by the Trustee and the Company from the representatives of holders of Designated Senior Indebtedness or the date of the acceleration referred to in clause (b) of the preceding paragraph, as the case may be, and shall end on the earliest to occur of the following events: (i) 179 days have elapsed since the receipt of such notice or the date of the acceleration referred to in clause (b) of the preceding paragraph (provided the maturity of such Designated Senior Indebtedness shall not theretofore have been accelerated), (ii) such default is cured or waived or ceases to exist or such Designated Senior Indebtedness is paid in full in cash or indefeasibly discharged, or (iii) such Payment Blockage Period shall have been terminated by written notice to the Company or the Trustee from the representatives of holders of Designated Senior Indebtedness initiating such Payment Blockage Period, after which the Company shall promptly resume making any and all required payments in respect of the Securities,

SECTION 14.4. Subrogation to Rights of Holders of Senior Indebtedness.

Subject to the payment in full in cash of all Senior Indebtedness, the Holders of the Securities shall be subrogated to the extent of the payments or distributions made to the holders of such Senior Indebtedness pursuant to the provisions of this Article XIV to the rights of the holders of such Senior Indebtedness to receive payments and distributions of cash, property and securities applicable to the Senior Indebtedness until the principal of, premium, if any, and interest on the Securities shall be paid in full. For purposes of such subrogation, no payments or distributions to the holders of the Senior Indebtedness of any cash, property or securities to which the Holders of the Securities or the Trustee would be entitled except for the provisions of this Article XIV, and no payments over pursuant to the provisions of this Article XIV to the holders of Senior Indebtedness by Holders of the Securities or the Trustee, shall, as among the Company, its creditors other than holders of Senior Indebtedness and the Holders of the Securities, be deemed to be a payment or distribution by the Company to or on account of the Senior Indebtedness.

SECTION 14.5. Provisions Solely to Define Relative Rights.

The provisions of this Article XIV are and are intended solely for the purpose of defining the relative rights of the Holders of the Securities on the one hand and the holders of Senior Indebtedness on the other hand. Nothing contained in this Article XIV or elsewhere in this Indenture or in the Securities is intended to or shall (a) impair, as among the Company, its creditors other than holders of Senior Indebtedness and the Holders of the Securities, the obligation of the Company, which is absolute and unconditional, to pay to the Holders of the Securities the principal of, premium, if any and interest on the Securities as and when the same shall become due and payable in accordance with their terms; (b) affect the relative rights against the Company of the Holders of the Securities and creditors of the Company other than the holders of Senior Indebtedness; or (c) prevent the Trustee or the Holder of any Securities from exercising all remedies otherwise permitted by applicable law upon default under this Indenture, subject to the rights, if any, under this Article XIV of the holders of Senior Indebtedness to receive cash, property and securities otherwise payable or deliverable to the Trustee or such Holder.

SECTION 14.6. Trustee to Effectuate Subordination.

Each Holder of a Security by its acceptance thereof authorizes and directs the Trustee on its behalf to take such action as may be necessary or appropriate to effectuate the subordination provided in this Article XIV and appoints the Trustee its attorney-in-fact for any and all such purposes.

SECTION 14.7. No Waiver of Subordination Provisions.

No right of any present or future holder of any Senior Indebtedness to enforce subordination as herein provided shall at any time in any way be prejudiced or impaired by any act or failure to act on the part of the Company or by any act or failure to act, in good faith, by any such holder, or by any nonWithout in any way limiting the generality of the foregoing paragraph, the holders of Senior Indebtedness may, at any time and from time to time, without the consent of or notice to the Trustee or the Holders of the Securities, without incurring responsibility to the Holders of the Securities and without impairing or releasing the subordination provided in this Article XIV or the obligations hereunder of the Holders of the Securities to the holders of Senior Indebtedness, do any one or more of the following: (i) change the manner, place or terms of payment or extend the time of payment of, or renew or alter, Senior Indebtedness, or otherwise amend or supplement in any manner Senior Indebtedness is outstanding; (ii) sell, exchange, release or otherwise deal with any property pledged, mortgaged or otherwise securing Senior Indebtedness; (iii) release any Person liable in any manner for the collection of Senior Indebtedness; and (iv) exercise or refrain from exercising any rights against the Company and any other Person.

SECTION 14.8. Notice to Trustee.

The Company shall give prompt written notice to the Trustee of any fact known to the Company which would prohibit the making of any payment to or by the Trustee in respect of the Securities. Notwithstanding the provisions of this Article XIV or any other provision of this Indenture, the Trustee shall not be charged with knowledge of the existence of any facts which would prohibit the making of any payment to or by the Trustee in respect of the Securities, unless and until the Trustee shall have received written notice thereof from the Company or a holder of Senior Indebtedness or from any trustee therefor; and, prior to the receipt of any such written notice, the Trustee, subject to the provisions of Section 6.1, shall be entitled in all respects to assume that no such facts exist; provided, however, that if the Trustee shall not have received at its Corporate Trust Office the notice provided for in this Section at least three Business Days prior to the date upon which by the terms hereof any money may become payable for any purpose (including, without limitation, the payment in cash of the principal of, premium, if any or interest on any Security), then, anything herein contained to the contrary notwithstanding, the Trustee shall have full power and authority to receive such money and to apply the same to the purpose for which such money was received and shall not be affected by any notice to the contrary which may be received by it within three Business Days prior to such date.

SECTION 14.9. Reliance on Judicial Order or Certificate of Liquidating Agent.

Upon any payment or distribution of assets of the Company referred to in this Article XIV, the Trustee, subject to the provisions of Section 6.1, and the Holders of the Securities shall be entitled to rely upon any order or decree entered by any court of competent jurisdiction in which such insolvency, bankruptcy, receivership, liquidation, reorganization, dissolution, winding up or similar case or proceeding is pending, or a certificate of the trustee in bankruptcy, receiver, liquidating trustee, custodian, assignee for the benefit of creditors, agent or other Person making such payment or distribution, delivered to the Trustee or to the Holders of Securities, for the purpose of ascertaining the Persons entitled to participate in such payment or distribution, the holders of Senior Indebtedness and other Indebtedness of the Company, the amount thereof or payable thereon, the amount or amounts paid or distributed thereon and all other facts pertinent thereto or to this Article XIV.

SECTION 14.10. Trustee Not Fiduciary for Holders of Senior Indebtedness.

The Trustee shall not be deemed to owe any fiduciary duty to the holders of Senior Indebtedness and shall not be liable to any such holders if it shall in good faith mistakenly pay over or distribute to Holders of Securities or to the Company or to any other Person cash, property or securities to which any holders of Senior Indebtedness shall be entitled by virtue of this Article XIV or otherwise, except in the case of gross negligence or wilful misconduct on the part of the Trustee.

SECTION 14.11. Rights of Trustee as Holder of Senior Indebtedness; Preservation of Trustee's Rights.

The Trustee in its individual capacity shall be entitled to all the rights set forth in this Article XIV with respect to any Senior Indebtedness which may at any time be held by it, to the same extent as any other holder of Senior Indebtedness, and nothing in this Indenture shall deprive the Trustee of any of its rights as such holder.

Nothing in this Article XIV shall apply to claims of, or payments to, the Trustee or its agent or counsel under or pursuant to Section 6.7.

SECTION 14.12. Article Applicable to Paying Agents.

In case at any time any Paying Agent other than the Trustee shall have been appointed by the Company and be then acting hereunder, the term "Trustee" as used in this Article XIV shall in such case (unless the context otherwise requires) be construed as extending to and including such Paying Agent within its meaning as fully for all intents and purposes as if such Paying Agent were named in this Article XIV in addition to or in place of the Trustee; provided, however, that Section 14.11 shall not apply to the Company or any Affiliate of the Company if it or such Affiliate acts as Paying Agent.

This Indenture may be executed in any number of counterparts, each of which so executed shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument.

IN WITNESS WHEREOF, the parties hereto have caused this Indenture to be duly executed as of the day and year first above written.

INTEGRATED ELECTRICAL SERVICES, INC.

By: /s/ H. DAVID RAMM _____ Name: H. David Ramm Title: President and Chief Executive Officer STATE STREET BANK AND TRUST COMPANY By: /s/ MICHAEL HOPKINS Name: Michael Hopkins Title: Vice President 1ST GROUP TELECOMMUNICATIONS, INC. ACE ELECTRIC, INC. ALADDIN WARD ELECTRIC & AIR, INC. ALADDIN WARD ELECTRIC, INC. AMBER ELECTRIC, INC. ANDERSON & WOOD CONSTRUCTION CO., INC. ARC ELECTRIC, INCORPORATED AXIS MANAGEMENT LLC BACHOFNER ELECTRIC, INC. BARTLEY & DEVARY ELECTRIC, INC. BEAR ACQUISITION CORPORATION BRINK ELECTRIC CONSTRUCTION CO. BRITT RICE ELECTRIC, INC. BRITT RICE MANAGEMENT LLC BRYANT ELECTRIC COMPANY, INC. BW CONSOLIDATED, INC. BW/BEC, INC. CANOVA ELECTRICAL CONTRACTING, INC. CARROLL MANAGEMENT LLC CARROLL SYSTEMS, INC. CHARLES P. BAGBY COMPANY, INC. COLLIER ELECTRIC COMPANY, INC. COMMERCIAL ELECTRICAL CONTRACTORS, INC. CROSS STATE ELECTRIC, INC. CYPRESS ELECTRICAL CONTRACTORS, INC. DANIEL ELECTRICAL CONTRACTORS, INC. DANIEL ELECTRICAL OF TREASURE COAST INC. DAVIS ELECTRICAL CONSTRUCTORS, INC. DELCO ELECTRIC, INC. ELECTRO-TECH, INC. EMC ACQUISITION CORPORATION ERNEST P. BREAUX ELECTRICAL, INC. FEDERAL COMMUNICATIONS GROUP, INC. FLORIDA INDUSTRIAL ELECTRIC, INC.

85

GENERAL PARTNER, INC. GOSS ELECTRIC COMPANY, INC. H.R. ALLEN, INC. HATFIELD REYNOLDS ELECTRIC COMPANY HOLLAND ELECTRICAL SYSTEMS, INC. HOUSTON-STAFFORD ELECTRIC, INC. HOUSTON-STAFFORD MANAGEMENT LLC HOWARD BROTHERS ELECTRIC CO., INC. I.C.G. ELECTRIC, INC. IES COMMUNICATIONS GROUP, INC. IES CONTRACTORS MANAGEMENT LLC IES ELECTRICAL GROUP, INC. IES RESIDENTIAL GROUP, INC IES SPECIALTY LIGHTING, INC. IES VENTURES INC. INVESTIGATED ELECTRIC COMPANY, INC. INTEGRATED ELECTRICAL FINANCE, INC. INTELLIGENT BUILDING SOLUTIONS, INC. J.W. GRAY ELECTRIC COMPANY, INC. J.W. GRAY MANAGEMENT, LLC KAYTON ELECTRIC, INC. KEY ELECTRICAL SUPPLY, INC. LINEMEN, INC. LINEMEN, INC. MARK HENDERSON, INCORPORATED MENNINGA ELECTRIC, INC. MIDLANDS ELECTRICAL CONTRACTORS, INC. MID-STATES ELECTRIC COMPANY, INC. MILLS ELECTRICAL CONTRACTORS, INC. MILLS MANAGEMENT LLC MITCHELL ELECTRIC COMPANY, INC. M-S SYSTEMS, INC. MURRAY ELECTRICAL CONTRACTORS, INC. MUTH ELECTRIC, INC. NEAL ELECTRIC MANAGEMENT LLC NEW TECHNOLOGY ELECTRICAL CONTRACTORS, INC. NEWCOMB ELECTRIC COMPANY, INC. PAN AMERICAN ELECTRIC COMPANY, INC. PAN AMERICAN ELECTRIC, INC. PAULIN ELECTRIC COMPANY, INC. POLLOCK ELECTRIC INC. PRIMENET, INC. PRIMO ELECTRIC COMPANY PUTZEL ELECTRICAL CONTRACTORS, INC. RAINES ELECTRIC CO., INC. RAINES MANAGEMENT LLC RKT ELECTRIC, INC. ROCKWELL ELECTRIC, INC. RODGERS ELECTRIC COMPANY, INC. RON'S ELECTRIC, INC. SPECTROL, INC. SPOOR ELECTRIC, INC. SUMMIT ELECTRIC OF TEXAS, INC.

```
T&H ELECTRICAL CORPORATION
TECH ELECTRIC CO., INC.
TEKNON ACQUISITION CORPORATION
TESLA POWER G.P., INC.
THOMAS POPP & COMPANY
VALENTINE ELECTRICAL, INC.
WOLFE ELECTRIC CO., INC.
WRIGHT ELECTRICAL CONTRACTING, INC.
By: /s/ NEIL J. DePASCAL, JR.
                                    Neil J. DePascal, Jr., Chief Accounting Officer
 AXIS COMMUNICATIONS LP
 By: Axis Management LLC, its general partner
 By: /s/ NEIL J. DePASCAL, JR.
                                 -----
                        - - - -
    Neil J. DePascal, Jr., Chief Accounting Officer
 B. RICE ELECTRIC LP
 By: Britt Rice Management LLC, its general partner
 By: /s/ NEIL J. DePASCAL, JR.
                                     Neil J. DePascal, Jr., Chief Accounting Officer
 BEXAR ELECTRIC COMPANY, LTD.
 By: BW/BEC, Inc., its general partner
 By: /s/ NEIL J. DePASCAL, JR.
                          - - -
```

Neil J. DePascal, Jr., Chief Accounting Officer

```
CARROLL SYSTEMS LP
By: Carroll Management LLC, its general partner
By: /s/ NEIL J. DePASCAL, JR.
                              Neil J. DePascal, Jr., Chief Accounting Officer
HAYMAKER ELECTRIC, LTD.
By: General Partner, Inc., its general partner
By: /s/ NEIL J. DePASCAL, JR.
                               -----
  Neil J. DePascal, Jr., Chief Accounting Officer
HOUSTON-STAFFORD ELECTRICAL CONTRACTORS, LP
By: Houston-Stafford Management LLC, its general
   partner
By: /s/ NEIL J. DePASCAL, JR.
  Neil J. DePascal, Jr., Chief Accounting Officer
ICS INTEGRATED COMMUNICATIONS SERVICES LP
By: Neal Electric Management LLC, its general
   partner
By: /s/ NEIL J. DePASCAL, JR.
  Neil J. DePascal, Jr., Chief Accounting Officer
IES CONTRACTORS LP
By: IES Contractors Management LLC, its general
   partner
By: /s/ NEIL J. DePASCAL, JR.
                                              - - - -
```

Neil J. DePascal, Jr., Chief Accounting Officer

```
IES MANAGEMENT LP
By: Integrated Electrical Finance, Inc.
By: /s/ NEIL J. DEPASCAL, JR.
                           Neil J. DePascal, Jr., Chief Accounting Officer
J.W. GRAY ELECTRICAL CONTRACTORS, LP
By: J.W. Gray Management, LLC, its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                         -----
  Neil J. DePascal, Jr., Chief Accounting Officer
MILLS ELECTRIC LP
By: Mills Management LLC, its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                              Neil J. DePascal, Jr., Chief Accounting Officer
NEAL ELECTRIC LP
By: BW/BEC, Inc., its general partner
```

By: /s/ NEIL J. DEPASCAL, JR. Neil J. DePascal, Jr., Chief Accounting Officer

83

```
POLLOCK SUMMIT ELECTRIC LP
By: Pollock Electric, Inc., its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                          -----
  Neil J. DePascal, Jr., Chief Accounting Officer
By: Summit Electric of Texas, Inc., its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                           Neil J. DePascal, Jr., Chief Accounting Officer
RAINES ELECTRIC LP
By: Raines Management LLC, its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                           Neil J. DePascal, Jr., Chief Accounting Officer
TESLA POWER AND AUTOMATION, LP
By: Tesla Power GP, Inc., its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                           -----
  Neil J. DePascal, Jr., Chief Accounting Officer
TESLA POWER PROPERTIES, LP
By: Tesla Power GP, Inc., its general partner
By: /s/ NEIL J. DEPASCAL, JR.
                           -----
```

Neil J. DePascal, Jr., Chief Accounting Officer

AXIS HOLDINGS LLC BRITT RICE HOLDINGS LLC BW/BEC, L.L.C. CARROLL HOLDINGS LLC HOUSTON-STAFFORD HOLDINGS LLC ICS HOLDINGS LLC IES CONTRACTORS HOLDINGS LLC IES HOLDINGS LLC J.W. GRAY HOLDINGS LLC MILLS ELECTRICAL HOLDINGS LLC POLLOCK SUMMIT HOLDINGS INC. RAINES HOLDINGS LLC By: /s/ ADRIANNE M. HORNE Adrianne M. Horne, President TESLA POWER (NEVADA), INC. DKD ELECTRIC COMPANY, INC. NBH HOLDING CO., INC.

84

By: /s/ ADRIANNE M. HORNE

Adrianne M. Horne, Chief Executive Officer

Schedule A

LIST OF GUARANTORS

1st Group Telecommunications, Inc. Ace Electric, Inc. Aladdin-Ward Electric & Air, Inc. Amber Electric, Inc. Anderson & Wood Construction Co., Inc. ARC Electric, Incorporated Axis Communications LP Axis Holdings LLC Axis Management LLC B. Rice Electric LP Bachofner Electric, Inc. Bardey & Devary Electric, Inc. Bear Acquisition Corporation Bexar Electric Company, Ltd. (GP is BW/BEC, Inc.) Brink Electric Construction Co. Britt Rice Electric, Inc. Britt Rice Holdings LLC Britt Rice Management LLC Bryant Electric Company, Inc. BW Consolidated, Inc. BW/BEC, Inc. BW/BEC, L.L.C. Canova Electrical Contracting, Inc. Carroll Holdings LLC Carroll Holdings LLC Carroll Management LLC Carroll Systems LP Carroll Systems, Inc. Charles P. Bagby Co., Inc. Collier Electric Company, Inc. Commercial Electrical Contractors, Inc. Cross State Electrical Contractors, Inc. Cypress Electrical Contractors, Inc. Daniel Electrical Contractors, Inc. Daniel Electrical of Treasure Coast, Inc. Davis Electrical Constructors, Inc. Delco Electric, Inc. DKD Electric Company, Inc. Electro-Tech, Inc. EMC Acquisition Corporation Ernest P. Breaux Electrical, Inc. Federal Communications Group, Inc. Florida Industrial Electric, Inc. General Partner, Inc. Goss Electric Company, Inc.

H. R. Allen, Inc. Hatfield Reynolds Electric Company Haymaker Electric, Ltd. (GP is General Partner, Inc.) Holland Electrical Systems, Inc. Houston-Stafford Electrical Contractors LP (GP is Houston Stafford Management LLC) Houston-Stafford Electric, Inc. Houston-Stafford Holdings LLC Houston-Stafford Management LLC Howard Brothers Electric Co., Inc. I.C.G. Electric, Inc. ICS Holdings LLC ICS Integrated Communication Services LP (GP is Neal Electric Management LLC) IES Communications Group, Inc. IES Contractors Holdings LLC IES Contractors LP (GP is IES Contractors Management LLC) IES Contractors Management LLC IES Electrical Group, Inc. IES Holdings LLC IES Management LP (GP is Integrated Electrical Finance, Inc.) IES Residential Group, Inc. IES Residential group, inc. IES Specialty Lighting, Inc. IES Ventures, Inc. Innovative Electric Company, Inc. Integrated Electric Company, Inc. Intelligent Building Solutions, Inc. J. W. Gray Electric Company, Inc. J. W. Gray Electrical Contractors LP J. W. Gray Electrical Contractors (GP is JW Gray Management LLC)
 J. W. Gray Holdings LLC
 J. W. Gray Management LLC
 Kayton Electric, Inc.
 Key Electrical Supply, Inc. Linemen, Inc. Mark Henderson, Incorporated Menninga Electric, Inc. Midlands Electrical Contractors, Inc. Mid-States Electric Company, Inc. Mills Electric LP (GP is Mills Management LLC) Mills Electrical Contractors, Inc. Mills Electrical Holdings LLC Mills Management LLC Mitchell Electric Company, Inc.

M-S Systems, Inc. Murray Electrical Contractors, Inc. Muth Electric, Inc. NBH Holding Co., Inc Neal Electric LP Neal Electric Management LLC New Technology Electrical Contractors, Inc. Newcomb Electric Company, Inc. Pan American Electric Company, Inc. Pan American Electric, Inc. Paulin Electric Company, Inc. Pollock Electric Inc. Pollock Summit Electric LP (GPs are Pollock Electric, Inc and Summit Electric of Texas, Inc.) Pollock Summit Holdings, Inc. PrimeNet, Inc. Primo Electric Company Putzel Electrical Contractors, Inc. Raines Electric Co., Inc. Raines Electric LP (GP is Raines Management LLC) Raines Holding LLC Raines Management LLC RKT Electric, Inc. Rockwell Electric, Inc. Rodgers Electric Company, Inc. Ron's Electric, Inc. Spectrol, Inc. Spoor Electric, Inc. Summit Electric of Texas, Incorporated T&H Electrical Corporation T&H Electrical Corporation Tech Electric Co., Inc. Teknon Acquisition Corporation Tesla Power G.P., Inc. Tesla Power (Nevada), Inc. Tesla Power and Automation, LP Tesla Power Properties, LP Thomas Popp & Company Valentine Electrical, Inc. Wolfe Electric Co., Inc. Wright Electrical Contracting, Inc.

FORM OF SECURITY

[PRIVATE PLACEMENT LEGEND FOR 144A SECURITIES AND IAI SECURITIES]

THIS SECURITY HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION. NEITHER THIS SECURITY NOR ANY INTEREST OR PARTICIPATION HEREIN MAY BE REOFFERED, SOLD, ASSIGNED, TRANSFERRED, PLEDGED, ENCUMBERED OR OTHERWISE DISPOSED OF IN THE ABSENCE OF SUCH REGISTRATION OR UNLESS SUCH TRANSACTION IS EXEMPT FROM, OR NOT SUBJECT TO, SUCH REGISTRATION.

THE HOLDER OF THIS SECURITY, BY ITS ACCEPTANCE HEREOF, AGREES ON ITS OWN BEHALF AND ON BEHALF OF ANY INVESTOR ACCOUNT FOR WHICH IT HAS PURCHASED TO OFFER, SELL OR OTHERWISE TRANSFER SUCH SECURITY, PRIOR TO THE SECURITIES. DATE (THE "RESALE RESTRICTION TERMINATION DATE") THAT IS TWO YEARS AFTER THE LATER OF THE ORIGINAL ISSUE DATE HEREOF AND THE LAST DATE ON WHICH THE ISSUER OR ANY AFFILIATE OF THE ISSUER WAS THE OWNER OF THIS SECURITY (OR ANY PREDECESSOR OF SUCH SECURITY), ONLY (A) TO THE ISSUER, (B) PURSUANT TO A REGISTRATION STATEMENT THAT HAS BEEN DECLARED EFFECTIVE UNDER THE SECURITIES ACT, (C) FOR SO LONG AS THE SECURITIES ARE ELIGIBLE FOR RESALE PURSUANT TO RULE 144A UNDER THE SECURITIES ACT, TO A PERSON IT REASONABLY BELIEVES IS A "QUALIFIED INSTITUTIONAL BUYER" AS DEFINED IN RULE 144A UNDER THE SECURITIES ACT THAT PURCHASES FOR ITS OWN ACCOUNT OR FOR THE ACCOUNT OF A QUALIFIED INSTITUTIONAL BUYER TO WHOM NOTICE IS GIVEN THAT THE TRANSFER IS BEING MADE IN RELIANCE ON RULE 144A, (D) PURSUANT TO OFFERS AND SALES THAT OCCUR OUTSIDE THE UNITED STATES WITHIN THE MEANING OF REGULATION S UNDER THE SECURITIES ACT, (E) TO AN INSTITUTIONAL "ACCREDITED INVESTOR" WITHIN THE MEANING OF RULE 501(a)(1), (2), (3) OR (7) UNDER THE SECURITIES ACT THAT IS AN INSTITUTIONAL ACCREDITED INVESTOR ACQUIRING THE SECURITY FOR ITS OWN ACCOUNT OR FOR THE ACCOUNT OF SUCH AN INSTITUTIONAL ACCREDITED INVESTOR, IN EACH CASE IN A MINIMUM PRINCIPAL AMOUNT OF THE SECURITIES OF \$250,000 OF SECURITIES, FOR INVESTMENT PURPOSES AND NOT WITH A VIEW TO OR FOR OFFER OR SALE IN CONNECTION WITH ANY DISTRIBUTION IN VIOLATION OF THE SECURITIES ACT, OR (F) PURSUANT TO ANOTHER AVAILABLE EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, SUBJECT TO THE ISSUER'S AND THE TRUSTEE'S RIGHT PRIOR TO ANY SUCH OFFER, SALE OR TRANSFER PURSUANT TO CLAUSES (D), (E) OR (F) TO REQUIRE THE DELIVERY OF AN OPINION OF COUNSEL, CERTIFICATION AND/OR OTHER INFORMATION SATISFACTORY TO EACH OF THEM. THIS LEGEND WILL BE REMOVED UPON THE REQUEST OF THE HOLDER AFTER THE RESALE RESTRICTION TERMINATION DATE.

[PRIVATE PLACEMENT LEGEND FOR REGULATION S SECURITIES]

THIS SECURITY HAS NOT BEEN REGISTERED UNDER THE U.S. SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), AND, ACCORDINGLY, MAY NOT BE OFFERED OR SOLD WITHIN THE UNITED STATES OR TO OR FOR THE ACCOUNT OR BENEFIT OF, U.S. PERSONS EXCEPT AS SET FORTH IN THE FOLLOWING SENTENCE. BY ITS ACQUISITION HEREOF, THE HOLDER (1) REPRESENTS THAT IT IS NOT A U.S. PERSON NOR IS IT PURCHASING FOR THE ACCOUNT OF A U.S. PERSON AND IS ACOUIRING THIS SECURITY IN AN OFFSHORE TRANSACTION IN ACCORDANCE WITH REGULATION S UNDER THE SECURITIES ACT ("REGULATION S"), (2) BY ITS ACCEPTANCE HEREOF AGREES TO OFFER, SELL OR OTHERWISE TRANSFER SUCH SECURITY, PRIOR TO THE DATE (THE "RESALE RESTRICTION TERMINATION DATE") THAT IS TWO YEARS AFTER THE LATER OF THE ORIGINAL ISSUE DATE HEREOF AND THE LAST DATE ON WHICH THE COMPANY OR ANY AFFILIATE OF THE COMPANY WAS THE OWNER OF THIS SECURITY (OR ANY PREDECESSOR OF SUCH SECURITY), ONLY (A) TO THE COMPANY, (B) PURSUANT TO A REGISTRATION STATEMENT THAT HAS BEEN DECLARED EFFECTIVE UNDER THE SECURITIES ACT, (C) FOR SO LONG AS THE SECURITIES ARE ELIGIBLE FOR RESALE PURSUANT TO RULE 144A UNDER THE SECURITIES ACT, TO A PERSON IT REASONABLY BELIEVES IS A "QUALIFIED INSTITUTIONAL BUYER" AS DEFINED IN RULE 144A UNDER THE SECURITIES ACT THAT PURCHASES FOR ITS OWN ACCOUNT OR FOR THE ACCOUNT OF A QUALIFIED INSTITUTIONAL BUYER TO WHOM NOTICE IS GIVEN THAT THE TRANSFER IS BEING MADE IN RELIANCE ON RULE 144A, (D) PURSUANT TO OFFERS AND SALES THAT OCCUR OUTSIDE THE UNITED STATES WITHIN THE MEANING OF REGULATION S, (E) TO AN INSTITUTIONAL ACCREDITED INVESTOR WITHIN THE MEANING OF RULE SO1(a)(1), (2), (3) OR (7) UNDER THE SECURITIES ACT THAT IS ACQUIRING THE SECURITY FOR ITS OWN ACCOUNT, OR FOR THE ACCOUNT OF SUCH AN INSTITUTIONAL ACCREDITED INVESTOR, IN EACH CASE IN A TRANSACTION INVOLVING A MINIMUM PRINCIPAL AMOUNT OF THE SECURITIES OF \$250,000, FOR INVESTMENT PURPOSES AND NOT WITH A VIEW TO OR FOR OFFER OR SALE IN CONNECTION WITH ANY DISTRIBUTION IN VIOLATION OF THE SECURITIES ACT, OR (F) PURSUANT TO ANY OTHER AVAILABLE EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, SUBJECT TO THE COMPANY'S AND THE TRUSTEE'S RIGHT PRIOR TO ANY SUCH OFFER, SALE OR TRANSFER PURSUANT TO CLAUSES (D), (E) OR (F) TO REQUIRE THE DELIVERY OF AN OPINION OF COUNSEL, CERTIFICATION AND/OR OTHER INFORMATION SATISFACTORY TO EACH OF THEM AND IN THE CASE OF THE FOREGOING CLAUSE (E), A CERTIFICATE OF TRANSFER IN THE FORM APPEARING ON THE OTHER SIDE OF THIS SECURITY IS COMPLETED AND DELIVERED BY THE TRANSFEROR TO THE COMPANY AND THE TRUSTEE. THIS LEGEND WILL BE REMOVED AFTER 40 CONSECUTIVE DAYS BEGINNING ON AND INCLUDING THE LATER OF (A) THE DAY ON WHICH THE SECURITIES ARE OFFERED TO PERSONS OTHER THAN DISTRIBUTORS (AS DEFINED IN REGULATION S) AND (B) THE DATE OF THE CLOSING OF THE ORIGINAL OFFERING. AS USED HEREIN, THE TERMS "OFFSHORE TRANSACTION," "UNITED STATES" AND "U.S. PERSON" HAVE THE MEANINGS GIVEN TO THEM BY REGULATION S UNDER THE SECURITIES ACT.

9[]% Senior Subordinated Note, Series C, due 2009

Original principal amount: \$

CUSIP NO.

Integrated Electrical Services, Inc., a corporation duly organized and existing under the laws of the State of Delaware (herein called the "Company," which term includes any successor Person under the Indenture hereinafter referred to), for value received, hereby promises to pay to ______ or registered assigns, the principal sum of ______ Dollars [(as increased or decreased in accordance with the terms of the Indenture)]1 on February 1, 2009 and to pay interest thereon from May 29, 2001 or from the most recent Interest Payment Date to which interest has been paid or duly provided for, semiannually in arrears on February 1 and August 1 in each year, commencing August 1, 2001 at the rate of 9[]% per annum, until the principal hereof is paid or duly provided for, provided that any principal and premium, and any such installment of interest, which is overdue shall bear interest shall be legally enforceable), from the dates such amounts are due until they are paid or duly provided for. The interest so payable and punctually paid or duly provided for, on any Interest Payment Date will, as provided in such Indenture, be paid to the Person in whose name this Security (or one or more predecessor Securities) is registered at the close of business on the Regular Record Date for such interest, which shall be the January 15 and July 15 (whether or not a Business Day), as the case may be, next preceding such Interest Payment Date. Any such interest not so punctually paid or duly provided for the payment of such Defaulted Interest to be fixed by the Trustee, notice whereof shall be given to be paid at any time in any other lawful manner not inconsistent with the requirements of any securities exchange on which the Securities may be listed, and upon such notice as may be required by such exchange, all as more fully provided in said Indenture.

Payment of the principal of (and premium, if any) and interest on this Security will be made at the office or agency of the Company maintained for that purpose, in such coin or currency of the United States of America as at the time of payment is legal tender for payment of public and private debts; provided, however, that, at the option of the Company, payment of interest may be made by check mailed to the address of the Person entitled thereto as such address shall appear in the Security Register (subject to the right of certain holders to request payment by wire transfer as provided in the Indenture).

Reference is hereby made to the further provisions of this Security set forth on the reverse hereof, which further provisions shall for all purposes have the same effect as if set forth at this place.

Unless the certificate of authentication hereon has been executed by the Trustee referred to on the reverse hereof by manual signature, this Security shall not be entitled to any benefit under the Indenture or be valid or obligatory for any purpose.



No.

^{*} For Global Securities only.

IN WITNESS WHEREOF, the Company h executed and attested.	as caused this instrument to be duly
Attest:	Integrated Electrical Services, Inc.
	By:
Title:	Title:
TILLE:	TILLE:

TRUSTEE'S CERTIFICATE OF AUTHENTICATION

This is one of the Securities referred to in the within-mentioned Indenture.

STATE STREET BANK AND TRUST COMPANY, as Trustee

Dated:

By: Authorized Signatory

FORM OF REVERSE OF SECURITY

This Security is one of a duly authorized issue of Securities of the Company designated as 9[]% Senior Subordinated Notes, Series C, due 2009 and 9[]% Senior Subordinated Notes, Series D, due 2009, issued and to be issued under an Indenture, dated as of May 29, 2001 (herein called the "Indenture," which term shall have the meaning assigned to it in such instrument), among the Company, the Guarantors named therein and State Street Bank and Trust Company, as Trustee (herein called the "Trustee," which term includes any successor trustee under the Indenture), and reference is hereby made to the Indenture for a statement of the respective rights, limitations of rights, duties and immunities thereunder of the Company, the Trustee and the Holders of the Securities and of the terms upon which the Securities are, and are to be, authenticated and delivered. The Securities include (i) \$125,000,000 aggregate principal amount of the Company's 9[]% Senior Subordinated Notes, Series C, due 2009 issued under the Indenture on May 29, 2001 (herein called "Initial Securities"), (ii) if and when issued, additional 9[]% Senior Subordinated Notes, Series C, due 2009 or 9[]% Senior Subordinated Notes, Series D, due 2009 of the Company that may be issued from time to time under the Indenture subsequent to May 29, 2001 (herein called "Additional Securities") and (iii) if and when issued, the Company's 9[]% Senior Subordinated Notes, Series D, due 2009 that may be issued from time to time under the Indenture in exchange for Initial Securities or Additional Securities in an offer registered under the Securities Act as provided in a Registration Rights Agreement (as referred to below) or as Private Exchange Securities as contemplated by the Registration Rights Agreement. The Initial Securities, Additional Securities and Exchange Securities are treated as a single class of securities under the Indenture.

The terms of the Securities include those stated in the Indenture and those made part of the Indenture by reference to the Trust Indenture Act of 1939 (15 U.S.C. Sections 7aaa - 77bbbb) (the "TIA"), as in effect on the date of the Indenture. Notwithstanding anything to the contrary herein, the Securities are subject to all such terms, and Holders of Securities are referred to the Indenture and the TIA for a statement of such terms. The Securities are unsecured senior subordinated obligations of the Company. The principal amount of Securities which may be issued under the Indenture is unlimited.

This Security is redeemable at the option of the Company, in whole or in part, at any time on or after February 1, 2004, at the Redemption Prices (expressed as percentages of principal amount) set forth below, plus accrued and unpaid interest, if any, thereon to the Redemption Date, if redeemed during the 12-month period beginning February 1 of the years indicated below:

Year	Redemption Price
2004 2005 2006 2007 and thereafter	

In addition, at any time, or from time to time, on or prior to February 1, 2002, the Company may, at its option, use the net cash proceeds of one or more Qualified Equity Offerings to redeem up to an aggregate of 35% of the principal amount of the Securities originally issued, at a redemption price equal to 109.375% of the principal amount thereof plus accrued and unpaid interest, if any, thereon to the Redemption Date; provided that at least 65% of the originally issued principal amount of Securities remains outstanding immediately after the occurrence of such redemption. In order to effect the foregoing redemption with the proceeds of any Qualified Equity Offering, the Company must consummate such redemption not later than 60 days after the closing of any such Qualified Equity Offering.

The Securities are not subject to any sinking fund.

The Indenture provides that the Company is obligated (a) upon the occurrence of a Change in Control to make an offer to purchase all outstanding Securities at a purchase price equal to 101% of the principal amount thereof, plus accrued and unpaid interest, if any, thereon to the date of purchase and (b) to make an offer to purchase Securities with a portion of the net cash proceeds of certain sales or other dispositions of assets (not applied as specified in the Indenture within the periods set forth therein) at a purchase price equal to 100% of the principal amount thereof plus accrued and unpaid interest, if any, to the date of purchase.

In the event of redemption or purchase of this Security in part only pursuant to a Change of Control Offer or an Asset Sale Offer, a new Security or Securities for the unredeemed or unpurchased portion hereof will be issued in the name of the Holder hereof upon the cancellation hereof.

The Indenture contains provisions for defeasance at any time of the entire indebtedness of this Security or of certain restrictive covenants and Events of Default with respect to this Security, in each case upon compliance with certain conditions set forth in the Indenture.

If an Event of Default shall occur and be continuing, there may be declared due and payable the principal of, premium, if any, and accrued and unpaid interest, if any, on all of the outstanding Securities, in the manner and with the effect provided in the Indenture.

The Indenture permits, with certain exceptions as therein provided, the amendment thereof and the modification of the rights and obligations of the Company and the rights of the Holders of the Securities under the Indenture at any time by the Company and the Trustee with the consent of the Holders of a majority in aggregate principal amount of the Securities at the time Outstanding. The Indenture also contains provisions permitting the Holders of specified percentages in aggregate principal amount of the Securities at the time Outstanding, on behalf of the Holders of all the Securities, to waive compliance by the Company with certain provisions of the Indenture and certain past defaults under the Indenture and their consequences. Any such consent or waiver by the Holder of this Security shall be conclusive and binding upon such Holder and upon all future Holders of this Security and of any Security issued upon the registration of transfer hereof or in exchange herefor or in lieu hereof, whether or not notation of such consent or waiver is made upon this Security.

As provided in and subject to the provisions of the Indenture, the Holder of this Security shall not have the right to institute any proceeding with respect to the Indenture or for the appointment of a receiver or trustee or for any other remedy thereunder, unless such Holder shall have previously given the Trustee written notice of a continuing Event of Default with respect to the Securities, the Holders of not less than 25% in principal amount of the Securities at the time Outstanding shall have made written request to the Trustee to institute proceedings in respect of such Event of Default as Trustee and offered the Trustee reasonable indemnity and the Trustee shall not have received from the Holders of a majority in principal amount of Securities at the time Outstanding a direction inconsistent with such request, and shall have failed to institute any such proceeding for 60 days after receipt of such notice, request and offer of indemnity. The foregoing shall not apply to certain suits described in the Indenture, including any suit

instituted by the Holder of this Security for the enforcement of any payment of principal hereof or any premium or interest hereon on or after the respective due dates expressed herein (or, in the case of redemption, on or after the Redemption Date or, in the case of any purchase of this Security required to be made pursuant to a Change of Control Offer or an Asset Sale Offer, on or after the relevant Purchase Date).

No reference herein to the Indenture and no provision of this Security or of the Indenture shall alter or impair the obligation of the Company, which is absolute and unconditional, to pay the principal of (and premium, if any) and interest on this Security at the times, place and rate, and in the coin or currency, herein prescribed.

As provided in the Indenture and subject to certain limitations therein set forth, the transfer of this Security is registrable in the Security Register, upon surrender of this Security for registration of transfer at the office or agency of the Company in the Borough of Manhattan, The City of New York, duly endorsed by, or accompanied by a written instrument of transfer in form satisfactory to the Company and the Security Registrar duly executed by, the Holder hereof or his attorney duly authorized in writing, and thereupon one or more new Securities, of authorized denominations and for the same aggregate principal amount, will be issued to the designated transferee or transferees.

This Security is issuable only in registered form without coupons in denominations of \$1,000 and any integral multiple thereof. As provided in the Indenture and subject to certain limitations therein set forth, Securities are exchangeable for a like aggregate principal amount of Securities of like tenor of a different authorized denomination, as requested by the Holder surrendering the same.

No service charge shall be made for any such registration of transfer or exchange, but the Company may require payment of a sum sufficient to cover any tax or other governmental charge payable in connection therewith.

Prior to due presentment of this Security for registration of transfer, the Company, the Trustee and any agent of the Company or the Trustee may treat the Person in whose name this Security is registered as the owner hereof for all purposes, whether or not this Security be overdue, and neither the Company, the Trustee nor any such agent shall be affected by notice to the contrary.

Pursuant to the Registration Rights Agreement relating to this Security by and among the Company and the other parties thereto, the Company will be obligated to consummate an exchange offer pursuant to which the Holder of this Security shall have the right to exchange this Security for 9[]% Senior Subordinated Notes, Series D, due 2009 of the Company, which have been registered under the Securities Act, in like principal amount and having identical terms as this Security (other than as set forth in this paragraph). The Holder of this Security shall be entitled to receive certain additional interest payments in the event such exchange offer is not consummated and upon certain other conditions, all pursuant to and in accordance with the terms of the Registration Rights Agreement.

Interest on this Security shall be computed on the basis of a 360-day year of twelve 30-day months.

The obligations of the Company under the Indenture and this Security are expressly subordinated to all Senior Indebtedness and senior in right of payment to all Subordinated Indebtedness, in each case to

the extent set forth in Article XIV of the Indenture, and reference is hereby made to such Indenture for the precise terms of such subordination.

As provided in the Indenture and subject to certain limitations therein set forth, the obligations of the Company under the Indenture and this Security are Guaranteed pursuant to Guarantees as provided in the Indenture. Each Holder, by holding this Security, agrees to all of the terms and provisions of said Guarantees. The Indenture provides that each Guarantor shall be released from its Guaranty upon compliance with certain conditions.

All terms used in this Security which are defined in the Indenture shall have the meanings assigned to them in the Indenture.

The Indenture and this Security shall be governed by and construed in accordance with the laws of the State of New York.

ASSIGNMENT FORM

If you, the Holder, want to assign this Security, fill in the form below and have your signature guaranteed:

I (or we) assign and transfer this Security to

(Insert assignee's social security or tax ID number)

- -----

(Print or type assignee's name, address and zip code) and irrevocably appoint

agent to transfer this Security on the books of the Company. The agent may substitute another to act for such agent.

In connection with any transfer or exchange of any of the Securities evidenced by this certificate occurring prior to the date that is two years after the later of the date of original issuance of such Securities and the last date, if any, on which such Securities were owned by the Company or any Affiliate of the Company, the undersigned confirms that such Securities are being:

CHECK ONE BOX BELOW:

- [] 1 acquired for the undersigned's own account, without transfer; or
 [] 2 transferred to the Company; or
- [] 3 transferred pursuant to and in compliance with Rule 144A under the Securities Act of 1933, as amended (the "Securities Act"); or
- [] 4 transferred pursuant to an effective registration statement under the Securities Act; or
- [] 5 transferred pursuant to and in compliance with Regulation S under the Securities Act; or
- [] 6 transferred to an institutional "accredited investor" (as defined in Rule 501(a)(1), (2), (3) or (7) under the Securities Act) that has furnished to the Trustee a signed letter containing certain representations and agreements (the form of which letter appears as Exhibit E of the Indenture); or

[] 7 transferred pursuant to another available exemption from the registration requirements of the Securities Act of 1933.

Unless one of the boxes is checked, the Trustee will refuse to register any of the Securities evidenced by this certificate in the name of any person other than the registered Holder thereof; provided, however, that if box (5), (6) or (7) is checked, the Trustee or the Company may require, prior to registering any such transfer of the Securities, in their sole discretion, such legal opinions, certifications and other information as the Trustee or the Company may reasonably request to confirm that such transfer is being made pursuant to an exemption from, or in a transaction not subject to, the registration requirements of the Securities Act of 1933, such as the exemption provided by Rule 144 under such Act.

Date: Your Signature: (Sign exactly as your name appears on the other side of this Security) By: NOTICE: To be executed by an executive officer

Signature Guarantee:

TO BE COMPLETED BY PURCHASER IF (3) ABOVE IS CHECKED.

The undersigned represents and warrants that it is purchasing this Security for its own account or an account with respect to which it exercises sole investment discretion and that it and any such account is a "qualified institutional buyer" within the meaning of Rule 144A under the Securities Act of 1933, as amended, and is aware that the sale to it is being made in reliance on Rule 144A and acknowledges that it has received such information regarding the Company as the undersigned has requested pursuant to Rule 144A or has determined not to request such information and that it is aware that the transferor is relying upon the undersigned's foregoing representations in order to claim the exemption from registration provided by Rule 144A.

Date:

OPTION OF HOLDER TO ELECT PURCHASE

If you want to elect to have this Security purchased in its entirety by the Company pursuant to Section 10.13 or 10.14 of the Indenture, check the applicable box:

Section 10.13 ____

Section 10.14 ____

If you want to elect to have only a part of the principal amount of this Security purchased by the Company pursuant to Section 10.13 or 10.14 of the Indenture, state the portion of such amount: \$_____

Date: Your Signature:

(Sign exactly as name appears on the other side of this Security)

Signature Guarantee:

(Signature must be guaranteed by a financial institution that is a member of the Securities Transfer Agent Medallion Program ("STAMP"), the Stock Exchange Medallion Program ("SEMP"), the New York Stock Exchange, Inc. Medallion Signature Program ("MSP") or such other signature guarantee program as may be determined by the Security Registrar in addition to, or in substitution for, STAMP, SEMP or MSP, all in accordance with the Securities Exchange Act of 1934, as amended.)

EXHIBIT A-2

INTEGRATED ELECTRICAL SERVICES, INC.

9[]% SENIOR SUBORDINATED NOTE DUE 2009, SERIES D

No.

Original principal amount: \$

CUSIP NO.

Integrated Electrical Services, Inc., a corporation duly organized and existing under the laws of the State of Delaware (herein called the "Company," which term includes any successor Person under the Indenture hereinafter referred to), for value received, hereby promises to pay to ______ or registered assigns, the principal sum of ______ Dollars [(as incre or decreased in accordance with the terms of the Indenture)]* on February 1, Dollars [(as increased 2009 and to pay interest thereon from May 29, 2001 or from the most recent Interest Payment Date to which interest has been paid or duly provided for, semiannually in arrears on February 1 and August 1 in each year, commencing August 1, 2001 at the rate of 9[]% per annum, until the principal hereof is paid or duly provided for, provided that any principal and premium, and any such installment of interest, which is overdue shall bear interest at the rate of 9[]% per annum (to the extent that the payment of such interest shall be legally enforceable), from the dates such amounts are due until they are paid or duly provided for. The interest so payable and punctually paid or duly provided for, on any Interest Payment Date will, as provided in such Indenture, be paid to the Person in whose name this Security (or one or more predecessor Securities) is registered at the close of business on the Regular Record Date for such interest, which shall be the January 15 and August 15 (whether or not a Business Day), as the case may be, next preceding such Interest Payment Date. Any such interest not so punctually paid or duly provided for will forthwith cease to be payable to the Holder on such Regular Record Date and may either be paid to the Person in whose name this Security (or one or more predecessor Securities) is registered at the close of business on a Special Record Date for the payment of such Defaulted Interest to be fixed by the Trustee, notice whereof shall be given to Holders of securities not less than 10 days prior to such Special Record Date, or be paid at any time in any other lawful manner not inconsistent with the requirements of any securities exchange on which the Securities may be listed, and upon such notice as may be required by such exchange, all as more fully provided in said Indenture.

Payment of the principal of (and premium, if any) and interest on this Security will be made at the office or agency of the Company maintained for that purpose, in such coin or currency of the United States of America as at the time of payment is legal tender for payment of public and private debts; provided, however, that, at the option of the Company, payment of interest may be made by check mailed to the address of the Person entitled thereto as such address shall appear in the Security Register (subject to the right of certain holders to request payment by wire transfer as provided in the Indenture).

Reference is hereby made to the further provisions of this Security set forth on the reverse hereof, which further provisions shall for all purposes have the same effect as if set forth at this place.

* For Global Securities only.

	IN WITNESS	WHEREOF,	the	Company	has	caused	this	instrument	to	be	duly
executed	and attest	ed.									

Attest:	Integrated Electrical Services, Inc.
	By:
Title:	Title:

TRUSTEE'S CERTIFICATE OF AUTHENTICATION

This is one of the Securities referred to in the within-mentioned Indenture.

STATE STREET BANK AND TRUST COMPANY, as Trustee

Dated:

By: Authorized Signatory

FORM OF REVERSE OF SECURITY

This Security is one of a duly authorized issue of Securities of the Company designated as 9[]% Senior Subordinated Notes, Series C, due 2009 and 9[]% Senior Subordinated Notes, Series D, due 2009, issued and to be issued under an Indenture, dated as of May 29, 2001 (herein called the "Indenture,' which term shall have the meaning assigned to it in such instrument), among the Company, the Guarantors named therein and State Street Bank and Trust Company, as Trustee (herein called the "Trustee," which term includes any successor trustee under the Indenture), and reference is hereby made to the Indenture for a statement of the respective rights, limitations of rights, duties and immunities thereunder of the Company, the Trustee and the Holders of the Securities and of the terms upon which the Securities are, and are to be, authenticated and delivered. The Securities include (i) \$125,000,000 aggregate principal amount of the Company's 9[]% Senior Subordinated Notes, Series C, due 2009 issued under the Indenture on May 29, 2001 (herein called "Initial Securities"), (ii) if and when issued, additional 9[]% Senior Subordinated Notes, Series C, due 2009 or 9[]% Senior Subordinated Notes, Series D, due 2009 of the Company that may be issued from time to time under the Indenture subsequent to May 29, 2001 (herein called "Additional Securities") and (iii) if and when issued, the Company's 9[]% Senior Subordinated Notes, Series D, due 2009 that may be issued from time to time under the Indenture in exchange for Initial Securities or Additional Securities in an offer registered under the Securities Act as provided in a Registration Rights Agreement (as referred to below) or as Private Exchange Securities as contemplated by the Registration Rights Agreement. The Initial Securities, Additional Securities and Exchange Securities are treated as a single class of securities under the Indenture.

The terms of the Securities include those stated in the Indenture and those made part of the Indenture by reference to the Trust Indenture Act of 1939 (15 U.S.C. Sections 7aaa - 77bbb) (the "TIA"), as in effect on the date of the Indenture. Notwithstanding anything to the contrary herein, the Securities are subject to all such terms, and Holders of Securities are referred to the Indenture and the TIA for a statement of such terms. The Securities are unsecured senior subordinated obligations of the Company. The principal amount of Securities which may be issued under the Indenture is unlimited.

This Security is redeemable at the option of the Company, in whole or in part, at any time on or after February 1, 2004, at the Redemption Prices (expressed as percentages of principal amount) set forth below, plus accrued and unpaid interest, if any, thereon to the Redemption Date, if redeemed during the 12-month period beginning February 1 of the years indicated below:

Year	Redemption Price
2004 2005 2006 2007 and thereafter	103.125% 101.563%

In addition, at any time, or from time to time, on or prior to February 1, 2002, the Company may, at its option, use the net cash proceeds of one or more Qualified Equity Offerings to redeem up to an aggregate of 35% of the principal amount of the Securities originally issued, at a redemption price equal to 109.375% of the principal amount thereof plus accrued and unpaid interest, if any, thereon to the Redemption Date; provided that at least 65% of the originally issued principal amount of Securities remains outstanding immediately after the occurrence of such redemption. In order to effect the

foregoing redemption with the proceeds of any Qualified Equity Offering, the Company shall consummate such redemption not later than 60 days after the closing of any such Qualified Equity Offering.

The Securities are not subject to any sinking fund.

The Indenture provides that the Company is obligated (a) upon the occurrence of a Change in Control to make an offer to purchase all outstanding Securities at a purchase price equal to 101% of the principal amount thereof, plus accrued and unpaid interest, if any, thereon to the date of purchase and (b) to make an offer to purchase Securities with a portion of the net cash proceeds of certain sales or other dispositions of assets (not applied as specified in the Indenture within the periods set forth therein) at a purchase price equal to 100% of the principal amount thereof plus accrued and unpaid interest, if any, to the date of purchase.

In the event of redemption or purchase of this Security in part only pursuant to a Change of Control Offer or an Asset Sale Offer, a new Security or Securities for the unredeemed or unpurchased portion hereof will be issued in the name of the Holder hereof upon the cancellation hereof.

The Indenture contains provisions for defeasance at any time of the entire indebtedness of this Security or of certain restrictive covenants and Events of Default with respect to this Security, in each case upon compliance with certain conditions set forth in the Indenture.

If an Event of Default shall occur and be continuing, there may be declared due and payable the principal of, premium, if any, and accrued and unpaid interest, if any, on all of the outstanding Securities, in the manner and with the effect provided in the Indenture.

The Indenture permits, with certain exceptions as therein provided, the amendment thereof and the modification of the rights and obligations of the Company and the rights of the Holders of the Securities under the Indenture at any time by the Company and the Trustee with the consent of the Holders of a majority in aggregate principal amount of the Securities at the time Outstanding. The Indenture also contains provisions permitting the Holders of specified percentages in aggregate principal amount of the Securities at the time Outstanding, on behalf of the Holders of all the Securities, to waive compliance by the Company with certain provisions of the Indenture and certain past defaults under the Indenture and their consequences. Any such consent or waiver by the Holder of this Security shall be conclusive and binding upon such Holder and upon all future Holders of this Security and of any Security issued upon the registration of transfer hereof or in exchange herefor or in lieu hereof, whether or not notation of such consent or waiver is made upon this Security.

As provided in and subject to the provisions of the Indenture, the Holder of this Security shall not have the right to institute any proceeding with respect to the Indenture or for the appointment of a receiver or trustee or for any other remedy thereunder, unless such Holder shall have previously given the Trustee written notice of a continuing Event of Default with respect to the Securities, the Holders of not less than 25% in principal amount of the Securities at the time Outstanding shall have made written request to the Trustee to institute proceedings in respect of such Event of Default as Trustee and offered the Trustee reasonable indemnity and the Trustee shall not have received from the Holders of a majority in principal amount of Securities at the time Outstanding a direction inconsistent with such request, and shall have failed to institute any such proceeding for 60 days after receipt of such notice, request and offer of indemnity. The foregoing shall not apply to certain suits described in the Indenture, including any suit

instituted by the Holder of this Security for the enforcement of any payment of principal hereof or any premium or interest hereon on or after the respective due dates expressed herein (or, in the case of redemption, on or after the Redemption Date or, in the case of any purchase of this Security required to be made pursuant to a Change of Control Offer or an Asset Sale Offer, on or after the relevant Purchase Date).

No reference herein to the Indenture and no provision of this Security or of the Indenture shall alter or impair the obligation of the Company, which is absolute and unconditional, to pay the principal of (and premium, if any) and interest on this Security at the times, place and rate, and in the coin or currency, herein prescribed.

The obligations of the Company under the Indenture and this Security are expressly subordinated to all Senior Indebtedness, in each case to the extent set forth in Article XIV of the Indenture, and reference is hereby made to such Indenture for the precise terms of such subordination.

As provided in the Indenture and subject to certain limitations therein set forth, the transfer of this Security is registrable in the Security Register, upon surrender of this Security for registration of transfer at the office or agency of the Company in the Borough of Manhattan, The City of New York, duly endorsed by, or accompanied by a written instrument of transfer in form satisfactory to the Company and the Security Registrar duly executed by, the Holder hereof or his attorney duly authorized in writing, and thereupon one or more new Securities, of authorized denominations and for the same aggregate principal amount, will be issued to the designated transferee or transferees.

This Security is issuable only in registered form without coupons in denominations of 1,000 and any integral multiple thereof. As provided in the Indenture and subject to certain limitations therein set forth, Securities are exchangeable for a like aggregate principal amount of Securities of like tenor of a different authorized denomination, as requested by the Holder surrendering the same.

No service charge shall be made for any such registration of transfer or exchange, but the Company may require payment of a sum sufficient to cover any tax or other governmental charge payable in connection therewith.

Prior to due presentment of this Security for registration of transfer, the Company, the Trustee and any agent of the Company or the Trustee may treat the Person in whose name this Security is registered as the owner hereof for all purposes, whether or not this Security be overdue, and neither the Company, the Trustee nor any such agent shall be affected by notice to the contrary.

Interest on this Security shall be computed on the basis of a 360-day year of twelve 30-day months.

As provided in the Indenture and subject to certain limitations therein set forth, the obligations of the Company under the Indenture and this Security are Guaranteed pursuant to Guarantees as provided in the Indenture. Each Holder, by holding this Security, agrees to all of the terms and provisions of said Guarantees. The Indenture provides that each Guarantor shall be released from its Guaranty upon compliance with certain conditions.

All terms used in this Security which are defined in the Indenture shall have the meanings assigned to them in the Indenture.

The Indenture and this Security shall be governed by and construed in accordance with the laws of the State of New York.

ASSIGNMENT FORM

If you, the Holder, want to assign this Security, fill in the form below and have your signature guaranteed:

 ${\tt I}$ (or we) assign and transfer this Security to

- -----(Insert assignee's social security or tax ID number) - -----_ _____ _____ (Print or type assignee's name, address and zip code) and irrevocably appoint agent to transfer this Security on the books of the Company. The agent may substitute another to act for such agent. Your Signature: Date: (Sign exactly as your name appears on the other side of this Security) By: -----NOTICE: To be executed by an executive officer Signature Guarantee:

OPTION OF HOLDER TO ELECT PURCHASE

If you want to elect to have this Security purchased in its entirety by the Company pursuant to Section 10.13 or 10.14 of the Indenture, check the applicable box:

Section 10.13 ____

Section 10.14 ____

If you want to elect to have only a part of the principal amount of this Security purchased by the Company pursuant to Section 10.13 or 10.14 of the Indenture, state the portion of such amount: \$_____

Date: Your Signature:

(Sign exactly as name appears on the other side of this Security)

Signature Guarantee:

(Signature must be guaranteed by a financial institution that is a member of the Securities Transfer Agent Medallion Program ("STAMP"), the Stock Exchange Medallion Program ("SEMP"), the New York Stock Exchange, Inc. Medallion Signature Program ("MSP") or such other signature guarantee program as may be determined by the Security Registrar in addition to, or in substitution for, STAMP, SEMP or MSP, all in accordance with the Securities Exchange Act of 1934, as amended.)

EXHIBIT B

FORM OF LEGEND FOR BOOK-ENTRY SECURITIES

Any Global Security authenticated and delivered hereunder shall bear a legend (which would be in addition to any other legends required in the case of a Restricted Security) in substantially the following form:

THIS SECURITY IS A GLOBAL SECURITY WITHIN THE MEANING OF THE INDENTURE HEREINAFTER REFERRED TO AND IS REGISTERED IN THE NAME OF A DEPOSITORY OR A NOMINEE OF A DEPOSITORY OR A SUCCESSOR DEPOSITORY. THIS SECURITY IS NOT EXCHANGEABLE FOR SECURITIES REGISTERED IN THE NAME OF A PERSON OTHER THAN THE DEPOSITORY OR ITS NOMINEE EXCEPT IN THE LIMITED CIRCUMSTANCES DESCRIBED IN THE INDENTURE, AND NO TRANSFER OF THIS SECURITY (OTHER THAN A TRANSFER OF THIS SECURITY AS A WHOLE BY THE DEPOSITORY TO A NOMINEE OF THE DEPOSITORY OR BY A NOMINEE OF THE DEPOSITORY TO THE DEPOSITORY OR ANOTHER NOMINEE OF THE DEPOSITORY) MAY BE REGISTERED EXCEPT IN THE LIMITED CIRCUMSTANCES DESCRIBED IN THE INDENTURE.

UNLESS THIS CERTIFICATE IS PRESENTED BY AN AUTHORIZED REPRESENTATIVE OF THE DEPOSITORY TRUST COMPANY, A NEW YORK CORPORATION ("D.T.C."), TO THE COMPANY OR ITS AGENT FOR REGISTRATION OF TRANSFER, EXCHANGE OR PAYMENT, AND ANY CERTIFICATE ISSUED IS REGISTERED IN THE NAME OF CEDE & CO. OR IN SUCH OTHER NAME AS IS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF D.T.C. (AND ANY PAYMENT IS MADE TO CEDE & CO. OR TO SUCH OTHER ENTITY AS IS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF D.T.C.), ANY TRANSFER, PLEDGE OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL INASMUCH AS THE REGISTERED OWNER HEREOF, CEDE & CO., HAS AN INTEREST HEREIN.

B-1

1

EXHIBIT C

FORM OF CERTIFICATE TO BE DELIVERED IN CONNECTION WITH TRANSFERS PURSUANT TO REGULATION S

State Street Bank and Trust Company Goodwin Square, 23rd Floor 225 Asylum Street Hartford, CT 06103 Attn: Corporate Trust Administration

> Re: Integrated Electrical Services, Inc. (the "Company") 9[]% Senior Subordinated Notes, Series C, due 2009 (the "Securities")

Ladies and Gentlemen:

In connection with our proposed sale of ______ aggregate principal amount of the Securities, we confirm that such sale has been effected pursuant to and in accordance with Regulation S under the U.S. Securities Act of 1933, as amended (the "Securities Act"), and, accordingly, we represent that:

(1) the offer of the Securities was not made to a Person in the United States;

(2) either (a) at the time the buy offer was originated, the transferee was outside the United States or we and any Person acting on our behalf reasonably believed that the transferee was outside the United States, or (b) the transaction was executed in, on or through the facilities of a designated offshore securities market described in Rule 902(a) of Regulation S and neither we nor any Person acting on our behalf knows that the transaction has been pre-arranged with a buyer in the United States;

(3) no directed selling efforts have been made in the United States in contravention of the requirements of Rule 903(b) or Rule 904(b) of Regulation S, as applicable;

(4) the transaction is not part of a plan or scheme to evade the registration requirements of the Securities Act;

(5) we have advised the transferee of the transfer restrictions applicable to the Securities;

(6) if the circumstances set forth in Rule 904(c) under the Securities Act are applicable, we have complied with the additional conditions therein, including (if applicable) sending a confirmation or other notice stating that the Securities may be offered and sold during the distribution compliance period specified in Rule 903(c)(2) or (3), as applicable, only in accordance with the provisions of Regulation S; pursuant to registration of the Securities under the Securities Act; or pursuant to another available exemption from the registration requirements under the Securities Act; and

C-1

(7) if the sale is made during a distribution compliance period and the provisions of Rule 903(c)(3) are applicable thereto, we confirm that such sale has been made in accordance with such provisions.

You and the Company are entitled to rely upon this letter and are irrevocably authorized to produce this letter or a copy hereof to any interested party in any administrative or legal proceedings or official inquiry with respect to the matters covered hereby. Terms used in this certificate have the meanings set forth in Peoulation S meanings set forth in Regulation S.

Very truly yours,

[Name of Transferor]

By: Authorized Signature

C-2

EXHIBIT D

FORM OF SUPPLEMENTAL INDENTURE RELATING TO ADDITIONAL GUARANTORS

SUPPLEMENTAL INDENTURE

This SUPPLEMENTAL INDENTURE, dated as of _____, ___, is among Integrated Electrical Services, Inc., a Delaware corporation (the "Company"), each of the parties identified under the caption "Guarantors" on the signature page hereto (the "Guarantors") and State Street Bank and Trust Company, as Trustee.

RECITALS

WHEREAS, the Company, the subsidiaries of the Company named therein and the Trustee entered into an Indenture, dated as of May 29, 2001 (as amended or supplemented, the "Indenture"), pursuant to which the Company has issued its 9[]% Senior Subordinated Notes, Series C, due 2009 and its 9[]% Senior Subordinated Notes, Series D, due 2009 (the "Securities"); and

WHEREAS, Section 9.1(i) of the Indenture provides that the Company and the Trustee may amend or supplement the Indenture in order to execute and deliver a guarantee (a "Guarantee") to comply with Section 10.17 thereof without the consent of the Holders of the Securities; and

WHEREAS, all acts and things prescribed by the Indenture, by law and by the certificate of incorporation and the bylaws (or comparable constituent documents) of the Company, of the Guarantors and of the Trustee necessary to make this Supplemental Indenture a valid instrument legally binding on the Company, the Guarantors and the Trustee, in accordance with its terms, have been duly done and performed;

NOW, THEREFORE, to comply with the provisions of the Indenture and in consideration of the above premises, the Company, the Guarantors and the Trustee covenant and agree for the equal and proportionate benefit of the respective Holders of the Securities as follows:

ARTICLE 1

Section 1.01. This Supplemental Indenture is supplemental to the Indenture and does and shall be deemed to form a part of, and shall be construed in connection with and as part of, the Indenture for any and all purposes.

Section 1.02. This Supplemental Indenture shall become effective immediately upon its execution and delivery by each of the Company, the Guarantors and the Trustee.

ARTICLE 2

From this date, in accordance with Section 10.17 and by executing this Supplemental Indenture, the Guarantors whose signatures appear below are Guarantors for all purposes under the Indenture and are subject to the provisions of the Indenture to the extent provided for in Article XIII thereunder.

D-1

ARTICLE 3

Section 3.01. Except as specifically modified herein, the Indenture and the Securities are in all respects ratified and confirmed (mutatis mutandis) and shall remain in full force and effect in accordance with their terms with all capitalized terms used herein without definition having the same respective meanings ascribed to them as in the Indenture.

Section 3.02. Except as otherwise expressly provided herein, no duties, responsibilities or liabilities are assumed, or shall be construed to be assumed, by the Trustee by reason of this Supplemental Indenture. This Supplemental Indenture is executed and accepted by the Trustee subject to all the terms and conditions set forth in the Indenture with the same force and effect as if those terms and conditions were repeated at length herein and made applicable to the Trustee with respect hereto.

Section 3.03. THE LAW OF THE STATE OF NEW YORK SHALL GOVERN AND BE USED TO CONSTRUE AND ENFORCE THIS SUPPLEMENTAL INDENTURE.

Section 3.04. The parties may sign any number of copies of this Supplemental Indenture. Each signed copy shall be an original, but all of such executed copies together shall represent the same agreement.

[NEXT PAGE IS SIGNATURE PAGE]

IN WITNESS WHEREOF, the parties hereto have caused this Supplemental Indenture to be duly executed, all as of the date first written above.

INTEGRATED ELECTRICAL SERVICES, INC.

Ву Name: Title:

GUARANTORS

[]

Ву Name: Title:

STATE STREET BANK AND TRUST COMPANY, as Trustee

Ву Name: Title:

D-3

1

EXHIBIT E

ANNEX A TO OFFERING MEMORANDUM

FORM OF CERTIFICATE FROM ACQUIRING INSTITUTIONAL ACCREDITED INVESTOR

[Date]

Integrated Electrical Services, Inc. 1800 West Loop South, Suite 500 Houston, Texas 77027-3290 Attention: General Counsel

Dear Sirs:

This certificate is delivered to request a transfer of \$_____ principal amount of the 9[]% Senior Subordinated Notes, Series C, due 2009 (the "Securities") of Integrated Electrical Services, Inc. (the "Company").

Upon transfer, the Securities would be registered in the name of the new beneficial owner as follows:

Name: _____

Taxpayer ID Number:

Address: ___

The undersigned represents and warrants to you that:

1. We are an institutional "accredited investor" (as defined in Rule 501(a)(1), (2), (3) or (7) under the Securities Act of 1933, as amended (the "Securities Act")) purchasing for our own account or for the account of such an institutional "accredited investor" at least \$250,000 principal amount of the Securities, and we are acquiring the Securities not with a view to, or for offer or sale in connection with, any distribution in violation of the Securities Act. We have such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risk of our investment in the Securities and we invest in or purchase securities similar to the Securities in the normal course of our business. We and any accounts for which we are acting are each able to bear the economic risk of our or its investment.

2. We understand that the Securities have not been registered under the Securities Act and, unless so registered, may not be sold except as permitted in the following sentence. We agree on our own behalf and on behalf of any investor account for which we are purchasing Securities to offer, sell or otherwise transfer such Securities prior to the date which is two years after the later of the date of original issue and the last date on which the Company or any affiliate of the Company was the owner of

E-1

such Securities (or any predecessor thereto) (the "Resale Restriction Termination Date") only (a) to the Company, (b) pursuant to a registration statement which has been declared effective under the Securities Act, (c) in a transaction complying with the requirements of Rule 144A under the Securities Act ("Rule 144A"), to a person we reasonably believe is a qualified institutional buyer under Rule 144A (a "QIB") that purchases for its own account or for the account of a QIB and to whom notice is given that the transfer is being made in reliance on Rule 144A, (d) pursuant to offers and sales that occur outside the United States within the meaning of Regulation S under the Securities Act, (e) to an institutional "accredited investor" (within the meaning of Rule 501(a)(1), (2), (3) or (7) under the Securities Act) that is purchasing for its own account or for the account of such an institutional "accredited investor," in each case in a minimum principal amount of Securities of \$250,000 or (f) pursuant to any other available exemption from the registration requirements of the Securities Act, subject in each of the foregoing cases to any requirement of law that the disposition of our property or the property of such investor account or accounts be at all times within our or their control and in compliance with any applicable state securities laws. The foregoing restrictions on resale will not apply subsequent to the Resale Restriction Termination Date. If any resale or other transfer of the Securities is proposed to be made pursuant to clause (e) above prior to the Resale Restriction Termination Date, the transferor shall deliver a letter from the transferee substantially in the form of this letter to the Company and the Trustee, which shall provide, among other things, that the transferee is an institutional "accredited investor" (within the meaning of Rule 501(a)(1), (2), (3) or (7) under the Securities Act) that is acquiring such Securities for investment purposes and not for distribution in violation of the Securities Act. We acknowledge that the Company and the Trustee reserve the right prior to any offer, sale or other transfer prior to the Resale Termination Date of the Securities pursuant to clauses (d), (e) or (f) above to require the delivery of an opinion of counsel, certifications and/or other information satisfactory to the Company and the Trustee.

TRANSFEREE:

		-	 -	 -	 -	 	 -	 	-	-	-	-	-	-
BY:														
	 		 -	 -	 _	 	 -	 	-	-	-		_	_

E-2

EXECUTION COPY

INTEGRATED ELECTRICAL SERVICES, INC.

\$125,000,000

9[]% Senior Subordinated Notes, Series C, due 2009

EXCHANGE AND REGISTRATION RIGHTS AGREEMENT

May 29, 2001

J.P. MORGAN SECURITIES INC.
Merrill Lynch, Pierce, Fenner & Smith Incorporated
CREDIT LYONNAIS SECURITIES (USA) INC.
SCOTIA CAPITAL (USA) INC.
TD SECURITIES (USA) INC.
c/o J.P. Morgan Securities Inc.
270 Park Avenue, 4th floor
New York, New York 10017

Ladies and Gentlemen:

Integrated Electrical Services, a Delaware corporation (the "Company"), proposes to issue and sell to J.P. Morgan Securities Inc., Merrill Lynch, Pierce, Fenner & Smith Incorporated, Credit Lyonnais Securities (USA) Inc., Scotia Capital (USA) Inc. and TD Securities (USA) Inc. (together, the "Initial Purchasers"), upon the terms and subject to the conditions set forth in a purchase agreement dated May 23, 2001 (the "Purchase Agreement"), \$125,000,000 aggregate principal amount of its 9[]% Senior Subordinated Notes, Series C, due 2009 (the "Securities") to be jointly and severally guaranteed on a senior subordinated basis by the subsidiaries of the Company listed on Schedule 1 and signatories hereto (collectively, the "Guarantors"). Capitalized terms used but not defined herein shall have the meanings given to such terms in the Purchase Agreement.

As an inducement to the Initial Purchasers to enter into the Purchase Agreement and in satisfaction of a condition to the obligations of the Initial Purchasers thereunder, the Company and the Guarantors agree with the Initial Purchasers, for the benefit of the holders (including the Initial Purchasers) of the Securities, the Exchange Securities (as defined herein) and the Private Exchange Securities (as defined herein) (collectively, the "Holders"), as follows:

1. Registered Exchange Offer

The Company shall (i) prepare and, not later than 60 days following the date of original issuance of the Securities (the "Issue Date"), file with the Commission a registration

statement (the "Exchange Offer Registration Statement") on an appropriate form under the Securities Act with respect to a proposed offer to the Holders of the Securities (the "Registered Exchange Offer") to issue and deliver to such Holders, in exchange for the Securities, a like aggregate principal amount of debt securities of the Company (the "Exchange Securities") that are identical in all material respects to the Securities, except for the transfer restrictions relating to the Securities, (ii) use its reasonable best efforts to cause the Exchange Offer Registration Statement to become effective under the Securities Act no later than 150 days after the Issue Date and the Registered Exchange Offer to be consummated no later than 180 days after the Issue Date and (iii) keep the Exchange Offer Registration Statement effective for not less than 20 business days (or longer, if required by applicable law) after the date on which notice of the Registered Exchange Offer Registration Period"). The Exchange Securities Will be issued under the Indenture or an indenture (the "Exchange Securities Indenture") between the Company, the Guarantors and the Trustee or such other bank or trust company that is reasonably satisfactory to the Initial Purchasers, as trustee (the "Exchange Securities Trustee"), such indenture to be identical in all material respects to the Indenture, except for the transfer restrictions relating to the Securities (as described above).

Upon the effectiveness of the Exchange Offer Registration Statement, the Company shall promptly commence the Registered Exchange Offer, it being the objective of such Registered Exchange Offer to enable each Holder electing to exchange Securities for Exchange Securities (assuming that such Holder (a) is not an affiliate of the Company or an Exchanging Dealer (as defined herein) not complying with the requirements of the next sentence, (b) is not an Initial Purchaser holding Securities that have, or that are reasonably likely to have, the status of an unsold allotment in an initial distribution, (c) acquires the Exchange Securities in the ordinary course of such Holder's business and (d) has no arrangements or understandings with any person to Exchange Securities from and after their receipt without any limitations or restrictions under the Securities Act and without material restrictions under the securities laws of the several states of the United States. The Company, the Guarantors, the Initial Purchasers and each Exchanging Dealer acknowledge that, pursuant to current interpretations by the Commission's staff of Section 5 of the Securities Act, each Holder that is a broker-dealer electing to exchange Securities, acquired for its own account as a result of market-making activities or other trading activities, for Exchange Securities (an "Exchanging Dealer"), is required to deliver a prospectus containing substantially the information set forth in Annex A hereto in the "Exchange Offer Procedures" section and the "Purpose of the Exchange Offer" section and in Annex B hereto in the "Plan of Distribution" section of such prospectus in connection with a sale of any such Exchange Securities received by such Exchanging Dealer pursuant to the Registered Exchange Offer.

If, prior to the consummation of the Registered Exchange Offer, any Holder holds any Securities acquired by it that have, or that are reasonably likely to be determined to have, the status of an unsold allotment in an initial distribution, or any Holder is not entitled to participate in the Registered Exchange Offer, the Company shall, upon the request of any such Holder, simultaneously with the delivery of the Exchange Securities in the Registered Exchange Offer, issue and deliver to any such Holder, in exchange for the Securities held by such Holder (the In connection with the Registered Exchange Offer, the Company shall:

 (a) mail to each Holder a copy of the prospectus forming part of the Exchange Offer Registration Statement, together with an appropriate letter of transmittal and related documents;

(b) keep the Registered Exchange Offer open for not less than 20 business days (or longer, if required by applicable law) after the date on which notice of the Registered Exchange Offer is mailed to the Holders;

(c) utilize the services of a depositary for the Registered Exchange Offer with an address in the Borough of Manhattan, The City of New York;

(d) permit Holders to withdraw tendered Securities at any time prior to the close of business, New York City time, on the last business day on which the Registered Exchange Offer shall remain open; and

(e) otherwise comply in all respects with all laws that are applicable to the Registered Exchange Offer.

As soon as practicable after the close of the Registered Exchange Offer and any Private Exchange, as the case may be, the Company shall:

 (a) accept for exchange all Securities tendered and not validly withdrawn pursuant to the Registered Exchange Offer and the Private Exchange;

(b) deliver to the Trustee for cancellation all Securities so accepted for exchange; and

(c) cause the Trustee or the Exchange Securities Trustee, as the case may be, promptly to authenticate and deliver to each Holder, Exchange Securities or Private Exchange Securities, as the case may be, equal in principal amount to the Securities of such Holder so accepted for exchange.

The Company shall use its reasonable best efforts to keep the Exchange Offer Registration Statement effective and to amend and supplement the prospectus contained therein in order to permit such prospectus to be used by all persons subject to the prospectus delivery requirements of the Securities Act for such period of time as such persons must comply with such requirements in order to resell the Exchange Securities; provided that (i) in the case where such prospectus and any amendment or supplement thereto must be delivered by an Exchanging

Dealer, such period shall be the lesser of 180 days and the date on which all Exchanging Dealers have sold all Exchange Securities held by them and (ii) the Company shall make such prospectus and any amendment or supplement thereto available to any broker-dealer for use in connection with any resale of any Exchange Securities for a period of not less than 90 days after the consummation of the Registered Exchange Offer.

The Indenture or the Exchange Securities Indenture, as the case may be, shall provide that the Securities, the Exchange Securities and the Private Exchange Securities shall vote and consent together on all matters as one class and that none of the Securities, the Exchange Securities or the Private Exchange Securities will have the right to vote or consent as a separate class on any matter.

Interest on each Exchange Security and Private Exchange Security issued pursuant to the Registered Exchange Offer and in the Private Exchange will accrue from the last interest payment date on which interest was paid on the Securities surrendered in exchange therefor or, if no interest has been paid on the Securities, from the Issue Date.

Each Holder participating in the Registered Exchange Offer shall be required to represent to the Company that at the time of the consummation of the Registered Exchange Offer (i) any Exchange Securities received by such Holder will be acquired in the ordinary course of business, (ii) such Holder will have no arrangements or understanding with any person to participate in the distribution of the Securities or the Exchange Securities within the meaning of the Securities Act and (iii) such Holder is not an affiliate of the Company or, if it is such an affiliate, such Holder will comply with the registration and prospectus delivery requirements of the Securities Act to the extent applicable.

Notwithstanding any other provisions hereof, the Company and the Guarantors will ensure that (i) any Exchange Offer Registration Statement and any amendment thereto and any prospectus forming part thereof and any supplement thereto complies in all material respects with the Securities Act and the rules and regulations of the Commission thereunder, (ii) any Exchange Offer Registration Statement and any amendment thereto does not, when it becomes effective, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading and (iii) any prospectus forming part of any Exchange Offer Registration Statement, and any supplement to such prospectus, does not, as of the consummation of the Registered Exchange Offer, include an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

2. Shelf Registration

If (i) because of any change in law or applicable interpretations thereof by the Commission's staff the Company is not permitted to effect the Registered Exchange Offer as contemplated by Section 1 hereof, or (ii) for any other reason the Registered Exchange Offer is not consummated within 180 days after the Issue Date, or (iii) any Securities validly tendered pursuant to the Registered Exchange Offer are not exchanged for Exchange Securities within 10 days of being accepted in the Registered Exchange Offer, or (iv) any Initial Purchaser so requests with respect to Securities or Private Exchange Securities not eligible to be exchanged for Exchange Securities in the Registered Exchange Offer and held by it following the consummation of the Registered Exchange Offer, or (v) any applicable law or interpretations do not permit any Holder to participate in the Registered Exchange Offer, or (vi) any Holder that participates in the Registered Exchange Offer does not receive freely transferable Exchange Securities in exchange for tendered Securities, then the following provisions shall apply:

(a) The Company and the Guarantors shall file as promptly as practicable (but in no event more than 20 days after so required or requested pursuant to this Section 2) with the Commission (the "Shelf Filing Date"), and thereafter shall use their reasonable best efforts to cause to be declared effective, a shelf registration statement on an appropriate form under the Securities Act relating to the offer and sale of the Transfer Restricted Securities (as defined below) by the Holders thereof from time to time in accordance with the methods of distribution set forth in such registration statement (hereafter, a "Shelf Registration Statement" and, together with any Exchange Offer Registration Statement, a "Registration Statement").

(b) The Company and the Guarantors shall use their reasonable best efforts to keep the Shelf Registration Statement continuously effective in order to permit the prospectus forming part thereof to be used by Holders of Transfer Restricted Securities (as defined below) for a period ending on the earlier of (i) two years from the Issue Date or such shorter period that will terminate when all the Transfer Restricted Securities covered by the Shelf Registration Statement have been sold pursuant thereto and (ii) the date on which the Securities become eligible for resale without volume restrictions pursuant to Rule 144 under the Securities Act (in any such case, such period being called the "Shelf Registration Period").

(c) Notwithstanding any other provisions hereof, the Company and the Guarantors will ensure that (i) any Shelf Registration Statement and any amendment thereto and any prospectus forming part thereof and any supplement thereto complies in all material respects with the Securities Act and the rules and regulations of the Commission thereunder, (ii) any Shelf Registration Statement and any amendment thereto (in either case, other than with respect to information included therein in reliance upon or in conformity with written information furnished to the Company by or on behalf of any Holder specifically for use therein (the "Holders' Information")) does not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading and (iii) any prospectus forming part of any Shelf Registration Statement, and any supplement to such prospectus (in either case, other than with respect to Holders' Information), does not include an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

(d) In the absence of the events described in clauses (i) through (vi) of the first paragraph of this Section 2, the Company and the Guarantors shall not be permitted to discharge their obligations hereunder by means of the filing of a Shelf Registration Statement.

5

3. Additional Interest

(a) The parties hereto agree that the Holders of Transfer Restricted Securities will suffer damages if the Company and the Guarantors fail to fulfill their obligations under Section 1 or Section 2, as applicable, and that it would not be feasible to ascertain the extent of such damages. Accordingly, if (i) the Exchange Offer Registration Statement is not filed with the Commission on or prior to 60 days after the Issue Date or the Shelf Registration Statement is not filed with the Commission on or before the Shelf Filing Date, (ii) the Exchange Offer Registration Statement is not declared effective within 150 days after the Issue Date or the Shelf Registration Statement is not declared effective within 120 days of the Shelf Filing Date, (iii) the Registered Exchange Offer is not consummated on or prior to 180 days after the Issue Date, or (iv) the Shelf Registration Statement is filed and declared effective within 120 days after the Shelf Filing Date but shall thereafter cease to be effective (at any time that the Company and the Guarantors are obligated to maintain the effectiveness thereof) without being succeeded within 30 days by an additional Registration Statement filed and declared effective (each such event referred to in clauses (i) through (iv), "Registration Default"), the Company and the Guarantors will be jointly and severally obligated to pay additional interest to each Holder of Transfer Restricted Securities, upon the occurrence of one or more such Registration Defaults, at the rate of 0.25% per annum on the principal amount of Transfer Restricted Securities held by such Holder plus, beginning on the 91st day after the occurrence of such Registration Default, an additional 0.25% per annum on the principal amount of Transfer Restricted Securities held by such Holder, until (i) the applicable Registration Statement is filed, (ii) the Exchange Offer Registration Statement is declared effective and the Registered Exchange Offer is consummated, (iii) the Shelf Registration Statement is declared effective or (iv) the Shelf Registration Statement again becomes effective, as the case may be. Following the cure of all Registration Defaults, the accrual of additional interest will cease. As used herein, the term "Transfer Restricted Securities" means each Security or Private Exchange Security, until the earliest to occur of: (i) the date on which such Security has been exchanged for a freely transferable Exchange Security in the Registered Exchange Offer, (ii) the date on which such Security or Private Exchange Security has been effectively registered under the Securities Act and disposed of in accordance with the Shelf Registration Statement or (iii) the date on which such Security or Private Exchange Security is distributed to the public pursuant to Rule 144 under the Securities Act or is saleable pursuant to Rule 144(k) under the Securities Act. Notwithstanding anything to the contrary in this Section 3(a), neither the Company nor the Guarantors shall be required to pay additional interest to a Holder of Transfer Restricted Securities if such Holder failed to comply with its obligations to make the representations set forth in the second to last paragraph of Section 1 or failed to provide the information required to be provided by it, if any, pursuant to Section 4(n).

(b) The Company shall notify the Trustee and the paying agent to be appointed pursuant to the Indenture (the "Paying Agent") under the Indenture within three business days of the happening of each and every Registration Default. The Company and the Guarantors shall pay the additional interest due on the Transfer Restricted Securities by depositing with the Paying Agent (which may not be the Company for these purposes), in trust, for the benefit of the Holders thereof, prior to 10:00 a.m., New York City time, on the next interest payment date specified by the Indenture and the Securities, sums sufficient to pay the additional interest then due. The additional interest due shall be payable on each interest payment date specified by the Indenture and the Securities to the record holder entitled to receive the interest payment to be made on such date. Each obligation to pay additional interest shall be deemed to accrue from and including the date of the applicable Registration Default.

7

(c) The parties hereto agree that the additional interest provided for in this Section 3 constitutes a reasonable estimate of and is intended to constitute the sole damages that will be suffered by Holders of Transfer Restricted Securities by reason of the failure of (i) the Shelf Registration Statement or the Exchange Offer Registration Statement to be filed, (ii) the Shelf Registration Statement to be declared effective or to remain effective or (iii) the Exchange Offer Registration Statement to be declared effective and the Registered Exchange Offer to be consummated, in each case to the extent required by this Agreement.

4. Registration Procedures

In connection with any Registration Statement, the following provisions shall apply:

(a) The Company shall (i) furnish to each Initial Purchaser, prior to the filing thereof with the Commission, a copy of the Registration Statement and each amendment thereof and each supplement, if any, to the prospectus included therein and shall use its reasonable best efforts to reflect in each such document, when so filed with the Commission, such comments as any Initial Purchaser may reasonably propose; (ii) include the information set forth in Annex A hereto in the "Exchange Offer Procedures" section and the "Purpose of the Exchange Offer" section and in Annex B hereto in the "Plan of Distribution" section of the prospectus forming a part of the Exchange Offer Registration Statement, and include the information set forth in Annex C hereto in the Letter of Transmittal delivered pursuant to the Registered Exchange Offer; and (iii) if requested by any Initial Purchaser, include the information required by Items 507 or 508 of Regulation S-K, as applicable, in the prospectus forming a part of the Exchange Offer Registration Statement.

(b) The Company shall advise each Initial Purchaser, each Exchanging Dealer and the Holders (if applicable) and, if requested by any such person, confirm such advice in writing (which advice pursuant to clauses (ii)-(v) hereof shall be accompanied by an instruction to suspend the use of the prospectus until the requisite changes have been made):

(i) when any Registration Statement and any amendment thereto has been filed with the Commission and when such Registration Statement or any post-effective amendment thereto has become effective;

(ii) of any request by the Commission for amendments or supplements to any Registration Statement or the prospectus included therein or for additional information; (iii) of the issuance by the Commission of any stop order suspending the effectiveness of any Registration Statement or the initiation of any proceedings for that purpose;

(iv) of the receipt by the Company of any notification with respect to the suspension of the qualification of the Securities, the Exchange Securities or the Private Exchange Securities for sale in any jurisdiction or the initiation or threatening of any proceeding for such purpose; and

 (ν) of the happening of any event that requires the making of any changes in any Registration Statement or the prospectus included therein in order that the statements therein are not misleading and do not omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading.

(c) The Company and the Guarantors will make every reasonable effort to obtain the withdrawal at the earliest possible time of any order suspending the effectiveness of any Registration Statement.

(d) The Company will furnish to each Holder of Transfer Restricted Securities included within the coverage of any Shelf Registration Statement, without charge, at least one conformed copy of such Shelf Registration Statement and any post-effective amendment thereto, including financial statements and schedules and, if any such Holder so requests in writing, all exhibits thereto (including those, if any, incorporated by reference).

(e) The Company will, during the Shelf Registration Period, promptly deliver to each Holder of Transfer Restricted Securities included within the coverage of any Shelf Registration Statement, without charge, as many copies of the prospectus (including each preliminary prospectus) included in such Shelf Registration Statement and any amendment or supplement thereto as such Holder may reasonably request; and the Company consents to the use of such prospectus or any amendment or supplement thereto by each of the selling Holders of Transfer Restricted Securities in connection with the offer and sale of the Transfer Restricted Securities covered by such prospectus or any amendment or supplement thereto.

(f) The Company will furnish to each Initial Purchaser and each Exchanging Dealer, and to any other Holder who so requests, without charge, at least one conformed copy of the Exchange Offer Registration Statement and any post-effective amendment thereto, including financial statements and schedules and, if any Initial Purchaser or Exchanging Dealer or any such Holder so requests in writing, all exhibits thereto (including those, if any, incorporated by reference).

(g) The Company will, during the Exchange Offer Registration Period or the Shelf Registration Period, as applicable, promptly deliver to each Initial Purchaser, each Exchanging Dealer and such other persons that are required to deliver a prospectus following the Registered Exchange Offer, without charge, as many copies of the final prospectus included in the Exchange Offer Registration Statement or the Shelf Registration Statement and any amendment or supplement thereto as such Initial Purchaser, Exchanging Dealer or other persons may reasonably request; and the Company and the Guarantors consent to the use of such prospectus or any amendment or supplement thereto by any such Initial Purchaser, Exchanging Dealer or other persons, as applicable, as aforesaid.

9

(h) Prior to the effective date of any Registration Statement, the Company and the Guarantors will use their reasonable best efforts to register or qualify, or cooperate with the Holders of Securities, Exchange Securities or Private Exchange Securities included therein and their respective counsel in connection with the registration or qualification of, such Securities, Exchange Securities or Private Exchange Securities for offer and sale under the securities or blue sky laws of such jurisdictions as any such Holder reasonably requests in writing and do any and all other acts or things necessary or advisable to enable the offer and sale in such jurisdictions of the Securities, Exchange Securities or Private Exchange Securities covered by such Registration Statement; provided that the Company and the Guarantors will not be required to qualify generally to do business in any jurisdiction where it is not then so qualified or to take any action which would subject it to general service of process or to taxation in any such jurisdiction where it is not then so subject.

(i) The Company and the Guarantors will cooperate with the Holders of Securities, Exchange Securities or Private Exchange Securities to facilitate the timely preparation and delivery of certificates representing Securities, Exchange Securities or Private Exchange Securities to be sold pursuant to any Registration Statement free of any restrictive legends and in such denominations and registered in such names as the Holders thereof may request in writing prior to sales of Securities, Exchange Securities or Private Exchange Securities pursuant to such Registration Statement.

(j) If any event contemplated by Section 4(b)(ii) through (v) occurs during the period for which the Company and the Guarantors are required to maintain an effective Registration Statement, the Company will promptly prepare and file with the Commission a post-effective amendment to the Registration Statement or a supplement to the related prospectus or file any other required document so that, as thereafter delivered to purchasers of the Securities, Exchange Securities or Private Exchange Securities from a Holder, the prospectus will not include an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

(k) Not later than the effective date of the applicable Registration Statement, the Company will provide a CUSIP number for the Securities, the Exchange Securities and the Private Exchange Securities, as the case may be, and provide the applicable trustee with printed certificates for the Securities, the Exchange Securities or the Private Exchange Securities, as the case may be, in a form eligible for deposit with The Depository Trust Company.

(1) The Company and the Guarantors will comply with all applicable rules and regulations of the Commission and will make generally available to its security holders as soon as practicable after the effective date of the applicable Registration Statement an earning statement satisfying the provisions of Section 11(a) of the Securities Act; provided that in no event shall such earning statement be delivered later than 45 days after the end of a 12-month period (or 90 days, if such period is a fiscal year) beginning with the first month of the Company's first fiscal quarter commencing after the effective date of the applicable Registration Statement, which statement shall cover such 12-month period.

(m) The Company and the Guarantors will cause the Indenture or the Exchange Securities Indenture, as the case may be, to be qualified under the Trust Indenture Act as required by applicable law in a timely manner.

(n) The Company may require each Holder of Transfer Restricted Securities to be registered pursuant to any Shelf Registration Statement to furnish to the Company such information concerning the Holder and the distribution of such Transfer Restricted Securities as the Company may from time to time reasonably require for inclusion in such Shelf Registration Statement, and the Company may exclude from such registration the Transfer Restricted Securities of any Holder that fails to furnish such information within a reasonable time after receiving such request.

(o) In the case of a Shelf Registration Statement, each Holder of Transfer Restricted Securities to be registered pursuant thereto agrees by acquisition of such Transfer Restricted Securities that, upon receipt of any notice from the Company pursuant to Section 4(b)(ii) through (v), such Holder will discontinue disposition of such Transfer Restricted Securities until such Holder's receipt of copies of the supplemental or amended prospectus contemplated by Section 4(j) or until advised in writing (the "Advice") by the Company that the use of the applicable prospectus may be resumed. If the Company shall give any notice under Section 4(b)(ii) through (v) during the period that the Company is required to maintain an effective Registration Statement (the "Effectiveness Period"), such Effectiveness Period shall be extended by the number of days during such period from and including the date of the giving of such notice to and including the date when each seller of Transfer Restricted Securities covered by such Registration Statement shall have received (x) the copies of the supplemental or amended prospectus contemplated by Section 4(j) (if an amended or supplemental prospectus is required).

(p) In the case of a Shelf Registration Statement, the Company and the Guarantors shall enter into such customary agreements (including, if requested, an underwriting agreement in customary form) and take all such other action, if any, as Holders of a majority in aggregate principal amount of the Securities, Exchange Securities and Private Exchange Securities being sold or the managing underwriters (if any) shall reasonably request in order to facilitate any disposition of Securities, Exchange Securities or Private Exchange Securities pursuant to such Shelf Registration Statement.

(q) In the case of a Shelf Registration Statement, the Company shall (i) make reasonably available for inspection by a representative of, and Special Counsel (as defined below) acting for, Holders of a majority in aggregate principal amount of the Securities, Exchange Securities and Private Exchange Securities being sold and any underwriter participating in any disposition of Securities, Exchange Securities or Private Exchange Securities pursuant to such Shelf Registration Statement, all relevant financial and other records, pertinent corporate documents and properties of the Company and its subsidiaries as are reasonably requested by such representative or Special Counsel and (ii) use its reasonable best efforts to have its officers, directors, employees, accountants and counsel supply all relevant information reasonably requested by such representative, Special Counsel or any such underwriter in connection with such Shelf Registration Statement.

(r) In the case of a Shelf Registration Statement, the Company shall, if requested by Holders of a majority in aggregate principal amount of the Securities, Exchange Securities and Private Exchange Securities being sold, their Special Counsel or the managing underwriters (if any) in connection with such Shelf Registration Statement, use its reasonable best efforts to cause (i) its counsel to deliver an opinion relating to the Shelf Registration Statement and the Securities, Exchange Securities or Private Exchange Securities, as applicable, in customary form, (ii) its officers to execute and deliver all customary documents and certificates requested by Holders of a majority in aggregate principal amount of the Securities, Exchange Securities and Private Exchange Securities being sold, their Special Counsel or the managing underwriters (if any) and (iii) its independent public accountants to provide a comfort letter or letters in customary form, subject to receipt of appropriate documentation as contemplated, and only if permitted, by Statement of Auditing Standards No. 72.

5. Registration Expenses

The Company and the Guarantors will bear all expenses incurred in connection with the performance of its obligations under Sections 1, 2, 3 and 4 and the Company will reimburse the Initial Purchasers and the Holders for the reasonable fees and disbursements of one firm of attorneys (in addition to any local counsel) chosen by the Holders of a majority in aggregate principal amount of the Securities, the Exchange Securities and the Private Exchange Securities, as the case may be, to be sold pursuant to each Registration Statement (the "Special Counsel") acting for the Initial Purchasers or Holders in connection therewith.

6. Indemnification

(a) In the event of a Shelf Registration Statement or in connection with any prospectus delivery pursuant to an Exchange Offer Registration Statement by an Initial Purchaser or Exchanging Dealer, as applicable, the Company and each of the Guarantors shall jointly and severally indemnify and hold harmless each Holder (including, without limitation, any such Initial Purchaser or Exchanging Dealer), its affiliates, their respective officers, directors, employees, representatives and agents, and each person, if any, who controls such Holder within the meaning of the Securities Act or the Exchange Act (collectively referred to for purposes of this Section 6 and Section 7 as a Holder) from and against any loss, claim, damage or liability, joint or several, or any action in respect thereof (including, without limitation, any loss, claim, damage, liability or action relating to purchases and sales of Securities, Exchange Act, any other federal or state statutory law or regulation, at common law or otherwise, insofar as such loss, claim, damage, liability or action arises out of, or is based upon, (i) any untrue statement or alleged untrue statement of a material fact contained in any such Registration

material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, and shall reimburse each Holder promptly upon demand for any legal or other expenses reasonably incurred by that Holder in connection with investigating or defending or preparing to defend against or appearing as a third party witness in connection with any such loss, claim, damage, liability or action as such expenses are incurred; provided, however, that the Company and the Guarantors shall not be liable in any such case to the extent that any such loss, claim, damage, liability or action arises out of, or is based upon, an untrue statement or alleged untrue statement in or omission or alleged omission from any of such documents in reliance upon and in conformity with any Holders Information; and provided, further, that with respect to any such untrue statement in or omission from any related preliminary prospectus, the indemnity agreement contained in this Section 6(a) shall not inure to the benefit of any Holder from whom the person asserting any such loss, claim, damage, liability or action received Securities, Exchange Securities or Private Exchange Securities to the extent that such loss, claim, damage, liability or action of or with respect to such Holder results from the fact that both (A) a copy of the final prospectus was not sent or given to such person at or prior to the written confirmation of the sale of such Securities, Exchange Securities or Private Exchange Securities to such person and (B) the untrue statement in or omission from the related preliminary prospectus was corrected in the final prospectus unless, in either case, such failure to deliver the final prospectus was a result of non-compliance by the Company with Section 4(d), 4(e), 4(f) or 4(g).

(b) In the event of a Shelf Registration Statement, each Holder shall indemnify and hold harmless the Company, each Guarantor and their respective affiliates, their respective officers, directors, employees, representatives and agents, and each person, if any, who controls the Company or any Guarantor within the meaning of the Securities Act or the Exchange Act (collectively referred to for purposes of this Section 6(b) and Section 7 as the Company), from and against any loss, claim, damage or liability, joint or several, or any action in respect thereof, to which the Company may become subject, whether commenced or threatened, under the Securities Act, the Exchange Act, any other federal or state statutory law or regulation, at common law or otherwise, insofar as such loss, claim, damage, liability or action arises out of, or is based upon, (i) any untrue statement or alleged untrue statement of a material fact contained in any such Registration Statement or any prospectus forming part thereof or in any amendment or supplement thereto or (ii) the omission or alleged omission to state therein a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, but in each case only to the extent that the untrue statement or alleged untrue statement or omission or alleged omission was made in reliance upon and in conformity with any Holders' Information furnished to the Company by such Holder, and shall reimburse the Company for any legal or other expenses reasonably incurred by the Company in connection with investigating or defending or preparing to defend against or appearing as a third party witness in connection with any such loss, claim, damage, liability or action as such expenses are incurred; provided, however, that no such Holder shall be liable for any indemnity claims hereunder in excess of the amount of net proceeds received by such Holder from the sale of Securities, Exchange Securities or Private Exchange Securities pursuant to such Shelf Registration Statement.

13

(c) Promptly after receipt by an indemnified party under this Section 6 of notice of any claim or the commencement of any action, the indemnified party shall, if a claim on the commencement of is to be made against the indemnifying party pursuant to Section 6(a) or 6(b), notify the indemnifying party in writing of the claim or the commencement of that action; provided, however, that the failure to notify the indemnifying party shall not relieve it from any liability which it may have under this Section 6 except to the extent that it has been materially prejudiced (through the forfeiture of substantive rights or defenses) by such failure; and provided, further, that the failure to notify the indemnifying party shall not relieve it from any liability which it may have to an indemnified party otherwise than under this Section 6. If any such claim or action shall be brought against an indemnified party, and it shall notify the indemnifying party thereof, the indemnifying party shall be entitled to participate therein and, to the extent that it wishes, jointly with any other similarly notified indemnifying party, to assume the defense thereof with counsel reasonably satisfactory to the indemnified party. After notice from the indemnifying party to the indemnified party of its election to assume the defense of such claim or action, the indemnifying party shall not be liable to the indemnified party under this Section 6 for any legal or other expenses subsequently incurred by the indemnified party in connection with the defense thereof other than the reasonable costs of investigation; provided, however, that an indemnified party shall have the right to employ its own counsel in any such action, but the fees, expenses and other charges of such counsel for the indemnified party will be at the expense of such indemnified party unless (1) the employment of counsel by the indemnified party has been authorized in writing by the indemnifying party, (2) the indemnified party has reasonably concluded (based upon advice of counsel to the indemnified party) that there may be legal defenses available to it or other indemnified parties that are different from or in addition to those available to the indemnifying party, (3) a conflict or potential conflict exists (based upon advice of counsel to the indemnified party) between the indemnified party and the indemnifying party (in which case the indemnifying party will not have the right to direct the defense of such action on behalf of the indemnified party) or (4) the indemnifying party has not in fact employed counsel reasonably satisfactory to the indemnified party to assume the defense of such action within a reasonable time after receiving notice of the commencement of the action, in each of which cases the reasonable fees, disbursements and other charges of counsel will be at the expense of the indemnifying party or parties. It is understood that the indemnifying party or parties shall not, in connection with any proceeding or related proceedings in the same jurisdiction, be liable for the reasonable fees, disbursements and other charges of more than one separate firm of attorneys (in addition to any local counsel) at any one time for all such indemnified party or parties. Each indemnified party, as a condition of the indemnity agreements contained in Sections 6(a) and 6(b), shall use all reasonable efforts to cooperate with the indemnifying party in the defense of any such action or claim. No indemnifying party shall be liable for any settlement of any such action effected without its written consent (which consent shall not be unreasonably withheld), but if settled with its written consent or if there be a final judgment for the plaintiff in any such action, the indemnifying party agrees to indemnify and hold harmless any indemnified party from and against any loss or liability by reason of such settlement or judgment. No indemnifying party shall, without the prior written consent of the indemnified party (which consent shall not be unreasonably withheld), effect any settlement of any pending or threatened proceeding in respect of which any indemnified party is or could have been a party and indemnity could have been sought hereunder by such indemnified party, unless

such settlement includes an unconditional release of such indemnified party from all liability on claims that are the subject matter of such proceeding.

7. Contribution

If the indemnification provided for in Section 6 is unavailable or insufficient to hold harmless an indemnified party under Section 6(a) or 6(b), then each indemnifying party shall, in lieu of indemnifying such indemnified party, contribute to the amount paid or payable by such indemnified party as a result of such loss, claim, damage or liability, or action in respect thereof, (i) in such proportion as shall be appropriate to reflect the relative benefits received by the Company and the Guarantors from the offering and sale of the Securities, on the one hand, and a Holder with respect to the sale by such Holder of Securities, Exchange Securities or Private Exchange Securities, on the other, or (ii) if the allocation provided by clause (i) above is not permitted by applicable law, in such proportion as is appropriate to reflect not only the relative benefits referred to in clause (i) above but also the relative fault of the Company and the Guarantors on the one hand and such Holder on the other with respect to the statements or omissions that resulted in such loss. claim, damage or liability, or action in respect thereof, as well as any other relevant equitable considerations. The relative benefits received by the Company and the Guarantors on the one hand and a Holder on the other with respect to such offering and such sale shall be deemed to be in the same proportion as the total net proceeds from the offering of the Securities (before deducting expenses) received by or on behalf of the Company, on the one hand, bear to the total proceeds received by such Holder with respect to its sale of Securities, Exchange Securities or Private Exchange Securities, on the other. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to the Company and the Guarantors or information supplied by the Company and the Guarantors on the one hand or to any Holders' Information supplied by such Holder on the other, the intent of the parties and their relative knowledge, access to information and opportunity to correct or prevent such untrue statement or omission. The parties hereto agree that it would not be just and equitable if contributions pursuant to this Section 7 were to be determined by pro rata allocation or by any other method of allocation that does not take into account the equitable considerations referred to herein. The amount paid or payable by an indemnified party as a result of the loss, claim, damage or liability, or action in respect thereof, referred to above in this Section 7 shall be deemed to include, for purposes of this Section 7, any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending or preparing to defend any such action or claim. Notwithstanding the provisions of this Section 7, an indemnifying party that is a Holder of Securities, Exchange Securities or Private Exchange Securities shall not be required to contribute any amount in excess of the amount by which the total price at which the Securities, Exchange Securities or Private Exchange Securities were sold by such indemnifying party to any purchaser exceeds the amount of any damages which such indemnifying party has otherwise paid or become liable to pay by reason of any untrue or alleged untrue statement or omission or alleged omission. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.

8. Rules 144 and 144A

So long as any Transfer Restricted Securities remain outstanding, the Company shall use its reasonable best efforts to file the reports required to be filed by it under Rule 144A(d)(4) under the Securities Act and the Exchange Act in a timely manner and, if at any time the Company is not required to file such reports, it will, upon the written request of any Holder of Transfer Restricted Securities, make publicly available other information so long as necessary to permit sales of such Holder's securities pursuant to Rules 144 and 144A. The Company and the Guarantors covenant that they will take such further action as any Holder of Transfer Restricted Securities may reasonably request, all to the extent required from time to time to enable such Holder to sell Transfer Restricted Securities without registration under the Securities Act within the limitation of the exemptions provided by Rules 144 and 144A (including, without limitation, the requirements of Rule 144A(d)(4)). Upon the written request of any Holder of Transfer Restricted Securities, the Company and the Guarantors shall deliver to such Holder a written statement as to whether it has complied with such requirements. Notwithstanding the foregoing, nothing in this Section 8 shall be deemed to require the Company to register any of its securities pursuant to the Exchange Act.

9. Underwritten Registrations

If any of the Transfer Restricted Securities covered by any Shelf Registration Statement are to be sold in an underwritten offering, the investment banker or investment bankers and manager or managers that will administer the offering will be selected by the Holders of a majority in aggregate principal amount of such Transfer Restricted Securities included in such offering, subject to the consent of the Company (which shall not be unreasonably withheld or delayed), and such Holders shall be responsible for all underwriting commissions and discounts in connection therewith.

No person may participate in any underwritten registration hereunder unless such person (i) agrees to sell such person's Transfer Restricted Securities on the basis reasonably provided in any underwriting arrangements approved by the persons entitled hereunder to approve such arrangements and (ii) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements.

10. Miscellaneous

(a) Amendments and Waivers. The provisions of this Agreement may not be amended, modified or supplemented, and waivers or consents to departures from the provisions hereof may not be given, unless the Company has obtained the written consent of Holders of a majority in aggregate principal amount of the Securities, the Exchange Securities and the Private Exchange Securities, taken as a single class. Notwithstanding the foregoing, a waiver or consent to depart from the provisions hereof with respect to a matter that relates exclusively to the rights of Holders whose Securities, Exchange Securities or Private Exchange Securities are being sold pursuant to a Registration Statement and that does not directly or indirectly affect the rights of other Holders may be given by Holders of a majority in aggregate principal amount of the Securities, the Exchange Securities and the Private Exchange Securities being sold by such Holders pursuant to such Registration Statement.

(b) Notices. All notices and other communications provided for or permitted hereunder shall be made in writing by hand-delivery, first-class mail, telecopier or air courier guaranteeing next-day delivery:

> (1) if to a Holder, at the most current address given by such Holder to the Company in accordance with the provisions of this Section 10(b), which address initially is, with respect to each Holder, the address of such Holder maintained by the Registrar under the Indenture, with a copy in like manner to each Initial Purchaser;

(2) if to an Initial Purchaser, initially at its address set forth in the Purchase Agreement;

(3) if to the Company, initially at the address of the Company set forth in the Purchase Agreement; and

(4) if to the Guarantors, initially at the address of the Guarantors set forth in the Purchase Agreement.

All such notices and communications shall be deemed to have been duly given: when delivered by hand, if personally delivered; one business day after being delivered to a next-day air courier; five business days after being deposited in the mail; and when receipt is acknowledged by the recipient's telecopier machine, if sent by telecopier.

(c) Successors And Assigns. This Agreement shall be binding upon the Company, the Guarantors and their respective successors and assigns.

(d) Counterparts. This Agreement may be executed in any number of counterparts (which may be delivered in original form or by telecopier) and by the parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

(e) Definition of Terms. For purposes of this Agreement, (a) the term "business day" means any day on which the New York Stock Exchange, Inc. is open for trading, (b) the term "subsidiary" has the meaning set forth in Rule 405 under the Securities Act and (c) except where otherwise expressly provided, the term "affiliate" has the meaning set forth in Rule 405 under the Securities Act.

(f) Headings. The headings in this Agreement are for convenience of reference only and shall not limit or otherwise affect the meaning hereof.

(g) Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of New York.

(i) No Inconsistent Agreements. Each of the Company and each Guarantor represents, warrants and agrees that (i) it has not entered into, shall not, on or after the date of this Agreement, enter into any agreement that is inconsistent with the rights granted to the Holders in this Agreement or otherwise conflicts with the provisions hereof, (ii) it has not previously entered into any agreement which remains in effect granting any registration rights with respect to any of its debt securities to any person and (iii) without limiting the generality of the foregoing, without the written consent of the Holders of a majority in aggregate principal amount of the then outstanding Transfer Restricted Securities, it shall not grant to any person the right to request the Company to register any debt securities of the Company under the Securities Act unless the rights so granted are not in conflict or inconsistent with the provisions of this Agreement.

(j) No Piggyback on Registrations. Neither the Company nor the Guarantors nor any of its security holders (other than the Holders of Transfer Restricted Securities in such capacity) shall have the right to include any securities of the Company in any Shelf Registration or Registered Exchange Offer other than Transfer Restricted Securities.

(k) Severability. The remedies provided herein are cumulative and not exclusive of any remedies provided by law. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, illegal, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions set forth herein shall remain in full force and effect and shall in no way be affected, impaired or invalidated, and the parties hereto shall use their reasonable best efforts to find and employ an alternative means to achieve the same or substantially the same result as that contemplated by such term, provision, covenant or restriction. It is hereby stipulated and declared to be the intention of the parties that they would have executed the remaining terms, provisions, covenants and restrictions without including any of such that may be hereafter declared invalid, illegal, void or unenforceable.

Please confirm that the foregoing correctly sets forth the agreement among the Company, the Guarantors and the Initial Purchasers.

Very truly yours,

INTEGRATED ELECTRICAL SERVICES, INC.

By /s/ William W. Reynolds Name: William W. Reynolds Title: Executive Vice President and Chief Financial Officer

1ST GROUP TELECOMMUNICATIONS, INC. ACE ELECTRIC, INC. ALADDIN WARD ELECTRIC & AIR, INC. AMBER ELECTRIC, INC. ANDERSON & WOOD CONSTRUCTION CO., INC. ARC ELECTRIC, INCORPORATED AXIS MANAGEMENT LLC BACHOFNER ELECTRIC, INC. BARTLEY & DEVARY ELECTRIC, INC. BEAR ACQUISITION CORPORATION BRINK ELECTRIC CONSTRUCTION CO. BRITT RICE ELECTRIC, INC. BRITT RICE MANAGEMENT LLC BRYANT ELECTRIC COMPANY, INC. BW CONSOLIDATED, INC. BW/BEC, INC. CANOVA ELECTRICAL CONTRACTING, INC. CARROLL MANAGEMENT LLC CARROLL SYSTEMS, INC. CHARLES P. BAGBY COMPANY, INC. COLLIER ELECTRIC COMPANY, INC. COMMERCIAL ELECTRICAL CONTRACTORS, INC. CROSS STATE ELECTRIC, INC. CYPRESS ELECTRICAL CONTRACTORS, INC. DANIEL ELECTRICAL CONTRACTORS, INC. DANIEL ELECTRICAL OF TREASURE COAST INC. DAVIS ELECTRICAL OF TREASURE COAST INC. DELCO ELECTRICAL CONSTRUCTORS, INC. DELCO ELECTRIC, INC. ELECTRO-TECH, INC. EMC ACQUISITION CORPORATION ERNEST P. BREAUX ELECTRICAL, INC. FEDERAL COMMUNICATIONS GROUP, INC.

FLORIDA INDUSTRIAL ELECTRIC, INC. GENERAL PARTNER, INC. GOSS ELECTRIC COMPANY, INC. H.R. ALLEN, INC. HATFIELD REYNOLDS ELECTRIC COMPANY HOLLAND ELECTRICAL SYSTEMS, INC. HOUSTON-STAFFORD ELECTRIC, INC. HOUSTON-STAFFORD MANAGEMENT LLC HOWARD BROTHERS ELECTRIC CO., INC. I.C.G. ELECTRIC, INC. IES CONTRACTORS MANAGEMENT LLC IES CONTRACTORS MANAGEMENT LLC IES ELECTRICAL GROUP, INC. IES RESIDENTIAL GROUP, INC. IES SPECIALTY LIGHTING, INC. IES VENTURES INC. IES VENIURES INC. INNOVATIVE ELECTRIC COMPANY, INC. INTEGRATED ELECTRICAL FINANCE, INC. INTELLIGENT BUILDING SOLUTIONS, INC. J.W. GRAY ELECTRIC COMPANY, INC. J.W. GRAY MANAGEMENT, LLC KAYTON ELECTRIC, INC. KEY ELECTRICAL SUPPLY, INC. LINEMEN, INC. LINEMEN, INC. MARK HENDERSON, INCORPORATED MENNINGA ELECTRIC, INC. MIDLANDS ELECTRICAL CONTRACTORS, INC. MID-STATES ELECTRIC COMPANY, INC. MILLS ELECTRICAL CONTRACTORS, INC. MILLS MANAGEMENT LLC MITCHELL ELECTRIC COMPANY, INC. M-S SYSTEMS, INC. MURRAY ELECTRICAL CONTRACTORS, INC. MUTH ELECTRIC, INC. NEAL ELECTRIC MANAGEMENT LLC NEW TECHNOLOGY ELECTRICAL CONTRACTORS, INC. NEWCOMB ELECTRIC COMPANY, INC. PAN AMERICAN ELECTRIC COMPANY, INC. PAN AMERICAN ELECTRIC, INC. PAULIN ELECTRIC COMPANY, INC. POLLOCK ELECTRIC INC. PRIMENET, INC. PRIMO ELECTRIC COMPANY PUTZEL ELECTRICAL CONTRACTORS, INC.

```
RAINES ELECTRIC CO., INC.
RAINES MANAGEMENT LLC
RKT ELECTRIC, INC.
ROCKWELL ELECTRIC, INC.
RODGERS ELECTRIC COMPANY, INC.
RON'S ELECTRIC, INC.
SPECTROL, INC.
SPOOR ELECTRIC, INC.
SUMMIT ELECTRIC OF TEXAS, INC.
T&H ELECTRICAL CORPORATION
TECH ELECTRIC CO., INC.
TECHNON ACQUISITION CORPORATION
TESLA POWER G.P., INC.
THOMAS POPP & COMPANY
VALENTINE ELECTRICAL, INC.
WOLFE ELECTRIC CO., INC.
WRIGHT ELECTRICAL CONTRACTING, INC.
By: /s/ Neil J. DePascal, Jr.
   Neil J. DePascal, Jr., Chief Accounting Officer
AXIS COMMUNICATIONS LP
By: Axis Management LLC, its general partner
By: /s/ Neil J. DePascal, Jr.
                                            Neil J. DePascal, Jr., Chief Accounting Officer
B. RICE ELECTRIC LP
By: Britt Rice Management LLC, its general partner
By: /s/ Neil J. DePascal, Jr.
                                         Neil J. DePascal, Jr., Chief Accounting Officer
BEXAR ELECTRIC COMPANY, LTD.
By: BW/BEC, Inc., its general partner
```

By: /s/ Neil J. DePascal, Jr. Neil J. DePascal, Jr., Chief Accounting Officer

```
CARROLL SYSTEMS LP
By: Carroll Management LLC, its general partner
By: /s/ Neil J. DePascal, Jr.
                                . . . . . . . . . . . .
   Neil J. DePascal, Jr., Chief Accounting Officer
HAYMAKER ELECTRIC, LTD.
By: General Partner, Inc., its general partner
By: /s/ Neil J. DePascal, Jr.
                          -----
               -----
   Neil J. DePascal, Jr., Chief Accounting Officer
HOUSTON-STAFFORD ELECTRICAL CONTRACTORS, LP
By: Houston-Stafford Management LLC,
    its general partner
By: /s/ Neil J. DePascal, Jr.
   Neil J. DePascal, Jr., Chief Accounting Officer
ICS INTEGRATED COMMUNICATIONS SERVICES LP
By: Neal Electric Management LLC,
    its general partner
By: /s/ Neil J. DePascal, Jr.
                               Neil J. DePascal, Jr., Chief Accounting Officer
IES CONTRACTORS LP
By: IES Contractors Management LLC,
    its general partner
By: /s/ Neil J. DePascal, Jr.
                                  . . . . . . . . . . . . . . . . . .
```

Neil J. DePascal, Jr., Chief Accounting Officer

```
IES MANAGEMENT LP
By: Integrated Electrical Finance, Inc.
By: /s/ Neil J. DePascal, Jr.
Neil J. DePascal, Jr., Chief Accounting Officer
J.W. GRAY ELECTRICAL CONTRACTORS, LP
By: J.W. Gray Management, LLC, its general partner
```

By: /s/ Neil J. DePascal, Jr. Neil J. DePascal, Jr., Chief Accounting Officer

MILLS ELECTRIC LP

By: Mills Management LLC, its general partner

By: /s/ Neil J. DePascal, Jr. Neil J. DePascal, Jr., Chief Accounting Officer

NEAL ELECTRIC LP

By: BW/BEC, Inc., its general partner

By: /s/ Neil J. DePascal, Jr. Neil J. DePascal, Jr., Chief Accounting Officer

23

```
POLLOCK SUMMIT ELECTRIC LP
By: Pollock Electric, Inc., its general partner
By: /s/ Neil J. DePascal, Jr.
                                  . . . . . . . . . . . . . . . . . . .
   Neil J. DePascal, Jr., Chief Accounting Officer
By: Summit Electric of Texas, Inc.,
its general partner
By: /s/ Neil J. DePascal, Jr.
                                -----
   Neil J. DePascal, Jr., Chief Accounting Officer
RAINES ELECTRIC LP
By: Raines Management LLC, its general partner
By: /s/ Neil J. DePascal, Jr.
   Neil J. DePascal, Jr., Chief Accounting Officer
TESLA POWER AND AUTOMATION, LP
By: Tesla Power GP, Inc., its general partner
By: /s/ Neil J. DePascal, Jr.
                . . . . . . . . . .
   Neil J. DePascal, Jr., Chief Accounting Officer
TESLA POWER PROPERTIES, LP
By: Tesla Power GP, Inc., its general partner
By: /s/ Neil J. DePascal, Jr.
            -----
   Neil J. DePascal, Jr., Chief Accounting Officer
AXIS HOLDINGS LLC
BRITT RICE HOLDINGS LLC
BW/BEC, L.L.C.
CARROLL HOLDINGS LLC
HOUSTON-STAFFORD HOLDINGS LLC
ICS HOLDINGS LLC
```

IES CONTRACTORS HOLDINGS LLC

IES HOLDINGS LLC J.W. GRAY HOLDINGS LLC MILLS ELECTRICAL HOLDINGS LLC POLLOCK SUMMIT HOLDINGS INC. RAINES HOLDINGS LLC By: /s/ Adrianne M. Horne Adrianne M. Horne, President TESLA POWER (NEVADA), INC. DKD ELECTRIC COMPANY, INC. NBH HOLDING CO., INC. By: /s/ Adrianne M. Horne Adrianne M. Horne, Chief Executive Officer

J.P. MORGAN SECURITIES INC., on behalf of the Initial Purchasers

By /s/ IRA GINSBURG

Authorized Signatory

GUARANTORS

LIST OF GUARANTORS

1st Group Telecommunications, Inc. Ace Electric, Inc. Aladdin-Ward Electric & Air, Inc. Amber Electric, Inc. Anderson & Wood Construction Co., Inc. ARC Electric, Incorporated Axis Communications LP Axis Holdings LLC Axis Management LLC B. Rice Electric LP Bachofner Electric, Inc. Bardey & Devary Electric, Inc. Bear Acquisition Corporation Bexar Electric Company, Ltd. (GP is BW/BEC, Inc.) Brink Electric Construction Co. Britt Rice Electric, Inc. Britt Rice Holdings LLC Britt Rice Management LLC Bryant Electric Company, Inc. BW Consolidated, Inc. BW/BEC, Inc. BW/BEC, L.L.C. Canova Electrical Contracting, Inc. Carroll Holdings LLC Carroll Management LLC Carroll Systems LP Carroll Systems LP Carroll Systems, Inc. Charles P. Bagby Co., Inc. Collier Electric Company, Inc. Commercial Electrical Contractors, Inc. Cross State Electric, Inc. Cypress Electrical Contractors, Inc. Daniel Electrical Contractors, Inc. Daniel Electrical of Treasure Coast, Inc. Davis Electrical Constructors, Inc. Delco Electric, Inc. DKD Electric Company, Inc. Electro-Tech, Inc. EMC Acquisition Corporation Ernest P. Breaux Electrical, Inc. Federal Communications Group, Inc. Florida Industrial Electric, Inc. General Partner, Inc. Goss Electric Company, Inc. H. R. Allen, Inc. Hatfield Reynolds Electric Company

Haymaker Electric, Ltd. (GP is General Partner, Inc.) Holland Electrical Systems, Inc. Houston-Stafford Electrical Contractors LP (GP is Houston Stafford Management LLC) Houston-Stafford Electric, Inc. Houston-Stafford Holdings LLC Houston-Stafford Management LLC Howard Brothers Electric Co., Inc. I.C.G. Electric, Inc. ICS Holdings LLC ICS Integrated Communication Services LP (GP is Neal Electric Management LLC) IES Communications Group, Inc. IES Contractors Holdings LLC IES Contractors LP (GP is IES Contractors Management LLC) IES Contractors Management LLC IES Electrical Group, Inc. IES Holdings LLC IES Management LP (GP is Integrated Electrical Finance, Inc.) IES Residential Group, Inc. IES Specialty Lighting, Inc. IES Ventures, Inc. Innovative Electric Company, Inc. Integrated Electric Company, Inc. Intelligent Building Solutions, Inc. J. W. Gray Electric Company, Inc. J. W. Gray Electrical Contractors LP (GP is JW Gray Management LLC) J. W. Gray Holdings LLC J. W. Gray Management LLC Kayton Electric, Inc. Key Electrical Supply, Inc. Linemen, Inc. Mark Henderson, Incorporated Menninga Electric, Inc. Midlands Electrical Contractors, Inc. Mid-States Electric Company, Inc. Mills Electric LP (GP is Mills Management LLC) Mills Electrical Contractors, Inc. Mills Electrical Holdings LLC Mills Management LLC Mitchell Electric Company, Inc. M-S Systems, Inc. Murray Electrical Contractors, Inc. Muth Electric, Inc. NBH Holding Co., Inc Neal Electric LP Neal Electric Management LLC

New Technology Electrical Contractors, Inc. Newcomb Electric Company, Inc. Pan American Electric Company, Inc. Pan American Electric, Inc. Paulin Electric Company, Inc. Pollock Electric Inc. Pollock Summit Electric LP (GPs are Pollock Electric, Inc and Summit Electric of Texas, Inc.) Pollock Summit Holdings, Inc. PrimeNet, Inc. Primo Electric Company Putzel Electrical Contractors, Inc. Raines Electric Co., Inc. Raines Electric LP (GP is Raines Management LLC) Raines Holding LLC Raines Management LLC RKT Electric, Inc. Rockwell Electric, Inc. Rodgers Electric Inc. Ron's Electric, Inc. Spectrol, Inc. Spoor Electric, Inc. Summit Electric of Texas, Incorporated T&H Electrical Corporation Tech Electric Co., Inc. Teknon Acquisition Corporation Tesla Power G.P., Inc. Tesla Power (Nevada), Inc. Tesla Power and Automation, LP Tesla Power Properties, LP Thomas Popp & Company Valentine Electrical, Inc. Wolfe Electric Co., Inc. Wright Electrical Contracting, Inc.

Each broker-dealer that receives Exchange Securities for its own account in exchange for Securities, where such Securities were acquired by such broker-dealer as a result of market-making activities or other trading activities, must acknowledge that it will deliver a prospectus in connection with any resale of such Exchange Securities. See "Plan of Distribution."

PLAN OF DISTRIBUTION

Each broker-dealer that receives Exchange Securities for its own account pursuant to the Registered Exchange Offer must acknowledge that it will deliver a prospectus in connection with any resale of such Exchange Securities. This Prospectus, as it may be amended or supplemented from time to time, may be used by a broker-dealer in connection with resales of Exchange Securities received in exchange for Securities where such Securities were acquired as a result of market-making activities or other trading activities. The Company has agreed that, for a period of 180 days after the Expiration Date, it will make this prospectus, as amended or supplemented, available to any broker-dealer for use in connection with any such resale. In addition, until ______, 200_, all dealers effecting transactions in the Exchange Securities may be required to deliver a prospectus.

The Company will not receive any proceeds from any sale of Exchange Securities by broker-dealers. Exchange Securities received by broker-dealers for their own account pursuant to the Registered Exchange Offer may be sold from time to time in one or more transactions in the over-the-counter market, in negotiated transactions, through the writing of options on the Exchange Securities or a combination of such methods of resale, at market prices prevailing at the time of resale, at prices related to such prevailing market prices or at negotiated prices. Any such resale may be made directly to purchasers or to or through brokers or dealers who may receive compensation in the form of commissions or concessions from any such broker-dealer or the purchasers of any such Exchange Securities. Any broker-dealer that resells Exchange Securities that were received by it for its own account pursuant to the Registered Exchange Offer and any broker or dealer that participates in a distribution of such Exchange Securities may be deemed to be an "underwriter" within the meaning of the Securities Act and any profit on any such resale of Exchange Securities and any commission or concessions received by any such persons may be deemed to be underwriting compensation under the Securities Act. The Letter of Transmittal states that, by acknowledging that it will deliver and by delivering a prospectus, a broker-dealer will not be deemed to admit that it is an "underwriter" within the meaning of the Securities Act.

For a period of 180 days after the Expiration Date the Company will promptly send additional copies of this Prospectus and any amendment or supplement to this Prospectus to any broker-dealer that requests such documents in the Letter of Transmittal. The Company has agreed to pay all expenses incident to the Registered Exchange Offer (including the expenses of one counsel for the Holders of the Securities) other than commissions or concessions of any broker-dealers and will indemnify the Holders of the Securities (including any broker-dealers) against certain liabilities, including liabilities under the Securities Act. [] CHECK HERE IF YOU ARE A BROKER-DEALER AND WISH TO RECEIVE 10 ADDITIONAL COPIES OF THE PROSPECTUS AND 10 COPIES OF ANY AMENDMENTS OR SUPPLEMENTS THERETO.

Name: Address:

If the undersigned is not a broker-dealer, the undersigned represents that it is not engaged in, and does not intend to engage in, a distribution of Exchange Securities. If the undersigned is a broker-dealer that will receive Exchange Securities for its own account in exchange for Securities that were acquired as a result of market-making activities or other trading activities, it acknowledges that it will deliver a prospectus in connection with any resale of such Exchange Securities; however, by so acknowledging and by delivering a prospectus, the undersigned will not be deemed to admit that it is an "underwriter" within the meaning of the Securities Act.

-6-

July 3, 2001

Board of Directors Integrated Electrical Services, Inc. 1800 West Loop South Post Oak Boulevard Suite 500 Houston, Texas 77027

Ladies and Gentlemen:

We have acted as counsel to Integrated Electrical Services, Inc., a Delaware corporation (the "Company") and are delivering this opinion in connection with the Company's Registration Statement on Form S-4 (the "Registration Statement") relating to the registration under the Securities Act of 1933, as amended (the "Securities Act"), of the offer by the Company to exchange up to \$125,000,000 aggregate principal amount of its 9 3/8% Senior Subordinated Notes Due 2009, Series D (the "Exchange Notes") for its existing 9 3/8% Senior Subordinated Notes Due 2009, Series C (the "Existing Notes"). The Exchange Notes are proposed to be issued in accordance with the provisions of the indenture (the "Indenture"), dated as of May 29, 2001, between the Company, the guarantors named therein (the "Guarantors") and State Street Bank and Trust Company, as Trustee.

In arriving at the opinions expressed below, we have examined the Registration Statement, the Prospectus contained therein, the Indenture which is filed as an exhibit to the Registration Statement, and the originals or copies certified or otherwise identified to our satisfaction of such other instruments and other certificates of public officials and officers and representatives of the Company. In such examination, we have assumed and have not verified (i) that the signatures on all documents that we have examined are genuine, (ii) the authenticity of all documents submitted to us as originals, (iii) the conformity with the authentic originals of all documents submitted to us as certified, photostatic or faxed copies, and (iv) that all documents in respect of which forms were filed with the Securities and Exchange Commission as exhibits to the Registration Statement will conform in all material respects to the forms thereof that we have examined. In addition, as the basis for the opinion hereinafter expressed, we have examined such statutes, regulations, corporate $% \left({{{\left[{{{\left[{{{c_{{\rm{s}}}} \right]}} \right]}_{\rm{s}}}}} \right)$ records and documents, certificates of corporate and public officials and other instruments as we have deemed necessary or advisable for the purposes of this opinion.

Based upon the foregoing, having due regard for such legal considerations as we deem relevant, we are of the opinion that the Exchange Notes and the guarantee of each of the Guarantors (the "Guarantees"), (a) when the Notes have been exchanged in the manner described in the Registration Statement, (b) when the Exchange Notes have been duly executed, authenticated, issued and delivered in accordance with the terms of the Indenture, (c) when the Indenture has been duly qualified under the Trust Indenture Act of 1939, as amended, and (d) when applicable provisions of "blue sky" laws have been complied with, will constitute valid and binding obligations of the Company and the Guarantors, as applicable, enforceable against the Company and the Guarantors, as applicable, in accordance with their terms, under the laws of the State of New York which are expressed to govern the same, except as the enforcement thereof may be limited by (a) bankruptcy, insolvency, reorganization, moratorium (including, without limitation, all laws relating to fraudulent transfers), (b) other similar laws relating to or affecting enforcement of creditors' rights generally, (c) general principles of equity (regardless of whether enforcement is considered in a proceeding in equity or at law) and (d) limitations on the waiver of rights under usury laws, and will be entitled to the benefits of the Indenture.

This opinion is limited in all respects to the laws of the State of Texas, the State of New York and the Delaware General Corporation Law. We express no opinion as to, and for the purposes of the opinions set forth herein, we have conducted no investigation of, and do not purport to be experts on, any other laws. With respect to each Guarantor which was not organized under the laws of the State of Texas or the State of Delaware, we have assumed that the laws of the jurisdiction, organization or formation of such Guarantor with respect to matters of authorization, execution and delivery do not differ in any material respect from the laws of the State of Texas in those regards and have undertaken no investigation of the laws of any jurisdiction or their effect, if any, on any legal conclusion herein expressed other than, the laws of the State of Texas, the State of New York and the Delaware General Corporation Law. We hereby consent to the use of this opinion as an exhibit to the Registration Statement.

Very truly yours,

/s/ VINSON & ELKINS L.L.P.

Vinson & Elkins L.L.P.

CREDIT AGREEMENT

Among

INTEGRATED ELECTRICAL SERVICES, INC., as Borrower,

THE FINANCIAL INSTITUTIONS NAMED IN THIS CREDIT AGREEMENT, as Banks,

CREDIT LYONNAIS and THE BANK OF NOVA SCOTIA, as Syndication Agents,

TORONTO DOMINION (TEXAS), INC., as Documentation Agent,

and

THE CHASE MANHATTAN BANK, as Administrative Agent

Dated as of May 22, 2001

Joint Bookrunners and Co-Lead Arrangers:

J.P. Morgan Securities Inc. and Credit Lyonnais Securities

PAGE

ARTICLE 1. DEFINITIONS AND ACCOUNTING TERMS	.1
1.1. Certain Defined Terms	.1
1.2. Computation of Time Periods	
1.3. Accounting Terms; Preparation of Financials	
1.4. Types	
1.5. Interpretation	
ARTICLE 2. CREDIT FACILITIES	20
2.1. Term Loan Facility	20
2.2. Revolving Loan Facility	
2.3. Letter of Credit Facility	
2.4. Swing Line Facility	
2.5. Fees	32
2.6. Interest	
2.7. Breakage Costs	35
2.8. Increased Costs	
2.9. Illegality	
2.10. Market Failure	
2.11. Payment Procedures and Computations	37
2.12. Taxes	39
2.13. Change of Lending Office	40
2.14. Conversion of the Term Loan	
ARTICLE 3. CONDITIONS PRECEDENT	
3.1. Conditions Precedent to Initial Extension of Credit	
3.2. Conditions Precedent to Each Extension of Credit	42
ARTICLE 4. REPRESENTATIONS AND WARRANTIES	
4.1. Organization	
4.2. Authorization	
4.3. Enforceability	
4.4. Absence of Conflicts and Approvals	
4.5. Investment Companies	
4.6. Public Utilities	
4.7. Financial Condition	
4.8. Condition of Assets	
4.9. Litigation	
4.10. Subsidiaries	
4.11. Laws and Regulations	
4.12. Environmental Compliance	
4.13. ERISA	
4.14. Taxes	
4.15. True and Complete Disclosure	
4.16. Senior Indebtedness	46

i

3

PAGE

4.17. Certain Documents
4.18. Shell Subsidiaries46
ARTICLE 5. COVENANTS
5.1. Organization
5.2. Reporting
5.3. Inspection
5.4. Use of Proceeds
5.5. Financial Covenants49
5.6. Debt
5.7. Liens
5.8. Other Obligations50
5.9. Corporate Transactions51
5.10. Distributions51
5.11. Transactions with Affiliates52
5.12. Insurance
5.13. Investments
5.14. Lines of Business53
5.15. Compliance with Laws53
5.16. Environmental Compliance53
5.17. ERISA Compliance53
5.18. Payment of Certain Claims53
5.19. Subsidiaries54
5.20. Changes in Fiscal Periods54
5.21. Negative Pledge Clauses54
5.22. Shell Subsidiaries54
ARTICLE 6. DEFAULT AND REMEDIES
6.1. Events of Default
6.2. Termination of Commitments
6.3. Acceleration of Credit Obligations
6.4. Cash Collateralization of Letters of Credit
6.5. Default Interest
6.6. Right of Setoff
6.7. Actions Under Credit Documents
6.8. Remedies Cumulative
6.9. Application of Payments
ARTICLE 7. THE ADMINISTRATIVE AGENT AND THE ISSUING BANK
7.1. Authorization and Action58
7.2. Reliance, Etc
7.3. Affiliates59
7.4. Bank Credit Decision59
7.5. Expenses
7.6. Indemnification60
7.7. Successor Administrative Agent and Issuing Bank

ii

ARTICLE 8. MISCELLANEOUS	
8.1. Expenses	
8.2. Indemnification	31
8.3. Modifications, Waivers, and Consents	32
8.4. Survival of Agreements	<u> </u> 32
8.5. Assignment and Participation	
8.6. Notice	35
8.7. CHOICE OF LAW	35
8.8. Submission To Jurisdiction; Waivers	35
8.9. WAIVER OF JURY TRIAL	36
8.10. Counterparts	36
8.11. No Further Agreements	<u> 3</u> 6

EXHIBITS

Exhibit /	A	- Form of	Compliance Certificate
Exhibit H	B-1	- Form of	Revolving Loan Borrowing Request
Exhibit H	B-2	- Form of	Term Loan Borrowing Request
Exhibit (С	- Form of	Continuation/Conversion Request
Exhibit [D-1	- Form of	Revolving Loan Note
Exhibit I	D-2	- Form of	Term Loan Note
Exhibit	D-3	- Form of	Swing Line Note
Exhibit H	E	- Form of	Assignment and Acceptance
Exhibit H	F	- Closing	Documents List
Exhibit (G	- Form of	Joinder Agreement
Exhibit H			Acquisition Certificate
Exhibit :	I	- Joinder	Deliveries
Exhibit .	J	- Form of	Borrowing Base Certificate
			5

SCHEDULES

Schedule	I-A	-	Commitments
Schedule	I-B	-	Administrative Information (Borrower; Administrative Agent;
			Banks)
Schedule	II	-	Disclosures (Existing Subsidiaries)
Schedule	III	-	Restricted Payment Terms (Qualified Preferred Stock)
Schedule	IV	-	Good Standing Exceptions

iii

CREDIT AGREEMENT

This Credit Agreement dated as of May 22, 2001, is among Integrated Electrical Services, Inc., a Delaware corporation, as Borrower, the financial institutions named herein, as Banks, Credit Lyonnais and The Bank of Nova Scotia, as syndication agents, Toronto Dominion (Texas), Inc., as documentation agent, and The Chase Manhattan Bank, as administrative agent.

The parties hereto agree as follows:

ARTICLE 1. DEFINITIONS AND ACCOUNTING TERMS

1.1. Certain Defined Terms. As used in this Agreement, the following terms shall have the following meanings (unless otherwise indicated, such meanings to be equally applicable to both the singular and plural forms of the terms defined):

"Acquisition" means the direct or indirect purchase or acquisition, whether in one or more related transactions, of any Person or group of Persons or any related group of assets, liabilities, or securities of any Person or group of Persons.

"Acquisition Certificate" means an acquisition certificate executed by a Responsible Officer of the Borrower in substantially the form of Exhibit H.

"Additional Senior Subordinated Notes" has the meaning set forth in the definition of "Senior Subordinated Notes".

"Administrative Agent" means The Chase Manhattan Bank, together with its affiliates, as the arranger of the Commitments and as the administrative agent for the Banks under this Agreement and the other Credit Documents, together with any of its successors.

"Affiliate" means, as to any Person, any other Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person or any Subsidiary of such Person. The term "control" (including the terms "controlled by" or "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership, by contract, or otherwise.

"Agreement" means this Credit Agreement.

"Annual Aggregate Acquisition Limit" means (i) \$40,000,000 for the fiscal year ending September 30, 2001, (ii) \$60,000,000 for the fiscal year ending September 30, 2002, (iii) \$60,000,000 for the fiscal year ending September 30, 2003, and (iv) \$60,000,000 for the fiscal year ending September 30, 2004.

"Applicable Lending Office" means, with respect to each Bank and for any particular type of transaction, the office of such Bank set forth in Schedule I-B to this Agreement (or in the applicable Assignment and Acceptance by which such Bank joined this Agreement) as its applicable lending office for such type of transaction or such other office of such Bank as such Bank may from time to time specify in writing to the Borrower and the Administrative Agent for such particular type of transaction.

"Applicable Margin" means, with respect to interest rates and letter of credit fees and as of any date of its determination, an amount equal to the percentage amount set forth in the table below opposite the applicable ratio of (a) the consolidated Total Debt of the Borrower as of the end of the fiscal quarter then most recently ended to (b) the consolidated EBITDA of the Borrower for the four fiscal quarters then most recently ended:

Total Debt to EBITDA	Applicable Margin LIBOR Tranches and Letter of Credit Fee	Applicable Margin Prime Rate Tranche
Less than or equal to 1.50	1.75%	0.25%
Greater than 1.50 but less than or equal to 2.00	2.00%	0.50%
Greater than 2.00 but less		
than or equal to 2.50 Greater than 2.50 but less	2.25%	0.75%
than or equal to 3.00	2.50%	1.00%
Greater than 3.00	2.75%	1.25%

The foregoing ratio and resulting Applicable Margin shall be based upon Schedule C of the most recent Compliance Certificate delivered to the Administrative Agent pursuant to Section 5.2(a) or Section 5.2(b) (provided that for the period from the date of this Agreement until the date when the Applicable Margin is reset based upon the Compliance Certificate for the period ending June 30, 2001, the ratio shall be deemed to be 2.01:1.00 and the Applicable Margin shall be set accordingly).

Any adjustments to the Applicable Margin shall become effective on the 45th day following the last day of each fiscal quarter or on the 90th day following the last day of each fiscal year as applicable; provided, however, that if any such Compliance Certificate is not delivered when required hereunder, the Applicable Margin shall be deemed to be the maximum percentage amount in each table from such 45th or 90th day until such Compliance Certificate is received by the Administrative Agent.

Upon any change in the Applicable Margin, the Administrative Agent shall promptly notify the Borrower and the Banks of the new Applicable Margin.

"Assignment and Acceptance" means an Assignment and Acceptance in substantially the form of Exhibit E executed by an assignor Bank, an assignee Bank, and the Administrative Agent, in accordance with Section 8.5.

"Available Revolving Loan Commitments" means an amount equal to the excess, if any, of (a) the Revolving Loan Commitments then in effect over (b) the aggregate outstanding

principal amount of the Revolving Loan Borrowings plus the aggregate outstanding principal amount of the Swing Line Loans plus the Letter of Credit Exposure.

"Bankruptcy Event of Default" has the meaning set forth in Section 6.1(h).

7

"Banks" means the lenders listed as Banks on the signature pages of this Agreement and each Eligible Assignee that shall become a party to this Agreement pursuant to Section 8.5(b).

"Base Rate" means, for any day, the fluctuating rate per annum of interest equal to the greater of (a) the Prime Rate in effect on such day or (b) the Federal Funds Rate in effect on such day plus 0.50%.

"Board" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" means Integrated Electrical Services, Inc., a $\ensuremath{\mathsf{Delaware}}$ corporation.

"Borrowing Base" means, on any date of determination thereof, an amount equal to 66 2/3% of the amount that would be set forth in a consolidated balance sheet of the Borrower and its Subsidiaries for trade accounts receivable, net of allowance for doubtful accounts (determined under United States generally accepted accounting principles), excluding any portion of such accounts receivable retained by a third party, at the date of, and as shown on, the most receivable retained by a third party, at the date of, and as shown on, the most receivable retained by a third party at the date of the Administrative Agent pursuant to Section 3.2(c) or 5.2(l). In the absence of the applicable Borrowing Base Certificate, the Administrative Agent shall determine the Borrowing Base from time to time in its reasonable discretion, taking into account all information reasonably available to it, and the Borrowing Base from time to time so determined shall be the Borrowing Base for all purposes of this Agreement until the applicable Borrowing Base Certificate is furnished to the Administrative Agent.

"Borrowing Base Certificate" means a certificate, substantially in the form of Exhibit J, or in such other form as the Administrative Agent shall from time to time request.

"Borrowings" means the collective reference to the Term Loan Borrowings and the Revolving Loan Borrowings.

"Business Day" means any Monday through Friday during which commercial banks are open for business in New York, New York, and, if the applicable Business Day relates to any LIBOR Tranche, on which dealings are carried on in the London interbank market.

"Capital Expenditures" means, with respect to any Person and any period of its determination, the consolidated expenditures of such Person during such period that are required to be included in or are reflected by the consolidated property, plant, or equipment accounts of such Person, or any similar fixed asset or long term capitalized asset accounts of such Person, on the consolidated balance sheet of such Person in conformity with generally accepted accounting principles.

"Capital Leases" means, with respect to any Person, any lease of any property by such Person which would, in accordance with generally accepted accounting principles, be required to be classified and accounted for as a capital lease on the balance sheet of such Person.

"Capital Lease Obligations" means, as to any Person, the obligations of such Person to pay rent or other amounts under any Capital Lease.

"Change of Control" means, with respect to the Borrower, (a) the direct or indirect acquisition after the date hereof by any Person or related Persons constituting a group of (i) beneficial ownership of issued and outstanding shares of Voting Securities of the Borrower, the result of which acquisition is that such Person or such group possesses 35% or more of the combined voting power of all then-issued and outstanding Voting Securities of the Borrower or (ii) the power to elect, appoint, or cause the election or appointment of at least a majority of the members of the board of directors of the Borrower, or (b) the individuals who, at the beginning of any period of 12 consecutive months, constitute the Borrower's board of directors (together with any new director whose election by the Borrower's board of directors or whose nomination for election by the Borrower's stockholders entitled to vote thereon was approved by a vote of at least a majority of the directors then still in office who either were directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason (other than death or disability) to constitute a majority of the Borrower's board of directors then in office.

"Closing Date" means the date on which the conditions precedent set forth in Section 3.1 shall have been satisfied, which date is May 22, 2001.

"Code" means the Internal Revenue Code of 1986, as amended, or any successor statute.

"Commitment" means, as to any Bank, the sum of the Term Loan Commitment and the Revolving Loan Commitment of such Bank.

"Commonly Controlled Entity" means, with respect to any Person, any other Person which is under common control with such Person within the meaning of Section 414 of the Code.

"Compliance Certificate" means a compliance certificate executed by a Responsible Officer of the Borrower in substantially the form of Exhibit A, including the following attached Schedules:

Schedule A: The applicable financial reports provided under Section 5.2(a) or 5.2(b) ending on the date of the computation of the financial covenants.

Schedule B: A schedule of any adjustments to the financial reports in Schedule A to the Compliance Certificate, listed on a company-by-company basis, that are requested by the Borrower to reflect the financial results of Acquisitions made prior to the end of the applicable period, together with the supporting financial reports of the Acquisitions from which the Borrower prepared such adjustments, prepared in accordance with Section 1.3(c) and otherwise in a form acceptable to the Administrative Agent.

Schedule C: The computation of the financial covenants under this Agreement that are based upon the financial reports in Schedule A or Schedule B, as applicable, to the Compliance Certificate, in a form acceptable to the Administrative Agent.

"Continuation/Conversion Request" means a Continuation/Conversion Request in substantially the form of Exhibit C executed by a Responsible Officer of the Borrower and delivered to the Administrative Agent.

"Contract Status Report" means a report, in form and substance acceptable to the Administrative Agent, detailing the status of each contract of any Restricted Entity which contract has a value equal to or greater than \$7,500,000.

"Credit Documents" means this Agreement, the Term Loan Notes, the Revolving Loan Notes, the Swing Line Note, the Letter of Credit Documents, the Guaranty, the Security Documents, the Interest Hedge Agreements, and each other agreement, instrument, or document executed at any time in connection with this Agreement.

"Credit Obligations" means all principal and interest (including interest accruing after the maturity of the loans made hereunder and interest accruing after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to the Borrower, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding) and fees, reimbursements, indemnifications, and other amounts now or hereafter owed by the Borrower to the Administrative Agent and the Banks (or with respect to the Interest Hedge Agreements, any Affiliates of the Banks) under this Agreement, the Term Loan Notes, the Revolving Loan Notes, the Swing Line Note, the Letter of Credit Documents, and the other Credit Documents and any increases, extensions, and rearrangements of those obligations under any amendments, supplements, and other modifications of the documents and agreements creating those obligations.

"Credit Parties" means the Borrower and the Guarantors.

"Debt" means, with respect to any Person, without duplication, (a) indebtedness of such Person for borrowed money, (b) obligations of such Person evidenced by bonds, debentures, notes, or other similar instruments, (c) obligations of such Person to pay the deferred purchase price of property or services (other than trade debt and normal operating liabilities incurred in the cordinary course of business), (d) obligations of such Person as lessee under Capital Leases, (e) obligations of such Person under or relating to letters of credit, guaranties, purchase agreements, or other creditor assurances, in each case, assuring a creditor against loss in respect of indebtedness or obligations of others of the kinds referred to in clauses (a) through (d) of this definition, (f) nonrecourse indebtedness or obligations of others of the kinds referred to in clauses (a) through (e) of this definition secured by any Lien on or in respect of any property of such Person, and (g) obligations of such Person evidenced by preferred stock or other equity interests in such Person which provide for mandatory redemption, mandatory payment of dividends, or similar rights to the payment of money. For the purposes of determining the amount of any Debt, the amount of any Debt described in clause (e) of the definition of Debt shall be valued at the maximum amount of the contingent liability thereunder, the amount of any Debt described in clause (f) that is not covered by clause (e) shall be valued at the lesser of the

amount of the Debt secured or the book value of the property securing such Debt, and the amount of any Debt described in clause (g) shall be valued at the stated redemption value of such Debt as of the date of determination.

"Default" means (a) an Event of Default or (b) any event or condition which with notice or lapse of time or both would, unless cured or waived, become an Event of Default.

"Default Rate" means, with respect to any amount due hereunder, a per annum interest rate equal to (a) if such amount is either outstanding principal accruing interest based upon a rate established elsewhere in this Agreement or accrued but unpaid interest thereon, the sum of (i) the interest rate established elsewhere in this Agreement from time to time for such principal amount, including any Applicable Margin, plus (ii) 2.00% per annum or (b) in all other cases, the Base Rate in effect from time to time plus the Applicable Margin for the Prime Rate Tranche in effect from time to time plus 2.00% per annum.

"Derivatives" means any swap, hedge, cap, collar, or similar arrangement providing for the exchange of risks related to price changes in any commodity, including money.

"Dollars or \$" means lawful money of the United States of America.

"EBIT" means, with respect to any Person and for any period of its determination, the consolidated net income (excluding any extraordinary gains or losses) of such Person for such period, plus the consolidated interest expense and income taxes of such Person for such period.

"EBITDA" means, with respect to any Person and for any period of its determination, the consolidated net income (excluding any extraordinary gains or losses) of such Person for such period, plus the consolidated interest expense and income taxes of such Person for such period, plus the consolidated depreciation and amortization of such Person for such period.

"Eligible Assignee" means, with respect to any assignment hereunder, at the time of such assignment, any commercial bank organized under the laws of the United States or any of the countries parties to the Organization for Economic Cooperation and Development or any political subdivision of any thereof which has primary capital (or its equivalent) of not less than \$250,000,000, is approved by the Administrative Agent, the Swing Line Lender and the Issuing Bank, and, so long as no Default or Event of Default exists, is approved by the Borrower, in each case, such approval not to be unreasonably withheld.

"Environmental Law" means all federal, state, and local laws, rules, regulations, ordinances, orders, decisions, agreements, and other requirements now or hereafter in effect relating to the pollution, destruction, loss, or injury of the environment, the presence of any contaminant in the environment, the protection, cleanup, remediation, or restoration of the environment, the creation, handling, transportation, use, or disposal of any waste product in the environment, exposure of persons to any contaminant, waste, or hazardous substance in the environment, and the health and safety of employees in relation to their environment.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Eurocurrency Reserve Requirements" means for any day as applied to a LIBOR Tranche, the aggregate (without duplication) of the maximum rates (expressed as a decimal fraction) of reserve requirements in effect on such day (including basic, supplemental, marginal and emergency reserves) under any regulations of the Board or other governmental authority having jurisdiction with respect thereto dealing with reserve requirements prescribed for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of the Board) maintained by a member bank of the Federal Reserve System.

"Eurodollar Base Rate" means, for any LIBOR Tranche for any Interest Period therefor, the rate per annum determined on the basis of the rate for deposits in Dollars for a period equal to such Interest Period commencing on the first day of such Interest Period appearing on Page 3750 of the Telerate screen as of 11:00 A.M., London time, two Business Days prior to the beginning of such Interest Period. In the event that such rate does not appear on Page 3750 of the Telerate screen (or otherwise on such screen), the "Eurodollar Base Rate" shall be determined by reference to such other comparable publicly available service for displaying eurodollar rates as may be selected by the Administrative Agent or, in the absence of such availability, by reference to the rate at which the Administrative Agent is offered Dollar deposits at or about 11:00 A.M., New York City time, two Business Days prior to the beginning of such Interest Period in the interbank eurodollar market where its eurodollar and foreign currency and exchange operations are then being conducted for delivery on the first day of such Interest Period for the number of days comprised therein.

"Event of Default" has the meaning specified in Section 6.1.

"Existing Credit Agreement" means the Credit Agreement dated as of January 30, 1998, among the Borrower, the financial institutions named therein and NationsBank of Texas, N.A., as agent, as amended, supplemented or otherwise modified from time to time.

"Facilities" means each of (a) the Term Loan Commitments and the Term Loan made thereunder and (b) the Revolving Loan Commitments and the extensions of credit made thereunder.

"Federal Funds Rate" means, for any period, a fluctuating per annum interest rate equal for each day during such period to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for any such day on such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by it.

"Federal Reserve Board" means the Board of Governors of the Federal Reserve System or any of its successors.

"Financial Statements" means the audited annual final statements of the Borrower for the period ending September 30, 2000, including the consolidated balance sheets of the Borrower as of the end of such fiscal year and the consolidated statements of income, stockholders' equity,

and cash flows for such fiscal year, as contained in the Form 10-K filed by the Borrower with the Securities and Exchange Commission for the period ending on such date.

"Guaranty" means the Guaranty dated as of May 22, 2001, made by the Subsidiaries of the Borrower in favor of the Administrative Agent guaranteeing the Credit Obligations.

"Guarantors" means (a) the Subsidiaries of the Borrower that have executed the Guaranty in connection with the execution of this Agreement and (b) any future Subsidiaries of the Borrower that join the Guaranty pursuant to Section 5.19.

"Hazardous Materials" means any substance or material identified as a hazardous substance pursuant to the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended and as now or hereafter in effect; any substance or material regulated as a hazardous waste pursuant to the Resource Conservation and Recovery Act of 1976, as amended and as now or hereafter in effect; and any substance or material designated as a hazardous substance or hazardous waste pursuant to any other Environmental Law.

"Highest Lawful Rate" means the maximum lawful interest rate, if any, that at any time or from time to time may be contracted for, charged, or received under the laws applicable to the relevant Bank which are presently in effect or, to the extent allowed by law, under such applicable laws which may hereafter be in effect and which allow a higher maximum nonusurious interest rate than applicable laws now allow.

"Interest Expense" means, with respect to any Person and for any period of its determination, the consolidated interest expense of such Person during such period.

"Interest Hedge Agreements" means any swap, hedge, cap, collar, or similar arrangement between the Borrower and any Bank (or any Affiliate of any Bank) providing for the exchange of risks related to price changes in the interest rate on the Loan Advances under this Agreement.

"Interest Period" means, with respect to each LIBOR Tranche, the period commencing on the date of such LIBOR Tranche and ending on the last day of the period selected by the Borrower pursuant to the provisions below. The duration of each such Interest Period shall be one, two, three, or six months, in each case as the Borrower may select in the applicable Term Loan Borrowing Request, Revolving Loan Borrowing Request or Continuation/Conversion Request (unless there shall exist any Default or Event of Default, in which case the Borrower may only select one month Interest Periods); provided, however, that:

> (a) whenever the last day of any Interest Period would otherwise occur on a day other than a Business Day, the last day of such Interest Period shall be extended to occur on the next succeeding Business Day; provided that if such extension would cause the last day of such Interest Period to occur in the next following calendar month, the last day of such Interest Period shall occur on the next preceding Business Day;

> (b) any Interest Period which begins on the last Business Day of the calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the

calendar month in which it would have ended if there were a numerically corresponding day in such calendar month; and

(c) the Borrower may not select an Interest Period for any LIBOR Tranche which ends after the Revolving Loan Maturity Date or after the date final payment is due on the Term Loan, as the case may be.

"Interim Financial Statements" means the consolidated financial statements of the Borrower dated as of March 31, 2001, including the consolidated balance sheets of the Borrower as of the end of such fiscal quarter and the consolidated statements of income and cash flows for such fiscal quarter and for the fiscal year to date period ending on the last day of such fiscal quarter.

"Issuing Bank" means The Chase Manhattan Bank and any successor issuing bank pursuant to Section 7.7.

"Joinder Agreement" has the meaning set forth in Section 5.19.

"Letter of Credit" means any commercial or standby letter of credit issued by the Issuing Bank for the account of the Borrower pursuant to the terms of this Agreement.

"Letter of Credit Application" means the Issuing Bank's standard form letter of credit application for either a commercial or standby letter of credit, as the case may be, which has been executed by the Borrower and accepted by the Issuing Bank in connection with the issuance of a Letter of Credit.

"Letter of Credit Application Amendment" means the Issuing Bank's standard form application to amend a letter of credit for either a commercial or standby letter of credit, as the case may be, which has been executed by the Borrower and accepted by the Issuing Bank in connection with the increase or extension of a Letter of Credit.

"Letter of Credit Collateral Account" means a special cash collateral account pledged to the Administrative Agent containing cash deposited pursuant to Section 2.2(f), 2.3(d) or 6.4 to be maintained with the Administrative Agent in accordance with Section 2.3(g).

"Letter of Credit Documents" means all Letters of Credit, Letter of Credit Applications, Letter of Credit Application Amendments, and agreements, documents, and instruments entered into in connection with or relating thereto.

"Letter of Credit Exposure" means, as of any date of its determination, the aggregate outstanding undrawn amount of all then outstanding Letters of Credit plus the aggregate then outstanding amount of the reimbursement obligations of the Borrower under the Letter of Credit Applications and this Agreement.

"Letter of Credit Sublimit" means \$25,000,000.

"LIBOR" means, for any LIBOR Tranche for any Interest Period therefor, a rate per annum determined in accordance with the following formula (rounded upward to the nearest 1/100th of 1%):

Eurodollar Base Rate 1.00 - Eurocurrency Reserve Requirements

"LIBOR Tranche" means any Tranche which bears interest based upon the LIBOR, as determined in accordance with Section 2.6.

"Lien" means any mortgage, lien, pledge, charge, deed of trust, security interest, encumbrance, or other type of preferential arrangement to secure or provide for the payment of any obligation of any Person, whether arising by contract, operation of law, or otherwise (including any title retention for such purposes under any conditional sale agreement, any Capital Lease, or any other title transfer or retention agreement).

"Loan Advances" means the collective reference to the Term Loan Advances and the Revolving Loan Advances.

"Majority Banks" means, at any time, Banks holding more than 50% of the sum of (a) the then aggregate unpaid principal amount of the Term Loan Notes held by the Banks (or, if no such principal amount is then outstanding, the aggregate amount of the Term Loan Commitments at such time) and (b) the aggregate amount of the Revolving Loan Commitments at such time (or, if the Revolving Loan Commitments have been terminated, the then aggregate unpaid principal amount of the Revolving Loan Notes held by the Banks and the Letter of Credit Exposure of the Banks at such time).

"Material Adverse Change" means any material adverse change in the business, assets, liabilities (actual or contingent), operations or condition (financial or otherwise) of the Borrower and its Subsidiaries on a consolidated basis.

"Net Worth" means, with respect to any Person and as of any date of its determination, the excess of (a) the assets of such Person over (b) the liabilities of such Person, provided that, notwithstanding anything herein to the contrary, for the purposes of calculating Net Worth, adjustments to goodwill in accordance with the February 14, 2001 FASB Exposure Draft relating to Business Combinations and Intangible Assets - Accounting for Goodwill, as adopted, shall be disregarded.

"Other Taxes" has the meaning set forth in Section 2.12(b).

"Partnership Pledge Agreement" means the Pledge Agreement dated as of May 22, 2001, made by the respective limited and general partners of each limited partnership Subsidiary of the Borrower, in favor of the Administrative Agent granting the Administrative Agent a security interest in the partnership interests of such partners in such limited partnerships, to secure the Credit Obligations.

"PBGC" means Pension Benefit Guaranty Corporation or its successor.

"Permitted Debt" means all of the following Debt:

(a) Debt in the form of the Credit Obligations;

(b) Debt in the form of indebtedness for borrowed money and letters of credit owed by any Subsidiary of the Borrower prior to the acquisition of such Subsidiary by the Borrower in an Acquisition transaction, or owed by any Person that is the subject of any Acquisition assumed by the Borrower or any Subsidiary of the Borrower in connection with such Acquisition, provided that with respect to any such indebtedness, arrangements satisfactory to the Administrative Agent for the repayment of such indebtedness within 90 days following the closing of the Acquisition are made prior to the closing of the Acquisition and such arrangements are executed;

(c) Debt in the form of (i) purchase money indebtedness and Capital Leases, (ii) indebtedness for borrowed money and letters of credit owed by any Subsidiary of the Borrower prior to the acquisition of such Subsidiary by the Borrower in an Acquisition transaction, or owed by any Person that is the subject of any Acquisition assumed by the Borrower or any Subsidiary of the Borrower in connection with such Acquisition, and (iii) other Debt, which Debt under clauses (i), (ii), and (iii) together are in an aggregate outstanding amount not to exceed \$20,000,000;

(d) Debt in the form of Subordinated Debt and the Senior Subordinated Notes;

(e) Debt in the form of Qualified Preferred Stock; and

(f) Debt in the form of reimbursement obligations for performance bonds issued in the ordinary course of business.

"Permitted Investments" means all of the following investments:

(a) investments (including investments in the form of loans) in wholly owned Subsidiaries of the Borrower;

(b) investments in the form of loans, guaranties, open accounts, and other extensions of trade credit in the ordinary course of business;

(c) investments in commercial paper, bankers' acceptances, loan participation agreements, and other similar investments, in each case, maturing in twelve months or less from the date of issuance and which, at the time of acquisition are rated A-2 or better by Standard & Poor's Ratings Services and P-2 or better by Moody's Investors Service, Inc;

(d) investments in direct obligations of the United States, or investments in any Person which investments are guaranteed by the full faith and credit of the United States, in either case maturing in twelve months or less from the date of acquisition thereof and repurchase agreements having a term of less than one year and fully collateralized by such obligations which are entered into with banks or trust companies described in clause (e) below or brokerage companies having net worth in excess of \$250,000,000;

(e) investments in time deposits or certificates of deposit maturing within one year from the date such investment is made, issued by a bank or trust company organized under the laws of the United States or any state thereof having capital, surplus, and undivided profits aggregating at least \$250,000,000 or a foreign branch thereof and whose long-term certificates of deposit are, at the time of acquisition thereof, rated A-2 by Standard & Poor's Ratings Services or Prime-2 by Moody's Investors Service, Inc.;

(f) investments in money market funds which invest solely in the types of investments described in paragraphs (c) through (e) above;

(g) loans and advances to directors, officers, and employees of the Credit Parties made in the ordinary course of business in an aggregate outstanding amount not to exceed 1,000,000; and

(h) other investments (including loans and advances) during any fiscal year of the Borrower not to exceed 5% of the consolidated Net Worth of the Borrower as of the last day of the immediately preceding fiscal year.

In valuing any investments for the purpose of applying the limitations set forth in this Agreement, such investments shall be taken at the original cost thereof (but without reduction for any subsequent appreciation or depreciation thereof) less any amount actually repaid or recovered on account of capital or principal (but without reduction for any offsetting investments made by the investee in the investor).

"Permitted Liens" means all of the following Liens:

(a) Liens securing the Credit Obligations;

(b) Liens securing purchase money debt, Capital Leases, and assumed or acquired indebtedness for borrowed money and letters of credit permitted under clause (b) of the definition of Permitted Debt provided that no such Lien is spread to cover any property not (i) purchased in connection with the incurrence of such Debt, in the case of purchase money debt, or (ii) covered by such Lien at the time of the assumption or acquisition of the indebtedness secured thereby, in the case of assumed or acquired indebtedness for borrowed money and letters of credit; and

(c) Liens arising in the ordinary course of business which are not incurred in connection with the borrowing of money, the obtaining of advances or credit, or payment of legal judgments and which do not materially detract from the value of any Restricted Entity's assets or materially interfere with any Restricted Entity's business, including such (i) Liens for taxes, assessments, or other governmental charges or levies not yet due or that are being contested in good faith by appropriate proceedings, provided that adequate reserves with respect thereto are maintained on the books of the Borrower or its Subsidiaries; (ii) Liens in connection with worker's compensation, unemployment insurance, or other social security, old age pension, or public liability obligations not yet due or that are being contested in good faith by appropriate proceedings; (iii) Liens in the form of legal or equitable encumbrances deemed to exist by reason of negative pledge covenants and other covenants or undertakings of like nature; (iv) Liens in the form of vendors', carriers', warehousemen's, repairmen's, mechanics', workmen's, materialmen's, construction, or other like Liens arising by operation of law in the ordinary course of business or incident to the construction or improvement of any property, including liens on such property securing reimbursement obligations for performance bonds issued in the ordinary course of business; and (v) Liens in the form of zoning restrictions, easements, licenses, and other restrictions on the use of real property or minor irregularities in title thereto which do not materially impair the use of such property in the operation of the business of the applicable Restricted Entity or the value of such property.

"Person" means an individual, partnership, corporation (including a business trust), joint stock company, trust, unincorporated association, joint venture, or other entity, or a government or any political subdivision or agency thereof, or any trustee, receiver, custodian, or similar official.

"Plan" means any (a) employee welfare benefit plan under Section 3(1) of ERISA, (b) employee pension benefit plan under Section 3(2) of ERISA, (c) multiemployer plan under Section 4001(a)(3) of ERISA, and (d) employee individual account benefit plan under Section 3(2) of ERISA, in each case, in which each Restricted Entity or any of their respective Commonly Controlled Entities is or would be deemed to be an "employer" as defined in Section 3(5) of ERISA.

"Pledge Agreements" means, collectively (i) the Stock Pledge Agreement, (ii) the Partnership Pledge Agreement, and (iii) any future agreements granting the Administrative Agent a security interest in capital stock, partnership interests, or other membership interests of any Subsidiary (direct or indirect) of the Borrower, in each case, to secure the Credit Obligations.

"Prime Rate" means, for any day, the fluctuating per annum interest rate in effect on such day equal to the rate of interest publicly announced by the Administrative Agent as its prime rate, whether or not the Borrower has notice thereof.

"Prime Rate Borrowing" shall mean that portion of any Borrowing which bears interest based upon the Base Rate as determined in accordance with Section 2.6.

"Prime Rate Tranche" shall mean the Tranche which bears interest based upon the Base Rate, as determined in accordance with Section 2.6.

"Prohibited Transaction" means any transaction set forth in Section 406 of ERISA or Section 4975 of the Code.

"Projections" has the meaning set forth in Section 5.2(f).

"Qualified Preferred Stock" means, with respect to the Borrower and as of any date of its issuance, any shares of preferred stock of the Borrower that (a) are issued after the date of this Agreement, (b) provide for their mandatory redemption on a date, if at all, that is on or after the first anniversary of the latest maturity of any of the Credit Obligations at the time issued, and (c) provide in the applicable certificate of designation for the redemption of such shares and for the blockage of Restricted Payments in respect of such shares during the existence of a Default or an

Event of Default (i) on the terms and conditions set forth on Schedule III, or (ii) on terms approved by the Administrative Agent and the Majority Banks in their sole discretion, all such provisions to be in form and content satisfactory to the Administrative Agent and the Majority Banks in their sole discretion.

"ratable share" or "pro rata share" means, with respect to any Bank and as of any date of its determination, either (a) until the Closing Date, the ratio of such Bank's Commitment at such time to the aggregate Commitments at such time or (b) thereafter, the ratio of such Bank's aggregate outstanding Term Loan Advance and/or such Bank's Revolving Loan Commitment at such time (or, if the Revolving Loan Commitments have been terminated, Revolving Loan Advances and share of the Letter of Credit Exposure at such time) to the aggregate outstanding Term Loan Advances and/or Revolving Loan Commitments at such time (or, if the Revolving Loan Commitments have been terminated, Revolving Loan Advances and Letter of Credit Exposure at such time), in each case as the context may require.

"Refunded Swing Line Loans" has the meaning set forth in Section 2.4(c)(i).

"Refunding Date" has the meaning set forth in Section 2.4(c)(ii).

"Related Parties" means, with respect to any Person, such Person's stockholders, directors, officers, employees, agents, Affiliates, successors, and assigns, and their respective stockholders, directors, officers, employees, and agents, and, with respect to any Person that is an individual, such Person's family relations and heirs.

"Reportable $\ensuremath{\mathsf{Event}}\xspace$ means any of the events set forth in Section 4043 of $\ensuremath{\mathsf{ERISA}}\xspace.$

"Responsible Officer" means, with respect to any Person, such Person's Chief Executive Officer, President, Chief Financial Officer, Secretary, Chief Accounting Officer, or any other officer of such Person designated by any of the foregoing in writing from time to time.

"Restricted Entities" means the Borrower and each Subsidiary of the Borrower.

"Restricted Payment" means the declaration or making by any Person of any (a) dividends; (b) purchase, redemption, retirement, or other acquisition for value any of its capital stock now or hereafter outstanding, or any distribution of assets to its stockholders as such, whether in cash, assets, or in obligations of it; (c) allocation or other setting apart of any sum for the payment of any dividend or distribution on, or for the purchase, redemption, or retirement of, any shares of its capital stock; or (d) making of any other distribution by reduction of capital or otherwise in respect of any shares of its capital stock; in each case, other than any such dividends, distributions, and payments payable solely in such Person's common stock.

"Revolving Loan" means the aggregate outstanding principal amount of the Revolving Loan Borrowings.

"Revolving Loan Advance" means the outstanding principal from a Bank which represents such Bank's ratable share of a Revolving Loan Borrowing. "Revolving Loan Borrowing" means any aggregate amount of principal advanced on the same day and pursuant to the same Revolving Loan Borrowing Request under the revolving loan facility created in Section 2.2.

"Revolving Loan Borrowing Request" means a Revolving Loan Borrowing Request in substantially the form of Exhibit B-1 executed by a Responsible Officer of the Borrower and delivered to the Administrative Agent.

"Revolving Loan Commitment" means, for any Bank, the amount set forth opposite such Bank's name on Schedule I-A as its Revolving Loan Commitment, or if such Bank has entered into any Assignment and Acceptance since the date of this Agreement, as set forth for such Bank as its Revolving Loan Commitment in the Register maintained by the Administrative Agent pursuant to Section 8.5(c), in each case as such amount may be increased pursuant to Section 2.14 or terminated pursuant to Section 6.2.

"Revolving Loan Maturity Date" means May 22, 2004.

"Revolving Loan Note" means a promissory note of the Borrower payable to the order of a Bank, in substantially the form of Exhibit D-1, evidencing the indebtedness of the Borrower to such Bank resulting from Revolving Loan Advances made by such Bank to the Borrower.

"Revolving Loan Percentage" means, as to any Bank at any time, the percentage which such Bank's Revolving Loan Commitment then constitutes of the aggregate Revolving Loan Commitments or, at any time after the Revolving Loan Commitments shall have expired or terminated, the percentage which the aggregate principal amount of such Lender's Revolving Loan Advances then outstanding constitutes of the aggregate principal amount of the Revolving Loan Advances then outstanding.

"Security Agreement" means the Security Agreement dated as of May 22, 2001, made by the Borrower and the Subsidiaries of the Borrower in favor of the Administrative Agent granting the Administrative Agent a security interest in the receivables of each such Credit Party to secure the Credit Obligations.

"Security Documents" means the Security Agreement, the Pledge Agreements, and any other document creating or consenting to Liens in favor of the Administrative Agent securing Credit Obligations.

"Senior Debt" means all Debt of the Borrower and the Subsidiaries of the Borrower other than Debt in the form of Subordinated Debt and the Senior Subordinated Notes.

"Senior Subordinated Note Indentures" means: (i) the Indenture, dated as of January 28, 1999, among the Borrower, certain of its Subsidiaries and State Street Bank and Trust Company, as trustee, together with all instruments and other agreements entered into by the Borrower or such Subsidiaries in connection therewith, and (ii) the indenture to be entered into by the Borrower and/or any of its Subsidiaries in connection with the issuance of the Additional Senior Subordinated Notes, together with all instruments and other agreements entered into by the Borrower and/or such Subsidiaries in connection therewith.

"Senior Subordinated Notes" means: (i) the outstanding \$150,000,000 9 3/8% Senior Subordinated Notes due 2009, Series B, issued by the Borrower, (together with the subsidiary guarantees thereof), and (ii) any additional Senior Subordinated Notes (the "Additional Senior Subordinated Notes"), in an aggregate principal amount not to exceed \$125,000,000, which may hereafter be issued by the Borrower (together with any subsidiary guarantees thereof); provided however that such Additional Senior Subordinated Notes have terms, maturities, covenants, and subordination substantially identical to the terms, maturities, covenants, and subordination of the Senior Subordinated Notes described in part (i) above, together with such changes thereto as the Administrative Agent shall approve.

"Shell Subsidiary" means any Subsidiary of the Borrower that is listed as a "Shell Subsidiary" on Schedule II.

"Stock Pledge Agreement" means the Pledge Agreement dated as of May 22, 2001, made by Borrower and certain Subsidiaries of Borrower in favor of the Administrative Agent granting the Administrative Agent a security interest in the capital stock or membership interests, as applicable, of each Subsidiary (whether direct or indirect) of Borrower, to secure the Credit Obligations.

"Subordinated Debt" means, with respect to the Borrower and as of any date of its issuance, any unsecured indebtedness for borrowed money, other than the Senior Subordinated Notes, for which the Borrower is directly and primarily obligated that (a) arises after the date of this Agreement, (b) does not have any stated maturity before the date six months following the latest maturity of any of the Credit Obligations at the time incurred, (c) has terms that are no more restrictive than the terms of the Credit Documents, which terms shall in any event (i) prohibit the acceleration of such indebtedness without 5 days' prior written notice to the Administrative Agent, (ii) not include a cross-default provision to other indebtedness (other than to a principal payment default of such other indebtedness), (iii) not include any financial covenants other than the type described in Section 5.5 and (iv) with respect any such financial covenants, be at least 20% less restrictive on the Borrower and its Subsidiaries than those contained herein, and (d) is expressly subordinated to the Credit Obligations (i) on the terms and conditions no less restrictive on the holders thereof than the subordination terms applicable to the Senior Subordinated Notes, or (ii) on terms approved by the Administrative Agent and the Majority Banks in their sole discretion, including payment subordination, remedy subordination, and related terms satisfactory to the Administrative Agent and the Majority Banks in their sole discretion.

"Subordinated Debt Event" means the date on which the Borrower issues Additional Senior Subordinated Notes or Subordinated Debt in an aggregate principal amount of at least \$75,000,000.

"Subordinated Debt Indenture" means any indenture or like agreement entered into by the Borrower in connection with the issuance of Subordinated Debt, together with all instruments and other agreements entered into by the Borrower in connection therewith. "Subsidiary" means, with respect to any Person, any other Person, a majority of whose outstanding Voting Securities (other than directors' qualifying shares) shall at any time be owned by such Person or one or more Subsidiaries of such person.

"Swing Line Borrowing" means any aggregate amount of principal advanced on the same day and pursuant to the same notice under the swing line facility created in Section 2.4.

"Swing Line Commitment" means the obligation of the Swing Line Lender to make Swing Line Loans pursuant to Section 2.4 in an aggregate principal amount at any one time outstanding not to exceed \$20,000,000.

"Swing Line Lender" means The Chase Manhattan Bank.

"Swing Line Loan" has the meaning set forth in Section 2.4(a).

"Swing Line Note" means the promissory note of the Borrower payable to the order of the Swing Line Lender, in substantially the form of Exhibit D-3, evidencing the indebtedness of the Borrower to the Swing Line Lender resulting from Swing Line Loans made by the Swing Line Lender to the Borrower.

"Swing Line Participation Amount" has the meaning set forth in Section $2.4(c)(\mbox{ii}).$

"Term Loan" means the aggregate outstanding principal amount of the Term Loan Borrowing.

"Term Loan Advance" means the outstanding principal from a Bank which represents such Bank's ratable share of the Term Loan Borrowing.

"Term Loan Borrowing" means the aggregate amount of principal advanced pursuant to the Term Loan Borrowing Request under the term loan facility created in Section 2.1.

"Term Loan Borrowing Request" means the Term Loan Borrowing Request in substantially the form of Exhibit B-2 executed by a Responsible Officer of the Borrower and delivered to the Administrative Agent.

"Term Loan Commitment" means, for any Bank, the amount set forth opposite such Bank's name on Schedule I-A as its Term Loan Commitment, or if such Bank has entered into any Assignment and Acceptance since the date of this Agreement, as set forth for such Bank as its Term Loan Commitment in the Register maintained by the Agent pursuant to Section 8.5(c), in each case as such amount may be terminated pursuant to Section 6.2.

"Term Loan Note" means a promissory note of the Borrower payable to the order of a Bank, in substantially the form of Exhibit D-2, evidencing the indebtedness of the Borrower to such Bank resulting from the Term Loan Advance made by such Bank to the Borrower.

"Term Loan Percentage" means, as to any Bank at any time, the percentage which such Bank's Term Loan Commitment then constitutes of the aggregate Term Loan Commitments (or, at any time after the Closing Date, the percentage which the aggregate principal amount of such Bank's Term Loan Advance then outstanding constitutes of the aggregate principal amount of the Term Loan Advances then outstanding).

"Total Debt" means all Debt of the Borrower and the Subsidiaries of the Borrower.

"Tranche" means any tranche of principal outstanding under the Term Loan or the Revolving Loan accruing interest on the same basis whether created in connection with new advances of principal under the Revolving Loan pursuant to Section 2.6(a)(i) or by the continuation or conversion of existing tranches of principal under the Term Loan or the Revolving Loan pursuant to Section 2.6(a)(ii) and shall include the Prime Rate Tranche or any LIBOR Tranche.

"Type" has the meaning set forth in Section 1.4.

"Voting Securities" means (a) with respect to any corporation, any capital stock of the corporation having general voting power under ordinary circumstances to elect directors of such corporation, (b) with respect to any partnership, any partnership interest having general voting power under ordinary circumstances to elect the general partner or other management of the partnership, and (c) with respect to any other Person, such ownership interests in such Person having general voting power under ordinary circumstances to elect the management of such Person, in each case irrespective of whether at the time any other class of stock, partnership interests, or other ownership interest might have special voting power or rights by reason of the happening of any contingency.

1.2. Computation of Time Periods. In this Agreement in the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each means "to but excluding."

1.3. Accounting Terms; Preparation of Financials.

(a) All accounting terms, definitions, ratios, and other tests described herein shall be construed in accordance with United States generally accepted accounting principles applied on a consistent basis with those applied in the preparation of the Financial Statements, except as expressly set forth in this Agreement.

(b) The Restricted Entities shall prepare their financial statements in accordance with United States generally accepted accounting principles applied on a consistent basis with those applied in the preparation of the Financial Statements, unless otherwise approved in writing by the Administrative Agent. In accordance with the foregoing, (i) any Acquisition which is permitted to be treated as a pooling transaction shall be treated as a pooling transaction, and following such an Acquisition the consolidated financial statements of the Borrower shall be adjusted to reflect the results of such Acquisition during the periods prior to such Acquisition in accordance with generally accepted accounting principles and (ii) any Acquisition which is not permitted to be treated as a pooling transaction shall be treated as an asset purchase, without adjustment for prior periods.

(c) The Borrower shall calculate the Applicable Margin and compliance with the financial covenants under this Agreement in Schedule C of the Compliance Certificate. Where

expressly permitted in this Agreement, the Borrower may elect to base such calculations upon the financial reports in Schedule B of such Compliance Certificate. In such case the accounting terms, definitions, ratios, and other tests used in making such calculation shall be construed as required by paragraph (a) above except that the consolidated financial results of the Borrower shall be deemed to be the adjusted consolidated financial results of the Borrower set forth in Schedule B of the Compliance Certificate. Unless otherwise expressly permitted in this Agreement, the Borrower shall calculate the Applicable Margin or compliance with a financial covenant under this Agreement based upon the financial reports in Schedule A of a Compliance Certificate. In such case the accounting terms, definitions, ratios, and other tests used in making such calculation shall be construed as required by paragraph (a) above. When calculating the Applicable Margin or compliance with a financial reports in Schedule B of the Construed as required by calculation shall be construed as required by paragraph (a) above. When calculating the Applicable Margin or compliance with a financial covenant under this Agreement based upon the financial reports in Schedule B of the Compliance with the following provisions:

(i) The Borrower may select one or more Acquisitions for inclusion in the adjustments permitted in Schedule B or C of the Compliance Certificate; provided that if the inclusion of any Acquisition results in an increase in the consolidated EBITDA of the Borrower over the consolidated EBITDA of the Borrower required to be reported in Schedule A of the Compliance Certificate, then the Borrower must include all Acquisitions which would cause a decrease in the consolidated EBITDA of the Borrower in Schedule B and C of the Compliance Certificate.

(ii) If the Acquisition is treated as a pooling transaction, the Borrower shall adjust the pooling accounting treatment of the Acquisition to reflect nonrecurring items (both positive and negative) that are permitted to be adjusted in accordance with the pro forma financial statement guidelines established by the Securities and Exchange Commission for acquisition accounting for reported acquisitions by public companies or as approved by the Majority Banks.

(iii) If the Acquisition is not treated as a pooling transaction, the financial results of the acquired Person or assets shall be added to the applicable financial results of the Borrower in the same manner as if such transaction were a pooling transaction with such adjustments thereto as are required to reflect nonrecurring items (both positive and negative) that are permitted to be adjusted in accordance with the pro forma financial statement guidelines established by the Securities and Exchange Commission for acquisition accounting for reported acquisitions by public companies or as approved by the Majority Banks.

(iv) No Acquisition may be included in Schedule B or C of the Compliance Certificate at the request of the Borrower unless the financial reports of the acquired Person or assets from which Schedule B and C are prepared are (A) audited financial reports prepared by an independent certified public accounting firm, or (B) with respect to unaudited financial reports, are otherwise approved by the Majority Banks.

1.4. Types. The "Type" of a Tranche refers to the determination whether such tranche is a LIBOR Tranche or the Prime Rate Tranche.

1.5. Interpretation. Article, Section, Schedule, and Exhibit references are to this Agreement, unless otherwise specified. All references to instruments, documents, contracts, and agreements are references to such instruments, documents, contracts, and agreements as the same may be amended, supplemented, and otherwise modified from time to time, unless otherwise specified. The word "including" shall mean "including but not limited to." The word "or" shall mean "and/or" wherever necessary to prevent interpretation of any provision against the Administrative Agent or the Banks. Whenever the Borrower has an obligation under this Agreement and the other Credit Documents the expense of complying with that obligation shall be an expense of the Borrower unless otherwise specified. Whenever any determination is to be made by the Administrative Agent or any Bank, such determination shall be in such Person's sole discretion unless otherwise specified in this Agreement. If any provision in this Agreement or the other Credit Documents is held to be illegal, invalid, not binding, or unenforceable, such provision shall be fully severable and this Agreement and the other Credit Documents shall be construed and enforced as if such illegal, invalid, not binding, or unenforceable provision had never comprised a part of this Agreement or the other Credit Documents, and the remaining provisions shall remain in full force and effect. This Agreement and the other Credit Documents have been reviewed and negotiated by sophisticated parties with access to legal counsel and shall not be construed against the drafter. In the event of a conflict between this Agreement and any other Credit Documents, this Agreement shall control.

ARTICLE 2. CREDIT FACILITIES

2.1. Term Loan Facility.

(a) Term Loan Commitments. Each Bank severally agrees, on the terms and conditions set forth in this Agreement and for the purposes set forth in Section 5.4, to make a Term Loan Advance to the Borrower as such Bank's ratable share of the Term Loan Borrowing requested by the Borrower on the Closing Date provided that the aggregate outstanding principal amount of the Term Loan Advance made by such Bank shall not exceed such Bank's Term Loan Commitment. The Term Loan Borrowing made on the Closing Date shall initially be a Prime Rate Borrowing, but may be immediately converted to a LIBOR Tranche in accordance with Section 2.6(a). The indebtedness of the Borrower to the Banks resulting from the Term Loan Advances made by the Banks shall be evidenced by Term Loan Notes made by the Borrower.

(b) Method of Advancing.

(i) The Term Loan Borrowing shall be made pursuant to a Term Loan Borrowing Request given by the Borrower to the Administrative Agent in writing or by telecopy not later than 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the anticipated Closing Date. The Term Loan Borrowing Request shall be fully completed and shall specify the information required therein, and shall be irrevocable and binding on the Borrower. Upon receipt of the Term Loan Borrowing Request by the Administrative Agent, the Administrative Agent shall promptly forward notice of the Term Loan Borrowing to the Banks. Each Bank shall, before 1:00 p.m. (local time at the Applicable Lending Office of the Administrative

Agent) on the anticipated Closing Date, make available from its Applicable Lending Office to the Administrative Agent at the Administrative Agent's Applicable Lending Office, in immediately available funds, such Bank's ratable share of the Term Loan Borrowing. Subject to the satisfaction of all applicable conditions precedent, after receipt by the Administrative Agent of such funds, the Administrative Agent shall, by 4:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent), on the Closing Date make the Term Loan Borrowing available to the Borrower in immediately available funds at any account of Borrower which is designated in writing by the Borrower to the Administrative Agent.

(ii) Unless the Administrative Agent shall have received notice from a Bank before the anticipated Closing Date that such Bank shall not make available to the Administrative Agent such Bank's ratable share of the Term Loan Borrowing, the Administrative Agent may assume that such Bank has made its ratable share of the Term Loan Borrowing available to the Administrative Agent on the Closing Date in accordance with paragraph (i) above, and the Administrative Agent may, in reliance upon such assumption, make available to the Borrower on such date a corresponding amount. If and to the extent that such Bank shall not have so made its ratable share of the Term Loan Borrowing available to the Administrative Agent, such Bank agrees that it shall pay interest on such amount for each day from the date such amount is made available to the Borrower by the Administrative Agent until the date such amount is paid to the Administrative Agent by such Bank at the Federal Funds Rate in effect from time to time, provided that with respect to such Bank if such amount is not paid by such Bank by the end of the second day after the Administrative Agent makes such amount available to the Borrower, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the third day and shall remain at such increased rate thereafter. Interest on such amount shall be due and payable by such Bank upon demand by the Administrative Agent. If such Bank shall pay to the Administrative Agent such amount and interest as provided above, such amount so paid shall constitute such Bank's Term Loan Advance as part of the Term Loan Borrowing for all purposes of this Agreement even though not made on the same day as the other Term Loan Advances comprising the Term Loan Borrowing. In the event that such Bank has not repaid such amount by the end of the fifth day after such amount was made available to the Borrower, the Borrower agrees to repay to the Administrative Agent on demand such amount, together with interest on such amount for each day from the date such amount was made available to the Borrower until the date such amount is repaid to the Administrative Agent at the interest rate charged to the Borrower for the Term Loan Borrowing under the terms of this Agreement.

(iii) The failure of any Bank to make available its ratable share of the Term Loan Borrowing shall not relieve any other Bank of its obligation, if any, to make available its ratable share of the Term Loan Borrowing. No Bank shall be responsible for the failure of any other Bank to honor such other Bank's obligations hereunder, including any failure to make available any funds as part of the Term Loan Borrowing.

(c) Prepayment.

(i) The Borrower may prepay the outstanding principal amount of the Term Loan pursuant to written notice given by the Borrower to the Administrative Agent in writing or by telecopy not later than (A) 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed prepayment, in the case of the prepayment of any portion of the Term Loan which is comprised of LIBOR Tranches, or (B) 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the same Business Day of the proposed prepayment, in the case of the prepayment of any portion of the Term Loan comprised solely of the Prime Rate Tranche. Each such notice shall specify the principal amount and Tranche or Tranches of the Term Loan which shall be prepaid, the date of the prepayment, and shall be irrevocable and binding on the Borrower. Prepayments of the Term Loan shall be made in integral multiples of \$500,000, in the case of prepayments of any LIBOR Tranches, or \$100,000 in the case of prepayments of the Prime Rate Tranche. If the prepayment would cause the aggregate outstanding principal amount of any LIBOR Tranche comprising all or any part of the Term Loan or the aggregate outstanding principal amount of the Prime Rate Tranche comprising all or any part of the Term Loan, to be less than \$1,000,000, in the case of any such LIBOR Tranche, or \$1,000,000, in the case of the Prime Rate Tranche, the prepayment must be in an amount equal to the entire outstanding principal amount of such LIBOR Tranche under the Term Loan or the entire outstanding principal amount of the Prime Rate Tranche under the Term Loan, as the case may be. Upon receipt of any notice of prepayment, the Administrative Agent shall give prompt notice of the intended prepayment to the Banks. For each such notice given by the Borrower, the Borrower shall prepay the Term Loan in the specified amount on the specified date as set forth in such notice. The Borrower shall have no right to prepay any principal amount of the Term Loan except as provided in this Section 2.1(c)(i).

(ii) Each prepayment of principal under the Term Loan pursuant to Section 2.1(c)(i) shall be accompanied by payment of all accrued but unpaid interest on the principal amount prepaid and any amounts required to be paid pursuant to Section 2.7 as a result of such prepayment. The amount of each principal prepayment under the Term Loan pursuant to Section 2.1(c)(i) shall be applied to reduce the then remaining installments of the Term Loan pro rata based upon the then remaining principal amount thereof. Amounts prepaid on account of the Term Loan may not be reborrowed.

(d) Repayment. The Term Loan Advance of each Bank shall mature in 12 quarterly installments, each of which shall be in an amount equal to such Bank's Term Loan Percentage multiplied by the amount set forth below opposite such installment:

Installment Principal Ar	mount
September 30, 2001 \$ 3,750,00	00
December 31, 2001 3,750,00	00
March 31, 2002 3,750,00	00
June 30, 2002 3,750,00	00
September 30, 2002 3,750,00	00
December 31, 2002 3,750,00	00
March 31, 2003 3,750,00	00
June 30, 2003 3,750,00	00
September 30, 2003 5,000,00	00
December 31, 2003 5,000,00	00
March 31, 2004 5,000,00	00
May 22, 2004 5,000,00	00

2.2. Revolving Loan Facility.

(a) Revolving Loan Commitments. Each Bank severally agrees, on the terms and conditions set forth in this Agreement and for the purposes set forth in Section 5.4, to make Revolving Loan Advances to the Borrower as such Bank's ratable share of Revolving Loan Borrowings requested by the Borrower from time to time on any Business Day during the period from the date of this Agreement until the Revolving Loan Maturity Date provided that the aggregate outstanding principal amount of Revolving Loan Advances made by such Bank plus such Bank's ratable share of the aggregate outstanding principal amount of the Swing Line Loans plus such Bank's ratable share of the Letter of Credit Exposure shall not exceed the lesser of (i) such Bank's Revolving Loan Commitment and (ii) such Bank's Revolving Loan Percentage of the Borrowing Base then in effect. Revolving Loan Borrowings must be made in an amount equal to or greater than \$1,000,000, in the case of any Revolving Loan Borrowing comprised of a LIBOR Tranche, or \$1,000,000, in the case of any Prime Rate Borrowing, and be made in multiples of \$500,000, in the case of any Revolving Loan Borrowing comprised of a LIBOR Tranche, or \$100,000, in the case of any Prime Rate Borrowing. Any Revolving Loan Borrowing made on the Closing Date shall initially be a Prime Rate Borrowing, but may be immediately converted to a LIBOR Tranche in accordance with Section 2.6(a). Within the limits expressed in this Agreement, the Borrower may from time to time borrow, prepay, and reborrow Revolving Loan Borrowings. The indebtedness of the Borrower to the Banks resulting from the Revolving Loan Advances made by the Banks shall be evidenced by Revolving Loan Notes made by the Borrower.

(b) Method of Advancing.

(i) Each Revolving Loan Borrowing shall be made pursuant to a Revolving Loan Borrowing Request given by the Borrower to the Administrative Agent in writing or by telecopy not later than the time required pursuant to Section 2.6(a)(i) to select the interest rate basis for the Revolving Loan Borrowing. Each Revolving Loan Borrowing Request shall be fully completed and shall specify the information required therein, and shall be irrevocable and binding on the Borrower. Upon receipt of the Revolving Loan Borrowing Request by the Administrative Agent, the Administrative Agent shall promptly forward notice of the Revolving Loan Borrowing to the Banks. Each Bank shall, before 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent at the Administrative Agent's Applicable Lending Office, in immediately available funds, such Bank's ratable share of such Revolving Loan Borrowing. Subject to the satisfaction of all applicable conditions precedent, after receipt by the Administrative Agent of such funds, the Administrative Agent shall, by 4:00 p.m. (local time at the Agent shall, by 4:00 p.m. (local time at the Agent shall, be funds, by 4:00 p.m.)

Administrative Agent), on the date requested for such Revolving Loan Borrowing make such Revolving Loan Borrowing available to the Borrower in immediately available funds at any account of Borrower which is designated in writing by the Borrower to the Administrative Agent.

(ii) Unless the Administrative Agent shall have received notice from a Bank before the date of any Revolving Loan Borrowing that such Bank shall not make available to the Administrative Agent such Bank's ratable share of such Revolving Loan Borrowing, the Administrative Agent may assume that such Bank has made its ratable share of such Revolving Loan Borrowing available to the Administrative Agent on the date of such Revolving Loan Borrowing in accordance with paragraph (i) above and the Administrative Agent may, in reliance upon such assumption, make available to the Borrower on such date a corresponding amount. If and to the extent that such Bank shall not have so made its ratable share of such Revolving Loan Borrowing available to the Administrative Agent, such Bank agrees that it shall pay interest on such amount for each day from the date such amount is made available to the Borrower by the Administrative Agent until the date such amount is paid to the Administrative Agent by such Bank at the Federal Funds Rate in effect from time to time, provided that with respect to such Bank if such amount is not paid by such Bank by the end of the second day after the Administrative Agent makes such amount available to the Borrower, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the third day and shall remain at such increased rate thereafter. Interest on such amount shall be due and payable by such Bank upon demand by the Administrative Agent. If such Bank shall pay to the Administrative Agent such amount and interest as provided above, such amount so paid shall constitute such Bank's Revolving Loan Advance as part of such Revolving Loan Borrowing for all purposes of this Agreement even though not made on Revolving Loan Borrowing. In the event that such Bank has not repaid such amount by the end of the fifth day after such amount was made available to the Borrower, the Borrower agrees to repay to the Administrative Agent on demand such amount, together with interest on such amount for each day from the date such amount was made available to the Borrower until the date such amount is repaid to the Administrative Agent at the interest rate charged to the Borrower for such Revolving Loan Borrowing under the terms of this Agreement.

(iii) The failure of any Bank to make available its ratable share of any Revolving Loan Borrowing shall not relieve any other Bank of its obligation, if any, to make available its ratable share of such Revolving Loan Borrowing. No Bank shall be responsible for the failure of any other Bank to honor such other Bank's obligations hereunder, including any failure to make available any funds as part of any Revolving Loan Borrowing.

(c) Prepayment.

(i) The Borrower may prepay the outstanding principal amount of the Revolving Loan pursuant to written notice given by the Borrower to the Administrative Agent in writing or by telecopy not later than (A) 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of

the proposed prepayment, in the case of the prepayment of any portion of the Revolving Loan which is comprised of LIBOR Tranches, or (B) 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the same Business Day of the proposed prepayment, in the case of the prepayment of any portion of the Revolving Loan comprised solely of the Prime Rate Tranche. Each such notice shall specify the principal amount and Tranche or Tranches of the Revolving Loan which shall be prepaid, the date of the prepayment, and shall be irrevocable and binding on the Borrower. Prepayments of the Revolving Loan shall be made in integral multiples of \$500,000, in the case of prepayments of any LIBOR Tranches, or \$100,000 in the case of prepayments of the Prime Rate Tranche. If the prepayment would cause the aggregate outstanding principal amount of any LIBOR Tranche comprising all or any part of the Revolving Loan or the aggregate outstanding principal amount of the Prime Rate Tranche comprising all or any part of the Revolving Loan, to be less than \$1,000,000, in the case of any such LIBOR Tranche, or \$1,000,000, in the case of the Prime Rate Tranche, the prepayment must be in an amount equal to the entire outstanding principal amount of such LIBOR Tranche under the Revolving Loan or the entire outstanding principal amount of the Prime Rate Tranche under the Revolving Loan, as the case may be. Upon receipt of any notice of prepayment, the Administrative Agent shall give prompt notice of the intended prepayment to the Banks. For each such notice given by the Borrower, the Borrower shall prepay the Revolving Loan in the specified amount on the specified date as set forth in such notice. The Borrower shall have no right to prepay any principal amount of the Revolving Loan except as provided in this Section 2.2(c)(i).

(ii) Each prepayment of principal of any LIBOR Tranche under the Revolving Loan pursuant to this Section 2.2(c) shall be accompanied by payment of all accrued but unpaid interest on the principal amount prepaid and any amounts required to be paid pursuant to Section 2.7 as a result of such prepayment.

(d) Repayment. The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks the aggregate outstanding principal amount of the Revolving Loan on the Revolving Loan Maturity Date.

(e) Reduction of Commitments. The Borrower shall have the right, upon at least three Business Days' irrevocable notice to the Administrative Agent, to terminate in whole or reduce ratably in part the Available Revolving Loan Commitments; provided that each partial reduction shall be in the aggregate amount of \$5,000,000 or in integral multiples of \$5,000,000 in excess thereof. Any reduction or termination of the Revolving Loan Commitments pursuant to this Section 2.2(e) shall be permanent, with no obligation of the Banks to reinstate such Revolving Loan Commitments, and the commitment fees provided for in Section 2.5(a) shall thereafter be computed on the basis of the Revolving Loan Commitments, as so reduced.

(f) Borrowing Base. If, at any time from the date of this Agreement until the Revolving Loan Maturity Date, the aggregate outstanding principal amount of the Revolving Loan Borrowings plus the aggregate outstanding principal amount of the Swing Line Loans plus the Letter of Credit Exposure exceeds the Borrowing Base then in effect, the Borrower shall, without notice or demand, immediately prepay the Revolving Loan and/or the Swing Line Loans in an aggregate principal amount equal to such excess, together with interest accrued to the date of

such payment or prepayment to the Administrative Agent. If, at any time from the date of this Agreement until the Revolving Loan Maturity Date, after giving effect to any prepayment pursuant to the preceding sentence, the Letter of Credit Exposure exceeds the Borrowing Base then in effect, the Borrower shall, without notice or demand, immediately pay to the Administrative Agent an amount equal to such excess to be held in the Letter of Credit Collateral Account for disposition in accordance with Section 2.3(g), provided that the Administrative Agent shall release to the Borrower from time to time such portion of the amount on deposit in the Letter of Credit Collateral Account which is equal to the amount by which the Borrowing Base at such time plus the amount on deposit in the Letter of Credit Collateral Account exceeds the Letter of Credit Exposure at such time.

2.3. Letter of Credit Facility.

(a) Commitment for Letters of Credit. The Issuing Bank shall, on the terms and conditions set forth in this Agreement and for the purposes set forth in Section 5.4, issue, increase, and extend Letters of Credit at the request of the Borrower from time to time on any Business Day during the period from the date of this Agreement until the Revolving Loan Maturity Date provided that (i) the Letter of Credit Exposure shall not exceed the Letter of Credit Sublimit and (ii) the aggregate outstanding principal amount of Revolving Loan Borrowings plus the aggregate outstanding principal amount of the Swing Line Loans plus the Letter of Credit Exposure shall not exceed the lesser of (a) the aggregate amount of the Revolving Loan Commitments and (b) the Borrowing Base then in effect. No Letter of Credit may have an expiration date later than 12 months after its issuance date, and each Letter of Credit which is self-extending beyond its expiration date must be cancelable upon no more than 30 days notice prior to each extension period given by the Issuing Bank to the beneficiary of such Letter of Credit. No Letter of Credit may have an expiration date later than 12 months after the Revolving Loan Maturity Date unless approved by the Issuing Bank, the Administrative Agent, and the Banks. Each Letter of Credit must be in form and substance acceptable to the Issuing Bank. The indebtedness of the Borrower to the Issuing Bank resulting from Letters of Credit requested by the Borrower shall be evidenced by the Letter of Credit Applications made by the Borrower.

(b) Requesting Letters of Credit. Each Letter of Credit shall be issued, increased, or extended pursuant to a Letter of Credit Application or Letter of Credit Application Amendment, as applicable, given by the Borrower to the Issuing Bank in writing or by telecopy promptly confirmed in writing, such Letter of Credit Application or Letter of Credit Application Amendment being given not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed issuance, increase, or extension of the Letter of Credit. Each Letter of Credit Application or Letter of Credit Application Amendment shall be fully completed and shall specify the information required therein (including the proposed form of the Letter of Credit or change thereto), and shall be irrevocable and binding on the Borrower. Upon receipt by the Issuing Bank of the Letter of Credit Application or Letter of Credit Application Amendment, the Issuing Bank shall give prompt notice thereof to the Administrative Agent, and the Administrative Agent shall promptly inform the Banks of the proposed Letter of Credit or change thereto. Subject to the satisfaction of all applicable conditions precedent, the Issuing Bank shall, by 4:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent), on the date requested by the Borrower for the issuance, increase, or extension of such Letter of Credit issue, increase, or extend such

Letter of Credit to the specified beneficiary. Upon the date of the issuance, increase, or extension of a Letter of Credit, the Issuing Bank shall be deemed to have sold to each other Bank and each other Bank shall be deemed to have purchased from the Issuing Bank a ratable participation in the related Letter of Credit or change thereto. The Issuing Bank shall notify the Administrative Agent of each Letter of Credit issued, increased, or extended and the date and amount of each Bank's participation in such Letter of Credit, and the Administrative Agent shall in turn notify the Banks.

(c) Reimbursements for Letters of Credit.

(i) With respect to any Letter of Credit and in accordance with the related Letter of Credit Application, the Borrower agrees to pay to the Issuing Bank on demand fees due with respect to such Letter of Credit as specified in Section 2.5(b). If the Borrower does not pay upon demand of the Issuing Bank any amount due to the Issuing Bank under any Letter of Credit Application, in addition to any rights the Issuing Bank may have under such Letter of Credit Application, the Issuing Bank may upon written notice to the Administrative Agent request the satisfaction of such obligation by the making of a Revolving Loan Borrowing. Concurrently with such notice to the Administrative Agent, the Issuing Bank will use reasonable efforts provide like notice to the Borrower, provided that failure to provide such notice to the Borrower at such time shall not invalidate the effectiveness of such request for a Revolving Loan Borrowing. Upon such request, the Borrower shall be deemed to have requested the making of a Revolving Loan Borrowing in the amount of such obligation and the transfer of the proceeds thereof to the Issuing Bank. Such Revolving Loan Borrowing shall be a Prime Rate Borrowing. The Administrative Agent shall promptly forward notice of such Revolving Loan Borrowing to the Borrower and the Banks, and each Bank shall, in accordance with the procedures of Section 2.2(b), other than limitations on the size of Revolving Loan Borrowings, and notwithstanding the failure of any conditions precedent, make available such Bank's ratable share of such Revolving Loan Borrowing to the Administrative Agent, and the Administrative Agent shall promptly deliver the proceeds thereof to the Issuing Bank for application to such Bank's ratable share of the obligations under such Letter of Credit. The Borrower hereby unconditionally and irrevocably authorizes, empowers, and directs the Issuing Bank to make such requests for Revolving Loan Borrowings on behalf of the Borrower, and the Banks to make Revolving Loan Advances to the Administrative Agent for the benefit of the Issuing Bank in satisfaction of such obligations. The Administrative Agent and each Bank may record and otherwise treat the making of such Revolving Loan Borrowings as the making of Revolving Loan Borrowings to the Borrower under this Agreement as if requested by the Borrower. Nothing herein is intended to release the Borrower's obligations under any Letter of Credit Application, but only to provide an additional method of payment therefor. The making of any Revolving Loan Borrowing under this Section 2.3(c) shall not constitute a cure or waiver of any Default or Event of Default, other than the payment Default or Event of Default which is satisfied by the application of the amounts deemed advanced hereunder, caused by the Borrower's failure to comply with the provisions of this Agreement or any Letter of Credit Application.

(ii) If prior to the time a Revolving Loan Advance would have otherwise been made pursuant to Section 2.3(c)(i), a Bankruptcy Event of Default shall have occurred and be continuing with respect to the Borrower or if for any other reason, as determined by the Issuing Bank in its sole discretion, Revolving Loan Advances may not be made as contemplated by Section 2.3(c)(i), each Bank shall, on the date such Revolving Loan Advance was to have been made pursuant to the notice referred to in Section 2.3(c)(i), pay to the Issuing Bank such Bank's ratable share of the obligations under such Letter of Credit.

(iii) Whenever, at any time after the Issuing Bank has made payment under any Letter of Credit and has received from any Bank its ratable share of such payment in accordance with this Section 2.3(c), the Issuing Bank receives any payment related to such Letter of Credit (whether directly from the Borrower or otherwise, including proceeds of collateral applied thereto by the Issuing Bank), or any payment of interest on account thereof, the Issuing Bank will distribute to such Bank its ratable share thereof; provided, however, that in the event that any such payment received by the Issuing Bank shall be required to be returned by the Issuing Bank, such Bank shall return to the Issuing Bank the portion thereof previously distributed by the Issuing Bank to it.

(iv) Each Bank's obligation to make the Revolving Loan Advances referred to in Section 2.3(c)(i) and to purchase participating interests pursuant to Section 2.3(b) shall be absolute and unconditional and shall not be affected by any circumstance, including (a) any setoff, counterclaim, recoupment, defense or other right that such Bank or the Borrower may have against the Issuing Bank, the Borrower or any other Person for any reason whatsoever; (b) the occurrence or continuance of a Default or an Event of Default or the failure to satisfy any of the other conditions specified in Section 3; (c) any adverse change in the condition (financial or otherwise) of the Borrower; (d) any breach of this Agreement or any other Credit Document by the Borrower, any other Credit Party or any other Bank; or (e) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing.

(d) Prepayments of Letters of Credit. In the event that any Letters of Credit shall be outstanding according to their terms after the Revolving Loan Maturity Date, the Borrower shall pay to the Administrative Agent an amount equal to the Letter of Credit Exposure allocable to such Letters of Credit to be held in the Letter of Credit Collateral Account and applied in accordance with paragraph (g) below.

(e) Obligations Unconditional. The obligations of the Borrower and each Bank under this Agreement and the Letter of Credit Applications to make payments as required to reimburse the Issuing Bank for draws under Letters of Credit and to make other payments due in respect of Letters of Credit shall be unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement and the Letter of Credit Applications under all circumstances, including: (i) any lack of validity or enforceability of any Letter of Credit Document; (ii) any amendment, waiver, or consent to departure from any Letter of Credit Document; (iii) the existence of any claim, set-off, defense, or other right which the Borrower or any Bank may have at any time against any beneficiary or transferee of any Letter of Credit (or any Persons for whom any such beneficiary or any such transferee may be acting), the Issuing Bank, or any other

person or entity, whether in connection with the transactions contemplated in this Agreement or any unrelated transaction; (iv) any statement or any other document presented under such Letter of Credit proving to be forged, fraudulent, invalid, or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or (v) payment by the Issuing Bank under any Letter of Credit against presentation of a draft or certificate which does not comply with the terms of such Letter of Credit; provided, however, that nothing contained in this paragraph (e) shall be deemed to constitute a waiver of any remedies of the Borrower or any Bank in connection with the Letters of Credit or the Borrower's or such Bank's rights under paragraph (f) below.

(f) Liability of Issuing Bank. The Issuing Bank shall not be liable or responsible for: (i) the use which may be made of any Letter of Credit or any acts or omissions of any beneficiary or transferee in connection therewith; (ii) the validity, sufficiency, or genuineness of documents related to Letters of Credit, or of any endorsement thereon, even if such documents should prove to be in any or all respects invalid, insufficient, fraudulent, or forged; (iii) payment by the Issuing Bank against presentation of documents which do not strictly comply with the terms of a Letter of Credit, including failure of any documents to bear any reference or adequate reference to the relevant Letter of Credit; or (iv) any other circumstances whatsoever in making or failing to make payment under any Letter of Credit (INCLUDING THE ISSUING BANK'S OWN NEGLIGENCE); except that the Issuing Bank shall be liable to the Borrower or any Bank to the extent of any direct, as opposed to consequential, damages suffered by the Borrower or such Bank which the Borrower or such Bank proves were caused by (A) the Issuing Bank's gross negligence or willful misconduct in determining whether documents presented under a Letter of Credit comply with the terms of such Letter of Credit, (B) the Issuing Bank's willful failure to make or delay in making lawful payment under any Letter of Credit after the presentation to it of documentation strictly complying with the terms and conditions of such Letter of Credit or the Issuing Bank's payment of greater than the maximum amount permitted under any Letter of Credit, or (C) the Issuing Bank's negligence in the handling of money.

(g) Letter of Credit Collateral Account.

(i) If the Borrower is required to deposit funds in the Letter of Credit Collateral Account pursuant to Section 2.2(f), 2.3(d) or 6.4, then the Borrower and the Administrative Agent shall establish the Letter of Credit Collateral Account, and the Borrower shall execute any documents and agreements, including the Administrative Agent's standard form assignment of deposit accounts, that the Administrative Agent reasonably requests in connection therewith to establish the Letter of Credit Collateral Account and grant the Administrative Agent a first priority security interest in such account and the funds therein. The Borrower hereby pledges to the Administrative Agent and grants the Administrative Agent a security interest in the Letter of Credit Collateral Account, whenever established, all funds held in the Letter of Credit Collateral Account from time to time, and all proceeds thereof as security for the payment of the Credit Obligations.

(ii) Funds held in the Letter of Credit Collateral Account shall be held as cash collateral for obligations with respect to Letters of Credit and promptly applied by the Administrative Agent at the request of the Issuing Bank to any reimbursement or other

obligations under Letters of Credit that exist or occur. To the extent that any surplus funds are held in the Letter of Credit Collateral Account above the Letter of Credit Exposure, during the existence of an Event of Default the Administrative Agent may (A) hold such surplus funds in the Letter of Credit Collateral Account as cash collateral for the Credit Obligations or (B) apply such surplus funds to any Credit Obligations in accordance with Section 6.9. If no Default exists, the Administrative Agent shall release to the Borrower at the Borrower's written request any funds held in the Letter of Credit Collateral Account above the amounts required by Section 2.3(d).

(iii) Funds held in the Letter of Credit Collateral Account shall be invested in money market funds of the Administrative Agent or in another investment if mutually agreed upon by the Borrower and the Administrative Agent, but the Administrative Agent shall have no other obligation to make any other investment of the funds therein. The Administrative Agent shall exercise reasonable care in the custody and preservation of any funds held in the Letter of Credit Collateral Account and shall be deemed to have exercised such care if such funds are accorded treatment substantially equivalent to that which the Administrative Agent accords its own property, it being understood that the Administrative Agent shall not have any responsibility for taking any necessary steps to preserve rights against any parties with respect to any such funds.

2.4. Swing Line Facility.

(a) Commitment.

(i) Subject to the terms and conditions hereof, the Swing Line Lender agrees to make a portion of the credit otherwise available to the Borrower under the Revolving Loan Commitments from time to time during the period from the date of this Agreement until the Revolving Loan Maturity Date by making swing line loans ("Swing Line Loans") to the Borrower; provided that (i) the aggregate principal amount of Swing Line Loans outstanding at any time shall not exceed the Swing Line Commitment then in effect (notwithstanding that the Swing Line Loans outstanding at any time, when aggregated with the Swing Line Lender's other outstanding Revolving Loan Advances, may exceed the Swing Line Commitment then in effect), (ii) the Borrower shall not request, and the Swing Line Lender shall not make, any Swing Line Loan if, after giving effect to the making of such Swing Line Loan, the aggregate amount of the Available Revolving Loan Commitments would be less than zero and (iii) at no time may the then outstanding Swing Line Loans plus the aggregate outstanding principal amount of Revolving Loan Borrowings plus the Letter of Credit Exposure exceed the Borrowing Base then in effect. Within the limits expressed in this Agreement, the Borrower may use the Swing Line Commitment by borrowing, repaying and reborrowing, all in accordance with the terms and conditions hereof. Each Swing Line Loan shall be a Prime Rate Borrowing. The indebtedness of the Borrower to the Swing Line Lender resulting from the Swing Line Loans made by the Swing Line Lender shall be evidenced by the Swing Line Note made by the Borrower.

(ii) The Borrower shall repay all outstanding Swing Line Loans on the Revolving Loan Maturity Date.

(b) Procedure for Swing Line Borrowing. Whenever the Borrower desires that the Swing Line Lender make Swing Line Loans it shall give the Swing Line Lender irrevocable telephonic notice confirmed promptly in writing (which telephonic notice must be received by the Swing Line Lender not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the proposed date of the Swing Line Borrowing), specifying (i) the amount to be borrowed and (ii) the requested date of the Swing Line Borrowing (which shall be a Business Day during the period from the date of this Agreement until the Revolving Loan Maturity Date). Each borrowing under the Swing Line Commitment shall be in an amount equal to \$500,000 or a whole multiple of \$100,000 in excess thereof. Not later than 3:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the date requested for a Swing Line Borrowing specified in a notice in respect of Swing Line Loans, the Swing Line Lender shall make available to the Administrative Agent at the Applicable Lending Office an amount in immediately available funds equal to the amount of the Swing Line Loan to be made by the Swing Line Lender. The Administrative Agent shall make the proceeds of such Swing Line Loan available to the Borrower on such date in immediately available funds at any account of Borrower which is designated in writing by the Borrower to the Administrative Agent.

(c) Reimbursements for Swing Line Loan Obligations.

(i) The Swing Line Lender, at any time and from time to time in its sole and absolute discretion may, on behalf of the Borrower (which hereby irrevocably directs the Swing Line Lender to act on its behalf), on one Business Day's notice given by the Swing Line Lender no later than 12:00 noon (local time at the Applicable Lending Office of the Administrative Agent) request each Bank to make, and each Bank hereby agrees to make, a Revolving Loan Advance, in an amount equal to such Bank's Revolving Loan Percentage of the aggregate amount of the Swing Line Loans (the "Refunded Swing Line Loans") outstanding on the date of such notice, to repay the Swing Line Lender. Upon such request, the Borrower shall be deemed to have requested the making of a Revolving Loan Borrowing in the amount of such request. Such Revolving Loan Borrowing shall be a Prime Rate Borrowing. Each Bank shall make the amount of its Revolving Loan Advance available to the Administrative Agent at the Administrative Agent's Applicable Lending Office in immediately available funds, not later than 10:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) one Business Day after the date of such notice. The proceeds of such Revolving Loan Borrowing shall be immediately made available by the Administrative Agent to the Swing Line Lender for application by the Swing Line Lender to the repayment of the Refunded Swing Line Loans. The Borrower irrevocably authorizes the Swing Line Lender to charge the 's accounts with the Administrative Agent (up to the amount Borrower available in each such account) in order to immediately pay the amount of such Refunded Swing Line Loans to the extent amounts received from the Banks are not sufficient to repay in full such Refunded Swing Line Loans.

(ii) If prior to the time a Revolving Loan Advance would have otherwise been made pursuant to Section 2.4(c)(i), a Bankruptcy Event of Default shall have occurred and be continuing with respect to the Borrower or if for any other reason, as determined by the Swing Line Lender in its sole discretion, Revolving Loan Advances may not be made as contemplated by Section 2.4(c)(i), each Bank shall, on the date such Revolving

Loan Advance was to have been made pursuant to the notice referred to in Section 2.4(c)(i) (the "Refunding Date"), purchase for cash an undivided participating interest in the then outstanding Swing Line Loans by paying to the Swing Line Lender an amount (the "Swing Line Participation Amount") equal to (a) such Bank's Revolving Percentage times (b) the sum of the aggregate principal amount of Swing Line Loans then outstanding that were to have been repaid with such Revolving Loan Borrowing.

(iii) Whenever, at any time after the Swing Line Lender has received from any Bank such Bank's Swing Line Participation Amount, the Swing Line Lender receives any payment on account of the Swing Line Loans, the Swing Line Lender will distribute to such Bank its Swing Line Participation Amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Bank's participating interest was outstanding and funded and, in the case of principal and interest payments, to reflect such Bank's pro rata portion of such payment if such payment is not sufficient to pay the principal of and interest on all Swing Line Loans then due); provided, however, that in the event that such payment received by the Swing Line Lender is required to be returned, such Bank will return to the Swing Line Lender any portion thereof previously distributed to it by the Swing Line Lender.

(iv) Each Bank's obligation to make the Revolving Loan Advances referred to in Section 2.4(c)(i) and to purchase participating interests pursuant to Section 2.4(c)(ii) shall be absolute and unconditional and shall not be affected by any circumstance, including (a) any setoff, counterclaim, recoupment, defense or other right that such Bank or the Borrower may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever; (b) the occurrence or continuance of a Default or an Event of Default or the failure to satisfy any of the other conditions specified in Section 3; (c) any adverse change in the condition (financial or otherwise) of the Borrower; (d) any breach of this Agreement or any other Credit Document by the Borrower, any other Credit Party or any other Bank; or (e) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing.

2.5. Fees.

(a) Commitment Fees. The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks an unused commitment fee of 0.50% per annum on the average daily amount by which (i) the aggregate amount of the Revolving Loan Commitments exceeds (ii) the aggregate outstanding principal amount of the Revolving Loan plus the Letter of Credit Exposure. The unused commitment fee shall accrue from the date of this Agreement and be due and payable in arrears on the last day of each calendar quarter and on the Revolving Loan Maturity Date.

(b) Fees for Letters of Credit. For each Letter of Credit issued by the Issuing Bank, the Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks a letter of credit fee equal to the Applicable Margin for letter of credit fees per annum on the face amount of such Letter of Credit for the stated term of such Letter of Credit, with a minimum fee of \$500. In addition, for each Letter of Credit issued by the Issuing Bank, the Borrower shall pay to the Administrative Agent for the benefit of the Issuing Bank a fronting fee of 0.125% per annum on

the face amount of such Letter of Credit for the stated term of such Letter of Credit, with a minimum fee of \$500. The Borrower shall pay each such letter of credit fee for each Letter of Credit quarterly in arrears within ten days after when billed therefor by the Issuing Bank.

2.6. Interest.

(a) Election of Interest Rate Basis. The Borrower may select the interest rate basis for the Term Loan and the Revolving Loan in accordance with the terms of this Section 2.6(a):

(i) Under the Revolving Loan Borrowing Request provided to the Administrative Agent in connection with the making of each Revolving Loan Borrowing, the Borrower shall select the amount and the Type of the Tranches, and for each LIBOR Tranche selected, any permitted Revolving Loan Borrowing, provided that (A) at no time shall there be more than ten separate LIBOR Tranches outstanding and (B) each LIBOR Tranche must be in a principal amount equal to or greater than \$1,000,000 and be made in multiples of \$500,000, and the Prime Rate Tranche must be in a principal amount equal to or greater than \$1,000,000 and be made in multiples of \$100,000. Such interest rate elections must be provided to the Administrative Agent in writing or by telecopy not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of any proposed Revolving Loan Borrowing comprised of a LIBOR Tranche or 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the same day of any proposed Revolving Loan Borrowing which is a Prime Rate Borrowing. The Administrative Agent shall promptly forward copies of such interest rate elections to the Banks. In the case of any Revolving Loan Borrowing comprised of a LIBOR Tranche, upon determination by the Administrative Agent, the Administrative Agent shall promptly notify the Borrower and the Banks of the applicable interest rate for such Tranche.

 $({\rm ii})$ With respect to any Tranche, the Borrower may continue or convert any portion of any LIBOR Tranche or the Prime Rate Tranche to form new LIBOR Tranches or increase or decrease the amount of the Prime Rate Tranche in accordance with this paragraph. Each such continuation or conversion shall be deemed to create a new LIBOR Tranche or increase or decrease the amount of the Prime Rate Tranche, as applicable, for all purposes of this Agreement. Each such continuation or conversion shall be made pursuant to a Continuation/Conversion Request given by the Borrower to the Administrative Agent in writing or by telecopy not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed continuation or conversion. Each Continuation/Conversion Request shall be fully completed and shall specify the information required therein, and shall be irrevocable and binding on the Borrower. The Administrative Agent shall promptly forward notice of the continuation or conversion to the Banks. In the case of any continuation or conversion into LIBOR Tranches, upon determination by the Administrative Agent, the Administrative Agent shall notify the Borrower and the Banks of the applicable interest rate. Continuations and conversions of LIBOR Tranches shall be made in integral multiples of \$500,000, and continuations and conversions of the Prime Rate Tranche shall be made in integral multiples of \$100,000.

No continuation or conversion shall be permitted if such continuation or conversion would cause the aggregate outstanding principal amount of any LIBOR Tranche which would remain outstanding to be less than \$1,000,000, or the aggregate outstanding principal amount of the Prime Rate Tranche which would remain outstanding to be less than \$1,000,000. At no time shall there be more than ten separate LIBOR Tranches outstanding. Any conversion of an existing LIBOR Tranche is subject to Section 2.7. Subject to the satisfaction of all applicable conditions precedent, the Administrative Agent and the Banks shall before close of business on the date requested by the Borrower for the continuation or conversion, make such continuation or conversion.

(iii) At the end of the Interest Period for any LIBOR Tranche if the Borrower has not continued or converted such LIBOR Tranche into new Tranches as provided for in paragraph (ii) above, the Borrower shall be deemed to have continued such LIBOR Tranche as a new LIBOR Tranche with an Interest Period of one month. All of the Prime Rate Tranche shall continue as the Prime Rate Tranche unless the Borrower converts such Prime Rate Tranche as provided for in paragraph (ii) above.

(b) LIBOR Tranches. Each LIBOR Tranche shall bear interest during its Interest Period at a per annum interest rate equal to the sum of the LIBOR for such Tranche plus the Applicable Margin for LIBOR Tranches in effect from time to time. The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks all accrued but unpaid interest on each LIBOR Tranche on the last day of the applicable Interest Period for such LIBOR Tranche (and with respect to LIBOR Tranches with Interest Periods of greater than three months, on the date which is three months after the first date of the Interest Period for such LIBOR Tranche), when required upon prepayment as specified elsewhere in this Agreement, on any date when such LIBOR Tranche is prepaid or repaid in full, and on the Revolving Loan Maturity Date.

(c) Prime Rate Tranche. The Prime Rate Tranche shall bear interest at a per annum interest rate equal to the Base Rate in effect from time to time plus the Applicable Margin for the Prime Rate Tranche in effect from time to time. The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks all accrued but unpaid interest on the aggregate outstanding principal amount of the Prime Rate Tranche on the last day of each calendar quarter, when required upon prepayment as specified elsewhere in this Agreement, on any date the Prime Rate Tranche is prepaid or repaid in full, and on the Revolving Loan Maturity Date.

(d) Usury Protection.

(i) If, with respect to any Bank and the Borrower, the effective rate of interest contracted for by such Bank with the Borrower under the Credit Documents, including the stated rates of interest contracted for hereunder and any other amounts contracted for under the Credit Documents which are deemed to be interest, at any time exceeds the Highest Lawful Rate, then the outstanding principal amount of the loans made by such Bank to the Borrower hereunder shall bear interest at a rate which would make the effective rate of interest on the loans made by such Bank to the Borrower under the Credit Documents equal the Highest Lawful Rate until the difference between the amounts which would have been due by the Borrower to such Bank at the stated rates and the amounts which were due by the Borrower to such Bank at the Highest Lawful Rate

(the "Lost Interest") has been recaptured by such Bank. If, when the loans made hereunder are repaid in full, the Lost Interest has not been fully recaptured by such Bank pursuant to the preceding paragraph, then, to the extent permitted by law, the interest rates charged by such Bank to the Borrower hereunder shall be retroactively increased such that the effective rate of interest on the loans made by such Bank to the Borrower under the Credit Documents was at the Highest Lawful Rate since the effectiveness of this Agreement to the extent necessary to recapture the Lost Interest not recaptured pursuant to the preceding sentence and, to the extent allowed by law, the Borrower shall pay to such Bank the amount of the Lost Interest remaining to be recaptured by such Bank.

(ii) In calculating all sums paid or agreed to be paid to any Bank by the Borrower for the use, forbearance, or detention of money under the Credit Documents, such amounts shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread in equal parts throughout the term of the Credit Documents.

(iii) NOTWITHSTANDING THE FOREGOING OR ANY OTHER TERM IN THIS AGREEMENT AND THE CREDIT DOCUMENTS TO THE CONTRARY, it is the intention of each Bank and the Borrower to conform strictly to any applicable usury laws. Accordingly, if any Bank contracts for, charges, or receives any consideration from the Borrower which constitutes interest in excess of the Highest Lawful Rate, then any such excess shall be canceled automatically and, if previously paid, shall at such Bank's option be applied to the outstanding amount of the loans made hereunder by such Bank to the Borrower or be refunded to the Borrower.

2.7. Breakage Costs. If (i) any payment of principal on or any conversion of any LIBOR Tranche is made on any date other than the last day of the Interest Period for such LIBOR Tranche, whether as a result of any voluntary or mandatory prepayment (other than a prepayment upon the occurrence of any event subject to Section 2.9 or 2.10), any acceleration of maturity, or any other cause, (ii) any payment of principal on any LIBOR Tranche is not made when due, or (iii) any LIBOR Tranche is not borrowed, converted, or prepaid in accordance with the respective notice thereof provided by the Borrower to the Administrative Agent, whether as a result of any failure to meet any applicable conditions precedent for borrowing, conversion, or prepayment, the permitted cancellation of any request for borrowing, conversion, or prepayment, the failure of the Borrower to provide the respective notice of borrowing, conversion, or prepayment, or any other cause not specified above which is created by the Borrower, then the Borrower shall pay to each Bank upon demand any amounts required to compensate such Bank for any losses, costs, or expenses, including lost profits and administrative expenses, which are reasonably allocable to such action, including losses, costs, and expenses related to the liquidation or redeployment of funds acquired or designated by such Bank to fund or maintain such Bank's ratable share of such LIBOR Tranche or related to the reacquisition or redesignation of funds by such Bank to fund or maintain such Bank's ratable share of such LIBOR Tranche following any liquidation or redeployment of such funds caused by such action. Such Bank need not prove matched funding of any particular funds, and a certificate as to the amount of such loss, cost, or expense detailing the calculation thereof and certifying that such Bank customarily charges such amounts to its other customers in similar circumstances submitted by such Bank to the Borrower shall be conclusive and binding for all purposes, absent manifest error.

Notwithstanding anything herein to the contrary, until the date 180 days after the Closing Date, the Borrower shall reimburse each Bank for all losses, costs and expenses of the type described in this Section 2.7 incurred by such Bank in connection with the syndication of the Facilities by the Administrative Agent in its sole discretion and the addition of new lenders.

2.8. Increased Costs.

(a) Cost of Funds. If due to either (i) any introduction of, change in, or change in the interpretation of, any law or regulation, in each case, after the date of this Agreement or (ii) compliance with any guideline or request from any central bank or other governmental authority having appropriate jurisdiction (whether or not having the force of law) given after the date of this Agreement, there shall be any increase in the costs of any Bank attributable to (x) committing to make any Term Loan Advance or Revolving Loan Advance or obtaining funds for the making, funding, or maintaining of such Bank's ratable share of any LIBOR Tranche in the relevant interbank market or (y) committing to make Letters of Credit or issuing, funding, or maintaining Letters of Credit (including any increase in any applicable reserve requirement specified by the Federal Reserve Board, including those for emergency, marginal, supplemental, or other reserves), then the Borrower shall pay to such Bank upon demand any amounts required to compensate such Bank for such increased costs, such amounts being due and payable upon demand by such Bank. A certificate as to the cause and amount of such increased cost detailing the calculation of such cost and certifying that such Bank customarily charges such amounts to its other customers in similar circumstances submitted by such Bank to the Borrower shall be conclusive and binding for all purpose, absent manifest error. No Bank may make any claim for compensation under this Section 2.8(a) for increased costs incurred more than 90 days prior to the delivery of any such certificate.

(b) Capital Adequacy. If, due to either (i) any introduction of, change in, or change in the interpretation of, any law or regulation, in each case, after the date of this Agreement or (ii) compliance with any guideline or request from any central bank or other governmental authority having appropriate jurisdiction (whether or not having the force of law) given after the date of this Agreement, there shall be any increase in the capital requirements of any Bank or its parent or holding company attributable to (x) committing to make Term Loan Advances or Revolving Loan Advances or making, funding, or maintaining Term Loan Advances or Revolving Loan Advances or (y) committing to make Letters of Credit or issuing, funding, or maintaining Letters of Credit, as such capital requirements are allocated by such Bank, then the Borrower shall pay to such Bank upon demand any amounts required to compensate such Bank or its parent or holding company for such increase in costs (including an amount equal to any reduction in the rate of return on assets or equity of such Bank or its parent or holding company), such amounts being due and payable upon demand by such Bank. A certificate as to the cause and amounts detailing the calculation of such amounts and certifying that such Bank customarily charges such amounts to its other customers in similar circumstances submitted by such Bank to the Borrower shall be conclusive and binding for all purposes, absent manifest error. No Bank may make any claim for compensation under this Section 2.8(b) for increased costs incurred more than 90 days prior to the delivery of any such certificate.

2.9. Illegality. Notwithstanding any other provision in this Agreement, if it becomes unlawful for any Bank to obtain deposits or other funds for making or funding such Bank's ratable share

of any LIBOR Tranche in the relevant interbank market, such Bank shall so notify the Borrower and the Administrative Agent, such Bank's commitment to create LIBOR Tranches shall be suspended until such condition has passed, all LIBOR Tranches applicable to such Bank shall be converted to the Prime Rate Tranche as of the end of each applicable Interest Period or earlier if necessary, and all subsequent requests for LIBOR Tranches shall be deemed to be requests for Prime Rate Borrowings or continuations and conversions of the Prime Rate Tranche, as applicable, with respect to such Bank.

2.10. Market Failure. Notwithstanding any other provision in this Agreement, if the Administrative Agent determines that: (a) quotations of interest rates for the relevant deposits referred to in the definition of "LIBOR" are not being provided in the relevant amounts, or maturities for purposes of determining the rate of interest referred to in the definition of "LIBOR" or (b) the relevant rates of interest referred to in the definition of "LIBOR" which are used as the basis to determine the rate of interest for LIBOR Tranches will not adequately cover the cost to any Bank of making or maintaining such Bank's ratable share of any LIBOR Tranche, then if the Administrative Agent so notifies the Borrower, the Banks' commitments to create LIBOR Tranches shall be suspended until such condition has passed, all LIBOR Tranches shall be converted to the Prime Rate Tranche as of the end of each applicable Interest Period or earlier if necessary, and all subsequent requests for LIBOR Tranches shall be deemed to be requests for Prime Rate Borrowings or continuations and conversions of the Prime Rate Tranche, as applicable, with respect to such Bank.

2.11. Payment Procedures and Computations.

(a) Payment Procedures. Time is of the essence in this Agreement and the Credit Documents. All payments hereunder shall be made in Dollars. The Borrower shall make each payment under this Agreement and under the Term Loan Notes and the Revolving Loan Notes not later than 12:00 noon (local time at the Applicable Lending Office of the Administrative Agent) on the day when due to the Administrative Agent at the Administrative Agent's Applicable Lending Office in immediately available funds. All payments by the Borrower hereunder shall be made without any offset, abatement, withholding, deduction, counterclaim, or reduction. Upon receipt of payment from the Borrower of any principal, interest, or fees due to the Banks, the Administrative Agent shall promptly after receipt thereof distribute to the Banks their ratable share of such payments for the account of their respective Applicable Lending Offices. If and to the extent that the Administrative Agent shall not have so distributed to any Bank its ratable share of such payments, the Administrative Agent agrees that it shall pay interest on such amount for each day after the day when such amount is made available to the Administrative Agent by the Borrower until the date such amount is paid to such Bank by the Administrative Agent at the Federal Funds Rate in effect from time to time, provided that if such amount is not paid by the Administrative Agent by the end of the third day after the Borrower makes such amount available to the Administrative Agent, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the fourth day and shall remain at such increased rate thereafter. Interest on such amount shall be due and payable by the Administrative Agent upon demand by such Bank. Upon receipt of other amounts due solely to the Administrative Agent, the Issuing Bank, the Swing Line Lender, or a specific Bank, the Administrative Agent shall distribute such amounts to the appropriate party to be applied in accordance with the terms of this Agreement.

(b) Administrative Agent Reliance. Unless the Administrative Agent shall have received written notice from the Borrower prior to any date on which any payment is due to the Banks that the Borrower shall not make such payment in full, the Administrative Agent may assume that the Borrower has made such payment in full to the Administrative Agent on such date, and the Administrative Agent may, in reliance upon such assumption, cause to be distributed to each Bank on such date an amount equal to the amount then due such Bank. If and to the extent the Borrower shall not have so made such payment in full to the Administrative Agent forthwith on demand such amount distributed to such Bank, together with interest thereon from the date such amount is distributed to such Bank until the date such amount to the Administrative Agent, at an interest rate equal to, the Federal Funds Rate in effect from time to time, provided that with respect to such Bank, if such amount is not repaid by such Bank by the end of the second day after the date of the Administrative Agent's demand, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the third day after the date of the Administrative Agent's demand and shall remain at such increased rate thereafter.

(c) Sharing of Payments. Each Bank agrees that if it should receive any payment (whether by voluntary payment, by realization upon security, by the exercise of the right of setoff or banker's lien, by counterclaim or cross action, by the enforcement of any right under the Credit Documents, or otherwise) in respect of any obligation of the Borrower to pay principal, interest, fees, or any other obligation incurred under the Credit Documents in a proportion greater than the total amount of such principal, interest, fees, or other obligation then owed and due by the Borrower to such Bank bears to the total amount of principal, interest, fees, or other obligation then owed and due by the Borrower to all of the Banks immediately prior to such receipt, then such Bank receiving such excess payment shall purchase for cash without recourse from the other Banks an interest in the obligations of the Borrower to such Banks, in proportion with the Banks' respective pro rata shares, in the aggregate unpaid amount of principal, interest, fees, or any such other obligation, as the case may be, owed by the Borrower to all of the Banks; provided that if all or any portion of such excess payment is thereafter recovered from such Bank, such purchase shall be rescinded and the purchase price restored to the extent of such recovery, in proportion with the Banks' respective pro rata shares, but without interest.

(d) Authority to Charge Accounts. The Administrative Agent, if and to the extent payment owed to the Administrative Agent or any Bank is not made when due, may charge from time to time against any account of the Borrower with the Administrative Agent any amount so due. The Administrative Agent agrees promptly to notify the Borrower after any such charge and application made by the Administrative Agent provided that the failure to give such notice shall not affect the validity of such charge and application.

(e) Interest and Fees. Unless expressly provided for in this Agreement, (i) all computations of interest based on the Prime Rate (including the Base Rate, when applicable) shall be made on the basis of a 365/366 day year, as the case may be, (ii) all computations of interest based on the Federal Funds Rate (including the Base Rate, when applicable) shall be made on the basis of a 360 day year, (iii) all computations of interest based upon the LIBOR shall be made on the basis of a 360 day year, and (iv) all computations of fees shall be made on the basis of a 360 day year, in each case for the actual number of days (including the first day,

but excluding the last day) occurring in the period for which such interest or fees are payable. Each determination by the Administrative Agent of an interest rate or fee shall be conclusive and binding for all purposes, absent manifest error.

(f) Payment Dates. Whenever any payment shall be stated to be due on a day other than a Business Day, such payment shall be made on the next succeeding Business Day, and such extension of time shall in such case be included in the computation of payment of interest or fees, as the case may be. If the time for payment for an amount payable is not specified in this Agreement or in any other Credit Document, the payment shall be due and payable on demand by the Administrative Agent or the applicable Bank.

2.12. Taxes.

(a) No Deduction for Certain Taxes. Any and all payments by the Borrower shall be made free and clear of and without deduction for any and all present or future taxes, levies, imposts, deductions, charges, or withholdings, and all liabilities with respect thereto, other than taxes imposed on the income of, and franchise taxes imposed in lieu of net income taxes on, the Administrative Agent, any Bank, or the Applicable Lending Office thereof by any jurisdiction in which any such entity is a present or former citizen or resident or any political subdivision of such jurisdiction (other than any such connection arising solely from the Administrative Agent, such Bank, or the Applicable Lending Office having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or any other Letter of Credit Document) (all such nonexcluded taxes, levies, imposts, deductions, charges, withholdings, and liabilities being hereinafter referred to as "Taxes"). If the Borrower shall be required by law to deduct any Taxes or Other Taxes form or in respect of any sum payable to the Administrative Agent, any Bank, or the Applicable Lending Office thereof, (i) the sum payable shall be increased as may be necessary so that, after making all required deductions (including deductions applicable to additional sums payable under this Section 2.12), such Person receives an amount equal to the sum it would have received had no such deductions been made; (ii) the Borrower shall make such deductions; and (iii) the Borrower shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable law.

(b) Other Taxes. The Borrower agrees to pay any present or future stamp or documentary taxes or any other excise or property taxes, charges, or similar levies ("Other Taxes") which arise from any payment made or from the execution, delivery, or registration of, or otherwise with respect to, this Agreement or the other Credit Documents (other than those which become due as a result of any Bank joining this Agreement as a result of any Assignment and Acceptance, which shall be paid by the Bank which becomes a Bank hereunder as a result of such Assignment and Acceptance).

(c) Foreign Bank Withholding Exemption. Each Bank and Issuing Bank that is not a "US Person" as defined in Section 7701(a)(30) of the Code (a "Non-US Bank") agrees that it shall deliver to the Borrower and the Administrative Agent (i) two duly completed copies of United States Internal Revenue Service Form W-8BEN, W-8ECI or other applicable form, as the case may be, certifying in each case that such Bank is entitled to receive payments under this Agreement, the Term Loan Notes and the Revolving Loan Notes payable to it, without deduction or withholding of any United States federal income taxes, (ii) if applicable, a statement

indicating that such Bank is entitled to the "portfolio interest" exception inducting that back back is the test of the code, and (iii) any other governmental forms which are necessary or required under an applicable tax treaty or otherwise by law to reduce or eliminate any withholding tax, which have been reasonably requested by the Borrower. Each Bank which delivers to the Borrower and the Administrative Agent a Form W-8BEN or W-8ECI pursuant to the next preceding sentence further undertakes to deliver to the Borrower and the Administrative Agent two further copies of Form W-8BEN or W-8ECI, or other applicable forms, or other manner of certification, as the case may be, on or before the date that any such form expires or becomes obsolete or after the occurrence of any event requiring a change in the most recent form previously delivered by it to the Borrower and the Administrative Agent, and such extensions or renewals thereof as may reasonably be requested by the Borrower and the Administrative Agent certifying in the case of a Form W-8BEN or W-8ECI that such Bank is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes. If an event (including without limitation any change in treaty, law or regulation) has occurred prior to the date on which any delivery required by the preceding sentence would otherwise be required which renders all such forms inapplicable or which would prevent any Bank from duly completing and delivering any such form with respect to it and such Bank advises the Borrower and the Administrative Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income tax, such Bank shall not be required to deliver such forms. The Borrower shall withhold tax at the rate and in the manner required by the laws of the United States with respect to payments made to a Bank failing to provide the requisite Internal Revenue Service forms in a timely manner. Each Bank which fails to provide to the Borrower in a timely manner such forms shall reimburse the Borrower upon demand for any penalties paid by the Borrower as a result of any failure of the Borrower to withhold the required amounts that are caused by such Bank's failure to provide the required forms in a timely manner. Notwithstanding any other provision of this paragraph, a Non-U.S. Bank shall not be required to deliver any form pursuant to this paragraph that such Non-U.S. Bank is not legally able to deliver.

2.13. Change of Lending Office.

(a) Each Bank agrees that if it makes any demand for payment under Section 2.8 or is required to pay additional amounts pursuant to Section 2.12(a), or if any adoption or change of the type described in Section 2.9 shall occur with respect to it, it will if requested by the Borrower use reasonable efforts (consistent with its internal policy and legal and regulatory restrictions and so long as such efforts would not be disadvantageous to it, as determined in its sole discretion) to designate a different lending office for any loan made hereunder affected if the making of such a designation would reduce or obviate the need for the Borrower to make payments under Section 2.8 or 2.12(a), or would eliminate or reduce the effect of any adoption or change described in Section 2.9.

(b) If any Bank (including any participant Bank under Section 8.5) shall assert that any adoption or change of the type described in Section 2.9 hereof has occurred with respect to it, or if any Bank (including any participant Bank under Section 8.5) requests compensation under Section 2.8, or if the Borrower is required to pay any additional amount to any Bank or any authority for the account of any Bank pursuant to Section 2.12, then the Borrower may, at its expense and effort, upon notice to such Bank and the Administrative Agent, require such Bank

to, and such Bank promptly shall, assign and delegate, without recourse (in accordance with and subject to the restrictions contained in Section 8.5 (provided that the Borrower shall be obligated to pay the administrative fee referred to therein) and subject to Section 2.7 (as if such assignment were a prepayment by the Borrower), all its interests, rights, and obligations under this Agreement to an assignee that shall assume such obligations (which assignee may be another Bank, if a Bank accepts such assignment); provided that (i) if such assignee is not a Bank or an Affiliate thereof, the Borrower shall have received the prior written consent of the Administrative Agent, the Swing Line Lender and the Issuing Bank (which consents shall not unreasonably be withheld or delayed), (ii) such Bank shall have received payment of an amount equal to the aggregate outstanding principal of such Bank's Term Loan Advances and Revolving Loan Advances and its participations in Letters of Credit, accrued interest thereon, accrued fees and all other amounts payable to it hereunder, from the assignee (at least to the extent of such outstanding principal) and the Borrower (in the case of all other amounts), (iii) in the case of any such assignment resulting from a claim for compensation under Section 2.8 or payment required to be made pursuant to Section 2.12, such assignment will result in a required to be made pursuant to Section 2.12, such assignment will result in a reduction in such compensation or payments compared to the compensation or payments payable to the assigning Bank, (iv) such assignment does not conflict with any requirement of law, (v) no Event of Default shall have occurred and be continuing at the time of such assignment, (vi) prior to any such assignment, such Bank shall have taken no action under Section 2.13(a) so as to eliminate the continued need for payment of amounts owing pursuant to Section 2.8 or 2.12(a), (vii) until such time as such assignment shall be consummated, the Borrower shall pay all additional amounts (if any) required pursuant to Section 2.8 or 2.12(a), as the case may be, and (viii) any such assignment shall not be deemed to be a waiver of any rights that the Borrower, the Administrative Agent or any other Bank shall have against the replaced Bank. A Bank shall not be required to make any such assignment and delegation if, prior thereto, as a result of a waiver by such Bank or otherwise, the circumstances entitling the Borrower to require such assignment and delegation no longer exist or cease to apply.

2.14. Conversion of the Term Loan. Upon the Subordinated Debt Event, the aggregate outstanding principal amount of the Term Loan Borrowing (the "Converted Term Loan Amount") and all Tranches thereunder shall be automatically converted into a Revolving Loan Borrowing of the same principal amount and with the same Tranches. Notwithstanding anything herein to the contrary, upon such conversion, the Term Loan shall be deemed to have been repaid in full, and the Revolving Loan Commitment of each Bank shall be automatically increased by an amount equal to such Bank's ratable share of the Converted Term Loan Amount. The Administrative Agent shall promptly notify each Bank of the increase in its Revolving Loan Commitment.

ARTICLE 3. CONDITIONS PRECEDENT

3.1. Conditions Precedent to Initial Extension of Credit. The obligation of each Bank to make the initial extension of credit under this Agreement, including the making of any Term Loan Advance or Revolving Loan Advance and the issuance of any Letter of Credit, and the obligation of the Swing Line Lender to make any Swing Line Loan shall be subject to the following conditions precedent: (a) Documents. The Borrower shall have delivered or shall have caused to be delivered the documents and other items listed on Exhibit F, together with any other documents requested by the Administrative Agent to document the agreements and intent of the Credit Documents, each in form and with substance satisfactory to the Administrative Agent;

(b) Material Adverse Change. No Material Adverse Change shall have occurred since December 31, 2000; and

(c) Existing Credit Agreement. (i) The Administrative Agent shall have received satisfactory evidence that the Existing Credit Agreement shall have been terminated and all amounts thereunder shall have been paid in full and (ii) satisfactory arrangements shall have been made for the termination of all Liens granted in connection therewith.

3.2. Conditions Precedent to Each Extension of Credit. The obligation of each Bank to make any extension of credit under this Agreement, including the making of any Term Loan Advance or Revolving Loan Advance and the issuance, increase, or extension of any Letter of Credit, and the obligation of the Swing Line Lender to make any Swing Line Loan shall be subject to the further conditions precedent that on the date of such extension of credit:

(a) Representations and Warranties. As of the date of the making of any extension of credit hereunder, the representations and warranties contained in each Credit Document shall be (i) if qualified as to materiality, true and correct or (ii) if not so qualified, true and correct in all material respects, on and as of such date as if made on and as of such date;

(b) Default. As of the date of the making of any extension of credit hereunder, there shall exist no Default or Event of Default, and the making of the extension of credit would not cause a Default or Event of Default; and

(c) Borrowing Base. After giving effect to the extensions of credit requested to be made on such date, the aggregate outstanding principal amount of the Revolving Loan Borrowings plus the aggregate outstanding principal amount of the Swing Line Loans plus the Letter of Credit Exposure shall not exceed the Borrowing Base then in effect. In addition, in the case of (i) a Revolving Loan Borrowing or a borrowing of Swing Line Loans and (ii) an issuance of a Letter of Credit, the Borrower shall have delivered a Borrowing Base Certificate as of the date of such extension of credit.

Each borrowing by and issuance of a Letter of Credit on behalf of the Borrower hereunder shall constitute a representation and warranty by the Borrower as of the date of such extension of credit that the conditions contained in this Section 3.2 have been satisfied.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES.

The Borrower represents and warrants to the Administrative Agent and each Bank, and with each request for any extension of credit hereunder, including the making of any Term Loan Advance or Revolving Loan Advance, and the issuance, increase, or extension of any Letter of

4.1. Organization. As of the date of this Agreement, each Restricted Entity (a) is duly organized, validly existing, and in good standing (or, with respect to any entity listed on Schedule IV, shall be in good standing within 30 days after the date of this Agreement) under the laws of such Person's respective jurisdiction of organization and (b) is duly licensed, qualified to do business, and in good standing in each jurisdiction in which such Person is organized, owns property, or conducts operations to the extent that any failure to be so licensed, qualified, or in good standing in accordance with this clause (b) could reasonably be expected to cause a Material Adverse Change.

4.2. Authorization. The execution, delivery, and performance by each Credit Party of the Credit Documents to which such Credit Party is a party and the consummation of the transactions contemplated thereby (a) do not contravene the organizational documents of such Credit Party, (b) have been duly authorized by all necessary corporate action of each Credit Party, and (c) are within each Credit Party's corporate powers.

4.3. Enforceability. Each Credit Document to which any Credit Party is a party has been duly executed and delivered by each Credit Party which is a party to such Credit Document and constitutes the legal, valid, and binding obligation of each such Credit Party, enforceable against each such Credit Party in accordance with such Credit Document's terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium, or similar laws at the time in effect affecting the rights of creditors generally and subject to the availability of equitable remedies.

4.4. Absence of Conflicts and Approvals. The execution, delivery, and performance by each Credit Party of the Credit Documents to which such Credit Party is a party and the consummation of the transactions contemplated thereby, (a) do not result in any violation or breach of any provisions of, or constitute a default under, any note, indenture, credit agreement, security agreement, credit support agreement, or other similar agreement to which such Credit Party is a party or any other material contract or agreement to which such Credit Party is a party, (b) do not violate any law or regulation binding on or affecting such Credit Party, (c) do not require any authorization, approval, or other action by, or any notice to or filing with, any governmental authority or any other Person, and (d) do not result in or require the creation or imposition of any Lien prohibited by this Agreement.

4.5. Investment Companies. No Restricted Entity or Affiliate thereof is an "investment company" or a company "controlled" by an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

4.6. Public Utilities. No Restricted Entity or Affiliate thereof is a "holding company," or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company" or of a "subsidiary company" of a "holding company," within the meaning of the Public Utility Holding Company Act of 1935, as amended. No Restricted Entity or Affiliate thereof is a regulated public utility.

4.7. Financial Condition.

(a) The Borrower has delivered to the Administrative Agent the Financial Statements, including therein the consolidated balance sheet of the Borrower and consolidated statement of income for the periods shown therein, which accurately and completely, in all material respects, present fairly the financial condition of Borrower as of such date and for such periods. The Borrower has further delivered to the Administrative Agent the Interim Financial Statements, and the Interim Financial Statements are accurate and complete in all material respects and present fairly the financial condition of Borrower as of their date and for their period in accordance with generally accepted accounting principles, as applicable to interim financial reports.

(b) As of the date of the Financial Statements, there were no material contingent obligations, liabilities for taxes, unusual forward or long-term commitments, or unrealized or anticipated losses of the Borrower or any of the Borrower's Subsidiaries, except as disclosed in the Financial Statements and the Interim Financial Statements, and adequate reserves for such items have been made in accordance with generally accepted accounting principles. No Material Adverse Change has occurred since the date of the Financial Statements. No Default exists.

4.8. Condition of Assets. Each Restricted Entity has good and indefeasible title to substantially all of its owned property and valid leasehold rights in all of its leased property, as reflected in the financial statements most recently provided to the Administrative Agent free and clear of all Liens except Permitted Liens. Each Restricted Entity possesses and has properly approved, recorded, and filed, where applicable, all permits, licenses, patents, patent rights or licenses, trademarks, trademark rights, trade names rights, and copyrights which are useful in the conduct of its business and which the failure to possess could reasonably be expected to cause a Material Adverse Change. The material properties used in the operations of each Restricted Entity are in good repair, working order, and condition, normal wear and tear excepted. The properties of each Restricted Entity have not been adversely affected as a result of any fire, explosion, earthquake, flood, drought, windstorm, accident, strike or other labor disturbance, embargo, requisition or taking of property or cancellation of contracts, permits, or concessions by a governmental authority, riot, activities of armed forces, or acts of God or of any public enemy in any manner which (after giving effect to any insurance proceeds) could reasonably be expected to cause a Material Adverse Change.

4.9. Litigation. There are no actions, suits, or proceedings pending or, to the knowledge of the Borrower, threatened against any Restricted Entity at law, in equity, or in admiralty, or by or before any governmental department, commission, board, bureau, agency, instrumentality, domestic or foreign, or any arbitrator which could reasonably be expected to cause a Material Adverse Change.

4.10. Subsidiaries. As of the date of this Agreement, (a) the Borrower has no Subsidiaries except as disclosed in Schedule II, which sets forth the name and jurisdiction of incorporation of each Subsidiary and, as to each such Subsidiary, the number of shares and percentage of each class of capital stock owned by any Credit Party, (b) the Borrower has no Subsidiaries which have not been disclosed in writing to the Administrative Agent and (c) there are no outstanding subscriptions, options, warrants, calls, rights or other agreements or commitments (other than

stock options granted to employees or directors and directors' qualifying shares) of any nature relating to any capital stock of the Borrower or any Subsidiary, except as created by the Credit Documents.

4.11. Laws and Regulations. Each Restricted Entity is in compliance with all federal, state, and local laws and regulations which are applicable to the operations and property of such Person where the failure to comply with the same could reasonably be expected to cause a Material Adverse Change. No part of the proceeds of any Loan Advances, and no other extensions of credit hereunder, will be used for "buying" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U of the Board as now and from time to time hereafter in effect or for any purpose that violates the provisions of the Regulations of the Board.

4.12. Environmental Compliance. Each Restricted Entity has been and is in compliance with all Environmental Laws and has obtained and is in compliance with all related permits necessary for the ownership and operation of any such Person's properties, in each case, where the failure to be in compliance with the same could reasonably be expected to cause a Material Adverse Change. Each Restricted Entity has not received notice of and has not been investigated for any violation or alleged violation of any Environmental Law in connection with any such Person's presently or previously owned properties which currently threaten action or suggest liabilities which could reasonably be expected to cause a Material Adverse Change. Each Restricted Entity does not and has not created, handled, transported, used, or disposed of any Hazardous Materials on or about any such Person's properties (nor has any such Person's properties been used for those purposes); has never been responsible for the release of any Hazardous Materials into the environment in connection with Hazardous Materials; and does not and has not contaminated any properties with Hazardous Materials; and does not and has not owned any properties contaminated by any Hazardous Materials; and materials, in each case in any manner which could reasonably be expected to cause a Material Adverse Change.

4.13. ERISA. Each Restricted Entity and each of their respective Commonly Controlled Entities are in compliance with all provisions of ERISA, the Code and other federal, state or local law with respect to any Plan to the extent that the failure to be in compliance could reasonably be expected to cause a Material Adverse Change. No Restricted Entity nor any of their respective Commonly Controlled Entities participates in or during the past five years has participated in any employee pension benefit plan covered by Title IV of ERISA or any multiemployer plan under Section 4001(a)(3) of ERISA. No Restricted Entity nor any of their respective Commonly Controlled Entities has any liability with respect to any Plan. With respect to the Plans of the Restricted Entities, no Reportable Event or Prohibited Transaction has occurred and exists that could reasonably be expected to cause a Material Adverse Change.

4.14. Taxes. Each Restricted Entity has filed all United States federal, state, and local income tax returns and all other domestic and foreign tax returns which are required to be filed by such Person and has paid, or provided for the payment before the same became delinquent of, all taxes due pursuant to such returns or pursuant to any assessment received by such Person except for tax payments being contested in good faith, for which adequate reserves have been established and reported in accordance with generally accepted accounting principals, and which could not reasonably be expected to cause a Material Adverse Change. The charges, accruals, and reserves

on the books of the Restricted Entities in respect of taxes are adequate in accordance with generally accepted accounting principles.

4.15. True and Complete Disclosure. All factual information furnished by or on behalf of any Credit Party in writing to the Administrative Agent or any Bank in connection with the Credit Documents and the transactions contemplated thereby is true and accurate in all material respects on the date as of which such information was dated or certified and does not contain any untrue statement of material fact or omit to state any material fact necessary to make the statements contained therein not misleading. All projections, estimates, and pro forma financial information furnished by any Credit Party were prepared on the basis of assumptions, data, information, tests, or conditions believed to be reasonable at the time such projections, estimates, and pro forma financial information were furnished.

4.16. Senior Indebtedness. The Credit Obligations constitute "Senior Indebtedness" (or any other defined term having a similar purpose) of the Borrower under and as defined in the Senior Subordinated Note Indentures and any Subordinated Debt Indenture. The obligations of each Guarantor under the Guaranty constitute "Guarantor Senior Indebtedness" (or any other defined term having a similar purpose) of such Guarantor under and as defined in the Senior Subordinated Note Indentures and any Subordinated Debt Indenture.

4.17. Certain Documents. The Borrower has delivered to the Administrative Agent a complete and correct copy of the Senior Subordinated Note Indentures and any Subordinated Debt Indenture, in each case including any amendments, supplements or modifications with respect to any of the foregoing.

4.18. Shell Subsidiaries. Each Shell Subsidiary is a "shell" company having (a) no assets (other than equity interests in a Subsidiary or other assets with an aggregate value not exceeding \$10,000) and (b) no operations.

ARTICLE 5. COVENANTS

Until the Administrative Agent and the Banks receive irrevocable payment of the Credit Obligations and have terminated this Agreement and each other Credit Document, the Borrower shall comply with and cause compliance with the following covenants:

5.1. Organization. The Borrower shall cause each Restricted Entity to (a) maintain itself as an entity duly organized, validly existing, and in good standing under the laws of such Person's respective jurisdiction of organization and (b) be duly licensed, qualified to do business, and in good standing in each jurisdiction in which such Person is organized, owns property, or conducts operations and which requires such licensing or qualification where failure to be so licensed, qualified, or in good standing as required by this clause (b) could reasonably be expected to cause a Material Adverse Change; provided, however, that nothing in this Section 5.1 shall be interpreted to be violated as a result of a transaction permitted by Section 5.9.

5.2. Reporting. The Borrower shall furnish to the Administrative Agent all of the following:

(a) Annual Reports. As soon as available and in any event not later than 90 days after the end of each fiscal year of the Borrower, (i) a copy of the annual audit report for such fiscal year for the Borrower, including therein the consolidated balance sheets of the Borrower as of the end of such fiscal year and the consolidated statements of income, stockholders' equity, and cash flows for the Borrower for such fiscal year, setting forth the consolidated financial position and results of the Borrower for such fiscal year and certified, without any qualification or limit of the scope of the examination of matters relevant to the financial statements, by a nationally recognized certified public accounting firm, (ii) a completed Compliance Certificate duly certified by a Responsible Officer of the Borrower, and (iii) a completed Contract Status Report duly certified by a Responsible Officer of the Borrower for the Borrower;

(b) Quarterly Reports. As soon as available and in any event not later than 45 days after the end of each of the first three fiscal quarters of the Borrower of each year, and in each case in form and substance acceptable to the Administrative Agent, (i) a copy of the internally prepared consolidated financial statements of the Borrower for such fiscal quarter and for the fiscal year to date period ending on the last day of such fiscal quarter, including therein the consolidated balance sheets of the Borrower as of the end of such fiscal quarter and the consolidated statements of income, and cash flows for such fiscal quarter and for such fiscal year to date period, setting forth the consolidated financial position and results of the Borrower for such fiscal quarter and fiscal year to date period, all in reasonable detail and duly certified by a Responsible Officer of the Borrower as having been prepared in accordance with generally accepted accounting principles, including those applicable to interim financial reports which permit normal year end adjustments and do not require complete financial notes, (ii) a completed Compliance Certificate duly certified by a Responsible Officer of the Borrower, and (iii) a completed Contract Status Report duly certified by a Responsible Officer of the Borrower;

(c) Semi-annual Subsidiary Update. As soon as available and in any event not later than 45 days after the end of the second fiscal quarter of the Borrower of each year and 90 days after the end of the fourth fiscal quarter of the Borrower of each year, an update of the information included in Schedule II, including all Subsidiaries acquired by the Borrower since the previous update of such scheduled information.

(d) Acquisition Information. As soon as available prior to the closing of any Acquisition requiring approval of the Majority Banks, and on or prior to the closing of any Acquisition not requiring such approval, a completed Acquisition Certificate duly certified by a Responsible Officer of the Borrower, which the Administrative Agent shall forward to the Banks for any Acquisition requiring approval of the Majority Banks (and prior to the consummation of the Acquisition, the Borrower shall, upon request by the Administrative Agent, make available to the Administrative Agent at the Borrower's offices in Houston, Texas, the acquisition documents regarding the acquired assets, including schedules reflecting litigation liabilities, environmental liabilities, and other assumed liabilities, and any other information regarding the acquired assets as the Administrative Agent may reasonably request);

(e) SEC Filings. As soon as available and in any event not later than 30 days after the filing or delivery thereof, copies of all financial statements, reports, and proxy statements which the Borrower shall have sent to its stockholders generally and copies of all regular and periodic

reports, if any, which any Restricted Entity shall have filed with the Securities and Exchange Commission;

52

(f) Consolidated Budget and Projections. As soon as available, and in any event no later than 45 days after the end of each fiscal year of the Borrower, a detailed consolidated budget for the following fiscal year (including a projected consolidated balance sheet of the Borrower and its Subsidiaries as of the end of the following fiscal year, the related consolidated statements of projected cash flow, projected changes in financial position and projected income and a description of the underlying assumptions applicable thereto), and, within 5 days after the availability thereof, significant revisions, if any, of such budget and projections with respect to such fiscal year (collectively, the "Projections"), which Projections shall in each case be accompanied by a certificate of a Responsible Officer of the Borrower stating that such Projections are based on reasonable estimates, information and assumptions and made in good faith;

(g) Defaults. Promptly, but in any event within 5 Business Days after the discovery thereof, a notice of any facts known to a Responsible Officer of the Borrower which constitute a Default, together with a statement of a Responsible Officer of the Borrower setting forth the details of such facts and the actions which the Borrower shall, promptly upon receipt from the Borrower of a notice pursuant to this Section 5.2(g), forward a copy of such notice to each Bank);

(h) Litigation. Promptly, but in any event within 10 Business Days after notice of commencement thereof, notice of all actions, suits, and proceedings before or brought by any court or governmental department, commission, board, bureau, agency, or instrumentality, domestic or foreign, affecting any Restricted Entity which, if determined adversely, could reasonably be expected to cause a Material Adverse Change;

(i) Material Agreement Default. Promptly, but in any event within 10 Business Days after a Responsible Officer of the Borrower obtains knowledge thereof, notice of any breach by any Restricted Entity of any contract or agreement which breach could reasonably be expected to cause a Material Adverse Change;

(j) Material Changes. Prompt written notice of any other condition or event of which a Responsible Officer of the Borrower has knowledge, which condition or event has resulted in or could reasonably be expected to cause a Material Adverse Change;

(k) Indentures; Senior Subordinated Notes. (i) As soon as available, a complete and correct copy of each of the Senior Subordinated Note Indentures and any Subordinated Debt Indenture; and (ii) no later than 5 Business Days prior to the effectiveness thereof, copies of substantially final drafts of any proposed amendment, supplement, waiver or other modification with respect to the Senior Subordinated Note Indentures or any Subordinated Debt Indentures.

(1) Borrowing Base Certificate. Within 25 days after the last day of each calendar month, a Borrowing Base Certificate setting forth the Borrowing Base as of such last day; and

(m) Other Information. Such other information respecting the business operations or property of any Restricted Entity, financial or otherwise, as the Administrative Agent or the Majority Banks may from time to time reasonably request.

5.3. Inspection. The Borrower shall cause each Restricted Entity to permit the Administrative Agent and the Banks to visit and inspect any of the properties of such Restricted Entity, to examine all of such Person's books of account, records, reports, and other papers, to make copies and extracts therefrom, and to discuss their respective affairs, finances, and accounts with their respective officers, employees, and independent public accountants all at such reasonable times and as often as may be reasonably requested, provided that the Borrower is given at least 3 Business Days' advance notice thereof and reasonable opportunity to be present when independent public accountants or other third parties are contacted, and provided further that so long as no Default or Event of Default exists, the Administrative Agent and the Banks shall not exercise the foregoing inspection right more often than once in any calendar year.

5.4. Use of Proceeds. The proceeds of the Term Loan Borrowing and the Revolving Loan Borrowings shall be used by the Borrower for Acquisitions, working capital needs, Capital Expenditures, and for other lawful corporate purposes.

5.5. Financial Covenants. The Administrative Agent shall determine compliance with the following financial covenants based upon the applicable Schedule of the most recent Compliance Certificate delivered to the Administrative Agent pursuant to Section 5.2(a) or 5.2(b).

(a) Net Worth. The Borrower shall not permit the consolidated Net Worth of the Borrower as of the last day of each fiscal quarter to be less than the sum of (i) \$456,974,000; plus (ii) 75% of the cumulative quarterly consolidated net income of the Borrower after September 30, 2000, for each fiscal quarter of the Borrower during which the Borrower has positive consolidated net earnings; plus (ii) 100% of the net proceeds received by Borrower after September 30, 2000, from any sale or issuance of any equity securities of, or any other additions to capital by, the Borrower or its Subsidiaries; plus (iv) to the extent that the required consolidated Net Worth under this Section 5.5(a) was not increased in clauses (i) through (iii) above as a result of any Acquisition, 100% of any increase in the consolidated Net Worth of the Borrower resulting from any Acquisition. Compliance with this paragraph (a) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule A of such Compliance Certificate.

(b) Maximum Leverage Ratios.

(i) Maximum Senior Debt to EBITDA Ratio. As of the last day of each fiscal quarter of the Borrower, the Borrower shall not permit the ratio of (a) the consolidated Senior Debt of the Borrower as of end of such fiscal quarter to (b) the consolidated EBITDA of the Borrower for the preceding four fiscal quarters then ended, to be greater than 2.00 to 1.00.

(ii) Maximum Total Debt to EBITDA Ratio. As of the last day of each fiscal quarter of the Borrower, the Borrower shall not permit the ratio of (a) the consolidated Total Debt of the Borrower as of end of such fiscal quarter to (b) the consolidated EBITDA of the Borrower for the preceding four fiscal quarters then ended, to be greater than 3.25 to 1.00.

Compliance with this paragraph (b) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

(c) Minimum Interest Coverage Ratio. As of the last day of each fiscal quarter, the Borrower shall not permit the ratio of (i) the consolidated EBIT of the Borrower for the preceding four fiscal quarters then ended to (ii) the consolidated Interest Expense of the Borrower for the preceding four fiscal quarters then ended to be less than 2.50 to 1.00. Compliance with this paragraph (c) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule A of such Compliance Certificate.

(d) Capital Expenditures. The Borrower shall not permit the consolidated Capital Expenditures (other than Capital Expenditures that are deemed to occur because of the making of an Acquisition) of the Borrower during any fiscal year to exceed an amount equal to 10% of the consolidated Net Worth of the Borrower as of the last day of the immediately preceding fiscal year of the Borrower; provided, that with respect to a determination including any fiscal quarter which involves Persons which were not Restricted Entities for the entire applicable period of determination, the Capital Expenditures of each such Person for each such quarterly period shall be deemed to be equal to the quotient obtained by dividing (i) the consolidated Capital Expenditures of such Person for the fiscal year then most recently ended, by (ii) 4. Compliance with this paragraph (d) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

5.6. Debt.

(a) The Borrower shall not permit any Restricted Entity to create, assume, incur, suffer to exist, or in any manner become liable, directly, indirectly, or contingently in respect of, any Debt other than Permitted Debt.

(b) The Borrower shall, as soon as available but in any event not less than 10 Business Days prior to the issuance of any preferred stock or subordinated indebtedness, deliver to the Administrative Agent a copy of the certificate of designation or subordinated debt documents, as applicable, together with a certificate, signed by a Responsible Officer of the Borrower, certifying that such preferred stock or subordinated indebtedness constitutes Qualified Preferred Stock or Subordinated Debt pursuant to the terms of this Agreement.

5.7. Liens. The Borrower shall not permit any Restricted Entity to create, assume, incur, or suffer to exist any Lien on any of its real or personal property whether now owned or hereafter acquired, or assign any right to receive its income, except for Permitted Liens.

5.8. Other Obligations.

(a) The Borrower shall not permit any Restricted Entity to create, incur, assume, or suffer to exist any obligations in respect of unfunded vested benefits under any pension Plan or deferred compensation agreement in an aggregate outstanding amount in excess of \$5,000,000.

(b) The Borrower shall not permit any Restricted Entity to create, incur, assume, or suffer to exist any obligations in respect of Derivatives, other than Derivatives used by any Restricted Entity in such Restricted Entity's respective business operations in aggregate notional quantities not to exceed the reasonably anticipated consumption of such Restricted Entity of the underlying commodity for the relevant period, but no Derivatives which are speculative in nature.

5.9. Corporate Transactions. The Borrower shall not, without the Majority Banks' consent, permit any Restricted Entity to (a) merge, consolidate, or amalgamate with another Person, or liquidate, wind up, or dissolve itself (or take any action towards any of the foregoing), (b) convey, sell, lease, assign, transfer, or otherwise dispose of any of its property, businesses, or other assets outside of the ordinary course of business, or (c) make any Acquisition except that:

(i) Any Subsidiary of the Borrower may merge, consolidate, or amalgamate into the Borrower or any wholly owned Subsidiary of the Borrower, or convey, sell, lease, assign, transfer, or otherwise dispose of any of its assets to the Borrower or any wholly owned Subsidiary of the Borrower (and if such disposition transfers all or substantially all of the assets of the transferring Subsidiary, such Subsidiary may then liquidate, wind up, or dissolve itself); provided that the Borrower or the wholly owned Subsidiary, as applicable, is the surviving or acquiring entity;

(ii) At any time after the Subordinated Debt Event, the Borrower or any Subsidiary of the Borrower may make any Acquisition (by purchase or merger) provided that (A) the Borrower or such Subsidiary of the Borrower is the acquiring or surviving entity, (B) the aggregate of all consideration (other than common stock of the Borrower) paid by the Restricted Entities in connection with any Acquisition does not exceed \$30,000,000 without the prior consent of the Majority Banks, (C) the aggregate of all consideration (other than common stock of the Borrower) paid by the Restricted Entities in connection with all Acquisitions during any fiscal year does not exceed the Annual Aggregate Acquisition Limit for the applicable year without the prior consent of the Majority Banks, (D) no Default or Event of Default exists and the Acquisition would not reasonably be expected to cause a Default or Event of Default after giving pro forma effect thereto (including any default under Section 5.5 with respect to historical and future pro forma financial status and results), and (E) the acquired assets are in substantially the same business as the Borrower;

(iii) The Borrower or any Subsidiary of the Borrower may sell, lease, assign, transfer, or otherwise dispose of any of its property, businesses, or other assets outside of the ordinary course of business in an amount not to exceed, in the aggregate, \$5,000,000 in any fiscal year of the Borrower.

5.10. Distributions. The Borrower shall not (a) declare or pay any dividends; (b) purchase, redeem, retire, or otherwise acquire for value any of its capital stock now or hereafter outstanding; (c) make any distribution of assets to its stockholders as such, whether in cash, assets, or in obligations of it; (d) allocate or otherwise set apart any sum for the payment of any dividend or distribution on, or for the purchase, redemption, or retirement of, any shares of its capital stock; or (e) make any other distribution by reduction of capital or otherwise in respect of any shares of its capital stock, except that the Borrower may make payments of dividends on Qualified Preferred Stock; provided, however, that at any time after the Subordinated Debt Event

and so long as no Default is continuing or would result therefrom, the Borrower may repurchase from the holders thereof capital stock of the Borrower in an aggregate amount during the term of this Agreement not to exceed an amount equal to the lesser of (i) the sum of (a) \$10,000,000 and (b) 50% of the cumulative quarterly consolidated net income (less 100% of the cumulative quarterly consolidated net loss) of the Borrower for each fiscal quarter of the Borrower commencing on the first day of the fiscal quarter in which the Subordinated Debt Event occurs and ending on the last day of the fiscal quarter then most recently ended prior to the date of any such repurchase for which financial statements have been delivered to the Banks pursuant to Section 5.2 and (ii) \$30,000,000.

5.11. Transactions with Affiliates. The Borrower shall not permit any Restricted Entity to enter into any transaction directly or indirectly with or for the benefit of an Affiliate except transactions with an Affiliate for the leasing of property, the rendering or receipt of services, or the purchase or sale of inventory or other assets in the ordinary course of business if the monetary or business consideration arising from such a transaction would be substantially as advantageous to such Restricted Entity as the monetary or business consideration which such Restricted Entity would obtain in a comparable arm's length transaction; provided that the Restricted Entities may make the investments described in item (g) of the definition of "Permitted Investments".

5.12. Insurance.

(a) The Borrower shall cause each Restricted Entity to maintain insurance with responsible and reputable insurance companies or associations reasonably acceptable to the Administrative Agent in such amounts and covering such risks as are usually carried by companies engaged in similar businesses and owning similar properties in the same general areas in which such Persons operate. The Borrower shall deliver to the Administrative Agent certificates evidencing such policies or copies of such policies at the Administrative Agent's request following a reasonable period to obtain such certificates taking into account the jurisdiction where the insurance is maintained.

(b) All policies representing liability insurance of the Restricted Entities shall name the Administrative Agent and the Banks as additional named insureds in a form satisfactory to the Administrative Agent. With respect to any insured liability suffered by the Administrative Agent or any Bank, all proceeds of such liability insurance coverage for the Administrative Agent and the Banks shall be paid as directed by the Administrative Agent to indemnify the Administrative Agent or the applicable Bank for the liability covered. In the event that proceeds of property or liability insurance are paid to any Restricted Entity in violation of the foregoing, the Restricted Entity shall hold the proceeds in trust for the Administrative Agent, segregate the proceeds from the other funds of such Restricted Entity, and promptly pay the proceeds to the Administrative Agent with any necessary endorsement. The Administrative Agent shall have the right, but not the obligation, during the existence of an Event of Default, to make proof of loss under, settle and adjust any claim under, and receive the proceeds under the insurance, and the reasonable expenses incurred by the Administrative Agent in the adjustment and collection of such proceeds shall be paid by the Borrower. The Borrower irrevocably appoints the Administrative Agent as its attorney in fact to take such actions in its name. If the Administrative Agent does not take such actions, the Borrower may take such actions subject to the approval of any final action by

the Administrative Agent. The Administrative Agent shall not be liable or responsible for failure to collect or exercise diligence in the collection of any proceeds.

5.13. Investments. The Borrower shall not permit any Restricted Entity to make or hold any direct or indirect investment in any Person, including capital contributions to the Person, investments in the debt or equity securities of the Person, and loans, guaranties, trade credit, or other extensions of credit to the Person, except for Permitted Investments.

5.14. Lines of Business. The Borrower shall not permit the Restricted Entities to change the character of their business as conducted on the date of this Agreement, or engage in any type of business not reasonably related to such business as presently and normally conducted. The parties agree that business entailing the construction, operation, or ownership of electrical infrastructure, electrical generation, electrical distribution, distributed generation, communications contracting, communications servicing, e-commerce ventures and other related ventures are reasonably related to the character of the Restricted Entities' business as conducted on the date hereof.

5.15. Compliance with Laws. The Borrower shall cause each Restricted Entity to comply with all federal, state, and local laws and regulations which are applicable to the operations and property of such Persons, in each case, where the failure to comply could reasonably be expected to cause a Material Adverse Change.

5.16. Environmental Compliance. The Borrower shall cause each Restricted Entity to comply with all Environmental Laws and obtain and comply with all related permits necessary for the ownership and operation of any such Person's properties, in each case, where the failure to comply could reasonably be expected to cause a Material Adverse Change. The Borrower shall cause each Restricted Entity to disclose promptly to the Administrative Agent any notice to or investigation of such Persons for any violation or alleged violation of any Environmental Law in connection with any such Person's presently or previously owned properties which represent liabilities which, if determined adversely to such Restricted Entity, could reasonably be expected to cause a Material Adverse Change. The Borrower shall not permit any Restricted Entity to create, handle, transport, use, or dispose of any Hazardous Materials on or about any such Person's properties; release any Hazardous Materials into the environment in connection with any such Person's operations or contaminate any properties with Hazardous Materials; or own properties contaminated by any Hazardous Materials, in each case in any manner that could reasonably be expected to cause a Materials, Adverse Change.

5.17. ERISA Compliance. The Borrower shall cause each Restricted Entity to (i) comply in all material respects with all applicable provisions of ERISA, the Code and other federal, state or local laws and prevent the occurrence of any Reportable Event or Prohibited Transaction with respect to, or the termination of, any of their respective Plans, in each case, where the failure to do so could reasonably be expected to cause a Material Adverse Change and (ii) not create or participate in any employee pension benefit plan covered by Title IV of ERISA or any multiemployer plan under Section 4001(a)(3) of ERISA.

5.18. Payment of Certain Claims. The Borrower shall cause each Restricted Entity to pay and discharge, before the same shall become delinquent, (a) all taxes, assessments, levies, and like

charges imposed upon any such Person or upon any such Person's income, profits, or property by authorities having competent jurisdiction prior to the date on which penalties attach thereto except for tax payments being contested in good faith for which adequate reserves have been established and reported in accordance with generally accepted accounting principals which could not reasonably be expected to cause a Material Adverse Change and (b) all trade payables and current operating liabilities, unless the same are less than 90 days past due or are being contested in good faith, have adequate reserves established and reported in accordance with generally accepted accounting principals, and could not reasonably be expected to cause a Material Adverse Change.

5.19. Subsidiaries. Upon the formation or acquisition of any new Subsidiary, the Borrower shall and shall cause such Subsidiary to promptly, but in any event within 30 days after the formation or acquisition of such new Subsidiary (a) execute and deliver to the Administrative Agent a Joinder Agreement (each, a "Joinder Agreement") in substantially the form of Exhibit G for the purpose of joining such Subsidiary as a party to the Guaranty, the Security Agreement, and, if applicable, the Stock Pledge Agreement or the Partnership Pledge Agreement, and (b) provide to the Administrative Agent the other certificates, documents, and items listed on Exhibit I.

5.20. Changes in Fiscal Periods. The Borrower shall not permit the fiscal year of the Borrower to end on a day other than September 30 or change the Borrower's method of determining fiscal quarters.

5.21. Negative Pledge Clauses. The Borrower shall not enter into or suffer to exist or become effective any agreement that prohibits or limits the ability of any Restricted Entity to create, incur, assume or suffer to exist any Lien upon any of its property or revenues, whether now owned or hereafter acquired, to secure its obligations under the Credit Documents to which it is a party other than (a) this Agreement and the other Credit Documents and (b) any agreements governing any purchase money Liens or Capital Lease Obligations otherwise permitted hereby (in which case, any prohibition or limitation shall only be effective against the assets financed thereby).

5.22. Shell Subsidiaries. The Borrower shall not permit any Shell Subsidiary to have (a) assets (other than equity interests in a Subsidiary or other assets with an aggregate value not exceeding \$10,000) or (b) operations.

ARTICLE 6. DEFAULT AND REMEDIES

6.1. Events of Default. Each of the following shall be an "Event of Default" for the purposes of this Agreement and for each of the Credit Documents:

(a) Payment Failure. The Borrower (i) fails to pay when due any principal amounts due under this Agreement or any other Credit Document or (ii) fails to pay when due any interest, fees, reimbursements, indemnifications, or other amounts due under this Agreement or any other Credit Document and such failure has not been cured within five Business Days;

(b) False Representation. Any written representation or warranty made by any Credit Party or any Responsible Officer thereof in this Agreement or in any other Credit Document proves to have been false or erroneous in any material respect at the time it was made or deemed made;

(c) Breach of Covenant. (i) Any breach by the Borrower of any of the covenants contained in Sections 5.1(a) (with respect to the Borrower), 5.2, 5.3, 5.4, 5.5, 5.6, 5.7, 5.8, 5.9, 5.10, 5.13 or 5.19 or (ii) any breach by any Credit Party of any other covenants contained in this Agreement, or any other Credit Document and such breach is not cured within 30 days following the earlier of knowledge of such breach by a Responsible Officer of such Credit Party or the receipt of written notice thereof from the Administrative Agent;

(d) Security and Support Documents. Any Security Document shall at any time and for any reason (other than one within the reasonable control of any Bank) cease to create the Lien on the property purported to be subject to such agreement in accordance with the terms of such agreement, or cease to be in full force and effect, or shall be contested by any party thereto;

(e) Guaranty. (i) The Guaranty shall at any time and for any reason cease to be in full force and effect with respect to any Guarantor (except as permitted under Section 5.9 hereof) or shall be contested by any Guarantor, or any Guarantor shall deny it has any further liability or obligation thereunder, or (ii) any breach by any Guarantor of any of the covenants contained in Section 1 of the Guaranty;

(f) Material Debt Default. (i) Any principal, interest, fees, or other amounts due on any Debt of any Restricted Entity (other than the Credit Obligations) is not paid when due, whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise, and such failure is not cured within the applicable grace period, if any, and the aggregate amount of all Debt of such Persons so in default exceeds \$2,000,000; (ii) any other event shall occur or condition shall exist under any agreement or instrument relating to any Debt of any such Person (other than the Credit Obligations) the effect of which is to accelerate or to permit the acceleration of the maturity of any such Debt, whether or not any such Debt is actually accelerated, and such event or condition shall not be cured within the applicable grace period, if any, and the aggregate amount of all Debt of such Persons so in default exceeds \$2,000,000; or (iii) any Debt of any such Person shall be declared to be due and payable, or required to be prepaid (other than by a regularly scheduled prepayment) prior to the stated maturity thereof, and the aggregate amount of all Debt of such Persons so accelerated exceeds \$2,000,000;

(g) Material Agreement Default; Material Plan Event. (i) There shall occur any breach by any Restricted Entity of any contract or agreement which breach could reasonably be expected to cause a Material Adverse Change and such breach is not cured within the applicable grace period, if any; or (ii) there shall occur or exist an event with respect to any Plan which could in the sole judgment of the Administrative Agent reasonably be expected to have a Material Adverse Change;

(h) Bankruptcy and Insolvency. (i) There shall have been filed against any Restricted Entity or any such Person's properties, without such Person's consent, any petition or other request for relief seeking an arrangement, receivership, reorganization, liquidation, or similar

relief under bankruptcy or other laws for the relief of debtors, and such request for relief (A) remains in effect for 60 or more days, whether or not consecutive, or (B) is approved by a final nonappealable order, or (ii) any such Person consents to or files any petition or other request for relief of the type described in clause (i) above seeking relief from creditors, makes any assignment for the benefit of creditors or other arrangement with creditors, or admits in writing such Person's inability to pay such Person's debts as they become due (the occurrence of any Event of Default under clause (i) or (ii) of this paragraph being a "Bankruptcy Event of Default");

(i) Adverse Judgment. The aggregate outstanding amount of judgments against the Restricted Parties not discharged or stayed pending appeal or other court action within 30 days following entry is greater than \$2,000,000;

(j) Change of Control. There shall occur any Change of Control; or

(k) Senior Subordinated Notes Default. (i) There shall occur any default or event of default (and such event or condition is not cured within the applicable grace period, if any), however denominated, under the Senior Subordinated Notes or either Senior Subordinated Note Indenture or any Subordinated Debt or any Subordinated Debt Indenture; (ii) any modification shall be made to the subordinated Debt Indenture; (iii) any modification shall be made to the subordinated Debt Indenture without terms of the Senior Subordinated Notes or either Senior Subordinated Note Indenture or any Subordinated Debt or any Subordinated Debt Indenture without the prior written consent of the Majority Banks; or (iii) any "Change of Control Offer" (or any other defined term having a similar purpose) as defined in either Senior Subordinated Note Indenture or any Subordinated Debt Indenture shall occur.

6.2. Termination of Commitments. Upon the occurrence of any Bankruptcy Event of Default, all of the Commitments of the Administrative Agent and the Banks hereunder shall terminate. During the existence of any Event of Default other than a Bankruptcy Event of Default, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower all of the Commitments of the Administrative Agent and the Banks hereunder terminated, whereupon the same shall immediately terminate.

6.3. Acceleration of Credit Obligations. Upon the occurrence of any Bankruptcy Event of Default, the aggregate outstanding principal amount of all loans made hereunder, all accrued interest thereon, and all other Credit Obligations shall immediately and automatically become due and payable. During the existence of any Event of Default other than a Bankruptcy Event of Default, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower the aggregate outstanding principal amount of all loans made hereunder, all accrued interest thereon, and all other Credit Obligations to be immediately due and payable, whereupon the same shall immediately become due and payable. In connection with the foregoing, except for the notice provided for above in this Article VI, the Borrower waives notice of any Default or Event of Default, grace, notice of intent to accelerate, notice of acceleration, presentment, demand, notice of nonpayment, protest, and all other notices.

6.4. Cash Collateralization of Letters of Credit. Upon the occurrence of any Bankruptcy Event of Default, the Borrower shall pay to the Administrative Agent an amount equal to the Letter of Credit Exposure to be held in the Letter of Credit Collateral Account for disposition in

accordance with Section 2.3(g). During the existence of any Event of Default other than a Bankruptcy Event of Default, the Administrative Agent shall at the request of the Majority Banks require by written notice to the Borrower that the Borrower pay to the Administrative Agent an amount equal to the Letter of Credit Exposure to be held in the Letter of Credit Collateral Account for disposition in accordance with Section 2.3(g), whereupon the Borrower shall pay to the Administrative Agent such amount for such purpose.

6.5. Default Interest. If any Event of Default exists based upon a default in the payment of any amounts owing hereunder, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower that the Credit Obligations specified in such notice shall bear interest beginning on the date specified in such notice until paid in full at the applicable Default Rate for such Credit Obligations, whereupon the Borrower shall pay such interest to the Administrative Agent for the benefit of the Administrative Agent and the Banks, as applicable, upon demand by the Administrative Agent. If any other Event of Default exists, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower that, unless such Event of Default is cured to the satisfaction of the Administrative Agent and the Majority Banks on or before the 30th day following the occurrence of such Event of Default, the Credit Obligations specified in such notice shall bear interest beginning on such 30th day until paid in full at the applicable Default Rate for such Credit Obligations, whereupon the Borrower shall, if such Event of Default is not cured by such date, pay such interest to the Administrative Agent for the benefit of the Administrative Agent and the Banks, as applicable, upon demand by the Administrative Agent after such date.

6.6. Right of Setoff. During the existence of an Event of Default, each of the Administrative Agent and the Banks is hereby authorized at any time, to the fullest extent permitted by law, to set off and apply any indebtedness owed by the Administrative Agent or such Bank to the Borrower against any and all of the obligations of the Borrower under this Agreement and the other Credit Documents, irrespective of whether or not the Administrative Agent or such Bank shall have made any demand under this Agreement or the other Credit Documents and although such obligations may be contingent and unmatured. Each of the Administrative Agent and the Banks agrees promptly to notify the Borrower after any such setoff and application made by such party provided that the failure to give such notice shall not affect the validity of such setoff and application.

6.7. Actions Under Credit Documents. Following an Event of Default, the Administrative Agent shall at the request of the Majority Banks take any and all actions permitted under the other Credit Documents, including the Guaranty and the Security Documents.

6.8. Remedies Cumulative. No right, power, or remedy conferred to the Administrative Agent or the Banks in this Agreement and the other Credit Documents, or now or hereafter existing at law, in equity, by statute, or otherwise, shall be exclusive, and each such right, power, or remedy shall to the full extent permitted by law be cumulative and in addition to every other such right, power, or remedy. No course of dealing and no delay in exercising any right, power, or remedy conferred to the Administrative Agent or the Banks in this Agreement and the other Credit Documents, or now or hereafter existing at law, in equity, by statute, or otherwise, shall operate as a waiver of or otherwise prejudice any such right, power, or remedy.

6.9. Application of Payments. Prior to the Revolving Loan Maturity Date or any acceleration of the Credit Obligations, all payments made hereunder shall be applied to the Credit Obligations as directed by the Borrower, subject to the rules regarding the application of payments to certain Credit Obligations provided for hereunder and in the other Credit Documents. Following the Revolving Loan Maturity Date or any acceleration of the Credit Obligations, all payments and collections shall be applied to the Credit Obligations in the following order:

First, to the payment of the costs, expenses, reimbursements (other than reimbursement obligations with respect to draws under Letters of Credit), and indemnifications of the Administrative Agent that are due and payable under the Credit Documents;

Then, ratably to the payment of the costs, expenses, reimbursements (other than reimbursement obligations with respect to draws under Letters of Credit), and indemnifications of the Banks that are due and payable under the Credit Documents;

Then, ratably to the payment of all outstanding principal of all loans made hereunder, all reimbursement obligations for draws under Letters of Credit, all accrued but unpaid interest and fees under this Agreement and all obligations then due and payable under Interest Hedge Agreements;

Then, ratably to the payment of any other amounts due and owing with respect to the Credit Obligations; and

Finally, any surplus held by the Administrative Agent and remaining after payment in full of all the Credit Obligations and reserve for Credit Obligations not yet due and payable shall be promptly paid over to the Borrower or to whomever may be lawfully entitled to receive such surplus. All applications shall be distributed in accordance with Section 2.11(a).

ARTICLE 7. THE ADMINISTRATIVE AGENT AND THE ISSUING BANK

7.1. Authorization and Action. Each Bank hereby appoints and authorizes the Administrative Agent to take such action as agent on its behalf and to exercise such powers under this Agreement as are delegated to the Administrative Agent by the terms hereof and of the other Credit Documents, together with such powers as are reasonably incidental thereto. Statements under the Credit Documents that the Administrative Agent may take certain actions, without further qualification, means that the Administrative Agent may take such actions with or without the consent of the Banks or the Majority Banks, but where the Credit Documents expressly require the determination of the Banks or the Majority Banks, the Administrative Agent shall not take any such action without the prior written consent thereof. As to any matters not expressly provided for by this Agreement or collection of the Term Loan Notes or the Revolving Loan Notes), the Administrative Agent shall not be required to exercise any discretion or take any action, but shall be required to act or to refrain from acting (and shall be fully protected in so acting or refraining from acting) upon the written instructions of the Majority Banks, and such instructions shall be binding upon all Banks and all holders of Term Loan Notes and Revolving Loan Notes; provided, however, that the Administrative Agent shall not be required to act or to refrain from acting Loan Notes; provided, however, that the Administrative Agent shall not be required to take any action which

exposes the Administrative Agent to personal liability or which is contrary to this Agreement, any other Credit Document, or applicable law.

7.2. Reliance, Etc. Neither the Administrative Agent, the Issuing Bank, nor any of their respective Related Parties (for the purposes of this Section 7.2, collectively, the "Indemnified Parties") shall be liable for any action taken or Agreement or the other Credit Documents, INCLUDING ANY INDEMNIFIED PARTY'S OWN NEGLIGENCE, except for any Indemnified Party's gross negligence or willful misconduct. Without limitation of the generality of the foregoing, the Administrative Agent and the Issuing Bank: (a) may treat the payee of any Term Loan Note or Revolving Loan Note as the holder thereof until the Administrative Agent receives written notice of the assignment or transfer thereof signed by such payee and in form satisfactory to the Administrative Agent; (b) may consult with legal counsel (including counsel for the Borrower), independent public accountants, and other experts selected by it and shall not be liable for any action taken or omitted to be taken in good faith by it in accordance with the advice of such counsel, accountants, or experts; (c) makes no warranty or representation to any Bank and shall not be responsible to any Bank for any Agreements, warranties, or representations made in or in connection with this Agreement or the other Credit Documents; (d) shall not have any duty to ascertain or to inquire as to the performance or observance of any of the terms, covenants, or conditions of this Agreement or any other Credit Document on the part of the Credit Parties or to inspect the property (including the books and records) of the Credit Parties; (e) shall not be responsible to any Bank for the due execution, legality, validity, enforceability, genuineness, sufficiency, or value of this Agreement or any other Credit Document; and (f) shall incur no liability under or in respect of this Agreement or any other Credit Document by acting upon any notice, consent, certificate, or other instrument or writing (which may be by telecopier or telex) reasonably believed by it to be genuine and signed or sent by the proper party or parties.

7.3. Affiliates. With respect to its Commitments, the Loan Advances made by it, its interests in the Letters of Credit, and the Term Loan Notes and the Revolving Loan Notes issued to it, the Administrative Agent and the Issuing Bank shall have the same rights and powers under this Agreement as any other Bank and may exercise the same as though it were not the Administrative Agent. The term "Bank" or "Banks" shall, unless otherwise expressly indicated, include the Administrative Agent, the Issuing Bank, and their respective Affiliates may accept deposits from, lend money to, act as trustee under indentures of, and generally engage in any kind of business with, any Credit Party, and any Person who may do business with or own securities of any Credit Party, all as if the Administrative Agent were not an agent hereunder and the Issuing Bank were not the issuer of Letters of Credit hereunder and without any duty to account therefor to the Banks.

7.4. Bank Credit Decision. Each Bank acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Bank and based on the Financial Statements, the Interim Financial Statements, and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Bank also acknowledges that it shall, independently and without reliance upon the Administrative Agent or any other Bank and based on such documents and information as it shall

deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement.

64

7.5. Expenses. To the extent not paid by the Borrower, each Bank severally agrees to pay to the Administrative Agent and the Issuing Bank on demand such Bank's ratable share of the following: (a) all reasonable out-of-pocket costs and expenses of the Administrative Agent and the Issuing Bank in connection with the preparation, execution, delivery, administration, modification, and amendment of this Agreement and the other Credit Documents, including the reasonable fees and expenses of outside counsel for the Administrative Agent and the Issuing Bank with respect to advising the Administrative Agent and the Issuing Bank as to their respective rights and responsibilities under this Agreement and the other Credit Documents, and (b) all out-of-pocket costs and expenses of the Administrative Agent and the Issuing Bank, and the Banks under this Agreement and the other Credit Documents, whether through negotiations, legal proceedings, or otherwise, including fees and expenses of this paragraph shall survive the repayment and termination of the credit provided for under this Agreement and any purported termination of this Agreement which does not expressly refer to this paragraph.

7.6. Indemnification. To the extent not reimbursed by the Borrower, each Bank severally agrees to protect, defend, indemnify, and hold harmless the Administrative Agent, the Issuing Bank, and each of their respective Related Parties (for the purposes of this Section 7.6, collectively, the "Indemnified Parties"), from and against all demands, claims, actions, suits, damages, judgments, fines, penalties, liabilities, and out-of-pocket costs and expenses, including reasonable costs of attorneys and related costs of experts such as accountants (collectively, the "Indemnified Liabilities"), actually incurred by any Indemnified Party which are related to any litigation or proceeding relating to this Agreement, the other Credit Documents, or the transactions contemplated thereunder, INCLUDING ANY INDEMNIFIED LIABILITIES CAUSED BY ANY INDEMNIFIED PARTY'S OWN NEGLIGENCE, but not Indemnified Liabilities which are a result of any Indemnified Party's gross negligence or willful misconduct. The provisions of this paragraph shall survive the repayment and termination of the credit provided for under this Agreement and any purported termination of this Agreement which does not expressly refer to this paragraph.

7.7. Successor Administrative Agent and Issuing Bank. The Administrative Agent or the Issuing Bank may resign at any time by giving written notice thereof to the Banks and the Borrower and may be removed at any time with or without cause by the Majority Banks upon receipt of written notice from the Majority Banks to such effect. Upon receipt of notice of any such resignation or removal, the Majority Banks shall have the right to appoint a successor Administrative Agent or Issuing Bank with the consent of the Borrower, which consent shall not be unreasonably withheld. If no successor Administrative Agent or Issuing Bank shall have been so appointed by the Majority Banks with the consent of the Borrower, and shall have accepted such appointment, within 30 days after the retiring Administrative Agent's or Issuing Bank's giving of notice of resignation or the Majority Banks' removal of the retiring Administrative Agent or Issuing Bank, then the retiring Administrative Agent or Issuing Bank may, on behalf of the Banks and the Borrower, appoint a successor Administrative Agent or Issuing Bank, which shall be, in the case of a successor agent, a commercial bank organized under the laws of the

United States of America or of any State thereof and having a combined capital and surplus of at least \$500,000,000 and, in the case of the Issuing Bank, a Bank. Upon the acceptance of any appointment as Administrative Agent or Issuing Bank by a successor Administrative Agent or Issuing Bank, such successor Administrative Agent or Issuing Bank shall thereupon succeed to and become vested with all the rights, powers, privileges, and duties of the retiring Administrative Agent or Issuing Bank, and the retiring Administrative Agent or Issuing Bank shall be discharged from any duties and obligations under this Agreement and the other Credit Documents after such acceptance, except that the retiring Issuing Bank shall remain the Issuing Bank with respect to any Letters of Credit outstanding on the effective date of its resignation or removal and the provisions affecting the Issuing Bank with respect to such Letters of Credit shall inure to the benefit of the retiring Issuing Bank until the termination of all such Letters of Credit. After any Administrative Agent's or Issuing Bank's resignation or removal hereunder as Administrative Agent or Issuing Bank, the provisions of this Article 7 shall inure to such Person's benefit as to any actions taken or omitted to be taken by such Person while such Person was Administrative Agent or Issuing Bank under this Agreement and the other Credit Documents.

ARTICLE 8. MISCELLANEOUS

8.1. Expenses. The Borrower shall pay on demand of the applicable party specified herein (a) all reasonable out-of-pocket costs and expenses of the Administrative Agent and the Issuing Bank in connection with the preparation, execution, delivery, administration, modification, and amendment of this Agreement and the other Credit Documents, including the reasonable fees and expenses of outside counsel for the Administrative Agent and the Issuing Bank, and (b) all out-of-pocket costs and expenses of the Administrative Agent, the Issuing Bank, and each Bank in connection with the preservation or enforcement of their respective rights under this Agreement and the other Credit Documents, whether through negotiations, legal proceedings, or otherwise, including fees and expenses of counsel for the Administrative Agent, the Issuing Bank, and each Bank. The provisions of this paragraph shall survive the repayment and termination of the credit provided for under this Agreement and any purported termination of this Agreement which does not expressly refer to this paragraph.

8.2. Indemnification. The Borrower agrees to protect, defend, indemnify, and hold harmless the Administrative Agent, the Issuing Bank, each Bank, and each of their respective Related Parties (for the purposes of this Section 8.2, collectively, the "Indemnified Parties"), from and against all demands, claims, actions, suits, damages, judgments, fines, penalties, liabilities, and out-of-pocket costs and expenses, including reasonable costs of attorneys and related costs of experts such as accountants (collectively, the "Indemnified Liabilities"), actually incurred by any Indemnified Party which are related to any litigation or proceeding relating to this Agreement, the other Credit Documents, or the transactions contemplated thereunder, INCLUDING ANY INDEMNIFIED LIABILITIES CAUSED BY ANY INDEMNIFIED PARTY'S OWN NEGLIGENCE, but not Indemnified Liabilities which are a result of any Indemnified Party's gross negligence or willful misconduct. The provisions of this paragraph shall survive the repayment and termination of the credit provided for under this Agreement to this paragraph.

8.3. Modifications, Waivers, and Consents. No provision of this Agreement, the Term Loan Notes or the Revolving Loan Notes may be waived, amended, or modified, nor shall any consent required under this Agreement, the Term Loan Notes or the Revolving Loan Notes be effective, except pursuant to an agreement or agreements in writing entered into by the Borrower and the Majority Banks or by the Borrower and the Administrative Agent with the consent of the Majority Banks; provided that no such agreement shall (a) waive any of the conditions specified in Section 3.1 or 3.2 without the written consent of the Administrative Agent, all the Banks, and the Borrower, (b) increase the Commitment of any Bank without the written consent of such Bank, (c) forgive or reduce the amount or rate of any principal, interest, or fees payable under the Credit Documents, or postpone or extend the time for payment thereof without the written consent of the Administrative Agent, all the Banks, and the Borrower, (d) extend the scheduled maturities or times of payment without the written consent of the Administrative Agent, all the Banks, and the Borrower, (e) release any Guarantor or all or substantially all of the collateral securing the Credit Obligations (except as otherwise permitted or required herein) without the written consent of the Administrative Agent, all the Banks, and the Borrower, or (f) change the percentage of Banks required to take any action under this Agreement, the Term Loan Notes, the Revolving Loan Notes, or the Security Documents, including any amendment of the definition of "Majority Banks" or this Section 8.3 without the written consent of the Administrative Agent, all the Banks, and the Borrower. No modification, waiver, or consent shall, unless in writing and signed by the Administrative Agent, the Issuing Bank or the Swing Line Lender affect the rights or obligations of the Administrative Agent, the Issuing Bank or the Swing Line Lender, as the case may be, under the Credit Documents. The Administrative Agent shall not modify or waive or grant any consent under any other Credit Document if such action would be prohibited under this Section 8.3 with respect to the Credit Agreement, the Term Loan Notes or the Revolving Loan Notes.

8.4. Survival of Agreements. All representations, warranties, and covenants of the Borrower in this Agreement and the other Credit Documents shall survive the execution of this Agreement and the other Credit Documents and any other document or agreement.

8.5. Assignment and Participation. This Agreement and the other Credit Documents shall bind and inure to the benefit of the Borrower and their respective successors and assigns and the Administrative Agent and the Banks and their respective successors and assigns. The Borrower may not assign its rights or delegate its duties under this Agreement or any other Credit Document.

(a) Assignments. Any Bank may assign to one or more banks or other entities all or any portion of its rights and obligations under this Agreement (including, without limitation, all or a portion of its Commitments, the Loan Advances owing to it, the Term Loan Notes and the Revolving Loan Notes held by it, and the participation interest in the Letters of Credit owned by it); provided, however, that (i) each such assignment shall be of a constant, and not a varying, percentage of all of such Bank's rights and obligations under this Agreement, (ii) assignments of Commitments shall be made in minimum amounts of \$5,000,000 and be made in integral multiples of \$1,000,000 and the assigning Bank, if it retains any Commitments, shall maintain at least \$5,000,000 in Commitments, (iii) each such assignment shall be ratable as between the Facilities, (iv) each such assignment shall be to an Eligible Assignee, (v) the parties to each such assignment shall execute and deliver to the Administrative Agent, for its acceptance and

recording in the Register, an Assignment and Acceptance, together with the Term Loan Notes and the Revolving Loan Notes subject to such assignment, and (vi) each Eligible Assignee (other than the Eligible Assignee of the Administrative Agent) shall pay to the Administrative Agent a \$3,500 administrative fee. Upon such execution, delivery, acceptance and recording, from and after the effective date specified in each Assignment and Acceptance, which effective date shall be at least three Business Days after the execution thereof, (A) the assignee thereunder shall be a party hereto for all purposes and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, have the rights and obligations of a Bank hereunder and (B) such Bank thereunder shall, to the extent that rights and obligations hereunder have been assigned by it pursuant to such Assignment and Acceptance, relinquish its rights and be released from its obligations under this Agreement (and, in the case of an Assignment and Acceptance covering all or the remaining portion of such Bank's rights and obligations under this Agreement, such Bank shall cease to be a party hereto).

(b) Term of Assignments. By executing and delivering an Assignment and Acceptance, the Bank thereunder and the assignee thereunder confirm to and agree with each other and the other parties hereto as follows: (i) other than as provided in such Assignment and Acceptance, such Bank makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or the execution, legality, validity, enforceability, genuineness, sufficiency of value of this Agreement or any other instrument or document furnished pursuant hereto; (ii) such Bank makes no representation or warranty and assumes no responsibility with respect to the financial condition of any Credit Party or the performance or observance by any Credit Party of any of its obligations under this Agreement or any other instrument or document furnished pursuant hereto; (iii) such assignee confirms that it has received a copy of this Agreement, together with copies of the Financial Statements, the Interim Financial Statements, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance; (iv) such assignee shall, independently and without reliance upon the Administrative Agent, such Bank or any other Bank and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement; (v) such assignee appoints and authorizes the Administrative Agent to take such action as agent on its behalf and to exercise such powers under this Agreement as are delegated to the Administrative Agent by the terms hereof, together with such powers as are reasonably incidental thereto; and (vi) such assignee agrees that it shall perform in accordance with their terms all of the obligations which by the terms of this Agreement are required to be performed by it as a Bank.

(c) The Register. The Administrative Agent shall maintain at its address referred to in Section 8.6 a copy of each Assignment and Acceptance delivered to and accepted by it and a register for the recordation of the names and addresses of the Banks and the Commitments of each Bank from time to time (the "Register"). The entries in the Register shall be conclusive and binding for all purposes, absent manifest error, and the Borrower, the Administrative Agent, the Issuing Bank, and the Banks may treat each Person whose name is recorded in the Register as a Bank hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower or any Bank at any reasonable time and from time to time upon reasonable prior notice. Any assignment of any loan made hereunder, whether or not evidenced by a note, shall be effective only upon appropriate entries with respect thereto being made in the

Register (and each note shall expressly so provide). Any assignment or transfer of all or part of a loan made hereunder evidenced by a note shall be registered on the Register only upon surrender for registration of assignment or transfer of the note evidencing such loan, and thereupon one or more new notes shall be issued to the designated assignee.

(d) Procedures. Upon its receipt of an Assignment and Acceptance executed by a Bank and an Eligible Assignee, together with the Term Loan Notes and the Revolving Loan Notes subject to such assignment, the Administrative Agent shall, if such Assignment and Acceptance has been completed in the appropriate form, (i) accept such Assignment and Acceptance, (ii) record the information contained therein in the Register, and (iii) give prompt notice thereof to the Borrower. Within five Business Days after its receipt of such notice, the Borrower shall execute and deliver to the Administrative Agent in exchange for the surrendered Term Loan Notes and Revolving Loan Notes a new Term Loan Note and Revolving Loan Note to the order of such Eligible Assignee in an amount equal to the share of the Term Loan and the Revolving Loan Commitment assumed by it pursuant to such Assignment and Acceptance and, if such Bank has retained any share of the Term Loan Note to the order of such Bank in an amount equal to the share of the Term Loan Note to the order of such Bank in an amount equal to the share of the Term Loan Note to the order of such Bank in an amount equal to the share of the Term Loan Note to the order of such Bank in an amount equal to the share of the Term Loan Notes and Revolving Loan Notes shall be dated the effective date of such Assignment and Acceptance and shall be in the appropriate form.

(e) Participation. Each Bank may sell participation to one or more banks or other entities in or to all or a portion of its rights and obligations under this Agreement (including, without limitation, all or a portion of its Commitments, the Loan Advances owing to it, its participation interest in the Letters of Credit, and the Term Loan Notes and the Revolving Loan Notes held by it); provided, however, that (i) such Bank's obligations under this Agreement (including, without limitation, its Commitments to the Borrower hereunder) shall remain unchanged, (ii) such Bank shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) such Bank shall remain the holder of any such Term Loan Notes and Revolving Loan Notes for all purposes of this Agreement, (iv) the Borrower, the Administrative Agent, and the Issuing Bank and the other Banks shall continue to deal solely and directly with such Bank in connection with such Bank's rights and obligations under this Agreement, and (v) such Bank shall not require the participant's consent to any matter under this Agreement, except that upon 10 days' written notice of such participation to the Administrative Agent and the Borrower, such Bank may permit the participant to possess consent rights with respect to changes in the principal amount of the Term Loan Notes and the Revolving Loan Notes, reductions in fees or interest, extensions of the applicable maturity date, or releases of all or substantially all of the collateral or any guarantor (except to the extent otherwise permitted herein or in any of the other Credit Documents). The Borrower hereby agrees that participants shall have the same rights under Sections 2.7, 2.8, 2.9, 2.10, 2.11, and 8.2 as a Bank to the extent of their respective participation.

(f) Assignments or Pledges to Federal Reserve Banks. In addition to the foregoing rights of assignment and participation, any Bank may assign or pledge any portion of its rights under this Agreement (including the Term Loan Advances and the Revolving Loan Advances owed to such Bank) to any Federal Reserve Bank in accordance with applicable law without notice to or the consent of the Borrower or the Administrative Agent, provided that (i) such Bank shall not be relieved of its obligations under this Agreement as a result thereof and (ii) in no event shall the

Federal Reserve Bank be entitled to direct the actions of the pledging or assigning Bank under this Agreement.

8.6. Notice. All notices and other communications under this Agreement, the Term Loan Notes and the Revolving Loan Notes shall be in writing and mailed by certified mail (return receipt requested), telecopied, telexed, hand delivered, or delivered by a nationally recognized overnight courier, to the address for the appropriate party specified in Schedule I-B or at such other address as shall be designated by such party in a written notice to the other parties. Mailed notices shall be effective when received. Telecopied or telexed notices shall be effective when transmission is completed or confirmed by telex answerback. Delivered notices shall be effective when delivered by messenger or courier. Notwithstanding the foregoing, notices and communications to the Administrative Agent pursuant to Article 2 or 7 shall not be effective until received by the Administrative Agent.

8.7. CHOICE OF LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.

 ${\tt 8.8.}$ Submission To Jurisdiction; Waivers. The Borrower hereby irrevocably and unconditionally:

(a) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Credit Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of the courts of the State of New York, the courts of the United States for the Southern District of New York, and appellate courts from any thereof;

(b) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;

(c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Borrower, as the case may be at its address set forth in Schedule I-B or at such other address of which the Administrative Agent shall have been notified pursuant thereto;

(d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and

(e) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this Section any special, exemplary, punitive or consequential damages.

8.9. WAIVER OF JURY TRIAL. THE BORROWER IRREVOCABLY WAIVES ANY RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THE CREDIT DOCUMENTS OR ANY TRANSACTIONS RELATING THERETO.

8.10. Counterparts. This Agreement may be executed in multiple counterparts which together shall constitute one and the same instrument.

8.11. NO FURTHER AGREEMENTS. THIS WRITTEN AGREEMENT AND THE CREDIT DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

EXECUTED as of the date first above written.

BORROWER:

INTEGRATED ELECTRICAL SERVICES, INC.

By: /s/ NEIL J. DePASCAL Name: Neil J. DePascal Title: Chief Accounting Officer

ADMINISTRATIVE AGENT:

THE CHASE MANHATTAN BANK, as Administrative Agent

By: /s/ JAMES R. DOLPHIN Name: James R. Dolphin Title: Senior Vice President

BANKS:

THE CHASE MANHATTAN BANK

By: /s/ JAMES R. DOLPHIN Name: James R. Dolphin Title: Senior Vice President

CREDIT LYONNAIS, NEW YORK BRANCH, as Syndication Agent and as a Bank

By: /s/ ATTILA KOC

Name: Attila Koc Title: Senior Vice President

THE BANK OF NOVA SCOTIA, as Syndication Agent and as a Bank

By: /s/ M. D. SMITH Name: M. D. Smith Title: Agent Operations

TORONTO DOMINION (TEXAS), INC., as Documentation Agent and as a Bank

By: /s/ ANN S. SLANIS Name: Ann S. Slanis Title: Vice President

BANK OF SCOTLAND

By: /s/ ANNIE GLYNN

Name:	Annie Glynn
Title:	Senior Vice President

FIRST BANK & TRUST

By: /s/ ALAN J. COTT Name: Alan J. Cott Title: Senior Vice President

FIRSTAR BANK, N.A.

By: /s/ GREGORY L. DRYDEN Name: Gregory L. Dryden Title: Vice President MERRILL LYNCH CAPITAL CORPORATION

By: /s/ CAROL J.E. FEELEY Name: Carol J.E. Feeley Title: Vice President

SOUTHWEST BANK OF TEXAS, N.A.

By: /s/ CARMEN DUMNIRE Name: Carmen Dumnire Title: Vice President

INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES RATIO OF EARNINGS TO FIXED CHARGES (IN THOUSANDS OF DOLLARS)

	Veer Ended	Nine Months Foded	Year Ended September 30,			
	Year Ended December 31, 1996	Nine Months Ended September 30, 1997	1997	1998	1999	2000
CONSOLIDATED Earnings: Income before income taxes Fixed charges		\$6,143 251	\$7,298 283	\$12,638 2,486	\$83,455 14,772	\$42,799 25,925
Fixed charges	\$6,359 ======	\$6,394	\$7,581	\$15,124	\$98,227	\$68,724
Fixed Charges: Interest expense Portion of rental cost representing interest	\$ 171 52	\$ 164 87	\$ 214 69	\$ 1,161 1,325	\$13,145 1,627	\$23,230 2,695
	\$ 223 ======	\$ 251 ======	\$ 283 ======	\$ 2,486	\$14,772 ======	\$25,925 ======
Ratio of earnings to fixed charges	28.5x =====	25.5x =====	26.8x =====	6.1x ======	6.6x ======	2.7x ======

	SIX MONTHS ENDED MARCH 31			
	2000	2001		
CONSOLIDATED Earnings: Income before income				
taxes	\$ 5,316 12,675	\$28,294 14,026		
	\$17,991 =====	\$42,320 =====		
Fixed Charges: Interest expense	\$ 11,395	\$ 12,494		
Portion of rental cost representing interest	1,280	1,532		
Tepresenting interest		1, 332		
	\$ 12,675 =====	\$ 14,026 =====		
Ratio of earnings to fixed charges	1.4x ======	3.0x ======		

CONSENT OF INDEPENDENT PUBLIC ACCOUNTANTS

_ _ _ _

As independent public accountants, we hereby consent to the incorporation by reference in this registration statement of our report dated November 8, 2000 included in Integrated Electrical Services, Inc.'s Form 10-K for the year ended September 30, 2000, and to all references to our Firm included in this Form S-4 Registration Statement.

/s/ ARTHUR ANDERSEN LLP

_ _ _ _ _ _ _ _ .

Houston, Texas July 9, 2001 SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM T-1

STATEMENT OF ELIGIBILITY UNDER THE TRUST INDENTURE ACT OF 1939 OF A CORPORATION DESIGNATED TO ACT AS TRUSTEE

Check if an Application to Determine Eligibility of a Trustee Pursuant to Section 305(b)(2)

STATE STREET BANK AND TRUST COMPANY (Exact name of trustee as specified in its charter)

Massachusetts04-1867445(Jurisdiction of incorporation or
organization if not a U.S. national bank)(I.R.S. EmployerIdentification No.)Identification No.)

225 Franklin Street, Boston, Massachusetts 02110 (Address of principal executive offices) (Zip Code)

Maureen Scannell Bateman, Esq. Executive Vice President and General Counsel 225 Franklin Street, Boston, Massachusetts 02110 (617) 654-3253 (Name, address and telephone number of agent for service)

INTEGRATED ELECTRICAL SERVICES, INC. (Exact name of obligor as specified in its charter)

DELAWARE

(State or other jurisdiction of incorporation or organization)

76-0542208 (I.R.S. Employer Identification No.)

1800 WEST LOOP SOUTH, SUITE 500 HOUSTON, TX 77027 (Address of principal executive offices) (Zip Code)

9 3/8% SENIOR NOTES DUE 2009

(Title of indenture securities)

GENERAL

ITEM 1. GENERAL INFORMATION.

FURNISH THE FOLLOWING INFORMATION AS TO THE TRUSTEE:

(a) NAME AND ADDRESS OF EACH EXAMINING OR SUPERVISORY AUTHORITY TO WHICH IT IS SUBJECT.

Department of Banking and Insurance of The Commonwealth of Massachusetts, 100 Cambridge Street, Boston, Massachusetts.

Board of Governors of the Federal Reserve System, Washington, D.C., Federal Deposit Insurance Corporation, Washington, D.C.

(b) WHETHER IT IS AUTHORIZED TO EXERCISE CORPORATE TRUST POWERS. Trustee is authorized to exercise corporate trust powers.

ITEM 2. AFFILIATIONS WITH OBLIGOR.

IF THE OBLIGOR IS AN AFFILIATE OF THE TRUSTEE, DESCRIBE EACH SUCH AFFILIATION.

The obligor is not an affiliate of the trustee or of its parent, State Street Corporation.

(See note on page 2.)

ITEM 3. THROUGH ITEM 15. NOT APPLICABLE.

ITEM 16. LIST OF EXHIBITS.

LIST BELOW ALL EXHIBITS FILED AS PART OF THIS STATEMENT OF ELIGIBILITY.

1. A COPY OF THE ARTICLES OF ASSOCIATION OF THE TRUSTEE AS NOW IN EFFECT.

A copy of the Articles of Association of the trustee, as now in effect, is on file with the Securities and Exchange Commission as Exhibit 1 to Amendment No. 1 to the Statement of Eligibility and Qualification of Trustee (Form T-1) filed with the Registration Statement of Morse Shoe, Inc. (File No. 22-17940) and is incorporated herein by reference thereto.

2. A COPY OF THE CERTIFICATE OF AUTHORITY OF THE TRUSTEE TO COMMENCE BUSINESS, IF NOT CONTAINED IN THE ARTICLES OF ASSOCIATION.

A copy of a Statement from the Commissioner of Banks of Massachusetts that no certificate of authority for the trustee to commence business was necessary or issued is on file with the Securities and Exchange Commission as Exhibit 2 to Amendment No. 1 to the Statement of Eligibility and Qualification of Trustee (Form T-1) filed with the Registration Statement of Morse Shoe, Inc. (File No. 22-17940) and is incorporated herein by reference thereto.

3. A COPY OF THE AUTHORIZATION OF THE TRUSTEE TO EXERCISE CORPORATE TRUST POWERS, IF SUCH AUTHORIZATION IS NOT CONTAINED IN THE DOCUMENTS SPECIFIED IN PARAGRAPH (1) OR (2), ABOVE.

A copy of the authorization of the trustee to exercise corporate trust powers is on file with the Securities and Exchange Commission as Exhibit 3 to Amendment No. 1 to the Statement of Eligibility and Qualification of Trustee (Form T-1) filed with the Registration Statement of Morse Shoe, Inc. (File No. 22-17940) and is incorporated herein by reference thereto.

4. A COPY OF THE EXISTING BY-LAWS OF THE TRUSTEE, OR INSTRUMENTS CORRESPONDING THERETO.

A copy of the by-laws of the trustee, as now in effect, is on file with the Securities and Exchange Commission as Exhibit 4 to the Statement of Eligibility and Qualification of Trustee (Form T-1) filed with the Registration Statement of Eastern Edison Company (File No. 33-37823) and is incorporated herein by reference thereto.

Not applicable.

6. THE CONSENTS OF UNITED STATES INSTITUTIONAL TRUSTEES REQUIRED BY SECTION 321(b) OF THE ACT.

The consent of the trustee required by Section 321(b) of the Act is annexed hereto as Exhibit 6 and made a part hereof.

7. A COPY OF THE LATEST REPORT OF CONDITION OF THE TRUSTEE PUBLISHED PURSUANT TO LAW OR THE REQUIREMENTS OF ITS SUPERVISING OR EXAMINING AUTHORITY.

A copy of the latest report of condition of the trustee published pursuant to law or the requirements of its supervising or examining authority is annexed hereto as Exhibit 7 and made a part hereof.

NOTES

In answering any item of this Statement of Eligibility which relates to matters peculiarly within the knowledge of the obligor or any underwriter for the obligor, the trustee has relied upon information furnished to it by the obligor and the underwriters, and the trustee disclaims responsibility for the accuracy or completeness of such information.

The answer furnished to Item 2. of this statement will be amended, if necessary, to reflect any facts which differ from those stated and which would have been required to be stated if known at the date hereof.

SIGNATURE

Pursuant to the requirements of the Trust Indenture Act of 1939, as amended, the trustee, State Street Bank and Trust Company, a corporation organized and existing under the laws of The Commonwealth of Massachusetts, has duly caused this statement of eligibility to be signed on its behalf by the undersigned, thereunto duly authorized, all in the City of Hartford and the State of Connecticut, on the 3rd of July 2001.

STATE STREET BANK AND TRUST COMPANY

By:

NAME: MICHAEL M. HOPKINS TITLE: VICE PRESIDENT

EXHIBIT 6

CONSENT OF THE TRUSTEE

Pursuant to the requirements of Section 321(b) of the Trust Indenture Act of 1939, as amended, in connection with the proposed issuance by INTEGRATED ELECTRICAL SERVICES, INC. of its 9 3/8% SENIOR NOTES DUE 2009, we hereby consent that reports of examination by Federal, State, Territorial or District authorities may be furnished by such authorities to the Securities and Exchange Commission upon request therefor.

STATE STREET BANK AND TRUST COMPANY

By:

4

NAME: MICHAEL M. HOPKINS TITLE: VICE PRESIDENT

DATED: JULY 3, 2001

EXHIBIT 7

Consolidated Report of Condition of State Street Bank and Trust Company, Massachusetts and foreign and domestic subsidiaries, a state banking institution organized and operating under the banking laws of this commonwealth and a member of the Federal Reserve System, at the close of business March 31, 2001 published in accordance with a call made by the Federal Reserve Bank of this District pursuant to the provisions of the Federal Reserve Act and in accordance with a call made by the Commissioner of Banks under General Laws, Chapter 172, Section 22(a).

	Thousands Dollars
SSETS	
ash and balances due from depository institutions:	
Noninterest-bearing balances and currency and coin	897,105
Interest-bearing balances	
ecurities	16,720,900
ederal funds sold and securities purchased under agreements to resell in domestic offices	
of the bank and its Edge subsidiary	15 060 110
oans and lease financing receivables:	
Loans and leases, net of unearned income	
Allowance for loan and lease losses 57,674	
Allocated transfer risk reserve	0 004 70
Loans and leases, net of unearned income and allowances	
remises and fixed assets	
ther real estate owned	
nvestments in unconsolidated subsidiaries	22,733
stomers' liability to this bank on acceptances outstanding	
ntangible assets	
ther assets	1,512,53.
otal assets	
IABILITIES eposits:	
In domestic offices	12,418,125
Noninterest-bearing	
Interest-bearing 5,145,260 In foreign offices and Edge subsidiary	25 631 712
Noninterest-bearing	
Interest-bearing	
ederal funds purchased and securities sold under	
agreements to repurchase in domestic offices of	10 541 000
the bank and of its Edge subsidiary	
	0
ading liabilities	 2,336,011
ading liabilities her borrowed money bordinated notes and debentures	2,336,012 184,267
ading liabilities her borrowed money ubordinated notes and debentures unk's liability on acceptances executed and outstanding	2,336,012 2,336,012 184,267 00 167,024
ading liabilities her borrowed money ubordinated notes and debentures unk's liability on acceptances executed and outstanding	2,336,012 2,336,012 184,267 00 167,024
rading liabilities ther borrowed money ubordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities	2,336,012 184,265 167,024 167,024 1,566,844
rading liabilities ther borrowed money ubordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities	2,336,012 2,336,012 184,265 167,024 1,566,844 58,845,912
ading liabilities her borrowed money ubordinated notes and debentures ank's liability on acceptances executed and outstanding her liabilities otal liabilities inority interest in consolidated subsidiaries QUITY CAPITAL	2,336,011 184,265 167,024 1,566,844 58,845,912 49,273
ading liabilities her borrowed money ubordinated notes and debentures unk's liability on acceptances executed and outstanding her liabilities tal liabilities unority interest in consolidated subsidiaries UITY CAPITAL erpetual preferred stock and related surplus	2, 336, 011 184, 265 167, 024 1, 566, 844 58, 845, 912 49, 273
ading liabilities	2, 336, 011 184, 267 167, 024 1, 566, 844 58, 845, 911 49, 273
ading liabilities her borrowed money bordinated notes and debentures nk's liability on acceptances executed and outstanding her liabilities tal liabilities nority interest in consolidated subsidiaries UITY CAPITAL repetual preferred stock and related surplus mmon stock urplus	2, 336, 011 184, 267 167, 022 1, 566, 844 58, 845, 911 49, 273 49, 273 4
ading liabilities ther borrowed money bordinated notes and debentures unk's liability on acceptances executed and outstanding ther liabilities that liabilities otal liabilities unority interest in consolidated subsidiaries puTY CAPITAL erpetual preferred stock and related surplus mmon stock rrplus etained Earnings	2, 336, 011 184, 265 167, 024 1, 566, 844 58, 845, 911 49, 273 49, 273 49, 273 67, 085 3, 140, 648
rading liabilities ther borrowed money ubordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities ther liabilities ther liabilities ptal liabilities unority interest in consolidated subsidiaries QUITY CAPITAL erpetual preferred stock and related surplus mmon stock urplus etained Earnings Accumulated other comprehensive income	2, 336, 011 184, 265 167, 022 1, 566, 844 58, 845, 912 49, 273 29, 933 29, 933 3, 140, 648 29, 833
rading liabilities ther borrowed money bordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities ther liabilities ther liabilities potal components potal components potal components potal components potal liabilities potal liabilities potal liabilities potal components potal components potal components potal components potal liabilities potal components potal components potal liabilities potal components potal components potal components potal components potal potal components potal liabilities potal components potal components potal potal components potal potal components potal potal components potal potal components potal potal potal potal components potal potal pota	2, 336, 011 184, 265 167, 022 1, 566, 844 58, 845, 912 49, 273 29, 933 29, 933 3, 140, 648 29, 833 29, 833 3, 140, 648 29, 833 29, 833 3, 140, 648 29, 833 29, 833 3, 140, 648 29, 833 3, 140, 648 3, 140, 140 3, 140 40, 1
rading liabilities ther borrowed money ubordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities that liabilities total liabilities potal liabilities total liabilities QUITY CAPITAL erpetual preferred stock and related surplus promon stock riplus etained Earnings Accumulated other comprehensive income ther equity capital components divided profits and capital reserves/Net unrealized holding gains (losses) Net unrealized holding gains (losses) on available-for-sale	2, 336, 011 184, 265 167, 024 1, 566, 844 58, 845, 911 49, 273 49, 273 29, 933 567, 088 3, 140, 648 29, 833 (0)
rading liabilities ther borrowed money bordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities ther liabilities total liabilities potal liabil	2, 336, 017 184, 266 167, 024 1, 566, 844 58, 845, 917 49, 273 49, 274 49, 274 49, 274 49, 275 49, 275 4
rading liabilities ther borrowed money ubordinated notes and debentures ank's liability on acceptances executed and outstanding ther liabilities that liabilities total liabilities potal liabilities total liabilities QUITY CAPITAL erpetual preferred stock and related surplus promon stock riplus etained Earnings Accumulated other comprehensive income ther equity capital components divided profits and capital reserves/Net unrealized holding gains (losses) Net unrealized holding gains (losses) on available-for-sale	2, 336, 011 184, 265 167, 024 1, 566, 844 58, 845, 911 49, 273 29, 933 29, 933 3, 140, 648 29, 833 3, 140, 648 29, 833 (0)

I, Frederick P. Baughman, Senior Vice President and Comptroller of the above named bank do hereby declare that this Report of Condition has been prepared in conformance with the instructions issued by the Board of Governors of the Federal Reserve System and is true to the best of my knowledge and belief.

Frederick P. Baughman

We, the undersigned directors, attest to the correctness of this Report of Condition and declare that it has been examined by us and to the best of our knowledge and belief has been prepared in conformance with the instructions issued by the Board of Governors of the Federal Reserve System and is true and correct.

6

Ronald E. Logue David A. Spina Truman S. Casner LETTER OF TRANSMITTAL TO TENDER OUTSTANDING 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009 of INTEGRATED ELECTRICAL SERVICES, INC. PURSUANT TO THE EXCHANGE OFFER AND PROSPECTUS DATED _____, 2001

THE EXCHANGE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., NEW YORK CITY TIME, ON _____, 2001 (THE "EXPIRATION DATE"), UNLESS THE EXCHANGE OFFER IS EXTENDED BY THE COMPANY.

THE EXCHANGE AGENT FOR THE EXCHANGE OFFER IS:

State Street Bank and Trust Company

By Registered or Certified Mail, by Overnight Courier or by Hand:

State Street Bank and Trust Company Corporate Trust-Securities Window Goodwin Square 225 Asylum Street, 23rd Floor Hartford, Connecticut

By Facsimile:

Confirm by Telephone:

Originals of all documents sent by facsimile should be sent promptly by registered or certified mail, by hand, or by overnight delivery service.

DELIVERY OF THIS LETTER OF TRANSMITTAL TO AN ADDRESS OR TRANSMISSION OF INSTRUCTIONS VIA FACSIMILE OTHER THAN AS SET FORTH ABOVE WILL NOT CONSTITUTE A VALID DELIVERY.

IF YOU WISH TO EXCHANGE CURRENTLY OUTSTANDING 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009 (THE "OLD NOTES") FOR AN EQUAL AGGREGATE PRINCIPAL AMOUNT AT MATURITY OF NEW 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009 PURSUANT TO THE EXCHANGE OFFER, YOU MUST VALIDLY TENDER (AND NOT WITHDRAW) OLD NOTES TO THE EXCHANGE AGENT PRIOR TO THE EXPIRATION DATE.

THE UNDERSIGNED HEREBY ACKNOWLEDGES RECEIPT AND REVIEW OF THE PROSPECTUS, DATED _____, 2001 (THE "PROSPECTUS"), OF INTEGRATED ELECTRICAL SERVICES, INC., A DELAWARE CORPORATION (THE "COMPANY"), AND THIS LETTER OF TRANSMITTAL (THE "LETTER OF TRANSMITTAL"), WHICH TOGETHER DESCRIBE THE COMPANY'S OFFER (THE "EXCHANGE OFFER") TO EXCHANGE ITS 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009 (THE "NEW NOTES") THAT HAVE BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), FOR A LIKE PRINCIPAL AMOUNT OF ITS ISSUED AND OUTSTANDING 9 3/8% SENIOR SUBORDINATED NOTES DUE 2009 (THE "OLD NOTES"). CAPITALIZED TERMS USED BUT NOT DEFINED HEREIN HAVE THE RESPECTIVE MEANING GIVEN TO THEM IN THE PROSPECTUS.

THE COMPANY RESERVES THE RIGHT, AT ANY TIME OR FROM TIME TO TIME, TO EXTEND THE EXCHANGE OFFER AT ITS DISCRETION, IN WHICH EVENT THE TERM "EXPIRATION DATE" SHALL MEAN THE LATEST DATE TO WHICH THE EXCHANGE OFFER IS EXTENDED. THE COMPANY SHALL NOTIFY THE EXCHANGE AGENT AND EACH REGISTERED HOLDER OF THE OLD NOTES OF ANY EXTENSION BY ORAL OR WRITTEN NOTICE PRIOR TO 9:00 A.M., NEW YORK CITY TIME, ON THE NEXT BUSINESS DAY AFTER THE PREVIOUSLY SCHEDULED EXPIRATION DATE.

THIS LETTER OF TRANSMITTAL IS TO BE USED BY A HOLDER OF OLD NOTES IF OLD NOTES ARE TO BE FORWARDED HEREWITH. AN AGENT'S MESSAGE (AS DEFINED IN THE NEXT SENTENCE) IS TO BE USED IF DELIVERY OF OLD NOTES IS TO BE MADE BY BOOK-ENTRY TRANSFER TO THE ACCOUNT MAINTAINED BY THE EXCHANGE AGENT AT THE DEPOSITORY TRUST COMPANY (THE "BOOK-ENTRY TRANSFER FACILITY") PURSUANT TO THE PROCEDURES SET FORTH IN THE PROSPECTUS UNDER THE CAPTION "EXCHANGE OFFER --PROCEDURES FOR TENDERING." THE TERM "AGENT'S MESSAGE" MEANS A MESSAGE, TRANSMITTED BY THE BOOK-ENTRY TRANSFER FACILITY AND RECEIVED BY THE EXCHANGE AGENT AND FORMING A PART OF THE CONFIRMATION OF A BOOK-ENTRY TRANSFER ("BOOK-ENTRY CONFIRMATION"), WHICH STATES THAT THE BOOK-ENTRY TRANSFER FACILITY HAS RECEIVED AN EXPRESS ACKNOWLEDGMENT FROM A PARTICIPANT TENDERING OLD NOTES THAT ARE THE SUBJECT OF SUCH BOOK-ENTRY CONFIRMATION AND THAT SUCH PARTICIPANT HAS RECEIVED AND AGREES TO BE BOUND BY THE TERMS OF THE LETTER OF TRANSMITTAL AND THAT THE COMPANY MAY ENFORCE SUCH AGREEMENT AGAINST SUCH PARTICIPANT. HOLDERS OF OLD NOTES WHOSE OLD NOTES ARE NOT IMMEDIATELY AVAILABLE, OR WHO ARE UNABLE TO DELIVER THEIR OLD NOTES AND ALL OTHER DOCUMENTS REQUIRED BY THIS LETTER OF TRANSMITTAL TO THE EXCHANGE AGENT ON OR PRIOR TO THE EXPIRATION DATE, OR WHO ARE UNABLE TO COMPLETE THE PROCEDURE FOR BOOK-ENTRY TRANSFER ON A TIMELY BASIS, MUST TENDER THEIR OLD NOTES ACCORDING TO THE GUARANTEED DELIVERY PROCEDURES SET FORTH IN THE PROSPECTUS UNDER THE CAPTION "EXCHANGE OFFER GUARANTEED DELIVERY PROCEDURES." DELIVERY OF DOCUMENTS TO THE BOOK-ENTRY TRANSFER FACILITY DOES NOT CONSTITUTE DELIVERY TO THE EXCHANGE AGENT.

THE TERM "HOLDER" WITH RESPECT TO THE EXCHANGE OFFER MEANS ANY PERSON IN WHOSE NAME OLD NOTES ARE REGISTERED ON THE BOOKS OF THE COMPANY OR ANY OTHER PERSON WHO HAS OBTAINED A PROPERLY COMPLETED BOND POWER FROM SUCH REGISTERED HOLDER. THE UNDERSIGNED HAS COMPLETED, EXECUTED AND DELIVERED THIS LETTER OF TRANSMITTAL TO INDICATE THE ACTION THE UNDERSIGNED DESIRES TO TAKE WITH RESPECT TO THE EXCHANGE OFFER. HOLDERS WHO WISH TO TENDER THEIR OLD NOTES MUST COMPLETE THIS LETTER OF TRANSMITTAL IN ITS ENTIRETY.

SIGNATURES MUST BE PROVIDED PLEASE READ THE ACCOMPANYING INSTRUCTIONS CAREFULLY.

LADIES AND GENTLEMEN:

1. THE UNDERSIGNED HEREBY TENDERS TO THE COMPANY THE OLD NOTES DESCRIBED IN THE BOX ENTITLED "DESCRIPTION OF OLD NOTES TENDERED" PURSUANT TO THE COMPANY'S OFFER OF \$1,000 PRINCIPAL AMOUNT AT MATURITY OF NEW NOTES IN EXCHANGE FOR EACH \$1,000 PRINCIPAL AMOUNT AT MATURITY OF THE OLD NOTES, UPON THE TERMS AND SUBJECT TO THE CONDITIONS CONTAINED IN THE PROSPECTUS, RECEIPT OF WHICH IS HEREBY ACKNOWLEDGED, AND IN THIS LETTER OF TRANSMITTAL.

2. THE UNDERSIGNED HEREBY REPRESENTS AND WARRANTS THAT IT HAS FULL AUTHORITY TO TENDER THE OLD NOTES DESCRIBED ABOVE. THE UNDERSIGNED WILL, UPON REQUEST, EXECUTE AND DELIVER ANY ADDITIONAL DOCUMENTS DEEMED BY THE COMPANY TO BE NECESSARY OR DESIRABLE TO COMPLETE THE TENDER OF OLD NOTES.

3. THE UNDERSIGNED UNDERSTANDS THAT THE TENDER OF THE OLD NOTES PURSUANT TO ALL OF THE PROCEDURES SET FORTH IN THE PROSPECTUS WILL CONSTITUTE AN AGREEMENT BETWEEN THE UNDERSIGNED AND THE COMPANY AS TO THE TERMS AND CONDITIONS SET FORTH IN THE PROSPECTUS.

-2-

4. THE UNDERSIGNED ACKNOWLEDGE(S) THAT THE EXCHANGE OFFER IS BEING MADE IN RELIANCE UPON INTERPRETATIONS CONTAINED IN NO-ACTION LETTERS ISSUED TO THIRD PARTIES BY THE STAFF OF THE SECURITIES AND EXCHANGE COMMISSION (THE "SEC"), INCLUDING EXXON CAPITAL HOLDINGS CORP., SEC NO-ACTION LETTER (AVAILABLE APRIL 13, 1989), MORGAN STANLEY & CO., INC., SEC NO-ACTION LETTER (AVAILABLE JUNE 5, 1991) AND SHEARMAN & STERLING, SEC NO-ACTION LETTER (AVAILABLE JULY 2, 1993), THAT THE NEW NOTES ISSUED IN EXCHANGE FOR THE OLD NOTES PURSUANT TO THE EXCHANGE OFFER MAY BE OFFERED FOR RESALE, RESOLD AND OTHERWISE TRANSFERRED BY HOLDERS THEREOF (OTHER THAN A BROKER-DEALER WHO PURCHASED OLD NOTES EXCHANGED FOR SUCH NEW NOTES DIRECTLY FROM THE COMPANY TO RESELL PURSUANT TO RULE 144A OR ANY OTHER AVAILABLE EXEMPTION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT") AND ANY SUCH HOLDER THAT IS AN "AFFILIATE" OF THE COMPANY WITHIN THE MEANING OF RULE 405 UNDER THE SECURITIES ACT), WITHOUT COMPLIANCE WITH THE REGISTRATION AND PROSPECTUS DELIVERY PROVISIONS OF THE SECURITIES ACT, PROVIDED THAT SUCH NEW NOTES ARE ACQUIRED IN THE ORDINARY COURSE OF SUCH HOLDERS' BUSINESS AND SUCH HOLDERS ARE NOT PARTICIPATING IN, AND HAVE NO ARRANGEMENT WITH ANY PERSON TO PARTICIPATE IN, THE DISTRIBUTION OF SUCH NEW NOTES.

5. Unless the box under the heading "Special Registration Instructions" is checked, the undersigned hereby represents and warrants that:

 (a) the New Notes acquired pursuant to the Exchange Offer are being obtained in the ordinary course of business of the undersigned, whether or not the undersigned is the holder;

(b) neither the undersigned nor any such other person is engaging in or intends to engage in a distribution of such New Notes;

(c) neither the undersigned nor any such other person has an arrangement or understanding with any person to participate in the distribution of such New Notes; and

(d) neither the holder nor any such other person is an "affiliate," as such term is defined under Rule 405 promulgated under the Securities Act, of the Company.

6. The undersigned may, if unable to make all of the representations and warranties contained in Item 5 above and as otherwise permitted in the Registration Rights Agreement (as defined below), elect to have its Old Notes registered in the shelf registration statement described in the Exchange and Registration Rights Agreement, dated as of May 29, 2001 (the "Registration Rights Agreement"), by and among the Company and the Initial Purchasers (as defined therein). Such election may be made by checking the box below entitled "Special Registration Instructions." By making such election, the undersigned agrees, as a holder of Old Notes participating in a shelf registration, to indemnify and hold harmless the Company, each of the directors of the Company, each of the officers of the Company who signs such shelf registration statement, each person who controls the Company within the meaning of either the Securities Act or the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and each other holder of Old Notes, from and against any and all losses, claims, damages or liabilities caused by any untrue statement or alleged untrue statement of a material fact contained in any shelf registration statement or prospectus, or in any supplement thereto or amendment thereof, or caused by the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading; but only with respect to information relating to the undersigned furnished in writing by or on behalf of the undersigned expressly for use in a shelf registration statement, a prospectus or any amendments or supplements thereto. Any such indemnification shall be governed by the terms and subject to the conditions set forth in the Registration Rights Agreement, including, without limitation, the provisions regarding notice, retention of counsel, contribution and payment of expenses set forth therein. The above summary of the indemnification provision of the Registration Rights Agreement is not intended to be exhaustive and is qualified in its entirety by the Registration Rights Agreement.

7. If the undersigned is a broker-dealer that will receive New Notes for its own account in exchange for Old Notes that were acquired as a result of market-making activities or other trading activities, it acknowledges that it will deliver a prospectus in connection with any resale of such New Notes; however, by so acknowledging and by delivering a prospectus, the undersigned will not be deemed to admit that it is an "underwriter" within the meaning of the Securities Act. If the undersigned is a broker-dealer and Old Notes held for its own account were

3

8. Any obligation of the undersigned hereunder shall be binding upon the successors, assigns, executors, administrators, trustees in bankruptcy and legal and personal representatives of the undersigned.

9. Unless otherwise indicated herein under "Special Delivery Instructions," please issue the certificates for the New Notes in the name of the undersigned.

List below the Old Notes to which this Letter of Transmittal relates. If the space below is inadequate, list the registered numbers and principal amounts on a separate signed schedule and affix the list to this Letter of Transmittal. DESCRIPTION OF OLD NOTES TENDERED NAME(S) AND ADDRESS(ES) OF REGISTERED HOLDER(S) EXACTLY AS NAME(S) APPEAR(S) ON OLD NOTES (PLEASE FILL IN, IF BLANK) OLD NOTE(S) TENDERED REGISTERED AGGREGATE PRINCIPAL AMOUNT PRINCIPAL NUMBER(S)* REPRESENTED BY OLD NOTE(S) AMOUNT TENDERED** ----------_____ _____ _____ _____ NEED NOTE BE COMPLETED BY BOOK-ENTRY HOLDERS. UNLESS OTHERWISE INDICATED, ANY TENDERING HOLDER OF OLD NOTES WILL BE DEEMED TO HAVE TENDERED THE ENTIRE AGGREGATE PRINCIPAL AMOUNT REPRESENTED BY SUCH OLD NOTES. ALL TENDERS MUST BE IN INTEGRAL MULTIPLES OF \$1,000. _____ METHOD OF DELIVERY Check here if tendered Old Notes are enclosed herewith. 0 Check here if tendered Old Notes are being delivered by book-entry transfer made to an account maintained by the Exchange Agent with a 0 Book-Entry Transfer Facility and complete the following: Name of Tendering Institution: Account Number: -----Transaction Code Number: -----Check here if tendered Old Notes are being delivered pursuant to a 0 Notice of Guaranteed Delivery and complete the following: Name(s) of Registered Holder(s): _____ Date of Execution of Notice of Guaranteed Delivery: Window Ticket Number (if available): -----Name of Eligible Institution that guaranteed delivery: Account Number (If delivered by book-entry transfer):

-5-

SPECIAL REGISTRATION INSTRUCTIONS

TO BE COMPLETED ONLY IF (i) THE UNDERSIGNED SATISFIES THE CONDITIONS SET FORTH IN ITEM 6 ABOVE, (ii) THE UNDERSIGNED ELECTS TO REGISTER ITS OLD NOTES IN THE SHELF REGISTRATION STATEMENT DESCRIBED IN THE REGISTRATION RIGHTS AGREEMENT AND (iii) THE UNDERSIGNED AGREES TO INDEMNIFY CERTAIN ENTITIES AND INDIVIDUALS AS SET FORTH IN ITEM 6 ABOVE. (SEE ITEM 6.)

[] By checking this box, the undersigned hereby (i) represents that it is unable to make all of the representations and warranties set forth in Item 5 above, (ii) elects to have its Old Notes registered pursuant to the shelf registration statement described in the Registration Rights Agreement and (iii) agrees to indemnify certain entities and individuals identified in, and to the extent provided in, Item 6 above.

SPECIAL BROKER-DEALER INSTRUCTIONS

[] Check here if you are a broker-dealer and wish to receive 10 additional copies of the Prospectus and 10 copies of any amendments or supplements thereto.

Name:

(Please Print)

Address:

(Including Zip Code)

including zip code)

IMPORTANT PLEASE SIGN HERE WHETHER OR NOT OLD NOTES ARE BEING PHYSICALLY TENDERED HEREBY (COMPLETE ACCOMPANYING SUBSTITUTE FORM W-9)

(Signature(s) of Registered Holders of Old Notes):

Dated:

aleu:

(THE ABOVE LINES MUST BE SIGNED BY THE REGISTERED HOLDER(S) OF OLD NOTES AS NAME(S) APPEAR(S) ON THE OLD NOTES OR ON A SECURITY POSITION LISTING, OR BY PERSON(S) AUTHORIZED TO BECOME REGISTERED HOLDER(S) BY A PROPERLY COMPLETED BOND POWER FROM THE REGISTERED HOLDER(S), A COPY OF WHICH MUST BE TRANSMITTED WITH THIS LETTER OF TRANSMITTAL. IF OLD NOTES TO WHICH THIS LETTER OF TRANSMITTAL RELATE ARE HELD OF RECORD BY TWO OR MORE JOINT HOLDERS, THEN ALL SUCH HOLDERS MUST SIGN THIS LETTER OF TRANSMITTAL. IF SIGNATURE IS BY A TRUSTEE, EXECUTOR, ADMINISTRATOR, GUARDIAN, ATTORNEY-IN-FACT, OFFICER OF A CORPORATION OR OTHER PERSON ACTING IN A FIDUCIARY OR REPRESENTATIVE CAPACITY, THEN SUCH PERSON MUST (I) SET FORTH HIS OR HER FULL TITLE BELOW AND (II) UNLESS WAIVED BY THE COMPANY, SUBMIT EVIDENCE SATISFACTORY TO THE COMPANY OF SUCH PERSON'S AUTHORITY SO TO ACT. SEE INSTRUCTION 5 REGARDING COMPLETION OF THIS LETTER OF TRANSMITTAL, PRINTED BELOW.)

-6-

Name(s): (Please Type or Print) Capacity: Address: (Include Zip Code) Area Code and Telephone Number:

> SIGNATURE GUARANTEE (IF REQUIRED BY INSTRUCTION 5) CERTAIN SIGNATURES MUST BE GUARANTEED BY AN ELIGIBLE INSTITUTION

(Name of Eligible Institution Guaranteeing Signatures)

(Address (including zip code) and Telephone Number (including area code) of Firm)

(Authorized Signature)

(Printed Name)

(Title)

Dated:

-7-

INSTRUCTIONS

FORMING PART OF THE TERMS AND CONDITIONS OF THE EXCHANGE OFFER

1. DELIVERY OF THIS LETTER OF TRANSMITTAL AND OLD NOTES OR BOOK-ENTRY CONFIRMATIONS.

All physically delivered Old Notes or any confirmation of a book-entry transfer to the Exchange Agent's account at the Book-Entry Transfer Facility of Old Notes tendered by book-entry transfer (a "Book-Entry Confirmation"), as well as a properly completed and duly executed copy of this Letter of Transmittal or Agent's Message or facsimile hereof, and any other documents required by this Letter of Transmittal, must be received by the Exchange Agent at its address set forth herein prior to 5:00 P.M., New York City time, on the Expiration Date.

THE METHOD OF DELIVERY OF THE TENDERED OLD NOTES, THIS LETTER OF TRANSMITTAL AND ALL OTHER REQUIRED DOCUMENTS TO THE EXCHANGE AGENT IS AT THE ELECTION AND RISK OF THE HOLDER AND, EXCEPT AS OTHERWISE PROVIDED BELOW, THE DELIVERY WILL BE DEEMED MADE ONLY WHEN ACTUALLY RECEIVED OR CONFIRMED BY THE EXCHANGE AGENT. IF SUCH DELIVERY IS BY MAIL, IT IS RECOMMENDED THAT REGISTERED MAIL, PROPERLY INSURED, WITH RETURN RECEIPT REQUESTED, BE USED. INSTEAD OF DELIVERY BY MAIL, IT IS RECOMMENDED THAT THE HOLDER USE AN OVERNIGHT OR HAND DELIVERY SERVICE. IN ALL CASES, SUFFICIENT TIME SHOULD BE ALLOWED TO ASSURE DELIVERY TO THE EXCHANGE AGENT BEFORE THE EXPIRATION DATE. NO LETTER OF TRANSMITTAL OR OLD NOTES SHOULD BE SENT TO THE COMPANY.

2. GUARANTEED DELIVERY PROCEDURES.

Holders who wish to tender their Old Notes and whose Old Notes are not immediately available or who cannot deliver their Old Notes, this Letter of Transmittal or any other documents required hereby to the Exchange Agent prior to the Expiration Date, or who cannot complete the procedure for book-entry transfer on a timely basis and deliver an Agent's Message, must tender their Old Notes according to the guaranteed delivery procedures set forth in the Prospectus. Pursuant to such procedures, a tender may be effected if the Exchange Agent has received at its office, on or prior to the Expiration Date, a letter, telegram or facsimile transmission from an Eligible Institution (as defined in the Prospectus) setting forth the name and address of the tendering holder, the name(s), in which the Old Notes are registered and the certificate number(s) of the Old Notes to be tendered, and stating that the tender is being made thereby and guaranteeing that, within three New York Stock Exchange trading days after the date of execution of such letter, telegram or facsimile transmission by the Eligible Institution, such Old Notes, in proper form for transfer (or a confirmation of book-entry transfer of such Old Notes into the Exchange Agent's account at DTC), will be delivered by such Eligible Institution together with a properly completed and duly executed Letter of Transmittal (and any other required documents). Unless Old Notes being tendered by the above-described method are deposited with the Exchange Agent within the time period set forth above (accompanied or preceded by a properly completed Letter of Transmittal and any other required documents), the Company may, at its option, reject the tender.

Any holder of Old Notes who wishes to tender Old Notes pursuant to the guaranteed delivery procedures described above must ensure that the Exchange Agent receives the Notice of Guaranteed Delivery prior to 5:00 P.M., New York City time, on the Expiration Date. Upon request of the Exchange Agent, a Notice of Guaranteed Delivery will be sent to holders who wish to tender their Old Notes according to the guaranteed delivery procedures set forth above. See "Exchange Offer--Guaranteed Delivery Procedures" section of the Prospectus.

3. TENDER BY HOLDER.

Only a registered holder of Old Notes may tender such Old Notes in the Exchange Offer. Any beneficial holder of Old Notes who is not the registered holder and who wishes to tender should arrange with the registered holder to execute and deliver this Letter of Transmittal on his behalf or must, prior to completing and executing this Letter of Transmittal and delivering his Old Notes, either make appropriate arrangements to register ownership of the Old Notes in such holder's name or obtain a properly completed bond power from the registered holder.

4. PARTIAL TENDERS.

Tenders of Old Notes will be accepted only in integral multiples of \$1,000. If less than the entire principal amount of any Old Notes is tendered, the tendering holder should fill in the principal amount tendered in the third

8

column of the box entitled "Description of Old Notes Tendered" above. The entire principal amount of Old Notes delivered to the Exchange Agent will be deemed to shave been tendered unless otherwise indicated. If the entire principal amount of all Old Notes is not tendered, then Old Notes for the principal amount of Old Notes not tendered and New Notes issued in exchange for any Old Notes accepted will be sent to the holder at his or her registered address, unless a different address is provided in the appropriate box on this Letter of Transmittal, promptly after the Old Notes are accepted for exchange.

5. SIGNATURES ON THIS LETTER OF TRANSMITTAL; BOND POWERS AND ENDORSEMENTS; GUARANTEE OF SIGNATURES.

If this Letter of Transmittal (or facsimile hereof) is signed by the registered holder(s) of the Old Notes tendered hereby, the signature must correspond with the name(s) as written on the face of the Old Notes without alteration, enlargement or any change whatsoever. If this Letter of Transmittal (or facsimile hereof) is signed by a participant in the Book-Entry Transfer Facility, the signature must correspond with the name as it appears on the security position listing as the holder of the Old Notes.

If this Letter of Transmittal (or facsimile hereof) is signed by the registered holder or holders of Old Notes listed and tendered hereby and the New Notes issued in exchange therefor are to be issued (or any untendered principal amount of Old Notes is to be reissued) to the registered holder, the said holder need not and should not endorse any tendered Old Notes, nor provide a separate bond power. In any other case, such holder must either properly endorse the Old Notes tendered or transmit a properly completed separate bond power with this Letter of Transmittal, with the signatures on the endorsement or bond power guaranteed by an Eligible Institution.

If this Letter of Transmittal (or facsimile hereof) is signed by a person other than the registered holder or holders of any Old Notes listed, such Old Notes must be endorsed or accompanied by appropriate bond powers, in each case signed as the name of the registered holder or holders appears on the Old Notes.

If this Letter of Transmittal (or facsimile hereof) or any Old Notes or bond powers are signed by trustees, executors, administrators, guardians, attorneys-in-fact, officers of corporations or others acting in a fiduciary or representative capacity, such persons should so indicate when signing, and, unless waived by the Company, evidence satisfactory to the Company of their authority to act must be submitted with this Letter of Transmittal.

Endorsements on Old Notes or signatures on bond powers required by this Instruction 5 must be guaranteed by an Eligible Institution.

No signature guarantee is required if (i) this Letter of Transmittal (or facsimile hereof) is signed by the registered holder(s) of the Old Notes tendered herein (or by a participant in the Book-Entry Transfer Facility whose name appears on a security position listing as the owner of the tendered Old Notes) and the New Notes are to be issued directly to such registered holder(s) (or, if signed by a participant in the Book-Entry Transfer Facility, deposited to such participant's account at such Book-Entry Transfer Facility) and neither the box entitled "Special Delivery Instructions" nor the box entitled "Special Registration Instructions" has been completed, or (ii) such Old Notes are tendered for the account of an Eligible Institution. In all other cases, all signatures on this Letter of Transmittal (or facsimile hereof) must be guaranteed by an Eligible Institution.

6. SPECIAL REGISTRATION AND DELIVERY INSTRUCTIONS.

Tendering holders should indicate, in the applicable box or boxes, the name and address (or account at the Book-Entry Transfer Facility) to which New Notes or substitute Old Notes for principal amounts not tendered or not accepted for exchange are to be issued or sent, if different from the name and address of the person signing this Letter of Transmittal. In the case of issuance in a different name, the taxpayer identification or social security number of the person named must also be indicated.

Tax law requires that a holder of any Old Notes that are accepted for exchange must provide the Company (as payor) with its correct taxpayer identification number ("TIN"), which, in the case of a holder who is an individual is his or her social security number. If the Company is not provided with the correct TIN, the holder may be subject to a \$50 penalty imposed by Internal Revenue Service. (If withholding results in an overpayment of

9

taxes, a refund may be obtained). Certain holders (including, among others, all corporations and certain foreign individuals) are not subject to these backup withholding and reporting requirements. See the enclosed "Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9" for additional instructions.

To prevent backup withholding, each tendering holder must provide such holder's correct TIN by completing the Substitute Form W-9 set forth herein, certifying that the TIN provided is correct (or that such holder is awaiting a TIN), and that (i) the holder has not been notified by the Internal Revenue Service that such holder is subject to backup withholding as a result of failure to report all interest or dividends or (ii) the Internal Revenue Service has notified the holder that such holder is no longer subject to backup withholding. If the Old Notes are registered in more than one name or are not in the name of the actual owner, see the enclosed "Guidelines for Certification of Taxpayer Identification Number of Substitute Form W-9" for information on which TIN to report.

The Company reserves the right in its sole discretion to take whatever steps necessary to comply with the Company's obligations regarding backup withholding.

7. VALIDITY OF TENDERS.

All questions as to the validity, form, eligibility (including time of receipt), acceptance, and withdrawal of tendered Old Notes will be determined by the Company, in its sole discretion, which determination will be final and binding. The Company reserves the absolute right to reject any or all tenders not in proper form or the acceptance for exchange of which may, in the opinion of counsel for the Company, be unlawful. The Company also reserves the absolute right to waive any of the conditions of the Exchange Offer or any defect or irregularity in the tender of any Old Notes. The Company's interpretation of the terms and conditions of the Exchange Offer (including the instructions on the Letter of Transmittal) will be final and binding on all parties. Unless waived, any defects or irregularities in connection with tenders of Old Notes must be cured within such time as the Company shall determine. Although the Company intends to notify holders of defects or irregularities with respect to tenders of Old Notes, neither the Company, the Exchange Agent, nor any other person shall be under any duty to give notification of any defects or irregularities in connection with tenders or incregularities in connection and the defects or irregularities in tenders or incur any liability for failure to give such notification. Tenders of Old Notes will not be deemed to have been made until such defects or irregularities have been cured or waived. Any Old Notes received by the Exchange Agent that are not properly tendered and as to which the defects or irregularities have not been cured or waived will be returned by the Exchange Agent to the tendering holders, unless otherwise provided in the Letter of Transmittal, as soon as practicable following the Expiration Date.

8. WAIVER OF CONDITIONS.

The Company reserves the absolute right to waive, in whole or part, any of the conditions to the Exchange Offer set forth in the Prospectus or in this Letter of Transmittal.

9. NO CONDITIONAL TENDER.

No alternative, conditional, irregular or contingent tender of Old Notes on transmittal of this Letter of Transmittal will be accepted.

10. MUTILATED, LOST, STOLEN OR DESTROYED OLD NOTES.

Any holder whose Old Notes have been mutilated, lost, stolen or destroyed should contact the Exchange Agent at the address indicated above for further instructions.

11. REQUEST FOR ASSISTANCE OR ADDITIONAL COPIES.

Requests for assistance or for additional copies of the Prospectus or this Letter of Transmittal may be directed to the Exchange Agent at the address or telephone number set forth on the cover page of this Letter of Transmittal. Holders may also contact their broker, dealer, commercial bank, trust company or other nominee for assistance concerning the Exchange Offer.

12. WITHDRAWAL.

11

Tenders may be withdrawn only pursuant to the limited withdrawal rights set forth in the Prospectus under the caption "Exchange Offer--Withdrawal of Tenders."

IMPORTANT: THIS LETTER OF TRANSMITTAL OR A MANUALLY SIGNED FACSIMILE HEREOF (TOGETHER WITH THE OLD NOTES DELIVERED BY BOOK-ENTRY TRANSFER OR IN ORIGINAL HARD COPY FORM) MUST BE RECEIVED BY THE EXCHANGE AGENT, OR THE NOTICE OF GUARANTEED DELIVERY MUST BE RECEIVED BY THE EXCHANGE AGENT, PRIOR TO THE EXPIRATION DATE.

SUBSTITUTE FORM W-9	PART I - PLEASE PROVIDE YOUR TIN IN THE BOX AT RIGHT AND CERTIFY BY	Social Security Number:
	SIGNING AND DATING BELOW.	OR Employer Identification Number:
DEPARTMENT OF THE TREASURY, INTERNAL REVENUE SERVICE	PART II - FOR PAYEES EXEMPT FROM BACKUP WI GUIDELINES" FOR CERTIFICATION OF TAXPAYER FORM W-9 ENCLOSED HEREWITH AND COMPLETE AS	IDENTIFICATION NUMBER ON SUBSTITUTE

PAYER'S REQUEST FOR TAXPAYER IDENTIFICATION NUMBER (TIN)

CERTIFICATION - Under penalties of perjury, I certify that:

(1) The number shown on this form is my correct taxpayer identification number or a taxpayer identification number has not been issued to me and either: (a) I have mailed or delivered an application to receive a taxpayer identification number to the appropriate Internal Revenue Service Center or Social Security Administration office, or (b) I intend to mail or deliver an application in the near future. I understand that if I do not provide a taxpayer identification number within 60 days, 31% of all reportable payments made to me thereafter will be withheld until I provide a number.

(2) I am not subject to backup withholding because I am exempt from backup withholding, I have not been notified by the Service that I am subject to backup withholding as a result of a failure to report all interest or dividends or the Service has notified me that I am no longer subject to backup withholding.

CERTIFICATION INSTRUCTION - You must cross out item (2) above if you have been notified by the Service that you are subject to backup withholding because of under reporting interest or dividends on your tax return unless you received a subsequent notification from the Service stating that you are no longer subject to backup withholding.

SIGNATURE: PART III NAME: Awaiting TIN y
DATE: Please complete the Certificate of
Awaiting Taxpayer Identification Number
below.

-11-

I CERTIFY UNDER PENALTIES OF PERJURY THAT A TAXPAYER IDENTIFICATION NUMBER HAS NOT BEEN ISSUED TO ME, AND EITHER (a) I HAVE MAILED OR DELIVERED AN APPLICATION TO RECEIVE A TAXPAYER IDENTIFICATION NUMBER TO THE APPROPRIATE INTERNAL REVENUE SERVICE CENTER OR SOCIAL SECURITY ADMINISTRATION OFFICE OR (b) I INTEND TO MAIL OR DELIVER AN APPLICATION IN THE NEAR FUTURE. I UNDERSTAND THAT IF I DO NOT PROVIDE A TAXPAYER IDENTIFICATION NUMBER TO THE PAYOR WITHIN 60 DAYS, 31% OF ALL REPORTABLE PAYMENTS MADE TO ME THEREAFTER WILL BE WITHHELD UNTIL I PROVIDE A NUMBER.

Signature

Date

CERTIFICATE OF FOREIGN RECORD HOLDERS

Under penalties of perjury, I certify that I am not a United States citizen or resident (or I am signing for a foreign corporation, partnership, estate or trust).

Signature

to Minors Act)

Date

GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9

GUIDELINES FOR DETERMINING THE PROPER IDENTIFICATION NUMBER TO GIVE THE PAYER. Social Security numbers have nine digits separated by two hyphens: i.e., 000-00-0000. Employer identification numbers have nine digits separated by only hyphen: i.e., 00-0000000. The table below will help determine the number to give the payer.

For this type of account:	Give the SOCIAL SECURITY number of:	For this type of account:	Give the EMPLOYER IDENTIFICATION number of:
1. An individual's account	The individual	6. A valid trust, estate, or pension trust	The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in the account title) (4)
2. Two or more individuals (joint account)	The actual owner of the account or, if combined funds, any one of the individuals(1)	7. Corporate account	The corporation
3. Custodian account of a minor (Uniform Gift	The minor (2)	8. Partnership	The partnership

-12-

4. a. The usual revocable savings trust account (grantor is also trustee)	The grantor-trustee (1)	9. Association, club, religious, charitable or educational or other tax-exempt organization	The organization
b. So-called trust account that is not a legal or valid trust under State law	The actual owner (1)		
5. Sole proprietorship account	The owner (3)	10. A broker or registered nominee	The broker or nominee
		11. Account with the Department of Agriculture in the name of a public entity (such as a State or local government, school district, or prison) that receives agricultural program payments	The public entity

11. Account with the Department of Agriculture in the name of a public entity (such as a State or local government, school district, or prison) that receives agricultural program payments The public entity

- (1) List first and circle the name of the person whose number you furnish. If only one person on a joint account has a social security number, that person's number must be furnished.
- (2) Circle the minor's name and furnish the minor's social security number.
- (3) You must show your individual name. You may also enter your business or "doing business as" name. You may use either your social security number or, if you have one, your employer identification number.
- (4) List first and circle the name of the legal trust, estate or pension trust.
- NOTE: If no name is circled when there is more than one name listed, the number will be considered to be that of the first name listed.

GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9

OBTAINING A NUMBER

If you do not have a taxpayer identification number or you don't know your number, obtain Form SS-5, Application for a Social Security Number Card, or Form SS-4, Application for Employer Identification Number, at the local office of the Social Security Administration or the Internal Revenue Service (the "IRS") and apply for a number. You may also obtain Form SS-4 by calling the IRS at 1-800-TAX-FORM.

If you do not have a TIN, but have applied for one, write "Applied for" in the space for the TIN, sign and date the form and return it to the Exchange Agent.

PAYEES EXEMPT FROM BACKUP WITHHOLDING

Payees specifically exempted from backup withholding on ALL payments include the following:

- An organization exempt from tax under section 501(a), or an individual retirement account.
 - o The United States or any wholly-owned agency or instrumentality thereof.
 - A State, the District of Columbia, a possession of the United States or any political subdivision or wholly-owned agency or instrumentality thereof.
 - A foreign government, a political subdivision of a foreign government, or any wholly-owned agency or instrumentality thereof.
 - An international organization or any wholly-owned agency or instrumentality thereof.

Payee specifically exempted from backup withholding on interest and dividend payments include the following:

- o A corporation.
 - o A financial institution.
 - A registered dealer in securities or commodities registered in the U.S., the District of Columbia, or a possession of the U.S.
 - o A real estate investment trust.
 - o A common trust fund operated by a bank under section 584(a).
 - o An exempt charitable remainder trust, or a non-exempt trust described in section 4947.
 - An entity registered at all times during the tax year under the Investment Company Act of 1940.
 - o A foreign central bank issue.
 - o A middleman known in the investment community as a nominee or who is listed in the most recent publication of the American Society of Corporate Secretaries, Inc., Nominee List.

Payments of dividends and patronage dividends not generally subject to backup withholding include the following:

- Payments to nonresident aliens subject to withholding under section 1441.
- Payments to partnerships not engaged in a trade or business in the U.S. and which have at least one nonresident partner.
- o Payments of patronage dividends not paid in money.
- Payments made by certain foreign organizations. o Section 404(k) payments made by an ESOP.

Payments of interest not generally subject to backup withholding include the following:

- Payments of interest on obligations issued by individuals. Note: You
 may be subject to backup withholding if this interest is \$600 or more
 and is paid in the course of the payer's trade or business and you
 have not provided your correct taxpayer identification number to the
 payer.
- Payments of tax-exempt interest (including exempt-interest dividends under section 852).
- o Payments described in section 6049(b)(5) to non-resident aliens.
- o Payments on tax-free covenant bonds under section 1451.
- o Payments made by certain foreign organizations.

Exempt payees described above should file Form W-9 to avoid possible erroneous backup withholding. FILE THE FORM W-9 WITH THE PAYER. FURNISH YOUR TAXPAYER IDENTIFICATION NUMBER. WRITE "EXEMPT" ON THE FACE OF THE FORM AND RETURN IT TO THE PAYER.

Certain payments other than dividends that are not subject to information reporting are also not subject to backup withholding. For details, see the regulations under sections 6041, 6041A(a), 6045, 6050A, 6050N and their regulations.

PRIVACY ACT NOTICE.--Section 6109 requires most recipients of dividend, interest or other payments to give taxpayer identification numbers to payers who must report the payments to the IRS. The IRS uses the number for identification purposes and to help verify the accuracy of tax returns. The IRS also may provide this information to the Department of Justice for civil and criminal litigation and to cities, states, and the District of Columbia to carry out their tax laws. Payers must be given the

-14-

numbers whether or not recipients are required to file tax returns. Payers must generally withhold 31% of taxable interest, dividends and certain other payments to a payee who does not furnish a taxpayer identification number to a payer. Certain penalties may also apply.

PENALTIES

- (1) PENALTY FOR FAILURE TO FURNISH TAXPAYER IDENTIFICATION NUMBER.--If you fail to furnish your taxpayer identification number to a payer, you are subject to a penalty of \$50 for each such failure unless your failure is due to reasonable cause and not to willful neglect.
- (2) CIVIL PENALTY FOR FALSE INFORMATION WITH RESPECT TO WITHHOLDING.--If you make a false statement with no reasonable basis which results in no imposition of backup withholding, you are subject to a penalty of \$500.
- (3) CRIMINAL PENALTY FOR FALSIFYING INFORMATION.--Willfully falsifying certifications or affirmations may subject you to criminal penalties including fines and/or imprisonment.

FOR ADDITIONAL INFORMATION CONTACT YOUR TAX CONSULTANT OR THE INTERNAL REVENUE SERVICE.

[FORM OF LETTER TO CLIENTS]

INTEGRATED ELECTRICAL SERVICES, INC.

LETTER TO CLIENTS FOR TENDER OF ALL OUTSTANDING 9[]% SENIOR SUBORDINATED NOTES DUE 2009 IN EXCHANGE FOR 9[]% SENIOR SUBORDINATED NOTES DUE 2009 THAT HAVE BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933

THE EXCHANGE OFFER WILL EXPIRE AT 5:00 P.M., NEW YORK CITY TIME, ON __, 2001, UNLESS EXTENDED (THE "EXPIRATION DATE").

NOTES TENDERED IN THE EXCHANGE OFFER MAY BE WITHDRAWN AT ANY TIME PRIOR TO 5:00 P.M., NEW YORK CITY TIME, ON THE EXPIRATION DATE UNLESS PREVIOUSLY ACCEPTED FOR EXCHANGE.

To Our Clients:

We have enclosed herewith a Prospectus, dated _____, 2001, of Integrated Electrical Services, Inc., a Delaware company (the "Company"), and a related Letter of Transmittal, which together constitute the Company's offer (the "Exchange Offer") to exchange its 9[]% Senior Subordinated Notes due 2009 (the "New Notes"), which have been registered under the Securities Act of 1933, as amended (the "Securities Act"), for a like principal amount of its issued and outstanding 9[]% Senior Subordinated Notes due 2009 (the "Old Notes"), upon the terms and subject to the conditions set forth in the Exchange Offer.

The Exchange Offer is not conditioned upon any minimum number of Old Notes being tendered.

We are the holder of record of Old Notes held by us for your own account. A tender of such Old Notes can be made only by us as the record holder and pursuant to your instructions. The Letter of Transmittal is furnished to you for your information only and cannot be used by you to tender Old Notes held by us for your account.

We request instructions as to whether you wish to tender any or all of the Old Notes held by us for your account pursuant to the terms and conditions of the Exchange Offer. We also request that you confirm that we may, on your behalf, make the representations and warranties contained in the Letter of Transmittal.

Very truly yours,

PLEASE RETURN YOUR INSTRUCTIONS TO US IN THE ENCLOSED ENVELOPE WITHIN AMPLE TIME TO PERMIT US TO SUBMIT A TENDER ON YOUR BEHALF PRIOR TO THE EXPIRATION DATE.

INSTRUCTION TO REGISTERED HOLDER AND/OR BOOK-ENTRY TRANSFER PARTICIPANT

To Registered Holder and/or Participant in the Book-Entry Transfer Facility:

The undersigned hereby acknowledges receipt of the Prospectus, dated ______, 2001 (the "Prospectus"), of Integrated Electrical Services, Inc., a Delaware company (the "Company"), and the accompanying Letter of Transmittal (the "Letter of Transmittal"), that together constitute the Company's offer (the "Exchange Offer") to exchange its 9[]% Senior Subordinated Notes due 2009 (the "New Notes") for all of its outstanding 9[]% Senior Subordinated Notes due 2009 (the "Old Notes"). Capitalized terms used, but not defined herein have the meanings ascribed to them in the Prospectus.

This will instruct you, the registered holder and/or book-entry transfer facility participant, as to the action to be taken by you relating to the Exchange Offer with respect to the Old Notes held by you for the account of the undersigned.

The aggregate face amount of the Old Notes held by you for the account of the undersigned is (FILL IN AMOUNT).

\$_____ of the 9[]% Senior Subordinated Notes due 2009

With respect to the Exchange Offer, the undersigned hereby instructs you (CHECK APPROPRIATE BOX):

[] To TENDER the following Old Notes held by you for the account of the undersigned (INSERT PRINCIPAL AMOUNT OF OLD NOTES TO BE TENDERED) (IF ANY):

\$_____ of the 9[]% Senior Notes due 2009

[] NOTE to TENDER any Old Notes held by you for the account of the undersigned.

If the undersigned instructs you to tender the Old Notes held by you for the account of the undersigned, it is understood that you are authorized to make, on behalf of the undersigned (and the undersigned by its signature below, hereby makes to you), the representations and warranties contained in the Letter of Transmittal that are to be made with respect to the undersigned as a beneficial owner, including, but not limited to, the representations, that (i) the New Notes acquired in exchange for the Old Notes pursuant to the Exchange Offer are being acquired in the ordinary course of business of the person receiving such New Notes, (ii) the undersigned is not engaging in and does not intend to engage in a distribution of the New Notes, (iii) the undersigned does not have any arrangement or understanding with any person to participate in the distribution of New Notes, and (iv) neither the undersigned nor any such other person is an "affiliate" (within the meaning of Rule 405 under the Securities Act of 1933, as amended) of the Company. If the undersigned is a broker-dealer that will receive New Notes for its own account in exchange for Old Notes, it acknowledges that it will deliver a prospectus in connection with any resale of such New Notes.

SIGN HERE
Name of beneficial owner(s):
Signature(s)
Name(s):
(Please Print)
Address:
Telephone number:
Taxpayer Identification or Social Security Number:
Date:

-3-

INTEGRATED ELECTRICAL SERVICES, INC.

LETTER TO REGISTERED HOLDERS AND DEPOSITORY TRUST COMPANY PARTICIPANTS FOR TENDER OF ALL OUTSTANDING 9[]% SENIOR SUBORDINATED NOTES DUE 2009 IN EXCHANGE FOR 9[]% SENIOR SUBORDINATED NOTES DUE 2009 THAT HAVE BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933

THE EXCHANGE OFFER WILL EXPIRE AT 5:00 P.M., NEW YORK CITY TIME, ON ____, 2001, UNLESS EXTENDED (THE "EXPIRATION DATE").

NOTES TENDERED IN THE EXCHANGE OFFER MAY BE WITHDRAWN AT ANY TIME PRIOR TO 5:00 P.M., NEW YORK CITY TIME, ON THE EXPIRATION DATE UNLESS PREVIOUSLY ACCEPTED FOR EXCHANGE.

To Registered Holders and Depository Trust Company Participants:

We are enclosing herewith the material listed below relating to the offer by Integrated Electrical Services, Inc., a Delaware company (the "Company"), to exchange its 9[]% Senior Subordinated Notes due 2009 (the "New Notes"), which have been registered under the Securities Act of 1933, as amended (the "Securities Act"), for a like principal amount of its issued and outstanding 9[]% Senior Subordinated Notes due 2009 (the "Old Notes") upon the terms and subject to the conditions set forth in the Company's Prospectus, dated ______, 2001, and the related Letter of Transmittal (which together constitute the "Exchange Offer").

Enclosed herewith are copies of the following documents:

- 1. Prospectus, dated _____, 2001;
- Letter of Transmittal (together with accompanying Substitute Form W-9 Guidelines);
- Notice of Guaranteed Delivery;
- Letter that may be sent to your clients for whose accounts you hold Old Notes in your name or in the name of your nominee; and
- 5. Letter that may be sent from your clients to you with such client's instruction with regard to the Exchange Offer.

We urge you to contact your clients promptly. Please note that the Exchange Offer will expire on the Expiration Date unless extended.

The Exchange Offer is not conditioned upon any minimum number of Old Notes being tendered.

Pursuant to the Letter of Transmittal, each holder of Old Notes will represent to the Company that (i) the New Notes acquired in exchange for Old Notes pursuant to the Exchange Offer are being acquired in the ordinary course of business of the person receiving such New Notes, (ii) the holder is not engaging in and does not intend to engage in a distribution of the New Notes, (iii) the holder does not have any arrangement or understanding with any person to participate in the distribution of New Notes, and (iv) neither the holder nor any such other person is an "affiliate" (within the meaning of Rule 405 under the Securities Act) of the Company. If the holder is a broker-dealer that will receive New Notes for its own account in exchange for Old Notes that were acquired as a result of market-making activities or other trading activities, it must acknowledge that it will deliver a prospectus in connection with any resale of such New Notes.

The enclosed Letter to Clients contains an authorization by the beneficial owners of the Old Notes for you to make the foregoing representations.

The Company will not pay any fee or commission to any broker or dealer or to any other person (other than the Exchange Agent) in connection with the solicitation of tenders of Old Notes pursuant to the Exchange Offer.

Additional copies of the enclosed material may be obtained from the undersigned.

Very truly yours,

INTEGRATED ELECTRICAL SERVICES, INC.

-2-

[FORM OF NOTICE OF GUARANTEED DELIVERY]

NOTICE OF GUARANTEED DELIVERY TO TENDER OUTSTANDING 9[]% SENIOR SUBORDINATED NOTES DUE 2009 OF INTEGRATED ELECTRICAL SERVICES, INC. PURSUANT TO THE EXCHANGE OFFER AND PROSPECTUS DATED _____, 2001

As set forth in the Prospectus, dated _____, 2001 (as the same may be amended or supplemented from time to time, the "Prospectus") of Integrated Electrical Services, Inc. (the "Company") under the caption "Exchange Offer --Guaranteed Delivery Procedures" and in the Letter of Transmittal to Tender 9[]% Senior Subordinated Notes due 2009 of Integrated Electrical Services, Inc., this form or one substantially equivalent hereto must be used to accept the Exchange Offer (as defined below) if: (i) certificates for outstanding 9[]% Senior Subordinated Notes due 2009 (the "Old Notes") of the Company are not immediately available, (ii) time will not permit all required documents to reach the Exchange Agent on or prior to the Expiration Date (as defined below), or (iii) the procedures for book-entry transfer cannot be completed on or prior to the Expiration Date. This form may be delivered by facsimile transmission, by registered or certified mail, by hand, or by overnight delivery service to the Exchange Agent. See "Exchange Offer -- Procedures for Tendering" in the Prospectus.

THE EXCHANGE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., NEW YORK CITY TIME, ON _____, 2001 (THE "EXPIRATION DATE"), UNLESS THE EXCHANGE OFFER IS EXTENDED BY THE COMPANY.

The Exchange Agent for the Exchange Offer is:

State Street Bank and Trust Company

By Registered or Certified Mail, by Overnight Courier or by Hand:

State Street Bank and Trust Company Corporate Trust-Securities Window Goodwin Square 225 Asylum Street, 23rd Floor Hartford, Connecticut

By Facsimile:

Confirm by Telephone:

(Originals of all documents sent by facsimile should be sent promptly by registered or certified mail, by hand, or by overnight delivery service.)

DELIVERY OF THIS NOTICE OF GUARANTEED DELIVERY TO AN ADDRESS OR TRANSMISSION OF INSTRUCTIONS VIA FACSIMILE OTHER THAN AS SET FORTH ABOVE WILL NOT CONSTITUTE A VALID DELIVERY. Ladies and Gentlemen:

The undersigned hereby tenders to the Company, upon the terms and conditions set forth in the Prospectus and in the Letter of Transmittal (which together constitute the "Exchange Offer"), receipt of which is hereby acknowledged, the principal amount of Old Notes set forth below pursuant to the guaranteed delivery procedures described in the Prospectus and in the Letter of Transmittal.

The undersigned understands and acknowledges that the Exchange Offer will expire at 5:00 P.M., New York City time, on _____, 2001, unless extended by the Company.

All authority herein conferred or agreed to be conferred by this Notice of Guaranteed Delivery shall survive the death or incapacity of the undersigned and every obligation of the undersigned under this Notice of Guaranteed Delivery shall be binding upon the heirs, personal representatives, executors, administrators, successors, assigns, trustees in bankruptcy and other legal representatives of the undersigned.

DESCRIPTION OF OLD NOTES TENDERED

Certificate Number(s) (if known) of Old Notes or Account Number at the Book-Entry Facility	Aggregate Principal Amount Represented by Old Notes	Principal Amount Tendered
	Total:	Total:
PLE	EASE SIGN AND COMPLETE	
Signatures(s):	Name(s):	
Address: (Zip Code)	Capacity (full title) in a representative c	
Area Code and Telephone Number:		
Dated:	Taxpayer Identificati Security Number:	on or Social

THE GUARANTEE ON THE FOLLOWING PAGE MUST BE COMPLETED

GUARANTEE (Not to be used for signature guarantees)

The undersigned, being a member firm of a registered national securities exchange, a member of the National Association of Securities Dealers, Inc., or a commercial bank or trust company having an office in the United States, hereby guarantees (a)that the above named person(s) "own(s)" the Old Notes tendered hereby within the meaning of Rule 14e-4 ("Rule 14e-4") under the Securities Exchange Act of 1934, as amended, (b)that such tender of such Old Notes complies with Rule 14e-4, and (c)to deliver to the Exchange Agent the certificates representing the Old Notes into the Exchange Agent's account at The Depository Trust Company, in proper form for transfer, together with the Letter of Transmittal (or facsimile thereof), properly completed and duly executed, with any required signature guarantees and any other required documents, within three New York Stock Exchange trading days after the Expiration Date.

Name of Firm:
Address:
Area Code and Telephone No.:
Authorized Signature:
Name:
Title:
Dated:

NOTE: DO NOT SEND CERTIFICATES OF OLD NOTES WITH THIS FORM. CERTIFICATES OF OLD NOTES SHOULD BE SENT ONLY WITH A LETTER OF TRANSMITTAL.